UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

\mathbf{T}^{\prime}		N/I	O	T/
н (JK		Ж.	- N

Current Report
Pursuant to Section 13 or 15(d)
of the Securities Exchange Act of 1934
July 12, 2021 (July 7, 2021)
Date of Report (Date of earliest event reported)

FIRST INDUSTRIAL REALTY TRUST, INC. FIRST INDUSTRIAL, L.P.

(Exact name of registrant as specified in its charter)

First Industrial Realty Trust, Inc. First Industrial, L.P.

Maryland
Delaware
(State or other jurisdiction of incorporation or organization)

1-13102 333-21873 (Commission File Number) 36-3935116 36-3924586 (I.R.S. Employer Identification No.)

1 North Wacker Drive, Suite 4200 Chicago, Illinois 60606 (Address of principal executive offices, zip code)

(312) 344-4300 (Registrant's telephone number, including area code)

Not Applicable

(Former name or former address, if changed since last report)

following provisions (see General Instruction A.2. bel	-	filing obligation of the registrant under any of the
 □ Written communications pursuant to Rule 425 ur □ Soliciting material pursuant to Rule 14a-12 unde □ Pre-commencement communications pursuant to □ Pre-commencement communications pursuant to 	r the Exchange Act (17 CFR 240.14a-12) Rule 14d-2(b) under the Exchange Act (1	* **
Securities registered pursuant to Section 12(b) of the A		
Title of each class Common Stock, par value \$.01 per share	Trading Symbol(s) FR	Name of each exchange on which registered New York Stock Exchange
		· ·
Indicate by check mark whether the registrant is an enchapter) or Rule 12b-2 of the Securities Act of 1934 (see Emerging growth company		e 405 of the Securities Act of 1933 (§230.405 of this
chapter) or Rule 12b-2 of the Securities Act of 1934 (Emerging growth company □	3240.12b-2 of this chapter). if the registrant has elected not to use the ext	e 405 of the Securities Act of 1933 (§230.405 of this ended transition period for complying with any new or revised

Item 1.01: Entry into a Material Definitive Agreement.

On July 7, 2021, First Industrial Realty Trust, Inc. (the "Company") and First Industrial, L.P. (the "Operating Partnership") entered into an amendment and restatement (the "New Credit Facility") of that certain Third Amended and Restated Unsecured Revolving Credit agreement, dated as of October 31, 2017, by and among the Operating Partnership, as borrower, the Company, as guarantor, Wells Fargo Bank, National Association ("Wells Fargo"), individually and as administrative agent, and the other lenders thereunder (the "Prior Credit Facility").

The New Credit Facility provides for up to \$750.0 million of revolving borrowings by the Operating Partnership, which is an increase of \$25.0 million to the maximum amount of revolving borrowings under the Prior Credit Facility. The Operating Partnership may request the New Credit Facility be increased by up to \$250.0 million, subject to certain conditions, to an aggregate \$1 billion of revolving borrowings, which is equal to the maximum amount to which the Prior Credit Facility could be expanded. The New Credit Facility provides for interest-only payments initially at LIBOR plus 77.5 basis points and a facility fee of 15 basis points. The rate and facility fee is subject to adjustment based on the Company's leverage and investment grade rating. As of July 7, 2021, the Prior Credit Facility provided for interest-only payments at LIBOR plus 110 basis points and a facility fee of 15 basis points. The New Credit Facility matures on July 7, 2025, unless extended pursuant to two six-month extension options at the Operating Partnership's election, subject to certain conditions. The Company has fully and unconditionally guaranteed payment of borrowings under the New Credit Facility.

In connection with the foregoing, on July 7, 2021, the Company and the Operating Partnership also entered into an amendment and restatement (the "2021 Term Loan") of that certain Unsecured Term Loan agreement, dated as of July 15, 2020, by and among the Operating Partnership, as borrower, the Company, as general partner and guarantor, Wells Fargo, individually and as administrative agent and the other lenders thereunder (the "Prior Term Loan").

The 2021 Term Loan provides for a \$200.0 million unsecured term loan which is equal to the amount of the Prior Term Loan. The 2021 Term Loan allows the Operating Partnership to, subject to certain conditions, request incremental term loans in an aggregate amount up to \$260.0 million which would expand the aggregate borrowings under the 2021 Term Loan to an amount of up to \$460.0 million, which is \$160.0 million more than the maximum aggregate amount to which the Prior Term Loan could be expanded. The 2021 Term Loan provides for interest-only payments initially at LIBOR plus 85 basis points. The rate is subject to adjustment based on the Company's leverage and investment grade rating. As of July 7, 2021, the Prior Term Loan provided for interest-only payments at LIBOR plus 150 basis points. The 2021 Term Loan matures on July 7, 2026. The Company has fully and unconditionally guaranteed payment of borrowings under the 2021 Term Loan.

Each of the New Credit Facility and the 2021 Term Loan also (i) incorporates customary methodology for determining a replacement benchmark rate in the event that LIBOR becomes unavailable to use as a benchmark rate and (ii) includes a sustainability-linked pricing metric which permits an interest rate reduction by meeting a target related to environmental sustainability, specifically at least 25% of the Company's aggregate investment incurred with respect to completed development projects in a particular year being incurred with respect to completed development projects that achieve LEED certification.

The initial interest rate on both the New Credit Facility and the 2021 Term Loan is based on the BBB+/Baa1 credit ratings level, even though the Company's current ratings are BBB/Baa2. This favorable interest rate will be maintained provided that the Company's consolidated leverage ratio, as defined in the respective agreements, remains less than 32.5%.

The Operating Partnership intends to use the New Credit Facility for general business purposes, including, without limitation, working capital needs, repayment of indebtedness and the acquisition and development of property. The 2021 Term Loan has been used to repay the Prior Term Loan scheduled to mature on July 15, 2021.

Each of the Company and the Operating Partnership has had or may have with one or more of the lenders party to the New Credit Facility and 2021 Term Loan customary banking relationships through which a variety of financial services are, were or will be provided, including investment banking, underwriting, lending, commercial banking, treasury management, trustee and other advisory services, and for which such lenders will receive or have received customary fees and expenses.

In connection with the foregoing, on July 7, 2021, the Company and the Operating Partnership also entered into an amendment (such amendment, the "2015 Term Loan Amendment") to that certain Unsecured Term Loan agreement, dated as of September 11, 2015 by and among the Operating Partnership, as borrower, the Company, as guarantor, Wells Fargo, individually and as administrative agent, and the other lenders thereunder (as amended from time to time, the "2015 Term Loan Agreement"). The 2015 Term Loan Amendment made certain changes to the covenants and other provisions of the 2015 Term Loan Agreement to conform those covenants and other provisions to the terms of the New Credit Facility and the 2021 Term

Loan. The 2015 Term Loan Amendment, however, did not modify the maturity date or pricing terms of the 2015 Term Loan Agreement.

The foregoing descriptions of the New Credit Facility, the 2021 Term Loan and the 2015 Term Loan Amendment do not purport to be complete and are qualified in their entirety by reference to the New Credit Facility filed as Exhibit 10.1 hereto, the 2021 Term Loan filed as Exhibit 10.2 hereto and the 2015 Term Loan Amendment filed as Exhibit 10.3 hereto.

Item 2.03: Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant .

The information set forth in Item 1.01 is incorporated herein by reference.

Item 7.01: Regulation FD Disclosure.

On July 7, 2021, the Company issued a press release with respect to the execution of the New Credit Facility and the 2021 Term Loan. A copy of the press release is attached and incorporated by reference as Exhibit 99.1.

The information furnished in this report under this Item 7.01, including the Exhibit attached hereto, shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, nor shall it be deemed incorporated by reference in any filing under the Securities Act of 1933, except as shall be expressly set forth by specific reference to such filing

Item 9.01. Financial Statements and Exhibits.

(d) Exhibits. The following are filed herewith:

Exhibit No.	Description
<u>10.1</u>	Fourth Amended and Restated Unsecured Revolving Credit Facility Agreement, dated as of July 7, 2021, among First Industrial, L.P., First Industrial Realty Trust, Inc., Wells Fargo Bank, National Association and the other lenders thereunder
<u>10.2</u>	Amended and Restated Unsecured Term Loan Agreement, dated as of July 7, 2021, among First Industrial, L.P., First Industrial Realty Trust, Inc., Wells Fargo Bank, National Association, PNC Bank, National Association, and the other lenders thereunder
<u>10.3</u>	Third Amendment, dated as of July 7, 2021, to Unsecured Term Loan Agreement, dated as of September 11, 2015, among First Industrial, L.P., First Industrial Realty Trust, Inc., Wells Fargo Bank, National Association and the other lenders party thereto
<u>99.1</u>	First Industrial Realty Trust, Inc. Press Release dated July 7, 2021 (furnished pursuant to Item 7.01)
104	Cover Page Interactive Data File (embedded within the Inline XBRL document)

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

FIRST INDUSTRIAL REALTY TRUST, INC.

By:	/s/ JENNIFER MATTHEWS RICE
	Jennifer Matthews Rice General Counsel

FIRST INDUSTRIAL, L.P.

By: FIRST INDUSTRIAL REALTY TRUST, INC.

as general partner

/s/ JENNIFER MATTHEWS RICE

Jennifer Matthews Rice
General Counsel By:

Date: July 12, 2021

FOURTH AMENDED AND RESTATED UNSECURED REVOLVING CREDIT AGREEMENT DATED AS OF JULY 7, 2021

AMONG

FIRST INDUSTRIAL, L.P., AS BORROWER
FIRST INDUSTRIAL REALTY TRUST, INC.,
AS GENERAL PARTNER AND GUARANTOR
THE LENDERS

AND

WELLS FARGO BANK, NATIONAL ASSOCIATION
AS ADMINISTRATIVE AGENT

AND

BANK OF AMERICA, N.A. and PNC BANK, NATIONAL ASSOCIATION AS CO-SYNDICATION AGENTS

AND

U.S. BANK NATIONAL ASSOCIATION,
JPMORGAN CHASE BANK, N.A.,
CITIBANK, N.A.,

REGIONS BANK and ROYAL BANK OF CANADA
AS CO-DOCUMENTATION AGENTS

WELLS FARGO SECURITIES, LLC,

BOFA SECURITIES, INC., PNC CAPITAL MARKETS LLC, REGIONS CAPITAL MARKETS and
U.S. BANK NATIONAL ASSOCIATION
AS JOINT-LEAD ARRANGERS

AND

WELLS FARGO SECURITIES, LLC,
BOFA SECURITIES, INC. and PNC CAPITAL MARKETS LLC
AS JOINT BOOK RUNNERS

<u>Page</u>

	TICLE I. DEFINITIONS AND ACCOUNTING TERMS	2
	Definitions	2
1.2.		<u>32</u>
1.3.		<u>32</u>
1.4.		<u>32</u>
1.5.		<u>32</u>
1.6.		<u>33</u>
1.7.		<u>34</u>
1.8.	Divisions	<u>36</u>
ART	FICLE II. THE FACILITY	<u>36</u>
2.1.	The Facility	<u>36</u>
2.2.	Increase in Revolving Commitment	<u>36</u>
2.3.	Principal Payments	<u>37</u>
2.4.		<u>37</u>
2.5.	Evidence of Credit Extensions	<u>37</u>
2.6.	Loans and Borrowings	<u>38</u>
2.7.	Requests for Revolving Borrowings	<u>38</u>
2.8.	Interest Elections	<u>39</u>
2.9.	Applicable Margins/Facility Fee Rate	<u>41</u>
	O. Other Fees	<u>42</u>
2.11.	. Minimum Amount of Each Borrowing	<u>42</u>
2.12.	2. Interest	<u>42</u>
2.13.	3. Method of Payment	<u>43</u>
2.14.		<u>43</u>
2.15.	5. Lending Offices	<u>43</u>
2.16.		<u>44</u>
2.17.	7. Swingline Loans	<u>44</u>
2.18.	B. Bid Rate Loans	<u>45</u>
2.19.	9. Extension of Maturity Date	<u>48</u>
2.20.). Pro Rata Treatment/Sharing of Payments	<u>49</u>
2.21.		<u>50</u>
2.22.	2. Application of Moneys Received	<u>50</u>
2.23.		<u>51</u>
2.24.	4. Voluntary Prepayments of Loans	<u>51</u>
ART	TICLE III. THE LETTER OF CREDIT SUBFACILITY	<u>51</u>
3.1.	Obligation to Issue	<u></u>
3.2.		<u></u>
3.3.		<u> </u>
3.4.		<u>53</u>
3.5.		<u>==</u> <u>54</u>
3.6.		55

(Continued)

<u>Page</u>

<u>72</u>

3.7.	Payment of Reimbursement Obligations	<u>56</u>
	Compensation for Facility Letters of Credit	57
	Letter of Credit Collateral Account	<u>58</u>
3.10.	Issuing Bank Agreements	<u>58</u>
ART	ICLE IV. CHANGE IN CIRCUMSTANCES	<u>58</u>
4.1.	Yield Protection	<u>58</u>
4.2.	Changes in Capital Adequacy Regulations	<u>59</u>
4.3.	[Reserved]	<u>59</u>
4.4.	Funding Indemnification	<u>59</u>
4.5.	Taxes	<u>60</u>
4.6.	Lender Statements; Survival of Indemnity	62
4.7.	Replacement of Lenders under Certain Circumstances	<u>62</u>
4.8.	Changed Circumstances	<u>63</u>
ART	ICLE V. CONDITIONS PRECEDENT	<u>65</u>
5.1.	Conditions Precedent to Closing	65
5.2.	Conditions Precedent to Subsequent Borrowings	67
ART	ICLE VI. REPRESENTATIONS AND WARRANTIES	<u>68</u>
6.1.	Existence	<u>68</u>
6.2.	Corporate/Partnership Powers	<u>68</u>
6.3.	Power of Officers	<u>68</u>
6.4.	Government and Other Approvals	<u></u>
6.5.	Solvency	<u>68</u>
6.6.	Compliance With Laws	<u>69</u>
6.7.	Enforceability of Agreement	<u>69</u>
6.8.	Title to Property	69
6.9.	Litigation	<u>69</u>
	Events of Default	69
6.11.	Investment Company Act of 1940	<u>70</u>
6.12.		70
6.13.	Regulation U	<u>70</u>
	No Material Adverse Financial Change	<u>70</u>
6.15.		<u>70</u>
6.16.	Factual Information	<u>70</u>
6.17.	ERISA	<u></u> <u>70</u>
	Taxes	<u>-</u> 70
6.19.	Environmental Matters	<u>70</u>
6.20.	Insurance	<u>7</u> 1
6.21.	No Brokers	<u>7</u> 1

6.22. No Violation of Usury Laws

(Continued)

	<u>Page</u>
5.23. Not a Foreign Person	<u>72</u>
5.24. No Trade Name	— <u>72</u>
5.25. Subsidiaries	
5.26. Unencumbered Assets	<u>72</u>
5.27. Anti-Corruption Laws and Sanctions	<u>73</u>
5.28. Affected Financial Institutions	<u>74</u>
ARTICLE VII. ADDITIONAL REPRESENTATIONS AND WARRANTIES	<u>74</u>
7.1. Existence	<u>74</u>
7.2. Corporate Powers	<u>74</u>
7.3. Power of Officers	<u>74</u>
7.4. Government and Other Approvals	<u>75</u>
7.5. Compliance With Laws	<u>75</u>
7.6. Enforceability of Agreement	<u>75</u>
7.7. Liens; Consents	<u>75</u>
7.8. Litigation	<u>75</u>
7.9. Events of Default	<u>75</u>
7.10. Investment Company Act of 1940	<u>75</u>
7.11. Public Utility Holding Company Act	<u>75</u>
7.12. No Material Adverse Financial Change	<u>75</u>
7.13. Financial Information	<u>75</u>
7.14. Factual Information	<u>76</u>
7.15. ERISA	<u>76</u>
7.16. Taxes	<u>76</u>
7.17. No Brokers	<u>76</u>
7.18. Subsidiaries	<u>76</u>
7.19. Status	<u>76</u>
7.20. Anti-Corruption Laws and Sanctions	<u>76</u>
ARTICLE VIII. AFFIRMATIVE COVENANTS	<u>77</u>
3.1. Notices	<u>77</u>
3.2. Financial Statements, Reports, Etc	<u>78</u>
3.3. Existence and Conduct of Operations	<u>80</u>
3.4. Maintenance of Properties	<u>80</u>
3.5. Insurance	<u>80</u>
3.6. Payment of Obligations	<u>81</u>
3.7. Compliance with Laws	<u>81</u>
8.8. Adequate Books	<u>81</u>
3.9. ERISA	<u>81</u>
3.10. Maintenance of Status	<u>81</u>
3.11. Use of Proceeds	<u>81</u>
3.12. Distributions	<u>81</u>

(Continued)

<u>Page</u>

90

ARTIO	ICLE IX. NEGATIVE COVENANTS	<u>82</u>
	Change in Business	<u></u>
	Change of Management of Properties	<u>82</u>
	Change of Borrower Ownership	<u>82</u>
	Use of Proceeds	<u>82</u>
9.5.	Liens	<u>83</u>
9.6.	Regulation U	<u>83</u>
9.7.	Indebtedness and Cash Flow Covenants	<u>83</u>
9.8.	Change of Control; Mergers and Dispositions	<u>85</u>
9.9.	Negative Pledge	<u>85</u>
ARTIO	ICLE X. DEFAULTS	<u>85</u>
10.1.	Nonpayment of Principal	<u>85</u>
10.2.	Certain Covenants	<u>85</u>
10.3.	Nonpayment of Interest and Other Obligations	<u>85</u>
10.4.	Cross Default	<u>85</u>
10.5.	Loan Documents	<u>85</u>
10.6.	Representation or Warranty	<u>86</u>
10.7.	Covenants, Agreements and Other Conditions	<u>86</u>
10.8.		<u>86</u>
10.9.	Material Adverse Financial Change	<u>86</u>
10.10.	D. Bankruptcy	<u>86</u>
10.11.	. Legal Proceedings	<u>87</u>
10.12.	2. ERISA	<u>87</u>
10.13.	3. Change in Control	<u>87</u>
10.14.	4. Failure to Satisfy Judgments	<u>87</u>
10.15.	5. Environmental Remediation	<u>87</u>
ARTIO	ICLE XI. ACCELERATION, WAIVERS, AMENDMENTS AND REMEDIES	<u>87</u>
11.1.	Acceleration	<u>87</u>
11.2.	Preservation of Rights; Amendments	88
ARTIO	ICLE XII. THE ADMINISTRATIVE AGENT	89
12.1.	Appointment	89
12.2.	Powers	<u>89</u>
12.3.	General Immunity	89
	No Responsibility for Loans, Recitals, etc	<u>89</u>
12.5.	Action on Instructions of Lenders	<u>89</u>
12.6.	Employment of Administrative Agents and Counsel	90
	Reliance on Documents; Counsel	90
12.8.	Administrative Agent's Reimbursement and Indemnification	90

12.9. Rights as a Lender

(Continued)

<u>Page</u>

12.10.	. Funds Transfer Disbursements	<u>91</u>
12.11.		91
12.12.		91
12.13.		91
12.14.		92
12.15.		92
12.16.		92
12.17.		95
12.18.		95
12.19.	. Erroneous Payments	95
ARTIO	CLE XIII. BENEFIT OF AGREEMENT; ASSIGNMENTS; PARTICIPATIONS	<u>98</u>
13.1.		98
13.2.	•	98
13.3.		100
	Register	101
13.5.		<u>101</u>
13.6	Confidentiality	<u>102</u>
13.7	Tax Treatment	<u>103</u>
ARTIO	ICLE XIV. GENERAL PROVISIONS	103
14.1.	Survival of Representations	103
14.2.		103
14.3.		<u>103</u>
14.4.	Headings	104
14.5.	No Third Party Beneficiaries	104
	Expenses; Indemnification	<u>104</u>
14.7.	Severability of Provisions	<u>104</u>
14.8.	Nonliability of the Lenders	<u>104</u>
14.9.	Choice of Law	<u>105</u>
14.10.	. Consent to Jurisdiction	<u>105</u>
14.11.	. Waiver of Jury Trial	<u>105</u>
14.12.	. Successors and Assigns	<u>105</u>
14.13.	Entire Agreement; Modification of Agreement	<u>105</u>
14.14.	. Dealings with the Borrower	<u>106</u>
14.15.	. Set-Off	<u>106</u>
14.16.	Counterparts; Integration, Effectiveness, Electronic Execution	<u>107</u>
14.17.	. Patriot Act CIP Notice	<u>108</u>
14.18.	Acknowledgement and Consent to Bail-In of Affected Financial Institutions	<u>108</u>
14.19.	. Acknowledgement Regarding Any Supported QFCs	<u>109</u>
14.20.	. Certain ERISA Matters	<u>110</u>

(Continued)

<u>Page</u>

ARTIO	CLE XV. NOTICES	<u>111</u>
15.1.	Giving Notice; Electronic Delivery	<u>111</u>
15.2.	Change of Address	<u>112</u>

(Continued)

EXHIBITS

- A Revolving Commitment Amounts
- B-1 Form of Note
- B-2 Form of Bid Rate Note
- C Disbursement Instruction Agreement
- D Form of Guaranty
- E Opinion of Borrower's Counsel
- F Opinion of General Partner's Counsel
- G [Intentionally Omitted]
- H Form of Compliance Certificate
- I Form of Assignment Agreement
- J Form of Notice of Borrowing
- K Form of Notice of Swingline Borrowing
- L Form of Designation Agreement
- M Form of Bid Rate Quote Request
- N Form of Bid Rate Quote
- O Form of Bid Rate Quote Acceptance
- P Form of Pricing Certificate

SCHEDULES

- 3.1 Outstanding Facility Letters of Credit
- 6.9 Litigation (Borrower)
- 6.19 Environmental Compliance
- 6.24 Trade Names
- 6.25 Subsidiaries (Borrower)
- 6.26 Unencumbered Assets
- 7.8 Litigation (General Partner)
- 7.18 Subsidiaries (General Partner)

FOURTH AMENDED AND RESTATED UNSECURED REVOLVING CREDIT AGREEMENT

THIS FOURTH AMENDED AND RESTATED UNSECURED REVOLVING CREDIT AGREEMENT is entered into as of July 7, 2021 by and among the following:

FIRST INDUSTRIAL, L.P., a Delaware limited partnership having its principal place of business at 1 N. Wacker Drive, Suite 4200, Chicago, Illinois 60606, the sole general partner of which is First Industrial Realty Trust, Inc., a Maryland corporation;

FIRST INDUSTRIAL REALTY TRUST, INC., a Maryland corporation that is qualified as a real estate investment trust whose principal place of business is 1 N. Wacker Drive, Suite 4200, Chicago, Illinois 60606 ("General Partner");

WELLS FARGO BANK, NATIONAL ASSOCIATION ("<u>Wells</u>"), a national bank organized under the laws of the United States of America having an office at 10 South Wacker Drive, 32nd Floor, Chicago, IL 60606, as Administrative Agent ("<u>Administrative Agent</u>") for the Lenders (as defined below); and

Those Lenders identified on the signature pages hereto.

RECITALS

- A. The Borrower is primarily engaged in the business of acquiring, developing, owning and operating bulk warehouse and light industrial properties.
- B. The Borrower, the General Partner, the lenders party thereto, the Departing Lenders (as defined below) and Wells, as administrative agent are currently party to the Third Amended and Restated Unsecured Revolving Credit Agreement, dated as of October 31, 2017 (as amended, modified or otherwise supplemented prior to the date hereof, the "Existing Credit Agreement").
- C. The Borrower, the Lenders and the Administrative Agent have agreed to enter into this Agreement in order to (i) amend and restate the Existing Credit Agreement in its entirety, (ii) re-evidence the Obligations under, and as defined in, the Existing Credit Agreement, which shall be repayable in accordance with the terms of this Agreement, and (iii) set forth the terms and conditions under which the Lenders will, from time to time, make loans to or for the benefit of the Borrower.
- D. The parties hereto intend that this Agreement not constitute a novation of the obligations and liabilities of the parties under the Existing Credit Agreement or be deemed to evidence or constitute full repayment of such obligations and liabilities, but that this Agreement amend and restate in its entirety the Existing Credit Agreement and re-evidence the obligations and liabilities of the Borrower outstanding thereunder, which shall be payable in accordance with the terms hereof.
- E. Each of the Borrower and the General Partner confirms that all obligations under the applicable "Loan Documents" (as referred to and defined in the Existing Credit Agreement) shall continue in full force and effect as modified or restated by the Loan Documents (as referred to and defined herein) and that, from and after the Agreement Execution Date, all references to the "Credit Agreement" contained in any such existing "Loan Documents" shall be deemed to refer to this Agreement.
- F. The General Partner is fully liable for the obligations of the Borrower hereunder by virtue of its status as the sole general partner of the Borrower and as guarantor under the Guaranty.

NOW, THEREFORE, in consideration of the mutual covenants and agreements herein contained, the parties hereto agree as follows:

ARTICLE I.

DEFINITIONS AND ACCOUNTING TERMS

- 1.1. <u>Definitions</u>. As used in this Agreement, the following terms have the meanings set forth below:
- "Absolute Rate" has the meaning given that term in Section 2.18(c)(ii)(C).
- "Absolute Rate Auction" means a solicitation of Bid Rate Quotes setting forth Absolute Rates pursuant to Section 2.18.
- "Absolute Rate Loan" means a Bid Rate Loan, the interest rate on which is determined on the basis of an Absolute Rate pursuant to an Absolute Rate Auction.
- "Adjusted Base Rate" means a floating interest rate equal to the sum of (i) the LIBOR Market Index Rate, plus (ii) the Base Rate Applicable Margin in effect from time to time.
 - "Adjusted Base Rate Borrowing" means a Borrowing that bears interest at the Adjusted Base Rate.
- "Adjusted EBITDA" means for any Person the sum of EBITDA for such Person and such Person's reported corporate overhead for itself and its Subsidiaries; provided that "Adjusted EBITDA" shall have deducted overhead related to specific properties.
- "Adjusted LIBOR Rate" means, with respect to a Eurocurrency Borrowing for the relevant Interest Period, the sum of (i) the Base LIBOR Rate applicable to such Interest Period, plus, (ii) in the case of ratable Eurocurrency Borrowings, the LIBOR Applicable Margin in effect from time to time during such Interest Period, or in the case of Eurocurrency Borrowings made as Bid Rate Loans, the LIBOR Margin established in the Bid Rate Quote applicable to such Bid Rate Loan.
- "<u>Administrative Agent</u>" means Wells, in its capacity as contractual representative of the Lenders pursuant to <u>Article XII</u>, and not in its individual capacity as a Lender, and any successor Administrative Agent appointed pursuant to <u>Article XII</u>.
- "<u>Administrative Office</u>" means the Administrative Agent's office designated on its signature page to this Agreement or such other office as may be designated by the Administrative Agent by written notice to the Borrower and the Lenders.
- "<u>Administrative Questionnaire</u>" means the Administrative Questionnaire completed by each Lender and delivered to the Administrative Agent in a form supplied by the Administrative Agent to the Lenders from time to time.
 - "Affected Financial Institution" means (a) any EEA Financial Institution or (b) any UK Financial Institution.
- "Affiliate" means any Person directly or indirectly controlling, controlled by or under direct or indirect common control with any other Person. A Person shall be deemed to control another Person if

the controlling Person owns ten percent (10%) or more of any class of voting securities of the controlled Person or possesses, directly or indirectly, the power to direct or cause the direction of the management or policies of the controlled Person, whether through ownership of stock, by contract or otherwise; <u>provided</u> that in no event shall the Administrative Agent or any Lender be deemed to be an Affiliate of the Borrower.

- "Aggregate Revolving Commitment" means, as of any date, the total of all Revolving Commitments, which as of the Agreement Execution Date is \$750,000,000, subject to the Borrower's right to increase the Aggregate Revolving Commitment pursuant to Section 2.2.
- "Agreement" means this Fourth Amended and Restated Unsecured Revolving Credit Agreement and all amendments, modifications and supplements hereto.
 - "Agreement Execution Date" shall mean July 7, 2021, the date on which all of the parties hereto have executed this Agreement.
- "<u>Anti-Corruption Laws</u>" means all Applicable Laws of any jurisdiction applicable to the Borrower or its Subsidiaries from time to time concerning or relating to bribery, corruption or money-laundering, including without limitation, the Foreign Corrupt Practices Act of 1977.
 - "Applicable Cap Rate" means 5.75%.
- "Applicable Law" means all international, foreign, federal, state and local statutes, treaties, rules, guidelines, regulations, ordinances, codes, directions, executive orders, and administrative or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case whether or not having the force of law.
- "Applicable Margin" means the applicable margins set forth in the table in Section 2.9 used in calculating the interest rate applicable to the various types of Borrowings, which shall vary from time to time in accordance with the long term, senior unsecured debt ratings of the Borrower and the General Partner in the manner set forth in Section 2.9. It is hereby understood and agreed that the Applicable Margin shall be adjusted from time to time based upon the Sustainability Margin Adjustment (to be calculated and applied as set forth in Section 1.7); provided that in no event shall the Applicable Margin be less than zero.
- "Approved Fund" means any Fund that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.
- "Arranger" means WFS, BofA Securities, PNC Capital Markets, Regions Capital Markets and U.S. Bank National Association and their successors in their capacity as Joint Lead Arrangers.
- "Asset Sale" means any sale or other disposition by the Consolidated Operating Partnership of any Property or other assets (excluding any proceeds resulting from the casualty or condemnation of such Property or other assets, to the extent such proceeds are used to rebuild such Properties or assets within 365 days of receipt of such proceeds) that yields gross proceeds equal to the aggregate of (i) all cash proceeds, plus (ii) the initial principal amount of any noncash proceeds consisting of notes or other debt securities, plus (iii) the fair market value of other non-cash proceeds, if and to the extent that the aggregate of (i), (ii) and (iii) exceeds \$500,000.

- "Assets Acquired Not In Service" means, as of any date of determination, any Project which has been acquired and owned for less than 12 months but has not yet been leased to 90%.
- "Assets Under Development" means, as of any date of determination, any Project which is under construction and then treated as an asset under development under GAAP.
- "Assignment and Assumption Agreement" means an Assignment and Assumption Agreement among a Lender, an Eligible Assignee and the Administrative Agent, substantially in the form of Exhibit I.
- "Available Tenor" means, as of any date of determination and with respect to the then-current Benchmark, as applicable, (a) if the then-current Benchmark is a term rate, any tenor for such Benchmark or (b) otherwise, any payment period for interest calculated with reference to such Benchmark, as applicable, that is or may be used for determining the length of an Interest Period pursuant to this Agreement as of such date and not including, for the avoidance of doubt, any tenor for such Benchmark that is then-removed from the definition of "Interest Period" pursuant to $\underline{\text{Section 4.8(c)(iv)}}$.
- "Bail-In Action" means the exercise of any Write-Down and Conversion Powers by the applicable Resolution Authority in respect of any liability of an Affected Financial Institution.

"Bail-In Legislation" means (a) with respect to any EEA Member Country implementing Article 55 of Directive 2014/59/EU of the European Parliament and of the Council of the European Union, the implementing law, regulation, rule or requirement for such EEA Member Country from time to time which is described in the EU Bail-In Legislation Schedule and (b) with respect to the United Kingdom, Part I of the United Kingdom Banking Act 2009 (as amended from time to time) and any other law, regulation or rule applicable in the United Kingdom relating to the resolution of unsound or failing banks, investment firms or other financial institutions or their affiliates (other than through liquidation, administration or other insolvency proceedings).

"Base LIBOR Rate" means, subject to the implementation of a Benchmark Replacement in accordance with Section 4.8, with respect to any Eurocurrency Borrowing for any Interest Period, the rate of interest obtained by dividing (i) the rate of interest per annum determined on the basis of the rate as set by the ICE Benchmark Administration ("ICE") (or the successor thereto if ICE is no longer making such rate available) for deposits in Dollars for a period equal to the applicable Interest Period which appears on Reuters Screen LIBOR01 Page (or any applicable successor page), at approximately 11:00 a.m. (London time) two (2) Business Days prior to the first day of the applicable Interest Period by (ii) a percentage equal to 1 minus the Eurodollar Reserve Percentage. If, for any reason, the rate referred to in the preceding clause (i) does not appear on Reuters Screen LIBOR01 Page (or any applicable successor page), then the rate to be used for such clause (i) shall be determined by the Administrative Agent to be the arithmetic average of the rate per annum at which deposits in Dollars would be offered by first class banks in the London interbank market to the Administrative Agent at approximately 11:00 a.m. (London time) two (2) Business Days prior to the first day of the applicable Interest Period for a period equal to such Interest Period. Any change in the Eurodollar Reserve Percentage shall result in a change in the Base LIBOR Rate on the date on which such change in such Eurodollar Reserve Percentage becomes effective. If the Base LIBOR Rate determined as provided above would be less than zero, the Base LIBOR Rate shall be deemed to be zero. Each calculation by the Administrative Agent of the Base LIBOR Rate shall be conclusive and binding for all purposes, absent manifest error. To the extent a comparable or successor rate is approved by the Administrative Agent in connection herewith, the approved rate shall be applied to the then applicable Interest Period in a manner consistent with market practice as reasonably determined by the Administrative Agent; provided that if such market practice is reasonably determined by the Administrative Agent to not be administratively feasible, such approved

rate shall be applied in a manner reasonably determined by the Administrative Agent. All approvals and determinations to be made by the Administrative Agent in the preceding sentence shall be done in consultation with the Borrower. Notwithstanding the foregoing, unless otherwise specified in any amendment to this Agreement entered into in accordance with Section 4.8, in the event that a Benchmark Replacement with respect to the Base LIBOR Rate is implemented then all references herein to the Base LIBOR Rate shall be deemed references to such Benchmark Replacement.

"Base Rate Applicable Margin" means the Applicable Margin in effect for an Adjusted Base Rate Borrowing as determined in accordance with Section 2.9 hereof.

"Benchmark" means, initially, USD LIBOR; provided that if a Benchmark Transition Event, a Term SOFR Transition Event, an Early Optin Election or an Other Benchmark Rate Election, as applicable, and its related Benchmark Replacement Date have occurred with respect to USD LIBOR or the then-current Benchmark, then "Benchmark" means the applicable Benchmark Replacement to the extent that such Benchmark Replacement has replaced such prior benchmark rate pursuant to Section 4.8(c)(i).

"Benchmark Replacement" means, for any Available Tenor,

or

- (a) with respect to any Benchmark Transition Event or Early Opt-in Election, the first alternative set forth in the order below that can be determined by the Administrative Agent for the applicable Benchmark Replacement Date; <u>provided</u>, that, if the Borrower has provided a notification to the Administrative Agent in writing on or prior to such Benchmark Replacement Date that the Borrower has a Rate Management Transaction in place with respect to any of the Loans as of the date of such notice (which such notification the Administrative Agent shall be entitled to rely upon and shall have no duty or obligation to ascertain the correctness or completeness of), then the Administrative Agent, in its sole discretion, may decide not to determine the Benchmark Replacement pursuant to clause (a)(1) or clause (a)(2) for such Benchmark Transition Event or Early Opt-in Election with respect to such Loan, and may instead select (A) the alternate benchmark referenced in such Rate Management Transaction, which may be the sum of a fallback rate and spread adjustment, for the entire balance of such Loan, or (B) the alternate benchmark referenced in such Rate Management Transaction, which may be the sum of a fallback rate and spread adjustment, for the portion of such Loan that is subject to a Rate Management Transaction, and the applicable Benchmark Replacement for the remaining portion of such Loan:
 - (1) the sum of: (A) Term SOFR and (B) the related Benchmark Replacement Adjustment;
 - (2) the sum of: (A) Daily Simple SOFR and (B) the related Benchmark Replacement Adjustment;
 - (3) the sum of: (A) the alternate benchmark rate that has been selected by the Administrative Agent and the Borrower as the replacement for the then-current Benchmark for the applicable Corresponding Tenor giving due consideration to (i) any selection or recommendation of a replacement benchmark rate or the mechanism for determining such a rate by the Relevant Governmental Body or (ii) any evolving or then-prevailing market convention for determining a benchmark rate as a replacement for the then-current Benchmark for Dollar-denominated syndicated credit facilities at such time and (B) the related Benchmark Replacement Adjustment;
 - (b) with respect to any Term SOFR Transition Event, the sum of (i) Term SOFR and (ii) the related Benchmark Replacement Adjustment;

- (c) with respect to any Other Benchmark Rate Election, the sum of: (i) the alternate benchmark rate that has been selected by the Administrative Agent and the Borrower as the replacement for the then-current Benchmark for the applicable Corresponding Tenor giving due consideration to any evolving or then-prevailing market convention for determining a benchmark rate as a replacement for the then-current Benchmark for Dollar-denominated syndicated credit facilities at such time and (ii) the related Benchmark Replacement Adjustment;
- <u>provided</u> that, (i) in the case of clause (a)(1), if the Administrative Agent decides that Term SOFR is not administratively feasible for the Administrative Agent, then Term SOFR will be deemed unable to be determined for purposes of this definition and (ii) in the case of clause (a)(1) or clause (b) of this definition, the applicable Unadjusted Benchmark Replacement is displayed on a screen or other information service that publishes such rate from time to time as selected by the Administrative Agent in its reasonable discretion. If the Benchmark Replacement as determined pursuant to clause (a)(1), (a)(2) or (a)(3), clause (b) or clause (c) of this definition would be less than the Floor, the Benchmark Replacement will be deemed to be the Floor for the purposes of this Agreement and the other Loan Documents.

"Benchmark Replacement Adjustment" means, with respect to any replacement of the then-current Benchmark with an Unadjusted Benchmark Replacement for any applicable Interest Period and Available Tenor for any setting of such Unadjusted Benchmark Replacement:

- (1) for purposes of clauses (a)(1) and (b) of the definition of "Benchmark Replacement," an amount equal to (A) 0.11448% (11.448 basis points) for an Available Tenor of one-month's duration, (B) 0.26161% (26.161 basis points) for an Available Tenor of three-months' duration and (C) 0.42826% (42.826 basis points) for an Available Tenor of six-months' duration;
 - (2) for purposes of clause (a)(2) of the definition of "Benchmark Replacement," an amount equal to 0.11448% (11.448 basis points);
- (3) for purposes of clause (a)(3) of the definition of "Benchmark Replacement," the spread adjustment, or method for calculating or determining such spread adjustment, (which may be a positive or negative value or zero) that has been selected by the Administrative Agent and the Borrower giving due consideration to (i) any selection or recommendation of a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Available Tenor of such Benchmark with the applicable Unadjusted Benchmark Replacement by the Relevant Governmental Body on the applicable Benchmark Replacement Date or (ii) any evolving or then-prevailing market convention for determining a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Available Tenor of such Benchmark with the applicable Unadjusted Benchmark Replacement for Dollar-denominated syndicated credit facilities; and
- (4) for purposes of clause (c) of the definition of "Benchmark Replacement," the spread adjustment, or method for calculating or determining such spread adjustment, (which may be a positive or negative value or zero) that has been selected by the Administrative Agent and the Borrower giving due consideration to any evolving or then-prevailing market convention for determining a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Available Tenor of such Benchmark with the applicable Unadjusted Benchmark Replacement for Dollar-denominated syndicated credit facilities.

"Benchmark Replacement Conforming Changes" means, with respect to any Benchmark Replacement, any technical, administrative or operational changes (including changes to the definition of "Adjusted Base Rate," the definition of "Business Day," the definition of "Interest Period," the definition of "LIBOR Market Index Rate", timing and frequency of determining rates and making payments of

interest, timing of borrowing requests or prepayment, conversion or continuation notices, length of lookback periods, the applicability of breakage provisions, and other technical, administrative or operational matters) that the Administrative Agent decides may be appropriate to reflect the adoption and implementation of such Benchmark Replacement and to permit the administration thereof by the Administrative Agent in a manner substantially consistent with market practice (or, if the Administrative Agent decides that adoption of any portion of such market practice is not administratively feasible or if the Administrative Agent determines that no market practice for the administration of such Benchmark Replacement exists, in such other manner of administration as the Administrative Agent decides is reasonably necessary in connection with the administration of this Agreement and the other Loan Documents).

"Benchmark Replacement Date" means the earliest to occur of the following events with respect to the then-current Benchmark:

- (a) in the case of clause (a) or (b) of the definition of "Benchmark Transition Event," the later of (a) the date of the public statement or publication of information referenced therein and (b) the date on which the administrator of such Benchmark (or the published component used in the calculation thereof) permanently or indefinitely ceases to provide all Available Tenors of such Benchmark (or such component thereof);
- (b) in the case of clause (c) of the definition of "Benchmark Transition Event," the date of the public statement or publication of information referenced therein;
- (c) in the case of a Term SOFR Transition Event, the date that is thirty (30) days after the Administrative Agent has provided the Term SOFR Notice to the Lenders and the Borrower pursuant to $\underline{Section 4.8(c)(i)(B)}$; or
- (d) in the case of an Early Opt-in Election or an Other Benchmark Rate Election, the sixth (6th) Business Day after the date notice of such Early Opt-in Election or Other Benchmark Rate Election, as applicable, is provided to the Lenders, so long as the Administrative Agent has not received, by 5:00 p.m. (New York City time) on the fifth (5th) Business Day after the date notice of such Early Opt-in Election or Other Benchmark Rate Election, as applicable, is provided to the Lenders, written notice of objection to such Early Opt-in Election or Other Benchmark Rate Election, as applicable, from Lenders comprising the Required Lenders.

For the avoidance of doubt, (i) if the event giving rise to the Benchmark Replacement Date occurs on the same day as, but earlier than, the Reference Time in respect of any determination, the Benchmark Replacement Date will be deemed to have occurred prior to the Reference Time for such determination and (ii) the "Benchmark Replacement Date" will be deemed to have occurred in the case of clause (a) or (b) with respect to any Benchmark upon the occurrence of the applicable event or events set forth therein with respect to all then-current Available Tenors of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Transition Event" means the occurrence of one or more of the following events with respect to the then-current Benchmark:

(a) a public statement or publication of information by or on behalf of the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that such administrator has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof), permanently or indefinitely, <u>provided</u> that, at the time of such statement or

publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof);

- (b) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof), the FRB, the Federal Reserve Bank of New York, an insolvency official with jurisdiction over the administrator for such Benchmark (or such component) or a court or an entity with similar insolvency or resolution authority over the administrator for such Benchmark (or such component), which states that the administrator of such Benchmark (or such component) has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof) permanently or indefinitely, provided that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof); or
- (c) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that all Available Tenors of such Benchmark (or such component thereof) are no longer representative.

For the avoidance of doubt, a "Benchmark Transition Event" will be deemed to have occurred with respect to any Benchmark if a public statement or publication of information set forth above has occurred with respect to each then-current Available Tenor of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Unavailability Period" means the period (if any) (x) beginning at the time that a Benchmark Replacement Date pursuant to clauses (a) or (b) of that definition has occurred if, at such time, no Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with $\underline{Section 4.8(c)}$ and (y) ending at the time that a Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with $\underline{Section 4.8(c)}$.

"Beneficial Ownership Certification" means a certification regarding beneficial ownership as required by the Beneficial Ownership Regulation.

"Beneficial Ownership Regulation" means 31 CFR § 1010.230.

"Benefit Plan" means any of (a) an "employee benefit plan" (as defined in ERISA) that is subject to Title I of ERISA, (b) a "plan" as defined in and subject to Section 4975 of the Code or (c) any Person whose assets include (for purposes of ERISA Section 3(42) or otherwise for purposes of Title I of ERISA or Section 4975 of the Code) the assets of any such "employee benefit plan" or "plan".

"BHC Act Affiliate" is defined in Section 14.19.

"Bid Rate Borrowing" has the meaning given that term in Section 2.18(b).

"Bid Rate Loan" means a loan made by a Lender under Section 2.18.

"Bid Rate Note" means a promissory note of the Borrower substantially in the form of Exhibit B-2 (or such other form approved by the Administrative Agent), payable to the order of a Lender as originally in effect and otherwise duly completed.

- "Bid Rate Quote" means an offer in accordance with Section 2.18(c) by a Lender to make a Bid Rate Loan with one single specified interest rate.
 - "Bid Rate Quote Request" has the meaning given that term in Section 2.18(b).
- "<u>BofA Securities</u>" means BofA Securities, Inc. (together with any affiliates it deems appropriate to provide the services contemplated herein).
 - "Borrower" means First Industrial, L.P., along with its permitted successors and assigns.
 - "Borrower EP Interest Accrual Date" has the meaning assigned thereto in Section 12.19(c).
- "Borrowing" means (a) Loans of the same Class and Type, made, converted or continued on the same date and, in the case of Eurocurrency Loans, as to which a single Interest Period is in effect, (b) a Swingline Loan, or (c) a Bid Rate Loan.
 - "Borrowing Date" means a Business Day on which a Borrowing is made to the Borrower.
 - "Borrowing Notice" is defined in Section 2.7 hereof.
- "Business Day." means a day, other than a Saturday, Sunday or holiday, on which banks are open for business in New York City; provided that with respect to any borrowings, disbursements and payments in respect of and calculations, interest rates and Interest Periods pertaining to any Eurocurrency Borrowings, such day is also a day on which banks are open for general business in London, England.
- "<u>Capital Stock</u>" means any and all shares, interests, participations or other equivalents (however designated) of capital stock of a corporation, any and all equivalent ownership interests in a Person which is not a corporation and any and all warrants or options to purchase any of the foregoing.
- "<u>Cash Collateralize</u>" means, to pledge and deposit with or deliver to the Administrative Agent, for the benefit of the applicable Issuing Bank or the Lenders, as collateral for LC Exposure or obligations of Lenders to fund participations in respect of LC Exposure, cash or deposit account balances or, if the Administrative Agent and the applicable Issuing Bank shall agree in their sole discretion, other credit support, in each case pursuant to documentation in form and substance satisfactory to the Administrative Agent and the applicable Issuing Bank. "Cash Collateral" shall have a meaning correlative to the foregoing and shall include the proceeds of such cash collateral and other credit support.
- "<u>Cash Equivalents</u>" shall mean (i) short term obligations of, or fully guaranteed by, the United States of America, (ii) commercial paper rated A 1 or better by Standard and Poor's Corporation or P-1 or better by Moody's Investors Service, Inc., or (iii) certificates of deposit issued by and time deposits with commercial banks (whether domestic or foreign) having capital and surplus in excess of \$100,000,000; provided in each case that the same provides for payment of both principal and interest (and not principal alone or interest alone) and is not subject to any contingency regarding the payment of principal or interest, <u>provided</u> that all such Cash Equivalents would qualify as cash equivalents in accordance with GAAP.
- "<u>Change in Control</u>" means (a) the acquisition of ownership, directly or indirectly, beneficially or of record, by any Person or group (within the meaning of the Securities Exchange Act of 1934 and the rules of the Securities and Exchange Commission thereunder as in effect on the date hereof), of more than 40% of the aggregate voting power represented by the then outstanding voting stock of the General Partner, (b) the occupation of a majority of the seats (other than vacant seats) on the board of directors of

the General Partner by Persons who were neither (i) nominated by the board of directors of the General Partner, (ii) appointed by directors so nominated, nor (iii) nominated by holders of the preferred stock in the General Partner pursuant to the terms of such stock; or (c) the General Partner shall cease to own, directly or indirectly, fifty-one percent (51%) or more of the Equity Interests of the Borrower.

"Change in Law" means the occurrence, after the date of this Agreement, of any of the following: (a) the adoption or taking effect of any law, rule, regulation or treaty, (b) any change in any law, rule, regulation or treaty or in the administration, interpretation or application thereof by any Governmental Authority, or (c) the making or issuance of any request, rules, guideline, requirement or directive (whether or not having the force of law) by any Governmental Authority; provided however, that notwithstanding anything herein to the contrary, (i) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines, requirements and directives thereunder, issued in connection therewith or in implementation thereof, and (ii) all requests, rules, guidelines, requirements and directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, shall in each case be deemed to be a "Change in Law" regardless of the date enacted, adopted, issued or implemented.

"Class" means when used in reference to any Loan or Borrowing, refers to whether such Loan, or the Loans comprising such Borrowing, are Revolving Loans, Swingline Loans or Bid Rate Loans.

"Closing Date" means the date on which the conditions precedent in Section 5.1 are satisfied in accordance therewith and this Agreement becomes effective.

"Code" means the Internal Revenue Code of 1986 as amended from time to time, or any replacement or successor statute, and the regulations promulgated thereunder from time to time.

"Consolidated Leverage Ratio" means, as of any date of determination, the ratio of Consolidated Total Indebtedness to Implied Capitalization Value of the Consolidated Operating Partnership.

"Consolidated Leverage Ratio Increase Period" is defined in Section 9.7(b).

"Consolidated Operating Partnership" means the Borrower, the General Partner and any other subsidiary partnerships or entities of either of them which are required under GAAP to be consolidated with the Borrower and the General Partner for financial reporting purposes.

"Consolidated Secured Debt" means as of any date of determination, the sum of (a) the aggregate outstanding principal amount of all Indebtedness of the Consolidated Operating Partnership outstanding at such date which is secured by a Lien on any asset or Capital Stock of Consolidated Operating Partnership, including without limitation loans secured by mortgages, stock, or partnership interests, but excluding Defeased Debt and (b) the amount by which the aggregate principal amount of all Indebtedness of the Subsidiaries of the Borrower or the General Partner outstanding at such date exceeds \$5,000,000 (for the avoidance of doubt, (x) excluding Indebtedness of the Borrower and (y) including Guarantee Obligations (other than customary non-recourse carveout obligations) of the Subsidiaries of the General Partner (other than the Borrower) in respect of primary obligations of the Borrower or the General Partner), without duplication of any Indebtedness included under clause (a). For clarification, Consolidated Secured Debt shall include the Borrower's or the General Partner's Ownership Share of any Investment Affiliate's Indebtedness.

"Consolidated Senior Unsecured Debt" means as of any date of determination, the aggregate outstanding principal amount of all Indebtedness of the Consolidated Operating Partnership (which will

include, without limitation, any Indebtedness that is secured by partnership interests and that is recourse to the Borrower or the Guarantor, where such recourse component applies only to the payment of principal and/or interest), outstanding at such date other than (a) Indebtedness which is contractually subordinated to the Indebtedness of the Consolidated Operating Partnership under the Loan Documents on terms acceptable to the Administrative Agent and (b) that portion of Consolidated Secured Debt described in clause (a) of that definition. For clarification, Consolidated Senior Unsecured Debt shall exclude the Borrower's or the General Partner's Ownership Share of any Investment Affiliate's Indebtedness.

"Consolidated Total Indebtedness" means as of any date of determination, all Indebtedness of the Consolidated Operating Partnership outstanding at such date, determined on a consolidated basis in accordance with GAAP, after eliminating intercompany items; provided that for purposes of defining "Consolidated Total Indebtedness" the term "Indebtedness" shall not include the short term debt (e.g. accounts payable, short term expenses) of the Borrower or the General Partner or Defeased Debt. For clarification, Consolidated Total Indebtedness shall include the Borrower's or the General Partner's Ownership Share of any Investment Affiliate's Indebtedness.

"Controlled Group" means all members of a controlled group of corporations and all trades or businesses (whether or not incorporated) under common control which, together with all or any of the entities in the Consolidated Operating Partnership, are treated as a single employer under Sections 414(b) or 414(c) of the Code.

"Conversion/Continuation Notice" is defined in Section 2.8.

"<u>Corresponding Tenor</u>" with respect to any Available Tenor means, as applicable, either a tenor (including overnight) or an interest payment period having approximately the same length (disregarding business day adjustment) as such Available Tenor.

"Covered Entity" is defined in Section 14.19.

"<u>Daily Simple SOFR</u>" means, for any day, SOFR, with the conventions for this rate (which will include a lookback) being established by the Administrative Agent in accordance with the conventions for this rate selected or recommended by the Relevant Governmental Body for determining "Daily Simple SOFR" for syndicated business loans; <u>provided</u>, that if the Administrative Agent decides that any such convention is not administratively feasible for the Administrative Agent, then the Administrative Agent may establish another convention in its reasonable discretion.

"<u>Debtor Relief Laws</u>" means the United States Bankruptcy Code, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar Applicable Laws relating to the relief of debtors in the United States of America or other applicable jurisdictions from time to time in effect.

"<u>Debt Service</u>" means for any period, (a) Interest Expense for such period <u>plus</u> (b) the aggregate amount of regularly scheduled principal payments of Indebtedness (excluding optional prepayments and balloon principal payments due on maturity in respect of any Indebtedness and mandatory excess cash flow sweeps) required to be made during such period by the Borrower, or any of its consolidated Subsidiaries <u>plus</u> (c) a percentage of all such regularly scheduled principal payments required to be made during such period by any Investment Affiliate on Indebtedness (excluding optional prepayments and balloon principal payments due on maturity in respect of any Indebtedness) taken into account in calculating Interest Expense, such percentage equal to the greater of (x) the percentage of the principal amount of such Indebtedness for which the Borrower or any consolidated Subsidiary is liable and (y) the

Ownership Share in such Investment Affiliate held by the Borrower and any consolidated Subsidiaries, in the aggregate, without duplication.

"Default" means an event which, with notice or lapse of time or both, would become an Event of Default.

"<u>Default Rate</u>" means with respect to any Borrowing, a rate equal to the interest rate applicable to such Borrowing plus three percent (3%) per annum.

"Default Right" is defined in Section 14.19.

"Defaulting Lender" means, subject to Section 12.16(f), any Revolving Lender that (a) has failed to (i) fund all or any portion of its Loans within two (2) Business Days of the date such Loans were required to be funded hereunder unless such Revolving Lender notifies the Administrative Agent and the Borrower in writing that such failure is the result of such Revolving Lender's determination that one or more conditions precedent to funding (each of which conditions precedent, together with any applicable default, shall be specifically identified in such writing) has not been satisfied, or (ii) pay to the Administrative Agent, any Issuing Bank, the Swingline Lender or any other Lender any other amount required to be paid by it hereunder (including in respect of its participation in Letters of Credit or Swingline Loans) within two (2) Business Days of the date when due, (b) has notified the Borrower, the Administrative Agent, any Issuing Bank or the Swingline Lender in writing that it does not intend to comply with its funding obligations hereunder, or has made a public statement to that effect (unless such writing or public statement relates to such Revolving Lender's obligation to fund a Loan hereunder and states that such position is based on such Revolving Lender's determination that a condition precedent to funding (which condition precedent, together with any applicable default, shall be specifically identified in such writing or public statement) cannot be satisfied), (c) has failed, within three (3) Business Days after written request by the Administrative Agent or the Borrower, to confirm in writing to the Administrative Agent and the Borrower that it will comply with its prospective funding obligations hereunder (provided that such Revolving Lender shall cease to be a Defaulting Lender pursuant to this clause (c) upon receipt of such written confirmation by the Administrative Agent and the Borrower), or (d) has, or has a direct or indirect parent company that has, (i) become the subject of (A) a proceeding under any Debtor Relief Law or (B) a Bail-In Action, or (ii) had appointed for it a receiver, custodian, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or assets, including the Federal Deposit Insurance Corporation or any other state or federal regulatory authority acting in such a capacity; provided that a Revolving Lender shall not be a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Revolving Lender or any direct or indirect parent company thereof by a Governmental Authority so long as such ownership interest does not result in or provide such Revolving Lender with immunity from the jurisdiction of courts within the United States of America or from the enforcement of judgments or writs of attachment on its assets or permit such Revolving Lender (or such Governmental Authority) to reject, repudiate, disayow or disaffirm any contracts or agreements made with such Revolving Lender. Any determination by the Administrative Agent that a Revolving Lender is a Defaulting Lender under clauses (a) through (d) above shall be conclusive and binding absent manifest error, and such Revolving Lender shall be deemed to be a Defaulting Lender (subject to Section 12.16(f)) upon delivery of written notice of such determination to the Borrower, any Issuing Bank, the Swingline Lender and each Revolving Lender.

"<u>Defeased Debt</u>" means that portion of debt which has already been defeased by depositing collateral in the form of obligations supported by the credit of the United States government in such amounts as are required and permitted under the terms of the applicable loan documents.

"<u>Departing Lender</u>" means each Lender under the Existing Credit Agreement that executes and delivers to the Administrative Agent a Departing Lender Signature Page.

"<u>Departing Lender Signature Page</u>" means each signature page to this Agreement on which it is indicated that the Departing Lender executing the same shall cease to be a party to the Existing Credit Agreement on the Agreement Execution Date.

"<u>Designated Lender</u>" means a special purpose corporation which is an Affiliate of, or sponsored by, a Lender, that is engaged in making, purchasing or otherwise investing in commercial loans in the ordinary course of its business and that issues (or the parent of which issues) commercial paper rated at least P-1 (or the then equivalent grade) by Moody's or A-1 (or the then equivalent grade) by S&P that, in either case, (a) is organized under the laws of the United States of America or any state thereof, (b) shall have become a party to this Agreement pursuant to <u>Section 13.5</u> and (c) is not otherwise a Lender.

"Designating Lender" has the meaning given that term in Section 13.5.

" $\underline{\text{Designation Agreement}}$ " means a Designation Agreement between a Lender and a Designated Lender and accepted by the Administrative Agent, substantially in the form of $\underline{\text{Exhibit L}}$ or such other form as may be agreed to by such Lender, such Designated Lender and the Administrative Agent.

"Disbursement Instruction Agreement" means an agreement substantially in the form of Exhibit \underline{C} to be executed and delivered by the Borrower pursuant to Section 12.10, as the same may be amended, restated or modified from time to time with the prior written approval of the Administrative Agent.

"Dollars" and "\$" mean United States Dollars.

"Early Opt-in Election" means, if the then-current Benchmark is USD LIBOR, the occurrence of:

- (a) a notification by the Administrative Agent to (or the request by the Borrower to the Administrative Agent to notify) each of the other parties hereto that at least five currently outstanding Dollar-denominated syndicated credit facilities at such time contain (as a result of amendment or as originally executed) a SOFR-based rate (including SOFR, a term SOFR or any other rate based upon SOFR) as a benchmark rate (and such syndicated credit facilities are identified in such notice and are publicly available for review), and
- (b) the joint election by the Administrative Agent and the Borrower to trigger a fallback from USD LIBOR and the provision by the Administrative Agent of written notice of such election to the Lenders.

"EBITDA" means, with respect to any Person, income before restructuring charges, non-cash impairment charges and other non-cash, non-recurring items determined in good faith by the Borrower and extraordinary items, without deduction of any losses related to initial offering costs of preferred stock which are written off due to the redemption of such preferred stock, and excluding any gains or losses from pay-off or retirement of debt and gains/losses on sales of Properties and excluding costs incurred in acquiring Properties, where such costs are required to be expensed under ASC 805 Business Combinations 805-10-25-23, as reported by such Person and its Subsidiaries on a consolidated basis in accordance with GAAP (reduced to eliminate any income from Investment Affiliates of such Person, any interest income and, with respect to the Consolidated Operating Partnership, any income from the assets used for Defeased Debt), plus Interest Expense, depreciation, amortization and income tax (if any) expense plus a percentage of such income (adjusted as described above) of any such Investment Affiliate

equal to the allocable economic interest in such Investment Affiliate held by such Person and any Subsidiaries, in the aggregate (provided that no item of income or expense shall be included more than once in such calculation even if it falls within more than one of the foregoing categories).

"EEA Financial Institution" means (a) any credit institution or investment firm established in any EEA Member Country which is subject to the supervision of an EEA Resolution Authority, (b) any entity established in an EEA Member Country which is a parent of an institution described in clause (a) of this definition, or (c) any financial institution established in an EEA Member Country which is a subsidiary of an institution described in clauses (a) or (b) of this definition and is subject to consolidated supervision with its parent.

"EEA Member Country" means any of the member states of the European Union, Iceland, Liechtenstein, and Norway.

"EEA Resolution Authority" means any public administrative authority or any Person entrusted with public administrative authority of any EEA Member Country (including any delegee) having responsibility for the resolution of any credit institution or investment firm established in any EEA Member Country.

"Effective Date" means each Borrowing Date (other than a continuation or conversion) and, if no Borrowing Date has occurred in the preceding calendar month, the first Business Day of each calendar month.

"Eligible Assignee" means (a) a Lender, (b) an Affiliate of a Lender, (c) an Approved Fund and (d) any other Person (other than a natural person) approved by the Administrative Agent (such approval not to be unreasonably withheld or delayed by any of the foregoing); provided that notwithstanding the foregoing, "Eligible Assignee" shall not include the Borrower or any of the Borrower's Affiliates or Subsidiaries.

"Environmental Laws" means any and all Federal, state, local or municipal laws, rules, orders, regulations, statutes, ordinances, codes, decrees, requirements of any Governmental Authority having jurisdiction over the Borrower, its Subsidiaries or Investment Affiliates, or their respective assets, and regulating or imposing liability or standards of conduct concerning protection of human health or the environment, as now or may at any time hereafter be in effect, in each case to the extent the foregoing are applicable to the operations of the Borrower, any Investment Affiliate, or any Subsidiary or any of their respective assets or Properties.

"Equity Interests" means, with respect to any Person, any share of capital stock of (or other ownership or profit interests in) such Person, any warrant, option or other right for the purchase or other acquisition from such Person of any share of capital stock of (or other ownership or profit interests in) such Person, any security convertible into or exchangeable for any share of capital stock of (or other ownership or profit interests in) such Person or warrant, right or option for the purchase or other acquisition from such Person of such shares (or such other interests), and any other ownership or profit interest in such Person (including, without limitation, partnership, member or trust interests therein), whether voting or nonvoting, and whether or not such share, warrant, option, right or other interest is authorized or otherwise existing on any date of determination.

"Equity Value" is defined in Section 10.10 hereof.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and regulations promulgated thereunder from time to time.

- "Erroneous Payment" is defined in Section 12.19(a) hereof.
- "EU Bail-In Legislation Schedule" means the EU Bail-In Legislation Schedule published by the Loan Market Association (or any successor thereto), as in effect from time to time.
- "<u>Eurocurrency</u>", when used in reference to any Loan or Borrowing, refers to whether such Loan, or the Loans comprising such Borrowing, are bearing interest at a rate determined by reference to the Adjusted LIBOR Rate.
- "<u>Eurodollar Reserve Percentage</u>" means, for any day, the percentage (stated as a decimal) which is in effect for such day as prescribed by the Board of Governors of the Federal Reserve System (or any successor) for determining the maximum reserve requirement (including without limitations, any basic, supplemental or emergency reserves) in respect of eurocurrency liabilities or any similar category.
 - "Event of Default" means any event set forth in Article X hereof.
- "Excluded Taxes" means, in the case of each Lender or applicable Lending Office and the Administrative Agent, (a) taxes imposed on its overall net income, and franchise taxes imposed on it, by (i) the jurisdiction under the laws of which such Lender or the Administrative Agent is incorporated or organized or (ii) the jurisdiction in which the Administrative Agent's or such Lender's principal executive office of such Lender's applicable Lending Office is located, and (b) any United States federal withholding taxes imposed by FATCA.
 - "Existing Credit Agreement" has the meaning specified in the preliminary statements of this Agreement
 - "Extended Letter of Credit" is defined in Section 3.2.
 - "Extension Request" is defined in Section 2.19(a).
 - "Facility" means the Revolving Commitments and the extensions of credit made thereunder.
 - "Facility Fee Rate" is defined in Section 2.10(b).
 - "Facility Letter of Credit" means a Financial Letter of Credit or Performance Letter of Credit issued under the Facility.
 - "Facility Letter of Credit Fee" is defined in Section 3.8.
- "FATCA" means Sections 1471 through 1474 of the Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with) and any current or future regulations or official interpretations thereof and any agreement entered into pursuant to Section 1471(b)(1) of the Code.
- "<u>Federal Funds Effective Rate</u>" means, for any period, a fluctuating interest rate per annum equal for each day during such period to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System, as published for such day (or, if such day is not a Business Day, for the immediately preceding Business Day) by the Federal Reserve Bank of New York, or, if such rate is not so published for any day which is a Business Day, the average of the quotations for such day on such transactions received by the Administrative Agent from three Federal funds brokers of recognized standing selected by the Administrative Agent in its sole discretion;

provided, that, if the Federal Funds Effective Rate shall be less than zero, such rate shall be deemed to be zero for purposes of this Agreement.

"<u>Financial Letter of Credit</u>" means any standby Letter of Credit which represents an irrevocable obligation to the beneficiary on the part of the applicable Issuing Bank (i) to repay money borrowed by or advanced to or for the account of the account party or (ii) to make any payment on account of any indebtedness undertaken by the account party, in the event the account party fails to fulfill its obligation to the beneficiary.

"Fixed Charge Coverage Ratio" is defined in Section 9.7(a).

"Floor" means the benchmark rate floor, if any, provided in this Agreement initially (as of the execution of this Agreement, the modification, amendment or renewal of this Agreement or otherwise) with respect to USD LIBOR.

"FRB" means the Board of Governors of the Federal Reserve System of the United States.

"Fronting Exposure" means, at any time there is a Defaulting Lender, (a) with respect to the applicable Issuing Bank, such Defaulting Lender's Percentage of the outstanding LC Exposure in respect of Letters of Credit issued by such Issuing Bank other than LC Exposure as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders or Cash Collateralized in accordance with the terms hereof, and (b) with respect to the Swingline Lender, such Defaulting Lender's Percentage of outstanding Swingline Loans other than Swingline Loans as to which such Defaulting Lender's participation obligation has been reallocated to other Lenders.

"Fund" means any Person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its business.

"<u>Funded Percentage</u>" means, with respect to any Lender at any time, a percentage equal to a fraction the numerator of which is the amount of the Aggregate Revolving Commitment actually disbursed and outstanding to the Borrower by such Lender at such time, and the denominator of which is the total amount of the Aggregate Revolving Commitment disbursed and outstanding to the Borrower by all of the Lenders at such time.

"GAAP" means those generally accepted accounting principles in the United States of America as in effect from time to time that are consistent with those utilized in preparing the audited financial statements of the Borrower required hereunder; provided that all financial computations shall be made in accordance with GAAP as in effect on the Agreement Execution Date.

"General Partner" means First Industrial Realty Trust, Inc., a Maryland corporation that is listed on a national securities exchange and is qualified as a real estate investment trust. The General Partner is the sole general partner of the Borrower.

"Governmental Authority" means the government of the United States of America or any other nation, or of any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government (including any supra-national bodies such as the European Union or the European Central Bank).

"Gross Negligence" means recklessness, or actions taken or omitted with conscious indifference to or the complete disregard of consequences or rights of others affected. Gross Negligence does not mean the absence of ordinary care or diligence, or an inadvertent act or inadvertent failure to act. If the term "gross negligence" is used with respect to the Administrative Agent, any Issuing Bank or any Lender or any indemnitee in any of the other Loan Documents, it shall have the meaning set forth herein.

"Ground Lease Payments" means, for any period, payment made in cash during such period in respect of any ground lease with respect to which the Borrower or any of its Subsidiaries is a lessee.

"Guarantee Obligation" means as to any Person (the "guaranteeing person"), any obligation (determined without duplication) of (a) the guaranteeing person or (b) another Person (including, without limitation, any bank under any letter of credit) to induce the creation of which the guaranteeing person has issued a reimbursement, counter indemnity or similar obligation, in either case guaranteeing or in effect guaranteeing (but only to the extent that the guaranty applies to the payment of principal or interest due under recourse Indebtedness) any Indebtedness, leases, dividends or other obligations (the "primary obligations") of any other third Person (the "primary obligor") in any manner, whether directly or indirectly, including, without limitation, any obligation of the guaranteeing person, whether or not contingent, (i) to purchase any such primary obligation or any property constituting direct or indirect security therefor, (ii) to advance or supply funds (1) for the purchase or payment of any such primary obligation or (2) to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor, (iii) to purchase property, securities or services primarily for the purpose of assuring the owner of any such primary obligation of the ability of the primary obligor to make payment of such primary obligation or (iv) otherwise to assure or hold harmless the owner of any such primary obligation against loss in respect thereof; provided, however, that the term Guarantee Obligation shall not include endorsements of instruments for deposit or collection in the ordinary course of business. The amount of any Guarantee Obligation of any guaranteeing person shall be deemed to be the maximum stated amount of the primary obligation relating to such Guarantee Obligation (or, if less, the maximum stated liability set forth in the instrument embodying such Guarantee Obligation), provided, that in the absence of any such stated amount or stated liability, the amount of such Guarantee Obligation shall be such guaranteeing person's maximum reasonably anticipated liability in respect thereof as determined by the Borrower in good faith. Notwithstanding the foregoing, a guaranty of customary non-recourse carveouts shall not be deemed a "Guarantee Obligation" for purposes of this Agreement.

"Guaranty" means the Guaranty executed by the General Partner in the form attached hereto as Exhibit D.

"IBA" has the meaning assigned thereto in Section 1.5.

"Implied Capitalization Value" means for any Person as of any date, the sum (without duplication) of (i) the quotient of (x) the Adjusted EBITDA for such Person during the most recent four fiscal quarters (which Adjusted EBITDA shall exclude any Adjusted EBITDA attributable to all assets of the type described in clause (ii), Assets Under Development, Assets Acquired Not in Service or Rollover Projects, and which Adjusted EBITDA attributable to each Project which was formerly a Rollover Project shall not be less than zero), and (y) the Applicable Cap Rate, plus (ii) the purchase price paid by such Person (less any amounts paid to such Person as a purchase price adjustment, held in escrow, retained as a contingency reserve or in connection with other similar arrangements) for any Property (other than Assets Under Development) acquired by such Person during the immediately preceding period of four consecutive fiscal quarters, plus (iii) an amount equal to the then current book value of each Asset Under Development, plus (iv) the then current book value of Unimproved Land, plus (v) with respect to each Rollover Project, an amount equal to 50% of the then current book value, determined in accordance with

GAAP, of such Rollover Project, plus (vi) an amount equal to 100% of unrestricted cash and unrestricted cash equivalents, including any cash on deposit with a qualified intermediary with respect to a deferred tax-free exchange (and specifically excluding any cash or cash equivalents being used to support Defeased Debt), plus (vii) an amount equal to 100% of the then current book value, determined in accordance with GAAP, of all first mortgage receivables on income producing commercial properties. For purposes of determining Implied Capitalization Value, to the extent the amount of Implied Capitalization Value attributable to unconsolidated joint ventures and partnerships and other assets of the type described in clauses (iii), (iv), (v) and (vii) would, in the aggregate, exceed 30% of Implied Capitalization Value, such excess shall be excluded. The Borrower's Ownership Share of assets held by Investment Affiliates (excluding assets of the type described in the immediately preceding clause (vi)) will be included in Implied Capitalization Value calculations consistent with the above described treatment for wholly owned assets. In the case of a newly formed Investment Affiliate, the Borrower's Ownership Share of assets held by the Investment Affiliate shall be calculated by multiplying (a) total assets plus accumulated depreciation of the Investment Affiliate by (b) the Ownership Share of such Investment Affiliate. This valuation methodology will be used for the first four quarters following the formation of any Investment Affiliate. For purposes of computing the Implied Capitalization Value, Adjusted EBITDA may be increased from quarter to quarter by the amount of (A) net cash flow from new leases of space at the Properties (where such net cash flow has not then been included in EBITDA) which have a minimum term of one year and (B) Properties which were previously Assets Under Development but which have been completed during such four quarter period and have at least some tenants in possession of the respective leased spaces and conducting business operations therein each will be included in the calculation of Implied Capitalization Value using pro forma EBITDA for such four quarter period, so long as a "new opening summary" form is submitted to, and approved by, Administrative Agent for each newly-opened Property during such four quarter period. "New opening summary" forms will only be required for developments to the extent the aggregate thereof exceeds \$600,000,000 during any fiscal year.

"Indebtedness" of any Person at any date means without duplication, (a) all indebtedness of such Person for borrowed money, (b) all obligations of such Person for the deferred purchase price of property or services (other than current trade liabilities and other accounts payable, and accrued expenses incurred in the ordinary course of business and payable in accordance with customary practices), repurchase obligations, takeout commitments or forward equity commitments, in each case evidenced by a binding agreement (excluding any such obligation to the extent the obligation can be satisfied by the issuance of Equity Interests (other than Mandatorily Redeemable Stock)), to the extent such obligations constitute indebtedness for the purposes of GAAP, (c) any other indebtedness of such Person which is evidenced by a note, bond, debenture or similar instrument, (d) all obligations of such Person under financing leases and capital leases, (e) all obligations of such Person in respect of acceptances issued or created for the account of such Person, (f) all Guarantee Obligations of such Person (excluding in any calculation of consolidated Indebtedness of the Consolidated Operating Partnership, Guarantee Obligations of any member of the Consolidated Operating Partnership in respect of primary obligations of any other member of the Consolidated Operating Partnership), (g) all reimbursement obligations of such Person for letters of credit and other contingent liabilities, (h) Net Mark-to-Market Exposure under Rate Management Transactions, (i) all liabilities secured by any lien (other than liens for taxes not yet due and payable) on any property owned by such Person even though such Person has not assumed or otherwise become liable for the payment thereof, (j) any repurchase obligation or liability of such Person or any of its Subsidiaries with respect to accounts or notes receivable sold by such Person or any of its Subsidiaries, (k) such Person's Ownership Share of debt in Investment Affiliates and any loans where such Person is liable as a general partner and (1) all obligations of such Person to purchase, redeem, retire, defease or otherwise make any payment in respect of any Mandatorily Redeemable Stock issued by such Person or any other Person, valued at the greater of its voluntary or involuntary liquidation preference plus accrued and unpaid dividends.

"Insolvent" means insolvent as defined in Section 101(32) of the United States Bankruptcy Code, as amended.

"Interest Expense" means all interest expense of the Consolidated Operating Partnership determined in accordance with GAAP <u>plus</u> (i) capitalized interest not covered by an interest reserve from a loan facility, <u>plus</u> (ii) the allocable portion (based on liability) of any interest incurred on any obligation for which the Consolidated Operating Partnership is wholly or partially liable under guaranties covering the payment of principal and/or interest, <u>plus</u> (iii) the allocable percentage of any interest incurred on any Indebtedness of any Investment Affiliate, whether recourse or nonrecourse, equal to the applicable Ownership Share in such Investment Affiliate held by the Consolidated Operating Partnership, in the aggregate, <u>provided</u> that no expense shall be included more than once in such calculation even if it falls within more than one of the foregoing categories; <u>provided</u>, <u>however</u>, that "Interest Expense" shall not include interest on loans after they become Defeased Debt.

"Interest Period" means (a) with respect to a ratable Eurocurrency Borrowing, a period commencing on the date such ratable Eurocurrency Borrowing is made (or in the case of the continuation of a ratable Eurocurrency Loan the last day of the preceding Interest Period for such Loan) and ending on the numerically corresponding day in the first, third or sixth calendar month thereafter (or twelfth calendar month thereafter, with the consent of each Lender), as the Borrower may select in the applicable Borrowing Notice or Conversion/Continuation Notice, as the case may be, except that each Interest Period that commences on the last Business Day of a calendar month (or on any day for which there is no numerically corresponding day in the appropriate subsequent calendar month) shall end on the last Business Day of the appropriate subsequent calendar month; and (b) with respect to each Bid Rate Loan, the period commencing on the date such Bid Rate Loan is made and ending on any Business Day not less than 7 nor more than 270 days thereafter, as the Borrower may select as provided in Section 2.18(b). Notwithstanding the foregoing or anything in this Agreement to the contrary: (i) if any Interest Period would otherwise end after the Maturity Date, such Interest Period shall end on the Maturity Date; and (ii) each Interest Period that would otherwise end on a day which is not a Business Day shall end on the immediately following Business Day (provided that if such immediately following Business Day falls in the next calendar month, such Interest Period shall end on the immediately preceding Business Day).

"Investment Affiliate" means any Person in which the Consolidated Operating Partnership, directly or indirectly, has an ownership interest, whose financial results are not consolidated under GAAP with the financial results of the Consolidated Operating Partnership on the consolidated financial statements of the Consolidated Operating Partnership.

"Investment Grade Rating" means, for any class of non-credit enhanced long-term senior unsecured debt issued by the Borrower, (a) a rating of BBB- or higher from S&P or (b) a rating of Baa3 or better from Moody's.

"ISP" means, with respect to any Letter of Credit, the "International Standby Practices 1998" published by the Institute of International Banking Law & Practice, Inc. (or such later version thereof as may be in effect at the time of issuance).

"Issuance Date" is defined in Section 3.4(a)(iii).

"Issuance Notice" is defined in Section 3.4(c).

"Issuing Bank" means, with respect to each Facility Letter of Credit, the Lender which issues such Facility Letter of Credit. Wells, Bank of America, N.A. and PNC Bank, National Association shall be the sole Issuing Banks.

"KPI Metric" shall mean, for any Reference Year, the ratio, stated as a percentage (rounded to the nearest one decimal point), of:

- (a) the Borrower's aggregate investment incurred (determined in accordance with GAAP, and whether incurred during or prior to such Reference Year) with respect to Substantially Completed Properties for such Reference Year that achieve LEED certification from Green Business Certification, Inc. (or certification under another similarly recognized rating system approved by the Administrative Agent and the Required Lenders) (such certification, the "Sustainability Certification") prior to the delivery of the applicable Pricing Certificate for such Reference Year, and in respect of which the Borrower shall have delivered to the Administrative Agent evidence reasonably acceptable to the Administrative Agent of such Sustainability Certification prior to or concurrently with the delivery of the applicable Pricing Certificate, to
- (b) the Borrower's aggregate investment incurred (determined in accordance with GAAP, and whether incurred during or prior to such Reference Year) with respect to Substantially Completed Properties for such Reference Year;

provided that, in the case of each of clause (a) and (b), such amounts shall be set forth in the Borrower's quarterly "Supplemental Information" report made available on the Borrower's website (or any similar report made publicly available on the Borrower's website) for the final quarterly period of such Reference Year. For purposes of this definition, aggregate investment incurred shall be determined for the Borrower and its Subsidiaries on a consolidated basis in accordance with GAAP and consistent with the determination of the aggregate investment incurred as set forth in the Borrower's quarterly "Supplemental Information" report available on the Borrower's website for the fiscal quarter ending March 31, 2021.

"KPI Metric Applicable Margin Adjustment Amount" means, with respect to any period between Sustainability Pricing Adjustment Dates, solely with respect to the LIBOR Applicable Margin and Base Rate Applicable Margin:

- (a) negative 0.01%, if (i) the then aggregate current book value of all Substantially Completed Properties for the applicable Reference Year shall be equal to or greater than \$100,000,000 as of the last day of such Reference Year and (ii) the KPI Metric for the applicable Reference Year as set forth in the Pricing Certificate for such Reference Year is greater than or equal to the KPI Metric Target for such Reference Year; and
- (b) 0.00%, if (i) the KPI Metric for the applicable Reference Year as set forth in the Pricing Certificate for such Reference Year is less than the KPI Metric Target for such Reference Year, (ii) the Borrower shall fail to deliver a Pricing Certificate (or omit to report the KPI Metric on any Pricing Certificate) for such Reference Year on or prior to the date the annual compliance certificate is required to be delivered pursuant to Section 8.2(iv) or (iii) the then aggregate current book value of all Substantially Completed Properties for the applicable Reference Year shall be less than \$100,000,000 as of the last day of such Reference Year.

"KPI Metric Target" means, with respect to any Reference Year, twenty-five percent (25.0%).

"LC Disbursement" means a payment made by the applicable Issuing Bank pursuant to a Facility Letter of Credit.

"LC Exposure" means at any time, the sum of (a) the aggregate undrawn amount of all outstanding Facility Letters of Credit, plus (b) the aggregate amount of all LC Disbursements that have

not yet been reimbursed by or on behalf of the Borrower at such time. The LC Exposure of any Revolving Lender at any time shall be its Percentage of the total LC Exposure at such time.

"LC Sublimit" is defined in Section 3.2(ii).

"<u>Lenders</u>" means, collectively, Wells, and the other Persons executing this Agreement in such capacity, or any Person which subsequently executes and delivers any amendment hereto in such capacity and each of their respective permitted successors and assigns. Where reference is made to "the Lenders" in any Loan Document it shall be read to mean "all of the Lenders". For the avoidance of doubt, "Lender" excludes the Departing Lenders (except to the extent of any claim made by a Departing Lender pursuant to <u>Section 14.6</u> in its capacity as a "Lender" under the Existing Credit Agreement).

"<u>Lending Office</u>" means, for each Lender and for each Type of Loan, the office of such Lender specified in such Lender's Administrative Questionnaire or in the applicable Assignment and Assumption Agreement or such other office of such Lender as such Lender may notify the Administrative Agent in writing from time to time.

"<u>Letter of Credit</u>" of a Person means a letter of credit or similar instrument which is issued upon the application of such Person or upon which such Person is an account party or for which such Person is in any way liable.

"Letter of Credit Collateral Account" is defined in Section 3.9.

"<u>Letter of Credit Documents</u>" means, with respect to any Letter of Credit, collectively, any application therefor, any certificate or other document presented in connection with a drawing under such Letter of Credit and any other agreement, instrument or other document governing or providing for (a) the rights and obligations of the parties concerned or at risk with respect to such Letter of Credit or (b) any collateral security for any of such obligations.

"Letter of Credit Liabilities" means, without duplication, at any time and in respect of any Letter of Credit, the sum of (a) the maximum undrawn amount of such Letter of Credit plus (b) the aggregate unpaid principal amount of all Reimbursement Obligations of the Borrower at such time due and payable in respect of all drawings made under such Letter of Credit. For purposes of this Agreement, a Lender (other than the Lender then acting as an Issuing Bank) shall be deemed to hold a Letter of Credit Liability in an amount equal to its participation interest under Section 3.6 in the related Letter of Credit, and the Lender then acting as an Issuing Bank shall be deemed to hold a Letter of Credit Liability in an amount equal to its retained interest in the related Letter of Credit after giving effect to the acquisition by the Lenders (other than the Lender then acting as an Issuing Bank) of their participation interests under such Section. For all purposes of this Agreement, if on any date of determination a Letter of Credit has expired by its terms but any amount may still be drawn thereunder by reason of the operation of Rule 3.14 of the ISP, such Letter of Credit shall be deemed to be "outstanding" in the amount so remaining available to be drawn.

"Letter of Credit Request" is defined in Section 3.4(a).

"<u>LIBOR Applicable Margin</u>" means, as of any date with respect to any Eurocurrency Borrowing, the Applicable Margin in effect for such Eurocurrency Borrowing as determined in accordance with <u>Section 2.9</u> hereof.

"<u>LIBOR Auction</u>" means a solicitation of Bid Rate Quotes setting forth LIBOR Margin Loans based on the Base LIBOR Rate pursuant to <u>Section 2.18</u>.

- "LIBOR Borrowing" means, the Eurocurrency Borrowings and Adjusted Base Rate Borrowings, collectively, of any Lender.
- "LIBOR Margin" has the meaning given that term in Section 2.18(c)(ii)(D).
- "LIBOR Margin Loan" means a Bid Rate Loan the interest rate on which is determined on the basis of the Base LIBOR Rate pursuant to a LIBOR Auction.
- "LIBOR Market Index Rate" means, for any day, the Base LIBOR Rate as of that day that would be applicable for a Eurocurrency Borrowing having a one month Interest Period determined at approximately 10:00 a.m. (Central Time) for such day (rather than 11:00 a.m. (London time) two (2) Business Days prior to the first day of such Interest Period as otherwise provided in the definition of Base LIBOR Rate), or if such day is not a Business Day, the immediately preceding Business Day. The LIBOR Market Index Rate shall be determined on a daily basis.
- "<u>Lien</u>" means any mortgage, pledge, security interest, encumbrance, lien or charge of any kind (including, without limitation, any conditional sale or other title retention agreement or lease in the nature thereof, any filing or agreement to file a financing statement as debtor under the Uniform Commercial Code on any property leased to any Person under a lease which is not in the nature of a conditional sale or title retention agreement, or any subordination agreement in favor of another Person).
 - "Loan" means, with respect to a Lender, such Lender's loan made pursuant to Article II (or any conversion or continuation thereof).
- "Loan Documents" means this Agreement, the Notes, the Guaranty and any and all other agreements or instruments required and/or provided to Lenders hereunder or thereunder, as any of the foregoing may be amended from time to time.
- "<u>Loan Party</u>" means each of the Borrower, the General Partner, each other Person who guarantees all or a portion of the Obligations and/or who at any time pledges any collateral to secure all or a portion of the Obligations.
- "London Banking Day" means any day on which dealings in Dollar deposits are conducted by and between banks in the London interbank Eurodollar market.
- "Mandatorily Redeemable Stock" means, with respect to any Person, any Equity Interest of such Person which by the terms of such Equity Interest (or by the terms of any security into which it is convertible or for which it is exchangeable or exercisable), upon the happening of any event or otherwise (a) matures or is mandatorily redeemable, pursuant to a sinking fund obligation or otherwise (other than an Equity Interest to the extent redeemable in exchange for common stock or other equivalent common Equity Interests), (b) is convertible into or exchangeable or exercisable for Indebtedness or Mandatorily Redeemable Stock, or (c) is redeemable at the option of the holder thereof, in whole or in part (other than an Equity Interest which is redeemable solely in exchange for common stock or other equivalent common Equity Interests); in each case, on or prior to the Maturity Date.
- "<u>Market Value Net Worth</u>" means at any time, the Implied Capitalization Value of a Person at such time minus the Indebtedness of such Person at such time.
- "<u>Material Acquisition</u>" means any acquisition (or series of related acquisitions) permitted by the Loan Documents and consummated in accordance with the terms of the Loan Documents if the aggregate

consideration paid in respect of such acquisition (including any Indebtedness assumed in connection therewith) exceeds 10% of the Implied Capitalization Value of the Consolidated Operating Partnership.

"Material Adverse Effect" means, with respect to any matter, that such matter in the Required Lenders' good faith judgment may (x) materially and adversely affect the business, properties, condition or results of operations of the Consolidated Operating Partnership taken as a whole, or (y) constitute a non-frivolous challenge to the validity or enforceability of any material provision of any Loan Document against any obligor party thereto.

"<u>Material Adverse Financial Change</u>" shall be deemed to have occurred if the Required Lenders, in their good faith judgment, determine that a material adverse financial, business or results of operations change has occurred which could prevent timely repayment of any Borrowing hereunder or materially impair the Borrower's ability to perform its obligations under any of the Loan Documents.

"Material Credit Facility" means, as to the Borrower, the General Partner and their Subsidiaries, this Agreement and any other agreement(s) creating or evidencing indebtedness for borrowed money (excluding any Indebtedness which is "non-recourse") entered into on or after the Agreement Execution Date by the Borrower, the General Partner or any of their Subsidiaries, or in respect of which the Borrower, the General Partner or any of their Subsidiaries is an obligor or otherwise provides a guarantee or other credit support (other than a guarantee of customary recourse exceptions) ("Credit Facility"), in a principal amount outstanding or available for borrowing equal to or greater than \$5,000,000 (or the equivalent of such amount in the relevant currency of payment, determined as of the date of the closing of such facility based on the exchange rate of such other currency); and if no Credit Facility or Credit Facilities equal or exceed such amounts, then the largest Credit Facility shall be deemed to be a Material Credit Facility.

"<u>Materials of Environmental Concern</u>" means any gasoline or petroleum (including crude oil or any fraction thereof) or petroleum products or any hazardous or toxic substances, materials or wastes, defined or regulated as such in or under any Environmental Law, including, without limitation, asbestos, radon, polychlorinated biphenyls and urea formaldehyde insulation.

"Maturity Date" means July 7, 2025 or such later date to which the Maturity Date has been extended pursuant to Section 2.19; or such earlier date on which the principal balance of the Facility and all other sums due in connection with the Facility shall be due as a result of the acceleration of the Facility.

"Monetary Default" means any Default involving the Borrower's failure to pay any of the Obligations when due.

"Moody's" means Moody's Investors Service, Inc. and its successors.

"Net Mark-to-Market Exposure" of a Person means, as of any date of determination, the excess (if any) of all unrealized losses over all unrealized profits of such Person arising from Rate Management Transactions. "Unrealized losses" means the fair market value of the cost to such Person of unwinding such Rate Management Transaction as of the date of determination (assuming the Rate Management Transaction were to be terminated as of that date), and "unrealized profits" means the fair market value of the gain to such Person of unwinding such Rate Management Transaction as of the date of determination (assuming such Rate Management Transaction were to be terminated as of that date).

"Non-Defaulting Lender" means, at any time, each Lender that is not a Defaulting Lender at such time.

- "Note" means, to the extent issued pursuant to <u>Section 2.5</u>, with respect to the Facility, the promissory note payable to the order of each Lender in the amount of such Lender's maximum applicable Revolving Commitment in substantially the form attached hereto as <u>Exhibit B-1</u> (or such other form approved by the Administrative Agent) (collectively, the "<u>Notes</u>").
- "Obligations" means the Borrowings, the LC Exposure and all accrued and unpaid fees and all other obligations of the Borrower to the Administrative Agent or any or all of the Lenders arising under this Agreement or any of the other Loan Documents.
 - "OFAC" means the U.S. Department of the Treasury's Office of Foreign Assets Control.
 - "Other Benchmark Rate Election" means, if the then-current Benchmark is USD LIBOR, the occurrence of:
- (a) a notification by the Administrative Agent to (or the request by the Borrower to the Administrative Agent to notify) each of the other parties hereto that at least five currently outstanding Dollar-denominated syndicated credit facilities at such time contain (as a result of amendment or as originally executed), in lieu of a USD LIBOR-based rate, a term benchmark rate that is not a SOFR-based rate as a benchmark rate (and such syndicated credit facilities are identified in such notice and are publicly available for review), and
- (b) the joint election by the Administrative Agent and the Borrower to trigger a fallback from USD LIBOR and the provision by the Administrative Agent of written notice of such election to the Lenders.
 - "Other Taxes" is defined in Section 4.5(ii).
- "Ownership Share" means, with respect to any Investment Affiliate, the pro rata share of the nominal ownership interests held by the Consolidated Operating Partnership, in the aggregate, in such Investment Affiliate, without duplication (e.g., if the Consolidated Operating Partnership owns 25% of an Investment Affiliate, but receives 90% of the economic benefits from such Investment Affiliate, then the Ownership Share shall be equal to 25%).
 - "Payment Date" means the last Business Day of each calendar quarter.
 - "Payment Recipient" has the meaning assigned thereto in Section 12.19(a).
 - "Parent" means, with respect to any Lender, any Person as to which such Lender is, directly or indirectly, a subsidiary.
 - "Participants" is defined in Section 13.2.1 hereof.
 - "PBGC" means the Pension Benefit Guaranty Corporation or any entity succeeding to any or all of its functions under ERISA.
- "Percentage" means, with respect to any Revolving Lender, the percentage of the total Revolving Commitments represented by such Lender's Revolving Commitment. If the Revolving Commitments have terminated or expired, the Percentages shall be determined based upon the Revolving Commitments most recently in effect, giving effect to any assignments and any increase of the Revolving Commitments pursuant to Section 2.2.

"<u>Performance Letter of Credit</u>" means any standby Letter of Credit which represents an irrevocable obligation to the beneficiary on the part of the applicable Issuing Bank to make payment on account of any default by the account party in the performance of a nonfinancial or commercial obligation.

"Permitted Liens" are defined in Section 9.5 hereof.

"<u>Permitted Negative Pledge</u>" means a "negative pledge" that (a) establishes a maximum ratio of unsecured debt to unencumbered assets, or of secured debt to total assets, or that otherwise conditions a Person's ability to encumber its assets upon the maintenance of one or more specified ratios that limit such Person's ability to encumber its assets but that does not generally prohibit the encumbrance of its assets, or the encumbrance of specific assets or (b) provides for the grant in favor of the holders of any unsecured debt of any equal and ratable Lien in connection with the pledge of any property or asset to secure the Facility.

"Person" means an individual, a corporation, a limited or general partnership, an association, a joint venture or any other entity or organization, including a governmental or political subdivision or an agent or instrumentality thereof.

"Plan" means an employee benefit plan as defined in Section 3(3) of ERISA, whether or not terminated, as to which the Borrower or any member of the Controlled Group may have any liability.

"PNC Capital Markets" means PNC Capital Markets LLC (together with any affiliates it deems appropriate to provide the services contemplated herein).

"Pricing Certificate" means a certificate substantially in the form of Exhibit P executed by an officer of the Borrower setting forth in reasonable detail the calculation of KPI Metric and the Sustainability Margin Adjustment, in each case, for the Reference Year covered thereby, and confirming the website for the concurrent public reporting of Borrower's quarterly "Supplemental Information" report for the final quarterly period of such Reference Year.

"Principal L/C Disbursements" is defined in Section 12.16.

"Project" means any real estate asset which is 100% owned by the Borrower or by any Wholly Owned Subsidiary and which is operated as an industrial property.

"Property" means each parcel of real property owned or operated by the Borrower, any Subsidiary or Investment Affiliate.

"Property Operating Income" means, with respect to any Property, for any period, earnings from rental operations (computed in accordance with GAAP but without deduction for reserves) attributable to such Property plus depreciation, amortization and interest expense with respect to such Property for such period, and, if such period is less than a year, adjusted by straight lining various ordinary operating expenses which are payable less frequently than once during every such period (e.g. real estate taxes and insurance). The earnings from rental operations reported for the immediately preceding fiscal quarter shall be adjusted to include pro forma earnings (as substantiated to the satisfaction of the Administrative Agent) for an entire quarter for any Property acquired or placed in service during such fiscal quarter and to exclude earnings during such quarter from any property not owned as of the end of the quarter.

"PTE" means a prohibited transaction class exemption issued by the U.S. Department of Labor, as any such exemption may be amended from time to time.

"Purpose Credit" has the meaning ascribed to it in Regulation U of the Board of Governors of the Federal Reserve System.

"OFC" is defined in Section 14.19.

"Qualified Officer" means, with respect to any entity, the chief financial officer, chief accounting officer, controller or assistant controller of such entity if it is a corporation or of such entity's general partner if it is a partnership.

"Rate Management Obligations" of a Person means any and all obligations of such Person, whether absolute or contingent and howsoever and whensoever created, arising, evidenced or acquired (including all renewals, extensions and modifications thereof and substitutions therefor), under (i) any and all Rate Management Transactions, and (ii) any and all cancellations, buy backs, reversals, terminations or assignments of any Rate Management Transactions.

"Rate Management Transaction" means any transaction (including an agreement with respect thereto) now existing or hereafter entered by the Borrower which is a rate swap, basis swap, forward rate transaction, commodity swap, commodity option, equity or equity index swap, equity or equity index option, interest rate option, foreign exchange transaction, cap transaction, floor transaction, collar transaction, forward transaction, currency swap transaction, cross-currency rate swap transaction, currency option or any other similar transaction (including any option with respect to any of these transactions) or any combination thereof, whether linked to one or more interest rates, foreign currencies, commodity prices, equity prices or other financial measures.

"Rating Agency" means S&P or Moody's.

"Recovery Event" means any settlement of or payment in respect of any property or casualty insurance claim or any condemnation proceeding relating to any Property.

"Reference Time" with respect to any setting of the then-current Benchmark means (1) if such Benchmark is USD LIBOR, 11:00 a.m. (London time) on the day that is two (2) London Banking Days preceding the date of such setting, and (2) if such Benchmark is not USD LIBOR, the time determined by the Administrative Agent in its reasonable discretion.

"Reference Year" means, with respect to any Pricing Certificate, the fiscal year ending immediately prior to the date of such Pricing Certificate.

"Register" is defined in Section 13.4.

"Regulation D" means Regulation D of the Board of Governors of the Federal Reserve System as from time to time in effect and any successor or other regulation or official interpretation of said Board of Governors relating to reserve requirements applicable to member banks of the Federal Reserve System.

"Reimbursement Obligations" means at any time, the aggregate of the Obligations of the Borrower to the Lenders, the Issuing Banks and the Administrative Agent in respect of all unreimbursed payments or disbursements made by the Lenders, the Issuing Banks and the Administrative Agent under or in respect of the Facility Letters of Credit.

"Relevant Governmental Body" means the FRB or the Federal Reserve Bank of New York, or a committee officially endorsed or convened by the FRB or the Federal Reserve Bank of New York, or any successor thereto.

"Reportable Event" means a reportable event as defined in Section 4043 of ERISA and the regulations issued under such Section, with respect to a Plan, excluding, however, such events as to which the PBGC by regulation waived the requirement of Section 4043(a) of ERISA that it be notified within 30 days of the occurrence of such event, <u>provided</u> that a failure to meet the minimum funding standard of Section 412 of the Code and of Section 302 of ERISA shall be a Reportable Event regardless of the issuance of any such waivers in accordance with either Section 4043(a) of ERISA or Section 412(d) of the Code.

"Required Lenders" means Lenders in the aggregate having at least 51% of the Aggregate Revolving Commitment then in effect or, if the Aggregate Revolving Commitment has been terminated, the Total Revolving Exposure then outstanding. In determining such percentage at any given time, all then existing Defaulting Lenders will be disregarded and excluded, and at all times when two or more Lenders (excluding Defaulting Lenders) are party to this Agreement, the term "Required Lenders" shall in no event mean less than two Lenders. For purposes of this definition, a Lender shall be deemed to hold a Swingline Loan or Reimbursement Obligations to the extent such Lender has acquired a participation therein under the terms of this Agreement and has not failed to perform its obligations in respect of such participation.

"Resolution Authority" means an EEA Resolution Authority or, with respect to any UK Financial Institution, a UK Resolution Authority.

"Revolving Commitment" means with respect to each Lender, the commitment, if any, of such Lender to make Revolving Loans and to acquire participations in Facility Letters of Credit and Swingline Loans hereunder, as such commitment may be changed from time to time pursuant to this Agreement (including, without limitation, in connection with any increases in accordance with Section 2.2). The amount of each Lender's Revolving Commitment as of Agreement Execution Date is set forth on Exhibit A. The aggregate amount of the Revolving Commitments is \$750,000,000 as of the Agreement Execution Date (subject to the Borrower's right to increase the Revolving Commitment in accordance with Section 2.2).

"Revolving Exposure" means with respect to any Lender at any time, the sum of the outstanding principal amount of such Lender's Revolving Loans and its LC Exposure and Swingline Exposure at such time.

"Revolving Lender" means a Lender with a Revolving Commitment or with Revolving Exposure.

"Revolving Loan" means a Loan made pursuant to Section 2.1 (and includes, as the context may require, a Bid Rate Loan).

"Rollover Projects" means those Projects which, due to no or low occupancy at such Project, have a value, determined by dividing the Property Operating Income for such a Project for the most recent four fiscal quarters by the Applicable Cap Rate, of less than 50% of book value, provided that a Project shall no longer be treated as a Rollover Project after: (i) a period of six consecutive full fiscal quarters has elapsed since such Project was first included as a Rollover Project, or (ii) such Project has a value, determined by dividing the Property Operating Income for such Project for the most recent four fiscal quarters by the Applicable Cap Rate, of greater than 50% of book value.

"S&P" means Standard & Poor's Ratings Group, a division of the McGraw Hill Companies and its successors.

"Sanctioned Person" means, at any time, (a) any Person that is, or is 50% or more owned by a Person or Persons that are, listed in any Sanctions related list of designated Persons maintained by any Governmental Authority of the United States of America, including without limitation, OFAC or the U.S. Department of State, or by the United Nations Security Council, Her Majesty's Treasury or the European Union or (b) a country, a region within a country, an agency of the government of a country, an organization controlled by a country or a Person resident in a country, that is subject to or the target of a sanctions program applicable to such country, region, agency, organization or Person and identified on any Sanctions-related list maintained by any Governmental Authority of the United States of America, including without limitation, OFAC or the U.S. Department of State.

"Sanctions" means economic or financial sanctions or trade embargoes imposed, administered or enforced by (a) any Governmental Authority of the United States of America, including without limitation, OFAC or the U.S. Department of State, or (b) the United Nations Security Council, Her Majesty's Treasury or the European Union.

"SOFR" means, with respect to any Business Day, a rate per annum equal to the secured overnight financing rate for such Business Day published by the SOFR Administrator on the SOFR Administrator's Website on the immediately succeeding Business Day.

"SOFR Administrator" means the Federal Reserve Bank of New York (or a successor administrator of the secured overnight financing rate).

"SOFR Administrator's Website" means the website of the Federal Reserve Bank of New York, currently at http://www.newyorkfed.org, or any successor source for the secured overnight financing rate identified as such by the SOFR Administrator from time to time.

"Specified Rate Management Obligations" means all indebtedness, liabilities, obligations, covenants and duties of the Borrower or its Subsidiaries under or in respect of any Specified Rate Management Transaction, whether direct or indirect, absolute or contingent, due or not due, liquidated or unliquidated, and whether or not evidenced by any written confirmation.

"Specified Rate Management Provider" means any Lender, or any Affiliate of a Lender that is a party to a Specified Rate Management Transaction at the time the Specified Rate Management Transaction is entered into.

"Specified Rate Management Transaction" means any Rate Management Transaction that is made or entered into at any time, or in effect at any time now or hereafter, whether as a result of an assignment or transfer or otherwise, between the Borrower or its Subsidiaries and any Specified Rate Management Provider.

"Subsidiary" means as to any Person, a corporation, partnership or other entity of which shares of stock or other ownership interests having ordinary voting power (other than stock or such other ownership interests having such power only by reason of the happening of a contingency) to elect a majority of the board of directors or other managers of such corporation, partnership or other entity are at the time owned, or the management of which is otherwise controlled, directly or indirectly through one or more intermediaries, or both, by such Person, and provided such corporation, partnership or other entity is consolidated with such Person for financial reporting purposes under GAAP.

"<u>Substantially Completed Properties</u>" means, for any Reference Year, Properties of the Borrower and its Subsidiaries which were previously Assets Under Development but (a) shall have been identified by the Borrower as being substantially completed (as determined on a basis consistent with the

identification of substantially completed Properties set forth on the Borrower's quarterly "Supplemental Information" report available on the Borrower's website for the fiscal period ending March 31, 2021) in the Borrower's quarterly "Supplemental Information" report made available on the Borrower's website) for the final quarterly period of such Reference Year, and (b) which (i) have been completed during such Reference Year and (ii) have been included in a "new opening summary" form submitted to, and approved by, the Administrative Agent for each newly-opened Property during such Reference Year. Notwithstanding the foregoing, "new opening summary" forms will only be required for developments to the extent the aggregate investment amount thereof exceeds \$600,000,000 during any fiscal year.

"Substitute Rate" means a floating rate of interest equal to (a) the Federal Funds Effective Rate from time to time <u>plus</u> one and one-half percent (1.50%) plus (b) the Applicable Margin for Adjusted Base Rate Loans. If the Substitute Rate determined as provided above would be less than 1.0%, the Substitute Rate shall be deemed to be 1.0%.

"Sustainability Pricing Adjustment Date" has the meaning specified in Section 1.7(a).

"Sustainability Margin Adjustment" means with respect to any Pricing Certificate for any period between Sustainability Pricing Adjustment Dates, an amount (whether negative or zero), expressed as a percentage, equal to the KPI Metric Applicable Margin Adjustment Amount.

"Swingline Borrowings" means, as of any date, collectively, all Swingline Loans then outstanding under this Facility.

"Swingline Exposure" means the outstanding principal balance of all Swingline Borrowings.

"Swingline Lender" shall mean Wells in its capacity as a Lender.

"Swingline Loan" means a Loan made by the Swingline Lender under the special availability provisions described in Section 2.17 hereof.

"Swingline Revolving Commitment" means the obligation of the Swingline Lender to make Swingline Loans not exceeding the aggregate amount of the lesser of (x) \$100,000,000 and (y) the unused commitment of the Swingline Lender in its capacity as a Lender in respect of the Facility.

"<u>Taxes</u>" means all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

"<u>Term SOFR</u>" means, for the applicable Corresponding Tenor as of the applicable Reference Time, the forward-looking term rate based on SOFR that has been selected or recommended by the Relevant Governmental Body.

"<u>Term SOFR Notice</u>" means a notification by the Administrative Agent to the Lenders and the Borrower of the occurrence of a Term SOFR Transition Event.

"<u>Term SOFR Transition Event</u>" means the determination by the Administrative Agent that (a) Term SOFR has been recommended for use by the Relevant Governmental Body, (b) the administration of Term SOFR is administratively feasible for the Administrative Agent and (c) a Benchmark Transition Event, an Early Opt-in Election or an Other Benchmark Rate Election, as applicable, has previously occurred resulting in the replacement of the then-current Benchmark for all purposes hereunder and under

any Loan Document in accordance with $\underline{Section\ 4.8(c)}$ with a Benchmark Replacement the Unadjusted Benchmark Replacement component of which is not Term SOFR.

"Total Revolving Exposure" means at any time, the sum of the aggregate Revolving Exposures for each of the Revolving Lenders.

"Transferee" is defined in Section 13.7 hereof.

"TRS" means a taxable REIT Subsidiary of the General Partner.

"Type" when used in reference to any Loan or Borrowing, refers to the rate by reference to which interest on such Loan, or on the Loans comprising such Borrowing, is determined. For purposes hereof, "rate" shall include the Adjusted LIBOR Rate and Adjusted Base Rate.

"<u>UK Financial Institution</u>" means any BRRD Undertaking (as such term is defined under the PRA Rulebook (as amended from time to time) promulgated by the United Kingdom Prudential Regulation Authority) or any person falling within IFPRU 11.6 of the FCA Handbook (as amended from time to time) promulgated by the United Kingdom Financial Conduct Authority, which includes certain credit institutions and investment firms, and certain affiliates of such credit institutions or investment firms.

"<u>UK Resolution Authority</u>" means the Bank of England or any other public administrative authority having responsibility for the resolution of any UK Financial Institution.

"<u>Unadjusted Benchmark Replacement</u>" means the applicable Benchmark Replacement excluding the related Benchmark Replacement Adjustment.

"Unencumbered Asset" means any Project which as of any date of determination, (a) is not subject to any Liens other than Permitted Liens set forth in Sections 9.5(i) through 9.5(v), (b) is not subject to any agreement (including any agreement governing Indebtedness incurred in order to finance or refinance the acquisition of such asset) which prohibits or limits the ability of the Borrower, or its Wholly-Owned Subsidiaries, as the case may be, to create, incur, assume or suffer to exist any Lien upon any assets or Capital Stock of the Borrower, or any of its Wholly-Owned Subsidiaries other than a Permitted Negative Pledge, (c) is not subject to any agreement (including any agreement governing Indebtedness incurred in order to finance or refinance the acquisition of such asset) which entitles any Person to the benefit of any Lien (but not subject to any Liens other than Permitted Liens set forth in Sections 9.5(i) through 9.5(v)) on any assets or Capital Stock of the Borrower or any of its Wholly-Owned Subsidiaries or would entitle any Person to the benefit of any Lien (but excluding the Permitted Liens set forth in Sections 9.5(i) through 9.5(v)) on such assets or Capital Stock upon the occurrence of any contingency (including, except in the case of a Material Credit Facility, pursuant to an "equal and ratable" clause), (d) is not the subject of any material architectural/engineering issue, as evidenced by a certification of the Borrower, and (e) is materially compliant with the representations and warranties in Article VI below. Notwithstanding the foregoing, if any Project is a "Superfund" site under federal law or a site identified in writing by the jurisdiction in which such Project is located as having significant environmental contamination under applicable state law, the Borrower shall so advise the Lenders in writing and the Required Lenders shall have the right to request from the Borrower a current detailed environmental assessment (or one which is not more than two years old for Unencumbered Assets owned as of the Agreement Execution Date), and, if applicable, a written estimate of any remediation costs from a recognized environmental contractor and to exclude any such Project from Unencumbered Assets at their election. No Project of a Wholly-Owned Subsidiary shall be deemed to be unencumbered unless such Project and all Capital Stock of such Wholly-Owned Subsidiary or any other intervening WhollyOwned Subsidiary between the Borrower and such Wholly-Owned Subsidiary is unencumbered and neither such Wholly-Owned Subsidiary nor any other intervening Wholly-Owned Subsidiary between the Borrower and such Wholly-Owned Subsidiary has any Indebtedness for borrowed money (other than Indebtedness due to the Borrower).

"Unencumbered Leverage Ratio" is defined in Section 9.7(c).

"Unencumbered Leverage Ratio Increase Period" is defined in Section 9.7(c).

"<u>Unimproved Land</u>" means land which constitutes a single tax parcel or separately platted lot and on which construction of an industrial building has not commenced.

"USD LIBOR" means the London interbank offered rate for Dollars.

"Value of Unencumbered Assets" means, for any Person as of any date, the sum (without duplication) of (a) the value of all Unencumbered Assets that are not Assets Under Development, Assets Acquired Not in Service or assets of the type described in clause (b) (determined in the manner set forth below), plus (b) the purchase price paid by such Person (less any amounts paid to such Person as a purchase price adjustment, held in escrow, retained as a contingency reserve or in connection with other similar arrangements) for any Property (other than Assets Under Development) that constitutes an Unencumbered Asset and acquired by such Person during the immediately preceding period of four consecutive fiscal quarters, plus (c) any unrestricted cash, including any cash on deposit with a qualified intermediary with respect to a deferred tax-free exchange, plus (d) an amount equal to 100% of the then-current book value, determined in accordance with GAAP, of each first mortgage receivable secured by an income producing commercial property, provided that such first mortgage receivable is not subject to any Lien, plus (e) 100% of the then current book value of each Asset Under Development that constitutes an Unencumbered Asset plus (f) with respect to each Rollover Project, an amount equal to 50% of the then current book value, determined in accordance with GAAP, of each Rollover Project; provided that to the extent the aggregate amount of Value of Unencumbered Assets from Assets Acquired Not in Service and the other items set forth in clauses (d), (e) and (f) exceed 20% of the total Value of Unencumbered Assets, such excess shall be excluded. Unencumbered Assets that are not Assets Under Development or assets of the type described in clause (b) above shall be valued by dividing the Property Operating Income for such Project for the most recent four fiscal quarters by the Applicable Cap Rate (provided that for the purpose of such calculation, the Property Operating Income of each Unencumbered Asset that was formerly a Rollover Project shall in no event be less than zero). If a Project is no longer owned as of the date of calculation, then no value shall be included based on capitalizing Property Operating Income from such Project, except for purposes of such financial covenant comparing the Property Operating Income from Unencumbered Assets during a quarter to Debt Service for such quarter.

"Wells" means Wells Fargo Bank, National Association.

"WFS" means Wells Fargo Securities, LLC.

"Wholly-Owned Subsidiary" means a member of the Consolidated Operating Partnership 100% of the ownership interests in which are owned, directly or indirectly, by the Borrower and the General Partner in the aggregate.

"Write-Down and Conversion Powers" means (a) with respect to any EEA Resolution Authority, the write-down and conversion powers of such EEA Resolution Authority from time to time under the Bail-In Legislation for the applicable EEA Member Country, which write-down and conversion powers are described in the EU Bail-In Legislation Schedule and (b) with respect to the United Kingdom, any

powers of the applicable Resolution Authority under the Bail-In Legislation to cancel, reduce, modify or change the form of a liability of any UK Financial Institution or any contract or instrument under which that liability arises, to convert all or part of that liability into shares, securities or obligations of that person or any other person, to provide that any such contract or instrument is to have effect as if a right had been exercised under it or to suspend any obligation in respect of that liability or any of the powers under that Bail-In Legislation that are related to or ancillary to any of those powers.

The foregoing definitions shall be equally applicable to both the singular and the plural forms of the defined terms.

- 1.2. Financial Standards. All financial computations required of a Person under this Agreement shall be made in accordance with GAAP as in effect on the date of this Agreement, and all financial information required under this Agreement shall be prepared in accordance with GAAP, except that if any Person's financial statements are not audited, such Person's financial statements shall be prepared in accordance with the same sound accounting principles utilized in connection with the financial information submitted to Lenders with respect to the Borrower or the General Partner or the Properties in connection with this Agreement and shall be certified by an authorized representative of such Person. Moreover, all financial computations required of a Person under this Agreement shall be calculated (i) without giving effect to any election under Accounting Standards Codification 825-10-25 (or any other Accounting Standards Codification or Financial Accounting Standard having a similar result or effect) to value any Indebtedness or other liabilities of the Borrower, the General Partner or any Wholly Owned Subsidiary at "fair value", as defined therein and (ii) without giving effect to any treatment of Indebtedness in respect of convertible debt instruments under Accounting Standards Codification 470-20 (or any other Accounting Standards Codification or Financial Accounting Standard having a similar result or effect) to value any such Indebtedness in a reduced or bifurcated manner as described therein, and such Indebtedness shall at all times be valued at the full stated principal amount thereof. To the extent the Consolidated Operating Partnership has Defeased Debt, both the underlying debt and interest payable thereon and the financial assets used to defease such debt and interest earned thereon shall be excluded from calculations of the foregoing financial covenants.
- 1.3. <u>Classification of Loans and Borrowings</u>. For purposes of this Agreement, Loans may be classified and referred to by Class (e.g., a "Revolving Loan") or by Type (e.g., a "Eurocurrency Revolving Loan"). Borrowings also may be classified and referred to by Class (e.g. a "Revolving Borrowing") or by Type (e.g., a "Eurocurrency Borrowing") or by Class and Type (e.g., a "Eurocurrency Revolving Borrowing").
- 1.4. <u>General References</u>. References in this Agreement to "Sections", "Articles", "Exhibits" and "Schedules" are to Sections, articles, exhibits and schedules herein and hereto unless otherwise indicated. References in this Agreement to any document, instrument or agreement (a) shall include all exhibits, schedules and other attachments thereto, (b) shall include all documents, instruments or agreements issued or executed in replacement thereof, to the extent permitted hereby and (c) shall mean such document, instrument or agreement, or replacement or predecessor thereto, as amended, supplemented, restated or otherwise modified from time to time to the extent not otherwise stated herein or prohibited hereby and in effect at any given time. Wherever from the context it appears appropriate, each term stated in either the singular or plural shall include the singular and plural, and pronouns stated in the masculine, feminine or neuter gender shall include the masculine, the feminine and the neuter. Unless otherwise indicated, all references to time are references to Central Time.
- 1.5. <u>Rates</u>. The interest rate on LIBOR Borrowings and Adjusted Base Rate Borrowings may be determined by reference to LIBOR, which is derived from the London interbank offered rate. The London interbank offered rate is intended to represent the rate at which contributing banks may obtain

short-term borrowings from each other in the London interbank market. On March 5, 2021, ICE Benchmark Administration ("IBA"), the administrator of the London interbank offered rate, and the Financial Conduct Authority (the "FCA"), the regulatory supervisor of IBA, announced in public statements (the "Announcements") that the final publication or representativeness date for the London interbank offered rate for Dollars for: (a) 1-week and 2-month tenor settings will be December 31, 2021 and (b) overnight, 1-month, 3-month, 6-month and 12-month tenor settings will be June 30, 2023. No successor administrator for IBA was identified in such Announcements. As a result, it is possible that commencing immediately after such dates, the London interbank offered rate for such tenors may no longer be available or may no longer be deemed a representative reference rate upon which to determine the interest rate on LIBOR Borrowings or Adjusted Base Rate Borrowings. There is no assurance that the dates set forth in the Announcements will not change or that IBA or the FCA will not take further action that could impact the availability, composition or characteristics of any London interbank offered rate. Public and private sector industry initiatives have been and continue, as of the date hereof, to be underway to implement new or alternative reference rates to be used in place of the London interbank offered rate. In the event that the London interbank offered rate or any other then-current Benchmark is no longer available or in certain other circumstances set forth in Section 4.8(c), such Section 4.8(c) provides a mechanism for determining an alternative rate of interest. The Administrative Agent will notify the Borrower, pursuant to Section 4.8(c), of any change to the reference rate upon which the interest rate on LIBOR Borrowings and Adjusted Base Rate Borrowings is based. However, the Administrative Agent does not warrant or accept any responsibility for, and shall not have any liability with respect to, (i) the continuation of, administration of, submission of, calculation of or any other matter related to the London interbank offered rate or other rates in the definition of "LIBOR" or with respect to any alternative, successor, or replacement rate thereto (including any then-current Benchmark or any Benchmark Replacement), including whether the composition or characteristics of any such alternative, successor or replacement rate (including any Benchmark Replacement), as it may or may not be adjusted pursuant to Section 4.8(c), will be similar to, or produce the same value or economic equivalence of, LIBOR or any other Benchmark, or have the same volume or liquidity as did the London interbank offered rate or any other Benchmark prior to its discontinuance or unavailability, or (ii) the effect, implementation or composition of any Benchmark Replacement Conforming Changes. The Administrative Agent and its Affiliates or other related entities may engage in transactions that affect the calculation of a Benchmark, any alternative, successor or replacement rate (including any Benchmark Replacement) or any relevant adjustments thereto and such transactions may be adverse to the Borrower. The Administrative Agent may select information sources or services in its reasonable discretion to ascertain any Benchmark, any component definition thereof or rates referenced in the definition thereof, in each case pursuant to the terms of this Agreement, and shall have no liability to the Borrower, any Lender or any other person or entity for damages of any kind, including direct or indirect, special, punitive, incidental or consequential damages, costs, losses or expenses (whether in tort, contract or otherwise and whether at law or in equity), for any error or calculation of any such rate (or component thereof) provided by any such information source or service.

1.6. Amendment and Restatement of the Existing Credit Agreement. The parties to this Agreement agree that, upon (i) the execution and delivery by each of the parties hereto of this Agreement and (ii) satisfaction of the conditions set forth in Section 5.1, the terms and provisions of the Existing Credit Agreement shall be and hereby are amended, superseded and restated in their entirety by the terms and provisions of this Agreement. This Agreement is not intended to and shall not constitute a novation. All "Loans" made and "Obligations" incurred under the Existing Credit Agreement which are outstanding on the Agreement Execution Date shall continue as Obligations under (and shall be governed by the terms of) this Agreement and the other Loan Documents. Without limiting the foregoing, upon the effectiveness hereof: (a) all references in the "Loan Documents" (as defined in the Existing Credit Agreement) to the "Administrative Agent", the "Credit Agreement" and the "Loan Documents" shall be deemed to refer to the Administrative Agent, this Agreement and the Loan Documents, (b) all obligations

constituting "Obligations" with any Lender or any Affiliate of any Lender which are outstanding on the Agreement Execution Date shall continue as Obligations under this Agreement and the other Loan Documents, (c) the Administrative Agent shall make such reallocations, sales, assignments or other relevant actions in respect of each Lender's credit and loan exposure under the Existing Credit Agreement as are necessary in order that each such Lender's outstanding Revolving Loans hereunder reflect such Lender's pro rata share of the outstanding aggregate Revolving Loans on the Agreement Execution Date, (d) the Borrower hereby agrees to compensate each Lender for any and all losses, costs and expenses incurred by such Lender in connection with the sale and assignment of any Eurocurrency Loans (including the "Eurocurrency Loans" under the Existing Credit Agreement) and such reallocation described above, in each case on the terms and in the manner set forth in Section 4.4 hereof and (e) each Departing Lender's outstanding "Loans" under (and as defined in) the Existing Credit Agreement as of the date hereof shall be repaid in full in cash in immediately available funds (accompanied by any accrued and unpaid interest and fees thereon and any other amounts or liabilities owing to each Departing Lender under the Existing Credit Agreement), each Departing Lender's "Revolving Commitment" under the Existing Credit Agreement shall be terminated and be of no further force and effect, each Departing Lender shall not be a Lender for any purpose hereunder (provided that each Departing Lender shall retain its respective rights as a "Lender" under the Existing Credit Agreement to expense reimbursement and indemnification pursuant to, and in accordance with, the terms of the Existing Credit Agreement), and such Departing Lender shall be released from any obligation or liability under the Existing Credit Agreement. Without limiting the forgoing, the parties hereto (including, without limitation, each Departing Lender) hereby agree that the consent of any Departing Lender shall be limited to the acknowledgements and agreements set forth in this Section 1.6 and shall not be required as a condition to the effectiveness of any other amendments, restatements, supplements or modifications to the Existing Credit Agreement or the Loan Documents.

1.7. Sustainability Adjustments.

- (a) Following the date on which Borrower provides a Pricing Certificate in respect of the most recently ended Reference Year, the Applicable Margin shall be adjusted, as applicable, pursuant to the Sustainability Margin Adjustment as set forth in such Pricing Certificate. For purposes of the foregoing, (i) the Sustainability Margin Adjustment shall be determined as of the fifth Business Day following receipt by the Administrative Agent of a Pricing Certificate delivered pursuant to Section 1.7(f) based upon the KPI Metric for the applicable Reference Year set forth in such Pricing Certificate and the calculations of the Sustainability Margin Adjustment, therein (such day, the "Sustainability Pricing Adjustment Date") and (ii) each change in the Applicable Margin resulting from a Pricing Certificate (or the non-delivery or delivery of an incomplete Pricing Certificate) shall be effective during the period commencing on and including the applicable Sustainability Pricing Adjustment Date and ending on the date immediately preceding the next such Sustainability Pricing Adjustment Date (or, in the case of non-delivery of a Pricing Certificate, the last day such Pricing Certificate could have been delivered pursuant to the terms of Section 1.7(f)).
- (b) For the avoidance of doubt, only one Pricing Certificate may be delivered in respect of any Reference Year. It is further understood and agreed that the LIBOR Applicable Margin and the Base Rate Applicable Margin will never be reduced by more than 0.01% pursuant to the Sustainability Margin Adjustment during any calendar year. For the avoidance of doubt, any adjustment to the Applicable Margin by reference to the KPI Metric Target in any Reference Year shall not be cumulative year-over-year. Each applicable adjustment shall only apply until the date on which the next adjustment is due to take place.
- (c) It is hereby understood and agreed that if no such Pricing Certificate is delivered, or any Pricing Certificate shall be incomplete and fail to include the KPI Metric for the applicable

Reference Year, within the period set forth in Section 1.7(f), the Sustainability Margin Adjustment will be made to the Applicable Margin commencing on the last day such Pricing Certificate could have been delivered pursuant to the terms of Section 1.7(f).

(d) If (i)(A) the Borrower or any Lender becomes aware of any material inaccuracy in the Sustainability Margin Adjustment, the KPI Metric as reported in a Pricing Certificate (any such material inaccuracy, a "Pricing Certificate Inaccuracy.") and, in the case of any Lender, such Lender delivers, not later than ten (10) Business Days after obtaining knowledge thereof, a written notice to the Administrative Agent describing such Pricing Certificate Inaccuracy in reasonable detail (which description shall be shared with each Lender and the Borrower), or (B) the Borrower and the Lenders agree that there was a Pricing Certificate Inaccuracy at the time of delivery of a Pricing Certificate, and (ii) a proper calculation of the Sustainability Margin Adjustment, the KPI Metric would have resulted in no adjustment to the Applicable Margin for any period, the Borrower shall be obligated to pay to the Administrative Agent for the account of the applicable Lenders or the applicable Issuing Banks, as the case may be, promptly on demand by the Administrative Agent (or, after the occurrence of an actual or deemed entry of an order for relief with respect to the Borrower under any Debtor Relief Laws, automatically and without further action by the Administrative Agent, any Lender or any Issuing Bank), but in any event within ten (10) Business Days after the Borrower has received written notice of, or has agreed in writing that there was, a Pricing Certificate Inaccuracy, an amount equal to the excess of (1) the amount of interest and fees that should have been paid for such period over (2) the amount of interest and fees actually paid for such period.

It is understood and agreed that any Pricing Certificate Inaccuracy shall not constitute a Default or Event of Default or otherwise result in the failure of any condition precedent to any advance or the issuance of any Letter of Credit; provided, that, the Borrower complies with the terms of this Section 1.7(d) with respect to such Pricing Certificate Inaccuracy. Notwithstanding anything to the contrary herein, unless such amounts shall be due upon the occurrence of an actual or deemed entry of an order for relief with respect to the Borrower under any Debtor Relief Laws, (a) any additional amounts required to be paid pursuant to the immediate preceding paragraph shall not be due and payable until a written demand is made for such payment by the Administrative Agent in accordance with such paragraph, (b) any nonpayment of such additional amounts prior to or concurrently with such demand for payment by the Administrative Agent shall not constitute a Default (whether retroactively or otherwise) and (c) none of such additional amounts shall be deemed overdue prior to such a demand or shall accrue interest at the default rate pursuant to Section 2.14 prior to such a demand.

- (e) Each party hereto hereby agrees that the Administrative Agent shall have no responsibility for (or liability in respect of) reviewing, auditing or otherwise evaluating any calculation by the Borrower of any Sustainability Margin Adjustment (or any of the data or computations that are part of or related to any such calculation) set forth in any Pricing Certificate (and the Administrative Agent may rely conclusively on any such certificate, without further inquiry).
- (f) As soon as available and in any event within 90 days following the end of each calendar year (commencing with the calendar year ending December 31, 2021), the Borrower may deliver a Pricing Certificate to the Administrative Agent (and the Administrative Agent shall promptly provide a copy to each Lender) for the most recently-ended Reference Year; provided, that, for any Reference Year the Borrower may elect not to deliver a Pricing Certificate, and such election shall not constitute a Default or Event of Default (but such failure to so deliver a Pricing Certificate by the end of such 90-day period shall result in the Sustainability Margin Adjustment being applied as set forth in Section 1.7(c)). Subject to the provisions of Section 1.7(b), the Borrower may, at its election exercisable by delivering written notice to the Administrative Agent, adjust the Reference Year and the timing of delivery of the Pricing Certificate in a manner intended to maintain consistency with the foregoing.

1.8. <u>Divisions</u>. For all purposes under the Loan Documents, in connection with any division or plan of division under Delaware law (or any comparable event under a different jurisdiction's laws): (a) if any asset, right, obligation or liability of any Person becomes the asset, right, obligation or liability of a different Person, then it shall be deemed to have been transferred from the original Person to the subsequent Person, and (b) if any new Person comes into existence, such new Person shall be deemed to have been organized on the first date of its existence by the holders of its Equity Interests at such time.

ARTICLE II.

THE FACILITY

2.1. The Facility.

- (a) Subject to the terms and conditions of this Agreement and in reliance upon the representations and warranties of the Borrower and the General Partner contained herein, each Lender agrees, severally and not jointly, to make Revolving Loans in Dollars to the Borrower from time to time prior to the Maturity Date in an aggregate principal amount that will not result in (x) such Lender's Revolving Exposure exceeding such Lender's Revolving Commitment, (y) the aggregate principal amount of all outstanding Bid Rate Loans would exceed 50.0% of the aggregate amount of the Revolving Commitments at such time, or (z) the Total Revolving Exposure to exceed the Aggregate Revolving Commitment. The Borrowings may be ratable Adjusted Base Rate Borrowings, ratable Eurocurrency Borrowings or nonpro rata Swingline Loans or non-pro rata Bid Rate Loans. This Facility is a revolving credit facility and, subject to the provisions of this Agreement, the Borrower may request Borrowings from the Facility, repay such Borrowings and reborrow such Borrowings at any time prior to the Maturity Date.
- (b) The Facility created by this Agreement, and the Revolving Commitment of each Lender to lend hereunder, shall terminate on the Maturity Date, unless sooner terminated in accordance with the terms of this Agreement.
- (c) In no event shall the Aggregate Revolving Commitment exceed Seven Hundred Fifty Million Dollars (\$750,000,000) (subject to the Borrower's right to increase pursuant to Section 2.2 below).
- 2.2. <u>Increase in Revolving Commitment</u>. The Borrower shall have the right to request increases in the Aggregate Revolving Commitment by providing written notice to the Administrative Agent, which notice shall be irrevocable once given; provided, however, that after giving effect to any such increases the Aggregate Revolving Commitment shall not exceed One Billion Dollars (\$1,000,000,000). Each such increase in the Revolving Commitments must be an aggregate minimum amount of \$25,000,000 and integral multiples of \$5,000,000 in excess thereof. The Administrative Agent, in consultation with the Borrower, shall manage all aspects of the syndication of such increase in the Revolving Commitments, including decisions as to the selection of the existing Lenders and/or other banks, financial institutions and other institutions of the increase in the Revolving Commitments among such existing Lenders and/or other banks, financial institutions and other institutional lenders. No Lender shall be obligated in any way whatsoever to increase its Revolving Commitment or provide a new Revolving Commitment, and any new Lender becoming a party to this Agreement in connection with any such requested increase must be an Eligible Assignee. If a new Lender becomes a party to this Agreement, or if any existing Lender is increasing its Revolving Commitment, such Lender shall on the date it becomes a Lender hereunder (or in the case of an existing Lender, increases its Revolving Commitment) (and as a condition thereto)

purchase from the other Lenders its Percentage (determined with respect to the Lenders' respective Revolving Commitments and after giving effect to the increase of Revolving Commitments) of any outstanding Loans, by making available to the Administrative Agent for the account of such other Lenders, in same day funds, an amount equal to the sum of (A) the portion of the outstanding principal amount of such Loans to be purchased by such Lender, plus (B) the aggregate amount of payments previously made by the other Revolving Lenders under Section 3.6 that have not been repaid, plus (C) interest accrued and unpaid to and as of such date on such portion of the outstanding principal amount of such Loans. The Borrower shall pay to the Revolving Lenders amounts payable, if any, to such Revolving Lenders under Section 4.4 as a result of the prepayment of any such Loans. Effecting the increase of the Revolving Commitments under this Section is subject to the following conditions precedent: (w) each of the Administrative Agent, each Issuing Bank and the Swingline Lender shall have consented thereto (such consent not to be unreasonably withheld or delayed), (x) no Default or Event of Default shall be in existence on the effective date of such increase, (y) the representations and warranties made or deemed made by the Borrower or any other Loan Party in any Loan Document to which such Loan Party is a party shall be true and correct on the effective date of such increase except to the extent that such representations and warranties expressly relate solely to an earlier date (in which case such representations and warranties shall have been true and correct on and as of such earlier date) and except for changes in factual circumstances specifically and expressly permitted hereunder, and (z) the Administrative Agent shall have received each of the following, in form and substance satisfactory to the Administrative Agent: (i) if not previously delivered to the Administrative Agent, copies certified by a Qualified Officer of the Borrower of (A) all partnership or other necessary action taken by the Borrower to authorize such increase and (B) all corporate or other necessary action taken by Guarantor authorizing the guaranty of such increase; and (ii) an opinion of counsel to the Borrower and the Guarantor, and addressed to the Administrative Agent and the Lenders covering such matters as reasonably requested by the Administrative Agent; and (iii) if requested, new Notes executed by the Borrower, payable to any new Lenders and replacement Notes, if requested by any applicable Lender, executed by the Borrower, payable to any existing Lenders increasing their Revolving Commitments, in the amount of each such Lender's Revolving Commitment at the time of the effectiveness of the applicable increase in the aggregate amount of the Revolving Commitments. In connection with any increase in the aggregate amount of the Revolving Commitments pursuant to this Section 2.2 any Lender becoming a party hereto shall execute such documents and agreements as the Administrative Agent may reasonably request.

- 2.3. <u>Principal Payments</u>. Any outstanding Borrowings (other than Bid Rate Loans) and all other unpaid Obligations shall be paid in full by the Borrower on the Maturity Date. Each Bid Rate Loan shall be paid in full on the last day of the applicable Interest Period of such Bid Rate Loan.
- 2.4. <u>Requests for Revolving Borrowings; Responsibility for Revolving Borrowings</u>. Ratable Revolving Borrowings shall be made available to the Borrower by Administrative Agent in accordance with <u>Section 2.1(a)</u> and <u>Section 2.7</u> hereof. The obligation of each Lender to fund its Percentage of each ratable Revolving Borrowing shall be several and not joint.
- 2.5. Evidence of Credit Extensions. Any Lender may request that Loans (other than Bid Rate Loans) made by it be evidenced by a Note. In such event, the Borrower shall prepare, execute and deliver to such Lender a Note payable to the order of such Lender (or, if requested by such Lender, to such Lender and its registered assigns). Thereafter, the Loans evidenced by such Note and interest thereon shall at all times (including after assignment pursuant to Section 13.3) be represented by one or more Notes in such form payable to the order of the payee named therein (or, if such Note is a registered note, to such payee and its registered assigns). Each Lender may record Borrowings and principal payments thereof on the schedule attached to its Note or, at its option, in its records, and each Lender's record thereof shall be conclusive absent the Borrower furnishing to such Lender conclusive and irrefutable evidence of an error made by such Lender with respect to that Lender's records. Except in the case of a

Lender that has notified the Administrative Agent in writing that it elects not to receive a Bid Rate Note, the Bid Rate Loans made by a Lender to the Borrower shall, in addition to this Agreement, also be evidenced by a Bid Rate Note payable to the order of such Lender. Notwithstanding the foregoing, the failure to make, or an error in making, a notation with respect to any Borrowing shall not limit or otherwise affect the obligations of the Borrower hereunder or under the Notes to pay the amount actually owed by the Borrower to Lenders.

2.6. Loans and Borrowings.

- (a) Each Loan (other than a Swingline Loan or Bid Rate Loan) shall be made as part of a Borrowing consisting of Loans of the same Class and Type made by the Lenders ratably in accordance with their respective Revolving Commitments of the applicable Class. The failure of any Lender to make any Loan required to be made by it shall not relieve any other Lender of its obligations hereunder.
- (b) Subject to Section 4.8(c), each Borrowing shall be comprised entirely of Adjusted Base Rate Loans or Eurocurrency Loans as the Borrower may request in accordance herewith. Each Swingline Loan shall be an Adjusted Base Rate Loan under the Facility. Each Bid Rate Loan shall be deemed to be outstanding under the Facility. Each Lender at its option may make any Eurocurrency Loan by causing any domestic or foreign branch or Affiliate of such Lender to make such Loan; provided that any exercise of such option shall not affect the obligation of the Borrower to repay such Loan in accordance with the terms of this Agreement.
 - (c) No more than ten (10) Eurocurrency Borrowings may be outstanding at any one time under the Facility.
- (d) Notwithstanding any other provision of this Agreement, the Borrower shall not be entitled to request, or to elect to convert or continue, any Borrowing if the Interest Period requested with respect thereto would end after the Maturity Date.

2.7. Requests for Revolving Borrowings.

(a) To request a Revolving Borrowing, the Borrower shall notify the Administrative Agent of such request by telephone (a "Borrowing Notice") (A) in the case of a Eurocurrency Borrowing, not later than 11:00 a.m. (Central Time), three (3) Business Days before the date of the proposed Revolving Borrowing, or (B) in the case of an Adjusted Base Rate Revolving Borrowing, not later than 11:00 a.m. (Central Time), one (1) Business Day before the date of the proposed Revolving Borrowing; provided that (x) any such notice of an Adjusted Base Rate Revolving Borrowing Borrowing to finance the reimbursement of an LC Disbursement as contemplated by Section 3.5 may be given not later than 11:00 a.m. (Central Time), on the date of the proposed Revolving Borrowing and (y) any such notice of a Borrowing of Swingline Loans may be given not later than 11:00 a.m. (Central Time), on the date of such proposed Borrowing. Each such telephonic Borrowing Notice shall be irrevocable and, if requested by the Administrative Agent, shall be confirmed promptly by delivery to the Administrative Agent of a written Borrowing Notice in substantially the form of Exhibit J in the case of any Borrowing (other than a Swingline Loan) or Exhibit K in the case of a Swingline Loan. Each such telephonic and written Borrowing Notice shall specify the following information in compliance with Section 2.7: (i) the Borrower requesting such Revolving Borrowing; (ii) the Class and Type of the requested Revolving Borrowing; (iii) the aggregate amount of such Revolving Borrowing; (iv) the date of such Revolving Borrowing, which shall be a Business Day; (v) in the case of a Eurocurrency Borrowing, the initial Interest Period to be applicable thereto; and (vi) the location and number of

the Borrower's account to which funds are to be disbursed, which shall comply with the requirements of Section 2.7. If no election as to the Type of Revolving Borrowing is specified, then the requested Revolving Borrowing shall be an Adjusted Base Rate Borrowing. If no Interest Period is specified with respect to any requested Eurocurrency Borrowing, then the Borrower shall be deemed to have selected an Interest Period of one month's duration. Promptly following receipt of a Borrowing Notice in accordance with this Section, the Administrative Agent shall advise each relevant Lender of the details thereof and of the amount of such Lender's Loan to be made as part of the requested Revolving Borrowing.

The Borrower shall also deliver together with each Borrowing Notice the compliance certificate required in <u>Section 5.2</u> and otherwise comply with the conditions set forth in <u>Section 5.2</u> for Borrowings.

Not later than 11:00 a.m. (Central Time) on each Borrowing Date, each Lender shall make available its Loan or Loans, in funds immediately available at the Administrative Office. Administrative Agent will promptly make the funds so received from the Lenders available to the Borrower in the account specified by the Borrower in the Disbursement Instruction Agreement.

- (b) Administrative Agent shall, as soon as practicable after receipt of a Borrowing Notice, determine the Adjusted LIBOR Rate applicable to the requested ratable Eurocurrency Borrowing and inform the Borrower and Lenders of the same. Each determination of the Adjusted LIBOR Rate by Administrative Agent shall be conclusive and binding upon the Borrower in the absence of manifest error.
- (c) If the Borrower shall prepay a Eurocurrency Borrowing other than on the last day of the Interest Period applicable thereto, the Borrower shall be responsible to pay all amounts due to Lenders as required by <u>Section 4.4</u> hereof. The Lenders shall not be obligated to match fund their Eurocurrency Borrowings.
- (d) The right of the Borrower to select the Adjusted LIBOR Rate, the Adjusted Base Rate and the LIBOR Market Index Rate for a Borrowing pursuant to this Agreement is subject to the availability to Lenders or the Swingline Lender, as applicable, of a similar option. If Administrative Agent determines that (i) deposits of Dollars in an amount approximately equal to the Borrowing for which the Borrower wishes to select the Adjusted LIBOR Rate, Adjusted Base Rate or LIBOR Market Index Rate are not generally available at such time in the London interbank eurodollar market, or (ii) the rate at which the deposits described in subsection (i) herein are being offered will not adequately and fairly reflect the costs to Lenders or the Swingline Lender, as applicable, of maintaining an Adjusted LIBOR Rate, Adjusted Base Rate or LIBOR Market Index Rate on a Borrowing or of funding the same in such market for such Interest Period, or (iii) reasonable means do not exist for determining an Adjusted LIBOR Rate, Adjusted Base Rate or LIBOR Market Index Rate, or (iv) the Adjusted LIBOR Rate, Adjusted Base Rate or LIBOR Market Index Rate would be in excess of the maximum interest rate which the Borrower may by law pay, then in any of such events, Administrative Agent shall so notify the Borrower and Lenders and notwithstanding anything herein to the contrary, such Borrowing shall bear interest at the Substitute Rate.

2.8. Interest Elections.

(a) Each Borrowing initially shall be of the Type specified in the applicable Borrowing Request and, in the case of a Eurocurrency Borrowing, shall have an initial Interest Period as specified in such Borrowing Request (or as set forth in Section 2.7 if no Interest Period

is specified). Thereafter, the Borrower may elect to convert such Borrowing to a different Type or to continue such Borrowing and, in the case of a Eurocurrency Borrowing, may elect Interest Periods therefor, all as provided in this Section. The Borrower may elect different options with respect to different portions of the affected Borrowing, in which case each such portion shall be allocated ratably among the Lenders holding the Loans comprising such Borrowing, and the Loans comprising each such portion shall be considered a separate Borrowing. Notwithstanding the foregoing, the Borrower may not (i) elect an Interest Period for Eurocurrency Loans that does not comply with Section 2.6(d), (ii) elect to convert any Adjusted Base Rate Loans to Eurocurrency Loans that would result in the number of Eurocurrency Borrowings exceeding the maximum number of Eurocurrency Borrowings permitted under Section 2.6(c), (iii) elect an Interest Period for Eurocurrency Loans unless the aggregate outstanding principal amount of Eurocurrency Loans to which such Interest Period will apply complies with the requirements as to minimum principal amount set forth in Section 2.11, or (iv) elect to convert or continue any Swingline Loans.

- (b) To make an election pursuant to this Section (an "<u>Interest Election Request</u>"), the Borrower shall notify the Administrative Agent of such election by telephone by the time that a Borrowing Request would be required under <u>Section 2.7</u> if the Borrower were requesting a Revolving Borrowing of the Type resulting from such election to be made on the effective date of such election. Each such telephonic Interest Election Request shall be irrevocable and shall be confirmed promptly by delivery to the Administrative Agent of a written Interest Election Request in a form approved by the Administrative Agent (hereinafter referred to as a "<u>Conversion/Continuation Notice</u>").
- (c) Each telephonic and written Conversion/Continuation Notice shall specify the following information in compliance with Section 2.6 and paragraph (a) of this Section: (i) the Borrowing to which such Conversion/Continuation Notice applies; (ii) the effective date of the election made pursuant to such Conversion/Continuation Notice, which shall be a Business Day; (iii) whether the resulting Borrowing is to be an Adjusted Base Rate Borrowing or a Eurocurrency Borrowing; and (iv) if the resulting Borrowing is a Eurocurrency Borrowing, the Interest Period to be applicable thereto after giving effect to such election. If any such Conversion/Continuation Notice requests a Eurocurrency Borrowing but does not specify an Interest Period, then the Borrower shall be deemed to have selected an Interest Period of one month's duration.
- (d) Promptly following receipt of a Conversion/Continuation Notice, the Administrative Agent shall advise each relevant Lender of the details thereof and of such Lender's portion of each resulting Borrowing.
- (e) If the Borrower fails to timely deliver a Conversion/Continuation Notice with respect to a Eurocurrency Borrowing in accordance with Section 2.8(b) prior to the end of the Interest Period applicable thereto, then, unless such Borrowing is repaid as provided herein, at the end of such Interest Period such Borrowing shall be converted to an Adjusted Base Rate Borrowing. The Administrative Agent shall advise each relevant Lender and the Borrower of the details of any such conversion under this clause (e).
- (f) Notwithstanding anything to the contrary contained in this <u>Section 2.8</u>, no Borrowing may be converted into a Eurocurrency Borrowing or continued as a Eurocurrency Borrowing (except with the consent of the Required Lenders) when any Monetary Default or Event of Default has occurred and is continuing.

2.9. Applicable Margins/Facility Fee Rate.

(a) [Reserved].

(b) The Base Rate Applicable Margin and the LIBOR Applicable Margin shall vary from time to time in accordance with the Investment Grade Rating as follows (such that the Applicable Margin shall change from time to time as and when the Investment Grade Rating changes, which changes shall be effective from and after the date that notice is delivered from the Borrower to the Administrative Agent of the applicable Investment Grade Rating change), other than in the case of Bid Rate Loans:

Rating Level	LIBOR Applicable Margin	Facility Fee Rate	Base Rate Applicable Margin
A/A2 (" <u>Level I</u> ")	0.70%	0.10%	0.70%
A-/A3 (" <u>Level II</u> ")	0.725%	0.125%	0.725%
BBB+/Baa1 ("Level III")	0.775%	0.15%	0.775%
BBB/Baa2 (" <u>Level IV</u> ")	0.85%	0.20%	0.85%
BBB-/Baa3 (" <u>Level V</u> ")	1.05%	0.25%	1.05%
Below BBB- and Baa3 (" <u>Level VI</u> ")	1.40%	0.30%	1.40%

Notwithstanding the Investment Grade Rating set forth in the above table, if: (i) the Consolidated Leverage Ratio as of the last day of the most recently ending fiscal quarter of the Borrower as set forth in the corresponding compliance certificate delivered pursuant to Section 8.2(iv) is less than 32.5% (or, so long as the Consolidated Leverage Ratio shall have been less than 32.5% during the immediately preceding fiscal quarter, greater than or equal to 32.5% but less than 37.5%; provided that the provisions of this parenthetical shall be applicable for only one fiscal quarter during the term of this Agreement), and (ii) (a) the Borrower's Investment Grade Rating with S&P is not lower than BBB and (b) the Borrower's Investment Grade Rating with Moody's is not lower than Baa2, the Base Rate Applicable Margin, the LIBOR Applicable Margin and the Facility Fee Rate shall be set at Level III. For the avoidance of doubt, (x) if the Consolidated Leverage Ratio is greater than or equal to 32.5% for two or more consecutive fiscal quarters, the Borrower will again satisfy the condition set forth in clause (i) above when and if the Consolidated Leverage Ratio as of the last day of the most recently ending fiscal quarter of the Borrower as set forth in the corresponding compliance certificate delivered pursuant to Section 8.2(iv) is less than 32.5%, and (y) the Borrower may qualify for pricing at Level I, II or III based solely upon its Investment Grade Rating even if it does not meet the Consolidated Leverage Ratio condition described in clause (i) above. Any increase or decrease in the Base Rate Applicable Margin, the LIBOR Applicable Margin and the Facility Fee Rate resulting from a change in the Consolidated Leverage Ratio in accordance with the foregoing proviso shall become effective as of the first Business Day immediately following the date a compliance certificate is delivered pursuant to Section 8.2(iv); provided, that if a compliance certificate is not delivered when due in accordance with such Section, then the Level corresponding to the Investment Grade Rating then in effect shall apply as of the fifth Business Day after the date on which such compliance certificate was required to have been delivered and shall remain in effect until the date on which such compliance certificate is delivered. The parties acknowledge and agree the Consolidated Leverage Ratio for the first fiscal quarter of 2021 is less than 32.5%.

The Facility Fee Rate shall also vary from time to time (as and when the Investment Grade Ratings change) in accordance with the foregoing table. Moreover, the Applicable Margin and the Facility

Fee Rate shall be determined by the higher of the two ratings from S&P or Moody's. In the event that such two ratings are more than one rating level apart and both are Investment Grade Ratings, then the rating level one level above the lower of the two ratings shall apply. If only one Investment Grade Ratings has been issued, the Applicable Margin and Facility Fee Rate shall be determined based on the sole Investment Grade Rating then in effect. If Investment Grade Ratings shall have been assigned by both rating agencies and thereafter the Borrower does not have an Investment Grade Rating from either Rating Agency, the Applicable Margin and Facility Fee Rate shall be determined based on Level VI of the forgoing table in this Section 2.9(b).

2.10. Other Fees. During the period from the Agreement Execution Date to but excluding the Maturity Date, the Borrower agrees to pay to the Administrative Agent for the account of the Revolving Lenders, a facility fee equal to the daily amount of the Aggregate Revolving Commitment (whether or not utilized) times a rate per annum equal to the applicable facility fee rate in effect from time to time, as shown in Section 2.9(b), as applicable (the "Facility Fee Rate"). Such foregoing fees (pursuant to subsection (b)) shall be payable quarterly in arrears on the first day of each January, April, July and October during the term of this Agreement and on the Maturity Date or any earlier date of termination of the Revolving Commitments or reduction of the Revolving Commitments to zero. The Borrower acknowledges that any fee payable hereunder is a bona fide commitment fee and is intended as reasonable compensation to the Lenders for committing to make funds available to the Borrower as described herein and for no other purposes.

The Borrower agrees to pay to the Administrative Agent a fee equal to \$2,500 at the time of each Bid Rate Quote Request made hereunder for services rendered by the Administrative Agent in connection with the Bid Rate Loans. Any fees due under this <u>Section 2.10</u> shall be computed on the basis of a year of 360 days and the actual number of days elapsed.

2.11. <u>Minimum Amount of Each Borrowing</u>. Each Eurocurrency Borrowing shall be in the minimum amount of \$2,000,000 (and in multiples of \$100,000 if in excess thereof), and each Adjusted Base Rate Borrowing shall be in the minimum amount of \$1,000,000 (and in multiples of \$100,000 if in excess thereof), provided, however, that any Adjusted Base Rate Borrowing may be in the amount of the unused Aggregate Revolving Commitment.

2.12. Interest.

- (a) Subject to Section 2.14, the outstanding principal balance of the Loans shall bear interest from time to time at a rate per annum equal to:
 - (i) the Adjusted Base Rate; or
 - (ii) at the election of the Borrower in accordance with $\underline{\text{Section 2.8}}$ with respect to all or portions of the Obligations, the Adjusted LIBOR Rate; or
 - (iii) if such Loan is an Absolute Rate Loan, at the Absolute Rate for such Loan for the Interest Period therefor quoted by the Lender making such Loan in accordance with <u>Section 2.18</u>; or
 - (iv) if such Loan is a LIBOR Margin Loan, at the Base LIBOR Rate for such Loan for the Interest Period therefor plus the LIBOR Margin quoted by the Lender making such Loan in accordance with <u>Section 2.18</u>.

(b) All interest shall be calculated for actual days elapsed on the basis of a 360-day year. Interest accrued on each Borrowing (other than Bid Rate Loans) shall be payable on the first day of each calendar month in arrears from time to time while such Borrowing is outstanding (or, if earlier, on the Maturity Date). Interest shall not be payable for the day of any payment on the amount paid if payment is received by Administrative Agent prior to 1:00 p.m. (Central Time). If any payment of principal or interest on the Loans, or fees hereunder or under any other Loan Document, shall become due on a day that is not a Business Day, such payment shall be made on the next succeeding Business Day and, in the case of a payment of principal, such extension of time shall be included in computing interest due in connection with such payment; provided that for purposes of Section 10.1 hereof, any payments of principal described in this sentence shall be considered to be "due" on such next succeeding Business Day.

2.13. Method of Payment.

- (a) All payments of the Obligations hereunder shall be made, without set off, deduction, or counterclaim, in immediately available funds to Administrative Agent at the Administrative Office or at any other Lending Office of Administrative Agent specified in writing by Administrative Agent to the Borrower, by 11:00 a.m. (Central Time) on the date when due and shall be applied ratably by Administrative Agent among Lenders in the Facility. Each such payment shall be made in Dollars. Each payment delivered to Administrative Agent for the account of any Lender shall be delivered promptly by Administrative Agent to such Lender in the same type of funds that Administrative Agent received at its address specified herein or at any Lending Office specified in a notice received by Administrative Agent from such Lender. Payments not made by Administrative Agent within one (1) Business Day after receipt shall accrue interest at Federal Funds Effective Rate. Administrative Agent is hereby authorized to charge the account of the Borrower maintained with Wells for each payment of principal, interest and fees as it becomes due hereunder.
- (b) If any Lender shall fail to make any payment required to be made by it pursuant to Sections 2.7, 2.16, 2.17, 3.1, 3.5 or 12.8, then the Administrative Agent shall, notwithstanding any contrary provision hereof, (i) apply any amounts thereafter received by the Administrative Agent for the account of such Lender for the benefit of the Administrative Agent, the Swingline Lender or the Issuing Banks to satisfy such Lender's obligations to it under such Section until all such unsatisfied obligations are fully paid, and/or (ii) hold any such amounts in a segregated account as cash collateral for, and application to, any future funding obligations of such Lender under any such Section, in the case of each of clauses (i) and (ii) above, in any order as determined by the Administrative Agent in its discretion.
- 2.14. <u>Default</u>. Notwithstanding the foregoing, during the continuance of a Monetary Default or an Event of Default, the Borrower shall not have the right to request a Eurocurrency Borrowing, request a Bid Rate Loan, select a new Interest Period for an existing ratable Eurocurrency Borrowing or convert any Adjusted Base Rate Borrowing to a ratable Eurocurrency Borrowing. During the continuance of a Monetary Default or an Event of Default, at the election of the Required Lenders, by notice to the Borrower, outstanding Borrowings shall bear interest at the applicable Default Rates until such Monetary Default or Event of Default ceases to exist or the Obligations are paid in full.
- 2.15. <u>Lending Offices</u>. Each Lender may book its Borrowings at any Lending Office selected by such Lender and may change its Lending Office from time to time. All terms of this Agreement shall apply to any such Lending Office and any Note shall be deemed held by the applicable Lender for the benefit of such Lending Office. Each Lender may, by written or telex notice to the Administrative Agent

and the Borrower, designate a Lending Office through which Borrowings will be made by it and for whose account payments are to be made.

- 2.16. Non Receipt of Funds by Administrative Agent. Unless the Borrower or a Lender, as the case may be, notifies Administrative Agent prior to the date on which it is scheduled to make payment to Administrative Agent of (i) in the case of a Lender, a Borrowing, or (ii) in the case of the Borrower, a payment of principal, interest or fees to the Administrative Agent for the account of the Lenders, that it does not intend to make such payment, Administrative Agent may assume that such payment has been made. Administrative Agent may, but shall not be obligated to, make the amount of such payment available to the intended recipient in reliance upon such assumption. If such Lender or the Borrower, as the case may be, has not in fact made such payment to Administrative Agent, the recipient of such payment shall, promptly after demand by Administrative Agent, repay to Administrative Agent the amount so made available together with interest thereon in respect of each day during the period commencing on the date such amount was so made available by Administrative Agent until the date Administrative Agent recovers such amount at a rate per annum equal to (i) in the case of payment by a Lender, the Federal Funds Effective Rate (as determined by Administrative Agent) or (ii) in the case of payment by the Borrower, the interest rate applicable to Adjusted Base Rate Borrowings. If the Borrower and any Lender shall each pay any such amount made available by the Administrative Agent on behalf of such Lender for the same or overlapping period, the Administrative Agent shall promptly remit to the Borrower the amount paid by the Borrower for such period. Any payment by the Borrower shall be without prejudice to any claim the Borrower may have against a Lender that shall have failed to make available the proceeds of a Loan to be made by such Lender.
- 2.17. Swingline Loans. In addition to the other options available to the Borrower hereunder, the Swingline Lender agrees, subject to the following terms and conditions, to make Swingline Loans in Dollars to the Borrower from time to time in an aggregate principal amount not to exceed the Swingline Revolving Commitment. Swingline Loans shall be made available for same day borrowings provided that notice is given in accordance with Section 2.7 hereof. All Swingline Loans shall bear interest at the LIBOR Market Index Rate and shall be deemed to be Adjusted Base Rate Borrowings. Swingline Loans shall be funded by Wells in an amount not to exceed the maximum amount it is required to disburse pursuant to the next sentence. In no event shall the Swingline Lender be required to fund a Swingline Loan if it would increase the total aggregate outstanding Swingline Loans to an amount in excess of the Swingline Revolving Commitment or if, after giving effect thereto, the Total Revolving Exposure would exceed the Aggregate Revolving Commitment. Upon request of the Swingline Lender made to all the Revolving Lenders, each Revolving Lender irrevocably agrees to purchase its Percentage of any Swingline Loan made by the Swingline Lender regardless of whether the conditions for disbursement are satisfied at the time of such purchase, including the existence of an Event of Default hereunder provided that such Event of Default did not exist at the time the Swingline Loan was made and provided further that no Lender shall be required to have its Revolving Exposure to be greater than its Revolving Commitment. Such purchase shall take place on the date of the request by Swingline Lender so long as such request is made by 11:00 a.m. (Central Time), otherwise on the Business Day following such request. All requests for purchase shall be in writing. From and after the date it is so purchased, each such Swingline Loan shall, to the extent purchased, (i) be treated as a Loan made by the purchasing Lenders and not by the selling Lender for all purposes under this Agreement and the payment of the purchase price by a Lender shall be deemed to be the making of a Loan by such Lender and shall constitute outstanding principal hereunder and under such Lender's Note, and (ii) shall no longer be considered a Swingline Loan except that all interest accruing on or attributable to such Swingline Loan for the period prior to the date of such purchase shall be paid when due by the Borrower to the Administrative Agent for the benefit of the Swingline Lender and all such amounts accruing on or attributable to such Loans for the period from and after the date of such purchase shall be paid when due by the Borrower to the Administrative Agent for the benefit of the purchasing Lenders. If prior to

purchasing its Percentage of a Swingline Loan one of the events described in <u>Section 10.10</u> shall have occurred and such event prevents the consummation of the purchase contemplated by preceding provisions, each Revolving Lender will purchase an undivided participating interest in the outstanding Swingline Loan in an amount equal to its Percentage of such Swingline Loan. From and after the date of each Lender's purchase of its participating interest in a Swingline Loan, if the Swingline Lender receive any payment on account thereof, the Swingline Lender will distribute to such Lender its participating interest in such amount (appropriately adjusted, in the case of interest payments, to reflect the period of time during which such Lender's participating interest was outstanding and funded); provided, however, that in the event that such payment was received by the Swingline Lender and is required to be returned to the Borrower, each Lender will return to the Swingline Lender any portion thereof previously distributed by the Swingline Lender to it. If any Revolving Lender fails to so purchase its Percentage of any Swingline Loan, such Lender shall be deemed to be a Defaulting Lender hereunder. No Swingline Loan shall be outstanding for more than five (5) days at a time (or such shorter period ending on the date any Revolving Loan shall be made subsequent to the making of such Swingline Loan) and Swingline Loans shall not be outstanding for more than a total of fifteen (15) days during any month.

2.18. Bid Rate Loans.

- (a) <u>Bid Rate Loans</u>. At any time during the period from the Closing Date to but excluding the Maturity Date, and so long as the Borrower continues to maintain an Investment Grade Rating from S&P or Moody's, the Borrower may, as set forth in this Section, request the Revolving Lenders to make offers to make Bid Rate Loans to the Borrower in Dollars. The Revolving Lenders may, but shall have no obligation to, make such offers and the Borrower may, but shall have no obligation to, accept any such offers in the manner set forth in this Section.
- (b) Requests for Bid Rate Loans. When the Borrower wishes to request from the Revolving Lenders offers to make Bid Rate Loans, it shall give the Administrative Agent notice (a "Bid Rate Quote Request") so as to be received no later than 11:00 a.m. (Central Time) on (x) the date two (2) Business Days prior to the proposed date of borrowing, in the case of an Absolute Rate Auction and (y) the date four (4) Business Days prior to the proposed date of borrowing, in the case of a LIBOR Auction. The Administrative Agent shall deliver to each Revolving Lender a copy of each Bid Rate Quote Request promptly upon receipt thereof by the Administrative Agent. The Borrower may request offers to make Bid Rate Loans for up to 3 different Interest Periods in any one Bid Rate Quote Request; provided that if granted each separate Interest Period shall be deemed to be a separate borrowing (a "Bid Rate Borrowing"). Each Bid Rate Quote Request shall be substantially in the form of Exhibit M and shall specify as to each Bid Rate Borrowing all of the following:
 - (i) the proposed date of such Bid Rate Borrowing, which shall be a Business Day;
 - (ii) the aggregate amount of such Bid Rate Borrowing which shall be in a minimum amount of \$10,000,000 and integral multiples of \$1,000,000 in excess thereof which shall not cause any of the limits specified in <u>Section 2.1(a)</u>, to be violated;
 - (iii) whether the Bid Rate Quote Request is for LIBOR Margin Loans or Absolute Rate Loans; and
 - (iv) the duration of the Interest Period applicable thereto, which shall not extend beyond the Maturity Date.

The Borrower shall not deliver any Bid Rate Quote Request within five (5) Business Days of the giving of any other Bid Rate Quote Request and the Borrower shall not deliver more than two (2) Bid Rate Quote Requests in any calendar month.

(c) Bid Rate Quotes.

- (i) Each Revolving Lender may submit one or more Bid Rate Quotes, each containing an offer to make a Bid Rate Loan in response to any Bid Rate Quote Request; provided that, if the Borrower's request under Section 2.18(b) specified more than one Interest Period, such Revolving Lender may make a single submission containing only one Bid Rate Quote for each such Interest Period. Each Bid Rate Quote must be submitted to the Administrative Agent not later than 10:00 a.m. (Central Time) (x) on the Business Day immediately preceding the proposed date of borrowing, in the case of an Absolute Rate Auction and (y) on the date three (3) Business Days prior to the proposed date of borrowing, in the case of a LIBOR Auction, and in either case the Administrative Agent shall disregard any Bid Rate Quote received after such time; provided that the Revolving Lender then acting as the Administrative Agent may submit a Bid Rate Quote only if it notifies the Borrower of the terms of the offer contained therein not later than 30 minutes prior to the latest time by which the Revolving Lenders must submit applicable Bid Rate Quotes. Any Bid Rate Quote so made shall be irrevocable except with the consent of the Administrative Agent given at the request of the Borrower. Such Bid Rate Loans may be funded by a Revolving Lender's Designated Lender (if any) as provided in Section 13.5; however, such Revolving Lender shall not be required to specify in its Bid Rate Quote whether such Bid Rate Loan will be funded by such Designated Lender.
 - (ii) Each Bid Rate Quote shall be substantially in the form of Exhibit N and shall specify:
 - (A) the proposed date of borrowing and the Interest Period therefor;
 - (B) the principal amount of the Bid Rate Loan for which each such offer is being made; provided that the aggregate principal amount of all Bid Rate Loans for which a Revolving Lender submits Bid Rate Quotes (x) may be greater or less than the Revolving Commitment of such Revolving Lender but (y) shall not exceed the principal amount of the Bid Rate Borrowing for a particular Interest Period for which offers were requested; provided further that any Bid Rate Quote shall be in a minimum amount of \$5,000,000 and integral multiples of \$1,000,000 in excess thereof;
 - (C) in the case of an Absolute Rate Auction, the rate of interest per annum (rounded upwards, if necessary, to the nearest one-hundredth of one percent (0.01%)) offered for each such Absolute Rate Loan (the "Absolute Rate");
 - (D) in the case of a LIBOR Auction, the margin above or below applicable Base LIBOR Rate (the "<u>LIBOR Margin</u>") offered for each such LIBOR Margin Loan, expressed as a percentage (rounded upwards, if necessary, to the nearest one-hundredth of one percent (0.01%)) to be added to (or subtracted from) the applicable Base LIBOR Rate; and
 - (E) the identity of the quoting Revolving Lender.

Unless otherwise agreed by the Administrative Agent and the Borrower, no Bid Rate Quote shall contain qualifying, conditional or similar language or propose terms other than or in addition to those set forth in the applicable Bid Rate Quote Request and, in particular, no Bid Rate Quote may be conditioned upon acceptance by the Borrower of all (or some specified minimum) of the principal amount of the Bid Rate Loan for which such Bid Rate Quote is being made.

(d) Notification by Administrative Agent. The Administrative Agent shall, as promptly as practicable after the Bid Rate Quotes are submitted (but in any event not later than 11:00 a.m. (Central Time) (x) on the Business Day immediately preceding the proposed date of borrowing, in the case of an Absolute Rate Auction or (y) on the date three (3) Business Days prior to the proposed date of borrowing, in the case of a LIBOR Auction), notify the Borrower of the terms (i) of any Bid Rate Quote submitted by a Revolving Lender that is in accordance with Section 2.18(c) and (ii) of any Bid Rate Quote that amends, modifies or is otherwise inconsistent with a previous Bid Rate Quote submitted by such Revolving Lender with respect to the same Bid Rate Quote Request. Any such subsequent Bid Rate Quote shall be disregarded by the Administrative Agent unless such subsequent Bid Rate Quote is submitted solely to correct a manifest error in such former Bid Rate Quote. The Administrative Agent's notice to the Borrower shall specify (A) the aggregate principal amount of the Bid Rate Borrowing for which offers have been received and (B) the principal amounts and Absolute Rates or LIBOR Margins, as applicable, so offered by each Revolving Lender (identifying the Revolving Lender that made such Bid Rate Quote).

(e) Acceptance by Borrower.

- (i) Not later than 12:00 p.m. (Central Time) (x) on the Business Day immediately preceding the proposed date of borrowing, in the case of an Absolute Rate Auction and (y) on the date three (3) Business Days prior to the proposed date of borrowing, in the case of a LIBOR Auction, the Borrower shall notify the Administrative Agent of its acceptance or nonacceptance of the Bid Rate Quotes so notified to it pursuant to Section 2.18(d) which notice shall be in the form of Exhibit O. In the case of acceptance, such notice shall specify the aggregate principal amount of Bid Rate Quotes for each Interest Period that are accepted. The failure of the Borrower to give such notice by such time shall constitute nonacceptance. The Borrower may accept any Bid Rate Quote in whole or in part; provided that:
 - (A) the aggregate principal amount of each Bid Rate Borrowing may not exceed the applicable amount set forth in the related Bid Rate Quote Request, but may be less;
 - (B) the aggregate principal amount of each Bid Rate Borrowing shall comply with the provisions of $\underline{\text{Section 2.18(b)(ii)}}$ and together with all other Bid Rate Loans then outstanding shall not cause the limits specified in $\underline{\text{Section 2.1(a)}}$ to be violated;
 - (C) acceptance of Bid Rate Quotes may be made only in ascending order of Absolute Rates or LIBOR Margins, as applicable, in each case beginning with the lowest rate so offered;
 - (D) any acceptance in part by the Borrower shall be in a minimum amount of \$5,000,000 and integral multiples of \$1,000,000 in excess thereof; and

- (E) the Borrower may not accept any Bid Rate Quote that fails to comply with <u>Section 2.18(c)</u> or otherwise fails to comply with the requirements of this Agreement.
- (ii) If Bid Rate Quotes are made by two or more Revolving Lenders with the same Absolute Rates or LIBOR Margins, as applicable, for a greater aggregate principal amount than the amount in respect of which Bid Rate Quotes are permitted to be accepted for the related Interest Period, the principal amount of Bid Rate Loans in respect of which such Bid Rate Quotes are accepted shall be allocated by the Administrative Agent among such Revolving Lenders in proportion to the aggregate principal amount of such Bid Rate Quotes. Determinations by the Administrative Agent of the amounts of Bid Rate Loans shall be conclusive in the absence of manifest error.
- (f) Obligation to Make Bid Rate Loans. The Administrative Agent shall promptly (and in any event not later than (x) 1:00 p.m. (Central Time) on the Business Day immediately preceding the proposed date of borrowing of Absolute Rate Loans and (y) on the date three (3) Business Days prior to the proposed date of borrowing of LIBOR Margin Loans) notify each Revolving Lender as to whose Bid Rate Quote has been accepted and the amount and rate thereof. A Revolving Lender who is notified that it has been selected to make a Bid Rate Loan may designate its Designated Lender (if any) to fund such Bid Rate Loan on its behalf, as described in Section 13.5. Any Designated Lender which funds a Bid Rate Loan shall on and after the time of such funding become the obligee in respect of such Bid Rate Loan and be entitled to receive payment thereof when due. No Revolving Lender shall be relieved of its obligation to fund a Bid Rate Loan, and no Designated Lender shall assume such obligation, prior to the time the applicable Bid Rate Loan is funded. Any Revolving Lender whose offer to make any Bid Rate Loan has been accepted shall, not later than 11:00 a.m. (Central Time) on the date specified for the making of such Loan, make the amount of such Loan available to the Administrative Agent at the Administrative Office in immediately available funds, for the account of the Borrower. The amount so received by the Administrative Agent shall, subject to the terms and conditions of this Agreement, be made available to the Borrower not later than 12:00 p.m. (Central Time) on such date by depositing the same, in immediately available funds, in an account of the Borrower designated in the Disbursement Instruction Agreement.
- (g) No Effect on Revolving Commitment. Except for the purpose and to the extent expressly stated in Section 2.1(a), the amount of any Bid Rate Loan made by any Revolving Lender shall not constitute a utilization of such Revolving Lender's Revolving Commitment.
- 2.19. Extension of Maturity Date. The Borrower shall have the right, exercisable two (2) times for the Facility, to request that the Administrative Agent and the Revolving Lenders agree to extend the Maturity Date by a six-month period. The Borrower may exercise such right only by executing and delivering to the Administrative Agent at least ninety (90) days but not more than one hundred eighty (180) days prior to the current Maturity Date, a written request for such extension (an "Extension Request"). The Administrative Agent shall notify the Revolving Lenders if it receives an Extension Request promptly upon receipt thereof. Subject to satisfaction of the following conditions, the Maturity Date shall be extended for a six-month period effective upon receipt by the Administrative Agent of the Extension Request and payment of the fee referred to in the following clause (ii): (i) immediately prior to such extension and immediately after giving effect thereto, (A) no Default or Event of Default shall exist and (B) the representations and warranties made or deemed made by the Borrower and each other Loan Party in the Loan Documents to which any of them is a party, shall be true and correct in all material respects (or, to the extent qualified by materiality or Material Adverse Effect, in all respects) on and as of the date of such extension with the same force and effect as if made on and as of such date except to the

extent that such representations and warranties expressly relate solely to an earlier date (in which case such representations and warranties shall have been true and correct in all material respects (or, to the extent qualified by materiality or Material Adverse Effect, in all respects) on and as of such earlier date) and except for changes in factual circumstances specifically and expressly permitted under the Loan Documents; and (ii) the Borrower shall have paid to Administrative Agent for the ratable account of each of the Lenders a fee equal to 0.0625% of the amount of the Aggregate Revolving Commitment on the effective date of the extension (which fee shall be fully earned on the date the Administrative Agent receives the Extension Request pursuant to this Section 2.19). At any time prior to the effectiveness of any such extension, upon the Administrative Agent's request, the Borrower shall deliver to the Administrative Agent a certificate from the chief executive officer or chief financial officer certifying the matters referred to in the immediately preceding clauses (i)(A) and (i)(B).

2.20. Pro Rata Treatment/Sharing of Payments.

- (a) Except to the extent otherwise provided herein: (i) each Borrowing from the Revolving Lenders under Sections 2.7, 2.17 and 3.4 shall be made from the Revolving Lenders, and each payment of the fees under Sections 2.10 and 2.19 shall be made for the account of the Revolving Lenders, pro rata according to the amounts of their respective Percentages; (ii) each payment or prepayment of principal of Revolving Loans (other than Bid Rate Loans) shall be made for the account of the Revolving Lenders pro rata in accordance with the respective unpaid principal amounts of the Revolving Loans held by them, provided that, subject to Section 12.16, if immediately prior to giving effect to any such payment in respect of any Revolving Loans the outstanding principal amount of the Revolving Loans shall not be held by the Revolving Lenders pro rata in accordance with their respective Percentages in effect at the time such Revolving Loans were made, then such payment shall be applied to the Revolving Loans in such manner as shall result, as nearly as is practicable, in the outstanding principal amount of the Revolving Loans being held by the Revolving Lenders pro rata in accordance with their respective Percentages; (iii) each payment of interest on Revolving Loans shall be made for the account of the Revolving Lenders, pro rata in accordance with the amounts of interest on such Revolving Loans, then due and payable to the respective Lenders; (iv) the making, conversion and continuation of Revolving Loans of a particular Type shall be made pro rata among the Revolving Lenders, according to the amounts of their respective Revolving Loans, and the then current Interest Period for each Lender's portion of each such Loan of such Type shall be coterminous; (v) each prepayment of principal of Bid Rate Loans pursuant to Section 2.9.(b)(iii) shall be made for account of the Lenders then owed Bid Rate Loans pro rata in accordance with the respective unpaid principal amounts of the Bid Rate Loans then owing to each such Lender; (vi) the Revolving Lenders' participation in, and payment obligations in respect of, Swingline Loans under Section 2.17, shall be in accordance with their respective Percentages; and (vii) the Revolving Lenders' participation in, and payment obligations in respect of, Letters of Credit under Article III, shall be in accordance with their respective Percentages. All payments of principal, interest, fees and other amounts in respect of the Swingline Loans shall be for the account of the Swingline Lender only (except to the extent any Lender shall have acquired a participating interest in any such Swingline Loan pursuant to Section 2.17, in which case such payments shall be pro rata in accordance with such participating interests).
- (b) If a Lender shall obtain payment of any principal of, or interest on, any Loan made by it to the Borrower under this Agreement or shall obtain payment on any other Obligation owing by the Borrower or any other Loan Party through the exercise of any right of set-off, banker's lien, counterclaim or similar right or otherwise or through voluntary prepayments directly to a Lender or other payments made by or on behalf the Borrower or any other Loan Party to a Lender (other than any payment in respect of Specified Rate Management Obligations)

not in accordance with the terms of this Agreement and such payment should be distributed to the Lenders in accordance with Section 2.20(a) or Section 2.22, as applicable, such Lender shall promptly purchase from the other Lenders participations in (or, if and to the extent specified by such Lender, direct interests in) the Loans made by the other Lenders or other Obligations owed to such other Lenders in such amounts, and make such other adjustments from time to time as shall be equitable, to the end that all the Lenders shall share the benefit of such payment (net of any reasonable expenses which may actually be incurred by such Lender in obtaining or preserving such benefit) in accordance with the requirements of Section 2.20(a) or Section 2.22, as applicable. To such end, all the Lenders shall make appropriate adjustments among themselves (by the resale of participations sold or otherwise) if such payment is rescinded or must otherwise be restored. The Borrower agrees that any Lender so purchasing a participation (or direct interest) in the Loans or other Obligations owed to such other Lenders may exercise all rights of set-off, banker's lien, counterclaim or similar rights with respect to such participation as fully as if such Lender were a direct holder of Loans in the amount of such participation. Nothing contained herein shall require any Lender to exercise any such right or shall affect the right of any Lender to exercise and retain the benefits of exercising, any such right with respect to any other indebtedness or obligation of the Borrower.

2.21. Reserved.

- 2.22. <u>Application of Moneys Received</u>. Subject to <u>Section 2.13(b)</u> hereof, all moneys collected or received by the Administrative Agent on account of the Facility directly or indirectly, shall be applied in the following order of priority:
 - (i) to the payment of all reasonable costs incurred in the collection of such moneys of which the Administrative Agent shall have given notice to the Borrower;
 - (ii) to the reimbursement of any yield protection due to any of the Lenders in accordance with Section 4.1;
 - (iii) first to the payment of any fee due pursuant to <u>Section 3.8(b)</u> in connection with the issuance of a Facility Letter of Credit to the applicable Issuing Bank until such fee is paid in full, then next to the payment of the Facility Fee and Facility Letter of Credit Fee to the Lenders, if then due, in that order on a pro rata basis in accordance with the respective amounts of such fees due to the Lenders and then finally to the payment of all fees then due to the Administrative Agent;
 - (iv) to payment of the full amount of interest and principal on the Swingline Loans;
 - (v) first to interest until paid in full and then to principal for all Lenders (x) as allocated by the Borrower (unless an Event of Default exists) between Bid Rate Loans and ratable Borrowings (the amount allocated to ratable Borrowings to be distributed in accordance with the applicable pro rata shares of the outstanding amounts of the Lenders) or (y) if an Event of Default exists, in accordance with the respective Funded Percentages of the Lenders until principal is paid in full, each Lenders' share of such payment to be allocated pro rata among the outstanding Classes and Types of Loans owed to such Lender and then to the Letter of Credit Collateral Account until the full amount of LC Exposures is on deposit therein; and

(vi) any other sums due to the Administrative Agent or any Lender under any of the Loan Documents.

2.23. Reserved.

2.24. <u>Voluntary Prepayments of Loans</u>.

- (a) The Borrower shall have the right at any time and from time to time to prepay any of its Borrowings (other than Bid Rate Loans) in whole or in part, subject to prior notice in accordance with paragraph (b) of this Section. A Bid Rate Loan may only be prepaid with the prior written consent of the Lender holding such Bid Rate Loan.
- (b) The Borrower shall notify the Administrative Agent (and, in the case of prepayment of a Swingline Loan, the Swingline Lender) by email of any prepayment under clause (a) above not later than 11:00 a.m. (Central Time) on the date of prepayment. Each such notice shall be irrevocable and shall specify the prepayment date and the principal amount of each Borrowing or portion thereof to be prepaid. Promptly following receipt of any such notice relating to a Borrowing, the Administrative Agent shall advise the Lenders of the contents thereof. Each partial prepayment pursuant to clause (a) above of any Borrowing shall be in an amount that would be permitted in the case of an advance of a Borrowing of the same Type as provided in Section 2.11 hereof. Each prepayment of a Borrowing shall be applied ratably to the Loans included in the applicable prepaid Borrowing. Prepayments shall be accompanied by accrued interest to the extent required by Section 2.12 hereof and by breakage costs to the extent required by Section 4.4 hereof.

ARTICLE III.

THE LETTER OF CREDIT SUBFACILITY

3.1. Obligation to Issue. Subject to the terms and conditions of this Agreement and in reliance upon the representations and warranties of the Borrower and the General Partner herein set forth, each Issuing Bank hereby agrees to issue for the account of the Borrower, one or more Facility Letters of Credit denominated in Dollars in accordance with this Article III, from time to time during the period commencing on the Agreement Execution Date and ending on a date one (1) Business Day prior to the Maturity Date. The parties acknowledge that there are certain Facility Letters of Credit that were issued under the Existing Credit Agreement which the Borrower has requested remain outstanding under this Agreement. Accordingly, from and after the date of the first Borrowing hereunder, the Facility Letters of Credit identified in Schedule 3.1 shall be deemed issued pursuant to the terms of this Agreement and shall be subject to all of the terms and conditions contained herein as if such Facility Letters of Credit were issued hereunder.

3.2. <u>Types and Amounts</u>. No Issuing Bank shall have any obligation to:

- (i) issue any Facility Letter of Credit if the aggregate maximum amount then available for drawing under Letters of Credit issued by such Issuing Bank, after giving effect to the Facility Letter of Credit requested hereunder, shall exceed any limit imposed by law or regulation upon such Issuing Bank;
- (ii) issue any Facility Letter of Credit if, after giving effect thereto, either (1) the Total Revolving Exposure would exceed the Aggregate Revolving Commitment, (2) the aggregate LC Exposure would exceed \$50,000,000 (the "LC Sublimit") or (3) the

LC Exposure of any Issuing Bank shall exceed the lesser of (a) one-third of the LC Sublimit at any time and (b) the Revolving Commitment of such Issuing Bank in its capacity as a Lender hereunder;

- (iii) issue any Facility Letter of Credit having an expiration date which is after the thirtieth (30th) Business Day immediately preceding the Maturity Date; provided that a Letter of Credit may contain an "evergreen" provision providing for the automatic extension of the expiration date but not to a date beyond the date that is thirty (30) Business Days prior to the Maturity Date; provided that any such "evergreen" Letter of Credit must permit the applicable Issuing Bank to prevent any such extension at least once in each twelve-month period (commencing with the date of issuance of such Letter of Credit) by giving prior notice to the beneficiary thereof not later than a day in each such twelve-month period to be agreed upon at the time such Letter of Credit is issued; or
- (iv) issue any Facility Letter of Credit having an expiration date, or containing automatic extension provisions to extend such date to a date, which is more than twelve (12) months after the date of its issuance.

Notwithstanding the provisions of clause (iii) above, a Facility Letter of Credit may, as a result of its express terms or as the result of the effect of an automatic extension provision, have an expiration date of not more than one year beyond the Maturity Date (any such Facility Letter of Credit being referred to as an "Extended Letter of Credit"), so long as the Borrower delivers to the Administrative Agent for its benefit and the benefit of the applicable Issuing Bank and the Lenders no later than thirty (30) days prior to the Maturity Date, Cash Collateral for such Facility Letter of Credit for deposit into the Letter of Credit Collateral Account in an amount equal to the stated amount of such Facility Letter of Credit; provided, that the obligations of the Borrower under this Section in respect of such Extended Letters of Credit shall survive the termination of this Agreement and shall remain in effect until no such Extended Letters of Credit remain outstanding. If the Borrower fails to provide Cash Collateral with respect to any Extended Letter of Credit by the date thirty (30) days prior to the Maturity Date, such failure shall be treated as a drawing under such Extended Letter of Credit (in an amount equal to the maximum stated amount of such Facility Letter of Credit) and as a failure to pay an Obligation, the provisions of Section 3.5(b) and the second paragraph of Section 11.1 shall apply thereto and such deemed drawing under such Extended Letter of Credit shall be reimbursed (or participations therein funded) by the Lenders in accordance with Section 3.6, with the proceeds being utilized to provide Cash Collateral for such Facility Letter of Credit.

- 3.3. <u>Conditions</u>. In addition to being subject to the satisfaction of the conditions contained in Article V hereof, the obligation of each Issuing Bank to issue any Facility Letter of Credit is subject to the satisfaction in full of the following conditions:
 - (i) the Borrower shall have delivered to such Issuing Bank at such times and in such manner as such Issuing Bank may reasonably prescribe such documents and materials as may be reasonably required pursuant to the terms of the proposed Facility Letter of Credit (it being understood that if any inconsistency exists between such documents and the Loan Documents, the terms of the Loan Documents shall control) and the proposed Facility Letter of Credit shall be reasonably satisfactory to such Issuing Bank as to form and content;
 - (ii) as of the date of issuance, no order, judgment or decree of any court, arbitrator or governmental authority shall purport by its terms to enjoin or restrain such Issuing Bank from issuing the requested Facility Letter of Credit and no law, rule or

regulation applicable to such Issuing Bank and no request or directive (whether or not having the force of law) from any governmental authority with jurisdiction over such Issuing Bank shall prohibit or request that such Issuing Bank refrain from the issuance of Letters of Credit generally or the issuance of the requested Facility Letter of Credit in particular;

- (iii) there shall not exist any Default or Event of Default; and
- (iv) the issuance of the proposed Facility Letter of Credit would not violate any policies of such Issuing Bank applicable to Letters of Credit generally.

3.4. Procedure for Issuance of Facility Letters of Credit.

- (a) The Borrower shall give the applicable Issuing Bank and the Administrative Agent at least five (5) Business Days' (or such shorter period as the Administrative Agent and such Issuing Bank may agree in their sole discretion) prior written notice of any requested issuance of a Facility Letter of Credit under this Agreement (a "Letter of Credit Request"), a copy of which shall be sent immediately to all Lenders (except that, in lieu of such written notice, the Borrower may give such Issuing Bank and the Administrative Agent telephonic notice of such request if confirmed in writing by delivery to such Issuing Bank and the Administrative Agent (i) immediately of an email of the written notice required hereunder which has been signed by an authorized officer and (ii) promptly (but in no event later than the requested date of issuance) of the written notice required hereunder containing the original signature of an authorized officer, the substance of which notice shall be promptly forwarded to all Lenders); such notice shall be irrevocable and shall specify:
 - (i) whether the requested Facility Letter of Credit is, in the Borrower's belief, a Financial Letter of Credit or a Performance Letter of Credit;
 - (ii) the stated amount of the Facility Letter of Credit requested (which stated amount shall not be less than \$50,000);
 - (iii) the effective date (which day shall be a Business Day) of issuance of such requested Facility Letter of Credit (the "Issuance Date");
 - (iv) the date on which such requested Facility Letter of Credit is to expire;
 - (v) the purpose for which such Facility Letter of Credit is to be issued;
 - (vi) the Person for whose benefit the requested Facility Letter of Credit is to be issued;
 - $(vii) \quad \text{any special language required to be included in the Facility Letter of Credit}; \\$

At the time such request is made, the Borrower shall also provide the Administrative Agent and such Issuing Bank with a copy of the form of the Facility Letter of Credit that the Borrower is requesting be issued. Such notice, to be effective, must be received by such Issuing Bank and the Administrative Agent not later than 1:00 p.m. (Central Time) on the last Business Day on which notice can be given under this <u>Section 3.4(a)</u>. Following receipt of such notice and prior to the issuance of the requested Facility Letter of Credit, the Administrative Agent shall notify the

Borrower, and the applicable Issuing Bank of the amount of the Total Revolving Exposure after giving effect to (i) the issuance of such Facility Letter of Credit, (ii) the issuance or expiration of any other Facility Letter of Credit that is to be issued or will expire prior to the requested date of issuance of such Facility Letter of Credit and (iii) the borrowing or repayment of any Revolving Loans or Swingline Loans that (based upon notices delivered to the Administrative Agent by the Borrower) are to be borrowed or repaid prior to the requested date of issuance of such Facility Letter of Credit. A Facility Letter of Credit shall be issued, amended, renewed or extended only if (and upon issuance, amendment, renewal or extension of each Facility Letter of Credit the Borrower shall be deemed to represent and warrant that), after giving effect to such issuance, amendment, renewal or extension (i) the LC Exposure shall not exceed \$50,000,000, and (ii) the Total Revolving Exposure shall not exceed the Aggregate Revolving Commitment. A Facility Letter of Credit shall not be issued, extended or renewed if the applicable Issuing Bank has received written notice from the Administrative Agent at least one (1) Business Day prior to the date of such requested issuance, extension or renewal, that one or more applicable conditions contained in Section 5.2 shall not be satisfied. Administrative Agent shall promptly give a copy of the Letter of Credit Request to the other Lenders.

- (b) Subject to the terms and conditions of this <u>Article III</u> and <u>provided</u> that the applicable conditions set forth in <u>Article V</u> hereof have been satisfied, the applicable Issuing Bank shall, on the Issuance Date, issue a Facility Letter of Credit on behalf of the Borrower in accordance with the Letter of Credit Request and such Issuing Bank's usual and customary business practices unless such Issuing Bank has actually received (i) written notice from the Borrower specifically revoking the Letter of Credit Request with respect to such Facility Letter of Credit, (ii) written notice from a Lender, which complies with the provisions of <u>Section 3.6(a)</u>, or (iii) written or telephonic notice from the Administrative Agent stating that the issuance of such Facility Letter of Credit would violate <u>Section 3.2</u>.
- (c) The applicable Issuing Bank shall give the Administrative Agent (who shall promptly notify Lenders) and the Borrower written or telex notice, or telephonic notice confirmed promptly thereafter in writing, of the issuance of a Facility Letter of Credit (the "Issuance Notice"), which shall indicate such Issuing Bank's reasonable determination as to whether such Facility Letter of Credit is a Financial Letter of Credit or a Performance Letter of Credit, which determination shall be conclusive absent manifest error.
- (d) The applicable Issuing Bank shall not extend or amend any Facility Letter of Credit unless the requirements of this <u>Section 3.4</u> are met as though a new Facility Letter of Credit was being requested and issued.
- (e) The Borrower shall promptly examine a copy of each Facility Letter of Credit and each amendment thereto that is delivered to it and, in the event of any claim of noncompliance with the Borrower's instructions or other irregularity, the Borrower will immediately notify the applicable Issuing Bank. The Borrower shall be conclusively deemed to have waived any such claim against any Issuing Bank and its correspondents unless such notice is given as aforesaid.

3.5. Reimbursement Obligations; Duties of Issuing Banks.

(a) If an Issuing Bank shall make any LC Disbursement in respect of a Facility Letter of Credit, the Borrower shall reimburse such LC Disbursement by paying to the Administrative Agent an amount equal to such LC Disbursement not later than 11:00 a.m. (Central Time) on the date that such LC Disbursement is made, if the Borrower shall have

received notice of such LC Disbursement prior to 10:00 a.m. (Central Time), as applicable, on such date, or, if such notice has not been received by the Borrower prior to such time on such date, then not later than 11:00 a.m. (Central Time), as applicable, on the Business Day immediately following the day that the Borrower receives such notice; <u>provided</u> that the Borrower may, subject to the conditions to borrowing set forth herein, request in accordance with <u>Section 2.4</u> or <u>2.7</u> that such payment be financed with an Adjusted Base Rate Borrowing under the Facility or Swingline Loan in an equivalent amount and, to the extent so financed, the Borrower's obligation to make such payment shall be discharged and replaced by the resulting Adjusted Base Rate Borrowing or Swingline Loan. Promptly following receipt by the Administrative Agent of any payment from the Borrower pursuant to this paragraph, the Administrative Agent shall distribute such payment to the applicable Issuing Bank or, to the extent that Revolving Lenders have made payments pursuant to this paragraph to reimburse the applicable Issuing Bank, then to such Lenders and such Issuing Bank as their interests may appear. Any payment made by a Revolving Lender pursuant to this paragraph to reimburse any Issuing Bank for any LC Disbursement (other than the funding of Adjusted Base Rate Revolving Loans or a Swingline Loan as contemplated above) shall not constitute a Loan and shall not relieve the Borrower of its obligation to reimburse such LC Disbursement.

(b) If an Issuing Bank shall make any LC Disbursement, then, unless the Borrower shall reimburse such LC Disbursement in full on the date such LC Disbursement is made (in the local time where the LC Disbursement is made regardless of when such reimbursement is due under Section 3.5(a)), the unpaid amount thereof shall bear interest, for each day from and including the date such LC Disbursement is made to, but excluding, the date that the Borrower reimburses such LC Disbursement, at the rate per annum then applicable to Adjusted Base Rate Borrowings; provided that, if the Borrower fails to reimburse such LC Disbursement when due pursuant to paragraph (a) of this Section, then Section 2.7 shall apply. Interest accrued pursuant to this paragraph shall be for the account of the applicable Issuing Bank, except that interest accrued on and after the date of payment by any Revolving Lender pursuant to Section 3.5 to reimburse such Issuing Bank shall be for the account of such Lender to the extent of such payment.

3.6. <u>Participation</u>.

(a) Immediately upon issuance by the applicable Issuing Bank of any Facility Letter of Credit in accordance with the procedures set forth in Section 3.4, each Revolving Lender shall be deemed to have irrevocably and unconditionally purchased and received from such Issuing Bank, without recourse, representation or warranty, an undivided interest and participation equal to such Lender's Percentage in such Facility Letter of Credit (including, without limitation, all obligations of the Borrower with respect thereto) and all related rights hereunder and under the Guaranty and other Loan Documents; provided that a Facility Letter of Credit issued by such Issuing Bank shall not be deemed to be a Facility Letter of Credit for purposes of this Section 3.6 if such Issuing Bank shall have received written notice from any Lender on or before the Business Day prior to the date of its issuance of such Facility Letter of Credit that one or more of the conditions contained in Section 5.2 is not then satisfied, and in the event such Issuing Bank receives such a notice it shall have no further obligation to issue any Facility Letter of Credit until such notice is withdrawn by that Lender or such Issuing Bank receives a notice from the Administrative Agent that such condition has been effectively waived in accordance with the provisions of this Agreement. Each Revolving Lender's obligation to make further Revolving Loans to the Borrower (other than any payments such Lender is required to make under subparagraph (b) below) or to purchase an interest from such Issuing Bank in any subsequent

letters of credit issued by such Issuing Bank on behalf of the Borrower shall be reduced by such Lender's Percentage of the undrawn portion of each Facility Letter of Credit outstanding.

- (b) In the event that an Issuing Bank makes any payment under any Facility Letter of Credit and the Borrower shall not have repaid such amount to such Issuing Bank pursuant to Section 3.7 hereof, such Issuing Bank shall promptly notify the Administrative Agent, which shall promptly notify each Revolving Lender of such failure, and each Revolving Lender shall promptly and unconditionally pay to the Administrative Agent for the account of such Issuing Bank the amount of such Lender's Percentage of each LC Disbursement made by such Issuing Bank. The failure of any Revolving Lender to make available to the Administrative Agent for the account of any Issuing Bank its Percentage of the unreimbursed amount of any such payment shall not relieve any other Lender of its obligation hereunder to make available to the Administrative Agent for the account of such Issuing Bank its Percentage of the unreimbursed amount of any payment on the date such payment is to be made, but no Lender shall be responsible for the failure of any other Lender to make available to the Administrative Agent its Percentage of the unreimbursed amount of any payment on the date such payment is to be made.
- (c) Whenever an Issuing Bank receives a payment on account of a Reimbursement Obligation, including any interest thereon, such Issuing Bank shall promptly pay to the Administrative Agent and the Administrative Agent shall promptly pay to each Lender which has funded its participating interest therein, in immediately available funds, an amount equal to such Lender's Percentage thereof.
- (d) Upon the request of the Administrative Agent or any Lender, any Issuing Bank shall furnish to such Administrative Agent or Lender copies of any Facility Letter of Credit to which such Issuing Bank is party and such other documentation as may reasonably be requested by the Administrative Agent or Lender.
- (e) The obligations of a Lender to make payments to the Administrative Agent for the account of any Issuing Bank with respect to a Facility Letter of Credit shall be absolute, unconditional and irrevocable, not subject to any counterclaim, set off, qualification or exception whatsoever other than a failure of such Issuing Bank to comply with the terms of this Agreement relating to the issuance of such Facility Letter of Credit, and such payments shall be made in accordance with the terms and conditions of this Agreement under all circumstances.

3.7. Payment of Reimbursement Obligations.

- (a) The Borrower agrees to pay to the Administrative Agent for the account of the applicable Issuing Bank the amount of all Reimbursement Obligations, interest and other amounts payable to such Issuing Bank under or in connection with any Facility Letter of Credit when due, irrespective of any claim, set off, defense or other right which the Borrower may have at any time against such Issuing Bank or any other Person, under all circumstances, including without limitation any of the following circumstances:
 - (i) any lack of validity or enforceability of this Agreement or any of the other Loan Documents;
 - (ii) the existence of any claim, setoff, defense or other right which the Borrower may have at any time against a beneficiary named in a Facility Letter of Credit or any transferee of any Facility Letter of Credit (or any Person for whom any such transferee may be acting), the Administrative Agent, any Issuing Bank, any Lender, or

any other Person, whether in connection with this Agreement, any Facility Letter of Credit, the transactions contemplated herein or any unrelated transactions (including any underlying transactions between the Borrower and the beneficiary named in any Facility Letter of Credit);

- (iii) any draft, certificate or any other document presented under the Facility Letter of Credit proving to be forged, fraudulent, invalid or insufficient in any respect of any statement therein being untrue or inaccurate in any respect;
- (iv) the surrender or impairment of any security for the performance or observance of any of the terms of any of the Loan Documents; or
 - (v) the occurrence of any Default or Event of Default.
- (b) In the event any payment by the Borrower received by the applicable Issuing Bank or the Administrative Agent with respect to a Facility Letter of Credit and distributed by the Administrative Agent to the Lenders on account of their participations is thereafter set aside, avoided or recovered from the Administrative Agent or such Issuing Bank in connection with any receivership, liquidation, reorganization or bankruptcy proceeding, each Lender which received such distribution shall, upon demand by the Administrative Agent, contribute such Lender's Percentage of the amount set aside, avoided or recovered together with interest at the rate required to be paid by such Issuing Bank or the Administrative Agent upon the amount required to be repaid by such Issuing Bank or the Administrative Agent.

3.8. <u>Compensation for Facility Letters of Credit</u>.

- (a) The Borrower shall pay to the Administrative Agent, for the ratable account of the Revolving Lenders, based upon such Lenders' respective Percentages, a per annum fee (the "Facility Letter of Credit Fee") with respect to each Facility Letter of Credit that is equal to (i) the LIBOR Applicable Margin in effect from time to time in the case of Financial Letters of Credit, and (ii) the LIBOR Applicable Margin from time to time minus 0.25% in the case of Performance Letters of Credit. The Facility Letter of Credit Fee relating to any Facility Letter of Credit shall be due and payable in arrears in equal installments on the first Business Day of each month following the issuance of any Facility Letter of Credit and, to the extent any such fees are then due and unpaid, on the Maturity Date. The Administrative Agent shall promptly remit such Facility Letter of Credit Fees, when paid, to the other Revolving Lenders in accordance with their Percentages thereof. The Borrower shall not have any liability to any Lender for the failure of the Administrative Agent to promptly deliver funds to any such Lender and shall be deemed to have made all such payments on the date the respective payment is made by the Borrower to the Administrative Agent, provided such payment is received by the time specified in Section 2.13 hereof.
- (b) Each Issuing Bank also shall have the right to receive solely for its own account an issuance fee equal to a percentage of the face amount of each Facility Letter of Credit issued by it in an amount mutually agreed in writing between the Borrower and such Issuing Bank, payable by the Borrower on the Issuance Date for each such Facility Letter of Credit. The applicable Issuing Bank shall also be entitled to receive its reasonable out-of-pocket costs and such Issuing Bank's standard charges of issuing, amending and servicing Facility Letters of Credit and processing draws thereunder.

- (f) Any fees due under this $\underline{\text{Section 3.8}}$ shall be computed on the basis of a year of 360 days and the actual number of days elapsed.
- 3.9. <u>Letter of Credit Collateral Account</u>. The Borrower hereby agrees that it will, until the Maturity Date, maintain a special collateral account (the "Letter of Credit Collateral Account") at the Administrative Agent's office at the address specified pursuant to Article XV, in the name of the Borrower but under the sole dominion and control of the Administrative Agent, for the benefit of the Lenders, and in which the Borrower shall have no interest other than as set forth in <u>Section 11.1</u>. In addition to the foregoing, the Borrower hereby grants to the Administrative Agent, for the benefit of the Revolving Lenders, a security interest in and to the Letter of Credit Collateral Account and any funds that may hereafter be on deposit in such account, including income earned thereon. The Lenders acknowledge and agree that the Borrower has no obligation to fund the Letter of Credit Collateral Account unless and until so required under Section 3.2 or Section 11.1 hereof.
- 3.10. <u>Issuing Bank Agreements</u>. Unless otherwise requested by the Administrative Agent, each Issuing Bank shall report in writing to the Administrative Agent (i) promptly following the end of each calendar month, the aggregate amount of Facility Letters of Credit issued by it and outstanding at the end of such month, (ii) on or prior to each Business Day on which such Issuing Bank expects to issue, amend, renew or extend any Facility Letter of Credit, the date of such issuance, amendment, renewal or extension, and the aggregate face amount of the Facility Letters of Credit to be issued, amended, renewed or extended by it and outstanding after giving effect to such issuance, amendment, renewal or extension occurred (and whether the amount thereof changed), it being understood that such Issuing Bank shall not permit any issuance, renewal, extension or amendment resulting in an increase in the amount of any Facility Letter of Credit to occur without first obtaining written confirmation from the Administrative Agent that it is then permitted under this Agreement, (iii) on each Business Day on which such Issuing Bank makes any LC Disbursement, the date of such LC Disbursement and the amount of such LC Disbursement, (iv) on any Business Day on which the Borrower fails to reimburse an LC Disbursement required to be reimbursed to such Issuing Bank on such day, the date of such failure and the amount of such LC Disbursement and (v) on any other Business Day, such other information as the Administrative Agent shall reasonably request.

ARTICLE IV.

CHANGE IN CIRCUMSTANCES

- 4.1. <u>Yield Protection</u>. If any Change in Law:
 - (i) subjects the Administrative Agent, any Lender or any applicable Lending Office to any taxes, duties, levies, imposts, deductions, assessments, fees, charges or withholdings, and all liabilities with respect thereto (other than (A) Taxes, (B) Excluded Taxes or (C) Other Taxes) on its loans, loan principal, letters of credit, commitments, or other obligations, or its deposits, reserves, other liabilities, capital or liquidity attributable to making, converting, funding or maintaining its Borrowings or its Revolving Commitment, or
 - (ii) imposes or increases or deems applicable any reserve, assessment, insurance charge, special deposit, liquidity or similar requirement against assets of, deposits with or for the account of, or credit extended by, any Lender or any applicable Lending Office (other than reserves and assessments taken into account in determining the interest rate applicable to Eurocurrency Borrowings), or

(iii) imposes any other condition, and the result is to increase the cost of any Lender or any applicable Lending Office of making, funding or maintaining loans or reduces any amount receivable by any Lender or any applicable Lending Office in connection with loans, or requires any Lender or any applicable Lending Office to make any payment calculated by reference to the amount of loans held, Letters of Credit issued or participated in or interest received by it, by an amount deemed material by such Lender,

then, within fifteen (15) days of demand by the Administrative Agent or such Lender, the Borrower shall pay the Administrative Agent or such Lender that portion of such increased expense incurred or reduction in an amount received which the Administrative Agent or such Lender determines is attributable to making, funding and maintaining its Borrowings and its Revolving Commitment (which determination shall be made in good faith (and not on an arbitrary or capricious basis) and consistent with similarly situated customers of the applicable Lender after consideration of such factors as such Lender then reasonably determines to be relevant).

Changes in Capital Adequacy Regulations. If a Lender determines the amount of capital or liquidity required or expected to be maintained by such Lender, any Lending Office of such Lender or any corporate entity controlling such Lender is increased as a result of a Change in Law (which determination shall be made in good faith (and not on an arbitrary or capricious basis) and consistent with similarly situated customers of the applicable Lender after consideration of such factors as such Lender then reasonably determines to be relevant), then, within fifteen (15) days of demand by such Lender, the Borrower shall pay such Lender the amount necessary to compensate for any shortfall in the rate of return on the portion of such increased capital which is attributable to this Agreement, its Borrowings, its interest in the Facility Letters of Credit, or its obligation to make Borrowings hereunder or participate in or issue Facility Letters of Credit hereunder (after taking into account such Lender's policies as to capital adequacy). Without in any way affecting the Borrower's obligation to pay compensation actually claimed by a Lender under this Section 4.2, the Borrower shall have the right to replace any Lender which has demanded such compensation provided that the Borrower notifies such Lender that it has elected to replace such Lender and notifies such Lender and the Administrative Agent of the identity of the proposed replacement Lender not more than six (6) months after the date of such Lender's most recent demand for compensation under this Section 4.2, and further provided that such replacement is otherwise in accordance with Section 4.7. The Lender being replaced shall assign its Revolving Commitment and its rights and obligations under this Facility to the replacement Lender in accordance with the requirements of Section 13.3 hereof and the replacement Lender shall assume such Percentage of the total Revolving Commitment and the related obligations under this Facility prior to the Maturity Date to be extended, all pursuant to an assignment agreement substantially in the form of Exhibit I hereto. The purchase by the replacement Lender shall be at par (plus all accrued and unpaid interest and any other sums owed to such Lender being replaced hereunder) which shall be paid to the Lender being replaced upon the execution and delivery of the assignment. The Lender being replaced shall continue to be entitled to the benefits of Sections 4.1, 4.2, 4.4, 4.5 and 14.6 for events recurring prior to assignment to the replacement Lender.

4.3. [Reserved].

4.4. <u>Funding Indemnification</u>. If any payment of a ratable LIBOR Borrowing or a Bid Rate Loan occurs on a date which is not the last day of the applicable Interest Period, whether because of acceleration, prepayment or otherwise, or a ratable LIBOR Borrowing or a Bid Rate Loan is not made, continued or converted on the date specified by the Borrower for any reason other than default by one or more of the Lenders, the Borrower will indemnify each Lender and each Designated Lender for any loss or cost incurred by such Lender or Designated Lender, as applicable, resulting therefrom, including,

without limitation, any loss or cost in liquidating or employing deposits acquired to fund or maintain the ratable LIBOR Borrowing or Bid Rate Loan.

4.5. <u>Taxes</u>.

- (i) All payments by the Borrower to or for the account of any Lender or the Administrative Agent hereunder or under any Note shall be made free and clear of and without deduction for any and all Taxes. If the Borrower shall be required by law to deduct any Taxes from or in respect of any sum payable hereunder to any Lender or the Administrative Agent, (a) the sum payable shall be increased as necessary so that after making all required deductions (including deductions applicable to additional sums payable under this Section 4.5) such Lender or the Administrative Agent (as the case may be) receives an amount equal to the sum it would have received had no such deductions been made, (b) the Borrower shall make such deductions, (c) the Borrower shall pay the full amount deducted to the relevant authority in accordance with Applicable Law, and (d) the Borrower shall furnish to the Administrative Agent the original copy of a receipt evidencing payment thereof within 30 days after such payment is made.
- (ii) In addition, the Borrower hereby agrees to pay any present or future stamp or documentary taxes and any other excise or property taxes, charges or similar levies which arise from any payment made hereunder or under any Note or from the execution or delivery of, or otherwise with respect to, this Agreement or any Note ("Other Taxes").
- (iii) The Borrower hereby agrees to indemnify the Administrative Agent and each Lender for the full amount of Taxes or Other Taxes (including, without limitation, any Taxes or Other Taxes imposed on amounts payable under this <u>Section 4.5</u>) paid by the Administrative Agent or such Lender as a result of its Revolving Commitment, any Loans made by it hereunder, or otherwise in connection with its participation in this Agreement and any liability (including penalties, interest and expenses) arising therefrom or with respect thereto. Payments due under this indemnification shall be made within 30 days of the date the Administrative Agent or such Lender makes demand therefor pursuant to Section 4.6.
- (iv) Each Lender that is not incorporated under the laws of the United States of America or a state thereof (each a "Non-U.S. Lender") agrees that it will, not more than ten (10) Business Days after the date of this Agreement, (i) deliver to the Administrative Agent two duly completed copies of United States Internal Revenue Service Form W-8BEN, W-8BEN-E or W-8ECI, certifying in either case that such Lender is entitled to receive payments under this Agreement without deduction or withholding of any United States federal income taxes, and (ii) deliver to the Administrative Agent a United States Internal Revenue Form W-8 or W-9, as the case may be, and certify that it is entitled to an exemption from United States backup withholding tax. Each Non-U.S. Lender further undertakes to deliver to each of the Borrower and the Administrative Agent (x) renewals or additional copies of such form (or any successor form) on or before the date that such form expires or becomes obsolete, and (y) after the occurrence of any event requiring a change in the most recent forms so delivered by it, such additional forms or amendments thereto as may be reasonably requested by the Borrower or the Administrative Agent. All forms or amendments described in the preceding sentence shall certify that such Lender is entitled to receive payments under this Agreement without deduction or withholding of any United States

federal income taxes, unless an event (including without limitation any change in treaty, law or regulation) has occurred prior to the date on which any such delivery would otherwise be required which renders all such forms inapplicable or which would prevent such Lender from duly completing and delivering any such form or amendment with respect to it and such Lender advises the Borrower and the Administrative Agent that it is not capable of receiving payments without any deduction or withholding of United States federal income tax.

- (v) For any period during which a Non-U.S. Lender has failed to provide the Borrower with an appropriate form pursuant to clause (iv), above (unless such failure is due to a change in treaty, law or regulation, or any change in the interpretation or administration thereof by any governmental authority, occurring subsequent to the date on which a form originally was required to be provided), such Non-U.S. Lender shall not be entitled to indemnification under this <u>Section 4.5</u> with respect to Taxes imposed by the United States; <u>provided</u> that, should a Non-U.S. Lender which is otherwise exempt from or subject to a reduced rate of withholding tax become subject to Taxes because of its failure to deliver a form required under clause (iv), above, the Borrower shall take such steps as such Non-U.S. Lender shall reasonably request to assist such Non-U.S. Lender to recover such Taxes.
- (vi) Any Lender that is entitled to an exemption from or reduction of withholding tax with respect to payments under this Agreement or any Note pursuant to the law of any relevant jurisdiction or any treaty shall deliver to the Borrower (with a copy to the Administrative Agent), at the time or times prescribed by Applicable Law, such properly completed and executed documentation prescribed by Applicable Law as will permit such payments to be made without withholding or at a reduced rate.
- (vii) If the U.S. Internal Revenue Service or any other governmental authority of the United States or any other country or any political subdivision thereof asserts a claim that the Administrative Agent did not properly withhold tax from amounts paid to or for the account of any Lender (because the appropriate form was not delivered or properly completed, because such Lender failed to notify the Administrative Agent of a change in circumstances which rendered its exemption from withholding ineffective, or for any other reason), such Lender shall indemnify the Administrative Agent fully for all amounts paid, directly or indirectly, by the Administrative Agent as tax, withholding therefor, or otherwise, including penalties and interest, and including taxes imposed by any jurisdiction on amounts payable to the Administrative Agent under this subsection, together with all costs and expenses related thereto (including attorneys fees and time charges of attorneys for the Administrative Agent, which attorneys may be employees of the Administrative Agent). The obligations of the Lenders under this Section 4.5(vii) shall survive the payment of the Obligations and termination of this Agreement.
- (viii) If a payment made to a Lender hereunder or under any Note would be subject to United States federal withholding tax imposed by FATCA if such Lender were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the Code, as applicable) such Lender shall deliver to the Borrower and the Administrative Agent at the time or times prescribed by law and at such time or times reasonably requested by the Borrower or the Administrative Agent such documentation prescribed by Applicable Law (including as prescribed by Section 1471(b)(3)(C)(i) of the Code) and such additional documentation reasonably requested by the Borrower or the Administrative Agent as may be necessary for the

Borrower and the Administrative Agent to comply with their obligations under FATCA and to determine that such Lender has complied with such Lender's obligations under FATCA or to determine the amount to deduct and withhold from such payment. Solely for purposes of this Section 4.5(viii), "FATCA" shall include any amendments made to FATCA after the date of this Agreement.

- (ix) Each of the Lenders represents that as of the Agreement Execution Date it is not aware of any facts that would give rise to a claim for additional payments under this Section 4.5.
- (x) For purposes of determining withholding taxes imposed under the FATCA, from and after the Agreement Execution Date, the Borrower and the Administrative Agent shall treat (and the Lenders hereby authorize the Administrative Agent to treat) the Agreement as not qualifying as a "grandfathered obligation" within the meaning of Treasury Regulation Section 1.1471-2(b)(2) (i).
- 4.6. <u>Lender Statements</u>; <u>Survival of Indemnity</u>. To the extent reasonably possible, each Lender shall designate an alternate Lending Office with respect to its LIBOR Borrowings to reduce any liability of the Borrower to such Lender under <u>Sections 4.1</u>, <u>4.2</u> and <u>4.5</u> or to avoid the unavailability of a LIBOR Borrowing, so long as such designation is not disadvantageous to such Lender. Each Lender shall deliver a written statement of such Lender as to the amount due, if any, under <u>Sections 4.1</u>, <u>4.2</u>, <u>4.4</u> and <u>4.5</u> hereof. Such written statement shall set forth in reasonable detail the calculations upon which such Lender determined such amount and shall be final, conclusive and binding on the Borrower in the absence of manifest error. The amount due in such statement shall not include amounts due under <u>Section 4.5</u> that are either attributable to facts known to the Lender as of the Agreement Execution Date or that relate to a time period more than ninety (90) days prior to the giving of such written statement. Determination of amounts payable under such Sections in connection with a LIBOR Borrowing shall be calculated as though each Lender funded its LIBOR Borrowing through the purchase of a deposit of the type and maturity corresponding to the deposit used as a reference in determining the Adjusted LIBOR Rate applicable to such Borrowing, whether in fact that is the case or not. Unless otherwise provided herein, the amount specified in the written statement shall be payable on demand after receipt by the Borrower of the written statement. The obligations of the Borrower under <u>Sections 4.1</u>, <u>4.2</u>, <u>4.4</u> and <u>4.5</u> hereof shall survive payment of the Obligations and termination of this Agreement.
- 4.7. Replacement of Lenders under Certain Circumstances. The Borrower shall be permitted to replace any Lender which (a) is not capable of receiving payments without any deduction or withholding of United States federal income tax pursuant to Section 4.5, or (b) cannot maintain its LIBOR Borrowings at a suitable Lending Office pursuant to Section 4.6, with a replacement bank or other financial institution or (c) becomes a Defaulting Lender; provided that (i) such replacement does not conflict with any applicable legal or regulatory requirements affecting the remaining Lenders, (ii) no Event of Default or (after notice thereof to the Borrower) no Default shall have occurred and be continuing at the time of such replacement, (iii) the replacement bank or institution shall repay other amounts owing to such replaced Lender prior to the date of replacement, (iv) the Borrower shall be liable to such replaced Lender under Sections 4.4 and 4.6 if any LIBOR Borrowing owing to such replaced Lender shall be prepaid (or purchased) other than on the last day of the Interest Period relating thereto, (v) the replacement bank or institution, if not already a Lender, and the terms and conditions of such replacement, shall be reasonably satisfactory to the Administrative Agent, the Swingline Lender and the Issuing Banks, (vi) the replaced Lender shall be obligated to make such replacement in accordance with the provisions of Section 13.3 (provided that the Borrower shall be obligated to pay the processing fee referred to therein and the replaced Lender's consent shall not be required), (vii) until such time as such

replacement shall be consummated, the Borrower shall continue to pay all amounts payable hereunder without setoff, deduction, counterclaim or withholding and (viii) any such replacement shall not be deemed to be a waiver of any rights which the Borrower, the Administrative Agent or any other Lender shall have against the replaced Lender.

4.8. Changed Circumstances.

- (a) <u>Circumstances Affecting Base LIBOR Rate Availability.</u> Subject to clause (c) below, in connection with any request for a LIBOR Borrowing or a conversion to or continuation thereof or otherwise, if for any reason (i) the Administrative Agent shall determine (which determination shall be conclusive and binding absent manifest error) that Dollar deposits are not being offered to banks in the London interbank eurodollar market for the applicable amount and Interest Period of such Loan, (ii) the Administrative Agent shall determine (which determination shall be conclusive and binding absent manifest error) that reasonable and adequate means do not exist for the ascertaining the LIBOR Rate for such Interest Period with respect to a proposed LIBOR Borrowing or (iii) the Required Lenders shall determine (which determination shall be conclusive and binding absent manifest error) that the LIBOR Rate does not adequately and fairly reflect the cost to such Lenders of making or maintaining such Loans during such Interest Period, then the Administrative Agent shall promptly give notice thereof to the Borrower. Thereafter, until the Administrative Agent notifies the Borrower that such circumstances no longer exist, the obligation of the Lenders to make LIBOR Borrowings and the right of the Borrower to convert any Loan to or continue any Loan as a LIBOR Borrowing shall be suspended, and the Borrower shall either (A) repay in full (or cause to be repaid in full) the then outstanding principal amount of each such LIBOR Borrowing; or (B) Borrowings shall, to the extent permissible under Applicable Law, bear interest at the Substitute Rate, rather than any Loans becoming payable.
- (b) <u>Laws Affecting Base LIBOR Rate Availability</u>. If, after the date hereof, the introduction of, or any change in, any Applicable Law or any change in the interpretation or administration thereof by any Governmental Authority, central bank or comparable agency charged with the interpretation or administration thereof, or compliance by any of the Lenders (or any of their respective Lending Offices) with any request or directive (whether or not having the force of law) of any such Governmental Authority, central bank or comparable agency, shall make it unlawful or impossible for any of the Lenders (or any of their respective Lending Offices) to honor its obligations hereunder to make or maintain any LIBOR Borrowing, such Lender shall promptly give notice the Administrative Agent and the Administrative Agent shall promptly give notice to the Borrower and the other Lenders. Thereafter, until the Administrative Agent notifies the Borrower that such circumstances no longer exist, (i) the obligations of the Lenders to make LIBOR Borrowings, and the right of the Borrower to convert any Loan to a LIBOR Borrowing or continue any Loan as a LIBOR Borrowing shall be suspended and thereafter the Borrower may select only Adjusted Base Rate Borrowings that shall, to the extent permissible under Applicable Law, bear interest at the Substitute Rate, and (ii) if any of the Lenders may not lawfully continue to maintain a LIBOR Borrowing to the end of the then current Interest Period applicable thereto, the applicable Loan shall immediately be converted to a Adjusted Base Rate Borrowing bearing interest at the Substitute Rate for the remainder of such Interest Period.

(c) Benchmark Replacement Setting.

(i) <u>Benchmark Replacement</u>. (A) Notwithstanding anything to the contrary herein or in any other Loan Document (and any documentation related to a Rate Management Transaction shall be deemed not to be a "Loan Document" for purposes of this <u>Section 4.8(c)</u>) if a Benchmark Transition Event, an Early Opt-in Election or an Other Benchmark

Rate Election, as applicable, and its related Benchmark Replacement Date have occurred prior to the Reference Time in respect of any setting of the then-current Benchmark, then (x) if a Benchmark Replacement is determined in accordance with clause (a)(1) or (a)(2) of the definition of "Benchmark Replacement" for such Benchmark Replacement Date, such Benchmark Replacement will replace such Benchmark for all purposes hereunder and under any Loan Document in respect of such Benchmark setting and subsequent Benchmark settings without any amendment to, or further action or consent of any other party to, this Agreement or any other Loan Document and (y) if a Benchmark Replacement is determined in accordance with clause (a)(3) or clause (c) of the definition of "Benchmark Replacement" for such Benchmark Replacement Date, such Benchmark Replacement will replace such Benchmark for all purposes hereunder and under any Loan Document in respect of any Benchmark setting at or after 5:00 p.m. (New York City time) on the fifth (5th) Business Day after the date notice of such Benchmark Replacement is provided to the Lenders without any amendment to, or further action or consent of any other party to, this Agreement or any other Loan Document so long as the Administrative Agent has not received, by such time, written notice of objection to such Benchmark Replacement from Lenders comprising the Required Lenders. If an Unadjusted Benchmark Replacement is Daily Simple SOFR, all interest payments will be payable on a monthly basis.

- (A) Notwithstanding anything to the contrary herein or in any other Loan Document, if a Term SOFR Transition Event and its related Benchmark Replacement Date have occurred prior to the Reference Time in respect of any setting of the thencurrent Benchmark, then the applicable Benchmark Replacement will replace the then-current Benchmark for all purposes hereunder or under any Loan Document in respect of such Benchmark setting and subsequent Benchmark settings, without any amendment to, or further action or consent of any other party to, this Agreement or any other Loan Document; provided that this clause (B) shall not be effective unless the Administrative Agent has delivered to the Lenders and the Borrower a Term SOFR Notice. For the avoidance of doubt, the Administrative Agent shall not be required to deliver a Term SOFR Notice after a Term SOFR Transition Event and may elect or not elect to do so in its sole discretion.
- (ii) <u>Benchmark Replacement Conforming Changes</u>. In connection with the implementation of a Benchmark Replacement, the Administrative Agent will have the right to make Benchmark Replacement Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Loan Document, any amendments implementing such Benchmark Replacement Conforming Changes will become effective without any further action or consent of any other party to this Agreement or any other Loan Document.
- (iii) Notices; Standards for Decisions and Determinations. The Administrative Agent will promptly notify the Borrower and the Lenders of (A) any occurrence of a Benchmark Transition Event, a Term SOFR Transition Event, an Early Opt-in Election or an Other Benchmark Rate Election, as applicable, and its related Benchmark Replacement Date, (B) the implementation of any Benchmark Replacement, (C) the effectiveness of any Benchmark Replacement Conforming Changes, (D) the removal or reinstatement of any tenor of a Benchmark pursuant to Section 4.8(c)(iv) below and (E) the commencement or conclusion of any Benchmark Unavailability Period. Any determination, decision or election that may be made by the Administrative Agent or, if applicable, any Lender (or group of Lenders) pursuant to this Section 4.8(c), including any determination with respect to a tenor, rate or adjustment or of the occurrence or non-occurrence of an event, circumstance or date and any decision to take or

refrain from taking any action or any selection, will be conclusive and binding absent manifest error and may be made in its or their sole discretion and without consent from any other party to this Agreement or any other Loan Document, except, in each case, as expressly required pursuant to this Section 4.8(c).

- (iv) <u>Unavailability of Tenor of Benchmark</u>. Notwithstanding anything to the contrary herein or in any other Loan Document, at any time (including in connection with the implementation of a Benchmark Replacement), (A) if the then-current Benchmark is a term rate (including Term SOFR or USD LIBOR) and either (1) any tenor for such Benchmark is not displayed on a screen or other information service that publishes such rate from time to time as selected by the Administrative Agent in its reasonable discretion or (2) the regulatory supervisor for the administrator of such Benchmark has provided a public statement or publication of information announcing that any tenor for such Benchmark is or will be no longer representative, then the Administrative Agent may modify the definition of "Interest Period" for any Benchmark settings at or after such time to remove such unavailable or non-representative tenor and (B) if a tenor that was removed pursuant to clause (A) above either (1) is subsequently displayed on a screen or information service for a Benchmark (including a Benchmark Replacement) or (2) is not, or is no longer, subject to an announcement that it is or will no longer be representative for a Benchmark (including a Benchmark Replacement), then the Administrative Agent may modify the definition of "Interest Period" for all Benchmark settings at or after such time to reinstate such previously removed tenor.
- (v) <u>Benchmark Unavailability Period</u>. Upon the Borrower's receipt of notice of the commencement of a Benchmark Unavailability Period, the Borrower may revoke any request for a borrowing of, conversion to or continuation of LIBOR Borrowings to be made, converted or continued during any Benchmark Unavailability Period and, failing that, the Borrower will be deemed to have converted any such request into a request for a borrowing of or conversion to Adjusted Base Rate Borrowings that shall, to the extent permissible under Applicable Law, bear interest at the Substitute Rate. During any Benchmark Unavailability Period, Borrowings bearing interest based on the Adjusted Base Rate shall instead bear interest at the Substitute Rate.
- (vi) London Interbank Offered Rate Benchmark Transition Event. On March 5, 2021, the IBA, the administrator of the London interbank offered rate, and the FCA, the regulatory supervisor of the IBA, made the Announcements that the final publication or representativeness date for Dollars for (I) 1-week and 2-month London interbank offered rate tenor settings will be December 31, 2021 and (II) overnight, 1-month, 3-month, 6-month and 12-month London interbank offered rate tenor settings will be June 30, 2023. No successor administrator for the IBA was identified in such Announcements. The parties hereto agree and acknowledge that the Announcements resulted in the occurrence of a Benchmark Transition Event with respect to the London interbank offered rate pursuant to the terms of this Agreement and that any obligation of the Administrative Agent to notify any parties of such Benchmark Transition Event pursuant to clause (iii) of this Section 4.8(c) shall be deemed satisfied.

ARTICLE V.

CONDITIONS PRECEDENT

5.1. <u>Conditions Precedent to Closing</u>. The Lenders shall not be required to make the initial Borrowing hereunder, nor shall any Issuing Bank be required to issue the initial Facility Letter of Credit hereunder, unless (i) the Borrower shall have paid all fees then due and payable to the Lenders, BofA

Securities, PNC Capital Markets, WFS and the Administrative Agent hereunder, including reasonable legal fees and out-of-pocket expenses, (ii) all of the conditions set forth in Section 5.2 are satisfied, and (iii) the Borrower shall have furnished to the Administrative Agent, in form and substance satisfactory to the Lenders and their counsel and with sufficient copies for the Lenders, the following:

- (a) <u>Certificates of Limited Partnership/Incorporation</u>. A copy of the Certificate of Limited Partnership for the Borrower and a copy of the articles of incorporation of the General Partner, each certified by the appropriate Secretary of State or equivalent state official.
- (b) <u>Agreements of Limited Partnership/Bylaws</u>. A copy of the Agreement of Limited Partnership for the Borrower and a copy of the bylaws of the General Partner, including all amendments thereto, each certified by the Secretary or an Assistant Secretary of the General Partner as being in full force and effect on the Agreement Execution Date.
- (c) <u>Good Standing Certificates</u>. A certified copy of a certificate from the Secretary of State or equivalent state official of the states where the Borrower and the General Partner are organized, dated as of the most recent practicable date, showing the good standing or partnership qualification (if issued) of (i) the Borrower, and (ii) the General Partner.
- (d) <u>Foreign Qualification Certificates</u>. A certified copy of a certificate from the Secretary of State or equivalent state official of the state where the Borrower and the General Partner maintain their principal place of business, dated as of the most recent practicable date, showing the qualification to transact business in such state as a foreign limited partnership or foreign corporation, as the case may be, for (i) the Borrower, and (ii) the General Partner.
- (e) <u>Resolutions</u>. A copy of a resolution or resolutions adopted by the Board of Directors of the General Partner, certified by the Secretary or an Assistant Secretary of the General Partner as being in full force and effect on the Agreement Execution Date, authorizing the Borrowings provided for herein and the execution, delivery and performance of the Loan Documents by the General Partner to be executed and delivered by it hereunder on behalf of itself and the Borrower.
- (f) <u>Incumbency Certificate</u>. A certificate, signed by the Secretary or an Assistant Secretary of the General Partner and dated the Agreement Execution Date, as to the incumbency, and containing the specimen signature or signatures, of the Persons authorized to execute and deliver the Loan Documents to be executed and delivered by it and the Borrower hereunder.
- (g) <u>Loan Documents</u>. (i) From each party hereto either (A) a counterpart of this Agreement signed on behalf of such party or (B) written evidence satisfactory to the Administrative Agent (which may include telecopy or electronic transmission of a signed signature page of this Agreement) that such party has signed a counterpart of this Agreement, (ii) duly executed copies of the other Loan Documents and (iii) the Departing Lenders shall have been repaid in full as contemplated by <u>Section 1.6</u>.
- (g) <u>Opinion of Borrower's Counsel</u>. A written opinion, dated the Agreement Execution Date, from outside counsel for the Borrower which counsel is reasonably satisfactory to Administrative Agent, substantially in the form attached hereto as <u>Exhibit E</u>.
- (h) <u>Opinion of General Partner's Counsel</u>. A written opinion, dated the Agreement Execution Date, from outside counsel for the General Partner which counsel is reasonably satisfactory to Administrative Agent, substantially in the form attached hereto as <u>Exhibit F.</u>

(i) Financial and Related Information. The following information:

- (i) The most recent financial statements of the Borrower and the General Partner and a certificate from a Qualified Officer of the Borrower that no change in the Borrower's financial condition that would have a Material Adverse Effect has occurred since December 31, 2020; and
- (ii) A Disbursement Instruction Agreement, in substantially the form of <u>Exhibit C</u> hereto, addressed to the Administrative Agent and signed by a Qualified Officer, together with such other related money transfer authorizations as the Administrative Agent may have reasonably requested.
- (j) <u>Closing Certificate</u>. A certificate, signed by an officer of the Borrower, stating that on the Agreement Execution Date no Default or Event of Default has occurred and is continuing and that all representations and warranties of the Borrower contained herein are true and correct as of the Agreement Execution Date as and to the extent set forth herein.

(k) [Reserved].

- (l) <u>PATRIOT Act, etc.</u> At least five (5) Business Days prior to the Closing Date, (i) all documentation and other information reasonably requested by the Administrative Agent or any Lender or required by regulatory authorities in order for the Administrative Agent and the Lenders to comply with requirements of any Anti-Corruption Laws, including the PATRIOT Act and any applicable "know your customer" rules and regulations, and (ii) to the extent requested by the Administrative Agent or any Lender, a Beneficial Ownership Certification in relation to the Borrower.
- (m) Other Evidence as any Lender May Require. Such other evidence as any Lender may reasonably request to establish the consummation of the transactions contemplated hereby, the taking of all necessary actions in any proceedings in connection herewith and compliance with the conditions set forth in this Agreement.

When all such conditions have been fulfilled (or, in the Lenders' sole discretion, waived by Lenders), the Administrative Agent shall confirm in writing to the Borrower that the initial Borrowing is then available to the Borrower hereunder.

- 5.2. <u>Conditions Precedent to Subsequent Borrowings</u>. Borrowings after the initial Borrowing (other than conversions or continuations) shall be made from time to time as requested by the Borrower, and the obligation of each Lender to make any Borrowing (including Swingline Loans and Bid Rate Loans) and the obligation of any Issuing Bank to issue, renew or extend a Facility Letter of Credit is subject to the following terms and conditions:
 - (a) prior to and after giving effect to each such Borrowing or Facility Letter of Credit issuance, renewal or extension, no Default or Event of Default shall have occurred and be continuing under this Agreement or any of the Loan Documents, and the Borrower shall deliver a certificate of the Borrower to such effect; and
 - (b) The representations and warranties contained in <u>Article VI</u> and <u>VII</u> are true and correct as of such borrowing date, Issuance Date (or date of renewal or extension of a Facility Letter of Credit), as and to the extent set forth therein, except to the extent any such

representation or warranty is stated to relate solely to an earlier date, in which case such representation or warranty shall be true and correct on and as of such earlier date.

Subject to the last grammatical paragraphs of <u>Article VI</u> and <u>VII</u> hereof, each Borrowing Notice and Letter of Credit Request shall constitute a representation and warranty by the Borrower that the conditions contained in <u>Sections 5.2(a)</u> and <u>(b)</u> have been satisfied. Unless set forth in writing to the contrary, the making of its initial Loan by a Lender shall constitute a certification by such Lender to the Administrative Agent and the other Lenders that the conditions precedent for initial Loans set forth in <u>Section 5.1</u> and <u>Section 5.2</u> (if applicable) that have not previously been waived by the Lenders in accordance with the terms of this Agreement have been satisfied.

ARTICLE VI.

REPRESENTATIONS AND WARRANTIES

The Borrower hereby represents and warrants to the Lenders that:

- 6.1. Existence. The Borrower is a limited partnership duly organized and existing under the laws of the State of Delaware, with its principal place of business in the State of Illinois, and is duly qualified as a foreign limited partnership, properly licensed (if required), in good standing and has all requisite authority to conduct its business in each jurisdiction in which it owns Properties and, except where the failure to be so qualified or to obtain such authority would not have a Material Adverse Effect, in each other jurisdiction in which its business is conducted. Each of its Subsidiaries is duly organized, validly existing and in good standing under the laws of its jurisdiction of organization and has all requisite authority to conduct its business in each jurisdiction in which it owns Property, and except where the failure to be so qualified or to obtain such authority would not have a Material Adverse Effect, in each other jurisdiction in which it conducts business.
- 6.2. <u>Corporate/Partnership Powers</u>. The execution, delivery and performance of the Loan Documents required to be delivered by the Borrower hereunder are within the partnership authority of such entity and the corporate powers of the general partners of such entity, have been duly authorized by all requisite action, and are not in conflict with the terms of any organizational instruments of such entity, or any instrument or agreement to which the Borrower or the General Partner is a party or by which the Borrower, the General Partner or any of their respective assets may be bound or affected.
- 6.3. <u>Power of Officers</u>. The officers of the General Partner executing the Loan Documents required to be delivered by such entities hereunder have been duly elected or appointed and were fully authorized to execute the same at the time each such agreement, certificate or instrument was executed.
- 6.4. <u>Government and Other Approvals</u>. No approval, consent, exemption or other action by, or notice to or filing with, any governmental authority is necessary in connection with the execution, delivery or performance of the Loan Documents required hereunder.

6.5. Solvency.

(i) Immediately after the Agreement Execution Date and immediately following the making of each Loan and after giving effect to the application of the proceeds of such Loans, (a) the fair value of the assets of the Borrower and its Subsidiaries on a consolidated basis, at a fair valuation, will exceed the debts and liabilities, subordinated, contingent or otherwise, of the Borrower and its Subsidiaries on a consolidated basis; (b) the present fair saleable value of the Properties of the Borrower

and its Subsidiaries on a consolidated basis will be greater than the amount that will be required to pay the probable liability of the Borrower and its Subsidiaries on a consolidated basis on their debts and other liabilities, subordinated, contingent or otherwise, as such debts and other liabilities become absolute and matured; (c) the Borrower and its Subsidiaries on a consolidated basis will be able to pay their debts and liabilities, subordinated, contingent or otherwise, as such debts and liabilities become absolute and matured; and (d) the Borrower and its Subsidiaries on a consolidated basis will not have unreasonably small capital with which to conduct the businesses in which they are engaged as such businesses are now conducted and are proposed to be conducted after the date hereof.

- (ii) The Borrower does not intend to, or to permit any of its Subsidiaries to incur debts beyond its ability to pay such debts as they mature, taking into account the timing of and amounts of cash to be received by it or any such Subsidiary and the timing of the amounts of cash to be payable on or in respect of its Indebtedness or the Indebtedness of any such Subsidiary.
- 6.6. <u>Compliance With Laws</u>. There is no judgment, decree or order or any law, rule or regulation of any court or governmental authority binding on the Borrower or any of its Subsidiaries which would be contravened by the execution, delivery or performance of the Loan Documents required hereunder.
- 6.7. <u>Enforceability of Agreement</u>. This Agreement is the legal, valid and binding agreement of the Borrower, and the Notes when executed and delivered will be the legal, valid and binding obligations of the Borrower, enforceable against the Borrower in accordance with their respective terms, and the Loan Documents required hereunder, when executed and delivered, will be similarly legal, valid, binding and enforceable except to the extent that such enforcement may be limited by applicable bankruptcy, insolvency, reorganization or other similar laws affecting the rights of creditors generally.
- 6.8. <u>Title to Property</u>. To the best of the Borrower's knowledge after due inquiry, the Borrower or its Subsidiaries has good and marketable title to the Properties and assets reflected in the financial statements as owned by it or any such Subsidiary free and clear of Liens except for the Permitted Liens. The execution, delivery or performance of the Loan Documents required to be delivered by the Borrower hereunder will not result in the creation of any Lien on the Properties. No consent to the transactions contemplated hereunder is required from any ground lessor or mortgagee or beneficiary under a deed of trust or any other party except as has been delivered to the Lenders.
- 6.9. <u>Litigation</u>. There are no suits, arbitrations, claims, disputes or other proceedings (including, without limitation, any civil, criminal, administrative or environmental proceedings), pending or, to the best of the Borrower's knowledge, threatened against or affecting the Borrower or any of the Properties, the adverse determination of which individually or in the aggregate would have a Material Adverse Effect on the Borrower and/or would cause a Material Adverse Financial Change of the Borrower or materially impair the Borrower's ability to perform its obligations hereunder or under any instrument or agreement required hereunder, except as disclosed on Schedule 6.9 hereto, or otherwise disclosed to Lenders in accordance with the terms hereof.
- 6.10. Events of Default. No Default or Event of Default has occurred and is continuing or would result from the incurring of obligations by the Borrower under any of the Loan Documents or any other document to which the Borrower is a party.

- 6.11. <u>Investment Company Act of 1940</u>. The Borrower is not and will by such acts as may be necessary continue not to be, an investment company within the meaning of the Investment Company Act of 1940.
- 6.12. <u>Public Utility Holding Company Act</u>. The Borrower is not a "holding company" or a "subsidiary company" of a "holding company," or an "affiliate" of a "holding company," or of a "subsidiary company" of a "holding company," within the definitions of the Public Utility Holding Company Act of 1935, as amended.
- 6.13. <u>Regulation U</u>. The proceeds of the Borrowings will not be used, directly or indirectly, in a manner which would cause the Facility to be treated as a "Purpose Credit."
- 6.14. <u>No Material Adverse Financial Change</u>. There has been no Material Adverse Financial Change in the condition of the Borrower since December 31, 2020.
- 6.15. <u>Financial Information</u>. All financial statements furnished to the Lenders by or at the direction of the Borrower and all other financial information and data furnished by the Borrower to the Lenders are complete and correct in all material respects as of the date thereof, and such financial statements have been prepared in accordance with GAAP and fairly present the consolidated financial condition and results of operations of the Borrower as of such date. The Borrower has no contingent obligations, liabilities for taxes or other outstanding financial obligations which are material in the aggregate, except as disclosed in such statements, information and data.
- 6.16. <u>Factual Information</u>. All factual information heretofore or contemporaneously furnished by or on behalf of the Borrower to the Lenders for purposes of or in connection with this Agreement and the other Loan Documents and the transactions contemplated therein is, and all other such factual information hereafter furnished by or on behalf of the Borrower to the Lenders will be, true and accurate (taken as a whole) in all material respects on the date as of which such information is dated or certified and not incomplete by omitting to state any material fact necessary to make such information (taken as a whole) not misleading at such time. As of the Closing Date, all of the information included in any Beneficial Ownership Certification is true and correct.
- 6.17. <u>ERISA</u>. (i) The Borrower is not an entity deemed to hold "plan assets" within the meaning of ERISA or any regulations promulgated thereunder of an employee benefit plan (as defined in Section 3(3) of ERISA) which is subject to Title I of ERISA or any plan within the meaning of Section 4975 of the Code, and (ii) the execution of this Agreement and the transactions contemplated hereunder do not give rise to a prohibited transaction within the meaning of Section 406 of ERISA or Section 4975 of the Code.
- 6.18. <u>Taxes</u>. All required Federal, state and other material tax returns have been filed by the Borrower with the appropriate authorities except (i) to the extent that extensions of time to file have been requested, granted and have not expired, and (ii) to the extent such taxes are being contested in good faith and for which adequate reserves, in accordance with GAAP, are being maintained.
- 6.19. <u>Environmental Matters</u>. Except as disclosed in Schedule 6.19, each of the following representations and warranties is true and correct except to the extent that the facts and circumstances giving rise to any such failure to be so true and correct, in the aggregate, could not reasonably be expected to have a Material Adverse Effect:
 - (i) To the knowledge of the Borrower, the Properties of the Borrower, its Subsidiaries, and Investment Affiliates do not contain any Materials of Environmental

Concern in amounts or concentrations which constitute a violation of, or could reasonably give rise to liability under, Environmental Laws.

- (ii) The Borrower has not received any written notice alleging that any or all of the Properties of the Borrower and its Subsidiaries and Investment Affiliates and all operations at the Properties are not currently in compliance with all applicable Environmental Laws. Further, the Borrower has not received any written notice alleging the current existence of any contamination at or under such Properties in amounts or concentrations which constitute a violation of any Environmental Law, or any violation of any Environmental Law with respect to such Properties for which the Borrower, its Subsidiaries or Investment Affiliates is or could be liable.
- (iii) Neither the Borrower nor any of its Subsidiaries or Investment Affiliates has received any written notice of current non-compliance, liability or potential liability regarding Environmental Laws with regard to any of the Properties, nor does it have knowledge that any such notice will be received or is being threatened.
- (iv) To the knowledge of the Borrower during the ownership of the Properties by any or all of the Borrower, its Subsidiaries and Investment Affiliates, Materials of Environmental Concern have not been transported or disposed of from the Properties of the Borrower and its Subsidiaries and Investment Affiliates in violation of, or in a manner or to a location which could reasonably give rise to liability of the Borrower, any Subsidiary, or any Investment Affiliate under, Environmental Laws, nor during the ownership of the Properties by any or all of the Borrower, its Subsidiaries and Investment Affiliates have any Materials of Environmental Concern been generated, treated, stored or disposed of at, on or under any of such Properties in violation of, or in a manner that could give rise to liability of the Borrower, any Subsidiary or any Investment Affiliate under, any applicable Environmental Laws.
- (v) No judicial proceedings or governmental or administrative action is pending, or, to the knowledge of the Borrower, threatened, under any Environmental Law to which the Borrower, any of its Subsidiaries, or any Investment Affiliate, is named as a party with respect to the Properties of such entity, nor are there any consent decrees or other decrees, consent orders, administrative order or other orders, or other administrative or judicial requirements outstanding under any Environmental Law with respect to such Properties for which the Borrower, its Subsidiaries, or any Investment Affiliate is or could be liable.
- (vi) To the knowledge of the Borrower during the ownership of the Properties by any or all of the Borrower, its Subsidiaries and Investment Affiliates, there has been no release or threat of release of Materials of Environmental Concern at or from the Properties of the Borrower and its Subsidiaries and Investment Affiliates, or arising from or related to the operations of such entity in connection with the Properties in violation of or in amounts or in a manner that could give rise to liability under Environmental Laws.
- 6.20. <u>Insurance</u>. The Borrower maintains insurance on its properties consistent with the insurance maintained by other institutional owners of similar properties.
- 6.21. <u>No Brokers</u>. The Borrower has dealt with no brokers in connection with this Facility, and no brokerage fees or commissions are payable by or to any Person in connection with this Agreement or

the Borrowings. Lenders shall not be responsible for the payment of any fees or commissions to any broker and the Borrower shall indemnify, defend and hold Lenders harmless from and against any claims, liabilities, obligations, damages, costs and expenses (including reasonable attorneys' fees and disbursements) made against or incurred by Lenders as a result of claims made or actions instituted by any broker or Person claiming by, through or under the Borrower in connection with the Facility.

- 6.22. <u>No Violation of Usury Laws</u>. No aspect of any of the transactions contemplated herein violate or will violate any usury laws or laws regarding the validity of agreements to pay interest in effect on the date hereof.
- 6.23. <u>Not a Foreign Person</u>. The Borrower is not a "foreign person" within the meaning of Section 1445 or 7701 of the Internal Revenue Code.
- 6.24. <u>No Trade Name</u>. Except for the name "First Industrial," and except as otherwise set forth on Schedule 6.24 attached hereto, the Borrower does not use any trade name and has not and does not do business under any name other than their actual names set forth herein. The principal place of business of the Borrower is as stated in the recitals hereto.
- 6.25. <u>Subsidiaries</u>. Schedule 6.25 hereto contains an accurate list of all of the presently existing Subsidiaries of the Borrower, setting forth the percentage of their respective Capital Stock owned by it or its Subsidiaries. All of the issued and outstanding shares of Capital Stock of such Subsidiaries have been duly authorized and issued and are fully paid and non-assessable.
- 6.26. <u>Unencumbered Assets</u>. Schedule 6.26 hereto contains a complete and accurate description of Unencumbered Assets as of March 31, 2021 and as supplemented from time to time including the entity that owns each Unencumbered Asset. With respect to each Project identified from time to time as an Unencumbered Asset, the Borrower hereby represents and warrants as follows except to the extent disclosed in writing to the Lenders and approved by the Required Lenders (which approval shall not be unreasonably withheld) or except to the extent the failure of such representation and warranty to be true would not materially adversely affect the use and operation of such Project for its intended use or its marketability or value:
 - (a) No portion of any improvement on the Unencumbered Asset is located in an area identified by the Secretary of Housing and Urban Development or any successor thereto as an area having special flood hazards pursuant to the National Flood Insurance Act of 1968 or the Flood Disaster Protection Act of 1973, as amended, or any successor law, or, if located within any such area, the Borrower has obtained and will maintain the insurance prescribed in <u>Section 6.20</u> hereof.
 - (b) To the Borrower's knowledge, the Unencumbered Asset and the present use and occupancy thereof are in material compliance with all Applicable Laws (including all Environmental Laws).
 - (c) The Unencumbered Asset is served by all utilities required for the current or contemplated use thereof. All utility service is provided by public utilities and the Unencumbered Asset has accepted or is equipped to accept such utility service.
 - (d) All public roads and streets necessary for service of and access to the Unencumbered Asset for the current or contemplated use thereof have been completed, are serviceable and all-weather and are physically and legally open for use by the public.

- (e) The Unencumbered Asset is served by public water and sewer systems or, if the Unencumbered Asset is not serviced by a public water and sewer system, such alternate systems are adequate and meet, in all material respects, all requirements and regulations of, and otherwise complies in all material respects with, all Applicable Laws with respect to such alternate systems.
- (f) The Borrower is not aware of any latent or patent structural or other significant deficiency of the Unencumbered Asset. The Unencumbered Asset is free of damage and waste that would materially and adversely affect the value of the Unencumbered Asset, is in good repair and there is no deferred maintenance other than ordinary wear and tear. The Unencumbered Asset is free from damage caused by fire or other casualty. There is no pending or, to the actual knowledge of the Borrower threatened condemnation proceedings affecting the Unencumbered Asset, or any material part thereof.
- (g) To the Borrower's knowledge, all liquid and solid waste disposal, septic and sewer systems located on the Unencumbered Asset are in a good and safe condition and repair and to the Borrower's knowledge, in material compliance with all Applicable Laws with respect to such systems.
- (h) All improvements on the Unencumbered Asset lie within the boundaries and building restrictions of the legal description of record of the Unencumbered Asset, no such improvements encroach upon easements benefiting the Unencumbered Asset other than encroachments that do not materially adversely affect the use or occupancy of the Unencumbered Asset and no improvements on adjoining properties encroach upon the Unencumbered Asset or easements benefiting the Unencumbered Asset other than encroachments that do not materially adversely affect the use or occupancy of the Unencumbered Asset. All amenities, access routes or other items that materially benefit the Unencumbered Asset are under direct control of the Borrower, constitute permanent easements that benefit all or part of the Unencumbered Asset or are public property, and the Unencumbered Asset, by virtue of such easements or otherwise, is contiguous to a physically open, dedicated all weather public street, and has the necessary permits for ingress and egress.
- (i) There are no delinquent taxes, ground rents, water charges, sewer rents, assessments, insurance premiums, leasehold payments, or other outstanding charges affecting the Unencumbered Asset except to the extent such items are being contested in good faith and as to which adequate reserves have been provided.
 - (j) The Unencumbered Asset satisfies each of the requirements for an Unencumbered Asset as set forth in the definition thereof.

A breach of any of the representations and warranties contained in this <u>Section 6.26</u> with respect to a Project shall disqualify such Project from being an Unencumbered Asset for so long as such breach continues (unless otherwise approved by the Required Lenders) but shall not constitute a Default (unless the elimination of such Property as an Unencumbered Asset results in a Default under one of the other provisions of this Agreement).

6.27. <u>Anti-Corruption Laws and Sanctions</u>.

None of the Borrower or any of its Subsidiaries, nor, to the knowledge of the Borrower, any of the directors, officers, employees or agents of the Borrower or of any of its Subsidiaries (i) is a Sanctioned Person, or (ii) to the knowledge of the Borrower, except as may be disclosed by the Borrower to the Administrative Agent from time to time, derives any of its assets or operating income from

investments in or transactions with any Sanctioned Person. The Borrower and its Subsidiaries and, to the knowledge of the Borrower, their respective officers, directors, employees and agents, are in compliance with Anti-Corruption Laws and applicable Sanctions in all material respects. No Loan or Facility Letter of Credit, use of the proceeds of any Loan or Facility Letter of Credit, or other transactions contemplated hereby will violate (x) Anti-Corruption Laws or (y) any Sanctions applicable to any party hereto. Neither the making of the Loans nor the use of the proceeds thereof will violate the Patriot Act, the Trading with the Enemy Act, as amended, or any of the foreign assets control regulations of the United States Treasury Department (31 C.F.R., Subtitle B, Chapter V, as amended) or any enabling legislation or executive order relating thereto or successor statute thereto. The Borrower and its Subsidiaries are in compliance in all material respects with the Patriot Act.

The Borrower agrees that all of its representations and warranties set forth in <u>Article VI</u> of this Agreement and elsewhere in this Agreement are true on the Agreement Execution Date, and will be true on each Effective Date in all material respects (except with respect to matters which have been disclosed in writing to and approved by the Required Lenders), and will be true in all material respects (except with respect to matters which have been disclosed in writing to and approved by the Required Lenders) upon the date which any extension of the Maturity Date is effectuated pursuant to <u>Section 2.19</u> and upon the date of each request for disbursement of a Borrowing, <u>provided</u> that the Borrower shall only be obligated to update any Schedules referred to in this <u>Article VI</u> and the financial statements required under <u>Section 8.2(i)</u> on a quarterly basis, unless any change otherwise required to be disclosed could reasonably be expected to have a Material Adverse Effect. Each request for disbursement and any request for an extension of the Maturity Date pursuant to <u>Section 2.19</u> hereunder shall constitute a reaffirmation of such representations and warranties as deemed modified in accordance with the disclosures made and approved, as aforesaid, as of the date of such request and disbursement.

6.28. <u>Affected Financial Institutions</u>. No Loan Party is an Affected Financial Institution.

ARTICLE VII.

ADDITIONAL REPRESENTATIONS AND WARRANTIES

The General Partner hereby represents and warrants to the Lenders that:

- 7.1. <u>Existence</u>. The General Partner is a corporation duly organized and existing under the laws of the State of Maryland, with its principal place of business in the State of Illinois, is duly qualified as a foreign corporation and properly licensed (if required) and in good standing in each jurisdiction where the failure to qualify or be licensed (if required) would constitute a Material Adverse Financial Change with respect to the General Partner or have a Material Adverse Effect on the business or properties of the General Partner.
- 7.2. <u>Corporate Powers</u>. The execution, delivery and performance of the Loan Documents required to be delivered by the General Partner hereunder are within the corporate powers of the General Partner, have been duly authorized by all requisite corporate action, and are not in conflict with the terms of any organizational instruments of the General Partner, or any instrument or agreement to which the General Partner is a party or by which the General Partner or any of its assets is bound or affected.
- 7.3. <u>Power of Officers</u>. The officers of the General Partner executing the Loan Documents required to be delivered by the General Partner hereunder have been duly elected or appointed and were fully authorized to execute the same at the time each such agreement, certificate or instrument was executed.

- 7.4. <u>Government and Other Approvals</u>. No approval, consent, exemption or other action by, or notice to or filing with, any governmental authority is necessary in connection with the execution, delivery or performance of the Loan Documents required hereunder.
- 7.5. <u>Compliance With Laws</u>. There is no judgment, decree or order or any law, rule or regulation of any court or governmental authority binding on the General Partner which would be contravened by the execution, delivery or performance of the Loan Documents required hereunder.
- 7.6. <u>Enforceability of Agreement</u>. This Agreement is the legal, valid and binding agreement of the General Partner, as the general partner of the Borrower, enforceable against the General Partner in accordance with its respective terms, and the Loan Documents required hereunder, when executed and delivered, will be similarly legal, valid, binding and enforceable except to the extent that such enforcement may be limited by applicable bankruptcy, insolvency, reorganization or other similar laws affecting the rights of creditors generally.
- 7.7. <u>Liens; Consents</u>. The execution, delivery or performance of the Loan Documents required to be delivered by the General Partner hereunder will not result in the creation of any Lien on the Properties other than in favor of the Lenders. No consent to the transactions hereunder is required from any ground lessor or mortgagee or beneficiary under a deed of trust or any other party except as has been delivered to the Lenders.
- 7.8. <u>Litigation</u>. There are no suits, arbitrations, claims, disputes or other proceedings (including, without limitation, any civil, criminal, administrative or environmental proceedings), pending or, to the best of the General Partner's knowledge, threatened against or affecting the General Partner or any of the Properties, the adverse determination of which individually or in the aggregate would have a Material Adverse Effect on the General Partner and/or would cause a Material Adverse Financial Change with respect to the General Partner or materially impair the General Partner's ability to perform its obligations hereunder or under any instrument or agreement required hereunder, except as disclosed on Schedule 7.8 hereto, or otherwise disclosed to Lenders in accordance with the terms hereof.
- 7.9. <u>Events of Default</u>. No Default or Event of Default has occurred and is continuing or would result from the incurring of obligations by the General Partner under any of the Loan Documents or any other document to which the General Partner is a party.
- 7.10. <u>Investment Company Act of 1940</u>. The General Partner is not, and will by such acts as may be necessary continue not to be, an investment company within the meaning of the Investment Company Act of 1940.
- 7.11. <u>Public Utility Holding Company Act</u>. The General Partner is not a "holding company" or a "subsidiary company" of a "holding company," or an "affiliate" of a "holding company," or of a "subsidiary company" of a "holding company," within the definitions of the Public Utility Holding Company Act of 1935, as amended.
- 7.12. <u>No Material Adverse Financial Change</u>. There has been no Material Adverse Financial Change in the condition of the General Partner since December 31, 2020.
- 7.13. <u>Financial Information</u>. All financial statements furnished to the Lenders by or on behalf of the General Partner and all other financial information and data furnished by or on behalf of the General Partner to the Lenders are complete and correct in all material respects as of the date thereof, and such financial statements have been prepared in accordance with GAAP and fairly present the consolidated financial condition and results of operations of the General Partner as of such date. The

General Partner has no contingent obligations, liabilities for taxes or other outstanding financial obligations which are material in the aggregate, except as disclosed in such statements, information and data.

- 7.14. <u>Factual Information</u>. All factual information heretofore or contemporaneously furnished by or on behalf of the General Partner to the Lenders for purposes of or in connection with this Agreement and the other Loan Documents and the transactions contemplated therein is, and all other such factual information hereafter furnished by or on behalf of the General Partner to the Lenders will be, true and accurate in all material respects (taken as a whole) on the date as of which such information is dated or certified and not incomplete by omitting to state any material fact necessary to make such information (taken as a whole) not misleading at such time.
- 7.15. <u>ERISA</u>. (i) The General Partner is not an entity deemed to hold "plan assets" within the meaning of ERISA or any regulations promulgated thereunder of an employee benefit plan (as defined in Section 3(3) of ERISA) which is subject to Title I of ERISA or any plan within the meaning of Section 4975 of the Code, and (ii) the execution of this Agreement and the transactions contemplated hereunder do not give rise to a prohibited transaction within the meaning of Section 406 of ERISA or Section 4975 of the Code.
- 7.16. <u>Taxes</u>. All required Federal, state and other material tax returns have been filed by the General Partner with the appropriate authorities except (i) to the extent that extensions of time to file have been requested, granted and have not expired and (ii) to the extent such taxes are being contested in good faith and for which adequate reserves, in accordance with GAAP, are being maintained.
- 7.17. No Brokers. The General Partner has dealt with no brokers in connection with this Facility, and no brokerage fees or commissions are payable by or to any Person in connection with this Agreement or the Borrowings. Lenders shall not be responsible for the payment of any fees or commissions to any broker and the General Partner shall indemnify, defend and hold Lender harmless from and against any claims, liabilities, obligations, damages, costs and expenses (including reasonable attorneys' fees and disbursements) made against or incurred by Lender as a result of claims made or actions instituted by any broker or Person claiming by, through or under the General Partner in connection with the Facility.
- 7.18. <u>Subsidiaries</u>. Schedule 7.18 hereto contains an accurate list of all of the presently existing Subsidiaries of the General Partner, setting forth their respective jurisdictions of formation and the percentage of their respective Capital Stock owned by it or its Subsidiaries. All of the issued and outstanding shares of Capital Stock of such Subsidiaries have been duly authorized and issued and are fully paid and non-assessable.
- 7.19. <u>Status</u>. General Partner is a corporation listed and in good standing on a national securities exchange and is currently qualified as a real estate investment trust under the Code.
- 7.20. Anti-Corruption Laws and Sanctions. None of the General Partner, the Borrower, any of the other Subsidiaries of the General Partner, or, to the knowledge of the General Partner, any other Affiliate of the General Partner or any of their respective directors, officers, employees or agents: (i) is a Sanctioned Person; or (ii) to the knowledge of the General Partner, except as may be disclosed by the General Partner to the Administrative Agent from time to time, derives any of its assets or operating income from investments in or transactions with any Sanctioned Person. The General Partner and its Affiliates and, to the knowledge of the General Partner, their respective officers, directors, employees and agents, are in compliance with Anti-Corruption Laws and applicable Sanctions in all material respects. No Loan or Facility Letter of Credit, use of the proceeds of any Loan or Facility Letter of Credit, or other

transactions contemplated hereby will violate (x) Anti-Corruption Laws or (y) any Sanctions applicable to any party hereto. Neither the making of the Loans nor the use of the proceeds thereof will violate the Patriot Act, the Trading with the Enemy Act, as amended, or any of the foreign assets control regulations of the United States Treasury Department (31 C.F.R., Subtitle B, Chapter V, as amended) or any enabling legislation or executive order relating thereto or successor statute thereto. The Borrower and its Subsidiaries are in compliance in all material respects with the Patriot Act.

The General Partner agrees that all of its representations and warranties set forth in <u>Article VII</u> of this Agreement and elsewhere in this Agreement are true as of the Agreement Execution Date, and will be true on each Effective Date in all material respects (except with respect to matters which have been disclosed in writing to and approved by the Required Lenders), and will be true in all material respects (except with respect to matters which have been disclosed in writing to and approved by the Required Lenders) upon each request for disbursement of a Borrowing, <u>provided</u> that the General Partner shall only be obligated to update any Schedules referred to in this <u>Article VII</u> and the financial statements required under <u>Section 8.2(i)</u> on a quarterly basis, unless any change otherwise required to be disclosed could reasonably be expected to have a Material Adverse Effect. Each request for disbursement hereunder shall constitute a reaffirmation of such representations and warranties as deemed modified in accordance with the disclosures made and approved, as aforesaid, as of the date of such request and disbursement.

ARTICLE VIII.

AFFIRMATIVE COVENANTS

The Borrower (and the General Partner, if expressly included in Sections contained in this Article) covenant and agree that so long as the Revolving Commitment of any Lender shall remain available and until the full and final payment of all Obligations incurred under the Loan Documents each will:

- 8.1. Notices. Promptly give written notice to Administrative Agent (who will promptly send such notice to Lenders) of:
- (a) all litigation or arbitration proceedings affecting the Borrower, the General Partner or any Subsidiary where the amount claimed is \$50,000,000 or more:
- (b) any Default or Event of Default, specifying the nature and the period of existence thereof and what action has been taken or been proposed to be taken with respect thereto;
- (c) all claims filed against any property owned by the Borrower or the General Partner which, if adversely determined, could have a Material Adverse Effect on the ability of the Borrower or the General Partner to meet any of their obligations under the Loan Documents:
- (d) the occurrence of any other event which might have a Material Adverse Effect or cause a Material Adverse Financial Change on or with respect to the Borrower or the General Partner;
- (e) any Reportable Event or any "prohibited transaction" (as such term is defined in Section 4975 of the Code) in connection with any Plan or any trust created thereunder, which may, singly or in the aggregate materially impair the ability of the Borrower or the General Partner to repay any of its obligations under the Loan Documents, describing the nature of each such event and the action, if any, the Borrower or the General Partner, as the case may be, proposes to take with respect thereto;

- (f) any notice from any Federal, state, local or foreign authority regarding any Hazardous Material, asbestos, or other environmental condition, proceeding, order, claim or violation affecting any of the Properties where such condition, proceeding, order, claim or violation would reasonably be expected to result in a liability to the Borrower, the General Partner or any Subsidiary in an amount, individually or in the aggregate, of \$50,000,000 or more.
- 8.2. <u>Financial Statements, Reports, Etc.</u>. The Borrower and the General Partner each shall maintain, for itself and each Subsidiary, a system of accounting established and administered in accordance with GAAP, and shall furnish to the Administrative Agent (and the Administrative Agent shall thereafter promptly furnish to the Lenders):
 - (i) quarterly financial statements (including a balance sheet, income statement, and cash flow statement) and related reports in form and substance satisfactory to the Lenders not later than forty-five (45) days after the end of each of the first three fiscal quarters (commencing with the fiscal quarter ending June 30, 2021), and not later than ninety (90) days after the end of each fiscal year (commencing with the fiscal year ending December 31, 2021), annual audited financial statements, audited by an accounting firm as reasonably approved by Administrative Agent (which audit report shall be without a "going concern" or like qualification or exception), provided, however, Administrative Agent shall only have the right to approve such accounting firm if the accounting firm is not a big 4 accounting firm, all certified by the Borrower's or the General Partner's, as applicable, chief financial officer, chief accounting officer or controller, calculation of the financial covenants described below, a description of Unencumbered Assets, a listing of capital expenditures (in the level of detail as currently disclosed in the Borrower's "Supplemental Information"), a report listing and describing all newly acquired Properties, including their cash flow, cost and secured or unsecured Indebtedness assumed in connection with such acquisition, if any, summary Property information for all Properties, including, without limitation, their Property Operating Income, occupancy rates, square footage, property type and date acquired or built, and such other information as may be requested to evaluate the quarterly compliance certificate delivered as provided below:
 - (ii) copies of all Form 10-Ks, 10-Qs, 8-Ks, and any other public information filed with the Securities Exchange Commission by the Borrower or the General Partner once a quarter simultaneously with delivering the compliance certificate described below, along with any other materials distributed to the shareholders of the General Partner or the partners of the Borrower from time to time, including a copy of the General Partner's annual report; provided that, to the extent any of such reports contains information required under the other subsections of this Section 8.2, the information need not be furnished separately under the other subsections; provided, further that the Borrower may comply with this Section 8.2(ii) by posting or causing to be posted, the foregoing information on either the Securities Exchange Commission public website or on the Borrower's or the General Partner's public website, so long as the Lenders have access to and are timely referred to any such website by the Borrower;
 - (iii) not later than forty-five (45) days after the end of the first three fiscal quarters (commencing with the fiscal quarter ending June 30, 2021), and not later than ninety (90) days after the end of the fiscal year (commencing with the fiscal year ending December 31, 2021), a report certified by the entity's chief financial officer, chief accounting officer or controller, containing Property Operating Income from individual

properties owned by the Borrower or a Wholly-Owned Subsidiary and included as Unencumbered Assets.

- (iv) Not later than forty-five (45) days after the end of each of the first three fiscal quarters (commencing with the fiscal quarter ending June 30, 2021), and not later than ninety (90) days after the end of the fiscal year (commencing with the fiscal year ending December 31, 2021), a compliance certificate in substantially the form of Exhibit H hereto signed by the Borrower's chief financial officer, chief accounting officer or controller confirming that the Borrower is in compliance with all of the covenants of the Loan Documents, showing the calculations and computations necessary to determine compliance with the financial covenants contained in this Agreement (including such schedules and backup information as may be necessary to demonstrate such compliance) and stating that to such officer's best knowledge, there is no other Default or Event of Default exists, or if any Default or Event of Default exists, stating the nature and status thereof;
- (v) As soon as possible and in any event within ten (10) Business Days after the Borrower knows that any Reportable Event has occurred with respect to any Plan, a statement, signed by the chief financial officer of the Borrower, describing said Reportable Event and within twenty (20) days after such Reportable Event, a statement signed by such chief financial officer describing the action which the Borrower proposes to take with respect thereto; and (b) within ten (10) Business Days of receipt, any notice from the Internal Revenue Service, PBGC or Department of Labor with respect to a Plan regarding any excise tax, proposed termination of a Plan, prohibited transaction or fiduciary violation under ERISA or the Code which could result in any liability to the Borrower or any member of the Controlled Group in excess of \$100,000; and (c) within ten (10) Business Days of filing, any Form 5500 filed by the Borrower with respect to a Plan, or any member of the Controlled Group which includes a qualified accountant's opinion;
- (vi) As soon as possible and in any event within thirty (30) days after receipt by the Borrower, a copy of (a) any notice or claim to the effect that the Borrower or any of its Subsidiaries is or may be liable to any Person as a result of the release by such entity, or any of its Subsidiaries, or any other Person of any toxic or hazardous waste or substance into the environment, and (b) any notice alleging any violation of any federal, state or local environmental, health or safety law or regulation by the Borrower or any of its Subsidiaries or Investment Affiliates, which, in either case, could be reasonably likely to have a Material Adverse Effect;

(vii) [Reserved]:

- (viii) Promptly upon the furnishing thereof to the shareholders of the Borrower, copies of all financial statements, reports and proxy statements so furnished; provided, further that the Borrower may comply with this Section 8.2(viii) by posting or having posted the foregoing information on either the Securities Exchange Commission public website or on the Borrower's or the General Partner's public website, so long as the Lenders have access to and are timely referred to any such website by the Borrower;
- (ix) Promptly upon the distribution thereof to the press or the public, copies of all press releases; provided, further that the Borrower may comply with this <u>Section 8.2(ix)</u> by posting or having posted the foregoing information on either the Securities

Exchange Commission public website or on the Borrower's or the General Partner's public website, so long as the Lenders have access to and are timely referred to any such website by the Borrower;

- (x) Promptly upon receipt thereof, notices with respect to the ratings for the Borrower's or the General Partner's long-term, senior unsecured debt, the effect of which may be to change the Base Rate Applicable Margin and/or the LIBOR Applicable Margin;
- (xi) As soon as possible, and in any event within ten (10) days after the Borrower knows of any fire or other casualty or any pending or threatened condemnation or eminent domain proceeding with respect to all or any material portion of any Unencumbered Asset, a statement signed by the Chief Financial Officer of the Borrower, describing such fire, casualty or condemnation and the action the Borrower intends to take with respect thereto; and
- (xii) Such other information (including, without limitation, nonfinancial information) as the Administrative Agent or any Lender may from time to time reasonably request, including, promptly upon the reasonable request of the Administrative Agent or any Lender, to the Administrative Agent or directly to such Lender, as the case may be, any information or documentation requested by it for purposes of complying with the Beneficial Ownership Regulation.
- 8.3. Existence and Conduct of Operations. Except as permitted herein, maintain and preserve its existence and all rights, privileges and franchises now enjoyed and necessary for the operation of its business, including remaining in good standing in each jurisdiction in which business is currently operated. The Borrower and the General Partner shall carry on and conduct their respective businesses in substantially the same manner and in substantially the same fields of enterprise as presently conducted. The Borrower will do, and will cause each of its Subsidiaries to do, all things necessary to remain duly incorporated and/or duly qualified, validly existing and in good standing as a real estate investment trust, corporation, general partnership, limited liability company or limited partnership, as the case may be, in its jurisdiction of incorporation/formation, except and to the extent that the Borrower determines that causing any of its Subsidiaries (other than any Loan Party) to remain duly incorporated and/or duly qualified, validly existing and in good standing is not in the best interests of the Borrower and such failure to remain duly incorporated and/or qualified, validly existing and in good standing could not reasonably be expected to have a Material Adverse Effect. The Borrower will maintain all requisite authority to conduct its business in each jurisdiction in which the Properties are located and, except where the failure to be so qualified would not have a Material Adverse Effect, in each jurisdiction required to carry on and conduct its businesses in substantially the same manner as it is presently conducted, and, specifically, neither the Borrower nor its Subsidiaries will undertake any business other than the acquisition, development, ownership, management, operation and leasing of industrial/warehouse properties and business activities ancillary, incidental or otherwise reasonably related thereto.
- 8.4. <u>Maintenance of Properties</u>. Maintain, preserve, protect and keep the Properties in good repair, working order and condition, and make all necessary and proper repairs, renewals and replacements, normal wear and tear excepted.
- 8.5. <u>Insurance</u>. Upon request of the Administrative Agent, provide a certificate of insurance from all insurance carriers who maintain policies with respect to the Properties within thirty (30) days after the end of each fiscal year, evidencing that the insurance required to be furnished to Lenders pursuant to <u>Section 6.20</u> hereof is in full force and effect. The Borrower shall timely pay, or cause to be

paid, all premiums on all insurance policies required under this Agreement from time to time. The Borrower shall promptly notify its insurance carrier or agent therefor (with a copy of such notification being provided simultaneously to Administrative Agent) if there is any occurrence which, under the terms of any insurance policy then in effect with respect to the Properties, requires such notification.

- 8.6. <u>Payment of Obligations</u>. Pay and discharge when due (i) all Federal, state and other material taxes, assessments, governmental charges and (ii) all other obligations which, if unpaid, would by law become a Lien upon its property, except, in the case of each of clauses (i) and (ii), such as may be contested in good faith or as to which a bona fide dispute may exist, and for which adequate reserves have been provided in accordance with sound accounting principles used by the Borrower on the date hereof.
- 8.7. <u>Compliance with Laws</u>. Comply in all material respects with all Applicable Laws, rules, regulations, orders and directions (including, without limitation, Anti-Corruption Laws and Sanctions) of any governmental authority having jurisdiction over the Borrower, the General Partner, or any of their respective businesses.
- 8.8. <u>Adequate Books</u>. Maintain adequate books, accounts and records in order to provide financial statements in accordance with GAAP and, if requested by any Lender, permit employees or representatives of such Lender at any reasonable time and upon reasonable notice to inspect and audit the properties of the Borrower and of the Consolidated Operating Partnership, and to examine or audit the inventory, books, accounts and records of each of them and make copies and memoranda thereof.
 - 8.9. ERISA. Comply in all material respects with all requirements of ERISA applicable to it with respect to each Plan.
- 8.10. <u>Maintenance of Status</u>. The General Partner shall at all times (i) remain as a corporation listed and in good standing on a national securities exchange, and (ii) take all steps to maintain the General Partner's status as a real estate investment trust in compliance with all applicable provisions of the Code.
- 8.11. <u>Use of Proceeds</u>. Use the proceeds of the Facility for the general business purposes of the Borrower, including without limitation working capital needs, closing costs, interim funding for property acquisitions and construction of new industrial properties, and/or payment of other debts and obligations of the Borrower.
- 8.12. <u>Distributions</u>. Provided there is no declared Monetary Default then existing and provided there is not an Event of Default relating to a breach of the financial covenants contained in <u>Section 9.7</u> below, the Borrower may make distributions to the General Partner and the General Partner may make distributions to its shareholders. Notwithstanding the foregoing, the Borrower shall be permitted to make distributions to the General Partner and the General Partner shall be permitted at all times to make distributions of whatever amount is necessary to maintain its tax status as a real estate investment trust.

ARTICLE IX.

NEGATIVE COVENANTS

The Borrower covenants and agrees that, so long as the Revolving Commitment shall remain available and until full and final payment of all Obligations incurred under the Loan Documents, without the prior written consent of either all of the Lenders pursuant to Section 14.13(b)(iii) or the consent of the Required Lenders in all other cases, it will not, and the General Partner will not and, in the case of Sections 9.9, the Borrower's Subsidiaries will not:

- 9.1. <u>Change in Business</u>. Engage in any business if, as a result, the general nature of the business in which the Borrower and the General Partner, taken as a whole, would then be engaged would be substantially changed from the general nature of the business in which the Borrower and the General Partner, taken as a whole, are engaged on the Closing Date.
- 9.2. <u>Change of Management of Properties</u>. Change the management of the Properties, except that (a) any Affiliate of the Borrower or the General Partner shall be permitted to manage any of the Properties and (b) the Borrower, the General Partner or any of their Subsidiaries may engage and/or replace third-party property managers with respect to any Property, *provided* that the aggregate book value of all Properties at any one time under third-party management shall not exceed 30% of the aggregate book value of all of the Properties at such time.
- 9.3. Change of Borrower Ownership. Without the consent of the Required Lenders, allow (i) the General Partner to own less than fifty-one percent (51%) of the partnership interests in the Borrower, (ii) the Borrower to be controlled by a Person other than the General Partner, (iii) a Change in Control to occur, or (iv) any pledge of, other encumbrance on, or conversion to limited partnership interests of, any of the general partnership interests in the Borrower. If any of the foregoing occurs with the consent of the Required Lenders, each non-consenting Lender shall have the option to terminate its Revolving Commitment and such termination shall be effective upon notice to Administrative Agent and the Borrower of the termination. Upon the termination of a Lender's Revolving Commitment in accordance with the foregoing provision, the Lender's obligation to fund Borrowings and to issue Facility Letters of Credit shall be terminated and the Borrower shall repay any outstanding Obligations due to such Lender prior to or concurrently with the occurrence of any of the foregoing events. Following the termination of any Revolving Commitments pursuant to this provision, the Aggregate Revolving Commitment shall be reduced by the amount of the Revolving Commitments terminated, and the pro rata shares of each remaining Lender shall be adjusted to reflect the new Aggregate Revolving Commitment.
- 9.4. <u>Use of Proceeds</u>. Apply or permit to be applied any proceeds of any Borrowing directly or indirectly, to the funding of any purchase of, or offer for, any share of capital stock of any publicly held corporation unless the board of directors of such corporation has consented to such offer prior to any public announcements relating thereto and all of the Lenders have consented to such use of the proceeds of the Facility. The Borrower shall not request any Loan or Facility Letter of Credit, shall not use, and shall ensure that its Subsidiaries and its or their respective directors, officers, employees and agents do not use the proceeds of any Loan or Facility Letter of Credit (i) in furtherance of an offer, payment, promise to pay, or authorization of the payment or giving of money, or anything else of value, to any Person in violation of any Anti-Corruption Laws, (ii) for the purpose of funding, financing or facilitating any activities, business or transaction of or with any Sanctioned Person to the extent such activities, business or transaction would be prohibited by Sanctions if conducted by a corporation incorporated in the United States or in any European Union member state or (iii) in any manner that would result in the violation of any Sanctions applicable to any party hereto.

- 9.5. <u>Liens</u>. Create, incur, or suffer to exist (or permit any of its Subsidiaries to create, incur, or suffer to exist) any Lien in, of or on the Property of any member of the Consolidated Operating Partnership other than:
 - (i) Liens for taxes, assessments or governmental charges or levies on their Property if the same shall not at the time be delinquent or thereafter can be paid without penalty, or are being contested in good faith and by appropriate proceedings and for which adequate reserves shall have been set aside on their books;
 - (ii) Liens which arise by operation of law, such as carriers', warehousemen's, landlords', materialmen and mechanics' liens and other similar liens arising in the ordinary course of business which secure payment of obligations not more than thirty (30) days past due or which are being contested in good faith by appropriate proceedings and for which adequate reserves shall have been set aside on its books;
 - (iii) Liens arising out of pledges or deposits under worker's compensation laws, unemployment insurance, old age pensions, or other social security or retirement benefits, or similar legislation;
 - (iv) Utility easements, building restrictions, zoning restrictions, easements and such other encumbrances or charges against real property as are of a nature generally existing with respect to properties of a similar character and which do not in any material way affect the marketability of the same or interfere with the use thereof in the business of the Borrower or its Subsidiaries;
 - (v) Liens of any Subsidiary in favor of the Borrower or the General Partner; and
 - (vi) So long as immediately prior to the creation, assumption or incurring of such Lien, and immediately thereafter, no Default or Event of Default would be in existence, Liens on Properties securing Indebtedness not prohibited hereunder.

Liens permitted pursuant to this <u>Section 9.5</u> shall be deemed to be "Permitted Liens".

- 9.6. <u>Regulation U.</u> Use the proceeds of the Loans for any other purpose than as provided for in <u>Section 8.11</u> and <u>Section 9.4</u> hereof. No part of the proceeds of any Loan will be used, whether directly or indirectly for any purpose that entails a violation of any of the regulations of the Board of Governors of the Federal Reserve System, including Regulations T, U and X. The proceeds will not be used in a manner which would cause the Facility to be treated as a "Purpose Credit."
 - 9.7. <u>Indebtedness and Cash Flow Covenants</u>. Permit or suffer:
 - (a) as of the last day of any fiscal quarter, the ratio of (A) the sum of (1) EBITDA of the Consolidated Operating Partnership plus (2) interest income (other than any interest income from assets being used to support Defeased Debt) to (B) the sum of (1) Debt Service plus, without duplication, (2) all payments on account of preferred stock or preferred partnership units of any member of the Consolidated Operating Partnership for such quarter plus (3) all Ground Lease Payments due from any member of the Consolidated Operating Partnership to the extent not deducted as an expense in calculating EBITDA of the Consolidated Operating Partnership (the "Fixed Charge Coverage Ratio"), to be less than (i) 1.50 to 1.0 from the Agreement Execution

Date through the Maturity Date, with all such calculations in clauses (A) and (B) above based on the results of the four (4) consecutive fiscal quarters then ended;

- (b) as of the last day of any fiscal quarter, the Consolidated Leverage Ratio to exceed sixty percent (60%); provided that, if any Material Acquisition shall occur then, at the election of the Borrower upon delivery of prior written notice to the Administrative Agent, concurrently with or prior to the delivery of a Compliance Certificate pursuant to Section 8.2(iv), and provided that no Default or Event of Default shall have occurred and be continuing (other than as a result of the Consolidated Leverage Ratio as of the end of such fiscal quarter being greater than sixty percent (60%) but less than or equal to sixty-five percent (65%)), the Consolidated Leverage Ratio shall be increased to sixty-five percent (65%) for such fiscal quarter and the next three succeeding fiscal quarters (any period with such increase a "Consolidated Leverage Ratio Increase Period"), provided that (i) no more than two (2) Consolidated Leverage Ratio Increase Periods may be elected by the Borrower during the term of this Agreement and (ii) any such Consolidated Leverage Ratio Increase Periods shall be nonconsecutive;
- (c) as of the last day of any fiscal quarter, the ratio of outstanding Consolidated Senior Unsecured Debt to Value of Unencumbered Assets (such ratio, the "<u>Unencumbered Leverage Ratio</u>") to exceed sixty percent (60%); <u>provided</u> that, if any Material Acquisition shall occur then, at the election of the Borrower upon delivery of prior written notice to the Administrative Agent, concurrently with or prior to the delivery of a Compliance Certificate pursuant to <u>Section 8.2(iv</u>), and provided that no Default or Event of Default shall have occurred and be continuing (other than as a result of the Unencumbered Leverage Ratio as of the end of such fiscal quarter being greater than sixty percent (60%) but less than or equal to sixty-five percent (65%)), the Unencumbered Leverage Ratio shall be increased to sixty-five percent (65%) for such fiscal quarter and the next three succeeding fiscal quarters (any period with such increase, an "<u>Unencumbered Leverage Ratio Increase Period</u>"), <u>provided</u> further that (i) no more than two (2) Unencumbered Leverage Ratio Increase Periods may be elected by the Borrower during the term of this Agreement and (ii) any such Unencumbered Leverage Ratio Increase Periods shall be non-consecutive;
- (d) as of the last day of any fiscal quarter, Consolidated Secured Debt to exceed 40% of Implied Capitalization Value of the Consolidated Operating Partnership; or
- (e) as of the last day of any fiscal quarter, the ratio of (A) Property Operating Income from Unencumbered Assets that are not Assets Under Development for such fiscal quarter to (B) interest expense on all Consolidated Senior Unsecured Debt for such fiscal quarter to be less than (x) 1.75 to 1.0 from the Agreement Execution Date through the Maturity Date.

The foregoing covenants set forth in paragraphs (c) and (e) above shall be tested at the end of each fiscal quarter (for the applicable reporting period), on a pro-forma basis at the time of each Borrowing under the Facility (using the latest quarterly financial statements then available and taking into account the proposed Borrowing), and on a pro-forma basis upon any Asset Sale or incurrence of any Indebtedness by the Consolidated Operating Partnership (using the latest quarterly financial statements then available and taking into account the proposed incurrence of Indebtedness). To the extent the Consolidated Operating Partnership has Defeased Debt, both the underlying debt and interest payable thereon and the financial assets used to defease such debt and interest earned thereon shall be excluded from calculations of the foregoing financial covenants. All financial computations required under this Section 9.7 shall be made in accordance with GAAP as in effect on the Agreement Execution Date.

- 9.8. Change of Control; Mergers and Dispositions. Enter into any merger, consolidation, reorganization, division or liquidation or transfer or otherwise dispose of all or a substantial portion of its properties, except for: such transactions that occur between wholly owned Subsidiaries; transactions where the Borrower and the General Partner are the surviving entities and there is no change in business conducted, loss of an Investment Grade Rating or Change in Control, and no Default or Event of Default under the Loan Documents results from such transaction; or as otherwise approved in advance by the Lenders. To the extent not reported on General Partner's Form 8-K filings, the Borrower will notify the Administrative Agent (who will promptly notify Lenders) of any single event acquisition (whether individually or consummated as a series of related acquisitions pursuant to documentation closed concurrently), disposition, merger, division or asset purchase involving assets valued in excess of 25% of the Consolidated Operating Partnership's then current Market Value Net Worth and certify compliance with covenants after giving effect to such proposed acquisition, disposition, merger, division or asset purchase.
- 9.9. <u>Negative Pledge</u>. The Borrower agrees that throughout the term of this Facility, no "negative pledge" on any Project then included in Unencumbered Assets restricting the Borrower's (or wholly owned Subsidiary's) right to sell or encumber such Project shall be given to any other lender or creditor other than a Permitted Negative Pledge or, if such a "negative pledge" is given, the Project affected shall be immediately excluded from Unencumbered Assets.

ARTICLE X.

DEFAULTS

The occurrence of any one or more of the following events shall constitute an Event of Default:

- 10.1. <u>Nonpayment of Principal</u>. The Borrower fails to pay any principal portion of the Obligations, including the Reimbursement Obligations, when due, whether on the Maturity Date or otherwise.
- 10.2. <u>Certain Covenants</u>. The Borrower, the General Partner and/or Consolidated Operating Partnership, as the case may be, is not in compliance with any one or more of <u>Sections 8.10</u>, <u>8.12</u>, <u>9.3</u>, <u>9.4</u>, <u>9.5</u>, <u>9.7</u>, <u>9.8</u>, or <u>9.9</u> hereof.
- 10.3. <u>Nonpayment of Interest and Other Obligations</u>. The Borrower fails to pay any interest or other portion of the Obligations, other than payments of principal, and such failure continues for a period of five (5) days after the date such payment is due.
- 10.4. <u>Cross Default</u>. Any monetary default occurs (after giving effect to any applicable cure period) under any other Indebtedness (which includes liability under Guaranties) of the Borrower or the General Partner, singly or in the aggregate, in excess of Seventy-Five Million Dollars (\$75,000,000), other than (i) Indebtedness arising from the purchase of personal property or the provision of services, the amount of which is being contested by the Borrower or (ii) Indebtedness which is "non recourse", i.e., which is not recoverable by the creditor thereof from the general assets of the Borrower, the General Partner or any of their Affiliates, but is limited to the proceeds of certain real estate, improvements and related personal property.
- 10.5. <u>Loan Documents</u>. Any Loan Document is not in full force and effect or a default has occurred and is continuing thereunder after giving effect to any cure or grace period in any such document.

- 10.6. <u>Representation or Warranty</u>. At any time or times hereafter any representation or warranty set forth in Articles VI or VII of this Agreement or in any other Loan Document or in any statement, report or certificate now or hereafter made by the Borrower or the General Partner to the Lenders or the Administrative Agent is not true and correct in any material respect.
- 10.7. <u>Covenants, Agreements and Other Conditions</u>. The Borrower or the General Partner fails to perform or observe any of the other covenants, agreements and conditions contained in Articles VIII and IX (except for <u>Sections 8.10</u>, <u>8.12</u>, <u>9.3</u>, <u>9.4</u>, <u>9.5</u>, <u>9.7</u>, <u>9.8</u>, or <u>9.9</u> hereof) and elsewhere in this Agreement or any of the other Loan Documents in accordance with the terms hereof or thereof, not specifically referred to herein, and such Default continues unremedied for a period of thirty (30) days after written notice from Administrative Agent, provided, however, that if such Default is susceptible of cure but cannot by the use of reasonable efforts be cured within such thirty (30) day period, such Default shall not constitute an Event of Default under this <u>Section 10.7</u> so long as (i) the Borrower or the General Partner, as the case may be, has commenced a cure within such thirty day period and (ii) thereafter, the Borrower or the General Partner, as the case may be, is proceeding to cure such default continuously and diligently and in a manner reasonably satisfactory to Lenders and (iii) such default is cured not later than sixty (60) days after the expiration of such thirty (30) day period.
 - 10.8. No Longer General Partner. The General Partner shall no longer be the sole general partner of the Borrower.
- 10.9. <u>Material Adverse Financial Change</u>. The Borrower or the General Partner has suffered a Material Adverse Financial Change or is Insolvent.

10.10. Bankruptcy.

- (a) The General Partner, the Borrower or one or more Subsidiaries in which the Equity Value of the Subsidiary or Subsidiaries is more than ten percent (10%), in the aggregate, of Implied Capitalization Value of the Consolidated Operating Partnership shall (i) have an order for relief entered with respect to it under the Federal bankruptcy laws as now or hereafter in effect, (ii) make an assignment for the benefit of creditors, (iii) apply for, seek, consent to, or acquiesce in, the appointment of a receiver, custodian, trustee, examiner, liquidator or similar official for it or any substantial portion of its Property, (iv) institute any proceeding seeking an order for relief under the Federal bankruptcy laws as now or hereafter in effect or seeking to adjudicate it as a bankrupt or insolvent, or seeking dissolution, winding up, liquidation, reorganization, arrangement, adjustment or composition of it or its debts under any law relating to bankruptcy, insolvency or reorganization or relief of debtors or fail to file an answer or other pleading denying the material allegations of any such proceeding filed against it, (v) take any corporate action to authorize or effect any of the foregoing actions set forth in this Section 10.10(a), (vi) fail to contest in good faith any appointment or proceeding described in Section 10.10(b) or (vii) not pay, or admit in writing its inability to pay, its debts generally as they become due. As used herein, the term "Equity Value" of a Subsidiary shall mean the lesser of: (1) Property Operating Income of such Subsidiary's Properties owned as of the Agreement Execution Date capitalized at a 6.5% rate, less any Indebtedness of such Subsidiary; or (2) the appraised value of any of such Subsidiary's Properties, less any Indebtedness of such Subsidiary.
- (b) A receiver, trustee, examiner, liquidator or similar official shall be appointed for (i) the General Partner, the Borrower or one or more Subsidiaries in which the Equity Value of the Subsidiary or Subsidiaries is more than ten percent (10%), in the aggregate, of Implied Capitalization Value of the Consolidated Operating Partnership or (ii) all or substantially all of the Consolidated Operating Partnership's Properties.

- (c) A proceeding described in <u>Section 10.10(a)(iv)</u> shall be instituted against the General Partner, the Borrower or one or more Subsidiaries in which the Equity Value of the Subsidiary or Subsidiaries is more than ten percent (10%), in the aggregate, of Implied Capitalization Value of the Consolidated Operating Partnership, and such proceeding continues undismissed or unstayed for a period of sixty (60) consecutive days.
- 10.11. <u>Legal Proceedings</u>. The Borrower or the General Partner is enjoined, restrained or in any way prevented by any court order or judgment or if a notice of lien, levy, or assessment is filed of record with respect to all or any part of the Properties by any governmental department, office or agency, which could materially adversely affect the performance of the obligations of such parties hereunder or under the Loan Documents, as the case may be, or if any proceeding is filed or commenced seeking to enjoin, restrain or in any way prevent the foregoing parties from conducting all or a substantial part of their respective business affairs and failure to vacate, stay, dismiss, set aside or remedy the same within ninety (90) days after the occurrence thereof.
- 10.12. <u>ERISA</u>. The Borrower or the General Partner is deemed to hold "plan assets" within the meaning of ERISA or any regulations promulgated thereunder of an employee benefit plan (as defined in Section 3(3) of ERISA) which is subject to Title I of ERISA or any plan (within the meaning of Section 4975 of the Code).
 - 10.13. Change in Control. A Change in Control has occurred.
- 10.14. <u>Failure to Satisfy Judgments</u>. The General Partner, the Borrower or any of its Subsidiaries shall fail within sixty (60) days to pay, bond or otherwise discharge any judgments or orders for the payment of money (other than with respect to any Indebtedness which is "non recourse", i.e., which is not recoverable by the creditor thereof from the general assets of the Borrower, the General Partner or any of their Affiliates, but is limited to the proceeds of certain real estate, improvements and related personal property) in an amount which, when added to all other judgments or orders outstanding against the General Partner, the Borrower or any Subsidiary would exceed Seventy-Five Million Dollars (\$75,000,000) in the aggregate, which have not been stayed on appeal or otherwise appropriately contested in good faith, unless the liability is insured against and the insurer has not challenged coverage of such liability.
- 10.15. <u>Environmental Remediation</u>. Failure to remediate (or pursue the remediation process with due diligence and good faith) within the time period required by law or governmental order, (or within a reasonable time in light of the nature of the problem if no specific time period is so established), environmental problems in violation of Applicable Law related to Properties of the Borrower and/or its Subsidiaries where the estimated cost of remediation is in the aggregate in excess of Seventy-Five Million Dollars (\$75,000,000), in each case after all administrative hearings and appeals have been concluded.

ARTICLE XI.

ACCELERATION, WAIVERS, AMENDMENTS AND REMEDIES

11.1. Acceleration. If any Event of Default described in Section 10.10 hereof occurs, or the Borrower or the General Partner becomes Insolvent, the Revolving Commitments and obligation of the Lenders to make Borrowings and of any Issuing Bank to issue Facility Letters of Credit hereunder shall automatically terminate and the Obligations shall immediately become due and payable without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by the Borrower. If any other Event of Default described in Article X hereof occurs and is continuing, the Administrative Agent, with the consent of the Required Lenders, may, and at the request of the Required

Lenders shall, by notice to the Borrower, take either or both of the following actions, at the same or different times: (i) terminate the Revolving Commitments, and thereupon the Revolving Commitments shall terminate immediately, and (ii) declare the Loans then outstanding to be due and payable in whole (or in part, in which case any principal not so declared to be due and payable may thereafter be declared to be due and payable), and thereupon the principal of the Loans so declared to be due and payable, together with accrued interest thereon and all fees and other Obligations of the Borrower accrued hereunder, shall become due and payable immediately, without presentment, demand, protest or other notice of any kind, all of which are hereby waived by the Borrower.

In addition to the foregoing, following the occurrence of an Event of Default and so long as any Facility Letter of Credit has not been fully drawn and has not been cancelled or expired by its terms, upon demand by the Required Lenders the Borrower shall deposit in the Letter of Credit Collateral Account cash in an amount equal to the LC Exposure as of such date plus any accrued and unpaid interest thereon provided that the obligation to deposit such cash collateral shall become effective immediately, and such deposit shall become immediately due and payable, without demand or other notice of any kind, upon the occurrence of any Default with respect to the Borrower described in Section 10.10. Each such deposit pursuant to this paragraph shall be held by the Administrative Agent as collateral for the payment and performance of the obligations of the Borrower under this Agreement. The Administrative Agent shall have exclusive dominion and control, including the exclusive right of withdrawal, over such account. Other than any interest earned on the investment of such deposits, which investments shall be made at the option and sole discretion of the Administrative Agent and at the Borrower's risk and expense, such deposits shall not bear interest. Interest or profits, if any, on such investments shall accumulate in such account. Moneys in such account shall be applied by the Administrative Agent to reimburse the applicable Issuing Bank for LC Disbursements for which it has not been reimbursed and, to the extent not so applied, shall be held for the satisfaction of the Reimbursement Obligations of the Borrower for the LC Exposure at such time or, if the maturity of the Loans has been accelerated (but subject to the consent of Revolving Lenders with LC Exposure representing at least 51% of the total LC Exposure), be applied to satisfy other obligations of the Borrower under this Agreement. If the Borrower is required to provide an amount of cash collateral hereunder as a result of the occurrence of a Default, such amount (to the extent not applied as aforesaid) shall be returned to the Borrower within three (3) Business Days after all Defaults have been cured or waived. The Borrower shall have no control over funds in the Letter of Credit Collateral Account, which funds shall be invested by the Administrative Agent from time to time in its discretion in certificates of deposit of Wells having a maturity not exceeding thirty (30) days. Such funds shall be promptly applied by the Administrative Agent to reimburse the applicable Issuing Bank for drafts drawn from time to time under the Facility Letters of Credit and to pay any fees or other amounts due with respect thereto. Such funds, if any, remaining in the Letter of Credit Collateral Account following the payment of all Obligations in full shall, unless the Administrative Agent is otherwise directed by a court of competent jurisdiction, be promptly paid over to the Borrower.

11.2. Preservation of Rights; Amendments. No delay or omission of the Lenders in exercising any right under the Loan Documents shall impair such right or be construed to be a waiver of any Default or an acquiescence therein, and the making of a Borrowing notwithstanding the existence of a Default or the inability of the Borrower to satisfy the conditions precedent to such Borrowing shall not constitute any waiver or acquiescence. Any single or partial exercise of any such right shall not preclude other or further exercise thereof or the exercise of any other right, and no waiver, amendment or other variation of the terms, conditions or provisions of the Loan Documents whatsoever shall be valid unless in writing signed by the Administrative Agent and the number of Lenders required hereunder and then only to the extent in such writing specifically set forth. All remedies contained in the Loan Documents or by law afforded shall be cumulative and all shall be available to the Lenders until the Obligations have been paid in full.

ARTICLE XII.

THE ADMINISTRATIVE AGENT

- 12.1. Appointment. Wells Fargo Bank, National Association is hereby appointed by each of the Lenders as its contractual representative (herein referred to as the "Administrative Agent") hereunder and under each other Loan Document, and each of the Lenders irrevocably authorizes the Administrative Agent to act as the contractual representative of such Lender with the rights and duties expressly set forth herein and in the other Loan Documents. The Administrative Agent agrees to act as such contractual representative upon the express conditions contained in this Article XII. Notwithstanding the use of the defined term "Administrative Agent," it is expressly understood and agreed that the Administrative Agent shall not have any fiduciary responsibilities to any Lender by reason of this Agreement or any other Loan Document and that the Administrative Agent is merely acting as the contractual representative of the Lenders with only those duties as are expressly set forth in this Agreement and the other Loan Documents. In its capacity as the Lenders' contractual representative, the Administrative Agent (i) does not hereby assume any fiduciary duties to any of the Lenders, (ii) is a "representative" of the Lenders within the meaning of the term "secured party" as defined in the Illinois Uniform Commercial Code and (iii) is acting as an independent contractor, the rights and duties of which are limited to those expressly set forth in this Agreement and the other Loan Documents. Each of the Lenders hereby agrees to assert no claim against the Administrative Agent on any agency theory or any other theory of liability for breach of fiduciary duty, all of which claims each Lender hereby waives.
- 12.2. <u>Powers</u>. The Administrative Agent shall have and may exercise such powers under the Loan Documents as are specifically delegated to the Administrative Agent by the terms of each thereof, together with such powers as are reasonably incidental thereto. The Administrative Agent shall have no implied duties to the Lenders, or any obligation to the Lenders to take any action thereunder except any action specifically provided by the Loan Documents to be taken by the Administrative Agent.
- 12.3. <u>General Immunity</u>. Neither the Administrative Agent (in its capacity as Administrative Agent) nor any of its directors, officers, agents or employees shall be liable to the Borrower, the Lenders or any Lender for any action taken or omitted to be taken by it or them hereunder or under any other Loan Document or in connection herewith or therewith, except for its or their own Gross Negligence or willful misconduct, as determined by a court of competent jurisdiction by a final and non-appealable judgment.
- 12.4. No Responsibility for Loans, Recitals, etc.. Neither the Administrative Agent (in its capacity as Administrative Agent) nor any of its directors, officers, agents or employees shall be responsible for or have any duty to ascertain, inquire into, or verify (i) any statement, warranty or representation made in connection with any Loan Document or any borrowing hereunder; (ii) the performance or observance of any of the covenants or agreements of any obligor under any Loan Document; (iii) the satisfaction of any condition specified in Article V, except receipt of items required to be delivered to the Administrative Agent; or (iv) the validity, effectiveness or genuineness of any Loan Document or any other instrument or writing furnished in connection therewith. Except as expressly set forth herein, the Administrative Agent shall not have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Subsidiaries that is communicated to or obtained by the bank serving as Administrative Agent or any of its Affiliates in any capacity.
- 12.5. <u>Action on Instructions of Lenders</u>. The Administrative Agent shall in all cases be fully protected in acting, or in refraining from acting, hereunder and under any other Loan Document in accordance with written instructions signed by the Required Lenders or all Lenders, as the case may be, and such instructions and any action taken or failure to act pursuant thereto shall be binding on all of the

Lenders and on all holders of Notes. The Administrative Agent shall be fully justified in failing or refusing to take any action hereunder and under any other Loan Document unless it shall be indemnified to its satisfaction by the Lenders pro rata against any and all liability, cost and expense that it may incur by reason of taking or continuing to take any such action.

- 12.6. <u>Employment of Administrative Agents and Counsel</u>. The Administrative Agent may execute any of its duties as Administrative Agent hereunder and under any other Loan Document by or through employees, agents, and attorneys in fact and shall not be answerable to the Lenders, except as to money or securities received by it or its authorized agents, for the default or misconduct of any such agents or attorneys in fact selected by it with reasonable care. The Administrative Agent shall be entitled to advice of counsel concerning all matters pertaining to the agency hereby created and its duties hereunder and under any other Loan Document. Without limitation of the extent of any liability any Lender may otherwise have pursuant to this Agreement, the Administrative Agent shall have no liability for any action taken or omitted in good faith as a result of advice of counsel.
- 12.7. <u>Reliance on Documents; Counsel</u>. The Administrative Agent shall be entitled to rely upon any Note, notice, consent, certificate, affidavit, letter, telegram, statement, paper or document believed by it to be genuine and correct and to have been signed or sent by the proper person or persons, and, in respect to legal matters, upon the opinion of outside counsel selected by the Administrative Agent.
- Administrative Agent's Reimbursement and Indemnification. The Lenders agree to reimburse and indemnify the Administrative Agent ratably in accordance with their respective Percentages (i) for any amounts not reimbursed by the Borrower (and without limiting the obligation of the Borrower to do so) for which the Administrative Agent is entitled to reimbursement by the Borrower under the Loan Documents, (ii) for any other reasonable expenses incurred by the Administrative Agent on behalf of the Lenders, in connection with the preparation, execution, delivery, administration and enforcement of the Loan Documents, if not paid by the Borrower (and without limiting the obligation of the Borrower to do so), and (iii) for any liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind and nature whatsoever which may be imposed on, incurred by or asserted against the Administrative Agent (in its capacity as Administrative Agent and not as a Lender) in any way relating to or arising out of the Loan Documents or any other document delivered in connection therewith or the transactions contemplated thereby, or the enforcement of any of the terms thereof or of any such other documents, provided that no Lender shall be liable for any of the foregoing to the extent they arise from the Gross Negligence or willful misconduct of the Administrative Agent as determined by a court of competent jurisdiction in a final, non-appealable judgment; provided, however, that no action taken in accordance with the directions of the Required Lenders (or all of the Lenders, if expressly required hereunder) shall be deemed to constitute Gross Negligence or willful misconduct for purposes of this Section. The agreements in this Section shall survive the payment of the Loans and all other amounts payable hereunder or under the other Loan Documents and the termination of this Agreement. If the Borrower shall reimburse the Administrative Agent for any indemnifiable amount set forth in Section 12.8(i), (ii) or (ii) following payment by any Lender to the Administrative Agent in respect of any such indemnifiable amount pursuant to this Section, the Administrative Agent shall share such reimbursement on a ratable basis with each Lender making any such payment.
- 12.9. Rights as a Lender. With respect to its Revolving Commitment, Borrowings made by it and any Note issued to it, the Administrative Agent shall have the same rights and powers hereunder and under any other Loan Document as any Lender and may exercise the same as though it were not the Administrative Agent, and the term "Lender" or "Lenders" shall, unless the context otherwise indicates, include the Administrative Agent in its individual capacity. The Administrative Agent, in its individual capacity, may accept deposits from, lend money to, and generally engage in any kind of trust, debt, equity

or other transaction, in addition to those contemplated by this Agreement or any other Loan Document, with the Borrower or any of its Subsidiaries in which the Borrower or such Subsidiary is not restricted hereby from engaging with any other Person. Each of the Lenders and each Issuing Bank acknowledges that the Administrative Agent's legal counsel in connection with the transactions contemplated by this Agreement is only acting as counsel to the Administrative Agent and is not acting as counsel to any Lender or any Issuing Bank.

- 12.10. <u>Funds Transfer Disbursements</u>. The Borrower hereby authorizes the Administrative Agent to disburse the proceeds of any Loan made by the Lenders or any of their Affiliates pursuant to the Loan Documents as requested by an authorized representative of the Borrower to any of the accounts designated in the Disbursement Instruction Agreement.
- 12.11. <u>Lender Credit Decision</u>. Each Lender acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Lender and based on the financial statements prepared by the Borrower and such other documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement and the other Loan Documents. Each Lender also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Agreement and the other Loan Documents.
- 12.12. Successor Administrative Agent. The Administrative Agent may resign at any time as Administrative Agent under the Loan Documents by giving written notice thereof to the Lenders and the Borrower. Upon any such resignation, the Required Lenders shall have the right to appoint a successor Administrative Agent which appointment shall, provided no Default or Event of Default exists, be subject to the Borrower's approval, which approval shall not be unreasonably withheld or delayed (except that the Borrower shall, in all events, be deemed to have approved each Lender and any of its Affiliates as a successor Administrative Agent). If no successor Administrative Agent shall have been so appointed in accordance with the immediately preceding sentence, and shall have accepted such appointment, within 30 days after the current Administrative Agent's giving of notice of resignation, then the current Administrative Agent may, on behalf of the Lenders and the Issuing Banks, appoint a successor Administrative Agent, which shall be a Lender, if any Lender shall be willing to serve, and otherwise shall be an Eligible Assignee. Upon the acceptance of any appointment as Administrative Agent hereunder by a successor Administrative Agent, such successor Administrative Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the current Administrative Agent, and the current Administrative Agent shall be discharged from its duties and obligations under the Loan Documents. Such successor Administrative Agent shall issue letters of credit in substitution for the Facility Letters of Credit, if any, outstanding at the time of such succession or shall make other arrangements satisfactory to the current Administrative Agent, in either case, to assume effectively the obligations of the current Administrative Agent with respect to such Facility Letters of Credit. After any Administrative Agent's resignation hereunder as Administrative Agent, the provisions of this Article XII shall continue to inure to its benefit as to any actions taken or omitted to be taken by it while it was Administrative Agent under the Loan Documents. Notwithstanding anything contained herein to the contrary, the Administrative Agent may assign its rights and duties under the Loan Documents to any of its Affiliates by giving the Borrower and each Lender prior written notice.
- 12.13. <u>Notice of Defaults</u>. The Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of a Default or Event of Default unless the Administrative Agent has received notice from a Lender or the Borrower referring to this Agreement, describing with reasonable specificity such Default or Event of Default and stating that such notice is a "notice of default." If a Lender becomes aware of a Default or Event of Default, such Lender shall notify the Administrative Agent of such fact.

Upon receipt of such notice that a Default or Event of Default has occurred, the Administrative Agent shall notify each of the Lenders of such fact.

- 12.14. Requests for Approval. If the Administrative Agent requests in writing the consent or approval of a Lender, such Lender shall, or, in the case of any request for consent or approval that is subject to Section 14.13(a) or (b), shall use commercially reasonable efforts to, respond and either approve or disapprove definitively in writing to the Administrative Agent within ten (10) Business Days (or sooner if such notice specifies a shorter period, but in no event less than five (5) Business Days for responses based on Administrative Agent's good faith determination that circumstances exist warranting its request for an earlier response) after such written request from the Administrative Agent provided that the request for approval states the time by which a response is needed before approval is deemed given. Solely with respect to any request for consent or approval that is subject to Section 14.13(c), if the Lender does not so respond, that Lender shall be deemed to have approved the request. Upon request, the Administrative Agent shall notify the Lenders which Lenders, if any, failed to respond to a request for approval.
- 12.15. <u>Copies of Documents</u>. Administrative Agent shall promptly deliver to each of the Lenders copies of all notices of default and other formal notices sent or received and according to <u>Section 15.1</u> of this Agreement. Administrative Agent shall deliver to Lenders within fifteen (15) Business Days following receipt, copies of all financial statements, certificates and notices received regarding the General Partner's ratings except to the extent such items are required to be furnished directly to the Lenders by the Borrower hereunder. Within fifteen (15) Business Days after a request by a Lender to the Administrative Agent for other documents furnished to the Administrative Agent by the Borrower, the Administrative Agent shall provide copies of such documents to such Lender except where this Agreement obligates Administrative Agent to provide copies in a shorter period of time.
- 12.16. <u>Defaulting Lenders</u>. Notwithstanding any provision of this Agreement to the contrary, if any Lender becomes a Defaulting Lender, then to the extent permitted by Applicable Law, the following provisions shall apply for so long as such Lender is a Defaulting Lender:
- (a) <u>Waivers and Amendments</u>. Such Defaulting Lender's right to approve or disapprove any amendment, waiver or consent with respect to this Agreement shall be restricted as set forth in the definition of Required Lenders.
- (b) <u>Defaulting Lender Waterfall</u>. Any payment of principal, interest, fees or other amounts received by the Administrative Agent for the account of such Defaulting Lender (whether voluntary or mandatory, at maturity, pursuant to <u>Section 11.1</u> or otherwise) or received by the Administrative Agent from a Defaulting Lender pursuant to <u>Section 2.20(b)</u> shall be applied at such time or times as may be determined by the Administrative Agent as follows: <u>first</u>, to the payment of any amounts owing by such Defaulting Lender to the Administrative Agent hereunder; <u>second</u>, to the payment on a pro rata basis of any amounts owing by such Defaulting Lender to any Issuing Bank or the Swingline Lender hereunder; <u>third</u>, to Cash Collateralize any Issuing Bank's Fronting Exposure ratably with respect to such Defaulting Lender in accordance with subsection (e) below; <u>fourth</u>, as the Borrower may request (so long as no Default or Event of Default exists), to the funding of any Loan in respect of which such Defaulting Lender has failed to fund its portion thereof as required by this Agreement, as determined by the Administrative Agent; <u>fifth</u>, if so determined by the Administrative Agent and the Borrower, to be held in a deposit account and released pro rata in order to (x) satisfy such Defaulting Lender's potential future funding obligations with respect to Loans under this Agreement and (y) Cash Collateralize any Issuing Bank's future Fronting Exposure ratably with respect to such Defaulting Lender with respect to future Facility Letters of Credit issued by such Issuing Bank under this Agreement, in accordance with subsection (e) below; <u>sixth</u>, to the payment of any amounts owing to the Lenders, the Issuing Banks or the

Swingline Lender as a result of any judgment of a court of competent jurisdiction obtained by any Lender, any Issuing Bank or the Swingline Lender against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; seventh, so long as no Default or Event of Default exists, to the payment of any amounts owing to the Borrower as a result of any judgment of a court of competent jurisdiction obtained by the Borrower against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; and eighth, to such Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if (x) such payment is a payment of the principal amount of any Loans or amounts owing by such Defaulting Lender under Section 3.6 in respect of Facility Letters of Credit (such amounts "Principal L/C Disbursements"), in respect of which such Defaulting Lender has not fully funded its appropriate share, and (y) such Loans were made or the related Facility Letters of Credit were issued at a time when the conditions set forth in Article IV were satisfied or waived, such payment shall be applied solely to pay the Loans of, and Principal L/C Disbursements owed to, all Non-Defaulting Lenders on a pro rata basis prior to being applied to the payment of any Loans of, or Principal L/C Disbursements owed to, such Defaulting Lender until such time as all Loans and funded and unfunded participations in LC Exposure and Swingline Loans are held by the Lenders pro rata in accordance with their respective Percentages (determined without giving effect to the immediately following subsection (d)). Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender or to post Cash Collateral pursuant to this subsection shall be deemed paid to and redirected by such Defaulting Lender, and each Lender irrevocably consents hereto.

(c) Certain Fees.

- (i) No Defaulting Lender shall be entitled to receive any fee payable under <u>Section 2.10</u> for any period that no Investment Grade Rating shall exist, during which period that Lender is a Defaulting Lender (and the Borrower shall not be required to pay any such fee that otherwise would have been required to have been paid to that Defaulting Lender). Each Defaulting Lender shall be entitled to receive the fee payable under <u>Section 2.10</u> for any period that an Investment Grade Rating shall exist, during which Period that Lender is a Defaulting Lender, but only to extent allocable to the sum of (1) the outstanding principal amount of the Revolving Loans funded by it, and (2) its Percentage of the face amount of Facility Letters of Credit for which it has provided Cash Collateral pursuant to the immediately following subsection (e).
- (ii) Each Defaulting Lender shall be entitled to receive any fee payable under <u>Section 3.8</u> for any period during which that Lender is a Defaulting Lender only to the extent allocable to its Percentage of the stated amount of Facility Letters of Credit for which it has provided Cash Collateral pursuant to the immediately following subsection (e).
- (iii) With respect to any fee not required to be paid to any Defaulting Lender pursuant to the immediately preceding clauses (i) or (ii), the Borrower shall (x) pay to each NonDefaulting Lender that portion of any such fee otherwise payable to such Defaulting Lender with respect to such Defaulting Lender's participation in LC Exposure or Swingline Loans that has been reallocated to such NonDefaulting Lender pursuant to the immediately following subsection (d), (y) pay to any Issuing Bank and Swingline Lender, as applicable, the amount of any such fee otherwise payable to such Defaulting Lender to the extent allocable to such Issuing Bank's or Swingline Lender's Fronting Exposure to such Defaulting Lender, and (z) not be required to pay the remaining amount of any such fee.
- (d) <u>Reallocation of Participations to Reduce Fronting Exposure</u>. All or any part of such Defaulting Lender's participation in LC Exposure and Swingline Loans shall be reallocated among the

Non-Defaulting Lenders in accordance with their respective Percentages (determined without regard to such Defaulting Lender's Revolving Commitment) but only to the extent that (x) the conditions set forth in Section 5.2 are satisfied at the time of such reallocation (and, unless the Borrower shall have otherwise notified the Administrative Agent at such time, the Borrower shall be deemed to have represented and warranted that such conditions are satisfied at such time), and (y) such reallocation does not cause the Revolving Exposure of any Non-Defaulting Lender to exceed such Non-Defaulting Lender's Revolving Commitment. No reallocation hereunder shall constitute a waiver or release of any claim of any party hereunder against a Defaulting Lender arising from that Lender having become a Defaulting Lender, including any claim of a Non-Defaulting Lender's increased exposure following such reallocation.

(e) Cash Collateral, Repayment of Swingline Loans.

- (i) If the reallocation described in the immediately preceding subsection (d) above cannot, or can only partially, be effected, the Borrower shall, without prejudice to any right or remedy available to it hereunder or under law, (x) first, prepay Swingline Loans in an amount equal to the Swingline Lender's Fronting Exposure and (y) second, Cash Collateralize any Issuing Bank's Fronting Exposure in accordance with the procedures set forth in this subsection.
- (ii) At any time that there shall exist a Defaulting Lender, within ten (10) Business Days following the written request of the Administrative Agent or any Issuing Bank (with a copy to the Administrative Agent), the Borrower shall Cash Collateralize such Issuing Bank's Fronting Exposure with respect to such Defaulting Lender (determined after giving effect to the immediately preceding subsection (d) and any Cash Collateral provided by such Defaulting Lender) in an amount not less than the aggregate Fronting Exposure of such Issuing Bank with respect to Facility Letters of Credit issued and outstanding by such Issuing Bank at such time.
- (iii) The Borrower, and to the extent provided by any Defaulting Lender, such Defaulting Lender, hereby grant to the Administrative Agent, for the benefit of each Issuing Bank, and agree to maintain, a first priority security interest in all such Cash Collateral as security for the Defaulting Lender's obligation to fund participations in respect of LC Exposure, to be applied pursuant to the immediately following clause (iv). If at any time the Administrative Agent determines that Cash Collateral is subject to any right or claim of any Person other than the Administrative Agent and any Issuing Bank as herein provided, or that the total amount of such Cash Collateral is less than the aggregate Fronting Exposure of such Issuing Bank with respect to Facility Letters of Credit issued and outstanding at such time, the Borrower will, promptly upon demand by the Administrative Agent, pay or provide to the Administrative Agent additional Cash Collateral in an amount sufficient to eliminate such deficiency (after giving effect to any Cash Collateral provided by the Defaulting Lender).
- (iv) Notwithstanding anything to the contrary contained in this Agreement, Cash Collateral provided under this Section in respect of Facility Letters of Credit shall be applied to the satisfaction of the Defaulting Lender's obligation to fund participations in respect of LC Exposure (including, as to Cash Collateral provided by a Defaulting Lender, any interest accrued on such obligation) for which the Cash Collateral was so provided, prior to any other application of such property as may otherwise be provided for herein.
- (v) Cash Collateral (or the appropriate portion thereof) provided to reduce ratably any Issuing Bank's Fronting Exposure shall no longer be required to be held as Cash Collateral pursuant to this subsection following (x) the elimination of the applicable Fronting Exposure (including by the termination of Defaulting Lender status of the applicable Lender), or (y) the

determination by the Administrative Agent and any Issuing Bank that there exists excess Cash Collateral; <u>provided</u> that, subject to the immediately preceding subsection (b), the Person providing Cash Collateral and the applicable Issuing Bank may agree that Cash Collateral shall be held to support future anticipated Fronting Exposure or other obligations.

- (f) <u>Defaulting Lender Cure</u>. If the Borrower, the Administrative Agent, the Swingline Lender and the Issuing Banks agree in writing that a Lender is no longer a Defaulting Lender, the Administrative Agent will so notify the parties hereto, whereupon as of the effective date specified in such notice and subject to any conditions set forth therein (which may include arrangements with respect to any Cash Collateral), that Lender will, to the extent applicable, purchase at par that portion of outstanding Loans of the other Lenders (other than Bid Rate Loans) or take such other actions as the Administrative Agent may determine to be necessary to cause the Loans (other than Bid Rate Loans) and funded and unfunded participations in Facility Letters of Credit and Swingline Loans to be held pro rata by the Lenders in accordance with their respective Percentages (determined without giving effect to the immediately preceding subsection (d)), whereupon such Lender will cease to be a Defaulting Lender; provided that no adjustments will be made retroactively with respect to fees accrued or payments made by or on behalf of the Borrower while that Lender was a Defaulting Lender; and provided, further, that except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Defaulting Lender to Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender's having been a Defaulting Lender.
- (g) New Swingline Loans/Facility Letters of Credit. So long as any Lender is a Defaulting Lender, (i) the Swingline Lender shall not be required to fund any Swingline Loans unless it is satisfied that it will have no Fronting Exposure after giving effect to such Swingline Loan and (ii) no Issuing Bank shall be required to issue, extend, renew or increase any Facility Letter of Credit unless it is satisfied that it will have no Fronting Exposure after giving effect thereto.
- 12.17. <u>Delegation to Affiliates</u>. The Borrower and the Lenders agree that the Administrative Agent may delegate any of its duties under this Agreement to any of its Affiliates. Any such Affiliate (and such Affiliate's directors, officers, agents and employees) which performs duties in connection with this Agreement shall be entitled to the same benefits of the indemnification, waiver and other protective provisions to which the Administrative Agent is entitled under <u>Articles XII</u> and <u>XIV</u>.
- 12.18. <u>Managing Agents, Co-Documentation Agents, Co-Syndication Agents, etc.</u>. Neither any of the Lenders identified in this Agreement as a "managing agent" nor the Co-Documentation Agents or the Co-Syndication Agents shall have any right, power, obligation, liability, responsibility or duty under this Agreement other than those applicable to all Lenders as such. Without limiting the foregoing, none of such Lenders shall have or be deemed to have a fiduciary relationship with any Lender. Each Lender hereby makes the same acknowledgments with respect to such Lenders as it makes with respect to the Administrative Agent in <u>Section 12.11</u>.

12.19. Erroneous Payments.

(a) Each Lender, each Issuing Bank, and any other party hereto hereby severally agrees that if (i) the Administrative Agent notifies (which such notice shall be conclusive absent manifest error) such Lender or Issuing Bank or any other Person that has received funds from the Administrative Agent or any of its Affiliates, either for its own account or on behalf of a Lender, or Issuing Bank (each such recipient, a "Payment Recipient") that the Administrative Agent has determined in its sole discretion that any funds received by such Payment Recipient were erroneously transmitted to, or otherwise erroneously or mistakenly received by, such Payment

Recipient (whether or not known to such Payment Recipient) or (ii) any Payment Recipient receives any payment from the Administrative Agent (or any of its Affiliates) (x) that is in a different amount than, or on a different date from, that specified in a notice of payment, prepayment or repayment sent by the Administrative Agent (or any of its Affiliates) with respect to such payment, prepayment or repayment, as applicable, (y) that was not preceded or accompanied by a notice of payment, prepayment or repayment sent by the Administrative Agent (or any of its Affiliates) with respect to such payment, prepayment or repayment, as applicable, or (z) that such Payment Recipient otherwise becomes aware was transmitted or received in error or by mistake (in whole or in part) then, in each case, an error in payment shall be presumed to have been made (any such amounts specified in clauses (i) or (ii) of this Section 12.19(a), whether received as a payment, prepayment or repayment of principal, interest, fees, distribution or otherwise; individually and collectively, an "Erroneous Payment"), then, in each case, such Payment Recipient is deemed to have knowledge of such error at the time of its receipt of such Erroneous Payment; provided that nothing in this Section shall require the Administrative Agent to provide any of the notices specified in clauses (i) or (ii) above. Each Payment Recipient agrees that it shall not assert any right or claim to any Erroneous Payment, and hereby waives any claim, counterclaim, defense or right of set-off or recoupment with respect to any demand, claim or counterclaim by the Administrative Agent for the return of any Erroneous Payments, including without limitation waiver of any defense based on "discharge for value" or any similar doctrine.

- (b) Without limiting the immediately preceding <u>clause (a)</u>, each Payment Recipient agrees that, in the case of clause (a)(ii) above, it shall promptly notify the Administrative Agent in writing of such occurrence.
- In the case of either clause (a)(i) or (a)(ii) above, such Erroneous Payment shall at all times remain the property of the Administrative Agent and shall be segregated by the Payment Recipient and held in trust for the benefit of the Administrative Agent, and upon demand from the Administrative Agent such Payment Recipient shall (or, shall cause any Person who received any portion of an Erroneous Payment on its behalf to), promptly, but in all events no later than two Business Days thereafter, return to the Administrative Agent the amount of any such Erroneous Payment (or portion thereof) as to which such a demand was made in same day funds and in the currency so received, together with interest thereon in respect of each day from and including the date such Erroneous Payment (or portion thereof) was received by such Payment Recipient to the date such amount is repaid to the Administrative Agent at the greater of the Federal Funds Effective Rate and a rate determined by the Administrative Agent in accordance with banking industry rules on interbank compensation from time to time in effect; provided that, if the Payment Recipient is the Borrower, (a) the accrual of such interest shall be at the Federal Funds Effective Rate and shall commence to accrue from and including the second Business Day following the date such Erroneous Payment (or portion thereof) was received by the Borrower (the "Borrower EP Interest Accrual Date") to the date such amount is repaid to the Administrative Agent and (b) if such Erroneous Payment (or portion thereof) shall have been returned to the Administrative Agent prior to the Borrower EP Interest Accrual Date, then the Borrower shall not have any further obligations or liability with respect to such Erroneous Payment, including for any interest, charges, fees, costs or expenses incurred as a result of the Administrative Agent making the Erroneous Payment to the Borrower, the Borrower holding the Erroneous Payment (or portion thereof) in trust for the benefit of the Administrative Agent and the Borrower returning the Erroneous Payment (or portion thereof) to the Administrative Agent.

- (d) In the event that an Erroneous Payment (or portion thereof) is not recovered by the Administrative Agent for any reason, after demand therefor by the Administrative Agent in accordance with immediately preceding clause (c), from any Lender that is a Payment Recipient or an Affiliate of a Payment Recipient (such unrecovered amount as to such Lender, an "Erroneous Payment Return <u>Deficiency</u>"), then at the sole discretion of the Administrative Agent and upon the Administrative Agent's written notice to such Lender (i) such Lender shall be deemed to have made a cashless assignment of the full face amount of the portion of its Loans (but not its Revolving Commitments) of the relevant Class with respect to which such Erroneous Payment was made (the "Erroneous Payment Impacted Class") to the Administrative Agent or, at the option of the Administrative Agent, the Administrative Agent's applicable lending affiliate in an amount that is equal to the Erroneous Payment Return Deficiency (or such lesser amount as the Administrative Agent may specify) (such assignment of the Loans (but not Revolving Commitments) of the Erroneous Payment Impacted Class, the "Erroneous Payment Deficiency Assignment") plus any accrued and unpaid interest on such assigned amount, without further consent or approval of any party hereto and without any payment by the Administrative Agent or its applicable lending affiliate as the assignee of such Erroneous Payment Deficiency Assignment. Without limitation of its rights hereunder, the Administrative Agent may cancel any Erroneous Payment Deficiency Assignment at any time by written notice to the applicable assigning Lender and upon such revocation all of the Loans assigned pursuant to such Erroneous Payment Deficiency Assignment shall be reassigned to such Lender without any requirement for payment or other consideration. The parties hereto acknowledge and agree that (1) any assignment contemplated in this clause (d) shall be made without any requirement for any payment or other consideration paid by the applicable assignee or received by the assignor, (2) the provisions of this clause (d) shall govern in the event of any conflict with the terms and conditions of Article XIII and (3) the Administrative Agent may reflect such assignments in the Register without further consent or action by any other Person.
- (e) Each party hereto hereby agrees that (x) in the event an Erroneous Payment (or portion thereof) is not recovered from any Payment Recipient that has received such Erroneous Payment (or portion thereof) for any reason, the Administrative Agent (1) shall be subrogated to all the rights of such Payment Recipient with respect to such amount and (2) is authorized to set off, net and apply any and all amounts at any time owing to such Payment Recipient under any Loan Document, or otherwise payable or distributable by the Administrative Agent to such Payment Recipient from any source, against any amount due to the Administrative Agent under this Section 12.19 or under the indemnification provisions of this Agreement, (y) the receipt of an Erroneous Payment by a Payment Recipient shall not for the purpose of this Agreement be treated as a payment, prepayment, repayment, discharge or other satisfaction of any Obligations owed by the Borrower or any other Loan Party, except, in each case, to the extent such Erroneous Payment is, and solely with respect to the amount of such Erroneous Payment that is, comprised of funds received by the Administrative Agent from the Borrower or any other Loan Party for the purpose of making for a payment on the Obligations and (z) to the extent that an Erroneous Payment was in any way or at any time credited as payment or satisfaction of any of the Obligations, the Obligations or any part thereof that were so credited, and all rights of the Payment Recipient, as the case may be, shall be reinstated and continue in full force and effect as if such payment or satisfaction had never been received.
- (f) Each party's obligations under this <u>Section 12.19</u> shall survive the resignation or replacement of the Administrative Agent or any transfer of right or obligations by, or the

replacement of, a Lender, the termination of the Revolving Commitments or the repayment, satisfaction or discharge of all Obligations (or any portion thereof) under any Loan Document.

(g) Nothing in this <u>Section 12.19</u> will constitute a waiver or release of any claim of any party hereunder arising from any Payment Recipient's receipt of an Erroneous Payment.

ARTICLE XIII.

BENEFIT OF AGREEMENT; ASSIGNMENTS; PARTICIPATIONS

Successors and Assigns. The terms and provisions of the Loan Documents shall be binding upon and inure to the benefit of the Borrower and the Lenders and their respective successors and assigns permitted hereby, except that (i) the Borrower shall not have the right to assign its rights or obligations under the Loan Documents without the prior written consent of each Lender, (ii) any assignment by any Lender must be made in compliance with Section 13.3, and (iii) any transfer by Participation must be made in compliance with Section 13.2. Any attempted assignment or transfer by any party not made in compliance with this Section 13.1 shall be null and void, unless such attempted assignment or transfer is treated as a participation in accordance with Section 13.3. The parties to this Agreement acknowledge that clause (ii) of this Section 13.1 relates only to absolute assignments and this <u>Section 13.1</u> does not prohibit assignments creating security interests, including, without limitation, (x) any pledge or assignment by any Lender of all or any portion of its rights under this Agreement and any Note to a Federal Reserve Bank or other central banking authority or (y) in the case of a Lender which is a Fund, any pledge or assignment of all or any portion of its rights under this Agreement and any Note to its trustee in support of its obligations to its trustee; provided, however, that no such pledge or assignment creating a security interest shall release the transferor Lender from its obligations hereunder unless and until the parties thereto have complied with the provisions of Section 13.3. The Administrative Agent may treat the Person which made any Loan or which holds any Note as the owner thereof for all purposes hereof unless and until such Person complies with Section 13.3; provided, however, that the Administrative Agent may in its discretion (but shall not be required to) follow instructions from the Person which made any Loan or which holds any Note to direct payments relating to such Loan or Note to another Person. Any assignee of the rights to any Loan or any Note agrees by acceptance of such assignment to be bound by all the terms and provisions of the Loan Documents. Any request, authority or consent of any Person, who at the time of making such request or giving such authority or consent is the owner of the rights to any Loan (whether or not a Note has been issued in evidence thereof), shall be conclusive and binding on any subsequent holder or assignee of the rights to such Loan.

13.2. Participations.

13.2.1 <u>Permitted Participants</u>; <u>Effect</u>. Any Lender may at any time sell to one or more banks or other entities (other than a natural Person, or a holding company, investment vehicle or trust for, or owned and operated for the primary benefit of, a natural Person, any Defaulting Lender, or the Borrower or any of the Borrower's Affiliates or Subsidiaries) ("<u>Participants</u>") participating interests in any Loan owing to such Lender, any Note held by such Lender, any Revolving Commitment of such Lender or any other interest of such Lender under the Loan Documents. In the event of any such sale by a Lender of participating interests to a Participant, such Lender's obligations under the Loan Documents shall remain unchanged, such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations, such Lender shall remain the owner of its Loans and the holder of any Note issued to it in evidence thereof for all purposes under the Loan Documents, all amounts payable by the Borrower under

this Agreement shall be determined as if such Lender had not sold such participating interests, and the Borrower and the Administrative Agent shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under the Loan Documents.

- 13.2.2 <u>Voting Rights</u>. Each Lender shall retain the sole right to approve, without the consent of any Participant, any amendment, modification or waiver of any provision of the Loan Documents other than any amendment, modification or waiver with respect to any Loan or Revolving Commitment in which such Participant has an interest which would require consent of all of the Lenders pursuant to the terms of <u>Section 14.13</u> or of any other Loan Document.
- 13.2.3 <u>Benefit of Certain Provisions</u>. The Borrower agrees that each Participant shall be deemed to have the right of setoff provided in <u>Section 14.15(a)</u> in respect of its participating interest in amounts owing under the Loan Documents to the same extent as if the amount of its participating interest were owing directly to it as a Lender under the Loan Documents, <u>provided</u> that each Lender shall retain the right of setoff provided in <u>Section 14.15(a)</u> with respect to the amount of participating interests sold to each Participant. The Lenders agree to share with each Participant, and each Participant, by exercising the right of setoff provided in <u>Section 14.15(a)</u>, agrees to share with each Lender, any amount received pursuant to the exercise of its right of setoff, such amounts to be shared in accordance with <u>Section 14.15(b)</u> as if each Participant were a Lender. The Borrower further agrees that each Participant shall be entitled to the benefits of <u>Sections 4.1, 4.2, 4.4</u> and <u>4.5</u> to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to <u>Section 13.3</u>, <u>provided</u> that (i) a Participant shall not be entitled to receive any greater payment under <u>Section 4.1, 4.2, 4.4</u> or <u>4.5</u> than the Lender who sold the participating interest to such Participant would have received had it retained such interest for its own account, unless the sale of such interest to such Participant is made with the prior written consent of the Borrower, and (ii) any Participant not incorporated under the laws of the United States of America or any State thereof agrees to comply with the provisions of <u>Section 4.5</u> to the same extent as if it were a Lender.
- 13.2.4 <u>Participant Register</u>. Each Lender that sells a participation shall, acting solely for this purpose as a non-fiduciary agent of the Borrower, maintain a register on which it enters the name and address of each Participant and the principal amounts (and stated interest) of each Participant's interest in the Loans or other obligations under this Agreement (the "<u>Participant Register</u>"); provided that no Lender shall have any obligation to disclose all or any portion of the Participant Register to any Person (including the identity of any Participant or any information relating to a Participant's interest in any Revolving Commitments, Loans, Facility Letters of Credit or its other obligations under this Agreement) except to the extent that such disclosure is necessary to establish that such Revolving Commitment, Loan, Facility Letter of Credit or other obligation is in registered form under Section 5f.103-1(c) of the United States Treasury Regulations. The entries in the Participant Register shall be conclusive absent manifest error, and such Lender shall treat each person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary.

13.3. <u>Assignments; Consents.</u> Any Lender may at any time assign to one or more Eligible Assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Revolving Commitment and the Loans at the time owing to it); provided that any such assignment shall be subject to the following conditions:

13.3.1 Minimum Amounts.

- (A) in the case of an assignment of the entire remaining amount of an assigning Revolving Lender's Revolving Commitment and the Loans at the time owing to it, or in the case of an assignment to a Lender, an Affiliate of a Lender or an Approved Fund, no minimum amount need be assigned; and
- (B) in any case not described in the immediately preceding subsection (A), the aggregate amount of the Revolving Commitment (which for this purpose includes Loans outstanding thereunder) or, if the applicable Revolving Commitment is not then in effect, the principal outstanding balance of the Loans of the assigning Lender subject to each such assignment (determined as of the date the Assignment and Assumption Agreement with respect to such assignment is delivered to the Administrative Agent or, if "Trade Date" is specified in the Assignment and Assumption Agreement, as of the Trade Date) shall not be less than \$5,000,000, unless each of the Administrative Agent, the Issuing Banks and, so long as no Event of Default has occurred and is continuing, the Borrower otherwise consents (each such consent not to be unreasonably withheld or delayed); provided, however, that if, after giving effect to such assignment, the amount of the Revolving Commitment held by such assigning Lender or the outstanding principal balance of the Loans of such assigning Lender, as applicable, would be less than \$5,000,000, then such assigning Lender shall assign the entire amount of its Revolving Commitment and the Loans at the time owing to it.
- 13.3.2 <u>Proportionate Amounts</u>. Each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Loan or the Revolving Commitment assigned.
- 13.3.3 <u>Required Consents</u>. No consent shall be required for any assignment except to the extent required by <u>Section 13.3.1(B)</u> and, in addition:
 - (A) the consent of the Borrower (such consent not to be unreasonably withheld or delayed) shall be required unless (x) an Event of Default has occurred and is continuing at the time of such assignment or (y) such assignment is to a Lender, an Affiliate of a Lender or an Approved Fund; provided that the Borrower shall be deemed to have consented to any such assignment unless it shall object thereto by written notice to the Administrative Agent within five (5) Business Days after having received notice thereof;
 - (B) the consent of the Administrative Agent (such consent not to be unreasonably withheld or delayed) shall be required for assignments in respect of a Revolving Commitment or a Loan if such assignment is to a Person that is not already a Lender, an Affiliate of such a Lender or an Approved Fund with respect to such a Lender; and

- (C) the consent of the Swingline Lender and the Issuing Banks (such consent not to be unreasonably withheld or delayed) shall be required for any assignment in respect of a Revolving Commitment.
- 13.3.4 <u>Assignment and Acceptance; Notes.</u> The parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption Agreement, together with a processing and recordation fee of \$3,500 for each assignment, and the assignee, if it is not a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire. If requested by the transferor Lender or the Assignee, upon the consummation of any assignment, the transferor Lender, the Administrative Agent and the Borrower shall make appropriate arrangements so that new Notes are issued to the Assignee and such transferor Lender, as appropriate.
- 13.3.5 <u>No Assignment to Borrower or Defaulting Lender.</u> No such assignment shall be made to (i) the Borrower or any of the Borrower's Affiliates or Subsidiaries or (ii) any Defaulting Lender.
- 13.3.6 <u>No Assignment to Natural Persons</u>. No such assignment shall be made to a natural person or a holding company, investment vehicle or trust for, or owned and granted for the primary benefit of, a natural Person.

Subject to acceptance and recording thereof by the Administrative Agent pursuant to Section 13.4, from and after the effective date specified in each Assignment and Assumption Agreement, the assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption Agreement, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Assumption Agreement, be released from its obligations under this Agreement (and, in the case of an Assignment and Assumption Agreement covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto) but shall continue to be entitled to the benefits of Sections 12.8 and 14.6 and the other provisions of this Agreement and the other Loan Documents that survive the termination hereof and thereof with respect to facts and circumstances occurring prior to the effective date of such assignment. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this paragraph shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with Section 13.2.

- 13.4. Register. The Administrative Agent, acting solely for this purpose as a non-fiduciary agent of the Borrower, shall maintain at the Administrative Office a copy of each Assignment and Assumption Agreement delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Revolving Commitments of, and principal amounts (and stated interest) of the Loans owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive, and the Borrower, the Administrative Agent and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by the Borrower and any Lender, at any reasonable time and from time to time upon reasonable prior notice.
- 13.5. <u>Designated Lenders</u>. Any Lender (each, a "Designating Lender") may at any time while the Borrower has been assigned an Investment Grade Rating from either S&P or Moody's and the Borrower shall have elected to have the provisions of <u>Section 2.9(b)</u> apply, designate one Designated Lender to fund Bid Rate Loans on behalf of such Designating Lender subject to the terms of this Section,

and the provisions in the immediately preceding Sections 13.2 and 13.3 shall not apply to such designation. No Lender may designate more than one Designated Lender. The parties to each such designation shall execute and deliver to the Administrative Agent for its acceptance a Designation Agreement. Upon such receipt of an appropriately completed Designation Agreement executed by a Designating Lender and a designee representing that it is a Designated Lender, the Administrative Agent will accept such Designation Agreement and give prompt notice thereof to the Borrower, whereupon (i) the Borrower shall execute and deliver to the Designating Lender a Bid Rate Note payable to the order of the Designated Lender, (ii) from and after the effective date specified in the Designation Agreement, the Designated Lender shall become a party to this Agreement with a right to make Bid Rate Loans on behalf of its Designating Lender pursuant to Section 2.18 after the Borrower has accepted a Bid Rate Loan (or portion thereof) of the Designating Lender, and (iii) the Designated Lender shall not be required to make payments with respect to any obligations in this Agreement except to the extent of excess cash flow of such Designated Lender which is not otherwise required to repay obligations of such Designated Lender which are then due and payable; provided, however, that regardless of such designation and assumption by the Designated Lender, the Designating Lender shall be and remain obligated to the Borrower, the Administrative Agent and the Lenders for each and every of the obligations of the Designating Lender and its related Designated Lender with respect to this Agreement, including, without limitation, any indemnification obligations under Section 12.3 and any sums otherwise payable to the Borrower by the Designated Lender. Each Designating Lender shall serve as the agent of the Designated Lender and shall on behalf of, and to the exclusion of, the Designated Lender: (i) receive any and all payments made for the benefit of the Designated Lender and (ii) give and receive all communications and notices and take all actions hereunder, including, without limitation, votes, approvals, waivers, consents and amendments under or relating to this Agreement and the other Loan Documents. Any such notice, communication, vote, approval, waiver, consent or amendment shall be signed by the Designating Lender as agent for the Designated Lender and shall not be signed by the Designated Lender on its own behalf and shall be binding on the Designated Lender to the same extent as if signed by the Designated Lender on its own behalf. The Borrower, the Administrative Agent and the Lenders may rely thereon without any requirement that the Designated Lender sign or acknowledge the same. No Designated Lender may assign or transfer all or any portion of its interest hereunder or under any other Loan Document, other than assignments to the Designating Lender which originally designated such Designated Lender. The Borrower, the Lenders and the Administrative Agent each hereby agrees that it will not institute against any Designated Lender or join any other Person in instituting against any Designated Lender any bankruptcy, reorganization, arrangement, insolvency or liquidation proceeding under any federal or state bankruptcy or similar law, until the later to occur of (x) one year and one day after the payment in full of the latest maturing commercial paper note issued by such Designated Lender and (y) the Maturity Date. In connection with any such designation, the Designating Lender shall pay to the Administrative Agent an administrative fee for processing such designation in the amount of \$3,500.

13.6. <u>Confidentiality</u>. Except as otherwise provided by Applicable Law, the Administrative Agent, each Issuing Bank and each Lender shall maintain the confidentiality of all Information (as defined below) in accordance with its customary procedure for handling confidential information of this nature and in accordance with safe and sound banking practices but in any event may make disclosure: (a) to its Affiliates and to its and its Affiliates' respective partners, directors, officers, employees, agents, advisors and other representatives (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential); (b) subject to an agreement containing provisions substantially the same as those of this Section, to (i) any actual or proposed Eligible Assignee, Participant or other transferee in connection with a potential transfer of any Revolving Commitment or participation therein as permitted hereunder, or (ii) any actual or prospective counterparty (or its advisors) to any swap or derivative transaction relating to the Borrower and its obligations; (c) as required or requested by any Governmental Authority or representative thereof or pursuant to legal process or in connection with any legal proceedings, or as

otherwise required by Applicable Law; (d) to the Administrative Agent's, such Issuing Bank's or such Lender's independent auditors and other professional advisors (provided they shall be notified of the confidential nature of the information); (e) in connection with the exercise of any remedies under any Loan Document or any action or proceeding relating to any Loan Document or the enforcement of rights hereunder or thereunder; (f) to the extent such Information (i) becomes publicly available other than as a result of a breach of this Section actually known by the Administrative Agent, such Issuing Bank or such Lender to be a breach of this Section or (ii) becomes available to the Administrative Agent, any Issuing Bank, any Lender or any Affiliate of the Administrative Agent, any Issuing Bank or any Lender on a nonconfidential basis from a source other than the Borrower or any Affiliate of the Borrower; (g) to the extent requested by, or required to be disclosed to, any nationally recognized rating agency or regulatory or similar authority (including any self-regulatory authority, such as the National Association of Insurance Commissioners) having or purporting to have jurisdiction over it; (h) to bank trade publications, such information to consist of deal terms and other information customarily found in such publications; (i) to any other party hereto; and (j) with the consent of the Borrower. Notwithstanding the foregoing, the Administrative Agent, each Issuing Bank and each Lender may disclose any such confidential information, without notice to the Borrower or the General Partner, to Governmental Authorities in connection with any regulatory examination of the Administrative Agent, such Issuing Bank or such Lender or in accordance with the regulatory compliance policy of the Administrative Agent, such Issuing Bank or such Lender. As used in this Section, the term "Information" means all information received from the Borrower, the General Partner, any Subsidiary or Affiliate of the Borrower relating to the Borrower, the General Partner or any of their Affiliates or any of their respective businesses, other than any such information that is available to the Administrative Agent, any Lender or any Issuing Bank on a nonconfidential basis prior to disclosure by the Borrower, the General Partner, or any Subsidiary or any Affiliate of the Borrower, provided that, in the case of any such information received from the Borrower, the General Partner, or any Subsidiary or any Affiliate of the Borrower after the date hereof, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information.

13.7. <u>Tax Treatment</u>. If any interest in any Loan Document is transferred to any Participant or Purchaser or any other Person acquiring an interest in the Loan Documents by operation of law (each a "Transferee") which is not incorporated under the laws of the United States or any State thereof, the transferor Lender shall cause such Transferee, concurrently with the effectiveness of such transfer, to comply with the provisions of <u>Section 4.5(iv)</u>.

ARTICLE XIV.

GENERAL PROVISIONS

- 14.1. <u>Survival of Representations</u>. All representations and warranties contained in this Agreement shall survive delivery of the Notes and the making of the Borrowings herein contemplated.
- 14.2. <u>Governmental Regulation</u>. Anything contained in this Agreement to the contrary notwithstanding, no Lender shall be obligated to extend credit to the Borrower in violation of any limitation or prohibition provided by any applicable statute or regulation.
- 14.3. <u>Taxes</u>. Any recording and other taxes (excluding franchise, income or similar taxes) or other similar assessments or charges payable or ruled payable by any governmental authority incurred in connection with the consummation of the transactions contemplated by this Agreement shall be paid by the Borrower, together with interest and penalties, if any.

- 14.4. <u>Headings</u>. Section headings in the Loan Documents are for convenience of reference only, and shall not govern the interpretation of any of the provisions of the Loan Documents.
- 14.5. <u>No Third Party Beneficiaries</u>. This Agreement shall not be construed so as to confer any right or benefit upon any Person other than the parties to this Agreement and their respective successors and assigns.
- Expenses; Indemnification. Subject to the provisions of this Agreement, the Borrower will pay (a) all out-of-pocket costs and expenses incurred by the Administrative Agent and the Arranger (including the reasonable fees, out-of-pocket expenses and other reasonable expenses of counsel) in connection with the preparation, execution and delivery of this Agreement, the Notes, the Loan Documents and any other agreements or documents referred to herein or therein and any amendments thereto, (b) all out-of-pocket costs and expenses incurred by the Administrative Agent and the Lenders (including the reasonable fees, out-of-pocket expenses and other reasonable expenses of counsel to the Administrative Agent and the Lenders) in connection with the enforcement and protection of the rights of the Lenders under this Agreement, the Notes, the Loan Documents or any other agreement or document referred to herein or therein, and (c) all reasonable and customary costs and expenses of periodic audits by the Administrative Agent's personnel of the Borrower's books and records provided that prior to an Event of Default, the Borrower shall be required to pay for only one such audit during any year. The Borrower further agrees to indemnify the Lenders, their Affiliates, and their respective directors, officers, employees, agents and advisors (each, an "Indemnified Party") against all losses, claims, damages, penalties, judgments, liabilities and reasonable expenses (including, without limitation, all expenses of litigation or preparation therefor whether or not the Indemnified Party is a party thereto) which any of them may pay or incur arising out of or relating to this Agreement, the other Loan Documents, the transactions contemplated hereby or the direct or indirect application or proposed application of the proceeds of any Borrowing hereunder, except that the foregoing indemnity shall not apply to any Indemnified Party to the extent that any losses, claims, etc. are determined by a court of competent jurisdiction by a final and non-appealable judgment to have resulted from such Indemnified Party's Gross Negligence or willful misconduct. In the case of an investigation, litigation or proceeding to which the indemnity in this Section applies, such indemnity shall be effective whether or not such investigation, litigation or proceeding is brought by the Borrower or the General Partner or the Borrower's or the General Partner's equity holders or creditors or an Indemnified Party is otherwise party thereto. The obligations of the Borrower under this Section shall survive the termination of this Agreement.
- 14.7. <u>Severability of Provisions</u>. Any provision in any Loan Document that is held to be inoperative, unenforceable, or invalid in any jurisdiction shall, as to that jurisdiction, be inoperative, unenforceable, or invalid without affecting the remaining provisions in that jurisdiction or the operation, enforceability, or validity of that provision in any other jurisdiction, and to this end the provisions of all Loan Documents are declared to be severable.
- 14.8. Nonliability of the Lenders. The relationship between the Borrower and the Lenders shall be solely that of borrower and lender. The Lenders shall not have any fiduciary responsibilities to the Borrower. The Lenders undertake no responsibility to the Borrower to review or inform the Borrower of any matter in connection with any phase of the Borrower's business or operations. Moreover, none of the Administrative Agent, any Issuing Bank or any Lender shall be liable to the Borrower or any other parties. None of the Administrative Agent, any Issuing Bank or any Lender shall have any liability with respect to, and the Borrower hereby waives, releases, and agrees not to sue any of them upon, any claim for any special, indirect, incidental, or consequential damages suffered or incurred by the Borrower in connection with, arising out of, or in any way related to, this Agreement, any of the

other Loan Documents or any fee letter delivered in connection herewith, or any of the transactions contemplated by this Agreement or any of the other Loan Documents.

- 14.9. Choice of Law. THE LOAN DOCUMENTS (OTHER THAN THOSE CONTAINING A CONTRARY EXPRESS CHOICE OF LAW PROVISION) SHALL BE CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS (AND NOT THE LAW OF CONFLICTS) OF THE STATE OF ILLINOIS, BUT GIVING EFFECT TO FEDERAL LAWS APPLICABLE TO NATIONAL BANKS.
- 14.10. <u>Consent to Jurisdiction</u>. THE BORROWER HEREBY IRREVOCABLY SUBMITS TO THE NON EXCLUSIVE JURISDICTION OF ANY UNITED STATES FEDERAL OR ILLINOIS STATE COURT SITTING IN CHICAGO, ILLINOIS IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO ANY LOAN DOCUMENTS AND THE BORROWER HEREBY IRREVOCABLY AGREES THAT ALL CLAIMS IN RESPECT OF SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN ANY SUCH COURT AND IRREVOCABLY WAIVES ANY OBJECTION IT MAY NOW OR HEREAFTER HAVE AS TO THE VENUE OF ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN SUCH A COURT OR THAT SUCH COURT IS AN INCONVENIENT FORUM. NOTHING HEREIN SHALL LIMIT THE RIGHT OF THE LENDERS TO BRING PROCEEDINGS AGAINST THE BORROWER IN THE COURTS OF ANY OTHER JURISDICTION. ANY JUDICIAL PROCEEDING BY THE BORROWER AGAINST THE LENDERS OR ANY AFFILIATE OF THE LENDERS INVOLVING, DIRECTLY OR INDIRECTLY, ANY MATTER IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH ANY LOAN DOCUMENT SHALL BE BROUGHT ONLY IN A COURT IN CHICAGO, ILLINOIS.
- 14.11. <u>Waiver of Jury Trial</u>. THE BORROWER, THE GENERAL PARTNER, THE ADMINISTRATIVE AGENT AND THE LENDERS HEREBY WAIVE TRIAL BY JURY IN ANY JUDICIAL PROCEEDING INVOLVING, DIRECTLY OR INDIRECTLY, ANY MATTER (WHETHER SOUNDING IN TORT, CONTRACT OR OTHERWISE) IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH ANY LOAN DOCUMENT OR THE RELATIONSHIP ESTABLISHED THEREUNDER.
- 14.12. <u>Successors and Assigns</u>. The terms and provisions of the Loan Documents shall be binding upon and inure to the benefit of the Borrower and the Lenders and their respective successors and assigns, except that the Borrower shall not have the right to assign its rights or obligations under the Loan Documents. Any assignee or transferee of rights or obligations hereunder or under the Notes agrees by acceptance thereof to be bound by all the terms and provisions of the Loan Documents. Any request, authority or consent of any Person, who at the time of making such request or giving such authority or consent is the holder of a Note, shall be conclusive and binding on any subsequent holder, transferee or assignee of such Note or of any Note or Notes issued in exchange therefor.
- 14.13. <u>Entire Agreement; Modification of Agreement</u>. The Loan Documents embody the entire agreement among the Borrower, the General Partner, Administrative Agent, and Lenders and supersede all prior conversations, agreements, understandings, commitments and term sheets among any or all of such parties with respect to the subject matter hereof. Any provisions of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is signed by the Borrower, and Administrative Agent if the rights or duties of Administrative Agent are affected thereby, and

- (a) each of the Lenders adversely affected thereby if such amendment or waiver:
- (i) subject to clause (d) below, reduces or forgives any payment of principal or interest on the Obligations or any fees payable by the Borrower to such Lender hereunder; or
- (ii) postpones the date fixed for any payment of principal of or interest on the Obligations or any fees payable by the Borrower to such Lender hereunder; or
- (iii) increases the amount of such Lender's Revolving Commitment (other than pursuant to an assignment permitted under <u>Section 13.3</u> or an increase in the Aggregate Revolving Commitment pursuant to <u>Section 2.2</u>) or the unpaid principal amount of such Lender's Loan; or
 - (iv) extends the Maturity Date (other than pursuant to Section 2.19 hereof); or
 - (v) modifies or waives all or any portion of Section 2.20;
- (b) all of the Lenders if such amendment or waiver:
 - (i) releases or limits the liability of the General Partner under the Loan Documents; or
 - (ii) changes the definition of Required Lenders, or modifies any requirement for consent by each of the Lenders; or
 - (iii) modifies or waives the second through fourth sentences of Section 9.3 hereof;
- (c) the Required Lenders, to the extent expressly provided for herein and in the case of all other waivers or amendments if no percentage of Lenders is specified herein, except that no waivers or amendments (1) affecting Section 2.17 shall be effective without the written consent of the Swingline Lender, (2) affecting Article III shall be effective without the written consent of each Issuing Bank or (3) affecting Section 12.16 shall be effective without the written consent of each of the Swingline Lender and each Issuing Bank; or
- (c) the Administrative Agent and the Borrower may, without the consent of any Lender, enter into amendments or modifications to this Agreement or any of the other Loan Documents or enter into additional Loan Documents as the Administrative Agent reasonably deems appropriate in order to implement any Benchmark Replacement or otherwise effectuate the terms of $\underline{\text{Section 4.8(c)}}$ in accordance with the terms of $\underline{\text{Section 4.8(c)}}$.
- 14.14. <u>Dealings with the Borrower</u>. The Lenders and their affiliates may accept deposits from, extend credit to and generally engage in any kind of banking, trust or other business with the Borrower or the General Partner or any of their Affiliates regardless of the capacity of the Lenders hereunder.

14.15. <u>Set-Off</u>.

(a) Subject to Section 2.20 and in addition to any rights now or hereafter granted under Applicable Law and not by way of limitation of any such rights, the Borrower hereby

authorizes the Administrative Agent, each Issuing Bank, each Lender, each Affiliate of the Administrative Agent, any Issuing Bank or any Lender, and each Participant, at any time, or from time to time while, an Event of Default exists, without notice to the Borrower or to any other Person, any such notice being hereby expressly waived, but in the case of an Issuing Bank, a Lender, an Affiliate of an Issuing Bank or a Lender, or a Participant, subject to receipt of the prior written consent of the Required Lenders exercised in their sole discretion, to set off and to appropriate and to apply any and all deposits (general or special, including, but not limited to, indebtedness evidenced by certificates of deposit, whether matured or unmatured) and any other indebtedness at any time held or owing by the Administrative Agent, such Issuing Bank, such Lender, any Affiliate of the Administrative Agent, such Issuing Bank or such Lender, or such Participant, to or for the credit or the account of the Borrower against and on account of any of the Obligations, irrespective of whether or not any or all of the Loans and all other Obligations have been declared to be, or have otherwise become, due and payable as permitted by Section 11.1, and although such Obligations shall be contingent or unmatured.

(b) Each Lender agrees that if it shall, by exercising any right of set off or counterclaim or otherwise, receive payment of a proportion of the aggregate amount of principal, interest or fees due with respect to any Loan held by it (other than payments received pursuant to Sections 4.1, 4.2 and 4.5) which is greater than the proportion received by any other Lender in respect of the aggregate amount of principal, interest or fees due with respect to any Loan held by such other Lender, the Lender receiving such proportionately greater payment shall purchase such participations in the Loans held by the other Lenders and such other adjustments shall be made as may be required so that all such payments of principal, interest or fees with respect to the Loans held by the Lenders shall be shared by the Lenders pro rata according to their respective Revolving Commitments.

14.16. <u>Counterparts; Integration, Effectiveness, Electronic Execution</u>.

- (a) This Agreement may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. This Agreement and the other Loan Documents, and any separate letter agreements with respect to fees payable to the Administrative Agent, any Issuing Bank, the Swingline Lender and/or any Arranger, constitute the entire contract among the parties relating to the subject matter hereof and supersede any and all previous agreements and understandings, oral or written, relating to the subject matter hereof. Except as provided in Section 5.1, this Agreement shall become effective when it shall have been executed by the Administrative Agent and when the Administrative Agent shall have received counterparts hereof that, when taken together, bear the signatures of each of the other parties hereto. Delivery of an executed counterpart of a signature page of this Agreement by facsimile or in electronic (i.e., "pdf" or "tif") format shall be effective as delivery of a manually executed counterpart of this Agreement.
- (b) The words "execute," "execution," "signed," "signature," "delivery" and words of like import in or related to this Agreement, any other Loan Document or any document, amendment, approval, consent, waiver, modification, information, notice, certificate, report, statement, disclosure, or authorization to be signed or delivered in connection with this Agreement or any other Loan Document or the transactions contemplated hereby shall be deemed to include Electronic Signatures or execution in the form of an Electronic Record, and contract formations on electronic platforms approved by the Administrative Agent, deliveries or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system,

as the case may be, to the extent and as provided for in any Applicable Law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act. Each party hereto agrees that any Electronic Signature or execution in the form of an Electronic Record shall be valid and binding on itself and each of the other parties hereto to the same extent as a manual, original signature. For the avoidance of doubt, the authorization under this paragraph may include, without limitation, use or acceptance by the parties of a manually signed paper which has been converted into electronic form (such as scanned into PDF format), or an electronically signed paper converted into another format, for transmission, delivery and/or retention. Notwithstanding anything contained herein to the contrary, the Administrative Agent is under no obligation to accept an Electronic Signature in any form or in any format unless expressly agreed to by the Administrative Agent pursuant to procedures approved by it; provided that without limiting the foregoing, (a) to the extent the Administrative Agent has agreed to accept such Electronic Signature from any party hereto, the Administrative Agent and the other parties hereto shall be entitled to rely on any such Electronic Signature purportedly given by or on behalf of the executing party without further verification and (b) upon the request of the Administrative Agent or any Lender, any Electronic Signature shall be promptly followed by an original manually executed counterpart thereof. Without limiting the generality of the foregoing, each party hereto hereby (i) agrees that, for all purposes, including without limitation, in connection with any workout, restructuring, enforcement of remedies, bankruptcy proceedings or litigation among the Administrative Agent, the Lenders and any of the Loan Parties, electronic images of this Agreement or any other Loan Document (in each case, including with respect to any signature pages thereto) shall have the same legal effect, validity and enforceability as any paper original, and (ii) waives any argument, defense or right to contest the validity or enforceability of the Loan Documents based solely on the lack of paper original copies of any Loan Documents, including with respect to any signature pages thereto.

- 14.17. <u>Patriot Act CIP Notice</u>. Each Lender hereby notifies the Borrower and the General Partner that pursuant to the requirements of the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "Act), it is from time to time required to obtain, verify and record information that identifies the Borrower and the General Partner, which information includes the name and address of the Borrower and the General Partner and other information that will allow such Lender to identify the Borrower and the General Partner in accordance with the Act.
- 14.18. Acknowledgement and Consent to Bail-In of Affected Financial Institutions. Notwithstanding anything to the contrary in any Loan Document or in any other agreement, arrangement or understanding among any such parties, each party hereto acknowledges that any liability of any Affected Financial Institution arising under any Loan Document, to the extent such liability is unsecured, may be subject to the Write-Down and Conversion Powers of the applicable Resolution Authority and agrees and consents to, and acknowledges and agrees to be bound by:
 - (a) the application of any Write-Down and Conversion Powers by the applicable Resolution Authority to any such liabilities arising hereunder which may be payable to it by any party hereto that is an Affected Financial Institution; and
 - (b) the effects of any Bail-In Action on any such liability, including, if applicable:
 - (i) a reduction in full or in part or cancellation of any such liability;
 - (ii) a conversion of all, or a portion of, such liability into shares or other instruments of ownership in such Affected Financial Institution, its parent undertaking, or

a bridge institution that may be issued to it or otherwise conferred on it, and that such shares or other instruments of ownership will be accepted by it in lieu of any rights with respect to any such liability under this Agreement or any other Loan Document; or

- (iii) the variation of the terms of such liability in connection with the exercise of the Write-Down and Conversion Powers of the applicable Resolution Authority.
- 14.19. Acknowledgement Regarding Any Supported QFCs. To the extent that the Loan Documents provide support, through a guarantee or otherwise, for Rate Management Transactions or any other agreement or instrument that is a QFC (such support, "QFC Credit Support" and, each such QFC, a "Supported QFC"), the parties acknowledge and agree as follows with respect to the resolution power of the FDIC under the Federal Deposit Insurance Act and Title II of the Dodd-Frank Wall Street Reform and Consumer Protection Act (together with the regulations promulgated thereunder, the "U.S. Special Resolution Regimes") in respect of such Supported QFC and QFC Credit Support (with the provisions below applicable notwithstanding that the Loan Documents and any Supported QFC may in fact be stated to be governed by the laws of the State of New York and/or of the United States or any other state of the United States):
 - (a) In the event a Covered Entity that is party to a Supported QFC (each, a "Covered Party") becomes subject to a proceeding under a U.S. Special Resolution Regime, the transfer of such Supported QFC and the benefit of such QFC Credit Support (and any interest and obligation in or under such Supported QFC and such QFC Credit Support, and any rights in property securing such Supported QFC or such QFC Credit Support) from such Covered Party will be effective to the same extent as the transfer would be effective under the U.S. Special Resolution Regime if the Supported QFC and such QFC Credit Support (and any such interest, obligation and rights in property) were governed by the laws of the United States or a state of the United States. In the event a Covered Party or a BHC Act Affiliate of a Covered Party becomes subject to a proceeding under a U.S. Special Resolution Regime, Default Rights under the Loan Documents that might otherwise apply to such Supported QFC or any QFC Credit Support that may be exercised against such Covered Party are permitted to be exercised to no greater extent than such Default Rights could be exercised under the U.S. Special Resolution Regime if the Supported QFC and the Loan Documents were governed by the laws of the United States or a state of the United States. Without limitation of the foregoing, it is understood and agreed that rights and remedies of the parties with respect to a Defaulting Lender shall in no event affect the rights of any Covered Party with respect to a Supported QFC or any QFC Credit Support.
 - (b) As used in this <u>Section 14.19</u>, the following terms have the following meanings:

"BHC Act Affiliate" of a party means an "affiliate" (as such term is defined under, and interpreted in accordance with, 12 U.S.C. 1841(k)) of such party.

"Covered Entity" means any of the following:

- (i) a "covered entity" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 252.82(b);
- (ii) a "covered bank" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 47.3(b); or
- (iii) a "covered FSI" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 382.2(b).

"<u>Default Right</u>" has the meaning assigned to that term in, and shall be interpreted in accordance with, 12 C.F.R. §§ 252.81, 47.2 or 382.1, as applicable.

"QFC" has the meaning assigned to the term "qualified financial contract" in, and shall be interpreted in accordance with, 12 U.S.C. 5390(c)(8) (D).

14.20. Certain ERISA Matters.

- (a) Each Lender (x) represents and warrants, as of the date such Person became a Lender party hereto, to, and (y) covenants, from the date such Person became a Lender party hereto to the date such Person ceases being a Lender party hereto, for the benefit of, the Administrative Agent, each Arranger and their respective Affiliates, and not, for the avoidance of doubt, to or for the benefit of the Borrower or any other Loan Party, that at least one of the following is and will be true:
 - (i) such Lender is not using "plan assets" (within the meaning of Section 3(42) of ERISA or otherwise) of one or more Benefit Plans with respect to such Lender's entrance into, participation in, administration of and performance of the Loans or the Revolving Commitments;
 - (ii) the transaction exemption set forth in one or more PTEs, such as PTE 84-14 (a class exemption for certain transactions determined by independent qualified professional asset managers), PTE 95-60 (a class exemption for certain transactions involving insurance company general accounts), PTE 90-1 (a class exemption for certain transactions involving insurance company pooled separate accounts), PTE 91-38 (a class exemption for certain transactions involving bank collective investment funds) or PTE 96-23 (a class exemption for certain transactions determined by in-house asset managers), is applicable with respect to such Lender's entrance into, participation in, administration of and performance of the Loans, the Revolving Commitments and this Agreement;
 - (iii) (A) such Lender is an investment fund managed by a "Qualified Professional Asset Manager" (within the meaning of Part VI of PTE 84-14), (B) such Qualified Professional Asset Manager made the investment decision on behalf of such Lender to enter into, participate in, administer and perform the Loans, the Revolving Commitments and this Agreement, (C) the entrance into, participation in, administration of and performance of the Loans, the Revolving Commitments and this Agreement satisfies the requirements of sub-Sections (b) through (g) of Part I of PTE 84-14 and (D) to the best knowledge of such Lender, the requirements of subsection (a) of Part I of PTE 84-14 are satisfied with respect to such Lender's entrance into, participation in, administration of and performance of the Loans, the Revolving Commitments and this Agreement; or
 - (iv) such other representation, warranty and covenant as may be agreed in writing between the Administrative Agent, in its sole discretion, and such Lender.

In addition, unless either (1) sub-clause (i) in the immediately preceding clause (a) is true with respect to a Lender or (2) a Lender has provided another representation, warranty and covenant in accordance with sub-clause (iv) in the immediately preceding clause (a), such Lender further (x) represents and warrants, as of the date such Person became a Lender party hereto, to, and (y) covenants, from the date such Person became a Lender party hereto to the date such Person ceases being a Lender party hereto, for the benefit of, the Administrative Agent, each Arranger and their respective Affiliates, and not, for the avoidance of

doubt, to or for the benefit of the Borrower or any other Loan Party, that none of the Administrative Agent, any Arranger and their respective Affiliates is a fiduciary with respect to the assets of such Lender involved in such Lender's entrance into, participation in, administration of and performance of the Loans, the Revolving Commitments and this Agreement (including in connection with the reservation or exercise of any rights by the Administrative Agent under this Agreement, any Loan Document or any documents related hereto or thereto).

ARTICLE XV.

NOTICES

15.19. <u>Giving Notice; Electronic Delivery.</u> All notices and other communications provided to any party hereto under this Agreement or any other Loan Document shall be in writing or by email and addressed or delivered to such party at its address set forth below or at such other address as may be designated by such party in a notice to the other parties. Any notice, if mailed and properly addressed with postage prepaid, shall be deemed given when received; any notice, if transmitted by email, shall be deemed given when transmitted (<u>provided</u> that, if such notice is transmitted by email, such notice is also promptly mailed in accordance with this paragraph). Notice may be given as follows:

To the Borrower:

First Industrial, L.P. c/o First Industrial Realty Trust, Inc. 1 N. Wacker Drive Suite 4200 Chicago, Illinois 60606 Attention: Mr. Scott Musil Email: smusil@firstindustrial.com

To the General Partner:

First Industrial Realty Trust, Inc. 1 N. Wacker Drive Suite 4200 Chicago, Illinois 60606 Attention: Scott A. Musil Email: smusil@firstindustrial.com

Each of the above with a copy to:

Barack Ferrazzano Kirschbaum & Nagelberg LLP 200 West Madison 39th Floor Chicago, Illinois 60606 Attention: Suzanne Bessette-Smith and Matthew A. Jackson Email: Suzanne.bessette-smith@bfkn.com; matt.jackson@bfkn.com

To each Lender:

At such address as set forth in its Administrative Questionnaire

To the Administrative Agent:

Wells Fargo Bank, National Association, as agent 10 South Wacker Drive, 32nd Floor Chicago, IL 60606 Attention: Scott S. Solis Email: scott.s.solis@wellsfargo.com

with a copy to:

Wells Fargo Bank, National Association, as agent Commercial Banking – Wholesale Loan Center MAC: N9300-085 600 S. 4th Street, Floor 08

Minneapolis, MN 55415-1526 Attention: David DeAngelis

Documents required to be delivered pursuant to the Loan Documents may be delivered by electronic communication and delivery, including, the Internet, e-mail or intranet websites to which the Administrative Agent and each Lender have access (including a commercial, third-party website such as www.sec.gov http://www.sec.gov or a website sponsored or hosted by the Administrative Agent or the Borrower) provided that the foregoing shall not apply to (i) notices to any Lender (or any Issuing Bank) pursuant to Article II and (ii) any Lender that has notified the Administrative Agent and the Borrower that it cannot or does not want to receive electronic communications. The Administrative Agent or the Borrower may, in its discretion, agree to accept notices and other communications to it hereunder by electronic delivery pursuant to procedures approved by it for all or particular notices or communications. Documents or notices delivered electronically shall be deemed to have been delivered twenty-four (24) hours after the date and time on which the Administrative Agent or the Borrower posts such documents or the documents become available on a commercial website and the Administrative Agent or Borrower notifies each Lender of said posting and provides a link thereto provided if such notice or other communication is not sent or posted during the normal business hours of the recipient, said posting date and time shall be deemed to have commenced as of 11:00 a.m. (Central Time) on the opening of business on the next business day for the recipient. Notwithstanding anything contained herein, in every instance the Borrower shall be required to provide paper copies of the certificate required by Section 8.2(iv) to the Administrative Agent and shall deliver paper copies of any documents to the Administrative Agent or to any Lender that requests such paper copies until a written request to cease delivering paper copies is given by the Administrative Agent or such Lender. Except for the certificates required by Section 8.2(iv), the Administrative Agent shall have no obligation to request the delivery of or to maintain paper copies of the documents delivered electronically, and in any event shall have no responsibility to monitor compliance by the Borrower with any such request for delivery. Each Lender shall be solely responsible for requesting delivery to it of paper copies and maintaining its paper or electronic documents.

Documents required to be delivered pursuant to <u>Article II</u> may be delivered electronically to a website provided for such purpose by the Administrative Agent pursuant to the procedures provided to the Borrower by the Administrative Agent.

15.20. <u>Change of Address</u>. Each party may change the address for service of notice upon it by a notice in writing to the other parties hereto.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, the undersigned have executed this Agreement as of the date first above written.

BORROWER: FIRST INDUSTRIAL, L.P.

By: FIRST INDUSTRIAL REALTY TRUST, INC., its General Partner

By: /s/ Scott A. Musil
Name: Scott A. Musil
Title: Chief Financial Officer

GENERAL PARTNER: FIRST INDUSTRIAL REALTY TRUST, INC.

By: /s/ Scott A. Musil
Name: Scott A. Musil
Title: Chief Financial Officer

LENDERS:

WELLS FARGO BANK, NATIONAL ASSOCIATION, individually as a Lender and as Administrative Agent $\,$

By: <u>/s/ Craig V. Koshkarian</u> Name: Craig V. Koshkarian

Title: Director

BANK OF AMERICA, N.A., individually as a Lender

By: /s/ Thomas W. Nowak

Name: Thomas W. Nowak Title: Vice President

PNC BANK, NATIONAL ASSOCIATION, individually as a Lender

By: /s/ Margaret K. Grady
Name: Margaret Kilner Grady
Title: Vice President

REGIONS BANK, individually as a Lender

By: <u>/s/ Susan Wolfe</u> Name: Susan Wolfe

Title: Assistant Vice President

U.S. BANK NATIONAL ASSOCIATION, individually as a Lender

By: /s/ Donald J. Pafford

Name: Donald J. Pafford
Title: Senior Vice President

JPMORGAN CHASE BANK, N.A., individually as a Lender

By: <u>/s/ Cody A. Canafax</u>
Name: Cody A. Canafax
Title: Vice President

CITIBANK, N.A., individually as a Lender

By: <u>/s/ David Bouton</u>
Name: David Bouton
Title: Managing Director

ROYAL BANK OF CANADA, individually as a Lender

By: /s/ William Behuniak

Name: William Behuniak
Title: Authorized Signatory

FIFTH THIRD BANK, NATIONAL ASSOCIATION, individually as a Lender

By: /s/ Michael Glandt

Name: Michael Glandt Title: Vice President

Accepted to and Agreed:

The undersigned is executing this signature page solely as a Departing Lender in its acceptance of the termination of its commitments and obligations under the Existing Credit Agreement as a "Lender" thereunder and not as a Lender party hereto. The undersigned hereby acknowledges that the Existing Credit Agreement shall be amended and restated in its entirety by this Agreement to which this signature page is attached and the undersigned shall not constitute a party thereto as a Lender other than for purposes of effectuating the amendment and restatement of the Existing Credit Agreement.

COMERICA BANK as a Departing Lender

By: <u>/s/ Casey L. Stevenson</u>
Name: Casey L. Stevenson
Title: Vice President

Accepted to and Agreed:

The undersigned is executing this signature page solely as a Departing Lender in its acceptance of the termination of its commitments and obligations under the Existing Credit Agreement as a "Lender" thereunder and not as a Lender party hereto. The undersigned hereby acknowledges that the Existing Credit Agreement shall be amended and restated in its entirety by this Agreement to which this signature page is attached and the undersigned shall not constitute a party thereto as a Lender other than for purposes of effectuating the amendment and restatement of the Existing Credit Agreement.

UBS AG, STAMFORD BRANCH as a Departing Lender

By: <u>/s/ Anthony N. Joseph</u> Name: Anthony N. Joseph Title: Associate Director

By: <u>/s/ Houssem Daly</u>
Name: Houssem Daly
Title: Director

EXHIBIT A

REVOLVING COMMITMENT AMOUNTS

Bank	Total Commitment
Wells Fargo Bank, National Association	\$100,000,000.00
Bank of America, N.A.	\$100,000,000.00
PNC Bank, National Association	\$100,000,000.00
Regions Bank	\$85,000,000.00
U.S. Bank National Association	\$85,000,000.00
JPMorgan Chase Bank, N.A.	\$75,000,000.00
Citibank, N.A.	\$75,000,000.00
Royal Bank of Canada	\$75,000,000.00
Fifth Third Bank	\$55,000,000.00
Total	\$750,000,000.00

EXHIBIT B-1

FORM OF NOTE

REVOLVING CREDIT NOTE

[1, 20

On or before the Maturity Date, as defined in that certain Fourth Amended and Restated Unsecured Revolving Credit Agreement
dated as of July 7, 2021 (the "Agreement") between FIRST INDUSTRIAL, L.P., a Delaware limited partnership ("Borrower"), First
Industrial Realty Trust, Inc., a Maryland corporation, Bank of America, N.A. and PNC Bank, National Association, each, individually and as
Co-Syndication Agents, Wells Fargo Bank, National Association, individually and as Administrative Agent for the Lenders (as such terms
are defined in the Agreement), and the other Lenders listed on the signature pages of the Agreement, Borrower promises to pay to the order
of [] (the "Lender"), or its successors and assigns, the aggregate unpaid principal amount of all Revolving Loans made by the
Lender to the Borrower pursuant to Section 2.1 of the Agreement, in immediately available funds at the office of the Administrative Agent in
Minneapolis, Minnesota, together with interest on the unpaid principal amount hereof at the rates and on the dates set forth in the Agreement.

The Lender shall, and is hereby authorized to record on the schedule attached hereto, or to otherwise record in accordance with its usual practice, the date and amount of each Revolving Borrowing and the date and amount of each principal payment hereunder; provided, however, that the failure of Lender to so record shall not affect the obligations of the Borrower hereunder or under the other Loan. Documents.

The Borrower shall pay this Promissory Note ("Note") in full on or before the Maturity Date in accordance with the terms of the Agreement.

This Note is issued pursuant to, and is entitled to the security under and benefits of, the Agreement and the other Loan Documents, to which Agreement and Loan Documents, as they may be amended from time to time, reference is hereby made for, inter alia, a statement of the terms and conditions under which this Note may be prepaid or its maturity date accelerated. Capitalized terms used herein and not otherwise defined herein are used with the meanings attributed to them in the Agreement.

If there is an Event of Default or Default under the Agreement or any other Loan Document and Lender exercises its remedies provided under the Agreement and/or any of the Loan Documents, then in addition to all amounts recoverable by the Lender under such documents, Lender shall be entitled to receive reasonable attorneys fees and expenses incurred by Lender in exercising such remedies.

Borrower and all endorsers severally waive presentment, protest and demand, notice of protest, demand and of dishonor and nonpayment of this Note (except as otherwise expressly provided for in the Agreement), and any and all lack of diligence or delays in collection or enforcement of this Note, and expressly agree that this Note, or any payment hereunder, may be extended from time to time, and expressly consent to the release of any party liable for the obligation secured by this Note, the release of any of the security of this Note, the acceptance of any other security therefor, or any other indulgence or forbearance whatsoever, all without notice to any party and without affecting the liability of the Borrower and any endorsers hereof.

This Note shall, be governed and construed under the internal laws of the State of Illinois.

BORROWER AND LENDER, BY ITS ACCEPTANCE HEREOF, EACH HEREBY WAIVE ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION OR PROCEEDING TO ENFORCE OR DEFEND ANY RIGHT UNDER THIS PROMISSORY NOTE OR ANY OTHER LOAN DOCUMENT OR RELATING THERETO OR ARISING FROM THE LENDING RELATIONSHIP WHICH IS THE SUBJECT OF THIS NOTE AND AGREE THAT ANY SUCH ACTION OR. PROCEEDING SHALL BE TRIED BEFORE A COURT AND NOT BEFORE A JURY.

[Signature on Following Page]

FIRST INDUSTRIAL, L.P.

By: First Industrial Realty Trust, Inc., its general partner

By: Name: Title:

PAYMENTS OF PRINCIPAL

Date Unpaid Principal Balance Notation Made by

EXHIBIT B-2

FORM OF BID RATE NOTE

FOR VALUE RECEIVED, the undersigned, FIRST INDUSTRIAL, L.P. (the "Borrower"), hereby promises to pay to the order of
(the "Lender") in care of Wells Fargo Bank National Association as Administrative Agent (the "Administrative

20

(the "Lender"), in care of Wells Fargo Bank, National Association, as Administrative Agent (the "Administrative Agent"), to Wells Fargo Bank, National Association at the office of the Administrative Agent in Minneapolis, Minnesota or at such other address as may be specified by the Administrative Agent to the Borrower, the aggregate unpaid principal amount of Bid Rate Loans made by the Lender to the Borrower under the Credit Agreement (as hereinafter defined), on the dates and in the principal amounts provided in the Credit Agreement, and to pay interest on the unpaid principal amount of each such Bid Rate Loan, at such office at the rates and on the dates provided in the Credit Agreement.

The date, amount, interest rate and maturity date of each Bid Rate Loan made by the Lender to the Borrower, and each payment made on account of the principal thereof, shall be recorded by the Lender on its books and, prior to any transfer of this Note, endorsed by the Lender on the schedule attached hereto or any continuation thereof, <u>provided</u> that the failure of the Lender to make any such recordation or endorsement shall not affect the obligations of the Borrower to make a payment when due of any amount owing under the Credit Agreement or hereunder in respect of the Bid Rate Loans made by the Lender.

This Note is one of the "Bid Rate Notes" referred to in the Fourth Amended and Restated Unsecured Revolving Credit Agreement dated as of July 7, 2021 (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement"), by and among the Borrower, the financial institutions party thereto and their assignees under Section 13.3. thereof, the Administrative Agent, and the other parties thereto, and evidences Bid Rate Loans made by the Lender thereunder. Terms used but not otherwise defined in this Note have the respective meanings assigned to them in the Credit Agreement.

The Credit Agreement provides for the acceleration of the maturity of this Note upon the occurrence of certain events and for prepayments of Bid Rate Loans upon the terms and conditions specified therein.

Except as permitted by Section 13.3. of the Credit Agreement, this Note may not be assigned by the Lender to any other Person.

This Note shall, be governed and construed under the internal laws of the State of Illinois.

BORROWER AND LENDER, BY ITS ACCEPTANCE HEREOF, EACH HEREBY WAIVE ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION OR PROCEEDING TO ENFORCE OR DEFEND ANY RIGHT UNDER THIS PROMISSORY NOTE OR ANY OTHER LOAN DOCUMENT OR RELATING THERETO OR ARISING FROM THE LENDING RELATIONSHIP WHICH IS THE SUBJECT OF THIS NOTE AND AGREE THAT ANY SUCH ACTION OR. PROCEEDING SHALL BE TRIED BEFORE A COURT AND NOT BEFORE A JURY.

[Signature on next page]

WITNESS WHEREOF, the undersigned has executed and delivered this Bid Rate Note under seal as of the date first written above	∕e.
FIRST INDUSTRIAL, L.P.	

By: First Industrial Realty Trust, Inc., its general partner

By: Name:

Title:

SCHEDULE OF BID RATE LOANS

This Note evidences Bid Rate Loans made under the within-described Credit Agreement to the Borrower, on the dates, in the principal amounts, bearing interest at the rates and maturing on the dates set forth below, subject to the payments and prepayments of principal set forth below:

	Principal		Maturity	Amount	Unpaid	
Date of	Amount of	Interest	Date of	Paid or	Principal	Notation
Loan	<u>Loan</u>	Rate	<u>Loan</u>	<u>Prepaid</u>	Amount	Made By

EXHIBIT C

FORM OF DISBURSEMENT INSTRUCTION AGREEMENT

Borrower: FIRST INDUSTRIAL, L.P.
Administrative Agent: Wells Fargo Bank, National Association
Loan: Loan number [] made pursuant to that certain "Fourth Amended and Restated Unsecured Revolving Credit Agreement" dated as of July 7, 2021 between Borrower, FIRST INDUSTRIAL REALTY TRUST, INC., a Maryland corporation, as General Partner, Administrative Agent, Wells Fargo Securities, LLC and Lenders, as amended from time to time
Effective Date: [], 2021
Check applicable box:
New – This is the first Disbursement Instruction Agreement submitted in connection with the Loan. Replace Previous Agreement – This is a replacement Disbursement Instruction Agreement. All prior instructions submitted in connection with this Loan are cancelled as of the Effective Date set forth above.

This Agreement must be signed by the Borrower and is used for the following purposes:

- (1) to designate an individual or individuals with authority to request disbursements of Loan proceeds, whether at the time of Loan closing/origination or thereafter;
- (2) to designate an individual or individuals with authority to request disbursements of funds from Restricted Accounts (as defined in the Terms and Conditions attached to this Agreement), if applicable; and
- (3) to provide Administrative Agent with specific instructions for wiring or transferring funds on Borrower's behalf.

Any of the disbursements, wires or transfers described above are referred to herein as a "Disbursement."

Specific dollar amounts for Disbursements must be provided to Administrative Agent at the time of the applicable Disbursement in the form of a signed closing statement, an email instruction or other written communication, or telephonic request pursuant to 2.5(b) of the Credit Agreement (each, a "**Disbursement Request**") from an applicable Authorized Representative (as defined in the Terms and Conditions attached to this Agreement).

A new Disbursement Instruction Agreement must be completed and signed by the Borrower if (i) all or any portion of a Disbursement is to be transferred to an account or an entity not described in this Agreement or (ii) Borrower wishes to add or remove any Authorized Representatives.

See the Additional Terms and Conditions attached hereto for additional information and for definitions of certain capitalized terms used in this Agreement.

Disbu	rsement of Loan Proceeds at Origination/Closing	
indivi		ed to accept one or more Disbursement Requests from any of the er") to disburse Loan proceeds on or about the date of the Loan with (each, a "Closing Disbursement"):
	Individual's Name	Title
1.		
2.		
3.		
etc.): DESC If the	CRIBE APPLICABLE RESTRICTIONS OR INDICATE "N/A"	ement Authorizers (dollar amount limits, wire/deposit destinations, ment Authorizer may submit a Disbursement Request for all
	TE FOLLOWING SECTION IF NO WIRE TRANSFERS AT C	ORIGINATION/CLOSING Subtractions by wire transfer must specify the amount
and ap		y such Disbursement Request must be listed below. Administrative ed directly to Administrative Agent by the Receiving Party or
	Names of Receiving Parties for the Closing Disbursement(s) (Receiving Party must be attached as the Closing Exhibit)	may include as many parties as needed; wire instructions for each
1.		
2.		
3.		
DELE:	ΓΕ FOLLOWING SECTION IF NO DEPOSITS INTO WFB A	CCOUNTS AT ORIGINATION/CLOSING
	<u>: Deposit:</u> Disbursement Requests for the Closing Disbursement y the amount and applicable account. Each account included in	(s) to be deposited into an account at Wells Fargo Bank, N.A. must any such Disbursement Request must be listed below.
Name	on Deposit Account:	

Wells Fargo Bank, N.A. Deposit Account Number:

Further Credit Information/Instructions:

Disbı	rrsements of Loan Proceeds Subsequent to Lo	an Closing/Origination
indivi	duals named below (each, a "Subsequent Disbu	e Agent is authorized to accept one or more Disbursement Requests from any of the arsement Authorizer") to disburse Loan proceeds after the date of the Loan nnection therewith (each, a "Subsequent Disbursement"):
	Individual's Name	Title
1.		
2.		
3.		
etc.): DESO If the	CRIBE APPLICABLE RESTRICTIONS OR INC	ubsequent Disbursement Authorizers (dollar amount limits, wire/deposit destinations, DICATE "N/A" sequent Disbursement Authorizer may submit a Disbursement Request for all
	TE FOLLOWING SECTION IF NO SUBSEQUI	
applic is aut attach	cable Receiving Party. Each Receiving Party includes to use the wire instructions that have been	Subsequent Disbursements to be made by wire transfer must specify the amount and uded in any such Disbursement Request must be listed below. Administrative Agent in provided directly to Administrative Agent by the Receiving Party or Borrower and wire instructions must be in the format specified on the Subsequent
	Names of Receiving Parties for Subsequent Dis Receiving Party must be attached as the Subseq	bursements (may include as many parties as needed; wire instructions for each quent Disbursement Exhibit)
1.		
2.		
3.		
DELE'	TE FOLLOWING SECTION IF NO SUBSEQUI	ENT DEPOSITS INTO WFB ACCOUNTS ANTICIPATED
		It Disbursements to be deposited into an account at Wells Fargo Bank, N.A. must unt included in any such Disbursement Request must be listed below.
Name	on Deposit Account:	

Wells Fargo Bank, N.A. Deposit Account Number:

Further Credit Information/Instructions:

Borrower acknowledges that all of the information is the Additional Terms and Conditions on the following	n this Agreement is correct and agrees to the terms and conditions set forth herein and in ng page.
FIRST INDUSTRIAL, L.P.	
By:	
Name:	
Title:	

Additional Terms and Conditions to the Disbursement Instruction Agreement

Definitions. The following capitalized terms shall have the meanings set forth below:

- "Authorized Representative" means any or all of the Closing Disbursement Authorizers, Subsequent Disbursement Authorizers and Restricted Account Disbursement Authorizers, as applicable.
- "Receiving Bank" means the financial institution where a Receiving Party maintains its account.
- "Receiving Party" means the ultimate recipient of funds pursuant to a Disbursement Request.
- "Restricted Account" means an account at Wells Fargo Bank, N.A. associated with the Loan to which Borrower's access is restricted.

Capitalized terms used in these Additional Terms and Conditions to Disbursement Instruction Agreement and not otherwise defined herein shall have the meanings given to such terms in the body of the Agreement.

Disbursement Requests. Except as expressly provided in the Credit Agreement, Administrative Agent must receive Disbursement Requests in writing. Disbursement Requests will only be accepted from the applicable Authorized Representatives designated in the Disbursement Instruction Agreement. Disbursement Requests will be processed subject to satisfactory completion of Administrative Agent's customer verification procedures. Administrative Agent is only responsible for making a good faith effort to execute each Disbursement Request and may use agents of its choice to execute Disbursement Requests. Funds disbursed pursuant to a Disbursement Request may be transmitted directly to the Receiving Bank, or indirectly to the Receiving Bank through another bank, government agency, or other third party that Administrative Agent considers to be reasonable. Administrative Agent will, in its sole discretion, determine the funds transfer system and the means by which each Disbursement will be made. Administrative Agent may delay or refuse to accept a Disbursement Request if the Disbursement would: (i) violate the terms of this Agreement; (ii) require use of a bank unacceptable to Administrative Agent or Lenders or prohibited by government authority; (iii) cause Administrative Agent or Lenders to violate any Federal Reserve or other regulatory risk control program or guideline; or (iv) otherwise cause Administrative Agent or Lenders to violate any Applicable Law or regulation.

Limitation of Liability. Administrative Agent, Issuing Banks, Swingline Lender and Lenders shall not be liable to Borrower or any other parties for: (i) errors, acts or failures to act of others, including other entities, banks, communications carriers or clearinghouses, through which Borrower's requested Disbursements may be made or information received or transmitted, and no such entity shall be deemed an agent of the Administrative Agent, Issuing Banks, Swingline Lender or any Lender; (ii) any loss, liability or delay caused by fires, earthquakes, wars, civil disturbances, power surges or failures, acts of government, labor disputes, failures in communications networks, legal constraints or other events beyond Administrative Agent's, any Issuing Bank's, Swingline Lender's or any Lender's control; or (iii) any special, consequential, indirect or punitive damages, whether or not (A) any claim for these damages is based on tort or contract or (B) Administrative Agent, any Issuing Bank, Swingline Lender or Borrower knew or should have known the likelihood of these damages in any situation. Neither Administrative Agent, any Issuing Bank, Swingline Lender nor any Lender makes any representations or warranties other than those expressly made in this Agreement. IN NO EVENT WILL ADMINISTRATIVE AGENT, ANY ISSUING BANK, SWINGLINE LENDER OR ANY LENDER BE LIABLE FOR DAMAGES ARISING DIRECTLY OR INDIRECTLY IF A DISBURSEMENT REQUEST IS EXECUTED BY ADMINISTRATIVE AGENT IN GOOD FAITH AN IN ACCORDANCE WITH THE TERMS OF THIS AGREEMENT.

Representative in or in accordance with this Agreement when executing a Disbursement Request until Administrative Agent has received a new Agreement signed by Borrower. Borrower agrees to be bound by any Disbursement Request: (i) authorized or transmitted by Borrower; or (ii) made in Borrower's name and accepted by Administrative Agent in good faith and in compliance with this Agreement, even if not properly authorized by Borrower. Administrative Agent may rely solely (i) on the account number of the Receiving Party, rather than the Receiving Party's name, and (ii) on the bank routing number of the Receiving Bank, rather than the Receiving Bank's name, in executing a Disbursement Request. Administrative Agent is not obligated or required in any way to take any actions to detect errors in information provided by Borrower or an Authorized Representative. If Administrative Agent takes any actions in an attempt to detect errors in the transmission or content of transfers or requests or takes any actions in an attempt to detect unauthorized Disbursement Requests, Borrower agrees that, no matter how many times Administrative Agent takes these actions, Administrative Agent will not in any situation be liable for failing to take or correctly perform these actions in the future, and such actions shall not become any part of the Disbursement procedures authorized herein, in the Loan Documents, or in any agreement between Administrative Agent and Borrower.

International Disbursements. A Disbursement Request expressed in US Dollars will be sent in US Dollars, even if the Receiving Party or Receiving Bank is located outside the United States. Administrative Agent will not execute Disbursement Requests expressed in foreign currency unless permitted by the Credit Agreement.

Errors. Borrower agrees to notify Administrative Agent of any errors in the Disbursement of any funds or of any unauthorized or improperly authorized Disbursement Requests within fourteen (14) days after Administrative Agent's confirmation to Borrower of such Disbursement.

Finality of Disbursement Requests. Disbursement Requests will be final and will not be subject to stop payment or recall; provided that Administrative Agent may, at Borrower's request, make an effort to effect a stop payment or recall but will incur no liability whatsoever for its failure or inability to do so.

CLOSING EXHIBIT WIRE INSTRUCTIONS

ADMINISTRATIVE AGENT TO ATTACH WIRE INSTRUCTIONS FROM RECEIVING PARTIES

All wire instructions must contain the following information:

Transfer/Deposit Funds to (Receiving Party Account Name)
Receiving Party Deposit Account Number
Receiving Bank Name, City and State
Receiving Bank Routing (ABA) Number
Further identifying information, if applicable (title escrow number, borrower name, loan number, etc.)

SUBSEQUENT DISBURSEMENT EXHIBIT WIRE INSTRUCTIONS

ADMINISTRATIVE AGENT TO ATTACH WIRE INSTRUCTIONS FROM RECEIVING PARTIES

All wire instructions must contain the following information:

Transfer/Deposit Funds to (Receiving Party Account Name)
Receiving Party Deposit Account Number
Receiving Bank Name, City and State
Receiving Bank Routing (ABA) Number
Further identifying information, if applicable (title escrow number, borrower name, loan number, etc.)

RESTRICTED ACCOUNT DISBURSEMENT EXHIBIT WIRE INSTRUCTIONS

ADMINISTRATIVE AGENT TO ATTACH WIRE INSTRUCTIONS FROM RECEIVING PARTIES

All wire instructions must contain the following information:

Transfer/Deposit Funds to (Receiving Party Account Name)
Receiving Party Deposit Account Number
Receiving Bank Name, City and State
Receiving Bank Routing (ABA) Number
Further identifying information, if applicable (title escrow number, borrower name, loan number, etc.)

EXHIBIT D

FORM OF GUARANTY

FOURTH AMENDED AND RESTATED GUARANTY

This Fourth Amended and Restated Guaranty (this "Guaranty") is made as of July 7, 2021, by First Industrial Realty Trust, Inc., a Maryland corporation ("Guarantor"), to and for the benefit of Wells Fargo Bank, National Association, a national banking association, individually ("Wells"), and as administrative agent for itself and the lenders listed on the signature pages of the Credit Agreement (as defined below) and their respective successors and assigns (collectively, "Lender").

RECITALS

- A. First Industrial, L.P., a Delaware limited partnership ("<u>Borrower</u>"), and Guarantor have requested that Lender make unsecured revolving credit loans available to Borrower in the aggregate principal amount of up to \$750,000,000 (subject to Borrower's right to increase such aggregate principal amount to \$1,000,000,000 in accordance with <u>Section 2.2</u> of the Credit Agreement (herein defined)) (the "<u>Facility</u>").
- B. Lender has agreed to make available the Facility to Borrower pursuant to the terms and conditions set forth in a Fourth Amended and Restated Unsecured Revolving Credit Agreement bearing even date herewith between Borrower, the Lenders and Guarantor (as amended, restated, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>"). All capitalized terms used herein and not otherwise defined shall have the meanings ascribed to such terms in the Credit Agreement.
- C. Guarantor is party to that certain Third Amended and Restated Guaranty, dated October 31, 2017 (as amended, restated, supplemented or otherwise modified from time to time, the "Existing Guaranty").
- D. The parties hereto wish to amend and restate the Existing Guaranty in its entirety and the Guarantor wishes to affirm its obligations under the Existing Guaranty.
- E. Guarantor is the sole general partner of Borrower and, therefore, Guarantor will derive financial benefit from the Facility evidenced by the Credit Agreement and the other Loan Documents. The execution and delivery of this Guaranty by Guarantor is a condition precedent to the performance by Lender of its obligations under the Credit Agreement.

AGREEMENTS

NOW, THEREFORE, Guarantor, in consideration of the matters described in the foregoing Recitals, which Recitals are incorporated herein and made a part hereof, and for other good and valuable consideration, hereby agrees as follows:

- 1. Guarantor affirms its obligations under, and the terms and conditions of, the Existing Guaranty and agrees that such obligations remain in full force and effect and are hereby ratified, reaffirmed and confirmed. Guarantor hereby absolutely, unconditionally, and irrevocably guarantees to Lender:
- (a) the full and prompt payment of the principal of and interest on the Loans when due, whether at stated maturity, upon acceleration or otherwise, and at all times thereafter, and the prompt

payment of all sums which may now be or may hereafter become due and owing under the Credit Agreement and the other Loan Documents;

- (b) the payment of all Enforcement Costs (as hereinafter defined in Paragraph 7 hereof);
- (c) the full, complete, and punctual observance, performance, and satisfaction of all of the obligations, duties, covenants, and agreements of Borrower under the Credit Agreement and the Loan Documents; and
- (d) All amounts due, debts, liabilities, and payment obligations described in subparagraphs (a) and (b) of this Paragraph 1 are referred to herein as the "Facility Indebtedness." All obligations described in subparagraph (c) of this Paragraph 1 are referred to herein as the "Obligations."
- 2. In the event of any default by Borrower in making payment of the Facility Indebtedness, or in performance of the Obligations, as aforesaid, in each case beyond the expiration of any applicable grace period, Guarantor agrees, on demand by Lender or any holder of any Note, to pay all the Facility Indebtedness and to perform all the Obligations as are or then or thereafter become due and owing or are to be performed under the terms of the Credit Agreement and the other Loan Documents, and to pay any reasonable expenses incurred by Lender in protecting, preserving, or defending its interest in the Property or in connection with the Facility or under any of the Loan Documents, including, without limitation, all reasonable attorneys' fees and costs. Lender shall have the right, at its option, either before, during or after pursuing any other right or remedy against Borrower or Guarantor, to perform any and all of the Obligations by or through any agent. contractor or subcontractor, or any of their agents, of its selection, all as Lender in its sole discretion deems proper, and Guarantor shall indemnify and hold Lender free and harmless from and against any and all loss, damage, cost, expense, injury, or liability Lender may suffer or incur in connection with the exercise of its rights under this Guaranty or the performance of the Obligations, except to the extent the same arises as determined by a court of competent jurisdiction by a final and non-appealable judgment to have resulted from the Gross Negligence or willful misconduct of Lender.

All of the remedies set forth herein and/or provided by any of the Loan Documents or law or equity shall be equally available to Lender, and the choice by Lender of one such alternative over another shall not be subject to question or challenge by Guarantor or any other person, nor shall any such choice be asserted as a defense, set-off, or failure to mitigate damages in any action, proceeding, or counteraction by Lender to recover or seeking any other remedy under this Guaranty, nor shall such choice preclude Lender from subsequently electing to exercise a different remedy. The parties have agreed to the alternative remedies hereinabove specified in part because they recognize that the choice of remedies in the event of a failure hereunder will necessarily be and should properly be a matter of business judgment, which the passage of time and events may or may not prove to have been the best choice to maximize recovery by Lender at the lowest cost to Borrower and/or Guarantor. It is the intention of the parties that such choice by Lender be given conclusive effect regardless of such subsequent developments.

3. Guarantor does hereby waive (i) notice of acceptance of this Guaranty by Lender and any and all notices and demands of every kind which may be required to be given by any statute, rule or law, (ii) any defense, right of set-off or other claim which Guarantor may have against the Borrower or which Guarantor or Borrower may have against Lender or the holders of any Note (other than defenses relating to payment of the Facility Indebtedness or the correctness of any allegation by Lender that Borrower was in default in the performance of the Obligations), (iii) presentment for payment, demand for payment (other than as provided for in Paragraph 2 above), notice of nonpayment (other than as provided for in

Paragraph 2 above) or dishonor, protest and notice of protest, diligence in collection and any and all formalities which otherwise might be legally required to charge Guarantor with liability, (iv) any failure by Lender to inform Guarantor of any facts Lender may now or hereafter know about Borrower, the Facility, or the transactions contemplated by the Credit Agreement, it being understood and agreed that Lender has no duty so to inform and that the Guarantor is fully responsible for being and remaining informed by the Borrower of all circumstances bearing on the existence or creation, or the risk of nonpayment of the Facility Indebtedness or the risk of nonperformance of the Obligations, and (v) any and all right to cause a marshalling of assets of the Borrower or any other action by any court or governmental body with respect thereto, or to cause Lender to proceed against any other security given to Lender in connection with the Facility Indebtedness or the Obligations. Credit may be granted or continued from time to time by Lender to Borrower without notice to or authorization from Guarantor, regardless of the financial or other condition of the Borrower at the time of any such grant or continuation. Lender shall have no obligation to disclose or discuss with Guarantor its assessment of the financial condition of Borrower. Guarantor acknowledges that no representations of any kind whatsoever have been made by Lender to Guarantor. No modification or waiver of any of the provisions of this Guaranty shall be binding upon Lender except as expressly set forth in a writing duly signed and delivered on behalf of Lender. Guarantor further agrees that any exculpatory language contained in the Credit Agreement and the Notes shall in no event apply to this Guaranty, and will not prevent Lender from proceeding against Guarantor to enforce this Guaranty.

- 4. Guarantor further agrees that Guarantor's liability as guarantor shall in no way be impaired by any renewals or extensions which may be made from time to time, with or without the knowledge or consent of Guarantor of the time for payment of interest or principal under the Credit Agreement and the other Loan Documents or by any forbearance or delay in collecting interest or principal under the Credit Agreement and the other Loan Documents, or by any waiver by Lender under the Credit Agreement or any other Loan Documents, or by Lender's failure or election not to pursue any other remedies it may have against Borrower, or by any change or modification in the Credit Agreement or any other Loan Documents, or by the acceptance by Lender of any additional security or any increase, substitution or change therein, or by the release by Lender of any security or any withdrawal thereof or decrease therein, or by the application of payments received from any source to the payment of any obligation other than the Facility Indebtedness, even though Lender might lawfully have elected to apply such payments to any part or all of the Facility Indebtedness, it being the intent hereof that Guarantor shall remain liable as principal for payment of the Facility Indebtedness and performance of the Obligations until all indebtedness has been paid in full and the other terms, covenants and conditions of the Credit Agreement and other Loan Documents and this Guaranty have been performed, notwithstanding any act or thing which might otherwise operate as a legal or equitable discharge of a surety. For the avoidance of doubt, it is understood and agreed that the Guarantor shall not be a "surety" for purposes of the Securities Act of Illinois (740 ILCS 155 et. seq.) and the Guarantor hereby waives any rights or benefits thereunder. Guarantor further understands and agrees that Lender may at any time enter into agreements with Borrower to amend and modify the Credit Agreement or other Loan Documents, or any thereof (including, without limitation in accordance with Borrower's right to increase the aggregate principal balance of the Loan pursuant to Section 2.2 of the Credit Agreement), and may waive or release any provision or provisions of the Credit Agreement and other Loan Documents or any thereof, and, with reference to such instruments, may make and enter into any such agreement or agreements as Lender and Borrower may deem proper and desirable, without in any manner impairing this Guaranty or any of Lender's rights hereunder or any of the Guarantor's obligations hereunder.
- 5. This is an absolute, unconditional, complete, present and continuing guaranty of payment and performance and not of collection. Guarantor agrees that this Guaranty may be enforced by Lender without the necessity at any time of resorting to or exhausting any other security or collateral given in

connection herewith or with the Credit Agreement or any of the other Loan Documents, or resorting to any other guaranties, and Guarantor hereby waives the right to require Lender to join Borrower in any action brought hereunder or to commence any action against or obtain any judgment against Borrower or to pursue any other remedy or enforce any other right. Guarantor further agrees that nothing contained herein or otherwise shall prevent Lender from pursuing concurrently or successively all rights and remedies available to it at law and/or in equity or under the Credit Agreement or any other Loan Documents, and the exercise of any of its rights or the completion of any of its remedies shall not constitute a discharge of any of Guarantor's obligations hereunder, it being the purpose and intent of the Guarantor that the obligations of such Guarantor hereunder shall be primary, absolute, independent and unconditional under any and all circumstances whatsoever. Neither Guarantor's obligations under this Guaranty nor any remedy for the enforcement thereof shall be impaired, modified, changed or released in any manner whatsoever by any impairment, modification, change, release or limitation of the liability of Borrower under the Credit Agreement or other Loan Documents or by reason of Borrower's bankruptcy or by reason of any creditor or bankruptcy proceeding instituted by or against Borrower. This Guaranty shall continue to be effective and be deemed to have continued in existence ore reinstated (as the case may be) if at any time payment of all or any part of any sum payable pursuant to the Credit Agreement or any other Loan Document is rescinded or otherwise required to be returned by the payee upon the insolvency, bankruptcy, or reorganization of the payor, all as though such payment to Lender had not been made, regardless of whether Lender contested the order requiring the return of such payment. The obligations of Guarantor pursuant to the preceding sentence shall survive any termination, cancellation, or relea

- 6. This Guaranty shall be assignable by Lender to any assignee of all or a portion of Lender's rights under the Loan Documents.
- 7. If: (i) this Guaranty, the Credit Agreement or any other Loan Document is placed in the hands of an attorney for collection or is collected through any legal proceeding; (ii) an attorney is retained to represent Lender in any bankruptcy, reorganization, receivership, or other proceedings affecting creditors' rights and involving a claim under this Guaranty, the Credit Agreement or any Loan Document; (iii) an attorney is retained to provide advice or other representation with respect to the Loan Documents in connection with an enforcement action or potential enforcement action; or (iv) an attorney is retained to represent Lender in any other legal proceedings whatsoever in connection with this Guaranty, the Credit Agreement, any of the Loan Documents, or any property subject thereto (other than any action or proceeding brought by any Lender or participant against the Administrative Agent (as defined in the Credit Agreement) alleging a breach by the Administrative Agent of its duties under the Loan Documents), then Guarantor shall pay to Lender upon demand all reasonable attorney's fees, costs and expenses, including, without limitation, court costs, filing fees, recording costs, expenses of foreclosure, title insurance premiums, survey costs, minutes of foreclosure, and all other costs and expenses incurred in connection therewith (all of which are referred to herein as "Enforcement Costs"), in addition to all other amounts due hereunder.
- 8. The parties hereto intend that each provision in this Guaranty comports with all applicable local, state and federal laws and judicial decisions. However, if any provision or provisions, or if any portion of any provision or provisions, in this Guaranty is found by a court of law to be in violation of any applicable local, state or federal ordinance, statute, law, administrative or judicial decision, or public policy, and if such court should declare such portion, provision or provisions of this Guaranty to be illegal, invalid, unlawful, void or unenforceable as written, then it is the intent of all parties hereto that such portion, provision or provisions shall be given force to the fullest possible extent that they are legal, valid and enforceable, that the remainder of this Guaranty shall be construed as if such illegal, invalid, unlawful, void or unenforceable portion, provision or provisions were not contained therein, and that the rights, obligations and interest of Lender or the holder of any Note under the remainder of this Guaranty

shall continue in full force and effect.

- 9. Any indebtedness of Borrower to Guarantor now or hereafter existing is hereby subordinated to the Facility Indebtedness. Guarantor agrees that until the entire Facility Indebtedness has been paid in full, (i) Guarantor will not seek, accept, or retain for Guarantor's own account any payment from Borrower on account of such subordinated debt, and (ii) any such payments to Guarantor on account of such subordinated debt shall be collected and received by Guarantor in trust for Lender and shall be paid over to Lender on account of the Facility Indebtedness without impairing or releasing the obligations of Guarantor hereunder.
- 10. Guarantor waives and releases any claim (within the meaning of 11 U.S.C. § 101) which Guarantor may have against Borrower arising from a payment made by Guarantor under this Guaranty and agrees not to assert or take advantage of any subrogation rights of Guarantor or Lender or any right of Guarantor or Lender to proceed against (i) Borrower for reimbursement, or (ii) any other guarantor or any collateral security or guaranty or right of offset held by Lender for the payment of the Facility Indebtedness and performance of the Obligations, nor shall Guarantor seek or be entitled to seek any contribution or reimbursement from Borrower or any other guarantor in respect of payments made by Guarantor hereunder. It is expressly understood that the waivers and agreements of Guarantor set forth above constitute additional and cumulative benefits given to Lender for its security and as an inducement for its extension of credit to Borrower. Nothing contained in this <u>Paragraph 10</u> is intended to prohibit Guarantor from making all distributions to its constituent shareholders which are required by law from time to time in order for Guarantor to maintain its status as a real estate investment trust in compliance with all applicable provisions of the Code (as defined in the Credit Agreement).
- 11. Any amounts received by Lender from any source on account of any indebtedness may be applied by Lender toward the payment of such indebtedness, and in such order of application, as Lender may from time to time elect.
- 12. The Guarantor hereby submits to personal jurisdiction in the State of Illinois for the enforcement of this Guaranty and waives any and all personal rights to object to such jurisdiction for the purposes litigation to enforce this Guaranty. Guarantor hereby consents to the jurisdiction of either the Circuit Court of Cook County, Illinois, or the United States District Court for the Northern District of Illinois, in any action, suit, or proceeding which Lender may at any time wish to file in connection with this Guaranty or any related matter. Guarantor hereby agrees that an action, suit, or proceeding to enforce this Guaranty may be brought in any state or federal court in the State of Illinois and hereby waives any objection which Guarantor may have to the laying of the venue of any such action, suit, or proceeding in any such court; provided, however, that the provisions of this Paragraph shall not be deemed to preclude Lender from filing any such action, suit, or proceeding in any other appropriate forum.
- 13. Anything contained herein to the contrary notwithstanding, the obligations of Guarantor hereunder at any time shall be limited to an aggregate amount equal to the largest amount that would not render its obligations hereunder subject to avoidance as a fraudulent transfer or conveyance under Section 548 of the United States Bankruptcy Code, as amended, or any comparable provisions of any similar federal or state law.
- 14. All notices and other communications provided to any party hereto under this Agreement or any other Loan Document shall be in writing or by email and addressed or delivered to such party at its address set forth below or at such other address as may be designated by such party in a notice to the other parties. Any notice, if mailed and properly addressed with postage prepaid, shall be deemed given when received; any notice, if transmitted by email, shall be deemed given when transmitted (provided)

that, if such notice is transmitted by email, such notice is also promptly mailed in accordance with this paragraph). Notice may be given as follows:

To the Guarantor: First Industrial Realty Trust, Inc. 1 N. Wacker Drive, Suite 4200 Chicago, Illinois 60606 Attention: Mr. Scott A. Musil Email: smusil@firstindustrial.com

With a copy to:

Barack Ferrazzano Kirschbaum & Nagelberg LLP 200 West Madison 39th Floor Chicago, Illinois 60606 Attention: Suzanne Bessette-Smith and Matthew A. Jackson Email: Suzanne.bessette-smith@bfkn.com; matt.jackson@bfkn.com

To the Lender:

c/o Wells Fargo Bank, National Association as agent 10 South Wacker Drive, 32nd Floor Chicago, IL 60606 Attention: Karen T. Skutt Email: berschkt@wellsfargo.com

or at such other address as the party to be served with notice may have furnished in writing to the party seeking or desiring to serve notice as a place for the service of notice.

- 14. This Guaranty shall be binding upon the heirs, executors, legal and personal representatives, successors and assigns of Guarantor and shall inure to the benefit of Lender's successors and assigns.
- 15. This Guaranty shall be construed and enforced under the internal law of the State of Illinois.
- 16. GUARANTOR AND LENDER, BY ITS ACCEPTANCE HEREOF, EACH HEREBY WAIVE ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION OR PROCEEDING TO ENFORCE OR DEFEND ANY RIGHT UNDER THIS GUARANTY OR ANY OTHER LOAN DOCUMENT OR RELATING THERETO OR ARISING FROM THE LENDING RELATIONSHIP WHICH IS THE SUBJECT OF THIS GUARANTY AND AGREE THAT ANY SUCH ACTION OR PROCEEDING SHALL BE TRIED BEFORE A COURT AND NOT BEFORE A JURY.

IN	WITNESS	WHEREOF	Guarantor has	delivered this	Guaranty in the	State of Illing	nis as of the	date first written	ahove

FIRST INDUSTRIAL REALTY TRUST, INC., a Maryland corporation
By:
Name:
Title:

STATE OF ILLINOIS)
) SS.
COUNTY OF COOK)
I, the undersigned, a Notary Public, in and for said County, in the State aforesaid, DO HEREBY CERTIFY, that of First Industrial Realty Trust, Inc., personally known to me to be the same person whose name is subscribed to the foregoing instrument, appeared before me this day in person and acknowledged that he signed and delivered the said instrument as his own free and voluntary act and as the free and voluntary act of said corporation, for the uses and purposes therein set forth.
GIVEN under my hand and Notarial Seal, this day of, 20
Notary Public

Acknowledged and Agreed by:
WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent
By: Name: Title:

EXHIBIT E

OPINION OF BORROWER'S COUNSEL

Attached

EXHIBIT F

OPINION OF GENERAL PARTNER'S COUNSEL

Attached

$\label{eq:exhibit} \mbox{EXHIBIT G}$ INTENTIONALLY OMITTED.

EXHIBIT H

FORM OF COMPLIANCE CERTIFICATE

To: The Administrative Agent and the Lenders who are parties to the Agreement described below

This Compliance Certificate is furnished pursuant to that certain Fourth Amended and Restated Unsecured Revolving Credit Agreement, dated as of July 7, 2021 (as amended, modified, renewed or extended from time to time, the "<u>Agreement</u>") among First Industrial, L.P. (the "<u>Borrower</u>"), First Industrial Realty Trust, Inc. (the "<u>General Partner</u>"), Wells Fargo Bank, National Association, individually and as Administrative Agent, and the Lenders named therein. Unless otherwise defined herein, capitalized terms used in this Compliance Certificate have the meanings ascribed thereto in the Agreement.

THE UNDERSIGNED HEREBY CERTIFIES THAT:

- 1. I am the duly elected [Chief Financial Officer] [Chief Accounting Officer] [Controller] of the [Borrower] [General Partner].
- 2. I have reviewed the terms of the Agreement and I have made, or have caused to be made under my supervision, a detailed review of the transactions and conditions of the General Partner, the Borrower and their respective Subsidiaries and Investment Affiliates during the accounting period covered by the financial statements attached (or most recently delivered to the Administrative Agent if none are attached).
- 3. The examinations described in paragraph 2 did not disclose, and I have no knowledge of, the existence of any condition or event which constitutes a Material Adverse Financial Change, Event of Default or Default during or at the end of the accounting period covered by the attached financial statements or as of the date of this Compliance Certificate, except as set forth below.
- 4. Schedule I (if attached) attached hereto sets forth financial data and computations and other information evidencing the General Partner's and the Borrower's compliance with certain covenants of the Agreement, all of which data, computations and information (or if no Schedule I is attached, the data, computations and information contained in the most recent Schedule I attached to a prior Compliance Certificate) are true, complete and correct in all material respects.
- 5. The financial statements and reports referred to in Section 8.2(i), 8.2(iii) or 8.2(viii), as the case may be, of the Agreement which are delivered concurrently with the delivery of this Compliance Certificate, if any, fairly present in all material respects the consolidated financial condition and operations of the General Partner, the Borrower and their respective Subsidiaries at such date and the consolidated results of their operations for the period then-ended, in accordance with GAAP applied consistently throughout such period and with prior periods and correctly state the amounts of Consolidated Total Indebtedness, Consolidated Secured Debt, Consolidated Senior Unsecured Debt and the Values of all Unencumbered Assets as determined pursuant to the Agreement.

Described below are the exceptions, if any, to paragraph 3 by listing, in detail, the nature of the condition or event, the period during which it has existed and the action which the Borrower has taken, is taking, or proposes to take with respect to each such condition or event:

The foregoing certifications, together with the computat	ions and information set forth in Sch	edule I hereto and the finan	cial statements
delivered with this Compliance Certificate in support hered	of, are made and delivered this	_ day of	, 20
	FIRST INDUSTRIAL, L.P.		
	By: FIRST INDUSTRIAL REALT	Y TRUST, INC.	
	General Partner		
	Ву:		
	Print Name:		
	Title:		

EXHIBIT I

FORM OF ASSIGNMENT AND ASSUMPTION AGREEMENT

THIS ASSIGNMENT AND ASSUMPTION AGREEMENT dated as of ______, 20__ (the "Agreement") by and among

	(the "Assignor"),		(the " <u>As</u>	ssignee"), FIRST I	NDUSTRIAL, L.P.	, a Delaware
limited partnership (the " <u>Administrative Agent</u> ").	" <u>Borrower</u> "), and WELLS					
of July 7, 2021 (as amend	nor is a Lender under that cert ed, restated, supplemented or stitutions party thereto and t	otherwise modified	d from time to	time, the "Credit A	Agreement"), by an	nd among the
9	nor desires to assign to the A ns and conditions set forth her		tion of the As	ssignor's Revolvin	g Commitment und	er the Credit
WHEREAS, the [B	orrower and the] Administrat	ive Agent consent[s] to such assig	anment on the terms	s and conditions set	forth herein.
	RE, for good and valuable coereto hereby agree as follows:		eipt and suffi	ciency of which h	iereby are acknowle	edged by the
Section 1. Assignm	<u>ient</u> .					
Assignor pursuant to Sect irrevocably sells, transfers Commitment") in and to the Agreement, such Assignor Revolving Commitments, i rights of the Assignor assorespect to the [Assigned Corespect to the Assigned Corespect to the Coredit Agreement assumes all obligations of signatory to the Credit Agreement.	e terms and conditions of this ion 2 of this Agreement, efficient and assigns to the Assigner e Assignor's Revolving Comes Note, and the other Loan Description of the Assigned Commitment and other rights mmitment, all as if the Assignation to the Assignor with respect to element having a Revolving Cobligation of the Assignor to	fective as ofee, without recours mitment, and all of occuments represent a principal amount mmitment all rights of the Assignor unnee were an originate gned Commitment. The Assigned Commitment equal to	e, a \$, 20_e, a \$, the other righting% of outstandings to receive inder the Credial Lender under the Assignee, mitment as if to the Assigned	(the "Assignme interest (such ts and obligations of in respect of the age Revolving Loans atterest on such amount and signatory to a subject to the term of the Assignee werll Commitment, while	ent Date") the Assich interest being the of the Assignor undurgeregate amount of equal to \$	ignor hereby he "Assigned her the Credit fall Lenders', all voting all fees with cuments with hent having a hereof, hereby her under and hinclude, but

and the obligation to indemnify the

Administrative Agent as provided in the Credit Agreement (the foregoing obligations, together with all other similar obligations more particularly set forth in the Credit Agreement and the other Loan Documents, shall be referred to hereinafter, collectively, as the "Assigned <u>Obligations</u>").

The assignment by the Assignor to the Assignee hereunder is without recourse to the Assignor. The Assignee makes and (b) confirms to the Administrative Agent, the Assignor, and the other Lenders all of the representations, warranties and covenants of a Lender under Article XII of the Credit Agreement. Not in limitation of the foregoing, the Assignee acknowledges and agrees that, except as set forth in Section 4. below, the Assignor is making no representations or warranties with respect to, and the Assignee hereby releases and discharges the Assignor for any responsibility or liability for: (i) the present or future solvency or financial condition of the Borrower, any other Loan Party or any other Subsidiary, (ii) any representations, warranties, statements or information made or furnished by the Borrower, any other Loan Party or any other Subsidiary in connection with the Credit Agreement or otherwise, (iii) the validity, efficacy, sufficiency, or enforceability of the Credit Agreement, any Loan Document or any other document or instrument executed in connection therewith, or the collectability of the Assigned Obligations, (iv) the perfection, priority or validity of any Lien with respect to any collateral at any time securing the Obligations or the Assigned Obligations under the Notes or the Credit Agreement and (v) the performance or failure to perform by the Borrower or any other Loan Party of any obligation under the Credit Agreement or any other Loan Document, Further, the Assignee acknowledges that it has, independently and without reliance upon the Administrative Agent, any other Lender or counsel to the Administrative Agent or any of their respective officers, directors, employees and agents and based on the financial statements supplied by the Borrower and such other documents and information as it has deemed appropriate, made its own credit analysis and decision to become a Lender under the Credit Agreement. The Assignee also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Credit Agreement or any Note or pursuant to any other obligation. The Administrative Agent shall have no duty or responsibility whatsoever, either initially or on a continuing basis, to provide the Assignee with any credit or other information with respect to the Borrower, any other Loan Party or any other Subsidiary or to notify the undersigned of any Default or Event of Default except as expressly provided in the Credit Agreement. The Assignee has not relied on the Administrative Agent as to any legal or factual matter in connection therewith or in connection with the transactions contemplated thereunder.

Section 2. <u>Payment by Assignee</u>. In consideration of the assignment made pursuant to Section 1. of this Agreement, the Assignee agrees to pay to the Assignor on the Assignment Date, an amount equal to \$______ representing the aggregate principal amount outstanding of the Revolving Loans owing to the Assignor under the Credit Agreement and the other Loan Documents being assigned hereby.

Section 3. <u>Payments by Assignor</u>. The Assignor agrees to pay to the Administrative Agent on the Assignment Date the administrative fee payable under Section 13.3 (c) of the Credit Agreement.

Section 4. Representations and Warranties of Assignor. The Assignor hereby represents and warrants to the Assignee that (a) as of the Assignment Date (i) the Assignor is a Lender under the Credit Agreement having a Revolving Commitment under the Credit Agreement immediately prior to the Assignment Date, equal to \$______ and that the Assignor is not in default of its obligations under the Credit Agreement; and (ii) the outstanding balance of Revolving Loans owing to the Assignor (without reduction by any assignments thereof which have not yet become effective) is \$______; and (b) it is the legal and beneficial owner of the Assigned Commitment which is free and clear of any adverse claim created by the Assignor.

Section 5. Representations, Warranties and Agreements of Assignee. The Assignee (a) represents and warrants that it is (i) legally authorized to enter into this Agreement; (ii) an "accredited investor" (as such term is used in Regulation D of the Securities Act) and (iii) an Eligible Assignee; (b) confirms that it has received a copy of the Credit Agreement, together with copies of the most recent financial statements delivered pursuant thereto and such other documents and information (including without limitation the Loan Documents) as it has deemed appropriate to make its own credit analysis and decision to enter into this Agreement; (c) appoints and authorizes the Administrative Agent to take such action as contractual representative on its behalf and to exercise such powers under the Loan Documents as are delegated to the Administrative Agent by the terms thereof together with such powers as are reasonably incidental thereto; (d) agrees that it will become a party to and shall be bound by the Credit Agreement and the other Loan Documents to which the other Lenders are a party on the Assignment Date and will perform in accordance therewith all of the obligations which are required to be performed by it as a Lender; and (e) is either (i) not organized under the laws of a jurisdiction outside the United States of America or (ii) has delivered to the Administrative Agent (with an additional copy for the Borrower) such items required under Section 4.5 of the Credit Agreement.

Section 6. Recording and Acknowledgment by the Administrative Agent. Following the execution of this Agreement, the Assignor will deliver to the Administrative Agent (a) a duly executed copy of this Agreement for acknowledgment and recording by the Administrative Agent and (b) the Assignor's Note. Upon such acknowledgment and recording, from and after the Assignment Date, the Administrative Agent shall make all payments in respect of the interest assigned hereby (including payments of principal, interest, fees and other amounts) to the Assignee. The Assignor and Assignee shall make all appropriate adjustments in payments under the Credit Agreement for periods prior to the Assignment Date directly between themselves.

below:	Section 7. <u>Addresses</u> . The Assignee specifies as its address for notices and its Lending Office for all Loans, the offices set forth
	Attention:
	Telephone No.:
	Telecopy No.:
be mad instruct	Section 8. <u>Payment Instructions</u> . All payments to be made to the Assignee under this Agreement by the Assignor, and all payments to le to the Assignee under the Credit Agreement, shall be made as provided in the Credit Agreement in accordance with the following tions:

Section 9. <u>Effectiveness of Assignment</u>. This Agreement, and the assignment and assumption contemplated herein, shall not be effective until (a) this Agreement is executed and delivered by each of the Assignor, the Assignee, the Administrative Agent and if required, the Borrower, and (b) the payment to the Assignor of the amounts owing by the Assignee pursuant to Section 2. hereof and (c) the payment to the Administrative Agent of the amounts owing by the Assignor pursuant to Section 3. hereof. Upon

recording and acknowledgment of this Agreement by the Administrative Agent, from and after the Assignment Date, (i) the Assignee shall be a party to the Credit Agreement and, to the extent provided in this Agreement, have the rights and obligations of a Lender thereunder and (ii) the Assignor shall, to the extent provided in this Agreement, relinquish its rights (except as otherwise provided in Section 13.3 of the Credit Agreement) and be released from its obligations under the Credit Agreement; provided, however, that if the Assignor does not assign its entire interest under the Loan Documents, it shall remain a Lender entitled to all of the benefits and subject to all of the obligations thereunder with respect to its Commitment.

Section 10. <u>Governing Law</u>. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF ILLINOIS APPLICABLE TO CONTRACTS EXECUTED, AND TO BE FULLY PERFORMED, IN SUCH STATE.

- Section 11. <u>Counterparts</u>. This Agreement may be executed in any number of counterparts each of which, when taken together, shall constitute one and the same agreement.
 - Section 12. <u>Headings</u>. Section headings have been inserted herein for convenience only and shall not be construed to be a part hereof.
- Section 13. <u>Amendments; Waivers</u>. This Agreement may not be amended, changed, waived or modified except by a writing executed by the Assignee and the Assignor.
- Section 14. <u>Entire Agreement</u>. This Agreement embodies the entire agreement between the Assignor and the Assignee with respect to the subject matter hereof and supersedes all other prior arrangements and understandings relating to the subject matter hereof.
- Section 15. <u>Binding Effect</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns.
- Section 16. <u>Definitions</u>. Terms not otherwise defined herein are used herein with the respective meanings given them in the Credit Agreement.

[Include this Section only if the Borrower's consent is required under Section 13.3 of the Credit Agreement]. Section 17. Agreements of the Borrower. The Borrower hereby agrees that the Assignee shall be a Lender under the Credit Agreement [having a Revolving Commitment equal to the Assigned Commitment]. The Borrower agrees that the Assignee shall have all of the rights and remedies of a Lender under the Credit Agreement and the other Loan Documents as if the Assignee were an original Lender under and signatory to the Credit Agreement, including, but not limited to, the right of a Lender to receive payments of principal and interest with respect to the Assigned Obligations, if any, and to the Revolving Loans made by the Lenders after the date hereof and to receive the fees payable to the Lenders as provided in the Credit Agreement. Further, the Assignee shall be entitled to the benefit of the indemnification provisions from the Borrower in favor of the Lenders as provided in the Credit Agreement and the other Loan Documents. The Borrower further agrees, upon the execution and delivery of this Agreement, to execute in favor of the Assignee a Note in an initial amount equal to the Assigned Commitment. Further, the Borrower agrees that, upon the execution and delivery of this Agreement, the Borrower shall owe the Assigned Obligations to the Assignee as if the Assignee were the Lender originally making such Loans and entering into such other obligations.

[Signatures on Following Page]

IN WITNESS WHEREOF, the particular written above.	es hereto have duly executed this Assignment and Assumption Agreement as of the date and year first
	ASSIGNOR:
	[NAME OF ASSIGNOR]
	By: Name: Title: Payment Instructions
	[Bank] [Address] ABA No.: Account No.: Account Name: Reference:
	ASSIGNEE:
	[NAME OF ASSIGNEE]
	By:

<u>Payment Instructions</u>

Title:

[Bank]
[Address]
ABA No.:
Account No.:
Account Name:
Reference:

[Signatures continued on Following Page]

	Agreed and Consented to as of the date first written above.
	[Include signature of the Borrower only if required under Section 13.3(c) of the Credit Agreement
	BORROWER:
	FIRST INDUSTRIAL, L.P.
	By: First Industrial Realty Trust, Inc., its general partner
	By: Name: Title:
Accepted as of the date first written abo	ve.
ADMINISTRATIVE AGENT:	
WELLS FARGO BANK, NATIONAL	ASSOCIATION, as Administrative Agent
By: Name: Title:	

EXHIBIT J

FORM OF NOTICE OF BORROWING

, 20

Wells Fargo Bank, National Association 600 S 4th St. Floor 09 Minneapolis, MN 55415-1526 Attention: David DeAngelis

Ladies and Gentlemen:

Reference is made to that certain Fourth Amended and Restated Unsecured Revolving Credit Agreement dated as of July 7, 2021 (as amended, restated, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>"), by and among FIRST INDUSTRIAL, L.P., a Delaware limited partnership (the "<u>Borrower</u>"), the financial institutions party thereto and their assignees under Section 13.3 thereof (the "<u>Lenders</u>"), Wells Fargo Bank, National Association, as Administrative Agent (the "<u>Administrative Agent</u>"), and the other parties thereto. Capitalized terms used herein, and not otherwise defined herein, have their respective meanings given them in the Credit Agreement.

1.	Pursuant to Section 2.7 of the Credit Agreement, the Borrower hereby requests that the Lenders make Revolving Loans to the Borrower in an aggregate amount equal to \$
2.	The Borrower requests that such Revolving Loans be made available to the Borrower on, 20
3.	The Borrower hereby requests that such Revolving Loans be of the following Type:
	 [Check one box only] □ Adjusted Base Rate Borrowing □ Eurocurrency Borrowing, with an initial Interest Period for a duration of:
	[Check one box only] one month two months

4. The location and number of Borrower's account to which proceeds of the Requested Revolving Loan are to be disbursed:

The Borrower hereby certifies to the Administrative Agent and the Lenders that as of the date hereof, as of the date of the making of the requested Revolving Loans, and after making such Revolving Loans, (a) no Default or Event of Default exists or would exist; and (b) the representations and warranties

three months six months other:

made or deemed made by the Borrower and each other Loan Party pursuant to Section 5.2(b) of the Credit Agreement, are and shall be true and correct with the same force and effect as if made on and as of such date except to the extent that such representations and warranties expressly relate solely to an earlier date (in which case such representations and warranties shall have been true and accurate on and as of such earlier date). In addition, the Borrower certifies to the Administrative Agent and the Lenders that all conditions to the making of the requested Revolving Loans contained in Section 5.2 of the Credit Agreement will have been satisfied at the time such Revolving Loans are made.

FIRST INDUSTRIAL, L.P.

By: FIRST INDUSTRIAL REALTY TRUST, INC.

General Partner

By: ____

Print Name: ____

Title: ____

EXHIBIT K

FORM OF NOTICE OF SWINGLINE BORROWING

, 20

Wells Fargo Bank, National Association 600 S 4th St. Floor 09 Minneapolis, MN 55415-1526 Attention: David DeAngelis

Ladies and Gentlemen:

Reference is made to the Fourth Amended and Restated Unsecured Revolving Credit Agreement dated as of July 7, 2021 (as amended, restated, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>"), by and among FIRST INDUSTRIAL, L.P., a Delaware limited partnership (the "<u>Borrower</u>"), the financial institutions party thereto and their assignees under Section 13.3 thereof (the "<u>Lenders</u>"), Wells Fargo Bank, National Association, as Administrative Agent (the "<u>Administrative Agent</u>"), and the other parties thereto. Capitalized terms used herein, and not otherwise defined herein, have their respective meanings given them in the Credit Agreement.

1.	Pursuant to Section 2.17 of the Credit Agreement, the Borrower hereby requests that the Swingline Lender make a Swingline Loan to the Borrower in an amount equal to \$
2.	The Borrower requests that such Swingline Loan be made available to the Borrower on, 20
3.	The Borrower requests that the proceeds of such Swingline Loan be made available to the Borrower by, 20
4.	The location and number of Borrower's account to which proceeds of such Swingline Loan are to be disbursed.

The Borrower hereby certifies to the Administrative Agent, the Swingline Lender and the Lenders that as of the date hereof, as of the date of the making of the requested Swingline Loan, and after making such Swingline Loan, (a) no Default or Event of Default exists or would exist; and (b) the representations and warranties made or deemed made by the Borrower and each other Loan Party pursuant to Section 5.2(b) of the Credit Agreement, are and shall be true and correct with the same force and effect as if made on and as of such date except to the extent that such representations and warranties expressly relate solely to an earlier date (in which case such representations and warranties shall have been true and accurate on and as of such earlier date). In addition, the Borrower certifies to the Administrative Agent and the Lenders that all conditions to the making of the requested Swingline Loan contained in Section 5.2 of the Credit Agreement will have been satisfied at the time such Swingline Loan is made.

[Continued on next page]

If notice of the requested	d borrowing of this	Swingline Loan w	vas previously	given by telephone,	this notice is to be	considered the written
confirmation of such teleph	none notice required	by Section 2.7(a) of the Credit	Agreement.		

FIRST INDUSTRIAL, L.P.
By: FIRST INDUSTRIAL REALTY TRUST, INC.
General Partner

By:	 	
Print Name:	 	
Title:		

EXHIBIT L

FORM OF DESIGNATION AGREEMENT

THIS	DESIGNATION	AGREEMENT	dated	as	of		, _	(1	the	"Agreem	nent")	by	and	among
	(th	e "Designating L	ender"),	,				(the	"Des	signated 1	Lender") and	Wells	Fargo
Bank, National Association, as Administrative Agent (the "Administrative Agent").														

WHEREAS, the Designating Lender is a Lender under that certain Fourth Amended and Restated Unsecured Revolving Credit Agreement dated as of July 7, 2021 (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement"), by and among FIRST INDUSTRIAL, L.P. (the "Borrower"), the financial institutions party thereto and their assignees under Section 13.3. thereof (the "Lenders"), Wells Fargo Bank, National Association, as Administrative Agent (the "Administrative Agent"), and the other parties thereto;

WHEREAS, pursuant to Section 13.5, the Designating Lender desires to designate the Designated Lender as its "Designated Lender" under and as defined in the Credit Agreement; and

WHEREAS, the Administrative Agent consents to such designation on the terms and conditions set forth herein.

NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which hereby are acknowledged by the parties hereto, the parties hereto hereby agree as follows:

Section 1. <u>Designation</u>. Subject to the terms and conditions of this Agreement, the Designating Lender hereby designates the Designated Lender, and the Designated Lender hereby accepts such designation, to have a right to make Bid Rate Loans on behalf of the Designating Lender pursuant to Section 2.18. of the Credit Agreement. Any assignment by the Designating Lender to the Designated Lender of rights to make a Bid Rate Loan shall only be effective at the time such Bid Rate Loan is funded by the Designated Lender. The Designated Lender, subject to the terms and conditions hereof, hereby agrees to make such accepted Bid Rate Loans and to perform such other obligations as may be required of it as a Designated Lender under the Credit Agreement.

Section 2. <u>Designating Lender Not Discharged</u>. Notwithstanding the designation of the Designated Lender hereunder, the Designating Lender shall be and remain obligated to the Borrower, the Administrative Agent and the Lenders for each and every of the obligations of the Designating Lender and its related Designated Lender with respect to the Credit Agreement and the other Loan Documents, including, without limitation, any indemnification obligations under Section 12.8. of the Credit Agreement and any sums otherwise payable to the Borrower by the Designated Lender.

Section 3. No Representations by Designating Lender. The Designating Lender makes no representation or warranty and, except as set forth in Section 8 below, assumes no responsibility pursuant to this Agreement with respect to (a) any statements, warranties or representations made in or in connection with any Loan Document or the execution, legality, validity, enforceability, genuineness, sufficiency or value of any Loan Document or any other instrument and document furnished pursuant thereto and (b) the financial condition of the Borrower, any other Loan Party or any other Subsidiary of the Borrower or the performance or observance by the Borrower or any other Loan Party of any of its

obligations under any Loan Document to which it is a party or any other instrument or document furnished pursuant thereto.

Section 4. Representations and Covenants of Designated Lender. The Designated Lender makes and confirms to the Administrative Agent, the Designating Lender, and the other Lenders all of the representations, warranties and covenants of a Lender under Article XII of the Credit Agreement. Not in limitation of the foregoing, the Designated Lender (a) represents and warrants that it (i) is legally authorized to enter into this Agreement; (ii) is an "accredited investor" (as such term is used in Regulation D of the Securities Act) and (iii) meets the requirements of a "Designated Lender" contained in the definition of such term contained in the Credit Agreement; (b) confirms that it has received a copy of the Credit Agreement, together with copies of the most recent financial statements delivered pursuant thereto and such other documents and information (including without limitation the Loan Documents) as it has deemed appropriate to make its own credit analysis and decision to enter into this Agreement; (c) confirms that it has, independently and without reliance upon the Administrative Agent, any other Lender or counsel to the Administrative Agent, or any of their respective officers, directors, employees, agents or counsel, and based on such financial statements and such other documents and information, made its own credit analysis and decision to become a Designated Lender under the Credit Agreement; (d) appoints and authorizes the Administrative Agent to take such action as contractual representative on its behalf and to exercise such powers under the Loan Documents as are delegated to the Administrative Agent by the terms thereof together with such powers as are reasonably incidental thereto; and (e) agrees that it will become a party to and shall be bound by the Credit Agreement, the other Loan Documents to which the other Lenders are a party on the Effective Date (as defined below) and will perform in accordance therewith all of the obligations which are required to be performed by it as a Designated Lender. The Designated Lender also acknowledges that it will, independently and without reliance upon the Administrative Agent, any other Lender or counsel to the Administrative Agent or any of their respective officers, directors, employees and agents, and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Credit Agreement or any Note or pursuant to any other obligation. The Designated Lender acknowledges and agrees that except as expressly required under the Credit Agreement, the Administrative Agent shall have no duty or responsibility whatsoever, either initially or on a continuing basis, to provide the Designated Lender with any credit or other information with respect to the Borrower, any other Loan Party or any other Subsidiary or to notify the Designated Lender of any Default or Event of Default.

Section 5. <u>Appointment of Designating Lender as AttorneyInFact</u>. The Designated Lender hereby appoints the Designating Lender as the Designated Lender's agent and attorneyinfact, and grants to the Designating Lender an irrevocable power of attorney, to receive any and all payments to be made for the benefit of the Designated Lender under the Credit Agreement, to deliver and receive all notices and other communications under the Credit Agreement and other Loan Documents and to exercise on the Designated Lender's behalf all rights to vote and to grant and make approvals, waivers, consents of amendments to or under the Credit Agreement or other Loan Documents. Any document executed by the Designating Lender on the Designated Lender's behalf in connection with the Credit Agreement or other Loan Documents shall be binding on the Designated Lender. The Borrower, each Administrative Agent and each of the Lenders may rely on and are beneficiaries of the preceding provisions.

Section 6. <u>Acceptance by the Administrative Agent</u>. Following the execution of this Agreement by the Designating Lender and the Designated Lender, the Designating Lender will (i) deliver to the Administrative Agent a duly executed original of this Agreement for acceptance by the Administrative Agent and (ii) pay to the Administrative Agent the fee, if any, payable under the applicable provisions of

the Credit Agreement whereupon this Agreement shall become effective as of the date of such acceptance or such other date as may be specified on the signature page hereof (the "Effective Date").

Section 7. Effect of Designation. Upon such acceptance and recording by the Administrative Agent, as of the Effective Date, the Designated Lender shall be a party to the Credit Agreement with a right to make Bid Rate Loans as a Lender pursuant to Section 2.18. of the Credit Agreement and the rights and obligations of a Lender related thereto; <u>provided, however</u>, that the Designated Lender shall not be required to make payments with respect to such obligations except to the extent of excess cash flow of the Designated Lender which is not otherwise required to repay obligations of the Designated Lender which are then due and payable. Notwithstanding the foregoing, the Designating Lender, as agent for the Designated Lender, shall be and remain obligated to the Borrower, the Administrative Agent and the Lenders for each and every of the obligations of the Designated Lender and the Designating Lender with respect to the Credit Agreement.

Section 8. <u>Indemnification of Designated Lender</u>. The Designating Lender unconditionally agrees to pay or reimburse the Designated Lender and save the Designated Lender harmless against all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever which may be imposed or asserted by any of the parties to the Loan Documents against the Designated Lender, in its capacity as such, in any way relating to or arising out of this Agreement or any other Loan Documents or any action taken or omitted by the Designated Lender hereunder or thereunder, provided that the Designating Lender shall not be liable for any portion of such liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements if the same results from the Designated Lender's gross negligence or willful misconduct.

- Section 9. Governing Law. This Agreement shall be construed and enforced under the internal law of the State of Illinois.
- Section 10. <u>Counterparts</u>. This Agreement may be executed in any number of counterparts each of which, when taken together, shall constitute one and the same agreement.
 - Section 11. <u>Headings</u>. Section headings have been inserted herein for convenience only and shall not be construed to be a part hereof.
- Section 12. <u>Amendments; Waivers</u>. This Agreement may not be amended, changed, waived or modified except by a writing executed by all parties hereto.
- Section 13. <u>Binding Effect</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns.
- Section 14. <u>Definitions</u>. Terms not otherwise defined herein are used herein with the respective meanings given them in the Credit Agreement.

[Signatures on Following Page]

IN WITNESS WHEREOF, the parties hereto have duly executed this Designation Agreement as of the date and year first written above.			
	EFFECTIVE DATE:		
	DESIGNATING LENDER:		
	[NAME OF DESIGNATING LENDER]		
	By: Name: Title:		
	DESIGNATED LENDER:		
	[NAME OF DESIGNATED LENDER]		
	By: Name: Title:		
Accepted as of the date first written abo	ve.		
ADMINISTRATIVE AGENT:			
WELLS FARGO BANK, NATIONAL	ASSOCIATION, as Administrative Agent		
By: Name: Title:			

EXHIBIT M

FORM OF BID RATE QUOTE REQUEST

	, 20
Afalla Faura Daula National Association	
Wells Fargo Bank, National Association	
500 S 4 th St.	
Floor 09	
Minneapolis, MN 55415-1526	
Attention: David DeAngelis	

Ladies and Gentlemen:

Reference is made to the Fourth Amended and Restated Unsecured Revolving Credit Agreement dated as of July 7, 2021 (as amended, restated, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>"), by and among FIRST INDUSTRIAL, L.P., a Delaware limited partnership (the "<u>Borrower</u>"), the financial institutions party thereto and their assignees under Section 13.3 thereof (the "<u>Lenders</u>"), Wells Fargo Bank, National Association, as Administrative Agent (the "<u>Administrative Agent</u>"), and the other parties thereto. Capitalized terms used herein, and not otherwise defined herein, have their respective meanings given them in the Credit Agreement.

1. The Borrower hereby requests Bid Rate Quotes for the following proposed Bid Rate Borrowings:

<u>Borrowing Date</u> , 20	<u>Amount</u> \$	<u>Type</u>	Interest Period days
2. The Borrower's Credit S&P Moody's	Rating, as applicable, as of the	e date hereof is:	
3. The proce	eeds of this Bid Ra	ate borrowing will be	used for the following purpose:
4. After giving effect to \$	the Bid Rate Borrowing red	quested herein, the total amou	nt of Bid Rate Loans outstanding shall be

The Borrower hereby certifies to the Administrative Agent and the Lenders that as of the date hereof, as of the date of the making of the requested Bid Rate Loans, and after making such Bid Rate Loans, (a) no Default or Event of Default exists or would exist; and (b) the representations and warranties made or deemed made by the Borrower and each other Loan Party pursuant to Section 5.2(b) of the Credit Agreement, are and shall be true and correct with the same force and effect as if made on and as of such date except to the extent that such representations and warranties expressly relate solely to an earlier date (in which case such representations and warranties shall have been true and accurate on and as of such earlier date). In addition, the Borrower certifies to the Administrative Agent and the Lenders that all

conditions to the making of the requested Bid Rate Loans con	itained in Section 5.2 of th	he Credit Agreement will h	ave been satisfied at the
time such Bid Rate Loans are made.			

FIRST INDUSTRIAL, L.P.

By: FIRST INDUSTRIAL REALTY TRUST, INC., its General Partner

Ву:	
Name:	
Title: _	

EXHIBIT N

FORM OF BID RATE QUOTE

1526			
emented or otherwise mod partnership (the "Borrower Fargo Bank, National Asso used herein, and not other rower's Bid Rate Quote Reg terms: er:	ified from time to time, r"), the financial institut ociation, as Administrativise defined herein, hav	the "Credit Agreement"), by a tions party thereto and their ass tive Agent (the "Administrative their respective meanings giv, 20, the undersigned here	and among FIRST INDUSTRIAL, signees under Section 13.3 thereof the Agent"), and the other parties them in the Credit Agreement. The eby makes the following Bid Rate
<u>Amount</u>	Туре	Interest Period	Bid Rate
\$ \$ \$		days days days	% % %
	emented or otherwise mod partnership (the "Borrowe Fargo Bank, National Assessused herein, and not other rower's Bid Rate Quote Reg terms: der:	de to the Fourth Amended and Restated Unsecuemented or otherwise modified from time to time, partnership (the "Borrower"), the financial instituted Fargo Bank, National Association, as Administrated used herein, and not otherwise defined herein, have rower's Bid Rate Quote Request dated	de to the Fourth Amended and Restated Unsecured Revolving Credit Agreememented or otherwise modified from time to time, the "Credit Agreement"), by a partnership (the "Borrower"), the financial institutions party thereto and their asserago Bank, National Association, as Administrative Agent (the "Administrative sused herein, and not otherwise defined herein, have their respective meanings give rower's Bid Rate Quote Request dated

The undersigned understands and agrees that the offer(s) set forth above, subject to satisfaction of the applicable conditions set forth in the Credit Agreement, irrevocably obligate[s] the undersigned to make the Bid Rate Loan(s) for which any offer(s) [is/are] accepted, in whole or in part.

By:	
Name:_	
Title:	

EXHIBIT O

FORM OF BID RATE QUOTE ACCEPTANCE

, 20

Type

Amount Accepted

\$_____

Wells Fargo Bank, National Association
600 S 4 th St.
Floor 09
Minneapolis, MN 55415-1526
Attention: David DeAngelis
Ladies and Gentlemen:
Reference is made to the Fourth Amended and Restated Unsecured Revolving Credit Agreement dated as of July 7, 2021 (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement"), by and among FIRST INDUSTRIAL, L.P., a Delaware limited partnership (the "Borrower"), the financial institutions party thereto and their assignees under Section 13.3 thereof (the "Lenders"), Wells Fargo Bank, National Association, as Administrative Agent (the "Administrative Agent"), and the other parties thereto. Capitalized terms used herein, and not otherwise defined herein, have their respective meanings given them in the Credit Agreement.
The Borrower hereby accepts the following offer(s) of Bid Rate Quotes to be made available to the Borrower on;

Quoting Lender

Quote Date

_____, 20___

_____, 20___ _____, 20___

time such Bid Rate Loans are made.

The Borrower hereby certifies to the Administrative Agent and the Lenders that as of the date hereof, as of the date of the making of the requested Bid Rate Loans, and after making such Bid Rate Loans, (a) no Default or Event of Default exists or would exist; and (b) the representations and warranties made or deemed made by the Borrower and each other Loan Party pursuant to Section 5.2(b) of the Credit Agreement, are and shall be true and correct with the same force and effect as if made on and as of such date except to the extent that such representations and warranties expressly relate solely to an earlier date (in which case such representations and warranties shall have been true and accurate on and as of such earlier date). In addition, the Borrower certifies to the Administrative Agent and the Lenders that all conditions to the making of the requested Bid Rate Loans contained in Section 5.2 of the Credit Agreement will have been satisfied at the

[Signature on next page]

FIRST INDUSTRIAL, L.P.

By: FIRST INDUSTRIAL REALTY TRUST, INC., its General Partner

By:		
Name:		
Title:		

EXHIBIT P

FORM OF PRICING CERTIFICATE

_____, 20___

Wells Fargo Bank, National Association 600 S 4th St. Floor 09 Minneapolis, MN 55415-1526 Attention: David DeAngelis

Ladies and Gentlemen:

Reference is made to that certain Fourth Amended and Restated Credit Agreement dated as of July 7, 2021 (as amended, restated, supplemented or otherwise modified from time to time, the "Loan Agreement"), by and among FIRST INDUSTRIAL, L.P., a Delaware limited partnership (the "Borrower"), the financial institutions party thereto and their assignees under Section 13.3 thereof (the "Lenders"), Wells Fargo Bank, National Association, as Administrative Agent (the "Administrative Agent"), and the other parties thereto. Capitalized terms used herein, and not otherwise defined herein, have their respective meanings given them in the Loan Agreement.

The undersigned hereby certifies solely in [his/her] capacity as [insert title of officer] of the Borrower and not in an individual capacity (and without personal liability) that:

- 1. I am the duly elected [insert title of officer] of the Borrower, and I am authorized to deliver this Certificate on behalf of the Borrower;
- 2. The calculation of the KPI Metric and the Sustainability Margin Adjustment, in each case, for the Reference Year ending ______, 20__ are set forth and attached as Annex A hereto, and are true and correct as of the date hereof and evidences the Borrower's qualification for a Sustainability Margin Adjustment with respect to Eurocurrency Borrowings and Adjusted Base Rate Borrowings equal to [____]% per annum. The LIBOR Applicable Margin and the Base Rate Applicable Margin will never be reduced by more than 0.01% pursuant to the Sustainability Margin Adjustment during any calendar year.
- 3. The public reporting of the Borrower's quarterly "Supplemental Information" report for the final quarterly period of such Reference Year can be found at the following website: [_______].

FIRST INDUSTRIAL, L.P.
By: FIRST INDUSTRIAL REALTY TRUST, INC.
General Partner
D .
By:
Print Name:
Title

Outstanding Facility Letters of Credit

Litigation (Borrower)

Environmental Compliance

Trade Names

Subsidiaries (Borrower)

Unencumbered Assets

Litigation (General Partner)

Subsidiaries (General Partner)

EXECUTION VERSION

AMENDED AND RESTATED UNSECURED TERM LOAN AGREEMENT DATED AS OF JULY 7, 2021

AMONG

FIRST INDUSTRIAL, L.P., AS BORROWER

FIRST INDUSTRIAL REALTY TRUST, INC.,

AS GENERAL PARTNER AND GUARANTOR

THE LENDERS

AND

WELLS FARGO BANK, NATIONAL ASSOCIATION
AS ADMINISTRATIVE AGENT

AND

PNC BANK, NATIONAL ASSOCIATION,
AS SYNDICATION AGENT

AND

FIFTH THIRD BANK, NATIONAL ASSOCIATION and REGIONS BANK,
AS CO-DOCUMENTATION AGENTS

WELLS FARGO SECURITIES, LLC,

PNC CAPITAL MARKETS LLC, FIFTH THIRD BANK, NATIONAL ASSOCIATION and REGIONS CAPITAL MARKETS

AS JOINT-LEAD ARRANGERS

AND

WELLS FARGO SECURITIES, LLC and PNC CAPITAL MARKETS LLC,
AS JOINT BOOK RUNNERS

		<u>Page</u>
ART	TICLE I. DEFINITIONS AND ACCOUNTING TERMS	<u>2</u>
1.1.	Definitions	<u>2</u>
1.2.	Financial Standards	<u>33</u>
1.3.	Classification of Loans and Borrowings	<u>33</u>
1.4.	General References	<u>33</u>
1.5.	Rates	<u>33</u>
1.6.	Divisions	<u>34</u>
1.7.	Sustainability Adjustments	<u>35</u>
1.8.	Amendment and Restatement of the Existing Term Loan Agreement	<u>36</u>
ART	TICLE II. THE FACILITY	<u>37</u>
2.1.	The Facility	<u>37</u>
2.2.	Incremental Loans	<u>37</u>
2.3.	Principal Payments	<u>38</u>
2.4.	Requests for Borrowings; Responsibility for Borrowings	<u>38</u>
2.5.	Evidence of Credit Extensions	<u>39</u>
2.6.	Loans and Borrowings	<u>39</u>
2.7.	Requests for Borrowings	<u>39</u>
2.8.	Interest Elections	<u>41</u>
2.9.	Applicable Margins and Fees	<u>42</u>
2.10.	. Funds Transfer Disbursements	<u>44</u>
2.11.	. Minimum Amount of Each Borrowing	<u>44</u>
2.12.	. Interest	<u>44</u>
2.13.	. Method of Payment	<u>44</u>
2.14.	. Default	<u>45</u>
2.15.	. Lending Offices	<u>45</u>
2.16.	. Non Receipt of Funds by Administrative Agent	<u>45</u>
2.17.	. [Reserved]	<u>46</u>
2.18.	. [Reserved]	<u>46</u>
2.19.	. [Reserved]	<u>46</u>
2.20.	. Pro Rata Treatment/Sharing of Payments	<u>46</u>
2.21.	. [Reserved]	<u>47</u>
2.22.	. Application of Moneys Received	<u>47</u>
2.23.	. [Reserved]	<u>47</u>
2.24.	. Voluntary Prepayments of Loans	<u>47</u>
ART	TCLE III. [RESERVED]	48
	TICLE IV. CHANGE IN CIRCUMSTANCES	<u>48</u>
4.1.	Yield Protection	<u>48</u>

(Continued)

<u>Page</u>

4.2.	Changes in Capital Adequacy Regulations	40
4.3.	[Reserved]	<u>49</u> 49
4.4.	Funding Indemnification	49
4.5.	Taxes	49
4.6.	Lender Statements; Survival of Indemnity	52
4. 7.	Replacement of Lenders under Certain Circumstances	52
4.8.	Changed Circumstances	52
4.0.	Changed Cheanistanices	<u> </u>
ARTI	ICLE V. CONDITIONS PRECEDENT	<u>56</u>
5.1.	Conditions Precedent to Closing	<u>56</u>
5.2.	Conditions Precedent to Subsequent Borrowings	<u>58</u>
л рті	ICLE VI. REPRESENTATIONS AND WARRANTIES	EQ
		<u>58</u>
6.1.	Existence Corporate (Portnership Portnership Portners	<u>58</u>
6.2.	Corporate/Partnership Powers Power of Officers	<u>59</u>
6.3.		<u>59</u>
6.4.	Government and Other Approvals	<u>59</u>
6.5.	Solvency Countries a With Language	<u>59</u>
6.6.	Compliance With Laws	60
6.7.	Enforceability of Agreement	60
6.8.	Title to Property	60
6.9.	Litigation	60
6.10.		<u>60</u>
6.11.		60
6.12.		<u>61</u>
6.13.	-	<u>61</u>
6.14.	9	<u>61</u>
6.15.		<u>61</u>
6.16.		<u>61</u>
6.17.	ERISA	<u>61</u>
6.18.		<u>61</u>
6.19.	Environmental Matters	<u>61</u>
6.20.	Insurance	63
6.21.		<u>63</u>
6.22.	No Violation of Usury Laws	<u>63</u>
6.23.	Not a Foreign Person	63
6.24.	No Trade Name	63
6.25.	Subsidiaries	63
6.26.	Unencumbered Assets	<u>63</u>
6.27.	Anti-Corruption Laws and Sanctions	<u>65</u>
6 28	Affected Financial Institutions	66

(Continued)

<u>Page</u>

ART	TCLE VII. ADDITIONAL REPRESENTATIONS AND WARRANTIES	<u>66</u>
7.1.	Existence	<u>66</u>
7.2.	Corporate Powers	<u>66</u>
7.3.	Power of Officers	<u>66</u>
7.4.	Government and Other Approvals	<u>66</u>
7.5.	Compliance With Laws	<u>66</u>
7.6.	Enforceability of Agreement	<u>66</u>
7.7.	Liens; Consents	<u>67</u>
7.8.	Litigation	<u>67</u>
7.9.	Events of Default	<u>67</u>
7.10.	. Investment Company Act of 1940	<u>67</u>
7.11.	. Public Utility Holding Company Act	<u>67</u>
7.12.	. No Material Adverse Financial Change	<u>67</u>
7.13.	. Financial Information	<u>67</u>
7.14.	. Factual Information	<u>67</u>
7.15.	. ERISA	<u>88</u>
7.16.	. Taxes	<u>88</u>
7.17.	. No Brokers	<u>68</u>
7.18.	. Subsidiaries	<u>68</u>
7.19.	. Status	<u>68</u>
7.20.	. Anti-Corruption Laws and Sanctions	<u>68</u>
ART	TICLE VIII. AFFIRMATIVE COVENANTS	<u>69</u>
8.1.	Notices	<u>69</u>
8.2.	Financial Statements, Reports, Etc.	<u>70</u>
8.3.	Existence and Conduct of Operations	<u>72</u>
8.4.	Maintenance of Properties	<u>73</u>
8.5.	Insurance	<u>73</u>
8.6.	Payment of Obligations	<u>73</u>
8.7.	Compliance with Laws	<u>73</u>
8.8.	Adequate Books	73
8.9.	ERISA	<u>74</u>
8.10.	. Maintenance of Status	<u>7</u> 4
8.11.	. Use of Proceeds	<u>7</u> 4
8.12.	. Distributions	<u>74</u>
8.13.	. Guarantors	<u>74</u>
ART	TICLE IX. NEGATIVE COVENANTS	<u>75</u>
9.1.	Change in Business	<u>75</u>
9.2.	Change of Management of Properties	<u>75</u>
9.3.	Change of Borrower Ownership	<u>75</u>
9.4.	Use of Proceeds	<u>75</u>
9.5.	Liens	76

(Continued)

<u>Page</u>

9.6.	Regulation U	<u>76</u>
	Indebtedness and Cash Flow Covenants	77
	Change of Control; Mergers and Dispositions	
	Negative Pledge	<u>78</u>
ARTI	CLE X. DEFAULTS	<u>78</u>
10.1.	Nonpayment of Principal	<u>78</u>
10.2.	Certain Covenants	<u>79</u>
10.3.	Nonpayment of Interest and Other Obligations	<u>79</u>
10.4.	Cross Default	<u>79</u>
10.5.	Loan Documents	<u>79</u>
10.6.	Representation or Warranty	<u>79</u>
10.7.	Covenants, Agreements and Other Conditions	<u>79</u>
10.8.	No Longer General Partner	<u>79</u>
10.9.	Material Adverse Financial Change	<u>79</u>
10.10.	. Bankruptcy	80
10.11.	. Legal Proceedings	<u>80</u>
10.12.	. ERISA	<u>81</u>
10.13.	-	<u>81</u>
10.14.	. Failure to Satisfy Judgments	<u>81</u>
10.15.	. Environmental Remediation	<u>81</u>
ARTI	CLE XI. ACCELERATION, WAIVERS, AMENDMENTS AND REMEDIES	<u>81</u>
11.1.	Acceleration	<u>81</u>
11.2.	Preservation of Rights; Amendments	<u>82</u>
ARTI	CLE XII. THE ADMINISTRATIVE AGENT	<u>82</u>
12.1.	Appointment	<u>82</u>
12.2.	Powers	<u>82</u>
12.3.	General Immunity	<u>82</u>
12.4.	No Responsibility for Loans, Recitals, etc.	<u>83</u>
12.5.	Action on Instructions of Lenders	<u>83</u>
12.6.	Employment of Administrative Agents and Counsel	<u>83</u>
12.7.	Reliance on Documents; Counsel	<u>83</u>
12.8.	Administrative Agent's Reimbursement and Indemnification	<u>83</u>
12.9.	Rights as a Lender	<u>84</u>
12.10.		<u>84</u>
12.11.		84
12.12.	5	<u>85</u>
12.13.		<u>85</u>
12.14.	•	<u>85</u>
12.15.	-	<u>86</u>
12.16.	. Defaulting Lenders	<u>86</u>

(Continued)

		<u>Page</u>
12.17.	. Delegation to Affiliates	<u>87</u>
12.17.		<u>87</u>
12.19.		<u>87</u>
12,120		<u>57.</u>
ARTIO	CLE XIII. BENEFIT OF AGREEMENT; ASSIGNMENTS; PARTICIPATIONS	<u>90</u>
13.1.	Successors and Assigns	<u>90</u>
13.2.	Participations	<u>90</u>
13.3.	Assignments; Consents	<u>92</u>
13.4.	Register	<u>93</u>
13.5.	[Reserved].	<u>94</u>
13.6	Confidentiality	<u>94</u>
13.7	Tax Treatment	<u>95</u>
ARTIC	CLE XIV. GENERAL PROVISIONS	<u>95</u>
14.1.	Survival of Representations	<u>95</u>
14.2.	Governmental Regulation	<u>95</u>
14.3.	Taxes	<u>95</u>
14.4.	Headings	<u>95</u>
14.5.	No Third Party Beneficiaries	<u>95</u>
14.6.	Expenses; Indemnification	<u>95</u>
14.7.	Severability of Provisions	<u>96</u>
14.8.	Nonliability of the Lenders	<u>96</u>
14.9.	Choice of Law	<u>96</u>
14.10.		<u>97</u>
14.11.	·	<u>97</u>
14.12.	5	<u>97</u>
14.13.	-	<u>97</u>
14.14.	. Dealings with the Borrower	<u>98</u>
14.15.		<u>99</u>
14.16.		<u>99</u>
14.17.		<u>100</u>
14.18. Institu		<u>101</u>
14.19.	. Acknowledgement Regarding Any Supported QFCs	<u>101</u>
14.20.	. Certain ERISA Matters	<u>102</u>
ARTIO	CLE XV. NOTICES	<u>104</u>
15.1.	Giving Notice; Electronic Delivery	<u>104</u>

<u>106</u>

15.2. Change of Address

(Continued)

EXHIBITS

- A Term Loan Commitment Amounts
- B Form of Note
- C Form of Disbursement Instruction Agreement
- D Form of Guaranty
- E Opinion of Borrower's Counsel
- F Opinion of General Partner's Counsel
- G [Reserved]
- H Form of Compliance Certificate
- I Form of Assignment Agreement
- J Form of Notice of Borrowing
- K Form of Pricing Certificate

SCHEDULES

- 6.9 Litigation (Borrower)
- 6.19 Environmental Compliance
- 6.24 Trade Names
- 6.25 Subsidiaries (Borrower)
- 6.26 Unencumbered Assets
- 7.8 Litigation (General Partner)
- 7.18 Subsidiaries (General Partner)

AMENDED AND RESTATED UNSECURED TERM LOAN AGREEMENT

THIS AMENDED AND RESTATED UNSECURED TERM LOAN AGREEMENT is entered into as of July 7, 2021 by and among the following:

FIRST INDUSTRIAL, L.P., a Delaware limited partnership having its principal place of business at 1 N. Wacker Drive, Suite 4200, Chicago, Illinois 60606, the sole general partner of which is First Industrial Realty Trust, Inc., a Maryland corporation;

FIRST INDUSTRIAL REALTY TRUST, INC., a Maryland corporation that is qualified as a real estate investment trust whose principal place of business is 1 N. Wacker Drive, Suite 4200, Chicago, Illinois 60606 ("General Partner");

WELLS FARGO BANK, NATIONAL ASSOCIATION ("<u>Wells</u>"), a national bank organized under the laws of the United States of America having an office at 10 South Wacker Drive, 32nd Floor, Chicago, IL 60606, as Administrative Agent ("<u>Administrative Agent</u>") for the Lenders (as defined below); and

Those Lenders identified on the signature pages hereto.

RECITALS

- A. The Borrower is primarily engaged in the business of acquiring, developing, owning and operating bulk warehouse and light industrial properties.
- B. The Borrower, the General Partner, the lenders party thereto and Wells, as administrative agent are currently party to the Unsecured Term Loan Agreement, dated as of July 15, 2020 (as amended, modified or otherwise supplemented prior to the date hereof, the "Existing Term Loan Agreement").
- C. The Borrower, the Lenders and the Administrative Agent have agreed to enter into this Agreement in order to (i) amend and restate the Existing Term Loan Agreement in its entirety, (ii) re-evidence the Obligations under, and as defined in, the Existing Term Loan Agreement, which shall be repayable in accordance with the terms of this Agreement, and (iii) set forth the terms and conditions under which the Lenders will, on the Closing Date, make term loans to or for the benefit of the Borrower.
- D. The parties hereto intend that this Agreement not constitute a novation of the obligations and liabilities of the parties under the Existing Term Loan Agreement or be deemed to evidence or constitute full repayment of such obligations and liabilities, but that this Agreement amend and restate in its entirety the Existing Term Loan Agreement and re-evidence the obligations and liabilities of the Borrower outstanding thereunder, which shall be payable in accordance with the terms hereof.
- E. Each of the Borrower and the General Partner confirms that all obligations under the applicable "Loan Documents" (as referred to and defined in the Existing Term Loan Agreement) shall continue in full force and effect as modified or restated by the Loan Documents (as referred to and defined herein) and that, from and after the Agreement Execution Date, all references to the

- "Credit Agreement" contained in any such existing "Loan Documents" shall be deemed to refer to this Agreement.
- F. The General Partner is fully liable for the obligations of the Borrower hereunder by virtue of its status as the sole general partner of the Borrower and as guarantor under the Guaranty.
- NOW, THEREFORE, in consideration of the mutual covenants and agreements herein contained, the parties hereto agree as follows:

ARTICLE I.

DEFINITIONS AND ACCOUNTING TERMS

- 1.1. <u>Definitions</u>. As used in this Agreement, the following terms have the meanings set forth below:
- "Adjusted Base Rate" means a floating interest rate equal to the sum of (i) the LIBOR Market Index Rate, plus (ii) the Base Rate Applicable Margin in effect from time to time.
 - "Adjusted Base Rate Borrowing" means a Borrowing that bears interest at the Adjusted Base Rate.
- "<u>Adjusted EBITDA</u>" means for any Person the sum of EBITDA for such Person and such Person's reported corporate overhead for itself and its Subsidiaries; <u>provided</u> that "Adjusted EBITDA" shall have deducted overhead related to specific properties.
- "Adjusted LIBOR Rate" means, with respect to a Eurocurrency Borrowing for the relevant Interest Period, the sum of (i) the Base LIBOR Rate applicable to such Interest Period, plus, (ii) the LIBOR Applicable Margin in effect from time to time during such Interest Period.
- "<u>Administrative Agent</u>" means Wells, in its capacity as contractual representative of the Lenders pursuant to <u>Article XII</u>, and not in its individual capacity as a Lender, and any successor Administrative Agent appointed pursuant to <u>Article XII</u>.
- "<u>Administrative Office</u>" means the Administrative Agent's office designated on its signature page to this Agreement or such other office as may be designated by the Administrative Agent by written notice to the Borrower and the Lenders.
- "<u>Administrative Questionnaire</u>" means the Administrative Questionnaire completed by each Lender and delivered to the Administrative Agent in a form supplied by the Administrative Agent to the Lenders from time to time.
 - "Affected Financial Institution" means (a) any EEA Financial Institution or (b) any UK Financial Institution.
- "Affiliate" means any Person directly or indirectly controlling, controlled by or under direct or indirect common control with any other Person. A Person shall be deemed to control another Person if the controlling Person owns ten percent (10%) or more of any class of voting securities of

the controlled Person or possesses, directly or indirectly, the power to direct or cause the direction of the management or policies of the controlled Person, whether through ownership of stock, by contract or otherwise; <u>provided</u> that in no event shall the Administrative Agent or any Lender be deemed to be an Affiliate of the Borrower.

- "Aggregate Term Loan Commitment" means, as of any date, the total of all Term Loan Commitments, which as of the Agreement Execution Date is \$200,000,000.
- "Agreement" means this Amended and Restated Unsecured Term Loan Agreement and all amendments, modifications and supplements hereto.
 - "Agreement Execution Date" shall mean July 7, 2021, the date on which all of the parties hereto have executed this Agreement.
- "<u>Anti-Corruption Laws</u>" means all Applicable Laws of any jurisdiction applicable to the Borrower or its Subsidiaries from time to time concerning or relating to bribery, corruption or money-laundering, including without limitation, the Foreign Corrupt Practices Act of 1977.
 - "Applicable Cap Rate" means 5.75 %.
- "Applicable Law" means all international, foreign, federal, state and local statutes, treaties, rules, guidelines, regulations, ordinances, codes, directions, executive orders, and administrative or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case whether or not having the force of law.
- "Applicable Margin" means the applicable margins set forth in the table in Section 2.9 used in calculating the interest rate applicable to the various types of Borrowings, which shall vary from time to time in accordance with the long term, senior unsecured debt ratings of the Borrower and the General Partner in the manner set forth in Section 2.9. It is hereby understood and agreed that the Applicable Margin shall be adjusted from time to time based upon the Sustainability Margin Adjustment (to be calculated and applied as set forth in Section 1.7); provided that in no event shall the Applicable Margin be less than zero.
- "<u>Approved Fund</u>" means any Fund that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.
- "Arranger" means Wells Fargo Securities, LLC, PNC Capital Markets LLC, Fifth Third Bank, National Association and Regions Capital Markets and their successors in their capacity as Joint Lead Arrangers.
- "Asset Sale" means any sale or other disposition by the Consolidated Operating Partnership of any Property or other assets (excluding any proceeds resulting from the casualty or condemnation of such Property or other assets, to the extent such proceeds are used to rebuild such Properties or assets within 365 days of receipt of such proceeds) that yields gross proceeds equal to the aggregate of (i) all cash proceeds, plus (ii) the initial principal amount of any noncash proceeds consisting of

notes or other debt securities, plus (iii) the fair market value of other non-cash proceeds, if and to the extent that the aggregate of (i), (ii) and (iii) exceeds \$500,000.

"Assets Acquired Not In Service" means, as of any date of determination, any Project which has been acquired and owned for less than 12 months but has not yet been leased to 90%.

"Assets Under Development" means, as of any date of determination, any Project which is under construction and then treated as an asset under development under GAAP.

"<u>Assignment and Assumption Agreement</u>" means an Assignment and Assumption Agreement among a Lender, an Eligible Assignee and the Administrative Agent, substantially in the form of <u>Exhibit I</u>.

"Available Tenor" means, as of any date of determination and with respect to the then-current Benchmark, as applicable, (a) if the then-current Benchmark is a term rate, any tenor for such Benchmark or (b) otherwise, any payment period for interest calculated with reference to such Benchmark, as applicable, that is or may be used for determining the length of an Interest Period pursuant to this Agreement as of such date and not including, for the avoidance of doubt, any tenor for such Benchmark that is then-removed from the definition of "Interest Period" pursuant to Section 4.8(c)(iv).

"Bail-In Action" means the exercise of any Write-Down and Conversion Powers by the applicable Resolution Authority in respect of any liability of an Affected Financial Institution.

"Bail-In Legislation" means (a) with respect to any EEA Member Country implementing Article 55 of Directive 2014/59/EU of the European Parliament and of the Council of the European Union, the implementing law, regulation, rule or requirement for such EEA Member Country from time to time which is described in the EU Bail-In Legislation Schedule and (b) with respect to the United Kingdom, Part I of the United Kingdom Banking Act 2009 (as amended from time to time) and any other law, regulation or rule applicable in the United Kingdom relating to the resolution of unsound or failing banks, investment firms or other financial institutions or their affiliates (other than through liquidation, administration or other insolvency proceedings).

"Base LIBOR Rate" means, subject to the implementation of a Benchmark Replacement in accordance with Section 4.8, with respect to any Eurocurrency Borrowing for any Interest Period, the rate of interest obtained by dividing (i) the rate of interest per annum determined on the basis of the rate as set by the ICE Benchmark Administration ("ICE") (or the successor thereto if ICE is no longer making such rate available) for deposits in Dollars for a period equal to the applicable Interest Period which appears on Reuters Screen LIBOR01 Page (or any applicable successor page), at approximately 11:00 a.m. (London time) two (2) Business Days prior to the first day of the applicable Interest Period by (ii) a percentage equal to 1 minus the Eurodollar Reserve Percentage. If, for any reason, the rate referred to in the preceding clause (i) does not appear on Reuters Screen LIBOR01 Page (or any applicable successor page), then the rate to be used for such clause (i) shall be determined by the Administrative Agent to be the arithmetic average of the rate per annum at which deposits in Dollars would be offered by first class banks in the London interbank market to the Administrative Agent at approximately 11:00 a.m. (London time) two (2) Business Days prior to the first day of the applicable Interest Period for a period equal to such Interest Period. Any change in the Eurodollar Reserve Percentage shall result in a change in the Base LIBOR Rate on the date on

which such change in such Eurodollar Reserve Percentage becomes effective. If the Base LIBOR Rate determined as provided above would be less than zero, the Base LIBOR Rate shall be deemed to be zero. Notwithstanding the foregoing, if the Borrower has delivered a written notice to the Administrative Agent certifying (i) that all or any portion of the Loans are subject to a Specified Rate Management Transaction (or any successor or replacement Rate Management Transaction) and (ii) that such Specified Rate Management Transaction (or any successor or replacement Rate Management Transaction) is not subject to a 0.0% interest rate floor, then Base LIBOR Rate shall not be subject to a floor of 0.0% with respect to such Loans. Each calculation by the Administrative Agent of the Base LIBOR Rate shall be conclusive and binding for all purposes, absent manifest error. To the extent a comparable or successor rate is approved by the Administrative Agent in connection herewith, the approved rate shall be applied to the then applicable Interest Period in a manner consistent with market practice as reasonably determined by the Administrative Agent to not be administratively feasible, such approved rate shall be applied in a manner reasonably determined by the Administrative Agent. All approvals and determinations to be made by the Administrative Agent in the preceding sentence shall be done in consultation with the Borrower. Notwithstanding the foregoing, unless otherwise specified in any amendment to this Agreement entered into in accordance with Section 4.8, in the event that a Benchmark Replacement with respect to the Base LIBOR Rate is implemented then all references herein to the Base LIBOR Rate shall be deemed references to such Benchmark Replacement.

"Base Rate Applicable Margin" means the Applicable Margin in effect for an Adjusted Base Rate Borrowing as determined in accordance with Section 2.9 hereof.

"Benchmark" means, initially, USD LIBOR; <u>provided</u> that if a Benchmark Transition Event, a Term SOFR Transition Event, an Early Opt-in Election or an Other Benchmark Rate Election, as applicable, and its related Benchmark Replacement Date have occurred with respect to USD LIBOR or the then-current Benchmark, then "Benchmark" means the applicable Benchmark Replacement to the extent that such Benchmark Replacement has replaced such prior benchmark rate pursuant to <u>Section 4.8(c)(i)</u>.

"Benchmark Replacement" means, for any Available Tenor,

(a) with respect to any Benchmark Transition Event or Early Opt-in Election, the first alternative set forth in the order below that can be determined by the Administrative Agent for the applicable Benchmark Replacement Date; <u>provided</u>, that, if the Borrower has provided a notification to the Administrative Agent in writing on or prior to such Benchmark Replacement Date that the Borrower has a Rate Management Transaction in place with respect to any of the Loans as of the date of such notice (which such notification the Administrative Agent shall be entitled to rely upon and shall have no duty or obligation to ascertain the correctness or completeness of), then the Administrative Agent, in its sole discretion, may decide not to determine the Benchmark Replacement pursuant to clause (a)(1) or clause (a)(2) for such Benchmark Transition Event or Early Opt-in Election with respect to such Loan, and may instead select (A) the alternate benchmark referenced in such Rate Management Transaction, which may be the sum of a fallback rate and spread adjustment, for the entire balance of such Loan, or (B) the alternate benchmark referenced in such Rate Management Transaction, which may be the sum of a fallback rate and spread adjustment,

for the portion of such Loan that is subject to a Rate Management Transaction, and the applicable Benchmark Replacement for the remaining portion of such Loan:

- (1) the sum of: (A) Term SOFR and (B) the related Benchmark Replacement Adjustment;
- (2) the sum of: (A) Daily Simple SOFR and (B) the related Benchmark Replacement Adjustment;
- (3) the sum of: (A) the alternate benchmark rate that has been selected by the Administrative Agent and the Borrower as the replacement for the then-current Benchmark for the applicable Corresponding Tenor giving due consideration to (i) any selection or recommendation of a replacement benchmark rate or the mechanism for determining such a rate by the Relevant Governmental Body or (ii) any evolving or then-prevailing market convention for determining a benchmark rate as a replacement for the then-current Benchmark for Dollar-denominated syndicated credit facilities at such time and (B) the related Benchmark Replacement Adjustment;
- (b) with respect to any Term SOFR Transition Event, the sum of (i) Term SOFR and (ii) the related Benchmark Replacement Adjustment; or
- (c) with respect to any Other Benchmark Rate Election, the sum of: (i) the alternate benchmark rate that has been selected by the Administrative Agent and the Borrower as the replacement for the then-current Benchmark for the applicable Corresponding Tenor giving due consideration to any evolving or then-prevailing market convention for determining a benchmark rate as a replacement for the then-current Benchmark for Dollar-denominated syndicated credit facilities at such time and (ii) the related Benchmark Replacement Adjustment;

provided that, (i) in the case of clause (a)(1), if the Administrative Agent decides that Term SOFR is not administratively feasible for the Administrative Agent, then Term SOFR will be deemed unable to be determined for purposes of this definition and (ii) in the case of clause (a)(1) or clause (b) of this definition, the applicable Unadjusted Benchmark Replacement is displayed on a screen or other information service that publishes such rate from time to time as selected by the Administrative Agent in its reasonable discretion. If the Benchmark Replacement as determined pursuant to clause (a)(1), (a)(2) or (a)(3), clause (b) or clause (c) of this definition would be less than the Floor, the Benchmark Replacement will be deemed to be the Floor for the purposes of this Agreement and the other Loan Documents.

"Benchmark Replacement Adjustment" means, with respect to any replacement of the then-current Benchmark with an Unadjusted Benchmark Replacement for any applicable Interest Period and Available Tenor for any setting of such Unadjusted Benchmark Replacement:

- (1) for purposes of clauses (a)(1) and (b) of the definition of "Benchmark Replacement," an amount equal to (A) 0.11448% (11.448 basis points) for an Available Tenor of one-month's duration, (B) 0.26161% (26.161 basis points) for an Available Tenor of three-months' duration and (C) 0.42826% (42.826 basis points) for an Available Tenor of six-months' duration:
- (2) for purposes of clause (a)(2) of the definition of "Benchmark Replacement," an amount equal to 0.11448% (11.448 basis points);
- (3) for purposes of clause (a)(3) of the definition of "Benchmark Replacement," the spread adjustment, or method for calculating or determining such spread adjustment, (which may be a positive or negative value or zero) that has been selected by the Administrative Agent and the Borrower giving due consideration to (i) any selection or recommendation of a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Available Tenor of such Benchmark with the applicable Unadjusted Benchmark Replacement by the Relevant Governmental Body on the applicable Benchmark Replacement Date or (ii) any evolving or then-prevailing market convention for determining a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Available Tenor of such Benchmark with the applicable Unadjusted Benchmark Replacement for Dollar-denominated syndicated credit facilities; and
- (4) for purposes of clause (c) of the definition of "Benchmark Replacement," the spread adjustment, or method for calculating or determining such spread adjustment, (which may be a positive or negative value or zero) that has been selected by the Administrative Agent and the Borrower giving due consideration to any evolving or then-prevailing market convention for determining a spread adjustment, or method for calculating or determining such spread adjustment, for the replacement of such Available Tenor of such Benchmark with the applicable Unadjusted Benchmark Replacement for Dollar-denominated syndicated credit facilities.

"Benchmark Replacement Conforming Changes" means, with respect to any Benchmark Replacement, any technical, administrative or operational changes (including changes to the definition of "Adjusted Base Rate," the definition of "Business Day," the definition of "Interest Period," the definition of "LIBOR Market Index Rate", timing and frequency of determining rates and making payments of interest, timing of borrowing requests or prepayment, conversion or continuation notices, length of lookback periods, the applicability of breakage provisions, and other technical, administrative or operational matters) that the Administrative Agent decides may be appropriate to reflect the adoption and implementation of such Benchmark Replacement and to permit the administration thereof by the Administrative Agent in a manner substantially consistent with market practice (or, if the Administrative Agent decides that adoption of any portion of such market practice is not administratively feasible or if the Administrative Agent determines that no market practice for the administration of such Benchmark Replacement exists, in such other manner of administration as the Administrative Agent decides is reasonably necessary in connection with the administration of this Agreement and the other Loan Documents).

"Benchmark Replacement Date" means the earliest to occur of the following events with respect to the then-current Benchmark:

- (a) in the case of clause (a) or (b) of the definition of "Benchmark Transition Event," the later of (a) the date of the public statement or publication of information referenced therein and (b) the date on which the administrator of such Benchmark (or the published component used in the calculation thereof) permanently or indefinitely ceases to provide all Available Tenors of such Benchmark (or such component thereof);
- (b) in the case of clause (c) of the definition of "Benchmark Transition Event," the date of the public statement or publication of information referenced therein;
- (c) in the case of a Term SOFR Transition Event, the date that is thirty (30) days after the Administrative Agent has provided the Term SOFR Notice to the Lenders and the Borrower pursuant to Section 4.8(c)(i)(B); or
- (d) in the case of an Early Opt-in Election or an Other Benchmark Rate Election, the sixth (6th) Business Day after the date notice of such Early Opt-in Election or Other Benchmark Rate Election, as applicable, is provided to the Lenders, so long as the Administrative Agent has not received, by 5:00 p.m. (New York City time) on the fifth (5th) Business Day after the date notice of such Early Opt-in Election or Other Benchmark Rate Election, as applicable, is provided to the Lenders, written notice of objection to such Early Opt-in Election or Other Benchmark Rate Election, as applicable, from Lenders comprising the Required Lenders.

For the avoidance of doubt, (i) if the event giving rise to the Benchmark Replacement Date occurs on the same day as, but earlier than, the Reference Time in respect of any determination, the Benchmark Replacement Date will be deemed to have occurred prior to the Reference Time for such determination and (ii) the "Benchmark Replacement Date" will be deemed to have occurred in the case of clause (a) or (b) with respect to any Benchmark upon the occurrence of the applicable event or events set forth therein with respect to all then-current Available Tenors of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Transition Event" means the occurrence of one or more of the following events with respect to the then-current Benchmark:

- (a) a public statement or publication of information by or on behalf of the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that such administrator has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof), permanently or indefinitely, <u>provided</u> that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof);
- (b) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof), the FRB, the Federal Reserve Bank of New York, an insolvency official with jurisdiction over the administrator for such Benchmark (or such component), a resolution authority with jurisdiction over the administrator for such Benchmark (or such component) or a court or an entity with similar insolvency or resolution authority over the administrator for such Benchmark (or such component),

which states that the administrator of such Benchmark (or such component) has ceased or will cease to provide all Available Tenors of such Benchmark (or such component thereof) permanently or indefinitely, <u>provided</u> that, at the time of such statement or publication, there is no successor administrator that will continue to provide any Available Tenor of such Benchmark (or such component thereof); or

(c) a public statement or publication of information by the regulatory supervisor for the administrator of such Benchmark (or the published component used in the calculation thereof) announcing that all Available Tenors of such Benchmark (or such component thereof) are no longer representative.

For the avoidance of doubt, a "Benchmark Transition Event" will be deemed to have occurred with respect to any Benchmark if a public statement or publication of information set forth above has occurred with respect to each then-current Available Tenor of such Benchmark (or the published component used in the calculation thereof).

"Benchmark Unavailability Period" means the period (if any) (x) beginning at the time that a Benchmark Replacement Date pursuant to clauses (a) or (b) of that definition has occurred if, at such time, no Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with Section 4.8(c) and (y) ending at the time that a Benchmark Replacement has replaced the then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with Section 4.8(c).

"Beneficial Ownership Certification" means a certification regarding beneficial ownership as required by the Beneficial Ownership Regulation.

"Beneficial Ownership Regulation" means 31 CFR § 1010.230.

"Benefit Plan" means any of (a) an "employee benefit plan" (as defined in ERISA) that is subject to Title I of ERISA, (b) a "plan" as defined in and subject to Section 4975 of the Code or (c) any Person whose assets include (for purposes of ERISA Section 3(42) or otherwise for purposes of Title I of ERISA or Section 4975 of the Code) the assets of any such "employee benefit plan" or "plan".

"BHC Act Affiliate" is defined in Section 14.19.

"Borrower" means First Industrial, L.P., along with its permitted successors and assigns.

"Borrower EP Interest Accrual Date" has the meaning assigned thereto in Section 12.19(c).

"Borrowing" means Loans of the same Type, made, converted or continued on the same date and, in the case of Eurocurrency Loans, as to which a single Interest Period is in effect.

"Borrowing Date" means a Business Day on which a Borrowing is made to the Borrower.

"Borrowing Notice" is defined in Section 2.7 hereof.

"Business Day" means a day, other than a Saturday, Sunday or holiday, on which banks are open for business in New York City; provided that with respect to any borrowings, disbursements and payments in respect of and calculations, interest rates and Interest Periods pertaining to any Eurocurrency Borrowings, such day is also a day on which banks are open for general business in London, England.

"<u>Capital Stock</u>" means any and all shares, interests, participations or other equivalents (however designated) of capital stock of a corporation, any and all equivalent ownership interests in a Person which is not a corporation and any and all warrants or options to purchase any of the foregoing.

"Cash Equivalents" shall mean (i) short term obligations of, or fully guaranteed by, the United States of America, (ii) commercial paper rated A 1 or better by Standard and Poor's Corporation or P-1 or better by Moody's Investors Service, Inc., or (iii) certificates of deposit issued by and time deposits with commercial banks (whether domestic or foreign) having capital and surplus in excess of \$100,000,000; provided in each case that the same provides for payment of both principal and interest (and not principal alone or interest alone) and is not subject to any contingency regarding the payment of principal or interest, provided that all such Cash Equivalents would qualify as cash equivalents in accordance with GAAP.

"Change in Control" means (a) the acquisition of ownership, directly or indirectly, beneficially or of record, by any Person or group (within the meaning of the Securities Exchange Act of 1934 and the rules of the Securities and Exchange Commission thereunder as in effect on the date hereof), of more than 40% of the aggregate voting power represented by the then outstanding voting stock of the General Partner, (b) the occupation of a majority of the seats (other than vacant seats) on the board of directors of the General Partner by Persons who were neither (i) nominated by the board of directors of the General Partner, (ii) appointed by directors so nominated, nor (iii) nominated by holders of the preferred stock in the General Partner pursuant to the terms of such stock; or (c) the General Partner shall cease to own, directly or indirectly, fifty-one percent (51%) or more of the Equity Interests of the Borrower.

"Change in Law" means the occurrence, after the date of this Agreement, of any of the following: (a) the adoption or taking effect of any law, rule, regulation or treaty, (b) any change in any law, rule, regulation or treaty or in the administration, interpretation or application thereof by any Governmental Authority, or (c) the making or issuance of any request, rules, guideline, requirement or directive (whether or not having the force of law) by any Governmental Authority; provided however, that notwithstanding anything herein to the contrary, (i) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines, requirements and directives thereunder, issued in connection therewith or in implementation thereof, and (ii) all requests, rules, guidelines, requirements and directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, shall in each case be deemed to be a "Change in Law" regardless of the date enacted, adopted, issued or implemented.

"Closing Date" means the date on which the conditions precedent in Section 5.1 are satisfied in accordance therewith and this Agreement becomes effective.

"Code" means the Internal Revenue Code of 1986 as amended from time to time, or any replacement or successor statute, and the regulations promulgated thereunder from time to time.

"Consolidated Leverage Ratio" means, as of any date of determination, the ratio of Consolidated Total Indebtedness to Implied Capitalization Value of the Consolidated Operating Partnership.

"Consolidated Leverage Ratio Increase Period" is defined in Section 9.7(b).

"Consolidated Operating Partnership" means the Borrower, the General Partner and any other subsidiary partnerships or entities of either of them which are required under GAAP to be consolidated with the Borrower and the General Partner for financial reporting purposes.

"Consolidated Secured Debt" means as of any date of determination, the sum of (a) the aggregate outstanding principal amount of all Indebtedness of the Consolidated Operating Partnership outstanding at such date which is secured by a Lien on any asset or Capital Stock of Consolidated Operating Partnership, including without limitation loans secured by mortgages, stock, or partnership interests, but excluding Defeased Debt and (b) the amount by which the aggregate principal amount of all Indebtedness of the Subsidiaries of the Borrower or the General Partner outstanding at such date exceeds \$5,000,000 (for the avoidance of doubt, (x) excluding Indebtedness of the Borrower and (y) including Guarantee Obligations (other than customary non-recourse carveout obligations) of the Subsidiaries of the General Partner (other than the Borrower) in respect of primary obligations of the Borrower or the General Partner), without duplication of any Indebtedness included under clause (a). For clarification, Consolidated Secured Debt shall include the Borrower's or the General Partner's Ownership Share of any Investment Affiliate's Indebtedness.

"Consolidated Senior Unsecured Debt" means as of any date of determination, the aggregate outstanding principal amount of all Indebtedness of the Consolidated Operating Partnership (which will include, without limitation, any Indebtedness that is secured by partnership interests and that is recourse to the Borrower or the Guarantor, where such recourse component applies only to the payment of principal and/or interest), outstanding at such date other than (a) Indebtedness which is contractually subordinated to the Indebtedness of the Consolidated Operating Partnership under the Loan Documents on terms acceptable to the Administrative Agent and (b) that portion of Consolidated Secured Debt described in clause (a) of that definition. For clarification, Consolidated Senior Unsecured Debt shall exclude the Borrower's or the General Partner's Ownership Share of any Investment Affiliate's Indebtedness.

"Consolidated Total Indebtedness" means as of any date of determination, all Indebtedness of the Consolidated Operating Partnership outstanding at such date, determined on a consolidated basis in accordance with GAAP, after eliminating intercompany items; provided that for purposes of defining "Consolidated Total Indebtedness" the term "Indebtedness" shall not include the short term debt (e.g. accounts payable, short term expenses) of the Borrower or the General Partner or Defeased Debt. For clarification, Consolidated Total Indebtedness shall include the Borrower's or the General Partner's Ownership Share of any Investment Affiliate's Indebtedness.

"Controlled Group" means all members of a controlled group of corporations and all trades or businesses (whether or not incorporated) under common control which, together with all or any of the entities in the Consolidated Operating Partnership, are treated as a single employer under Sections 414(b) or 414(c) of the Code.

"Conversion/Continuation Notice" is defined in Section 2.8.

"Corresponding Tenor" with respect to any Available Tenor means, as applicable, either a tenor (including overnight) or an interest payment period having approximately the same length (disregarding business day adjustment) as such Available Tenor.

"Covered Entity" is defined in Section 14.19.

"<u>Daily Simple SOFR</u>" means, for any day, SOFR, with the conventions for this rate (which will include a lookback) being established by the Administrative Agent in accordance with the conventions for this rate selected or recommended by the Relevant Governmental Body for determining "Daily Simple SOFR" for syndicated business loans; <u>provided</u>, that if the Administrative Agent decides that any such convention is not administratively feasible for the Administrative Agent, then the Administrative Agent may establish another convention in its reasonable discretion.

"<u>Debtor Relief Laws</u>" means the United States Bankruptcy Code, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar Applicable Laws relating to the relief of debtors in the United States of America or other applicable jurisdictions from time to time in effect.

"Debt Service" means for any period, (a) Interest Expense for such period <u>plus</u> (b) the aggregate amount of regularly scheduled principal payments of Indebtedness (excluding optional prepayments and balloon principal payments due on maturity in respect of any Indebtedness and mandatory excess cash flow sweeps) required to be made during such period by the Borrower, or any of its consolidated Subsidiaries <u>plus</u> (c) a percentage of all such regularly scheduled principal payments required to be made during such period by any Investment Affiliate on Indebtedness (excluding optional prepayments and balloon principal payments due on maturity in respect of any Indebtedness) taken into account in calculating Interest Expense, such percentage equal to the greater of (x) the percentage of the principal amount of such Indebtedness for which the Borrower or any consolidated Subsidiary is liable and (y) the Ownership Share in such Investment Affiliate held by the Borrower and any consolidated Subsidiaries, in the aggregate, without duplication.

"Default" means an event which, with notice or lapse of time or both, would become an Event of Default.

"<u>Default Rate</u>" means with respect to any Borrowing, a rate equal to the interest rate applicable to such Borrowing plus three percent (3%) per annum.

"Default Right" is defined in Section 14.19.

"Defaulting Lender" means, subject to Section 12.16(f), any Lender that (a) has failed to (i) fund all or any portion of its Loans within two (2) Business Days of the date such Loans were required to be funded hereunder unless such Lender notifies the Administrative Agent and the Borrower in writing that such failure is the result of such Lender's determination that one or more conditions precedent to funding (each of which conditions precedent, together with any applicable default, shall be specifically identified in such writing) has not been satisfied, or (ii) pay to the Administrative Agent or any other Lender any other amount required to be paid by it hereunder within two (2) Business Days of the date when due, (b) has notified the Borrower or the Administrative Agent in writing that it does not intend to comply with its funding obligations hereunder, or has made a public statement to that effect (unless such writing or public statement relates to such Lender's obligation to fund a Loan hereunder and states that such position is based on such Lender's determination that a condition precedent to funding (which condition precedent, together with any applicable default, shall be specifically identified in such writing or public statement) cannot be satisfied), (c) has failed, within three (3) Business Days after written request by the Administrative Agent or the Borrower, to confirm in writing to the Administrative Agent and the Borrower that it will comply with its prospective funding obligations hereunder (provided that such Lender shall cease to be a Defaulting Lender pursuant to this clause (c) upon receipt of such written confirmation by the Administrative Agent and the Borrower), (d) has, or has a direct or indirect parent company that has, (i) become the subject of (A) a proceeding under any Debtor Relief Law or (B) a Bail-In Action, or (ii) had appointed for it a receiver, custodian, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or assets, including the Federal Deposit Insurance Corporation or any other state or federal regulatory authority acting in such a capacity; provided that a Lender shall not be a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Lender or any direct or indirect parent company thereof by a Governmental Authority so long as such ownership interest does not result in or provide such Lender with immunity from the jurisdiction of courts within the United States of America or from the enforcement of judgments or writs of attachment on its assets or permit such Lender (or such Governmental Authority) to reject, repudiate, disayow or disaffirm any contracts or agreements made with such Lender, (e) has purchased or otherwise acquired all or any portion of another Lender's Term Loan Commitment, excluding any purchase or acquisition in accordance with the terms and conditions set forth in Section 13.1; or (f) has sold or otherwise transferred all or any portion of its Term Loan Commitment, excluding any sale or transfer in accordance with the terms and conditions set forth in Section 13.1. Any determination by the Administrative Agent that a Lender is a Defaulting Lender under clauses (a) through (f) above shall be conclusive and binding absent manifest error, and such Lender shall be deemed to be a Defaulting Lender (subject to Section 12.16(f)) upon delivery of written notice of such determination to the Borrower and each Lender.

"<u>Defeased Debt</u>" means that portion of debt which has already been defeased by depositing collateral in the form of obligations supported by the credit of the United States government in such amounts as are required and permitted under the terms of the applicable loan documents.

"<u>Disbursement Instruction Agreement</u>" means an agreement substantially in the form of <u>Exhibit C</u> to be executed and delivered by the Borrower pursuant to <u>Section 5.1</u>, as the same may be amended, restated or modified from time to time with the prior written approval of the Administrative Agent.

"Dollars" and "\$" mean United States Dollars.

"Early Opt-in Election" means, if the then-current Benchmark is USD LIBOR, the occurrence of:

- (a) a notification by the Administrative Agent to (or the request by the Borrower to the Administrative Agent to notify) each of the other parties hereto that at least five currently outstanding Dollar-denominated syndicated credit facilities at such time contain (as a result of amendment or as originally executed) a SOFR-based rate (including SOFR, a term SOFR or any other rate based upon SOFR) as a benchmark rate (and such syndicated credit facilities are identified in such notice and are publicly available for review), and
- (b) the joint election by the Administrative Agent and the Borrower to trigger a fallback from USD LIBOR and the provision by the Administrative Agent of written notice of such election to the Lenders.

"EBITDA" means, with respect to any Person, income before restructuring charges, non-cash impairment charges and other non-cash, non-recurring items determined in good faith by the Borrower and extraordinary items, without deduction of any losses related to initial offering costs of preferred stock which are written off due to the redemption of such preferred stock, and excluding any gains or losses from pay-off or retirement of debt and gains/losses on sales of Properties and excluding costs incurred in acquiring Properties, where such costs are required to be expensed under ASC 805 Business Combinations 805-10-25-23, as reported by such Person and its Subsidiaries on a consolidated basis in accordance with GAAP (reduced to eliminate any income from Investment Affiliates of such Person, any interest income and, with respect to the Consolidated Operating Partnership, any income from the assets used for Defeased Debt), plus Interest Expense, depreciation, amortization and income tax (if any) expense plus a percentage of such income (adjusted as described above) of any such Investment Affiliate equal to the allocable economic interest in such Investment Affiliate held by such Person and any Subsidiaries, in the aggregate (provided that no item of income or expense shall be included more than once in such calculation even if it falls within more than one of the foregoing categories).

"EEA Financial Institution" means (a) any credit institution or investment firm established in any EEA Member Country which is subject to the supervision of an EEA Resolution Authority, (b) any entity established in an EEA Member Country which is a parent of an institution described in clause (a) of this definition, or (c) any financial institution established in an EEA Member Country which is a subsidiary of an institution described in clauses (a) or (b) of this definition and is subject to consolidated supervision with its parent.

"EEA Member Country" means any of the member states of the European Union, Iceland, Liechtenstein, and Norway.

"EEA Resolution Authority" means any public administrative authority or any Person entrusted with public administrative authority of any EEA Member Country (including any delegee) having responsibility for the resolution of any credit institution or investment firm established in any EEA Member Country.

"Effective Date" means each Borrowing Date (other than a continuation or conversion).

"<u>Eligible Assignee</u>" means (a) a Lender, (b) an Affiliate of a Lender, (c) an Approved Fund and (d) any other Person (other than a natural person) approved by the Administrative Agent (such approval not to be unreasonably withheld or delayed by any of the foregoing); provided that notwithstanding the foregoing, "Eligible Assignee" shall not include the Borrower or any of the Borrower's Affiliates or Subsidiaries.

"Environmental Laws" means any and all Federal, state, local or municipal laws, rules, orders, regulations, statutes, ordinances, codes, decrees, requirements of any Governmental Authority having jurisdiction over the Borrower, its Subsidiaries or Investment Affiliates, or their respective assets, and regulating or imposing liability or standards of conduct concerning protection of human health or the environment, as now or may at any time hereafter be in effect, in each case to the extent the foregoing are applicable to the operations of the Borrower, any Investment Affiliate, or any Subsidiary or any of their respective assets or Properties.

"Equity Interests" means, with respect to any Person, any share of capital stock of (or other ownership or profit interests in) such Person, any warrant, option or other right for the purchase or other acquisition from such Person of any share of capital stock of (or other ownership or profit interests in) such Person, any security convertible into or exchangeable for any share of capital stock of (or other ownership or profit interests in) such Person or warrant, right or option for the purchase or other acquisition from such Person of such shares (or such other interests), and any other ownership or profit interest in such Person (including, without limitation, partnership, member or trust interests therein), whether voting or nonvoting, and whether or not such share, warrant, option, right or other interest is authorized or otherwise existing on any date of determination.

"Equity Value" is defined in Section 10.10 hereof.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and regulations promulgated thereunder from time to time.

"Erroneous Payment" is defined in Section 12.19(a) hereof.

"<u>EU Bail-In Legislation Schedule</u>" means the EU Bail-In Legislation Schedule published by the Loan Market Association (or any successor person), as in effect from time to time.

"<u>Eurocurrency</u>", when used in reference to any Loan or Borrowing, refers to whether such Loan, or the Loans comprising such Borrowing, are bearing interest at a rate determined by reference to the Adjusted LIBOR Rate.

"<u>Eurodollar Reserve Percentage</u>" means, for any day, the percentage (stated as a decimal) which is in effect for such day as prescribed by the Board of Governors of the Federal Reserve System (or any successor) for determining the maximum reserve requirement (including without limitations, any basic, supplemental or emergency reserves) in respect of eurocurrency liabilities or any similar category.

"Event of Default" means any event set forth in Article X hereof.

"Excluded Taxes" means, in the case of each Lender or applicable Lending Office and the Administrative Agent, (a) taxes imposed on its overall net income, and franchise taxes imposed on it, by (i) the jurisdiction under the laws of which such Lender or the Administrative Agent is incorporated or organized or (ii) the jurisdiction in which the Administrative Agent's or such Lender's principal executive office of such Lender's applicable Lending Office is located, and (b) any United States federal withholding taxes imposed by FATCA.

"<u>Existing Revolving Credit Agreement</u>" means that certain Fourth Amended and Restated Unsecured Revolving Credit Agreement dated as of July 7, 2021 among the Borrower, the General Partner, the lenders party thereto and Wells, as administrative agent, and all amendments, modifications, supplements, refinancing and replacements thereto.

"Existing Term Loan Agreement" has the meaning specified in the preliminary statements of this Agreement.

"Facility" means the Term Loan Commitments and the extensions of credit made thereunder.

"<u>FATCA</u>" means Sections 1471 through 1474 of the Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with) and any current or future regulations or official interpretations thereof and any agreement entered into pursuant to Section 1471(b)(1) of the Code.

"Federal Funds Effective Rate" means, for any period, a fluctuating interest rate per annum equal for each day during such period to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System, as published for such day (or, if such day is not a Business Day, for the immediately preceding Business Day) by the Federal Reserve Bank of New York, or, if such rate is not so published for any day which is a Business Day, the average of the quotations for such day on such transactions received by the Administrative Agent from three Federal funds brokers of recognized standing selected by the Administrative Agent in its sole discretion; provided, that, if the Federal Funds Effective Rate shall be less than zero, such rate shall be deemed to be zero for purposes of this Agreement.

"Fixed Charge Coverage Ratio" is defined in Section 9.7(a).

"Floor" means the benchmark rate floor, if any, provided in this Agreement initially (as of the execution of this Agreement, the modification, amendment or renewal of this Agreement or otherwise) with respect to USD LIBOR. Notwithstanding the foregoing, if the Borrower has delivered a written notice to the Administrative Agent certifying (i) that all or any portion of the Loans are subject to a Specified Rate Management Transaction (or any successor or replacement Rate Management Transaction) and (ii) that such Specified Rate Management Transaction (or any

successor or replacement Rate Management Transaction) is not subject to a 0.0% interest rate floor, then the Benchmark Replacement shall not be subject to a Floor of 0.0% with respect to such Loans.

"FRB" means the Board of Governors of the Federal Reserve System of the United States.

"<u>Fund</u>" means any Person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its business.

"GAAP" means those generally accepted accounting principles in the United States of America as in effect from time to time that are consistent with those utilized in preparing the audited financial statements of the Borrower required hereunder; provided that all financial computations shall be made in accordance with GAAP as in effect on the Agreement Execution Date.

"General Partner" means First Industrial Realty Trust, Inc., a Maryland corporation that is listed on a national securities exchange and is qualified as a real estate investment trust. The General Partner is the sole general partner of the Borrower.

"Governmental Authority" means the government of the United States of America or any other nation, or of any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government (including any supra-national bodies such as the European Union or the European Central Bank).

"Gross Negligence" means recklessness, or actions taken or omitted with conscious indifference to or the complete disregard of consequences or rights of others affected. Gross Negligence does not mean the absence of ordinary care or diligence, or an inadvertent act or inadvertent failure to act. If the term "gross negligence" is used with respect to the Administrative Agent or any Lender or any indemnitee in any of the other Loan Documents, it shall have the meaning set forth herein.

"Ground Lease Payments" means, for any period, payment made in cash during such period in respect of any ground lease with respect to which the Borrower or any of its Subsidiaries is a lessee.

"Guarantee Obligation" means as to any Person (the "guaranteeing person"), any obligation (determined without duplication) of (a) the guaranteeing person or (b) another Person (including, without limitation, any bank under any letter of credit) to induce the creation of which the guaranteeing person has issued a reimbursement, counter indemnity or similar obligation, in either case guaranteeing or in effect guaranteeing (but only to the extent that the guaranty applies to the payment of principal or interest due under recourse Indebtedness) any Indebtedness, leases, dividends or other obligations (the "primary obligations") of any other third Person (the "primary obligor") in any manner, whether directly or indirectly, including, without limitation, any obligation of the guaranteeing person, whether or not contingent, (i) to purchase any such primary obligation or any property constituting direct or indirect security therefor, (ii) to advance or supply funds (1) for the purchase or payment of any such primary obligation or (2) to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor, (iii) to purchase property, securities or services primarily for the purpose of assuring the

owner of any such primary obligation of the ability of the primary obligor to make payment of such primary obligation or (iv) otherwise to assure or hold harmless the owner of any such primary obligation against loss in respect thereof; <u>provided</u>, <u>however</u>, that the term Guarantee Obligation shall not include endorsements of instruments for deposit or collection in the ordinary course of business. The amount of any Guarantee Obligation of any guaranteeing person shall be deemed to be the maximum stated amount of the primary obligation relating to such Guarantee Obligation (or, if less, the maximum stated liability set forth in the instrument embodying such Guarantee Obligation), <u>provided</u>, that in the absence of any such stated amount or stated liability, the amount of such Guarantee Obligation shall be such guaranteeing person's maximum reasonably anticipated liability in respect thereof as determined by the Borrower in good faith. Notwithstanding the foregoing, a guaranty of customary non-recourse carveouts shall not be deemed a "Guarantee Obligation" for purposes of this Agreement.

"<u>Guaranty</u>" means the Guaranty executed by the General Partner in the form attached hereto as <u>Exhibit D</u> and each guaranty executed and delivered pursuant to <u>Section 8.13</u> hereof.

"IBA" has the meaning assigned thereto in Section 1.5.

"Implied Capitalization Value" means for any Person as of any date, the sum (without duplication) of (i) the quotient of (x) the Adjusted EBITDA for such Person during the most recent four fiscal quarters (which Adjusted EBITDA shall exclude any Adjusted EBITDA attributable to all assets of the type described in clause (ii), Assets Under Development, Assets Acquired Not in Service or Rollover Projects, and which Adjusted EBITDA attributable to each Project which was formerly a Rollover Project shall not be less than zero), and (y) the Applicable Cap Rate, plus (ii) the purchase price paid by such Person (less any amounts paid to such Person as a purchase price adjustment, held in escrow, retained as a contingency reserve or in connection with other similar arrangements) for any Property (other than Assets Under Development) acquired by such Person during the immediately preceding period of four consecutive fiscal quarters, plus (iii) an amount equal to the then current book value of each Asset Under Development, plus (iv) the then current book value of Unimproved Land, plus (v) with respect to each Rollover Project, an amount equal to 50% of the then current book value, determined in accordance with GAAP, of such Rollover Project, plus (vi) an amount equal to 100% of unrestricted cash and unrestricted cash equivalents, including any cash on deposit with a qualified intermediary with respect to a deferred tax-free exchange (and specifically excluding any cash or cash equivalents being used to support Defeased Debt), plus (vii) an amount equal to 100% of the then current book value, determined in accordance with GAAP, of all first mortgage receivables on income producing commercial properties. For purposes of determining Implied Capitalization Value, to the extent the amount of Implied Capitalization Value attributable to unconsolidated joint ventures and partnerships and other assets of the type described in clauses (iii), (iv), (v) and (vii) would, in the aggregate, exceed 30% of Implied Capitalization Value, such excess shall be excluded. The Borrower's Ownership Share of assets held by Investment Affiliates (excluding assets of the type described in the immediately preceding clause (vi)) will be included in Implied Capitalization Value calculations consistent with the above described treatment for wholly owned assets. In the case of a newly formed Investment Affiliate, the Borrower's Ownership Share of assets held by the Investment Affiliate shall be calculated by multiplying (a) total assets plus accumulated depreciation of the Investment Affiliate by (b) the Ownership Share of such Investment Affiliate. This valuation methodology will be used for the first four quarters following the formation of any Investment Affiliate. For purposes of computing the Implied

Capitalization Value, Adjusted EBITDA may be increased from quarter to quarter by the amount of (A) net cash flow from new leases of space at the Properties (where such net cash flow has not then been included in EBITDA) which have a minimum term of one year and (B) Properties which were previously Assets Under Development but which have been completed during such four quarter period and have at least some tenants in possession of the respective leased spaces and conducting business operations therein each will be included in the calculation of Implied Capitalization Value using pro forma EBITDA for such four quarter period, so long as a "new opening summary" form is submitted to, and approved by, Administrative Agent for each newly-opened Property during such four quarter period. "New opening summary" forms will only be required for developments to the extent the aggregate thereof exceeds \$600,000,000 during any fiscal year.

"Incremental Loans" is defined in Section 2.2 hereof.

"Indebtedness" of any Person at any date means without duplication, (a) all indebtedness of such Person for borrowed money, (b) all obligations of such Person for the deferred purchase price of property or services (other than current trade liabilities and other accounts payable, and accrued expenses incurred in the ordinary course of business and payable in accordance with customary practices), repurchase obligations, takeout commitments or forward equity commitments, in each case evidenced by a binding agreement (excluding any such obligation to the extent the obligation can be satisfied by the issuance of Equity Interests (other than Mandatorily Redeemable Stock)), to the extent such obligations constitute indebtedness for the purposes of GAAP, (c) any other indebtedness of such Person which is evidenced by a note, bond, debenture or similar instrument, (d) all obligations of such Person under financing leases and capital leases, (e) all obligations of such Person in respect of acceptances issued or created for the account of such Person, (f) all Guarantee Obligations of such Person (excluding in any calculation of consolidated Indebtedness of the Consolidated Operating Partnership, Guarantee Obligations of any member of the Consolidated Operating Partnership in respect of primary obligations of any other member of the Consolidated Operating Partnership), (g) all reimbursement obligations of such Person for letters of credit and other contingent liabilities, (h) Net Mark-to-Market Exposure under Rate Management Transactions, (i) all liabilities secured by any lien (other than liens for taxes not yet due and payable) on any property owned by such Person even though such Person has not assumed or otherwise become liable for the payment thereof, (j) any repurchase obligation or liability of such Person or any of its Subsidiaries with respect to accounts or notes receivable sold by such Person or any of its Subsidiaries, (k) such Person's Ownership Share of debt in Investment Affiliates and any loans where such Person is liable as a general partner and (1) all obligations of such Person to purchase, redeem, retire, defease or otherwise make any payment in respect of any Mandatorily Redeemable Stock issued by such Person or any other Person, valued at the greater of its voluntary or involuntary liquidation preference plus accrued and unpaid dividends.

"Insolvent" means insolvent as defined in Section 101(32) of the United States Bankruptcy Code, as amended.

"Interest Expense" means all interest expense of the Consolidated Operating Partnership determined in accordance with GAAP <u>plus</u> (i) capitalized interest not covered by an interest reserve from a loan facility, <u>plus</u> (ii) the allocable portion (based on liability) of any interest incurred on any obligation for which the Consolidated Operating Partnership is wholly or partially liable under guaranties covering the payment of principal and/or interest, <u>plus</u> (iii) the allocable percentage of any interest incurred on any Indebtedness of any Investment Affiliate, whether recourse or non-recourse, equal to the applicable Ownership Share in such Investment Affiliate held by the Consolidated Operating Partnership, in the aggregate, <u>provided</u> that no expense shall be included more than once in such calculation even if it falls within more than one of the foregoing categories; <u>provided</u>, <u>however</u>, that "Interest Expense" shall not include interest on loans after they become Defeased Debt.

"Interest Period" means with respect to a Eurocurrency Borrowing, (a) in the case of a Eurocurrency Borrowing made on the Closing Date or any continuation thereof (i) initially a period commencing on (and including) the Closing Date and ending on (but excluding) August 2, 2021 (the "Stub Period") and (ii) thereafter, unless the Borrower shall elect otherwise in a Conversion/Continuation Notice (which election shall be governed by clause (b) below), a period commencing on (and including) the first Business Day of each calendar month, beginning with August 2, 2021, and ending on (but excluding) the first Business Day in the immediately subsequent calendar month; and (b) in the case of any other Eurocurrency Borrowing, a period commencing on (and including) the date such Eurocurrency Borrowing is made (or in the case of the continuation of a Eurocurrency Loan, the last day of the preceding Interest Period for such Loan) and ending on (but excluding) the first Business Day in the first, third or sixth calendar month thereafter, as the Borrower may select in the applicable Borrowing Notice or Conversion/Continuation Notice, as the case may be. For the avoidance of doubt, except with respect to the Stub Period, each Interest Period shall begin on the first Business Day of a calendar month, and any Borrowing made on a date other than the first Business Day of a calendar month shall be an Adjusted Base Rate Borrowing unless and until the Borrower shall deliver a Conversion/Continuation Notice providing for the conversion of such Loan to a Eurocurrency Borrowing beginning on the first Business Day of a calendar month. Notwithstanding the foregoing or anything in this Agreement to the contrary, if any Interest Period would otherwise end after the Maturity Date, such Interest Period shall end on the Maturity Date.

"Investment Affiliate" means any Person in which the Consolidated Operating Partnership, directly or indirectly, has an ownership interest, whose financial results are not consolidated under GAAP with the financial results of the Consolidated Operating Partnership on the consolidated financial statements of the Consolidated Operating Partnership.

"Investment Grade Rating" means, for any class of non-credit enhanced long-term senior unsecured debt issued by the Borrower, (a) a rating of BBB- or higher from S&P or (b) a rating of Baa3 or better from Moody's.

"KPI Metric" shall mean, for any Reference Year, the ratio, stated as a percentage (rounded to the nearest one decimal point), of:

- (a) the Borrower's aggregate investment incurred (determined in accordance with GAAP, and whether incurred during or prior to such Reference Year) with respect to Substantially Completed Properties for such Reference Year that achieve LEED certification from Green Business Certification, Inc. (or certification under another similarly recognized rating system approved by the Administrative Agent and the Required Lenders) (such certification, the "Sustainability Certification") prior to the delivery of the applicable Pricing Certificate for such Reference Year, and in respect of which the Borrower shall have delivered to the Administrative Agent evidence reasonably acceptable to the Administrative Agent of such Sustainability Certification prior to or concurrently with the delivery of the applicable Pricing Certificate, to
- (b) the Borrower's aggregate investment incurred (determined in accordance with GAAP, and whether incurred during or prior to such Reference Year) with respect to Substantially Completed Properties for such Reference Year;

provided that, in the case of each of clause (a) and (b), such amounts shall be set forth in the Borrower's quarterly "Supplemental Information" report made available on the Borrower's website (or any similar report made publicly available on the Borrower's website) for the final quarterly period of such Reference Year. For purposes of this definition, aggregate investment incurred shall be determined for the Borrower and its Subsidiaries on a consolidated basis in accordance with GAAP and consistent with the determination of the aggregate investment incurred as set forth in the Borrower's quarterly "Supplemental Information" report available on the Borrower's website for the fiscal quarter ending March 31, 2021.

"KPI Metric Applicable Margin Adjustment Amount" means, with respect to any period between Sustainability Pricing Adjustment Dates, solely with respect to the LIBOR Applicable Margin and Base Rate Applicable Margin:

- (a) negative 0.01%, if (i) the then aggregate current book value of all Substantially Completed Properties for the applicable Reference Year shall be equal to or greater than \$100,000,000 as of the last day of such Reference Year and (ii) the KPI Metric for the applicable Reference Year as set forth in the Pricing Certificate for such Reference Year is greater than or equal to the KPI Metric Target for such Reference Year; and
- (b) 0.00%, if (i) the KPI Metric for the applicable Reference Year as set forth in the Pricing Certificate for such Reference Year is less than the KPI Metric Target for such Reference Year, (ii) the Borrower shall fail to deliver a Pricing Certificate (or omit to report the KPI Metric on any Pricing Certificate) for such Reference Year on or prior to the date the annual compliance certificate is required to be delivered

pursuant to <u>Section 8.2(iv)</u> or (iii) the then aggregate current book value of all Substantially Completed Properties for the applicable Reference Year shall be less than \$100,000,000 as of the last day of such Reference Year.

"KPI Metric Target" means, with respect to any Reference Year, twenty-five percent (25.0%).

"Lenders" means, collectively, Wells, and the other Persons executing this Agreement in such capacity, or any Person which subsequently executes and delivers any amendment hereto in such capacity and each of their respective permitted successors and assigns. Where reference is made to "the Lenders" in any Loan Document it shall be read to mean "all of the Lenders".

"<u>Lending Office</u>" means, for each Lender and for each Type of Loan, the office of such Lender specified in such Lender's Administrative Questionnaire or in the applicable Assignment and Assumption Agreement or such other office of such Lender as such Lender may notify the Administrative Agent in writing from time to time.

"<u>LIBOR Applicable Margin</u>" means, as of any date with respect to any Eurocurrency Borrowing, the Applicable Margin in effect for such Eurocurrency Borrowing as determined in accordance with <u>Section 2.9</u> hereof.

"LIBOR Borrowing" means, the Eurocurrency Borrowings and Adjusted Base Rate Borrowings, collectively, of any Lender.

"LIBOR Market Index Rate" means, for any day, the Base LIBOR Rate as of that day that would be applicable for a Eurocurrency Borrowing having a one month Interest Period determined at approximately 10:00 a.m. (Central Time) for such day (rather than 11:00 a.m. (London time) two (2) Business Days prior to the first day of such Interest Period as otherwise provided in the definition of Base LIBOR Rate), or if such day is not a Business Day, the immediately preceding Business Day. The LIBOR Market Index Rate shall be determined on a daily basis.

"<u>Lien</u>" means any mortgage, pledge, security interest, encumbrance, lien or charge of any kind (including, without limitation, any conditional sale or other title retention agreement or lease in the nature thereof, any filing or agreement to file a financing statement as debtor under the Uniform Commercial Code on any property leased to any Person under a lease which is not in the nature of a conditional sale or title retention agreement, or any subordination agreement in favor of another Person).

"Loan" means, with respect to a Lender, such Lender's loan (including any Incremental Loans) made pursuant to Article II (or any conversion or continuation thereof).

"Loan Documents" means this Agreement, the Notes, the Guaranty and any and all other agreements or instruments required and/or provided to Lenders hereunder or thereunder, as any of the foregoing may be amended from time to time.

"<u>Loan Party</u>" means each of the Borrower, the General Partner, each other Person who guarantees all or a portion of the Obligations and/or who at any time pledges any collateral to secure all or a portion of the Obligations.

"London Banking Day" means any day on which dealings in Dollar deposits are conducted by and between banks in the London interbank Eurodollar market.

"Mandatorily Redeemable Stock" means, with respect to any Person, any Equity Interest of such Person which by the terms of such Equity Interest (or by the terms of any security into which it is convertible or for which it is exchangeable or exercisable), upon the happening of any event or otherwise (a) matures or is mandatorily redeemable, pursuant to a sinking fund obligation or otherwise (other than an Equity Interest to the extent redeemable in exchange for common stock or other equivalent common Equity Interests), (b) is convertible into or exchangeable or exercisable for Indebtedness or Mandatorily Redeemable Stock, or (c) is redeemable at the option of the holder thereof, in whole or in part (other than an Equity Interest which is redeemable solely in exchange for common stock or other equivalent common Equity Interests); in each case, on or prior to the Maturity Date.

"Market Value Net Worth" means at any time, the Implied Capitalization Value of a Person at such time minus the Indebtedness of such Person at such time.

"Material Acquisition" means any acquisition (or series of related acquisitions) permitted by the Loan Documents and consummated in accordance with the terms of the Loan Documents if the aggregate consideration paid in respect of such acquisition (including any Indebtedness assumed in connection therewith) exceeds 10% of the Implied Capitalization Value of the Consolidated Operating Partnership.

"Material Adverse Effect" means, with respect to any matter, that such matter in the Required Lenders' good faith judgment may (x) materially and adversely affect the business, properties, condition or results of operations of the Consolidated Operating Partnership taken as a whole, or (y) constitute a non-frivolous challenge to the validity or enforceability of any material provision of any Loan Document against any obligor party thereto.

"<u>Material Adverse Financial Change</u>" shall be deemed to have occurred if the Required Lenders, in their good faith judgment, determine that a material adverse financial change has occurred which could prevent timely repayment of any Borrowing hereunder or materially impair the Borrower's ability to perform its obligations under any of the Loan Documents.

"Material Credit Facility" means, as to the Borrower, the General Partner and their Subsidiaries, this Agreement and any other agreement(s) creating or evidencing indebtedness for borrowed money (excluding any Indebtedness which is "non-recourse") entered into on or after the Agreement Execution Date by the Borrower, the General Partner or any of their Subsidiaries, or in respect of which the Borrower, the General Partner or any of their Subsidiaries is an obligor or otherwise provides a guarantee or other credit support (other than a guarantee of customary recourse exceptions) ("Credit Facility"), in a principal amount outstanding or available for borrowing equal to or greater than \$5,000,000 (or the equivalent of such amount in the relevant currency of payment, determined as of the date of the closing of such facility based on the exchange rate of such other

currency); and if no Credit Facility or Credit Facilities equal or exceed such amounts, then the largest Credit Facility shall be deemed to be a Material Credit Facility.

- "<u>Materials of Environmental Concern</u>" means any gasoline or petroleum (including crude oil or any fraction thereof) or petroleum products or any hazardous or toxic substances, materials or wastes, defined or regulated as such in or under any Environmental Law, including, without limitation, asbestos, radon, polychlorinated biphenyls and urea formaldehyde insulation.
- "Maturity Date" means July 7, 2026; or such earlier date on which the principal balance of the Facility and all other sums due in connection with the Facility shall be due as a result of the acceleration of the Facility.
 - "Monetary Default" means any Default involving the Borrower's failure to pay any of the Obligations when due.
 - "Moody's" means Moody's Investors Service, Inc. and its successors.
- "Net Mark-to-Market Exposure" of a Person means, as of any date of determination, the excess (if any) of all unrealized losses over all unrealized profits of such Person arising from Rate Management Transactions. "Unrealized losses" means the fair market value of the cost to such Person of unwinding such Rate Management Transaction as of the date of determination (assuming the Rate Management Transaction were to be terminated as of that date), and "unrealized profits" means the fair market value of the gain to such Person of unwinding such Rate Management Transaction as of the date of determination (assuming such Rate Management Transaction were to be terminated as of that date).
 - "Non-Defaulting Lender" means, at any time, each Lender that is not a Defaulting Lender at such time.
- "Note" means, to the extent issued pursuant to <u>Section 2.5</u>, with respect to the Facility, the promissory note payable to the order of each Lender in the amount of such Lender's maximum applicable Term Loan Commitment in substantially the form attached hereto as <u>Exhibit B</u> (or such other form approved by the Administrative Agent) (collectively, the "<u>Notes</u>").
- "Obligations" means the Borrowings and all accrued and unpaid fees and all other obligations of the Borrower to the Administrative Agent or any or all of the Lenders arising under this Agreement or any of the other Loan Documents.
 - "OFAC" means the U.S. Department of the Treasury's Office of Foreign Assets Control.
- "Other Benchmark Rate Election" means, with respect to a Benchmark with respect to any Obligations, interest, fees, commissions or other amounts denominated in Dollars or calculated with respect thereto, if such Benchmark is USD LIBOR, the occurrence of:
- (a) a notification by the Administrative Agent to (or the request by the Borrower to the Administrative Agent to notify) each of the other parties hereto that at least five currently outstanding Dollar-denominated syndicated credit facilities at such time contain (as a result of amendment or as originally executed), in lieu of a USD LIBOR-based rate, a term benchmark rate that is not a SOFR-

based rate as a benchmark rate (and such syndicated credit facilities are identified in such notice and are publicly available for review), and

- (b) the joint election by the Administrative Agent and the Borrower to trigger a fallback from USD LIBOR and the provision by the Administrative Agent of written notice of such election to the Lenders.
 - "Other Taxes" is defined in Section 4.5(ii).
- "Ownership Share" means, with respect to any Investment Affiliate, the pro rata share of the nominal ownership interests held by the Consolidated Operating Partnership, in the aggregate, in such Investment Affiliate, without duplication (e.g., if the Consolidated Operating Partnership owns 25% of an Investment Affiliate, but receives 90% of the economic benefits from such Investment Affiliate, then the Ownership Share shall be equal to 25%).
 - "Parent" means, with respect to any Lender, any Person as to which such Lender is, directly or indirectly, a subsidiary.
 - "Participants" is defined in Section 13.2.1 hereof.
 - "Payment Recipient" has the meaning assigned thereto in Section 12.19(a).
 - "PBGC" means the Pension Benefit Guaranty Corporation or any entity succeeding to any or all of its functions under ERISA.
- "<u>Percentage</u>" means, with respect to any Lender, the percentage of the Total Term Loan Exposure represented by such Lender's Term Loan Exposure at such time giving effect to any assignments and any Incremental Loans pursuant to <u>Section 2.2</u>.
 - "Permitted Liens" are defined in Section 9.5 hereof.
- "<u>Permitted Negative Pledge</u>" means a "negative pledge" that (a) establishes a maximum ratio of unsecured debt to unencumbered assets, or of secured debt to total assets, or that otherwise conditions a Person's ability to encumber its assets upon the maintenance of one or more specified ratios that limit such Person's ability to encumber its assets but that does not generally prohibit the encumbrance of its assets, or the encumbrance of specific assets or (b) provides for the grant in favor of the holders of any unsecured debt of any equal and ratable Lien in connection with the pledge of any property or asset to secure the Facility.
- "<u>Person</u>" means an individual, a corporation, a limited or general partnership, an association, a joint venture or any other entity or organization, including a governmental or political subdivision or an agent or instrumentality thereof.
- "Plan" means an employee benefit plan as defined in Section 3(3) of ERISA, whether or not terminated, as to which the Borrower or any member of the Controlled Group may have any liability.

"Pricing Certificate" means a certificate substantially in the form of Exhibit K executed by an officer of the Borrower setting forth in reasonable detail the calculation of KPI Metric and the Sustainability Margin Adjustment, in each case, for the Reference Year covered thereby, and confirming the website for the concurrent public reporting of Borrower's quarterly "Supplemental Information" report for the final quarterly period of such Reference Year.

"Project" means any real estate asset which is 100% owned by the Borrower or by any Wholly Owned Subsidiary and which is operated as an industrial property.

"Property" means each parcel of real property owned or operated by the Borrower, any Subsidiary or Investment Affiliate.

"Property Operating Income" means, with respect to any Property, for any period, earnings from rental operations (computed in accordance with GAAP but without deduction for reserves) attributable to such Property <u>plus</u> depreciation, amortization and interest expense with respect to such Property for such period, and, if such period is less than a year, adjusted by straight lining various ordinary operating expenses which are payable less frequently than once during every such period (e.g. real estate taxes and insurance). The earnings from rental operations reported for the immediately preceding fiscal quarter shall be adjusted to include pro forma earnings (as substantiated to the satisfaction of the Administrative Agent) for an entire quarter for any Property acquired or placed in service during such fiscal quarter and to exclude earnings during such quarter from any property not owned as of the end of the quarter.

"PTE" means a prohibited transaction class exemption issued by the U.S. Department of Labor, as any such exemption may be amended from time to time.

"Purpose Credit" has the meaning ascribed to it in Regulation U of the Board of Governors of the Federal Reserve System.

"QFC" is defined in Section 14.19.

"Qualified Officer" means, with respect to any entity, the chief financial officer, chief accounting officer, controller or assistant controller of such entity if it is a corporation or of such entity's general partner if it is a partnership.

"Rate Management Obligations" of a Person means any and all obligations of such Person, whether absolute or contingent and howsoever and whensoever created, arising, evidenced or acquired (including all renewals, extensions and modifications thereof and substitutions therefor), under (i) any and all Rate Management Transactions, and (ii) any and all cancellations, buy backs, reversals, terminations or assignments of any Rate Management Transactions.

"Rate Management Transaction" means any transaction (including an agreement with respect thereto) now existing or hereafter entered by the Borrower which is a rate swap, basis swap, forward rate transaction, commodity swap, commodity option, equity or equity index swap, equity or equity index option, interest rate option, foreign exchange transaction, cap transaction, floor transaction, collar transaction, forward transaction, currency swap transaction, cross-currency rate swap transaction, currency option or any other similar transaction (including any option with respect to any

of these transactions) or any combination thereof, whether linked to one or more interest rates, foreign currencies, commodity prices, equity prices or other financial measures.

"Rating Agency" means S&P or Moody's.

"Recovery Event" means any settlement of or payment in respect of any property or casualty insurance claim or any condemnation proceeding relating to any Property.

"Reference Time" with respect to any setting of the then-current Benchmark means (1) if such Benchmark is USD LIBOR, 11:00 a.m. (London time) on the day that is two (2) London Banking Days preceding the date of such setting, and (2) if such Benchmark is not USD LIBOR, the time determined by the Administrative Agent in its reasonable discretion.

"Reference Year" means, with respect to any Pricing Certificate, the fiscal year ending immediately prior to the date of such Pricing Certificate.

"Register" is defined in Section 13.4.

"Regulation D" means Regulation D of the Board of Governors of the Federal Reserve System as from time to time in effect and any successor or other regulation or official interpretation of said Board of Governors relating to reserve requirements applicable to member banks of the Federal Reserve System.

"Relevant Governmental Body" means the FRB or the Federal Reserve Bank of New York, or a committee officially endorsed or convened by the FRB or the Federal Reserve Bank of New York, or any successor thereto.

"Reportable Event" means a reportable event as defined in Section 4043 of ERISA and the regulations issued under such section, with respect to a Plan, excluding, however, such events as to which the PBGC by regulation waived the requirement of Section 4043(a) of ERISA that it be notified within 30 days of the occurrence of such event, <u>provided</u> that a failure to meet the minimum funding standard of Section 412 of the Code and of Section 302 of ERISA shall be a Reportable Event regardless of the issuance of any such waivers in accordance with either Section 4043(a) of ERISA or Section 412(d) of the Code.

"Required Lenders" means Lenders in the aggregate having at least 51% of the Aggregate Term Loan Commitment then in effect or, if the Aggregate Term Loan Commitment has been terminated, the Total Term Loan Exposure then outstanding. In determining such percentage at any given time, all then existing Defaulting Lenders will be disregarded and excluded, and at all times when two or more Lenders (excluding Defaulting Lenders) are party to this Agreement, the term "Required Lenders" shall in no event mean less than two Lenders.

"Resolution Authority" means an EEA Resolution Authority or, with respect to any UK Financial Institution, a UK Resolution Authority.

"Rollover Projects" means those Projects which, due to no or low occupancy at such Project, have a value, determined by dividing the Property Operating Income for such a Project for the most recent four fiscal quarters by the Applicable Cap Rate, of less than 50% of book value, <u>provided</u> that

a Project shall no longer be treated as a Rollover Project after: (i) a period of six consecutive full fiscal quarters has elapsed since such Project was first included as a Rollover Project, or (ii) such Project has a value, determined by dividing the Property Operating Income for such Project for the most recent four fiscal quarters by the Applicable Cap Rate, of greater than 50% of book value.

"S&P" means Standard & Poor's Ratings Group, a division of the McGraw Hill Companies and its successors.

"Sanctioned Person" means, at any time, (a) any Person that is, or is 50% or more owned by a Person or Persons that are, listed in any Sanctions related list of designated Persons maintained by any Governmental Authority of the United States of America, including without limitation, OFAC or the U.S. Department of State, or by the United Nations Security Council, Her Majesty's Treasury or the European Union or (b) a country, a region within a country, an agency of the government of a country, an organization controlled by a country or a Person resident in a country, that is subject to or the target of a sanctions program applicable to such country, region, agency, organization or Person and identified on any Sanctions-related list maintained by any Governmental Authority of the United States of America, including without limitation, OFAC or the U.S. Department of State.

"Sanctions" means economic or financial sanctions or trade embargoes imposed, administered or enforced by (a) any Governmental Authority of the United States of America, including without limitation, OFAC or the U.S. Department of State, or (b) the United Nations Security Council, Her Majesty's Treasury or the European Union.

"SOFR" means, with respect to any Business Day, a rate per annum equal to the secured overnight financing rate for such Business Day published by the SOFR Administrator on the SOFR Administrator's Website on the immediately succeeding Business Day.

"SOFR Administrator" means the Federal Reserve Bank of New York (or a successor administrator of the secured overnight financing rate).

"SOFR Administrator's Website" means the website of the Federal Reserve Bank of New York, currently at http://www.newyorkfed.org, or any successor source for the secured overnight financing rate identified as such by the SOFR Administrator from time to time.

"Specified Rate Management Obligations" means all indebtedness, liabilities, obligations, covenants and duties of the Borrower or its Subsidiaries under or in respect of any Specified Rate Management Transaction, whether direct or indirect, absolute or contingent, due or not due, liquidated or unliquidated, and whether or not evidenced by any written confirmation.

"<u>Specified Rate Management Provider</u>" means any Lender, or any Affiliate of a Lender that is a party to a Specified Rate Management Transaction at the time the Specified Rate Management Transaction is entered into.

"Specified Rate Management Transactions" means the Rate Management Transactions occurring (i) on February 26, 2020 between the Borrower and Wells Fargo Bank, National Association, as agreed to pursuant to a trade confirmation dated as of the same date between the Borrower and Wells Fargo Bank, National Association, (ii) on February 26, 2020 between the Borrower and Regions Bank, as agreed to pursuant to a trade confirmation dated as of the same date

between the Borrower and Regions Bank, and (iii) on March 3, 2020 between the Borrower and U.S. Bank National Association, as agreed to pursuant to a trade confirmation dated as of the same date between the Borrower and U.S. Bank National Association. The Administrative Agent acknowledges that it is aware of the above Specified Rate Management Transactions and to the extent this Agreement requires giving notice of any Rate Management transaction, the Administrative Agent shall be deemed to have received such notice of such Specified Rate Management Transactions on the Agreement Execution Date.

"Subsidiary" means as to any Person, a corporation, partnership or other entity of which shares of stock or other ownership interests having ordinary voting power (other than stock or such other ownership interests having such power only by reason of the happening of a contingency) to elect a majority of the board of directors or other managers of such corporation, partnership or other entity are at the time owned, or the management of which is otherwise controlled, directly or indirectly through one or more intermediaries, or both, by such Person, and provided such corporation, partnership or other entity is consolidated with such Person for financial reporting purposes under GAAP.

"Substantially Completed Properties" means, for any Reference Year, Properties of the Borrower and its Subsidiaries which were previously Assets Under Development but (a) shall have been identified by the Borrower as being substantially completed (as determined on a basis consistent with the identification of substantially completed Properties set forth on the Borrower's quarterly "Supplemental Information" report available on the Borrower's website for the fiscal period ending March 31, 2021) in the Borrower's quarterly "Supplemental Information" report made available on the Borrower's website (or any similar report made publicly available on the Borrower's website) for the final quarterly period of such Reference Year, and (b) which (i) have been completed during such Reference Year and (ii) have been included in a "new opening summary" form submitted to, and approved by, the Administrative Agent for each newly-opened Property during such Reference Year. Notwithstanding the foregoing, "new opening summary" forms will only be required for developments to the extent the aggregate investment amount thereof exceeds \$600,000,000 during any fiscal year.

"Substitute Rate" means a floating rate of interest equal to (a) the Federal Funds Effective Rate from time to time <u>plus</u> one and one-half percent (1.50%) plus (b) the Applicable Margin for Adjusted Base Rate Loans. If the Substitute Rate determined as provided above would be less than 1.0%, the Substitute Rate shall be deemed to be 1.0%. Notwithstanding the foregoing, if the Borrower has delivered a written notice to the Administrative Agent certifying (i) that all or any portion of the Loans are subject to a Specified Rate Management Transaction (or any successor or replacement Rate Management Transaction) and (ii) that such Specified Rate Management Transaction (or any successor or replacement Rate Management Transaction) is not subject to a 1.0% interest rate floor, then Substitute Rate shall not be subject to a floor of 1.0% with respect to such Loans.

"Sustainability Pricing Adjustment Date" has the meaning specified in Section 1.7(a).

"Sustainability Margin Adjustment" means with respect to any Pricing Certificate for any period between Sustainability Pricing Adjustment Dates, an amount (whether negative or zero), expressed as a percentage, equal to the KPI Metric Applicable Margin Adjustment Amount.

"<u>Taxes</u>" means all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

"<u>Term Loan Commitment</u>" means with respect to each Lender, the commitment, if any, of such Lender to make a Loan hereunder. The amount of each Lender's Term Loan Commitment as of the Agreement Execution Date is set forth on <u>Exhibit A</u>. The aggregate amount of the Term Loan Commitments is \$200,000,000 as of the Agreement Execution Date.

"<u>Term Loan Exposure</u>" means with respect to any Lender at any time, the sum of the outstanding principal amount of such Lender's Loan at such time.

"<u>Term SOFR</u>" means, for the applicable Corresponding Tenor as of the applicable Reference Time, the forward-looking term rate based on SOFR that has been selected or recommended by the Relevant Governmental Body.

"<u>Term SOFR Notice</u>" means a notification by the Administrative Agent to the Lenders and the Borrower of the occurrence of a Term SOFR Transition Event.

"<u>Term SOFR Transition Event</u>" means the determination by the Administrative Agent that (a) Term SOFR has been recommended for use by the Relevant Governmental Body, (b) the administration of Term SOFR is administratively feasible for the Administrative Agent and (c) a Benchmark Transition Event, an Early Opt-in Election or an Other Benchmark Rate Election, as applicable, has previously occurred resulting in the replacement of the then-current Benchmark for all purposes hereunder and under any Loan Document in accordance with <u>Section 4.8(c)</u> with a Benchmark Replacement the Unadjusted Benchmark Replacement component of which is not Term SOFR.

"Total Term Loan Exposure" means at any time, the sum of the aggregate Term Loan Exposures for each of the Lenders.

"Transferee" is defined in Section 13.7 hereof.

"TRS" means a taxable REIT Subsidiary of the General Partner.

"Type" when used in reference to any Loan or Borrowing, refers to the rate by reference to which interest on such Loan, or on the Loans comprising such Borrowing, is determined. For purposes hereof, "rate" shall include the Adjusted LIBOR Rate and Adjusted Base Rate.

"<u>UK Financial Institution</u>" means any BRRD Undertaking (as such term is defined under the PRA Rulebook (as amended from time to time) promulgated by the United Kingdom Prudential Regulation Authority) or any person falling within IFPRU 11.6 of the FCA Handbook (as amended from time to time) promulgated by the United Kingdom Financial Conduct Authority, which includes

certain credit institutions and investment firms, and certain affiliates of such credit institutions or investment firms.

"<u>UK Resolution Authority</u>" means the Bank of England or any other public administrative authority having responsibility for the resolution of any UK Financial Institution.

"<u>Unadjusted Benchmark Replacement</u>" means the applicable Benchmark Replacement excluding the related Benchmark Replacement Adjustment.

"Unencumbered Asset" means any Project which as of any date of determination, (a) is not subject to any Liens other than Permitted Liens set forth in Sections 9.5(i) through 9.5(y), (b) is not subject to any agreement (including any agreement governing Indebtedness incurred in order to finance or refinance the acquisition of such asset) which prohibits or limits the ability of the Borrower, or its Wholly-Owned Subsidiaries, as the case may be, to create, incur, assume or suffer to exist any Lien upon any assets or Capital Stock of the Borrower, or any of its Wholly-Owned Subsidiaries other than a Permitted Negative Pledge, (c) is not subject to any agreement (including any agreement governing Indebtedness incurred in order to finance or refinance the acquisition of such asset) which entitles any Person to the benefit of any Lien (but not subject to any Liens other than Permitted Liens set forth in Sections 9.5(i) through 9.5(y)) on any assets or Capital Stock of the Borrower or any of its Wholly-Owned Subsidiaries or would entitle any Person to the benefit of any Lien (but excluding the Permitted Liens set forth in Sections 9.5(i) through 9.5(y)) on such assets or Capital Stock upon the occurrence of any contingency (including, except in the case of a Material Credit Facility, pursuant to an "equal and ratable" clause), (d) is not the subject of any material architectural/engineering issue, as evidenced by a certification of the Borrower, and (e) is materially compliant with the representations and warranties in Article VI below. Notwithstanding the foregoing, if any Project is a "Superfund" site under federal law or a site identified in writing by the jurisdiction in which such Project is located as having significant environmental contamination under applicable state law, the Borrower shall so advise the Lenders in writing and the Required Lenders shall have the right to request from the Borrower a current detailed environmental assessment (or one which is not more than two years old for Unencumbered Assets owned as of the Agreement Execution Date), and, if applicable, a written estimate of any remediation costs from a recognized environmental contractor and to exclude any such Project from Unencumbered Assets at their election. No Project of a Wholly-Owned Subsidiary shall be deemed to be unencumbered unless such Project and all Capital Stock of such Wholly-Owned Subsidiary or any other intervening Wholly-Owned Subsidiary between the Borrower and such Wholly-Owned Subsidiary is unencumbered and neither such Wholly-Owned Subsidiary nor any other intervening Wholly-Owned Subsidiary between the Borrower and such Wholly-Owned Subsidiary has any Indebtedness for borrowed money (other than Indebtedness due to the Borrower).

"Unencumbered Leverage Ratio" is defined in Section 9.7(c).

"Unencumbered Leverage Ratio Increase Period" is defined in Section 9.7(c).

"<u>Unimproved Land</u>" means land which constitutes a single tax parcel or separately platted lot and on which construction of an industrial building has not commenced.

"USD LIBOR" means the London interbank offered rate for Dollars.

"Value of Unencumbered Assets" means, for any Person as of any date, the sum (without duplication) of (a) the value of all Unencumbered Assets that are not Assets Under Development, Assets Acquired Not in Service or assets of the type described in clause (b) (determined in the manner set forth below), plus (b) the purchase price paid by such Person (less any amounts paid to such Person as a purchase price adjustment, held in escrow, retained as a contingency reserve or in connection with other similar arrangements) for any Property (other than Assets Under Development) that constitutes an Unencumbered Asset and acquired by such Person during the immediately preceding period of four consecutive fiscal quarters, plus (c) any unrestricted cash, including any cash on deposit with a qualified intermediary with respect to a deferred tax-free exchange, plus (d) an amount equal to 100% of the then-current book value, determined in accordance with GAAP, of each first mortgage receivable secured by an income producing commercial property. provided that such first mortgage receivable is not subject to any Lien, plus (e) 100% of the then current book value of each Asset Under Development that constitutes an Unencumbered Asset plus (f) with respect to each Rollover Project, an amount equal to 50% of the then current book value, determined in accordance with GAAP, of each Rollover Project; provided that to the extent the aggregate amount of Value of Unencumbered Assets from Assets Acquired Not in Service and the other items set forth in clauses (d), (e) and (f) exceed 20% of the total Value of Unencumbered Assets, such excess shall be excluded. Unencumbered Assets that are not Assets Under Development or assets of the type described in clause (b) above shall be valued by dividing the Property Operating Income for such Project for the most recent four fiscal quarters by the Applicable Cap Rate (provided that for the purpose of such calculation, the Property Operating Income of each Unencumbered Asset that was formerly a Rollover Project shall in no event be less than zero). If a Project is no longer owned as of the date of calculation, then no value shall be included based on capitalizing Property Operating Income from such Project, except for purposes of such financial covenant comparing the Property Operating Income from Unencumbered Assets during a quarter to Debt Service for such quarter.

"Wells" means Wells Fargo Bank, National Association.

"<u>Wholly-Owned Subsidiary</u>" means a member of the Consolidated Operating Partnership 100% of the ownership interests in which are owned, directly or indirectly, by the Borrower and the General Partner in the aggregate.

"Write-Down and Conversion Powers" means (a) with respect to any EEA Resolution Authority, the write-down and conversion powers of such EEA Resolution Authority from time to time under the Bail-In Legislation for the applicable EEA Member Country, which write-down and conversion powers are described in the EU Bail-In Legislation Schedule and (b) with respect to the United Kingdom, any powers of the applicable Resolution Authority under the Bail-In Legislation to cancel, reduce, modify or change the form of a liability of any UK Financial Institution or any contract or instrument under which that liability arises, to convert all or part of that liability into shares, securities or obligations of that person or any other person, to provide that any such contract or instrument is to have effect as if a right had been exercised under it or to suspend any obligation in respect of that liability or any of the powers under that Bail-In Legislation that are related to or ancillary to any of those powers.

The foregoing definitions shall be equally applicable to both the singular and the plural forms of the defined terms.

- 1.2. Financial Standards. All financial computations required of a Person under this Agreement shall be made in accordance with GAAP as in effect on the date of this Agreement, and all financial information required under this Agreement shall be prepared in accordance with GAAP, except that if any Person's financial statements are not audited, such Person's financial statements shall be prepared in accordance with the same sound accounting principles utilized in connection with the financial information submitted to Lenders with respect to the Borrower or the General Partner or the Properties in connection with this Agreement and shall be certified by an authorized representative of such Person. Moreover, all financial computations required of a Person under this Agreement shall be calculated (i) without giving effect to any election under Accounting Standards Codification 825-10-25 (or any other Accounting Standards Codification or Financial Accounting Standard having a similar result or effect) to value any Indebtedness or other liabilities of the Borrower, the General Partner or any Wholly Owned Subsidiary at "fair value", as defined therein and (ii) without giving effect to any treatment of Indebtedness in respect of convertible debt instruments under Accounting Standards Codification 470-20 (or any other Accounting Standards Codification or Financial Accounting Standard having a similar result or effect) to value any such Indebtedness in a reduced or bifurcated manner as described therein, and such Indebtedness shall at all times be valued at the full stated principal amount thereof. To the extent the Consolidated Operating Partnership has Defeased Debt, both the underlying debt and interest payable thereon and the financial assets used to defease such debt and interest earned thereon shall be excluded from calculations of the foregoing financial covenants.
- 1.3. <u>Classification of Loans and Borrowings</u>. For purposes of this Agreement, Loans may be classified and referred to by Type (e.g., a "<u>Eurocurrency Loan</u>"). Borrowings also may be classified and referred to by Type (e.g., a "<u>Eurocurrency Borrowing</u>").
- 1.4. General References. References in this Agreement to "Sections", "Articles", "Exhibits" and "Schedules" are to sections, articles, exhibits and schedules herein and hereto unless otherwise indicated. References in this Agreement to any document, instrument or agreement (a) shall include all exhibits, schedules and other attachments thereto, (b) shall include all documents, instruments or agreements issued or executed in replacement thereof, to the extent permitted hereby and (c) shall mean such document, instrument or agreement, or replacement or predecessor thereto, as amended, supplemented, restated or otherwise modified from time to time to the extent not otherwise stated herein or prohibited hereby and in effect at any given time. Wherever from the context it appears appropriate, each term stated in either the singular or plural shall include the singular and plural, and pronouns stated in the masculine, feminine or neuter gender shall include the masculine, the feminine and the neuter. Unless otherwise indicated, all references to time are references to Central Time.
- 1.5. <u>Rates</u>. The interest rate on LIBOR Borrowings and Adjusted Base Rate Borrowings may be determined by reference to LIBOR, which is derived from the London interbank offered rate. The London interbank offered rate is intended to represent the rate at which contributing banks may obtain short-term borrowings from each other in the London interbank market. On March 5, 2021, ICE Benchmark Administration ("<u>IBA</u>"), the administrator of the London interbank offered rate, and the Financial Conduct Authority (the "<u>FCA</u>"), the regulatory supervisor of IBA, announced in public

statements (the "Announcements") that the final publication or representativeness date for the London interbank offered rate for Dollars for: (a) 1-week and 2-month tenor settings will be December 31, 2021 and (b) overnight, 1-month, 3-month, 6-month and 12month tenor settings will be June 30, 2023. No successor administrator for IBA was identified in such Announcements. As a result, it is possible that commencing immediately after such dates, the London interbank offered rate for such tenors may no longer be available or may no longer be deemed a representative reference rate upon which to determine the interest rate on LIBOR Borrowings or Adjusted Base Rate Borrowings. There is no assurance that the dates set forth in the Announcements will not change or that IBA or the FCA will not take further action that could impact the availability, composition or characteristics of any London interbank offered rate. Public and private sector industry initiatives have been and continue, as of the date hereof, to be underway to implement new or alternative reference rates to be used in place of the London interbank offered rate. In the event that the London interbank offered rate or any other then-current Benchmark is no longer available or in certain other circumstances set forth in Section 4.8(c), such Section 4.8(c) provides a mechanism for determining an alternative rate of interest. The Administrative Agent will notify the Borrower, pursuant to Section 4.8(c), of any change to the reference rate upon which the interest rate on LIBOR Borrowings and Adjusted Base Rate Borrowings is based. However, the Administrative Agent does not warrant or accept any responsibility for, and shall not have any liability with respect to, (i) the continuation of, administration of, submission of, calculation of or any other matter related to the London interbank offered rate or other rates in the definition of "LIBOR" or with respect to any alternative, successor or replacement rate thereto (including any then-current Benchmark or any Benchmark Replacement), including whether the composition or characteristics of any such alternative, successor or replacement rate (including any Benchmark Replacement), as it may or may not be adjusted pursuant to Section 4.8(c), will be similar to, or produce the same value or economic equivalence of, LIBOR or any other Benchmark, or have the same volume or liquidity as did the London interbank offered rate or any other Benchmark prior to its discontinuance or unavailability, or (ii) the effect, implementation or composition of any Benchmark Replacement Conforming Changes. The Administrative Agent and its Affiliates or other related entities may engage in transactions that affect the calculation of a Benchmark, any alternative, successor or replacement rate (including any Benchmark Replacement) or any relevant adjustments thereto and such transactions may be adverse to the Borrower. The Administrative Agent may select information sources or services in its reasonable discretion to ascertain any Benchmark, any component definition thereof or rates referenced in the definition thereof, in each case pursuant to the terms of this Agreement, and shall have no liability to the Borrower, any Lender or any other person or entity for damages of any kind, including direct or indirect, special, punitive, incidental or consequential damages, costs, losses or expenses (whether in tort, contract or otherwise and whether at law or in equity), for any error or calculation of any such rate (or component thereof) provided by any such information source or service.

1.6. <u>Divisions</u>. For all purposes under the Loan Documents, in connection with any division or plan of division under Delaware law (or any comparable event under a different jurisdiction's laws): (a) if any asset, right, obligation or liability of any Person becomes the asset, right, obligation or liability of a different Person, then it shall be deemed to have been transferred from the original Person to the subsequent Person, and (b) if any new Person comes into existence, such new Person shall be deemed to have been organized on the first date of its existence by the holders of its Equity Interests at such time.

1.7. Sustainability Adjustments.

- (a) Following the date on which Borrower provides a Pricing Certificate in respect of the most recently ended Reference Year, the Applicable Margin shall be adjusted, as applicable, pursuant to the Sustainability Margin Adjustment as set forth in such Pricing Certificate. For purposes of the foregoing, (i) the Sustainability Margin Adjustment shall be determined as of the fifth Business Day following receipt by the Administrative Agent of a Pricing Certificate delivered pursuant to Section 1.7(f) based upon the KPI Metric for the applicable Reference Year set forth in such Pricing Certificate and the calculations of the Sustainability Margin Adjustment, therein (such day, the "Sustainability Pricing Adjustment Date") and (ii) each change in the Applicable Margin resulting from a Pricing Certificate (or the non-delivery or delivery of an incomplete Pricing Certificate) shall be effective during the period commencing on and including the applicable Sustainability Pricing Adjustment Date and ending on the date immediately preceding the next such Sustainability Pricing Adjustment Date (or, in the case of non-delivery of a Pricing Certificate, the last day such Pricing Certificate could have been delivered pursuant to the terms of Section 1.7(f)).
- (b) For the avoidance of doubt, only one Pricing Certificate may be delivered in respect of any Reference Year. It is further understood and agreed that the LIBOR Applicable Margin and the Base Rate Applicable Margin will never be reduced by more than 0.01% pursuant to the Sustainability Margin Adjustment during any calendar year. For the avoidance of doubt, any adjustment to the Applicable Margin by reference to the KPI Metric Target in any Reference Year shall not be cumulative year-over-year. Each applicable adjustment shall only apply until the date on which the next adjustment is due to take place.
- (c) It is hereby understood and agreed that if no such Pricing Certificate is delivered, or any Pricing Certificate shall be incomplete and fail to include the KPI Metric for the applicable Reference Year, within the period set forth in Section 1.7(f), the Sustainability Margin Adjustment will be made to the Applicable Margin commencing on the last day such Pricing Certificate could have been delivered pursuant to the terms of Section 1.7(f).
- (d) If (i)(A) the Borrower or any Lender becomes aware of any material inaccuracy in the Sustainability Margin Adjustment, the KPI Metric as reported in a Pricing Certificate (any such material inaccuracy, a "Pricing Certificate Inaccuracy.") and, in the case of any Lender, such Lender delivers, not later than ten (10) Business Days after obtaining knowledge thereof, a written notice to the Administrative Agent describing such Pricing Certificate Inaccuracy in reasonable detail (which description shall be shared with each Lender and the Borrower), or (B) the Borrower and the Lenders agree that there was a Pricing Certificate Inaccuracy at the time of delivery of a Pricing Certificate, and (ii) a proper calculation of the Sustainability Margin Adjustment, the KPI Metric would have resulted in no adjustment to the Applicable Margin for any period, the Borrower shall be obligated to pay to the Administrative Agent for the account of the applicable Lenders, as the case may be, promptly on demand by the Administrative Agent (or, after the occurrence of an actual or deemed entry of an order for relief with respect to the Borrower under any Debtor Relief Laws, automatically and without further action by the Administrative Agent, any Lender), but in any event within ten (10) Business Days after the Borrower has received written notice of, or has agreed in writing that there was, a Pricing Certificate Inaccuracy, an amount equal to the excess of (1) the amount of interest and fees that should have been paid for such period over (2) the amount of interest and fees actually paid for such period.

It is understood and agreed that any Pricing Certificate Inaccuracy shall not constitute a Default or Event of Default or otherwise result in the failure of any condition precedent to any advance or the issuance of any Letter of Credit; provided, that, the Borrower complies with the terms of this Section 1.7(d) with respect to such Pricing Certificate Inaccuracy. Notwithstanding anything to the contrary herein, unless such amounts shall be due upon the occurrence of an actual or deemed entry of an order for relief with respect to the Borrower under any Debtor Relief Laws, (a) any additional amounts required to be paid pursuant to the immediate preceding paragraph shall not be due and payable until a written demand is made for such payment by the Administrative Agent in accordance with such paragraph, (b) any nonpayment of such additional amounts prior to or concurrently with such demand for payment by the Administrative Agent shall not constitute a Default (whether retroactively or otherwise) and (c) none of such additional amounts shall be deemed overdue prior to such a demand or shall accrue interest at the default rate pursuant to Section 2.14 prior to such a demand.

- (e) Each party hereto hereby agrees that the Administrative Agent shall have no responsibility for (or liability in respect of) reviewing, auditing or otherwise evaluating any calculation by the Borrower of any Sustainability Margin Adjustment (or any of the data or computations that are part of or related to any such calculation) set forth in any Pricing Certificate (and the Administrative Agent may rely conclusively on any such certificate, without further inquiry).
- (f) As soon as available and in any event within 90 days following the end of each calendar year (commencing with the calendar year ending December 31, 2021), the Borrower may deliver a Pricing Certificate to the Administrative Agent (and the Administrative Agent shall promptly provide a copy to each Lender) for the most recently-ended Reference Year; provided, that, for any Reference Year the Borrower may elect not to deliver a Pricing Certificate, and such election shall not constitute a Default or Event of Default (but such failure to so deliver a Pricing Certificate by the end of such 90-day period shall result in the Sustainability Margin Adjustment being applied as set forth in Section 1.7(c)). Subject to the provisions of Section 1.7(b), the Borrower may, at its election exercisable by delivering written notice to the Administrative Agent, adjust the Reference Year and the timing of delivery of the Pricing Certificate in a manner intended to maintain consistency with the foregoing.
- 1.8. Amendment and Restatement of the Existing Term Loan Agreement. The parties to this Agreement agree that, upon (i) the execution and delivery by each of the parties hereto of this Agreement and (ii) satisfaction of the conditions set forth in Section 5.1, the terms and provisions of the Existing Term Loan Agreement shall be and hereby are amended, superseded and restated in their entirety by the terms and provisions of this Agreement. This Agreement is not intended to and shall not constitute a novation. All "Loans" made and "Obligations" incurred under the Existing Term Loan Agreement which are outstanding on the Agreement Execution Date shall continue as Obligations under (and shall be governed by the terms of) this Agreement and the other Loan Documents. Without limiting the foregoing, upon the effectiveness hereof: (a) all references in the "Loan Documents" (as defined in the Existing Term Loan Agreement) to the "Administrative Agent", the "Loan Agreement" and the "Loan Documents" shall be deemed to refer to the Administrative Agent, this Agreement and the Loan Documents, (b) all obligations constituting "Obligations" with any Lender or any Affiliate of any Lender which are outstanding on the Agreement Execution Date shall continue as Obligations under this Agreement and the other Loan

Documents, (c) the Administrative Agent shall make such reallocations, sales, assignments or other relevant actions in respect of each Lender's credit and loan exposure under the Existing Term Loan Agreement as are necessary in order that each such Lender's outstanding Loans hereunder reflect such Lender's pro rata share of the outstanding aggregate Loans on the Agreement Execution Date, (d) the Borrower hereby agrees to compensate each Lender for any and all losses, costs and expenses incurred by such Lender in connection with the sale and assignment of any Eurocurrency Loans (including the "Eurocurrency Loans" under the Existing Term Loan Agreement) and such reallocation described above, in each case on the terms and in the manner set forth in Section 4.4 hereof.

ARTICLE II.

THE FACILITY

2.1. The Facility.

- (a) Subject to the terms and conditions of this Agreement and in reliance upon the representations and warranties of the Borrower and the General Partner contained herein, each Lender agrees, severally and not jointly, to make Loans in Dollars to the Borrower on the Closing Date in an aggregate principal amount that will not result in (x) such Lender's Term Loan Exposure exceeding such Lender's Term Loan Commitment or (y) the Total Term Loan Exposure exceeding the Aggregate Term Loan Commitment. The Borrowings may be Adjusted Base Rate Borrowings or Eurocurrency Borrowings.
- (b) The Facility created by this Agreement shall terminate on the Maturity Date, unless sooner terminated in accordance with the terms of this Agreement. The Term Loan Commitment of each Lender shall terminate upon the earlier of (i) the funding of such Lender's Loan hereunder on the Closing Date and (ii) 5:00 p.m. (Central Time) on the Closing Date. Amounts repaid or prepaid in respect of the Loans may not be reborrowed.
- 2.2. <u>Incremental Loans</u>. The Borrower shall have the right to request one or more additional tranches of additional Loans (the "<u>Incremental Loans</u>") by providing written notice to the Administrative Agent, which notice shall be irrevocable once given; provided, however, that the aggregate amount of such Incremental Loans shall not exceed Two Hundred Sixty Million Dollars (\$260,000,000). Each such Incremental Loan must be an aggregate minimum amount of \$20,000,000 and integral multiples of \$5,000,000 in excess thereof. The Administrative Agent and the Arrangers, in consultation with the Borrower, shall manage all aspects of such Incremental Loans, including decisions as to the selection of the existing Lenders and/or other banks, financial institutions and other institutional lenders to be approached with respect to such increase and the allocations of Incremental Loans among such existing Lenders and/or other banks, financial institutions and other institutional lenders. No Lender shall be obligated in any way whatsoever to provide an Incremental Loan, and any new Lender becoming a party to this Agreement in connection with any such requested Incremental Loan must be an Eligible Assignee. Effecting the Incremental Loans under this Section is subject to the following conditions precedent: (w) the Administrative Agent shall have consented thereto (such consent not to be unreasonably withheld or delayed), (x) no Default or Event of Default shall be in existence on the effective date of such Incremental Loans,

- (y) the representations and warranties made or deemed made by the Borrower or any other Loan Party in any Loan Document to which such Loan Party is a party shall be true and correct on the effective date of such Incremental Loans except to the extent that such representations and warranties expressly relate solely to an earlier date (in which case such representations and warranties shall have been true and correct on and as of such earlier date) and except for changes in factual circumstances specifically and expressly permitted hereunder, and (z) the Administrative Agent shall have received each of the following, in form and substance satisfactory to the Administrative Agent; (i) if not previously delivered to the Administrative Agent, copies certified by a Qualified Officer of the Borrower of (A) all partnership or other necessary action taken by the Borrower to authorize such Incremental Loans and (B) all corporate or other necessary action taken by Guarantor authorizing the guaranty of such Incremental Loans; and (ii) an opinion of counsel to the Borrower and the Guarantor, and addressed to the Administrative Agent and the Lenders covering such matters as reasonably requested by the Administrative Agent; and (iii) if requested, new Notes executed by the Borrower, payable to any new Lenders and replacement Notes, if requested by any applicable Lender, executed by the Borrower, payable to any existing Lenders making Incremental Loans, in the amount of each such Lender's Term Loan Exposure at the time of the effectiveness of the applicable Incremental Loans. In connection with any Incremental Loans pursuant to this Section 2.2 any Lender becoming a party hereto shall execute such documents and agreements as the Administrative Agent may reasonably request. The Incremental Loans (i) shall rank pari passu in right of payment with the existing Loans, (ii) shall not mature earlier than the Maturity Date and (iii) shall be treated substantially the same as (and in any event no more favorably than) the existing Loans. Incremental Loans may be made hereunder pursuant to an amendment or restatement of this Agreement and, as appropriate, the other Loan Documents, executed by the Borrower, each new Lender participating in such tranche (if any), each existing Lender participating in such tranche (if any) and the Administrative Agent. The Administrative Agent may, without the consent of any other Lenders (except as expressly required pursuant to Section 14.3), effect such amendments to this Agreement and the other Loan Documents as may be necessary or appropriate, in the reasonable opinion of the Administrative Agent, to effect the provisions of this Section 2.2.
- 2.3. <u>Principal Payments</u>. Any outstanding Borrowings and all other unpaid Obligations shall be paid in full by the Borrower on the Maturity Date.
- 2.4. <u>Requests for Borrowings; Responsibility for Borrowings</u>. Borrowings shall be made available to the Borrower by Administrative Agent in accordance with <u>Section 2.1(a)</u> and <u>Section 2.7</u> hereof. The obligation of each Lender to fund its Percentage of each Borrowing shall be several and not joint.

2.5. Evidence of Credit Extensions. Any Lender may request that Loans made by it be evidenced by a Note. In such event, the Borrower shall prepare, execute and deliver to such Lender a Note payable to the order of such Lender (or, if requested by such Lender, to such Lender and its registered assigns). Thereafter, the Loans evidenced by such Note and interest thereon shall at all times (including after assignment pursuant to Section 13.3) be represented by one or more Notes in such form payable to the order of the payee named therein (or, if such Note is a registered note, to such payee and its registered assigns). Each Lender may record Borrowings and principal payments thereof on the schedule attached to its Note or, at its option, in its records, and each Lender's record thereof shall be conclusive absent the Borrower furnishing to such Lender conclusive and irrefutable evidence of an error made by such Lender with respect to that Lender's records. Notwithstanding the foregoing, the failure to make, or an error in making, a notation with respect to any Borrowing shall not limit or otherwise affect the obligations of the Borrower hereunder or under the Notes to pay the amount actually owed by the Borrower to Lenders.

2.6. Loans and Borrowings.

- (a) Each Loan shall be made as part of a Borrowing consisting of Loans of the same Type made by the Lenders ratably in accordance with their respective Percentages. The failure of any Lender to make any Loan required to be made by it shall not relieve any other Lender of its obligations hereunder.
- (b) Subject to Section 4.8, each Borrowing shall be comprised entirely of Adjusted Base Rate Loans or Eurocurrency Loans as the Borrower may request in accordance herewith. Each Lender at its option may make any Eurocurrency Loan by causing any domestic or foreign branch or Affiliate of such Lender to make such Loan; provided that any exercise of such option shall not affect the obligation of the Borrower to repay such Loan in accordance with the terms of this Agreement.
 - (c) No more than ten (10) Eurocurrency Borrowings may be outstanding at any one time under the Facility.
- (d) Notwithstanding any other provision of this Agreement, the Borrower shall not be entitled to request, or to elect to convert or continue, any Borrowing if the Interest Period requested with respect thereto would end after the Maturity Date.

2.7. Requests for Borrowings.

(a) To request a Borrowing, the Borrower shall notify the Administrative Agent of such request by telephone (a "Borrowing Notice") (A) in the case of a Eurocurrency Borrowing, not later than 11:00 a.m. (Central Time), three (3) Business Days before the date of the proposed Borrowing, or (B) in the case of an Adjusted Base Rate Borrowing, not later than 11:00 a.m. (Central Time), one (1) Business Day before the date of the proposed Borrowing. Each such telephonic Borrowing Notice shall be irrevocable and, if requested by the Administrative Agent, shall be confirmed promptly by delivery to the Administrative Agent of a written Borrowing Notice in substantially the form of Exhibit J in the case of any Borrowing. Each such telephonic and written Borrowing Notice shall specify the following information in compliance with Section 2.7: (i) the Borrower requesting such Borrowing; (ii) the Type of the requested Borrowing; (iii) the aggregate amount of such Borrowing;

(iv) the date of such Borrowing, which shall be a Business Day; (v) in the case of a Eurocurrency Borrowing, the initial Interest Period to be applicable thereto; and (vi) the location and number of the Borrower's account to which funds are to be disbursed, which shall comply with the requirements of Section 2.7. If no election as to the Type of Borrowing is specified, then the requested Borrowing shall be an Adjusted Base Rate Borrowing. If no Interest Period is specified with respect to any requested Eurocurrency Borrowing, then the Borrower shall be deemed to have selected an Interest Period of one month's duration. Promptly following receipt of a Borrowing Notice in accordance with this Section, the Administrative Agent shall advise each relevant Lender of the details thereof and of the amount of such Lender's Loan to be made as part of the requested Borrowing.

The Borrower shall also deliver together with each Borrowing Notice the compliance certificate required in <u>Section 5.2</u> and otherwise comply with the conditions set forth in Section 5.2 for Borrowings.

Not later than 11:00 a.m. (Central Time) on each Borrowing Date, each Lender shall make available its Loan or Loans, in funds immediately available at the Administrative Office. Administrative Agent will promptly make the funds so received from the Lenders available to the Borrower in the account specified by the Borrower in the Disbursement Instruction Agreement.

- (b) Administrative Agent shall, as soon as practicable after receipt of a Borrowing Notice, determine the Adjusted LIBOR Rate applicable to the requested Eurocurrency Borrowing and inform the Borrower and Lenders of the same. Each determination of the Adjusted LIBOR Rate by Administrative Agent shall be conclusive and binding upon the Borrower in the absence of manifest error.
- (c) If the Borrower shall prepay a Eurocurrency Borrowing other than on the last day of the Interest Period applicable thereto, the Borrower shall be responsible to pay all amounts due to Lenders as required by <u>Section 4.4</u> hereof. The Lenders shall not be obligated to match fund their Eurocurrency Borrowings.
- (d) The right of the Borrower to select the Adjusted LIBOR Rate and the Adjusted Base Rate for a Borrowing pursuant to this Agreement is subject to the availability to Lenders, as applicable, of a similar option. If Administrative Agent determines that (i) deposits of Dollars in an amount approximately equal to the Borrowing for which the Borrower wishes to select the Adjusted LIBOR Rate or Adjusted Base Rate are not generally available at such time in the London interbank eurodollar market, or (ii) the rate at which the deposits described in subsection (i) herein are being offered will not adequately and fairly reflect the costs to Lenders, as applicable, of maintaining an Adjusted LIBOR Rate or Adjusted Base Rate on a Borrowing or of funding the same in such market for such Interest Period, or (iii) reasonable means do not exist for determining an Adjusted LIBOR Rate or Adjusted Base Rate would be in excess of the maximum interest rate which the Borrower may by law pay, then in any of such events, Administrative Agent shall so notify the Borrower and Lenders and notwithstanding anything herein to the contrary, such Borrowing shall bear interest at the Substitute Rate.

2.8. <u>Interest Elections</u>.

- (a) Each Borrowing initially shall be of the Type specified in the applicable Borrowing Request and, in the case of a Eurocurrency Borrowing, shall have an initial Interest Period as specified in such Borrowing Request (or as set forth in Section 2.7 if no Interest Period is specified). Thereafter, the Borrower may elect to convert such Borrowing to a different Type or to continue such Borrowing and, in the case of a Eurocurrency Borrowing, may elect Interest Periods therefor, all as provided in this Section. The Borrower may elect different options with respect to different portions of the affected Borrowing, in which case each such portion shall be allocated ratably among the Lenders holding the Loans comprising such Borrowing, and the Loans comprising each such portion shall be considered a separate Borrowing. Notwithstanding the foregoing, the Borrower may not (i) elect an Interest Period for Eurocurrency Loans that does not comply with Section 2.6(d), (ii) elect to convert any Adjusted Base Rate Loans to Eurocurrency Loans that would result in the number of Eurocurrency Borrowings exceeding the maximum number of Eurocurrency Borrowings permitted under Section 2.6(c) or (iii) elect an Interest Period for Eurocurrency Loans unless the aggregate outstanding principal amount of Eurocurrency Loans to which such Interest Period will apply complies with the requirements as to minimum principal amount set forth in Section 2.11.
- (b) To make an election pursuant to this Section (an "<u>Interest Election Request</u>"), the Borrower shall notify the Administrative Agent of such election by telephone by the time that a Borrowing Request would be required under <u>Section 2.7</u> if the Borrower were requesting a Borrowing of the Type resulting from such election to be made on the effective date of such election. Each such telephonic Interest Election Request shall be irrevocable and shall be confirmed promptly by delivery to the Administrative Agent of a written Interest Election Request in a form approved by the Administrative Agent (hereinafter referred to as a "<u>Conversion/Continuation Notice</u>").
- (c) Each telephonic and written Conversion/Continuation Notice shall specify the following information in compliance with Section 2.6 and paragraph (a) of this Section: (i) the Borrowing to which such Conversion/Continuation Notice applies; (ii) the effective date of the election made pursuant to such Conversion/Continuation Notice, which shall be a Business Day; (iii) whether the resulting Borrowing is to be an Adjusted Base Rate Borrowing or a Eurocurrency Borrowing; and (iv) if the resulting Borrowing is a Eurocurrency Borrowing, the Interest Period to be applicable thereto after giving effect to such election. If any such Conversion/Continuation Notice requests a Eurocurrency Borrowing but does not specify an Interest Period, then the Borrower shall be deemed to have selected an Interest Period of one month's duration.
- (d) Promptly following receipt of a Conversion/Continuation Notice, the Administrative Agent shall advise each relevant Lender of the details thereof and of such Lender's portion of each resulting Borrowing.
- (e) If the Borrower fails to timely deliver a Conversion/Continuation Notice with respect to a Eurocurrency Borrowing in accordance with <u>Section 2.8(b)</u> prior to the end of the Interest Period applicable thereto, then, unless such Borrowing is repaid as provided herein,

at the end of such Interest Period such Borrowing shall be continued as a Eurocurrency Borrowing with the same Interest Period as the Eurocurrency Borrowing being continued (unless such deemed election shall result in an Interest Period ending after the Maturity Date, in which case the Borrower shall be deemed to have elected a one (1) month Interest Period which will not end after the Maturity Date; provided however that if less than 30 days remain until the Maturity Date, such Eurocurrency Borrowing shall bear interest at the Adjusted Base Rate). The Administrative Agent shall advise each relevant Lender and the Borrower of the details of any such deemed continuation under this clause (e).

(f) Notwithstanding anything to the contrary contained in this <u>Section 2.8</u>, no Borrowing may be converted into a Eurocurrency Borrowing or continued as a Eurocurrency Borrowing (except with the consent of the Required Lenders) when any Monetary Default or Event of Default has occurred and is continuing.

2.9. <u>Applicable Margins and Fees</u>.

- (a) [Reserved].
- (b) The Base Rate Applicable Margin and the LIBOR Applicable Margin shall vary from time to time in accordance with the Investment Grade Rating as follows (such that the Applicable Margin shall change from time to time as and when the Investment Grade Rating changes, which changes shall be effective from and after the date that notice is delivered from the Borrower to the Administrative Agent of the applicable Investment Grade Rating change):

Rating Level	LIBOR Applicable Margin	Base Rate Applicable Margin
A/A2 (" <u>Level I</u> ")	0.75%	0.75%
A-/A3 (" <u>Level II</u> ")	0.80%	0.80%
BBB+/Baa1 (" <u>Level III</u> ")	0.85%	0.85%
BBB/Baa2 (" <u>Level IV</u> ")	0.95%	0.95%
BBB-/Baa3 (" <u>Level V</u> ")	1.20%	1.20%
Below BBB- and Baa3 (" <u>Level VI</u> ")	1.60%	1.60%

Notwithstanding the Investment Grade Rating set forth in above table, if: (i) the Consolidated Leverage Ratio as of the last day of the most recently ending fiscal quarter of the Borrower as set forth in the corresponding compliance certificate delivered pursuant to Section 8.2(iv) is less than 32.5% (or, so long as the Consolidated Leverage Ratio shall have been less than 32.5% during the immediately preceding fiscal quarter, greater than or equal to 32.5% but less than 37.5%; provided that the provisions of this parenthetical shall be applicable for only one fiscal quarter during the term of this Agreement), and (ii) (a) the Borrower's Investment Grade Rating with S&P is not lower than BBB and (b) the Borrower's Investment Grade

Rating with Moody's is not lower than Baa2, the Base Rate Applicable Margin, the LIBOR Applicable Margin shall be set at Level III. For the avoidance of doubt, (x) if the Consolidated Leverage Ratio is greater than or equal to 32.5% for two or more consecutive fiscal quarters, the Borrower will again satisfy the condition set forth in clause (i) above when and if the Consolidated Leverage Ratio as of the last day of the most recently ending fiscal quarter of the Borrower as set forth in the corresponding compliance certificate delivered pursuant to Section 8.2(iv) is less than 32.5%, and (y) the Borrower may qualify for pricing at Level I, II or III based solely upon its Investment Grade Rating even if it does not meet the Consolidated Leverage Ratio condition described in clause (i) above. Any increase or decrease in the Base Rate Applicable Margin, the LIBOR Applicable Margin resulting from a change in the Consolidated Leverage Ratio in accordance with the foregoing proviso shall become effective as of the first Business Day immediately following the date a compliance certificate is delivered pursuant to Section 8.2(iv); provided, that if a compliance certificate is not delivered when due in accordance with such Section, then the Level corresponding to the Investment Grade Rating then in effect shall apply as of the fifth Business Day after the date on which such compliance certificate was required to have been delivered and shall remain in effect until the date on which such compliance certificate is delivered. The parties acknowledge and agree the Consolidated Leverage Ratio for the first fiscal quarter of 2021 is less than 32.5%.

The Applicable Margin shall be determined by the higher of the two ratings from S&P or Moody's. In the event that such two ratings are more than one rating level apart and both are Investment Grade Ratings, then the rating level one level above the lower of the two ratings shall apply. If only one Investment Grade Ratings has been issued, the Applicable Margin shall be determined based on the sole Investment Grade Rating then in effect. If Investment Grade Ratings shall have been assigned by both rating agencies and thereafter the Borrower does not have an Investment Grade Rating from either Rating Agency, the Applicable Margin shall be determined based on Level VI of the foregoing table in this Section 2.9(b).

(c) [Reserved].

- 2.10. <u>Funds Transfer Disbursements</u>. The Borrower hereby authorizes the Administrative Agent to disburse the proceeds of any Loan made by the Lenders or any of their Affiliates pursuant to the Loan Documents as requested by an authorized representative of the Borrower to any of the accounts designated in the Disbursement Instruction Agreement.
- 2.11. <u>Minimum Amount of Each Borrowing</u>. Each Eurocurrency Borrowing shall be in the minimum amount of \$2,000,000 (and in multiples of \$100,000 if in excess thereof), and each Adjusted Base Rate Borrowing shall be in the minimum amount of \$1,000,000 (and in multiples of \$100,000 if in excess thereof).

2.12. Interest.

- (a) Subject to <u>Section 2.14</u>, the outstanding principal balance of the Loans shall bear interest from time to time at a rate per annum equal to:
 - (i) the Adjusted Base Rate; or
 - (ii) at the election of the Borrower in accordance with <u>Section 2.8</u> with respect to all or portions of the Obligations, the Adjusted LIBOR Rate.
- (b) All interest shall be calculated for actual days elapsed on the basis of a 360-day year. Interest accrued on each Borrowing shall be payable on the first day of each calendar month in arrears from time to time while such Borrowing is outstanding (or, if earlier, on the Maturity Date). Interest shall not be payable for the day of any payment on the amount paid if payment is received by Administrative Agent prior to 1:00 p.m. (Central Time). If any payment of principal or interest on the Loans, or fees hereunder or under any other Loan Document, shall become due on a day that is not a Business Day, such payment shall be made on the next succeeding Business Day and, in the case of a payment of principal, such extension of time shall be included in computing interest due in connection with such payment; provided that for purposes of Section 10.1 hereof, any payments of principal described in this sentence shall be considered to be "due" on such next succeeding Business Day.

2.13. Method of Payment.

(a) All payments of the Obligations hereunder shall be made, without set off, deduction, or counterclaim, in immediately available funds to Administrative Agent at the Administrative Office or at any other Lending Office of Administrative Agent specified in writing by Administrative Agent to the Borrower, by 11:00 a.m. (Central Time) on the date when due and shall be applied ratably by Administrative Agent among Lenders in the Facility. Each such payment shall be made in Dollars. Each payment delivered to Administrative Agent for the account of any Lender shall be delivered promptly by Administrative Agent to such Lender in the same type of funds that Administrative Agent received at its address specified herein or at any Lending Office specified in a notice received by Administrative Agent from such Lender. Payments not made by Administrative Agent within one (1) Business Day after receipt shall accrue interest at Federal Funds Effective Rate. Administrative Agent is hereby authorized to charge the account of the Borrower

maintained with Wells for each payment of principal, interest and fees as it becomes due hereunder.

- (b) If any Lender shall fail to make any payment required to be made by it pursuant to Sections 2.7, 2.16 or 12.8, then the Administrative Agent shall, notwithstanding any contrary provision hereof, (i) apply any amounts thereafter received by the Administrative Agent for the account of such Lender for the benefit of the Administrative Agent to satisfy such Lender's obligations to it under such Section until all such unsatisfied obligations are fully paid, and/or (ii) hold any such amounts in a segregated account as cash collateral for, and application to, any future funding obligations of such Lender under any such Section, in the case of each of clauses (i) and (ii) above, in any order as determined by the Administrative Agent in its discretion.
- 2.14. <u>Default</u>. Notwithstanding the foregoing, during the continuance of a Monetary Default or an Event of Default, the Borrower shall not have the right to request a Eurocurrency Borrowing, select a new Interest Period for an existing Eurocurrency Borrowing or convert any Adjusted Base Rate Borrowing to a Eurocurrency Borrowing. During the continuance of a Monetary Default or an Event of Default, at the election of the Required Lenders, by notice to the Borrower, outstanding Borrowings shall bear interest at the applicable Default Rates until such Monetary Default or Event of Default ceases to exist or the Obligations are paid in full.
- 2.15. <u>Lending Offices</u>. Each Lender may book its Borrowings at any Lending Office selected by such Lender and may change its Lending Office from time to time. All terms of this Agreement shall apply to any such Lending Office and any Note shall be deemed held by the applicable Lender for the benefit of such Lending Office. Each Lender may, by written or telex notice to the Administrative Agent and the Borrower, designate a Lending Office through which Borrowings will be made by it and for whose account payments are to be made.
- 2.16. Non Receipt of Funds by Administrative Agent. Unless the Borrower or a Lender, as the case may be, notifies Administrative Agent prior to the date on which it is scheduled to make payment to Administrative Agent of (i) in the case of a Lender, a Borrowing, or (ii) in the case of the Borrower, a payment of principal, interest or fees to the Administrative Agent for the account of the Lenders, that it does not intend to make such payment, Administrative Agent may assume that such payment has been made. Administrative Agent may, but shall not be obligated to, make the amount of such payment available to the intended recipient in reliance upon such assumption. If such Lender or the Borrower, as the case may be, has not in fact made such payment to Administrative Agent, the recipient of such payment shall, promptly after demand by Administrative Agent, repay to Administrative Agent the amount so made available together with interest thereon in respect of each day during the period commencing on the date such amount was so made available by Administrative Agent until the date Administrative Agent recovers such amount at a rate per annum equal to (i) in the case of payment by a Lender, the Federal Funds Effective Rate (as determined by Administrative Agent) or (ii) in the case of payment by the Borrower, the interest rate applicable to Adjusted Base Rate Borrowings. If the Borrower and any Lender shall each pay any such amount made available by the Administrative Agent on behalf of such Lender for the same or overlapping period, the Administrative Agent shall promptly remit to the Borrower the amount paid by the Borrower for such period. Any payment by the Borrower shall be without prejudice to any claim the Borrower may

have against a Lender that shall have failed to make available the proceeds of a Loan to be made by such Lender.

- 2.17. [<u>Reserved</u>].
- 2.18. [<u>Reserved</u>].
- 2.19. [<u>Reserved</u>].
- 2.20. Pro Rata Treatment/Sharing of Payments.
- (a) Except to the extent otherwise provided herein: (i) each Borrowing from the Lenders under <u>Sections 2.7</u> shall be made from the Lenders pro rata according to the amounts of their respective Percentages; (ii) each payment or prepayment of principal of Loans shall be made for the account of the Lenders pro rata in accordance with the respective unpaid principal amounts of the Loans held by them, provided that, subject to <u>Section 12.16</u>, if immediately prior to giving effect to any such payment in respect of any Loans the outstanding principal amount of the Loans shall not be held by the Lenders pro rata in accordance with their respective Percentages in effect at the time such Loans were made, then such payment shall be applied to the Loans in such manner as shall result, as nearly as is practicable, in the outstanding principal amount of the Loans being held by the Lenders pro rata in accordance with their respective Percentages; (iii) each payment of interest on Loans shall be made for the account of the Lenders, pro rata in accordance with the amounts of interest on such Loans, then due and payable to the respective Lenders and (iv) the making, conversion and continuation of Loans of a particular Type shall be made pro rata among the Lenders, according to the amounts of their respective Loans, and the then current Interest Period for each Lender's portion of each such Loan of such Type shall be coterminous.
- (b) If a Lender shall obtain payment of any principal of, or interest on, any Loan made by it to the Borrower under this Agreement or shall obtain payment on any other Obligation owing by the Borrower or any other Loan Party through the exercise of any right of set-off, banker's lien, counterclaim or similar right or otherwise or through voluntary prepayments directly to a Lender or other payments made by or on behalf the Borrower or any other Loan Party to a Lender (other than any payment in respect of Specified Rate Management Obligations) not in accordance with the terms of this Agreement and such payment should be distributed to the Lenders in accordance with Section 2.20(a) or Section 2.22, as applicable, such Lender shall promptly purchase from the other Lenders participations in (or, if and to the extent specified by such Lender, direct interests in) the Loans made by the other Lenders or other Obligations owed to such other Lenders in such amounts, and make such other adjustments from time to time as shall be equitable, to the end that all the Lenders shall share the benefit of such payment (net of any reasonable expenses which may actually be incurred by such Lender in obtaining or preserving such benefit) in accordance with the requirements of Section 2.20(a) or Section 2.22, as applicable. To such end, all the Lenders shall make appropriate adjustments among themselves (by the resale of participations sold or otherwise) if such payment is rescinded or must otherwise be restored. The Borrower agrees that any Lender so purchasing a participation (or direct interest) in the Loans or other Obligations owed to such other Lenders may exercise all rights of set-off,

banker's lien, counterclaim or similar rights with respect to such participation as fully as if such Lender were a direct holder of Loans in the amount of such participation. Nothing contained herein shall require any Lender to exercise any such right or shall affect the right of any Lender to exercise and retain the benefits of exercising, any such right with respect to any other indebtedness or obligation of the Borrower.

2.21. [<u>Reserved</u>].

- 2.22. <u>Application of Moneys Received</u>. Subject to <u>Section 2.13(b)</u> hereof, all moneys collected or received by the Administrative Agent on account of the Facility directly or indirectly, shall be applied in the following order of priority:
 - (i) to the payment of all reasonable costs incurred in the collection of such moneys of which the Administrative Agent shall have given notice to the Borrower;
 - (ii) to the reimbursement of any yield protection due to any of the Lenders in accordance with Section 4.1;
 - (iii) to the payment of all fees then due to the Administrative Agent;
 - (iv) [reserved];
 - (v) first to interest until paid in full and then to principal for all Lenders (x) for Borrowings (unless an Event of Default exists) (the amount allocated to Borrowings to be distributed in accordance with the applicable pro rata shares of the outstanding amounts of the Lenders) or (y) if an Event of Default exists, in accordance with the respective Percentages of the Lenders until principal is paid in full, each Lenders' share of such payment to be allocated pro rata among the outstanding Types of Loans owed to such Lender; and
 - (vi) any other sums due to the Administrative Agent or any Lender under any of the Loan Documents.

2.23. [<u>Reserved</u>].

2.24. <u>Voluntary Prepayments of Loans</u>.

- (a) The Borrower shall have the right at any time and from time to time to prepay any of its Borrowings in whole or in part, subject to prior notice in accordance with paragraph (b) of this Section.
- (b) The Borrower shall notify the Administrative Agent by email of any prepayment under clause (a) above not later than 11:00 a.m. (Central Time) on the date of prepayment. Each such notice shall be irrevocable and shall specify the prepayment date and the principal amount of each Borrowing or portion thereof to be prepaid. Promptly following receipt of any such notice relating to a Borrowing, the Administrative Agent shall advise the Lenders of the contents thereof. Each partial prepayment pursuant to clause (a) above of any

Borrowing shall be in an amount that would be permitted in the case of an advance of a Borrowing of the same Type as provided in <u>Section 2.11</u> hereof. Each prepayment of a Borrowing shall be applied ratably to the Loans included in the applicable prepaid Borrowing. Prepayments shall be accompanied by accrued interest to the extent required by <u>Section 2.12</u> hereof and by breakage costs to the extent required by <u>Section 4.4</u> hereof.

ARTICLE III.

[RESERVED]

ARTICLE IV.

CHANGE IN CIRCUMSTANCES

4.1. <u>Yield Protection</u>. If any Change in Law:

- (i) subjects the Administrative Agent, any Lender or any applicable Lending Office to any taxes, duties, levies, imposts, deductions, assessments, fees, charges or withholdings, and all liabilities with respect thereto (other than (A) Taxes, (B) Excluded Taxes or (C) Other Taxes) on its loans, loan principal, letters of credit, commitments, or other obligations, or its deposits, reserves, other liabilities, capital or liquidity attributable to making, converting, funding or maintaining its Borrowings or its Term Loan Commitment, or
- (ii) imposes or increases or deems applicable any reserve, assessment, insurance charge, special deposit, liquidity or similar requirement against assets of, deposits with or for the account of, or credit extended by, any Lender or any applicable Lending Office (other than reserves and assessments taken into account in determining the interest rate applicable to Eurocurrency Borrowings), or
- (iii) imposes any other condition, and the result is to increase the cost of any Lender or any applicable Lending Office of making, funding or maintaining loans or reduces any amount receivable by any Lender or any applicable Lending Office in connection with loans, or requires any Lender or any applicable Lending Office to make any payment calculated by reference to the amount of loans held or interest received by it, by an amount deemed material by such Lender,

then, within fifteen (15) days of demand by the Administrative Agent or such Lender, the Borrower shall pay the Administrative Agent or such Lender that portion of such increased expense incurred or reduction in an amount received which the Administrative Agent or such Lender determines is attributable to making, funding and maintaining its Borrowings and its Term Loan Commitment (which determination shall be made in good faith (and not on an arbitrary or capricious basis) and consistent with similarly situated customers of the applicable Lender after consideration of such factors as such Lender then reasonably determines to be relevant).

4.2. <u>Changes in Capital Adequacy Regulations</u>. If a Lender determines the amount of capital or liquidity required or expected to be maintained by such Lender, any Lending Office of such Lender or any corporate entity controlling such Lender is increased as a result of a Change in Law (which determination shall be made in good faith (and not on an arbitrary or capricious basis) and consistent with similarly situated customers of the applicable Lender after consideration of such factors as such Lender then reasonably determines to be relevant), then, within fifteen (15) days of demand by such Lender, the Borrower shall pay such Lender the amount necessary to compensate for any shortfall in the rate of return on the portion of such increased capital which is attributable to this Agreement, its Borrowings or its obligation to make Borrowings hereunder (after taking into account such Lender's policies as to capital adequacy). Without in any way affecting the Borrower's obligation to pay compensation actually claimed by a Lender under this Section 4.2, the Borrower shall have the right to replace any Lender which has demanded such compensation provided that the Borrower notifies such Lender that it has elected to replace such Lender and notifies such Lender and the Administrative Agent of the identity of the proposed replacement Lender not more than six (6) months after the date of such Lender's most recent demand for compensation under this Section 4.2, and further provided that such replacement is otherwise in accordance with Section 4.7. The Lender being replaced shall assign its Term Loan Exposure and its rights and obligations under this Facility to the replacement Lender in accordance with the requirements of Section 13.3 hereof and the replacement Lender shall assume such Term Loan Exposure, all pursuant to an assignment agreement substantially in the form of <u>Exhibit I</u> hereto. The purchase by the replacement Lender shall be at par (plus all accrued and unpaid interest and any other sums owed to such Lender being replaced hereunder) which shall be paid to the Lender being replaced upon the execution and delivery of the assignment. The Lender being replaced shall continue to be entitled to the benefits of Sections 4.1, 4.2, 4.4, 4.5 and 14.6 for events recurring prior to assignment to the replacement Lender.

4.3. [<u>Reserved</u>].

4.4. <u>Funding Indemnification</u>. If any payment of a LIBOR Borrowing occurs on a date which is not the last day of the applicable Interest Period, whether because of acceleration, prepayment or otherwise, or a LIBOR Borrowing is not made, continued or converted on the date specified by the Borrower for any reason other than default by one or more of the Lenders, the Borrower will indemnify each Lender for any loss or cost incurred by such Lender, as applicable, resulting therefrom, including, without limitation, any loss or cost in liquidating or employing deposits acquired to fund or maintain the LIBOR Borrowing.

4.5. Taxes.

(i) All payments by the Borrower to or for the account of any Lender or the Administrative Agent hereunder or under any Note shall be made free and clear of and without deduction for any and all Taxes. If the Borrower shall be required by law to deduct any Taxes from or in respect of any sum payable hereunder to any Lender or the Administrative Agent, (a) the sum payable shall be increased as necessary so that after making all required deductions (including deductions applicable to additional sums payable under this Section 4.5) such Lender or the Administrative Agent (as the case may be) receives an amount equal to the sum it would have received had no such deductions been made, (b) the Borrower shall make such deductions, (c) the Borrower shall pay the full amount deducted to the relevant authority in accordance with Applicable

Law, and (d) the Borrower shall furnish to the Administrative Agent the original copy of a receipt evidencing payment thereof within 30 days after such payment is made.

- (ii) In addition, the Borrower hereby agrees to pay any present or future stamp or documentary taxes and any other excise or property taxes, charges or similar levies which arise from any payment made hereunder or under any Note or from the execution or delivery of, or otherwise with respect to, this Agreement or any Note ("Other Taxes").
- (iii) The Borrower hereby agrees to indemnify the Administrative Agent and each Lender for the full amount of Taxes or Other Taxes (including, without limitation, any Taxes or Other Taxes imposed on amounts payable under this <u>Section 4.5</u>) paid by the Administrative Agent or such Lender as a result of its Term Loan Commitment, any Loans made by it hereunder, or otherwise in connection with its participation in this Agreement and any liability (including penalties, interest and expenses) arising therefrom or with respect thereto. Payments due under this indemnification shall be made within 30 days of the date the Administrative Agent or such Lender makes demand therefor pursuant to <u>Section 4.6</u>.
- (iv) Each Lender that is not incorporated under the laws of the United States of America or a state thereof (each a "Non-U.S. Lender") agrees that it will, not more than ten (10) Business Days after the date of this Agreement, (i) deliver to the Administrative Agent two duly completed copies of United States Internal Revenue Service Form W-8BEN, W-8BEN-E or W-8ECI, certifying in either case that such Lender is entitled to receive payments under this Agreement without deduction or withholding of any United States federal income taxes, and (ii) deliver to the Administrative Agent a United States Internal Revenue Form W-8 or W-9, as the case may be, and certify that it is entitled to an exemption from United States backup withholding tax. Each Non-U.S. Lender further undertakes to deliver to each of the Borrower and the Administrative Agent (x) renewals or additional copies of such form (or any successor form) on or before the date that such form expires or becomes obsolete, and (y) after the occurrence of any event requiring a change in the most recent forms so delivered by it, such additional forms or amendments thereto as may be reasonably requested by the Borrower or the Administrative Agent. All forms or amendments described in the preceding sentence shall certify that such Lender is entitled to receive payments under this Agreement without deduction or withholding of any United States federal income taxes, unless an event (including without limitation any change in treaty, law or regulation) has occurred prior to the date on which any such delivery would otherwise be required which renders all such forms inapplicable or which would prevent such Lender from duly completing and delivering any such form or amendment with respect to it and such Lender advises the Borrower and the Administrative Agent that it is not capable of receiving payments without any deduction or withholding of United States federal income tax.
- (v) For any period during which a Non-U.S. Lender has failed to provide the Borrower with an appropriate form pursuant to clause (iv), above (unless such failure is due to a change in treaty, law or regulation, or any change in the interpretation or administration thereof by any governmental authority, occurring subsequent to the date on which a form originally was required to be provided), such Non-U.S. Lender shall not be entitled to indemnification under this <u>Section 4.5</u> with respect to Taxes imposed by the United States; <u>provided</u> that, should a Non-U.S. Lender which is otherwise exempt from or subject to a reduced rate of withholding tax become subject to Taxes because of its failure to deliver a form required under clause (iv), above, the Borrower shall take such

steps as such Non-U.S. Lender shall reasonably request to assist such Non-U.S. Lender to recover such Taxes.

- (vi) Any Lender that is entitled to an exemption from or reduction of withholding tax with respect to payments under this Agreement or any Note pursuant to the law of any relevant jurisdiction or any treaty shall deliver to the Borrower (with a copy to the Administrative Agent), at the time or times prescribed by Applicable Law, such properly completed and executed documentation prescribed by Applicable Law as will permit such payments to be made without withholding or at a reduced rate.
- (vii) If the U.S. Internal Revenue Service or any other governmental authority of the United States or any other country or any political subdivision thereof asserts a claim that the Administrative Agent did not properly withhold tax from amounts paid to or for the account of any Lender (because the appropriate form was not delivered or properly completed, because such Lender failed to notify the Administrative Agent of a change in circumstances which rendered its exemption from withholding ineffective, or for any other reason), such Lender shall indemnify the Administrative Agent fully for all amounts paid, directly or indirectly, by the Administrative Agent as tax, withholding therefor, or otherwise, including penalties and interest, and including taxes imposed by any jurisdiction on amounts payable to the Administrative Agent under this subsection, together with all costs and expenses related thereto (including attorneys fees and time charges of attorneys for the Administrative Agent, which attorneys may be employees of the Administrative Agent). The obligations of the Lenders under this Section 4.5(vii) shall survive the payment of the Obligations and termination of this Agreement.
- (viii) If a payment made to a Lender hereunder or under any Note would be subject to United States federal withholding tax imposed by FATCA if such Lender were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the Code, as applicable) such Lender shall deliver to the Borrower and the Administrative Agent at the time or times prescribed by law and at such time or times reasonably requested by the Borrower or the Administrative Agent such documentation prescribed by Applicable Law (including as prescribed by Section 1471(b)(3)(C)(i) of the Code) and such additional documentation reasonably requested by the Borrower or the Administrative Agent as may be necessary for the Borrower and the Administrative Agent to comply with their obligations under FATCA and to determine that such Lender has complied with such Lender's obligations under FATCA or to determine the amount to deduct and withhold from such payment. Solely for purposes of this Section 4.5(viii), "FATCA" shall include any amendments made to FATCA after the date of this Agreement.
- (ix) Each of the Lenders represents that as of the Agreement Execution Date it is not aware of any facts that would give rise to a claim for additional payments under this <u>Section 4.5</u>.

- 4.6. Lender Statements; Survival of Indemnity. To the extent reasonably possible, each Lender shall designate an alternate Lending Office with respect to its LIBOR Borrowings to reduce any liability of the Borrower to such Lender under Sections 4.1, 4.2 and 4.5 or to avoid the unavailability of a LIBOR Borrowing, so long as such designation is not disadvantageous to such Lender. Each Lender shall deliver a written statement of such Lender as to the amount due, if any, under Sections 4.1, 4.2, 4.4 and 4.5 hereof. Such written statement shall set forth in reasonable detail the calculations upon which such Lender determined such amount and shall be final, conclusive and binding on the Borrower in the absence of manifest error. The amount due in such statement shall not include amounts due under Section 4.5 that are either attributable to facts known to the Lender as of the Agreement Execution Date or that relate to a time period more than ninety (90) days prior to the giving of such written statement. Determination of amounts payable under such Sections in connection with a LIBOR Borrowing shall be calculated as though each Lender funded its LIBOR Borrowing through the purchase of a deposit of the type and maturity corresponding to the deposit used as a reference in determining the Adjusted LIBOR Rate applicable to such Borrowing, whether in fact that is the case or not. Unless otherwise provided herein, the amount specified in the written statement shall be payable on demand after receipt by the Borrower of the written statement. The obligations of the Borrower under Sections 4.1, 4.2, 4.4 and 4.5 hereof shall survive payment of the Obligations and termination of this Agreement.
- Replacement of Lenders under Certain Circumstances. The Borrower shall be permitted to replace any Lender which (a) is not capable of receiving payments without any deduction or withholding of United States federal income tax pursuant to Section 4.5, or (b) cannot maintain its LIBOR Borrowings at a suitable Lending Office pursuant to Section 4.6, with a replacement bank or other financial institution or (c) becomes a Defaulting Lender; provided that (i) such replacement does not conflict with any applicable legal or regulatory requirements affecting the remaining Lenders, (ii) no Event of Default or (after notice thereof to the Borrower) no Default shall have occurred and be continuing at the time of such replacement, (iii) the replacement bank or institution shall purchase at par all Loans and the Borrower or the replacement bank or institution shall repay other amounts owing to such replaced Lender prior to the date of replacement, (iv) the Borrower shall be liable to such replaced Lender under Sections 4.4 and 4.6 if any LIBOR Borrowing owing to such replaced Lender shall be prepaid (or purchased) other than on the last day of the Interest Period relating thereto, (v) the replacement bank or institution, if not already a Lender, and the terms and conditions of such replacement, shall be reasonably satisfactory to the Administrative Agent, (vi) the replaced Lender shall be obligated to make such replacement in accordance with the provisions of Section 13.3 (provided that the Borrower shall be obligated to pay the processing fee referred to therein and the replaced Lender's consent shall not be required), (vii) until such time as such replacement shall be consummated, the Borrower shall continue to pay all amounts payable hereunder without setoff, deduction, counterclaim or withholding and (viii) any such replacement shall not be deemed to be a waiver of any rights which the Borrower, the Administrative Agent or any other Lender shall have against the replaced Lender.

4.8. Changed Circumstances.

(a) <u>Circumstances Affecting Base LIBOR Rate Availability</u>. Subject to clause (c) below, in connection with any request for a LIBOR Borrowing or a conversion to or continuation thereof or otherwise, if for any reason (i) the Administrative Agent shall determine (which determination shall be conclusive and binding absent manifest error) that Dollar deposits are not being offered to banks

in the London interbank eurodollar market for the applicable amount and Interest Period of such Loan, (ii) the Administrative Agent shall determine (which determination shall be conclusive and binding absent manifest error) that reasonable and adequate means do not exist for the ascertaining the LIBOR Rate for such Interest Period with respect to a proposed LIBOR Borrowing or (iii) the Required Lenders shall determine (which determination shall be conclusive and binding absent manifest error) that the LIBOR Rate does not adequately and fairly reflect the cost to such Lenders of making or maintaining such Loans during such Interest Period, then the Administrative Agent shall promptly give notice thereof to the Borrower. Thereafter, until the Administrative Agent notifies the Borrower that such circumstances no longer exist, the obligation of the Lenders to make LIBOR Borrowings and the right of the Borrower to convert any Loan to or continue any Loan as a LIBOR Borrowing shall be suspended, and the Borrower shall either (A) repay in full (or cause to be repaid in full) the then outstanding principal amount of each such LIBOR Borrowing together with accrued interest thereon (subject to Section 4.4), on the last day of the then current Interest Period applicable to such LIBOR Borrowing; or (B) Borrowings shall, to the extent permissible under Applicable Law, bear interest at the Substitute Rate, rather than any Loans becoming payable.

(b) Laws Affecting Base LIBOR Rate Availability. If, after the date hereof, the introduction of, or any change in, any Applicable Law or any change in the interpretation or administration thereof by any Governmental Authority, central bank or comparable agency charged with the interpretation or administration thereof, or compliance by any of the Lenders (or any of their respective Lending Offices) with any request or directive (whether or not having the force of law) of any such Governmental Authority, central bank or comparable agency, shall make it unlawful or impossible for any of the Lenders (or any of their respective Lending Offices) to honor its obligations hereunder to make or maintain any LIBOR Borrowing, such Lender shall promptly give notice thereof to the Administrative Agent and the Administrative Agent shall promptly give notice to the Borrower and the other Lenders. Thereafter, until the Administrative Agent notifies the Borrower that such circumstances no longer exist, (i) the obligations of the Lenders to make LIBOR Borrowings, and the right of the Borrower to convert any Loan to a LIBOR Borrowing or continue any Loan as a LIBOR Borrowing shall be suspended and thereafter the Borrower may select only Adjusted Base Rate Borrowings that shall, to the extent permissible under Applicable Law, bear interest at the Substitute Rate, and (ii) if any of the Lenders may not lawfully continue to maintain a LIBOR Borrowing to the end of the then current Interest Period applicable thereto, the applicable Loan shall immediately be converted to a Adjusted Base Rate Borrowing bearing interest at the Substitute Rate for the remainder of such Interest Period.

(c) Benchmark Replacement Setting.

(i) <u>Benchmark Replacement.</u> (A) Notwithstanding anything to the contrary herein or in any other Loan Document (and any documentation related to a Rate Management Transaction shall be deemed not to be a "Loan Document" for purposes of this <u>Section 4.8(c)</u>) if a Benchmark Transition Event, an Early Opt-in Election or an Other Benchmark Rate Election, as applicable, and its related Benchmark Replacement Date have occurred prior to the Reference Time in respect of any setting of the then-current Benchmark, then (x) if a Benchmark Replacement is determined in accordance with clause (a)(1) or (a)(2) of the definition of "Benchmark Replacement" for such Benchmark Replacement Date, such Benchmark Replacement will replace such Benchmark for all purposes hereunder and under any Loan Document in respect

of such Benchmark setting and subsequent Benchmark settings without any amendment to, or further action or consent of any other party to, this Agreement or any other Loan Document and (y) if a Benchmark Replacement is determined in accordance with clause (a)(3) or clause (c) of the definition of "Benchmark Replacement" for such Benchmark Replacement Date, such Benchmark Replacement will replace such Benchmark for all purposes hereunder and under any Loan Document in respect of any Benchmark setting at or after 5:00 p.m. (New York City time) on the fifth (5th) Business Day after the date notice of such Benchmark Replacement is provided to the Lenders without any amendment to, or further action or consent of any other party to, this Agreement or any other Loan Document so long as the Administrative Agent has not received, by such time, written notice of objection to such Benchmark Replacement from Lenders comprising the Required Lenders. If an Unadjusted Benchmark Replacement is Daily Simple SOFR, all interest payments will be payable on a monthly basis.

- (B) Notwithstanding anything to the contrary herein or in any other Loan Document, if a Term SOFR Transition Event and its related Benchmark Replacement Date have occurred prior to the Reference Time in respect of any setting of the then-current Benchmark, then the applicable Benchmark Replacement will replace the then-current Benchmark for all purposes hereunder or under any Loan Document in respect of such Benchmark setting and subsequent Benchmark settings, without any amendment to, or further action or consent of any other party to, this Agreement or any other Loan Document; provided that this clause (B) shall not be effective unless the Administrative Agent has delivered to the Lenders and the Borrower a Term SOFR Notice. For the avoidance of doubt, the Administrative Agent shall not be required to deliver a Term SOFR Notice after a Term SOFR Transition Event and may elect or not elect to do so in its sole discretion.
- (ii) <u>Benchmark Replacement Conforming Changes</u>. In connection with the implementation of a Benchmark Replacement, the Administrative Agent will have the right to make Benchmark Replacement Conforming Changes from time to time and, notwithstanding anything to the contrary herein or in any other Loan Document, any amendments implementing such Benchmark Replacement Conforming Changes will become effective without any further action or consent of any other party to this Agreement or any other Loan Document.
- (iii) Notices; Standards for Decisions and Determinations. The Administrative Agent will promptly notify the Borrower and the Lenders of (A) any occurrence of a Benchmark Transition Event, a Term SOFR Transition Event, an Early Opt-in Election or an Other Benchmark Rate Election, as applicable, and its related Benchmark Replacement Date, (B) the implementation of any Benchmark Replacement, (C) the effectiveness of any Benchmark Replacement Conforming Changes, (D) the removal or reinstatement of any tenor of a Benchmark pursuant to Section 4.8(c)(iv) below and (E) the commencement or conclusion of any Benchmark Unavailability Period. Any determination, decision or election that may be made by the Administrative Agent or, if applicable, any Lender (or group of Lenders) pursuant to this Section 4.8(c), including any determination with respect to a tenor, rate or adjustment or of the occurrence or non-occurrence of an event, circumstance or date

and any decision to take or refrain from taking any action or any selection, will be conclusive and binding absent manifest error and may be made in its or their sole discretion and without consent from any other party to this Agreement or any other Loan Document, except, in each case, as expressly required pursuant to this <u>Section 4.8(c)</u>.

- (iv) <u>Unavailability of Tenor of Benchmark.</u> Notwithstanding anything to the contrary herein or in any other Loan Document, at any time (including in connection with the implementation of a Benchmark Replacement), (A) if the then-current Benchmark is a term rate (including Term SOFR or USD LIBOR) and either (1) any tenor for such Benchmark is not displayed on a screen or other information service that publishes such rate from time to time as selected by the Administrative Agent in its reasonable discretion or (2) the regulatory supervisor for the administrator of such Benchmark has provided a public statement or publication of information announcing that any tenor for such Benchmark is or will be no longer representative, then the Administrative Agent may modify the definition of "Interest Period" for any Benchmark settings at or after such time to remove such unavailable or non-representative tenor and (B) if a tenor that was removed pursuant to clause (A) above either (1) is subsequently displayed on a screen or information service for a Benchmark (including a Benchmark Replacement) or (2) is not, or is no longer, subject to an announcement that it is or will no longer be representative for a Benchmark (including a Benchmark Replacement), then the Administrative Agent may modify the definition of "Interest Period" for all Benchmark settings at or after such time to reinstate such previously removed tenor.
- (v) <u>Benchmark Unavailability Period</u>. Upon the Borrower's receipt of notice of the commencement of a Benchmark Unavailability Period, the Borrower may revoke any request for a borrowing of, conversion to or continuation of LIBOR Borrowings to be made, converted or continued during any Benchmark Unavailability Period and, failing that, the Borrower will be deemed to have converted any such request into a request for a borrowing of or conversion to Adjusted Base Rate Borrowings that shall, to the extent permissible under Applicable Law, bear interest at the Substitute Rate. During any Benchmark Unavailability Period, Borrowings bearing interest based on the Adjusted Base Rate shall instead bear interest at the Substitute Rate.
- (vi) London Interbank Offered Rate Benchmark Transition Event. On March 5, 2021, the IBA, the administrator of the London interbank offered rate, and the FCA, the regulatory supervisor of the IBA, made the Announcements that the final publication or representativeness date for Dollars for (I) 1-week and 2-month London interbank offered rate tenor settings will be December 31, 2021 and (II) overnight, 1-month, 3-month, 6-month and 12-month London interbank offered rate tenor settings will be June 30, 2023. No successor administrator for the IBA was identified in such Announcements. The parties hereto agree and acknowledge that the Announcements resulted in the occurrence of a Benchmark Transition Event with respect to the London interbank offered rate pursuant to the terms of this Agreement and that any obligation of the Administrative Agent to notify any parties of such Benchmark Transition Event pursuant to clause (iii) of this Section 4.8(c) shall be deemed satisfied.

ARTICLE V.

CONDITIONS PRECEDENT

- 5.1. <u>Conditions Precedent to Closing</u>. The Lenders shall not be required to make the initial Loans pursuant to <u>Section 2.1</u> hereunder unless (i) the Borrower shall have paid all fees then due and payable to the Lenders, PNC Capital Markets LLC, Wells Fargo Securities, LLC and the Administrative Agent hereunder, including reasonable legal fees and out-of-pocket expenses, (ii) all of the conditions set forth in <u>Section 5.2</u> are satisfied, and (iii) the Borrower shall have furnished to the Administrative Agent, in form and substance satisfactory to the Lenders and their counsel and with sufficient copies for the Lenders, the following:
 - (a) <u>Certificates of Limited Partnership/Incorporation</u>. A copy of the Certificate of Limited Partnership for the Borrower and a copy of the articles of incorporation of the General Partner, each certified by the appropriate Secretary of State or equivalent state official.
 - (b) <u>Agreements of Limited Partnership/Bylaws</u>. A copy of the Agreement of Limited Partnership for the Borrower and a copy of the bylaws of the General Partner, including all amendments thereto, each certified by the Secretary or an Assistant Secretary of the General Partner as being in full force and effect on the Agreement Execution Date.
 - (c) <u>Good Standing Certificates</u>. A certified copy of a certificate from the Secretary of State or equivalent state official of the states where the Borrower and the General Partner are organized, dated as of the most recent practicable date, showing the good standing or partnership qualification (if issued) of (i) the Borrower, and (ii) the General Partner.
 - (d) <u>Foreign Qualification Certificates</u>. A certified copy of a certificate from the Secretary of State or equivalent state official of the state where the Borrower and the General Partner maintain their principal place of business, dated as of the most recent practicable date, showing the qualification to transact business in such state as a foreign limited partnership or foreign corporation, as the case may be, for (i) the Borrower, and (ii) the General Partner.
 - (e) <u>Resolutions</u>. A copy of a resolution or resolutions adopted by the Board of Directors of the General Partner, certified by the Secretary or an Assistant Secretary of the General Partner as being in full force and effect on the Agreement Execution Date, authorizing the Borrowings provided for herein and the execution, delivery and performance of the Loan Documents by the General Partner to be executed and delivered by it hereunder on behalf of itself and the Borrower.
 - (f) <u>Incumbency Certificate</u>. A certificate, signed by the Secretary or an Assistant Secretary of the General Partner and dated the Agreement Execution Date, as to the incumbency, and containing the specimen signature or signatures, of the Persons authorized to execute and deliver the Loan Documents to be executed and delivered by it and the Borrower hereunder.

- (g) <u>Loan Documents</u>. (i) From each party hereto either (A) a counterpart of this Agreement signed on behalf of such party or (B) written evidence satisfactory to the Administrative Agent (which may include telecopy or electronic transmission of a signed signature page of this Agreement) that such party has signed a counterpart of this Agreement and (ii) duly executed copies of the other Loan Documents.
- (h) <u>Opinion of Borrower's Counsel</u>. A written opinion, dated the Agreement Execution Date, from outside counsel for the Borrower which counsel is reasonably satisfactory to Administrative Agent, substantially in the form attached hereto as <u>Exhibit E</u>.
- (i) <u>Opinion of General Partner's Counsel</u>. A written opinion, dated the Agreement Execution Date, from outside counsel for the General Partner which counsel is reasonably satisfactory to Administrative Agent, substantially in the form attached hereto as <u>Exhibit F</u>.

(j) <u>Financial and Related Information</u>. The following information:

- (i) The most recent financial statements of the Borrower and the General Partner and a certificate from a Qualified Officer of the Borrower that no change in the Borrower's financial condition that would have a Material Adverse Effect has occurred since December 31, 2020; and
- (ii) A Disbursement Instruction Agreement, in substantially the form of <u>Exhibit C</u> hereto, addressed to the Administrative Agent and signed by a Qualified Officer, together with such other related money transfer authorizations as the Administrative Agent may have reasonably requested.
- (k) <u>Closing Certificate</u>. A certificate, signed by an officer of the Borrower, stating that on the Agreement Execution Date no Default or Event of Default has occurred and is continuing and that all representations and warranties of the Borrower contained herein are true and correct as of the Agreement Execution Date as and to the extent set forth herein.

(l) [Reserved].

- (m) <u>PATRIOT Act, etc.</u> At least five (5) Business Days prior to the Closing Date, (i) all documentation and other information reasonably requested by the Administrative Agent or any Lender or required by regulatory authorities in order for the Administrative Agent and the Lenders to comply with requirements of any Anti-Corruption Laws, including the PATRIOT Act and any applicable "know your customer" rules and regulations, and (ii) to the extent requested by the Administrative Agent or any Lender, a Beneficial Ownership Certification in relation to the Borrower.
- (n) <u>Other Evidence as any Lender May Require</u>. Such other evidence as any Lender may reasonably request to establish the consummation of the transactions contemplated hereby, the taking of all necessary actions in any proceedings in connection herewith and compliance with the conditions set forth in this Agreement.

When all such conditions have been fulfilled (or, in the Lenders' sole discretion, waived by Lenders), the Administrative Agent shall confirm in writing to the Borrower that the initial Borrowings are then available to the Borrower hereunder.

- 5.2. <u>Conditions Precedent to Subsequent Borrowings</u>. Borrowings after the initial Borrowing (other than conversions or continuations) shall be made from time to time as requested by the Borrower, and the obligation of each Lender to make any Borrowing is subject to the following terms and conditions:
 - (a) prior to and after giving effect to each such Borrowing, no Default or Event of Default shall have occurred and be continuing under this Agreement or any of the Loan Documents, and the Borrower shall deliver a certificate of the Borrower to such effect; and
 - (b) The representations and warranties contained in <u>Article VI</u> and <u>VII</u> are true and correct as of such borrowing date, as and to the extent set forth therein, except to the extent any such representation or warranty is stated to relate solely to an earlier date, in which case such representation or warranty shall be true and correct on and as of such earlier date.

Subject to the last grammatical paragraphs of <u>Article VI</u> and <u>VII</u> hereof, each Borrowing Notice shall constitute a representation and warranty by the Borrower that the conditions contained in <u>Sections 5.2(a)</u> and <u>(b)</u> have been satisfied. Unless set forth in writing to the contrary, the making of its initial Loan or Incremental Loan by a Lender shall constitute a certification by such Lender to the Administrative Agent and the other Lenders that the conditions precedent for initial Loans set forth in <u>Section 5.1</u> and <u>Section 5.2</u> (if applicable) that have not previously been waived by the Lenders in accordance with the terms of this Agreement have been satisfied.

ARTICLE VI.

REPRESENTATIONS AND WARRANTIES

The Borrower hereby represents and warrants to the Lenders that:

6.1. Existence. The Borrower is a limited partnership duly organized and existing under the laws of the State of Delaware, with its principal place of business in the State of Illinois, and is duly qualified as a foreign limited partnership, properly licensed (if required), in good standing and has all requisite authority to conduct its business in each jurisdiction in which it owns Properties and, except where the failure to be so qualified or to obtain such authority would not have a Material Adverse Effect, in each other jurisdiction in which its business is conducted. Each of its Subsidiaries is duly organized, validly existing and in good standing under the laws of its jurisdiction of organization and has all requisite authority to conduct its business in each jurisdiction in which it owns Property, and except where the failure to be so qualified or to obtain such authority would not have a Material Adverse Effect, in each other jurisdiction in which it conducts business.

- 6.2. <u>Corporate/Partnership Powers</u>. The execution, delivery and performance of the Loan Documents required to be delivered by the Borrower hereunder are within the partnership authority of such entity and the corporate powers of the general partners of such entity, have been duly authorized by all requisite action, and are not in conflict with the terms of any organizational instruments of such entity, or any instrument or agreement to which the Borrower or the General Partner is a party or by which the Borrower, the General Partner or any of their respective assets may be bound or affected.
- 6.3. <u>Power of Officers</u>. The officers of the General Partner executing the Loan Documents required to be delivered by such entities hereunder have been duly elected or appointed and were fully authorized to execute the same at the time each such agreement, certificate or instrument was executed.
- 6.4. <u>Government and Other Approvals</u>. No approval, consent, exemption or other action by, or notice to or filing with, any governmental authority is necessary in connection with the execution, delivery or performance of the Loan Documents required hereunder.

6.5. Solvency.

- (i) Immediately after the Agreement Execution Date and immediately following the making of each Loan and after giving effect to the application of the proceeds of such Loans, (a) the fair value of the assets of the Borrower and its Subsidiaries on a consolidated basis, at a fair valuation, will exceed the debts and liabilities, subordinated, contingent or otherwise, of the Borrower and its Subsidiaries on a consolidated basis; (b) the present fair saleable value of the Properties of the Borrower and its Subsidiaries on a consolidated basis will be greater than the amount that will be required to pay the probable liability of the Borrower and its Subsidiaries on a consolidated basis on their debts and other liabilities, subordinated, contingent or otherwise, as such debts and other liabilities become absolute and matured; (c) the Borrower and its Subsidiaries on a consolidated basis will be able to pay their debts and liabilities, subordinated, contingent or otherwise, as such debts and liabilities become absolute and matured; and (d) the Borrower and its Subsidiaries on a consolidated basis will not have unreasonably small capital with which to conduct the businesses in which they are engaged as such businesses are now conducted and are proposed to be conducted after the date hereof.
- (ii) The Borrower does not intend to, or to permit any of its Subsidiaries to incur debts beyond its ability to pay such debts as they mature, taking into account the timing of and amounts of cash to be received by it or any such Subsidiary and the timing of the amounts of cash to be payable on or in respect of its Indebtedness or the Indebtedness of any such Subsidiary.

- 6.6. <u>Compliance With Laws</u>. There is no judgment, decree or order or any law, rule or regulation of any court or governmental authority binding on the Borrower or any of its Subsidiaries which would be contravened by the execution, delivery or performance of the Loan Documents required hereunder.
- 6.7. Enforceability of Agreement. This Agreement is the legal, valid and binding agreement of the Borrower, and the Notes when executed and delivered will be the legal, valid and binding obligations of the Borrower, enforceable against the Borrower in accordance with their respective terms, and the Loan Documents required hereunder, when executed and delivered, will be similarly legal, valid, binding and enforceable except to the extent that such enforcement may be limited by applicable bankruptcy, insolvency, reorganization or other similar laws affecting the rights of creditors generally.
- 6.8. <u>Title to Property.</u> To the best of the Borrower's knowledge after due inquiry, the Borrower or its Subsidiaries has good and marketable title to the Properties and assets reflected in the financial statements as owned by it or any such Subsidiary free and clear of Liens except for the Permitted Liens. The execution, delivery or performance of the Loan Documents required to be delivered by the Borrower hereunder will not result in the creation of any Lien on the Properties. No consent to the transactions contemplated hereunder is required from any ground lessor or mortgagee or beneficiary under a deed of trust or any other party except as has been delivered to the Lenders.
- 6.9. <u>Litigation</u>. There are no suits, arbitrations, claims, disputes or other proceedings (including, without limitation, any civil, criminal, administrative or environmental proceedings), pending or, to the best of the Borrower's knowledge, threatened against or affecting the Borrower or any of the Properties, the adverse determination of which individually or in the aggregate would have a Material Adverse Effect on the Borrower and/or would cause a Material Adverse Financial Change of the Borrower or materially impair the Borrower's ability to perform its obligations hereunder or under any instrument or agreement required hereunder, except as disclosed on <u>Schedule 6.9</u> hereto, or otherwise disclosed to Lenders in accordance with the terms hereof.
- 6.10. Events of Default. No Default or Event of Default has occurred and is continuing or would result from the incurring of obligations by the Borrower under any of the Loan Documents or any other document to which the Borrower is a party.
- 6.11. <u>Investment Company Act of 1940</u>. The Borrower is not and will by such acts as may be necessary continue not to be, an investment company within the meaning of the Investment Company Act of 1940.

- 6.12. <u>Public Utility Holding Company Act</u>. The Borrower is not a "holding company" or a "subsidiary company" of a "holding company," or an "affiliate" of a "holding company," or of a "subsidiary company" of a "holding company," within the definitions of the Public Utility Holding Company Act of 1935, as amended.
- 6.13. Regulation U. The proceeds of the Borrowings will not be used, directly or indirectly, in a manner which would cause the Facility to be treated as a "Purpose Credit."
- 6.14. <u>No Material Adverse Financial Change</u>. To the best knowledge of the Borrower, there has been no Material Adverse Financial Change in the condition of the Borrower since the date of the financial and/or operating statements most recently submitted to the Lenders.
- 6.15. <u>Financial Information</u>. All financial statements furnished to the Lenders by or at the direction of the Borrower and all other financial information and data furnished by the Borrower to the Lenders are complete and correct in all material respects as of the date thereof, and such financial statements have been prepared in accordance with GAAP and fairly present the consolidated financial condition and results of operations of the Borrower as of such date. The Borrower has no contingent obligations, liabilities for taxes or other outstanding financial obligations which are material in the aggregate, except as disclosed in such statements, information and data.
- 6.16. <u>Factual Information</u>. All factual information heretofore or contemporaneously furnished by or on behalf of the Borrower to the Lenders for purposes of or in connection with this Agreement and the other Loan Documents and the transactions contemplated therein is, and all other such factual information hereafter furnished by or on behalf of the Borrower to the Lenders will be, true and accurate (taken as a whole) in all material respects on the date as of which such information is dated or certified and not incomplete by omitting to state any material fact necessary to make such information (taken as a whole) not misleading at such time. As of the Closing Date, all of the information included in any Beneficial Ownership Certification is true and correct.
- 6.17. <u>ERISA</u>. (i) The Borrower is not an entity deemed to hold "plan assets" within the meaning of ERISA or any regulations promulgated thereunder of an employee benefit plan (as defined in Section 3(3) of ERISA) which is subject to Title I of ERISA or any plan within the meaning of Section 4975 of the Code, and (ii) the execution of this Agreement and the transactions contemplated hereunder do not give rise to a prohibited transaction within the meaning of Section 406 of ERISA or Section 4975 of the Code.
- 6.18. <u>Taxes</u>. All required Federal, state and other material tax returns have been filed by the Borrower with the appropriate authorities except (i) to the extent that extensions of time to file have been requested, granted and have not expired and (ii) to the extent such taxes are being contested in good faith and for which adequate reserves, in accordance with GAAP, are being maintained.
- 6.19. <u>Environmental Matters</u>. Except as disclosed in <u>Schedule 6.19</u>, each of the following representations and warranties is true and correct except to the extent that the facts and circumstances giving rise to any such failure to be so true and correct, in the aggregate, could not reasonably be expected to have a Material Adverse Effect:

- (i) To the knowledge of the Borrower, the Properties of the Borrower, its Subsidiaries, and Investment Affiliates do not contain any Materials of Environmental Concern in amounts or concentrations which constitute a violation of, or could reasonably give rise to liability under, Environmental Laws.
- (ii) The Borrower has not received any written notice alleging that any or all of the Properties of the Borrower and its Subsidiaries and Investment Affiliates and all operations at the Properties are not currently in compliance with all applicable Environmental Laws. Further, the Borrower has not received any written notice alleging the current existence of any contamination at or under such Properties in amounts or concentrations which constitute a violation of any Environmental Law, or any violation of any Environmental Law with respect to such Properties for which the Borrower, its Subsidiaries or Investment Affiliates is or could be liable.
- (iii) Neither the Borrower nor any of its Subsidiaries or Investment Affiliates has received any written notice of current non-compliance, liability or potential liability regarding Environmental Laws with regard to any of the Properties, nor does it have knowledge that any such notice will be received or is being threatened.
- (iv) To the knowledge of the Borrower during the ownership of the Properties by any or all of the Borrower, its Subsidiaries and Investment Affiliates, Materials of Environmental Concern have not been transported or disposed of from the Properties of the Borrower and its Subsidiaries and Investment Affiliates in violation of, or in a manner or to a location which could reasonably give rise to liability of the Borrower, any Subsidiary, or any Investment Affiliate under, Environmental Laws, nor during the ownership of the Properties by any or all of the Borrower, its Subsidiaries and Investment Affiliates have any Materials of Environmental Concern been generated, treated, stored or disposed of at, on or under any of such Properties in violation of, or in a manner that could give rise to liability of the Borrower, any Subsidiary or any Investment Affiliate under, any applicable Environmental Laws.
- (v) No judicial proceedings or governmental or administrative action is pending, or, to the knowledge of the Borrower, threatened, under any Environmental Law to which the Borrower, any of its Subsidiaries, or any Investment Affiliate, is named as a party with respect to the Properties of such entity, nor are there any consent decrees or other decrees, consent orders, administrative order or other orders, or other administrative or judicial requirements outstanding under any Environmental Law with respect to such Properties for which the Borrower, its Subsidiaries, or any Investment Affiliate is or could be liable.
- (vi) To the knowledge of the Borrower during the ownership of the Properties by any or all of the Borrower, its Subsidiaries and Investment Affiliates, there has been no release or threat of release of Materials of Environmental Concern at or from the Properties of the Borrower and its Subsidiaries and Investment Affiliates, or arising from or related to the operations of such entity in connection

with the Properties in violation of or in amounts or in a manner that could give rise to liability under Environmental Laws

- 6.20. <u>Insurance</u>. The Borrower maintains insurance on its properties consistent with the insurance maintained by other institutional owners of similar properties.
- 6.21. <u>No Brokers</u>. The Borrower has dealt with no brokers in connection with this Facility, and no brokerage fees or commissions are payable by or to any Person in connection with this Agreement or the Borrowings. Lenders shall not be responsible for the payment of any fees or commissions to any broker and the Borrower shall indemnify, defend and hold Lenders harmless from and against any claims, liabilities, obligations, damages, costs and expenses (including reasonable attorneys' fees and disbursements) made against or incurred by Lenders as a result of claims made or actions instituted by any broker or Person claiming by, through or under the Borrower in connection with the Facility.
- 6.22. <u>No Violation of Usury Laws</u>. No aspect of any of the transactions contemplated herein violate or will violate any usury laws or laws regarding the validity of agreements to pay interest in effect on the date hereof.
- 6.23. <u>Not a Foreign Person</u>. The Borrower is not a "foreign person" within the meaning of Section 1445 or 7701 of the Internal Revenue Code.
- 6.24. <u>No Trade Name</u>. Except for the name "First Industrial," and except as otherwise set forth on <u>Schedule 6.24</u> attached hereto, the Borrower does not use any trade name and has not and does not do business under any name other than their actual names set forth herein. The principal place of business of the Borrower is as stated in the recitals hereto.
- 6.25. <u>Subsidiaries</u>. <u>Schedule 6.25</u> hereto contains an accurate list of all of the presently existing Subsidiaries of the Borrower, setting forth the percentage of their respective Capital Stock owned by it or its Subsidiaries. All of the issued and outstanding shares of Capital Stock of such Subsidiaries have been duly authorized and issued and are fully paid and nonassessable.
- 6.26. <u>Unencumbered Assets</u>. <u>Schedule 6.26</u> hereto contains a complete and accurate description of Unencumbered Assets as of March 31, 2021 and as supplemented from time to time including the entity that owns each Unencumbered Asset. With respect to each Project identified from time to time as an Unencumbered Asset, the Borrower hereby represents and warrants as follows except to the extent disclosed in writing to the Lenders and approved by the Required Lenders (which approval shall not be unreasonably withheld) or except to the extent the failure of such representation and warranty to be true would not materially adversely affect the use and operation of such Project for its intended use or its marketability or value:
 - (a) No portion of any improvement on the Unencumbered Asset is located in an area identified by the Secretary of Housing and Urban Development or any successor thereto as an area having special flood hazards pursuant to the National Flood Insurance Act of 1968 or the Flood Disaster Protection Act of 1973, as amended, or any successor law, or, if located within any such area, the Borrower has obtained and will maintain the insurance prescribed in <u>Section 6.20</u> hereof.

- (b) To the Borrower's knowledge, the Unencumbered Asset and the present use and occupancy thereof are in material compliance with all Applicable Laws (including all Environmental Laws).
- (c) The Unencumbered Asset is served by all utilities required for the current or contemplated use thereof. All utility service is provided by public utilities and the Unencumbered Asset has accepted or is equipped to accept such utility service.
- (d) All public roads and streets necessary for service of and access to the Unencumbered Asset for the current or contemplated use thereof have been completed, are serviceable and all-weather and are physically and legally open for use by the public.
- (e) The Unencumbered Asset is served by public water and sewer systems or, if the Unencumbered Asset is not serviced by a public water and sewer system, such alternate systems are adequate and meet, in all material respects, all requirements and regulations of, and otherwise complies in all material respects with, all Applicable Laws with respect to such alternate systems.
- (f) The Borrower is not aware of any latent or patent structural or other significant deficiency of the Unencumbered Asset. The Unencumbered Asset is free of damage and waste that would materially and adversely affect the value of the Unencumbered Asset, is in good repair and there is no deferred maintenance other than ordinary wear and tear. The Unencumbered Asset is free from damage caused by fire or other casualty. There is no pending or, to the actual knowledge of the Borrower threatened condemnation proceedings affecting the Unencumbered Asset, or any material part thereof.
- (g) To the Borrower's knowledge, all liquid and solid waste disposal, septic and sewer systems located on the Unencumbered Asset are in a good and safe condition and repair and to the Borrower's knowledge, in material compliance with all Applicable Laws with respect to such systems.
- (h) All improvements on the Unencumbered Asset lie within the boundaries and building restrictions of the legal description of record of the Unencumbered Asset, no such improvements encroach upon easements benefiting the Unencumbered Asset other than encroachments that do not materially adversely affect the use or occupancy of the Unencumbered Asset and no improvements on adjoining properties encroach upon the Unencumbered Asset or easements benefiting the Unencumbered Asset other than encroachments that do not materially adversely affect the use or occupancy of the Unencumbered Asset. All amenities, access routes or other items that materially benefit the Unencumbered Asset are under direct control of the Borrower, constitute permanent easements that benefit all or part of the Unencumbered Asset or are public property, and the Unencumbered Asset, by virtue of such easements or otherwise, is contiguous to a physically open, dedicated all weather public street, and has the necessary permits for ingress and egress.
- (i) There are no delinquent taxes, ground rents, water charges, sewer rents, assessments, insurance premiums, leasehold payments, or other outstanding charges affecting

the Unencumbered Asset except to the extent such items are being contested in good faith and as to which adequate reserves have been provided.

(j) The Unencumbered Asset satisfies each of the requirements for an Unencumbered Asset as set forth in the definition thereof.

A breach of any of the representations and warranties contained in this <u>Section 6.26</u> with respect to a Project shall disqualify such Project from being an Unencumbered Asset for so long as such breach continues (unless otherwise approved by the Required Lenders) but shall not constitute a Default (unless the elimination of such Property as an Unencumbered Asset results in a Default under one of the other provisions of this Agreement).

6.27. Anti-Corruption Laws and Sanctions(a) .

None of the Borrower or any of its Subsidiaries, nor, to the knowledge of the Borrower, any of the directors, officers, employees or agents of the Borrower or of any of its Subsidiaries (i) is a Sanctioned Person, or (ii) to the knowledge of the Borrower, except as may be disclosed by the Borrower to the Administrative Agent from time to time, derives any of its assets or operating income from investments in or transactions with any Sanctioned Person. The Borrower and its Subsidiaries and, to the knowledge of the Borrower, their respective officers, directors, employees and agents, are in compliance with Anti-Corruption Laws and applicable Sanctions in all material respects. No Loan, use of the proceeds of any Loan, or other transactions contemplated hereby will violate (x) Anti-Corruption Laws or (y) any Sanctions applicable to any party hereto. Neither the making of the Loans nor the use of the proceeds thereof will violate the Patriot Act, the Trading with the Enemy Act, as amended, or any of the foreign assets control regulations of the United States Treasury Department (31 C.F.R., Subtitle B, Chapter V, as amended) or any enabling legislation or executive order relating thereto or successor statute thereto. The Borrower and its Subsidiaries are in compliance in all material respects with the Patriot Act.

The Borrower agrees that all of its representations and warranties set forth in Article VI of this Agreement and elsewhere in this Agreement are true as of the Agreement Execution Date, and will be true on each Effective Date in all material respects (except with respect to matters which have been disclosed in writing to and approved by the Required Lenders), and will be true in all material respects (except with respect to matters which have been disclosed in writing to and approved by the Required Lenders) upon the date of each request for disbursement of a Borrowing, provided that the Borrower shall only be obligated to update any Schedules referred to in this Article VI and the financial statements required under Section 8.2(i) on a quarterly basis, unless any change otherwise required to be disclosed could reasonably be expected to have a Material Adverse Effect. Each request for disbursement hereunder shall constitute a reaffirmation of such representations and warranties as deemed modified in accordance with the disclosures made and approved, as aforesaid, as of the date of such request and disbursement.

6.28. <u>Affected Financial Institutions</u>. No Loan Party is an Affected Financial Institution.

ARTICLE VII.

ADDITIONAL REPRESENTATIONS AND WARRANTIES

The General Partner hereby represents and warrants to the Lenders that:

- 7.1. Existence. The General Partner is a corporation duly organized and existing under the laws of the State of Maryland, with its principal place of business in the State of Illinois, is duly qualified as a foreign corporation and properly licensed (if required) and in good standing in each jurisdiction where the failure to qualify or be licensed (if required) would constitute a Material Adverse Financial Change with respect to the General Partner or have a Material Adverse Effect on the business or properties of the General Partner.
- 7.2. <u>Corporate Powers</u>. The execution, delivery and performance of the Loan Documents required to be delivered by the General Partner hereunder are within the corporate powers of the General Partner, have been duly authorized by all requisite corporate action, and are not in conflict with the terms of any organizational instruments of the General Partner, or any instrument or agreement to which the General Partner is a party or by which the General Partner or any of its assets is bound or affected.
- 7.3. <u>Power of Officers</u>. The officers of the General Partner executing the Loan Documents required to be delivered by the General Partner hereunder have been duly elected or appointed and were fully authorized to execute the same at the time each such agreement, certificate or instrument was executed.
- 7.4. <u>Government and Other Approvals</u>. No approval, consent, exemption or other action by, or notice to or filing with, any governmental authority is necessary in connection with the execution, delivery or performance of the Loan Documents required hereunder.
- 7.5. <u>Compliance With Laws</u>. There is no judgment, decree or order or any law, rule or regulation of any court or governmental authority binding on the General Partner which would be contravened by the execution, delivery or performance of the Loan Documents required hereunder.
- 7.6. <u>Enforceability of Agreement</u>. This Agreement is the legal, valid and binding agreement of the General Partner, as the general partner of the Borrower, enforceable against the General Partner in accordance with its respective terms, and the Loan Documents required hereunder, when executed and delivered, will be similarly legal, valid, binding and enforceable except to the extent that such enforcement may be limited by applicable bankruptcy, insolvency, reorganization or other similar laws affecting the rights of creditors generally.

- 7.7. <u>Liens; Consents</u>. The execution, delivery or performance of the Loan Documents required to be delivered by the General Partner hereunder will not result in the creation of any Lien on the Properties other than in favor of the Lenders. No consent to the transactions hereunder is required from any ground lessor or mortgagee or beneficiary under a deed of trust or any other party except as has been delivered to the Lenders.
- 7.8. <u>Litigation</u>. There are no suits, arbitrations, claims, disputes or other proceedings (including, without limitation, any civil, criminal, administrative or environmental proceedings), pending or, to the best of the General Partner's knowledge, threatened against or affecting the General Partner or any of the Properties, the adverse determination of which individually or in the aggregate would have a Material Adverse Effect on the General Partner and/or would cause a Material Adverse Financial Change with respect to the General Partner or materially impair the General Partner's ability to perform its obligations hereunder or under any instrument or agreement required hereunder, except as disclosed on <u>Schedule 7.8</u> hereto, or otherwise disclosed to Lenders in accordance with the terms hereof.
- 7.9. <u>Events of Default</u>. No Default or Event of Default has occurred and is continuing or would result from the incurring of obligations by the General Partner under any of the Loan Documents or any other document to which the General Partner is a party.
- 7.10. <u>Investment Company Act of 1940</u>. The General Partner is not, and will by such acts as may be necessary continue not to be, an investment company within the meaning of the Investment Company Act of 1940.
- 7.11. <u>Public Utility Holding Company Act</u>. The General Partner is not a "holding company" or a "subsidiary company" of a "holding company," or an "affiliate" of a "holding company," or of a "subsidiary company" of a "holding company," within the definitions of the Public Utility Holding Company Act of 1935, as amended.
- 7.12. <u>No Material Adverse Financial Change</u>. There has been no Material Adverse Financial Change in the condition of the General Partner since the last date on which the financial and/or operating statements were submitted to the Lenders.
- 7.13. Financial Information. All financial statements furnished to the Lenders by or on behalf of the General Partner and all other financial information and data furnished by or on behalf of the General Partner to the Lenders are complete and correct in all material respects as of the date thereof, and such financial statements have been prepared in accordance with GAAP and fairly present the consolidated financial condition and results of operations of the General Partner as of such date. The General Partner has no contingent obligations, liabilities for taxes or other outstanding financial obligations which are material in the aggregate, except as disclosed in such statements, information and data.
- 7.14. <u>Factual Information</u>. All factual information heretofore or contemporaneously furnished by or on behalf of the General Partner to the Lenders for purposes of or in connection with this Agreement and the other Loan Documents and the transactions contemplated therein is, and all other such factual information hereafter furnished by or on behalf of the General Partner to the Lenders will be, true and accurate in all material respects (taken as a whole) on the date as of which

such information is dated or certified and not incomplete by omitting to state any material fact necessary to make such information (taken as a whole) not misleading at such time.

- 7.15. <u>ERISA</u>. (i) The General Partner is not an entity deemed to hold "plan assets" within the meaning of ERISA or any regulations promulgated thereunder of an employee benefit plan (as defined in Section 3(3) of ERISA) which is subject to Title I of ERISA or any plan within the meaning of Section 4975 of the Code, and (ii) the execution of this Agreement and the transactions contemplated hereunder do not give rise to a prohibited transaction within the meaning of Section 406 of ERISA or Section 4975 of the Code.
- 7.16. <u>Taxes</u>. All required Federal, state and other material tax returns have been filed by the General Partner with the appropriate authorities except (i) to the extent that extensions of time to file have been requested, granted and have not expired and (ii) to the extent such taxes are being contested in good faith and for which adequate reserves, in accordance with GAAP, are being maintained.
- 7.17. No Brokers. The General Partner has dealt with no brokers in connection with this Facility, and no brokerage fees or commissions are payable by or to any Person in connection with this Agreement or the Borrowings. Lenders shall not be responsible for the payment of any fees or commissions to any broker and the General Partner shall indemnify, defend and hold Lender harmless from and against any claims, liabilities, obligations, damages, costs and expenses (including reasonable attorneys' fees and disbursements) made against or incurred by Lender as a result of claims made or actions instituted by any broker or Person claiming by, through or under the General Partner in connection with the Facility.
- 7.18. <u>Subsidiaries</u>. <u>Schedule 7.18</u> hereto contains an accurate list of all of the presently existing Subsidiaries of the General Partner, setting forth their respective jurisdictions of formation and the percentage of their respective Capital Stock owned by it or its Subsidiaries. All of the issued and outstanding shares of Capital Stock of such Subsidiaries have been duly authorized and issued and are fully paid and non-assessable.
- 7.19. <u>Status</u>. General Partner is a corporation listed and in good standing on a national securities exchange and is currently qualified as a real estate investment trust under the Code.
- 7.20. Anti-Corruption Laws and Sanctions. None of the General Partner, the Borrower, any of the other Subsidiaries of the General Partner, or, to the knowledge of the General Partner, any other Affiliate of the General Partner or any of their respective directors, officers, employees or agents: (i) is a Sanctioned Person; or (ii) to the knowledge of the General Partner, except as may be disclosed by the General Partner to the Administrative Agent from time to time, derives any of its assets or operating income from investments in or transactions with any Sanctioned Person. The General Partner and its Affiliates and, to the knowledge of the General Partner, their respective officers, directors, employees and agents, are in compliance with Anti-Corruption Laws and applicable Sanctions in all material respects. No Loan, use of the proceeds of any Loan, or other transactions contemplated hereby will violate (x) Anti-Corruption Laws or (y) any Sanctions applicable to any party hereto. Neither the making of the Loans nor the use of the proceeds thereof will violate the Patriot Act, the Trading with the Enemy Act, as amended, or any of the foreign assets control regulations of the United States Treasury Department (31 C.F.R., Subtitle B, Chapter V, as

amended) or any enabling legislation or executive order relating thereto or successor statute thereto. The Borrower and its Subsidiaries are in compliance in all material respects with the Patriot Act.

The General Partner agrees that all of its representations and warranties set forth in Article VII of this Agreement and elsewhere in this Agreement are true as of the Agreement Execution Date, and will be true on each Effective Date in all material respects (except with respect to matters which have been disclosed in writing to and approved by the Required Lenders), and will be true in all material respects (except with respect to matters which have been disclosed in writing to and approved by the Required Lenders) upon each request for disbursement of a Borrowing, provided that the General Partner shall only be obligated to update any Schedules referred to in this Article VII and the financial statements required under Section 8.2(i) on a quarterly basis, unless any change otherwise required to be disclosed could reasonably be expected to have a Material Adverse Effect. Each request for disbursement hereunder shall constitute a reaffirmation of such representations and warranties as deemed modified in accordance with the disclosures made and approved, as aforesaid, as of the date of such request and disbursement.

ARTICLE VIII.

AFFIRMATIVE COVENANTS

The Borrower (and the General Partner, if expressly included in Sections contained in this Article) covenant and agree that so long as the Term Loan Commitment of any Lender shall remain available and until the full and final payment of all Obligations incurred under the Loan Documents each will:

- 8.1. Notices. Promptly give written notice to Administrative Agent (who will promptly send such notice to Lenders) of:
- (a) all litigation or arbitration proceedings affecting the Borrower, the General Partner or any Subsidiary where the amount claimed is \$50,000,000 or more;
- (b) any Default or Event of Default, specifying the nature and the period of existence thereof and what action has been taken or been proposed to be taken with respect thereto;
- (c) all claims filed against any property owned by the Borrower or the General Partner which, if adversely determined, could have a Material Adverse Effect on the ability of the Borrower or the General Partner to meet any of their obligations under the Loan Documents;
- (d) the occurrence of any other event which might have a Material Adverse Effect or cause a Material Adverse Financial Change on or with respect to the Borrower or the General Partner;
- (e) any Reportable Event or any "prohibited transaction" (as such term is defined in Section 4975 of the Code) in connection with any Plan or any trust created thereunder, which may, singly or in the aggregate materially impair the ability of the Borrower or the General Partner to repay any of its obligations under the Loan Documents, describing the

nature of each such event and the action, if any, the Borrower or the General Partner, as the case may be, proposes to take with respect thereto;

- (f) any notice from any Federal, state, local or foreign authority regarding any Hazardous Material, asbestos, or other environmental condition, proceeding, order, claim or violation affecting any of the Properties where such condition, proceeding, order, claim or violation would reasonably be expected to result in a liability to the Borrower, the General Partner or any Subsidiary in an amount, individually or in the aggregate, of \$50,000,000 or more.
- 8.2. <u>Financial Statements</u>, <u>Reports</u>, <u>Etc.</u>. The Borrower and the General Partner each shall maintain, for itself and each Subsidiary, a system of accounting established and administered in accordance with GAAP, and shall furnish to the Administrative Agent (and the Administrative Agent shall thereafter promptly furnish to the Lenders):
 - (i) quarterly financial statements (including a balance sheet income statement, and cash flow statement) and related reports in form and substance satisfactory to the Lenders not later than forty-five (45) days after the end of each of the first three fiscal quarters (commencing with the fiscal quarter ending June 30, 2021), and not later than ninety (90) days after the end of each fiscal year (commencing with the fiscal year ending December 31, 2021), annual audited financial statements, audited by an accounting firm as reasonably approved by Administrative Agent (which audit report shall be without a "going concern" or like qualification or exception), provided, however, Administrative Agent shall only have the right to approve such accounting firm if the accounting firm is not a big 4 accounting firm, all certified by the Borrower's or the General Partner's, as applicable, chief financial officer, chief accounting officer or controller, calculation of the financial covenants described below, a description of Unencumbered Assets, a listing of capital expenditures (in the level of detail as currently disclosed in the Borrower's "Supplemental Information"), a report listing and describing all newly acquired Properties, including their cash flow, cost and secured or unsecured Indebtedness assumed in connection with such acquisition, if any, summary Property information for all Properties, including, without limitation, their Property Operating Income, occupancy rates, square footage, property type and date acquired or built, and such other information as may be requested to evaluate the quarterly compliance certificate delivered as provided below;
 - (ii) copies of all Form 10-Ks, 10-Qs, 8-Ks, and any other public information filed with the Securities Exchange Commission by the Borrower or the General Partner once a quarter simultaneously with delivering the compliance certificate described below, along with any other materials distributed to the shareholders of the General Partner or the partners of the Borrower from time to time, including a copy of the General Partner's annual report; provided that, to the extent any of such reports contains information required under the other subsections of this Section 8.2, the information need not be furnished separately under the other subsections; provided, further that the Borrower may comply with this Section 8.2(ii) by posting or causing to be posted, the foregoing information on either the Securities

Exchange Commission public website or on the Borrower's or the General Partner's public website, so long as the Lenders have access to and are timely referred to any such website by the Borrower;

- (iii) not later than forty-five (45) days after the end of the first three fiscal quarters (commencing with the fiscal quarter ending June 30, 2021), and not later than ninety (90) days after the end of the fiscal year (commencing with the fiscal year ending December 31, 2021), a report certified by the entity's chief financial officer, chief accounting officer or controller, containing Property Operating Income from individual properties owned by the Borrower or a Wholly-Owned Subsidiary and included as Unencumbered Assets;
- (iv) Not later than forty-five (45) days after the end of each of the first three fiscal quarters (commencing with the fiscal quarter ending June 30, 2021), and not later than ninety (90) days after the end of the fiscal year (commencing with the fiscal year ending December 31, 2021), a compliance certificate in substantially the form of Exhibit H hereto signed by the Borrower's chief financial officer, chief accounting officer or controller confirming that the Borrower is in compliance with all of the covenants of the Loan Documents, showing the calculations and computations necessary to determine compliance with the financial covenants contained in this Agreement (including such schedules and backup information as may be necessary to demonstrate such compliance) and stating that to such officer's best knowledge, there is no other Default or Event of Default exists, or if any Default or Event of Default exists, stating the nature and status thereof;
- (v) As soon as possible and in any event within ten (10) Business Days after the Borrower knows that any Reportable Event has occurred with respect to any Plan, a statement, signed by the chief financial officer of the Borrower, describing said Reportable Event and within twenty (20) days after such Reportable Event, a statement signed by such chief financial officer describing the action which the Borrower proposes to take with respect thereto; and (b) within ten (10) Business Days of receipt, any notice from the Internal Revenue Service, PBGC or Department of Labor with respect to a Plan regarding any excise tax, proposed termination of a Plan, prohibited transaction or fiduciary violation under ERISA or the Code which could result in any liability to the Borrower or any member of the Controlled Group in excess of \$100,000; and (c) within ten (10) Business Days of filing, any Form 5500 filed by the Borrower with respect to a Plan, or any member of the Controlled Group which includes a qualified accountant's opinion;
- (vi) As soon as possible and in any event within thirty (30) days after receipt by the Borrower, a copy of (a) any notice or claim to the effect that the Borrower or any of its Subsidiaries is or may be liable to any Person as a result of the release by such entity, or any of its Subsidiaries, or any other Person of any toxic or hazardous waste or substance into the environment, and (b) any notice alleging any violation of any federal, state or local environmental, health or safety law or regulation by the Borrower or any of its Subsidiaries or Investment Affiliates, which, in either case, could be reasonably likely to have a Material Adverse Effect;

(vii) [Reserved];

- (viii) Promptly upon the furnishing thereof to the shareholders of the Borrower, copies of all financial statements, reports and proxy statements so furnished; provided, further that the Borrower may comply with this Section 8.2(viii) by posting or having posted the foregoing information on either the Securities Exchange Commission public website or on the Borrower's or the General Partner's public website, so long as the Lenders have access to and are timely referred to any such website by the Borrower;
- (ix) Promptly upon the distribution thereof to the press or the public, copies of all press releases; provided, further that the Borrower may comply with this <u>Section 8.2(ix)</u> by posting or having posted the foregoing information on either the Securities Exchange Commission public website or on the Borrower's or the General Partner's public website, so long as the Lenders have access to and are timely referred to any such website by the Borrower;
- (x) Promptly upon receipt thereof, notices with respect to the ratings for the Borrower's or the General Partner's long-term, senior unsecured debt, the effect of which may be to change the Base Rate Applicable Margin and/or the LIBOR Applicable Margin;
- (xi) As soon as possible, and in any event within ten (10) days after the Borrower knows of any fire or other casualty or any pending or threatened condemnation or eminent domain proceeding with respect to all or any material portion of any Unencumbered Asset, a statement signed by the Chief Financial Officer of the Borrower, describing such fire, casualty or condemnation and the action the Borrower intends to take with respect thereto; and
- (xii) Such other information (including, without limitation, nonfinancial information) as the Administrative Agent or any Lender may from time to time reasonably request, including, promptly upon the reasonable request of the Administrative Agent or any Lender, to the Administrative Agent or directly to such Lender, as the case may be, any information or documentation requested by it for purposes of complying with the Beneficial Ownership Regulation.
- 8.3. Existence and Conduct of Operations. Except as permitted herein, maintain and preserve its existence and all rights, privileges and franchises now enjoyed and necessary for the operation of its business, including remaining in good standing in each jurisdiction in which business is currently operated. The Borrower and the General Partner shall carry on and conduct their respective businesses in substantially the same manner and in substantially the same fields of enterprise as presently conducted. The Borrower will do, and will cause each of its Subsidiaries to do, all things necessary to remain duly incorporated and/or duly qualified, validly existing and in good standing as a real estate investment trust, corporation, general partnership, limited liability company or limited partnership, as the case may be, in its jurisdiction of incorporation/formation, except and to the extent that the Borrower determines that causing any of its Subsidiaries (other than any Loan Party) to remain duly incorporated and/or duly qualified, validly existing and in good

standing is not in the best interests of the Borrower and such failure to remain duly incorporated and/or qualified, validly existing and in good standing could not reasonably be expected to have a Material Adverse Effect. The Borrower will maintain all requisite authority to conduct its business in each jurisdiction in which the Properties are located and, except where the failure to be so qualified would not have a Material Adverse Effect, in each jurisdiction required to carry on and conduct its businesses in substantially the same manner as it is presently conducted, and, specifically, neither the Borrower nor its Subsidiaries will undertake any business other than the acquisition, development, ownership, management, operation and leasing of industrial/warehouse properties and business activities ancillary, incidental or otherwise reasonably related thereto.

- 8.4. <u>Maintenance of Properties</u>. Maintain, preserve, protect and keep the Properties in good repair, working order and condition, and make all necessary and proper repairs, renewals and replacements, normal wear and tear excepted.
- 8.5. <u>Insurance</u>. Upon request of the Administrative Agent, provide a certificate of insurance from all insurance carriers who maintain policies with respect to the Properties within thirty (30) days after the end of each fiscal year, evidencing that the insurance required to be furnished to Lenders pursuant to <u>Section 6.20</u> hereof is in full force and effect. The Borrower shall timely pay, or cause to be paid, all premiums on all insurance policies required under this Agreement from time to time. The Borrower shall promptly notify its insurance carrier or agent therefor (with a copy of such notification being provided simultaneously to Administrative Agent) if there is any occurrence which, under the terms of any insurance policy then in effect with respect to the Properties, requires such notification.
- 8.6. <u>Payment of Obligations</u>. Pay and discharge when due (i) all Federal, state and other material taxes, assessments, governmental charges and (ii) all other obligations which, if unpaid, would by law become a Lien upon its property, except, in the case of each of clauses (i) and (ii), such as may be contested in good faith or as to which a bona fide dispute may exist, and for which adequate reserves have been provided in accordance with sound accounting principles used by the Borrower on the date hereof.
- 8.7. <u>Compliance with Laws</u>. Comply in all material respects with all Applicable Laws, rules, regulations, orders and directions (including, without limitation, Anti-Corruption Laws and Sanctions) of any governmental authority having jurisdiction over the Borrower, the General Partner, or any of their respective businesses.
- 8.8. <u>Adequate Books</u>. Maintain adequate books, accounts and records in order to provide financial statements in accordance with GAAP and, if requested by any Lender, permit employees or representatives of such Lender at any reasonable time and upon reasonable notice to inspect and audit the properties of the Borrower and of the Consolidated Operating Partnership, and to examine or audit the inventory, books, accounts and records of each of them and make copies and memoranda thereof.

- 8.9. <u>ERISA</u>. Comply in all material respects with all requirements of ERISA applicable to it with respect to each Plan.
- 8.10. <u>Maintenance of Status</u>. The General Partner shall at all times (i) remain as a corporation listed and in good standing on a national securities exchange, and (ii) take all steps to maintain the General Partner's status as a real estate investment trust in compliance with all applicable provisions of the Code.
- 8.11. <u>Use of Proceeds</u>. Use the proceeds of the Facility to re-evidence Indebtedness outstanding under the Existing Term Loan Agreement and for the general business purposes of the Borrower, including without limitation working capital needs, closing costs, interim funding for property acquisitions and construction of new industrial properties, and/or payment of other debts and obligations of the Borrower.
- 8.12. <u>Distributions</u>. Provided there is no declared Monetary Default then existing and provided there is not an Event of Default relating to a breach of the financial covenants contained in <u>Section 9.7</u> below, the Borrower may make distributions to the General Partner and the General Partner may make distributions to its shareholders. Notwithstanding the foregoing, the Borrower shall be permitted to make distributions to the General Partner and the General Partner shall be permitted at all times to make distributions of whatever amount is necessary to maintain its tax status as a real estate investment trust.
- 8.13. <u>Guarantors</u>. To the extent any Person executes and delivers a guaranty of, or otherwise becomes obligated in respect of, the Indebtedness of the Parent or the Borrower under the Existing Revolving Credit Agreement, the Borrower shall, within five (5) Business Days (or such longer period as the Administrative Agent may reasonably determine) cause such Person to deliver to the Administrative Agent each of the following in form and substance satisfactory to the Administrative Agent: (a) a Guaranty executed by such Person substantially similar to the form of guaranty delivered by such Person with respect to the Existing Revolving Credit Agreement and (b) to the extent reasonably requested by the Administrative Agent, supporting documentation comparable in scope and type to any supporting documentation delivered by such Person with respect to the Existing Revolving Credit Agreement.

ARTICLE IX.

NEGATIVE COVENANTS

The Borrower covenants and agrees that, so long as the Term Loan Commitment shall remain available and until full and final payment of all Obligations incurred under the Loan Documents, without the prior written consent of either all of the Lenders pursuant to Section 14.13(b)(iii) or the consent of the Required Lenders in all other cases, it will not, and the General Partner will not and, in the case of Sections 9.9, the Borrower's Subsidiaries will not:

- 9.1. <u>Change in Business</u>. Engage in any business if, as a result, the general nature of the business in which the Borrower and the General Partner, taken as a whole, would then be engaged would be substantially changed from the general nature of the business in which the Borrower and the General Partner, taken as a whole, are engaged on the Closing Date.
- 9.2. <u>Change of Management of Properties</u>. Change the management of the Properties, except that (a) any Affiliate of the Borrower or the General Partner shall be permitted to manage any of the Properties and (b) the Borrower, the General Partner or any of their Subsidiaries may engage and/or replace third-party property managers with respect to any Property, provided that the aggregate book value of all Properties at any one time under third-party management shall not exceed 30% of the aggregate book value of all of the Properties at such time.
- 9.3. <u>Change of Borrower Ownership</u>. Without the consent of the Required Lenders, allow (i) the General Partner to own less than fifty-one percent (51%) of the partnership interests in the Borrower, (ii) the Borrower to be controlled by a Person other than the General Partner, (iii) a Change in Control to occur, or (iv) any pledge of, other encumbrance on, or conversion to limited partnership interests of, any of the general partnership interests in the Borrower. If any of the foregoing occurs with the consent of the Required Lenders, each non-consenting Lender shall have the option to demand repayment of any outstanding Obligations due to such Lender prior to or concurrent with the occurrence of any of the foregoing events. Upon any Lender's demand in accordance with the foregoing provision, the Borrower shall repay any outstanding Obligations due to such Lender prior to or concurrently with the occurrence of any of the foregoing events.
- 9.4. <u>Use of Proceeds</u>. Apply or permit to be applied any proceeds of any Borrowing directly or indirectly, to the funding of any purchase of, or offer for, any share of capital stock of any publicly held corporation unless the board of directors of such corporation has consented to such offer prior to any public announcements relating thereto and all of the Lenders have consented to such use of the proceeds of the Facility. The Borrower shall not request any Loan, shall not use, and shall ensure that its Subsidiaries and its or their respective directors, officers, employees and agents do not use the proceeds of any Loan (i) in furtherance of an offer, payment, promise to pay, or authorization of the payment or giving of money, or anything else of value, to any Person in violation of any Anti-Corruption Laws, (ii) for the purpose of funding, financing or facilitating any activities, business or transaction of or with any Sanctioned Person to the extent such activities, business or transaction would be prohibited by Sanctions if conducted by a corporation incorporated in the United States or in any European Union member state or (iii) in any manner that would result in the violation of any Sanctions applicable to any party hereto.

- 9.5. <u>Liens</u>. Create, incur, or suffer to exist (or permit any of its Subsidiaries to create, incur, or suffer to exist) any Lien in, of or on the Property of any member of the Consolidated Operating Partnership other than:
 - (i) Liens for taxes, assessments or governmental charges or levies on their Property if the same shall not at the time be delinquent or thereafter can be paid without penalty, or are being contested in good faith and by appropriate proceedings and for which adequate reserves shall have been set aside on their books;
 - (ii) Liens which arise by operation of law, such as carriers', warehousemen's, landlords', materialmen and mechanics' liens and other similar liens arising in the ordinary course of business which secure payment of obligations not more than thirty (30) days past due or which are being contested in good faith by appropriate proceedings and for which adequate reserves shall have been set aside on its books;
 - (iii) Liens arising out of pledges or deposits under worker's compensation laws, unemployment insurance, old age pensions, or other social security or retirement benefits, or similar legislation;
 - (iv) Utility easements, building restrictions, zoning restrictions, easements and such other encumbrances or charges against real property as are of a nature generally existing with respect to properties of a similar character and which do not in any material way affect the marketability of the same or interfere with the use thereof in the business of the Borrower or its Subsidiaries;
 - (v) Liens of any Subsidiary in favor of the Borrower or the General Partner; and
 - (vi) So long as immediately prior to the creation, assumption or incurring of such Lien, and immediately thereafter, no Default or Event of Default would be in existence, Liens on Properties securing Indebtedness not prohibited hereunder.

Liens permitted pursuant to this <u>Section 9.5</u> shall be deemed to be "Permitted Liens".

9.6. <u>Regulation U</u>. Use the proceeds of the Loans for any other purpose than as provided for in <u>Section 8.11</u> and <u>Section 9.4</u> hereof. No part of the proceeds of any Loan will be used, whether directly or indirectly for any purpose that entails a violation of any of the regulations of the Board of Governors of the Federal Reserve System, including Regulations T, U and X. The proceeds will not be used in a manner which would cause the Facility to be treated as a "Purpose Credit."

9.7. Indebtedness and Cash Flow Covenants. Permit or suffer:

- (a) as of the last day of any fiscal quarter, the ratio of (A) the sum of (1) EBITDA of the Consolidated Operating Partnership plus (2) interest income (other than any interest income from assets being used to support Defeased Debt) to (B) the sum of (1) Debt Service plus, without duplication, (2) all payments on account of preferred stock or preferred partnership units of any member of the Consolidated Operating Partnership for such quarter plus (3) all Ground Lease Payments due from any member of the Consolidated Operating Partnership to the extent not deducted as an expense in calculating EBITDA of the Consolidated Operating Partnership (the "Fixed Charge Coverage Ratio"), to be less than (i) 1.50 to 1.0 from the Agreement Execution Date through the Maturity Date, with all such calculations in clauses (A) and (B) above based on the results of the four (4) consecutive fiscal quarters then ended;
- (b) as of the last day of any fiscal quarter, the Consolidated Leverage Ratio to exceed sixty percent (60%); provided that, if any Material Acquisition shall occur then, at the election of the Borrower upon delivery of prior written notice to the Administrative Agent, concurrently with or prior to the delivery of a Compliance Certificate pursuant to Section 8.2(iv), and provided that no Default or Event of Default shall have occurred and be continuing (other than as a result of the Consolidated Leverage Ratio as of the end of such fiscal quarter being greater than sixty percent (60%) but less than or equal to sixty-five percent (65%)), the Consolidated Leverage Ratio shall be increased to sixty-five percent (65%) for such fiscal quarter and the next three succeeding fiscal quarters (any period with such increase a "Consolidated Leverage Ratio Increase Period"), provided that (i) no more than two (2) Consolidated Leverage Ratio Increase Periods may be elected by the Borrower during the term of this Agreement and (ii) any such Consolidated Leverage Ratio Increase Periods shall be non-consecutive;
- (c) as of the last day of any fiscal quarter, the ratio of outstanding Consolidated Senior Unsecured Debt to Value of Unencumbered Assets (such ratio, the "<u>Unencumbered Leverage Ratio</u>") to exceed sixty percent (60%); provided that, if any Material Acquisition shall occur then, at the election of the Borrower upon delivery of prior written notice to the Administrative Agent, concurrently with or prior to the delivery of a Compliance Certificate pursuant to <u>Section 8.2(iv)</u>, and provided that no Default or Event of Default shall have occurred and be continuing (other than as a result of the Unencumbered Leverage Ratio as of the end of such fiscal quarter being greater than sixty percent (60%) but less than or equal to sixty-five percent (65%)), the Unencumbered Leverage Ratio shall be increased to sixty-five percent (65%) for such fiscal quarter and the next three succeeding fiscal quarters (any period with such increase, an "<u>Unencumbered Leverage Ratio Increase Period</u>"), provided further that (i) no more than two (2) Unencumbered Leverage Ratio Increase Periods may be elected by the Borrower during the term of this Agreement and (ii) any such Unencumbered Leverage Ratio Increase Periods shall be non-consecutive;
- (d) as of the last day of any fiscal quarter, Consolidated Secured Debt to exceed 40% of Implied Capitalization Value of the Consolidated Operating Partnership; or

(e) as of the last day of any fiscal quarter, the ratio of (A) Property Operating Income from Unencumbered Assets that are not Assets Under Development for such fiscal quarter to (B) interest expense on all Consolidated Senior Unsecured Debt for such fiscal quarter to be less than (x) 1.75 to 1.0 from the Agreement Execution Date through the Maturity Date.

The foregoing covenants set forth in paragraphs (c) and (e) above shall be tested at the end of each fiscal quarter (for the applicable reporting period), on a pro-forma basis at the time of each Borrowing under the Facility (using the latest quarterly financial statements then available and taking into account the proposed Borrowing), and on a pro-forma basis upon any Asset Sale or incurrence of any Indebtedness by the Consolidated Operating Partnership (using the latest quarterly financial statements then available and taking into account the proposed incurrence of Indebtedness). To the extent the Consolidated Operating Partnership has Defeased Debt, both the underlying debt and interest payable thereon and the financial assets used to defease such debt and interest earned thereon shall be excluded from calculations of the foregoing financial covenants. All financial computations required under this <u>Section 9.7</u> shall be made in accordance with GAAP as in effect on the Agreement Execution Date.

- 9.8. <u>Change of Control</u>; <u>Mergers and Dispositions</u>. Enter into any merger, consolidation, reorganization, division or liquidation or transfer or otherwise dispose of all or a substantial portion of its properties, except for: such transactions that occur between wholly owned Subsidiaries; transactions where the Borrower and the General Partner are the surviving entities and there is no change in business conducted, loss of an Investment Grade Rating or Change in Control, and no Default or Event of Default under the Loan Documents results from such transaction; or as otherwise approved in advance by the Lenders. To the extent not reported on General Partner's Form 8-K filings, the Borrower will notify the Administrative Agent (who will promptly notify Lenders) of any single event acquisition (whether individually or consummated as a series of related acquisitions pursuant to documentation closed concurrently), disposition, merger, division or asset purchase involving assets valued in excess of 25% of the Consolidated Operating Partnership's then current Market Value Net Worth and certify compliance with covenants after giving effect to such proposed acquisition, disposition, merger, division or asset purchase.
- 9.9. <u>Negative Pledge</u>. The Borrower agrees that throughout the term of this Facility, no "negative pledge" on any Project then included in Unencumbered Assets restricting the Borrower's (or wholly owned Subsidiary's) right to sell or encumber such Project shall be given to any other lender or creditor other than a Permitted Negative Pledge or, if such a "negative pledge" is given, the Project affected shall be immediately excluded from Unencumbered Assets.

ARTICLE X.

DEFAULTS

The occurrence of any one or more of the following events shall constitute an Event of Default:

10.1. <u>Nonpayment of Principal</u>. The Borrower fails to pay any principal portion of the Obligations when due, whether on the Maturity Date or otherwise.

- 10.2. <u>Certain Covenants</u>. The Borrower, the General Partner and/or Consolidated Operating Partnership, as the case may be, is not in compliance with any one or more of <u>Sections 8.10</u>, <u>8.12</u>, <u>9.3</u>, <u>9.4</u>, <u>9.5</u>, <u>9.7</u>, <u>9.8</u>, or <u>9.9</u> hereof.
- 10.3. <u>Nonpayment of Interest and Other Obligations</u>. The Borrower fails to pay any interest or other portion of the Obligations, other than payments of principal, and such failure continues for a period of five (5) days after the date such payment is due.
- 10.4. <u>Cross Default</u>. Any monetary default occurs (after giving effect to any applicable cure period) under any other Indebtedness (which includes liability under Guaranties) of the Borrower or the General Partner, singly or in the aggregate, in excess of Seventy-Five Million Dollars (\$75,000,000), other than (i) Indebtedness arising from the purchase of personal property or the provision of services, the amount of which is being contested by the Borrower or (ii) Indebtedness which is "non recourse", i.e., which is not recoverable by the creditor thereof from the general assets of the Borrower, the General Partner or any of their Affiliates, but is limited to the proceeds of certain real estate, improvements and related personal property.
- 10.5. <u>Loan Documents</u>. Any Loan Document is not in full force and effect or a default has occurred and is continuing thereunder after giving effect to any cure or grace period in any such document.
- 10.6. <u>Representation or Warranty</u>. At any time or times hereafter any representation or warranty set forth in <u>Articles VI</u> or <u>VII</u> of this Agreement or in any other Loan Document or in any statement, report or certificate now or hereafter made by the Borrower or the General Partner to the Lenders or the Administrative Agent is not true and correct in any material respect.
- 10.7. <u>Covenants, Agreements and Other Conditions</u>. The Borrower or the General Partner fails to perform or observe any of the other covenants, agreements and conditions contained in <u>Articles VIII</u> and <u>IX</u> (except for <u>Sections 8.10, 8.12, 9.3, 9.4, 9.5, 9.7, 9.8, or <u>9.9</u> hereof) and elsewhere in this Agreement or any of the other Loan Documents in accordance with the terms hereof or thereof, not specifically referred to herein, and such Default continues unremedied for a period of thirty (30) days after written notice from Administrative Agent, <u>provided, however</u>, that if such Default is susceptible of cure but cannot by the use of reasonable efforts be cured within such thirty (30) day period, such Default shall not constitute an Event of Default under this <u>Section 10.7</u> so long as (i) the Borrower or the General Partner, as the case may be, has commenced a cure within such thirty day period and (ii) thereafter, the Borrower or the General Partner, as the case may be, is proceeding to cure such default continuously and diligently and in a manner reasonably satisfactory to Lenders and (iii) such default is cured not later than sixty (60) days after the expiration of such thirty (30) day period.</u>
 - 10.8. No Longer General Partner. The General Partner shall no longer be the sole general partner of the Borrower.
- 10.9. <u>Material Adverse Financial Change</u>. The Borrower or the General Partner has suffered a Material Adverse Financial Change or is Insolvent.

10.10. Bankruptcy.

- (a) The General Partner, the Borrower or one or more Subsidiaries in which the Equity Value of the Subsidiary or Subsidiaries is more than ten percent (10%), in the aggregate, of Implied Capitalization Value of the Consolidated Operating Partnership shall (i) have an order for relief entered with respect to it under the Federal bankruptcy laws as now or hereafter in effect, (ii) make an assignment for the benefit of creditors, (iii) apply for, seek, consent to, or acquiesce in, the appointment of a receiver, custodian, trustee, examiner, liquidator or similar official for it or any substantial portion of its Property, (iv) institute any proceeding seeking an order for relief under the Federal bankruptcy laws as now or hereafter in effect or seeking to adjudicate it as a bankrupt or insolvent, or seeking dissolution, winding up, liquidation, reorganization, arrangement, adjustment or composition of it or its debts under any law relating to bankruptcy, insolvency or reorganization or relief of debtors or fail to file an answer or other pleading denying the material allegations of any such proceeding filed against it, (v) take any corporate action to authorize or effect any of the foregoing actions set forth in this Section 10.10(a), (vi) fail to contest in good faith any appointment or proceeding described in Section 10.10(b) or (vii) not pay, or admit in writing its inability to pay, its debts generally as they become due. As used herein, the term "Equity Value" of a Subsidiary shall mean the lesser of: (1) Property Operating Income of such Subsidiary's Properties owned as of the Agreement Execution Date capitalized at a 6.5% rate, less any Indebtedness of such Subsidiary; or (2) the appraised value of any of such Subsidiary's Properties, less any Indebtedness of such Subsidiary.
- (b) A receiver, trustee, examiner, liquidator or similar official shall be appointed for (i) the General Partner, the Borrower or one or more Subsidiaries in which the Equity Value of the Subsidiary or Subsidiaries is more than ten percent (10%), in the aggregate, of Implied Capitalization Value of the Consolidated Operating Partnership or (ii) all or substantially all of the Consolidated Operating Partnership's Properties.
- (c) A proceeding described in Section 10.10(a)(iv) shall be instituted against the General Partner, the Borrower or one or more Subsidiaries in which the Equity Value of the Subsidiary or Subsidiaries is more than ten percent (10%), in the aggregate, of Implied Capitalization Value of the Consolidated Operating Partnership, and such proceeding continues undismissed or unstayed for a period of sixty (60) consecutive days.
- 10.11. <u>Legal Proceedings</u>. The Borrower or the General Partner is enjoined, restrained or in any way prevented by any court order or judgment or if a notice of lien, levy, or assessment is filed of record with respect to all or any part of the Properties by any governmental department, office or agency, which could materially adversely affect the performance of the obligations of such parties hereunder or under the Loan Documents, as the case may be, or if any proceeding is filed or commenced seeking to enjoin, restrain or in any way prevent the foregoing parties from conducting all or a substantial part of their respective business affairs and failure to vacate, stay, dismiss, set aside or remedy the same within ninety (90) days after the occurrence thereof.

- 10.12. <u>ERISA</u>. The Borrower or the General Partner is deemed to hold "plan assets" within the meaning of ERISA or any regulations promulgated thereunder of an employee benefit plan (as defined in Section 3(3) of ERISA) which is subject to Title I of ERISA or any plan (within the meaning of Section 4975 of the Code).
 - 10.13. Change in Control. A Change in Control has occurred.
- 10.14. <u>Failure to Satisfy Judgments</u>. The General Partner, the Borrower or any of its Subsidiaries shall fail within sixty (60) days to pay, bond or otherwise discharge any judgments or orders for the payment of money (other than with respect to any Indebtedness which is "non recourse", i.e., which is not recoverable by the creditor thereof from the general assets of the Borrower, the General Partner or any of their Affiliates, but is limited to the proceeds of certain real estate, improvements and related personal property) in an amount which, when added to all other judgments or orders outstanding against the General Partner, the Borrower or any Subsidiary would exceed Seventy-Five Million Dollars (\$75,000,000) in the aggregate, which have not been stayed on appeal or otherwise appropriately contested in good faith, unless the liability is insured against and the insurer has not challenged coverage of such liability.
- 10.15. <u>Environmental Remediation</u>. Failure to remediate (or pursue the remediation process with due diligence and good faith) within the time period required by law or governmental order, (or within a reasonable time in light of the nature of the problem if no specific time period is so established), environmental problems in violation of Applicable Law related to Properties of the Borrower and/or its Subsidiaries where the estimated cost of remediation is in the aggregate in excess of Seventy-Five Million Dollars (\$75,000,000), in each case after all administrative hearings and appeals have been concluded.

ARTICLE XI.

ACCELERATION, WAIVERS, AMENDMENTS AND REMEDIES

11.1. Acceleration. If any Event of Default described in Section 10.10 hereof occurs, or the Borrower or the General Partner becomes Insolvent, the Term Loan Commitments and obligation of the Lenders to make Borrowings hereunder shall automatically terminate and the Obligations shall immediately become due and payable without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by the Borrower. If any other Event of Default described in Article X hereof occurs and is continuing, the Administrative Agent, with the consent of the Required Lenders, may, and at the request of the Required Lenders shall, by notice to the Borrower, take either or both of the following actions, at the same or different times: (i) terminate the Term Loan Commitments, and thereupon the Term Loan Commitments shall terminate immediately, and (ii) declare the Loans then outstanding to be due and payable in whole (or in part, in which case any principal not so declared to be due and payable may thereafter be declared to be due and payable, together with accrued interest thereon and all fees and other Obligations of the Borrower accrued hereunder, shall become due and payable immediately, without presentment, demand, protest or other notice of any kind, all of which are hereby waived by the Borrower.

11.2. Preservation of Rights; Amendments. No delay or omission of the Lenders in exercising any right under the Loan Documents shall impair such right or be construed to be a waiver of any Default or an acquiescence therein, and the making of a Borrowing notwithstanding the existence of a Default or the inability of the Borrower to satisfy the conditions precedent to such Borrowing shall not constitute any waiver or acquiescence. Any single or partial exercise of any such right shall not preclude other or further exercise thereof or the exercise of any other right, and no waiver, amendment or other variation of the terms, conditions or provisions of the Loan Documents whatsoever shall be valid unless in writing signed by the Administrative Agent and the number of Lenders required hereunder and then only to the extent in such writing specifically set forth. All remedies contained in the Loan Documents or by law afforded shall be cumulative and all shall be available to the Lenders until the Obligations have been paid in full.

ARTICLE XII.

THE ADMINISTRATIVE AGENT

- 12.1. <u>Appointment</u>. Wells Fargo Bank, National Association is hereby appointed by each of the Lenders as its contractual representative (herein referred to as the "<u>Administrative Agent</u>") hereunder and under each other Loan Document, and each of the Lenders irrevocably authorizes the Administrative Agent to act as the contractual representative of such Lender with the rights and duties expressly set forth herein and in the other Loan Documents. The Administrative Agent agrees to act as such contractual representative upon the express conditions contained in this <u>Article XII</u>. Notwithstanding the use of the defined term "Administrative Agent," it is expressly understood and agreed that the Administrative Agent shall not have any fiduciary responsibilities to any Lender by reason of this Agreement or any other Loan Document and that the Administrative Agent is merely acting as the contractual representative of the Lenders with only those duties as are expressly set forth in this Agreement and the other Loan Documents. In its capacity as the Lenders' contractual representative, the Administrative Agent (i) does not hereby assume any fiduciary duties to any of the Lenders, (ii) is a "representative" of the Lenders within the meaning of the term "secured party" as defined in the Illinois Uniform Commercial Code and (iii) is acting as an independent contractor, the rights and duties of which are limited to those expressly set forth in this Agreement and the other Loan Documents. Each of the Lenders hereby agrees to assert no claim against the Administrative Agent on any agency theory or any other theory of liability for breach of fiduciary duty, all of which claims each Lender hereby waives.
- 12.2. <u>Powers</u>. The Administrative Agent shall have and may exercise such powers under the Loan Documents as are specifically delegated to the Administrative Agent by the terms of each thereof, together with such powers as are reasonably incidental thereto. The Administrative Agent shall have no implied duties to the Lenders, or any obligation to the Lenders to take any action thereunder except any action specifically provided by the Loan Documents to be taken by the Administrative Agent.
- 12.3. <u>General Immunity</u>. Neither the Administrative Agent (in its capacity as Administrative Agent) nor any of its directors, officers, agents or employees shall be liable to the Borrower, the Lenders or any Lender for any action taken or omitted to be taken by it or them hereunder or under any other Loan Document or in connection herewith or therewith, except for its

or their own Gross Negligence or willful misconduct, as determined by a court of competent jurisdiction by a final and non-appealable judgment.

- 12.4. No Responsibility for Loans, Recitals, etc.. Neither the Administrative Agent (in its capacity as Administrative Agent) nor any of its directors, officers, agents or employees shall be responsible for or have any duty to ascertain, inquire into, or verify (i) any statement, warranty or representation made in connection with any Loan Document or any borrowing hereunder; (ii) the performance or observance of any of the covenants or agreements of any obligor under any Loan Document; (iii) the satisfaction of any condition specified in Article V, except receipt of items required to be delivered to the Administrative Agent; or (iv) the validity, effectiveness or genuineness of any Loan Document or any other instrument or writing furnished in connection therewith. Except as expressly set forth herein, the Administrative Agent shall not have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Subsidiaries that is communicated to or obtained by the bank serving as Administrative Agent or any of its Affiliates in any capacity.
- 12.5. Action on Instructions of Lenders. The Administrative Agent shall in all cases be fully protected in acting, or in refraining from acting, hereunder and under any other Loan Document in accordance with written instructions signed by the Required Lenders or all Lenders, as the case may be, and such instructions and any action taken or failure to act pursuant thereto shall be binding on all of the Lenders and on all holders of Notes. The Administrative Agent shall be fully justified in failing or refusing to take any action hereunder and under any other Loan Document unless it shall be indemnified to its satisfaction by the Lenders pro rata against any and all liability, cost and expense that it may incur by reason of taking or continuing to take any such action.
- 12.6. <u>Employment of Administrative Agents and Counsel</u>. The Administrative Agent may execute any of its duties as Administrative Agent hereunder and under any other Loan Document by or through employees, agents, and attorneys in fact and shall not be answerable to the Lenders, except as to money or securities received by it or its authorized agents, for the default or misconduct of any such agents or attorneys in fact selected by it with reasonable care. The Administrative Agent shall be entitled to advice of counsel concerning all matters pertaining to the agency hereby created and its duties hereunder and under any other Loan Document. Without limitation of the extent of any liability any Lender may otherwise have pursuant to this Agreement, the Administrative Agent shall have no liability for any action taken or omitted in good faith as a result of advice of counsel.
- 12.7. <u>Reliance on Documents; Counsel</u>. The Administrative Agent shall be entitled to rely upon any Note, notice, consent, certificate, affidavit, letter, telegram, statement, paper or document believed by it to be genuine and correct and to have been signed or sent by the proper person or persons, and, in respect to legal matters, upon the opinion of outside counsel selected by the Administrative Agent.
- 12.8. <u>Administrative Agent's Reimbursement and Indemnification</u>. The Lenders agree to reimburse and indemnify the Administrative Agent ratably in accordance with their respective Percentages (i) for any amounts not reimbursed by the Borrower (and without limiting the obligation of the Borrower to do so) for which the Administrative Agent is entitled to reimbursement by the Borrower under the Loan Documents, (ii) for any other reasonable expenses incurred by the Administrative Agent on behalf of the Lenders, in connection with the preparation, execution,

delivery, administration and enforcement of the Loan Documents, if not paid by the Borrower (and without limiting the obligation of the Borrower to do so), and (iii) for any liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind and nature whatsoever which may be imposed on, incurred by or asserted against the Administrative Agent (in its capacity as Administrative Agent and not as a Lender) in any way relating to or arising out of the Loan Documents or any other document delivered in connection therewith or the transactions contemplated thereby, or the enforcement of any of the terms thereof or of any such other documents, provided that no Lender shall be liable for any of the foregoing to the extent they arise from the Gross Negligence or willful misconduct of the Administrative Agent as determined by a court of competent jurisdiction in a final, non-appealable judgment; provided, however, that no action taken in accordance with the directions of the Required Lenders (or all of the Lenders, if expressly required hereunder) shall be deemed to constitute Gross Negligence or willful misconduct for purposes of this Section. The agreements in this Section shall survive the payment of the Loans and all other amounts payable hereunder or under the other Loan Documents and the termination of this Agreement. If the Borrower shall reimburse the Administrative Agent for any indemnifiable amount set forth in Section 12.8(i), (ii) or (ii) following payment by any Lender to the Administrative Agent in respect of any such indemnifiable amount pursuant to this Section, the Administrative Agent shall share such reimbursement on a ratable basis with each Lender making any such payment.

12.9. Rights as a Lender. With respect to its Term Loan Commitment, Borrowings made by it and any Note issued to it, the Administrative Agent shall have the same rights and powers hereunder and under any other Loan Document as any Lender and may exercise the same as though it were not the Administrative Agent, and the term "Lender" or "Lenders" shall, unless the context otherwise indicates, include the Administrative Agent in its individual capacity. The Administrative Agent, in its individual capacity, may accept deposits from, lend money to, and generally engage in any kind of trust, debt, equity or other transaction, in addition to those contemplated by this Agreement or any other Loan Document, with the Borrower or any of its Subsidiaries in which the Borrower or such Subsidiary is not restricted hereby from engaging with any other Person. Each of the Lenders acknowledges that the Administrative Agent's legal counsel in connection with the transactions contemplated by this Agreement is only acting as counsel to the Administrative Agent and is not acting as counsel to any Lender.

12.10. [<u>Reserved</u>].

12.11. <u>Lender Credit Decision</u>. Each Lender acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Lender and based on the financial statements prepared by the Borrower and such other documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement and the other Loan Documents. Each Lender also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Agreement and the other Loan Documents.

- 12.12. <u>Successor Administrative Agent</u>. The Administrative Agent may resign at any time as Administrative Agent under the Loan Documents by giving written notice thereof to the Lenders and the Borrower. Upon any such resignation, the Required Lenders shall have the right to appoint a successor Administrative Agent which appointment shall, provided no Default or Event of Default exists, be subject to the Borrower's approval, which approval shall not be unreasonably withheld or delayed (except that the Borrower shall, in all events, be deemed to have approved each Lender and any of its Affiliates as a successor Administrative Agent). If no successor Administrative Agent shall have been so appointed in accordance with the immediately preceding sentence, and shall have accepted such appointment, within 30 days after the current Administrative Agent's giving of notice of resignation, then the current Administrative Agent may, on behalf of the Lenders, appoint a successor Administrative Agent, which shall be a Lender, if any Lender shall be willing to serve, and otherwise shall be an Eligible Assignee. Upon the acceptance of any appointment as Administrative Agent hereunder by a successor Administrative Agent, such successor Administrative Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the current Administrative Agent, and the current Administrative Agent shall be discharged from its duties and obligations under the Loan Documents. After any Administrative Agent's resignation hereunder as Administrative Agent, the provisions of this Article XII. shall continue to inure to its benefit as to any actions taken or omitted to be taken by it while it was Administrative Agent under the Loan Documents. Notwithstanding anything contained herein to the contrary, the Administrative Agent may assign its rights and duties under the Loan Documents to any of its Affiliates by giving the Borrower and each Lender prior written notice.
- 12.13. <u>Notice of Defaults</u>. The Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of a Default or Event of Default unless the Administrative Agent has received notice from a Lender or the Borrower referring to this Agreement, describing with reasonable specificity such Default or Event of Default and stating that such notice is a "notice of default." If a Lender becomes aware of a Default or Event of Default, such Lender shall notify the Administrative Agent of such fact. Upon receipt of such notice that a Default or Event of Default has occurred, the Administrative Agent shall notify each of the Lenders of such fact.
- 12.14. Requests for Approval. If the Administrative Agent requests in writing the consent or approval of a Lender, such Lender shall, or, in the case of any request for consent or approval that is subject to Section 14.13(a) or (b), shall use commercially reasonable efforts to, respond and either approve or disapprove definitively in writing to the Administrative Agent within ten (10) Business Days (or sooner if such notice specifies a shorter period, but in no event less than five (5) Business Days for responses based on Administrative Agent's good faith determination that circumstances exist warranting its request for an earlier response) after such written request from the Administrative Agent provided that the request for approval states the time by which a response is needed before approval is deemed given. Solely with respect to any request for consent or approval that is subject to Section 14.13(c), if the Lender does not so respond, that Lender shall be deemed to have approved the request. Upon request, the Administrative Agent shall notify the Lenders which Lenders, if any, failed to respond to a request for approval.

- 12.15. <u>Copies of Documents</u>. Administrative Agent shall promptly deliver to each of the Lenders copies of all notices of default and other formal notices sent or received and according to <u>Section 15.1</u> of this Agreement. Administrative Agent shall deliver to Lenders within fifteen (15) Business Days following receipt, copies of all financial statements, certificates and notices received regarding the General Partner's ratings except to the extent such items are required to be furnished directly to the Lenders by the Borrower hereunder. Within fifteen (15) Business Days after a request by a Lender to the Administrative Agent for other documents furnished to the Administrative Agent by the Borrower, the Administrative Agent shall provide copies of such documents to such Lender except where this Agreement obligates Administrative Agent to provide copies in a shorter period of time.
- 12.16. <u>Defaulting Lenders</u>. Notwithstanding any provision of this Agreement to the contrary, if any Lender becomes a Defaulting Lender, then to the extent permitted by Applicable Law, the following provisions shall apply for so long as such Lender is a Defaulting Lender:
- (a) <u>Waivers and Amendments</u>. Such Defaulting Lender's right to approve or disapprove any amendment, waiver or consent with respect to this Agreement shall be restricted as set forth in the definition of Required Lenders.
- <u>Defaulting Lender Waterfall</u>. Any payment of principal, interest, fees or other amounts received by the Administrative Agent for the account of such Defaulting Lender (whether voluntary or mandatory, at maturity, pursuant to Section 11.1 or otherwise) or received by the Administrative Agent from a Defaulting Lender pursuant to Section 2.20(b) shall be applied at such time or times as may be determined by the Administrative Agent as follows: first, to the payment of any amounts owing by such Defaulting Lender to the Administrative Agent hereunder; second, [reserved]; third, [reserved]; fourth, as the Borrower may request (so long as no Default or Event of Default exists), to the funding of any Loan in respect of which such Defaulting Lender has failed to fund its portion thereof as required by this Agreement, as determined by the Administrative Agent; fifth,[reserved]; sixth, to the payment of any amounts owing to the Lenders as a result of any judgment of a court of competent jurisdiction obtained by any Lender against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; seventh, so long as no Default or Event of Default exists, to the payment of any amounts owing to the Borrower as a result of any judgment of a court of competent jurisdiction obtained by the Borrower against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; and eighth, to such Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if such Loans were made at a time when the conditions set forth in <u>Article IV</u> were satisfied or waived, such payment shall be applied solely to pay the Loans of all Non-Defaulting Lenders on a pro rata basis prior to being applied to the payment of any Loans of such Defaulting Lender until such time as all Loans are held by the Lenders pro rata in accordance with their respective Percentages (determined without giving effect to the immediately following subsection (d)). Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender pursuant to this subsection shall be deemed paid to and redirected by such Defaulting Lender, and each Lender irrevocably consents hereto.

(c) [Reserved]

- (d) [Reserved]
- (e) [Reserved]
- (f) <u>Defaulting Lender Cure</u>. If the Borrower and the Administrative Agent agree in writing that a Lender is no longer a Defaulting Lender, the Administrative Agent will so notify the parties hereto, whereupon such Lender will cease to be a Defaulting Lender; <u>provided</u> that except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Defaulting Lender to Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender's having been a Defaulting Lender.
- 12.17. <u>Delegation to Affiliates</u>. The Borrower and the Lenders agree that the Administrative Agent may delegate any of its duties under this Agreement to any of its Affiliates. Any such Affiliate (and such Affiliate's directors, officers, agents and employees) which performs duties in connection with this Agreement shall be entitled to the same benefits of the indemnification, waiver and other protective provisions to which the Administrative Agent is entitled under Articles XII and XIV.
- 12.18. <u>Managing Agents</u>, <u>Syndication Agent</u>, <u>etc.</u>. Neither any of the Lenders identified in this Agreement as a "managing agent" nor the Syndication Agent shall have any right, power, obligation, liability, responsibility or duty under this Agreement other than those applicable to all Lenders as such. Without limiting the foregoing, none of such Lenders shall have or be deemed to have a fiduciary relationship with any Lender. Each Lender hereby makes the same acknowledgments with respect to such Lenders as it makes with respect to the Administrative Agent in <u>Section 12.11</u>.

12.19. Erroneous Payments.

(a) Each Lender and any other party hereto hereby severally agrees that if (i) the Administrative Agent notifies (which such notice shall be conclusive absent manifest error) such Lender or any other Person that has received funds from the Administrative Agent or any of its Affiliates, either for its own account or on behalf of a Lender (each such recipient, a "Payment Recipient") that the Administrative Agent has determined in its sole discretion that any funds received by such Payment Recipient were erroneously transmitted to, or otherwise erroneously or mistakenly received by, such Payment Recipient (whether or not known to such Payment Recipient) or (ii) any Payment Recipient receives any payment from the Administrative Agent (or any of its Affiliates) (x) that is in a different amount than, or on a different date from, that specified in a notice of payment, prepayment or repayment sent by the Administrative Agent (or any of its Affiliates) with respect to such payment, prepayment or repayment, as applicable, (y) that was not preceded or accompanied by a notice of payment, prepayment or repayment sent by the Administrative Agent (or any of its Affiliates) with respect to such payment, prepayment or repayment sent by the Administrative Agent (or any of its Affiliates) with respect to such payment, prepayment or repayment Recipient otherwise becomes aware was transmitted or received in error or by mistake (in whole or in part) then, in each case, an error in payment shall be presumed to have been made (any such amounts specified in clauses (i) or (ii) of this Section 12.19(a), whether received as a payment, prepayment or repayment of principal, interest, fees, distribution or otherwise; individually and collectively, an "Erroneous Payment"), then, in each case, such Payment Recipient is deemed to have knowledge of such error at the time of its receipt of such Erroneous Payment; provided that

nothing in this Section shall require the Administrative Agent to provide any of the notices specified in clauses (i) or (ii) above. Each Payment Recipient agrees that it shall not assert any right or claim to any Erroneous Payment, and hereby waives any claim, counterclaim, defense or right of set-off or recoupment with respect to any demand, claim or counterclaim by the Administrative Agent for the return of any Erroneous Payments, including without limitation waiver of any defense based on "discharge for value" or any similar doctrine.

(b) Without limiting the immediately preceding <u>clause (a)</u>, each Payment Recipient agrees that, in the case of clause (a)(ii) above, it shall promptly notify the Administrative Agent in writing of such occurrence.

In the case of either clause (a)(i) or (a)(ii) above, such Erroneous Payment shall at all times remain the property of the Administrative Agent and shall be segregated by the Payment Recipient and held in trust for the benefit of the Administrative Agent, and upon demand from the Administrative Agent such Payment Recipient shall (or, shall cause any Person who received any portion of an Erroneous Payment on its behalf to), promptly, but in all events no later than two Business Days thereafter, return to the Administrative Agent the amount of any such Erroneous Payment (or portion thereof) as to which such a demand was made in same day funds and in the currency so received, together with interest thereon in respect of each day from and including the date such Erroneous Payment (or portion thereof) was received by such Payment Recipient to the date such amount is repaid to the Administrative Agent at the greater of the Federal Funds Effective Rate and a rate determined by the Administrative Agent in accordance with banking industry rules on interbank compensation from time to time in effect; provided that, if the Payment Recipient is the Borrower, (a) the accrual of such interest shall be at the Federal Funds Effective Rate and shall commence to accrue from and including the second Business Day following the date such Erroneous Payment (or portion thereof) was received by the Borrower (the "Borrower EP Interest Accrual <u>Date</u>") to the date such amount is repaid to the Administrative Agent and (b) if such Erroneous Payment (or portion thereof) shall have been returned to the Administrative Agent prior to the Borrower EP Interest Accrual Date, then the Borrower shall not have any further obligations or liability with respect to such Erroneous Payment, including for any interest, charges, fees, costs or expenses incurred as a result of the Administrative Agent making the Erroneous Payment to the Borrower, the Borrower holding the Erroneous Payment (or portion thereof) in trust for the benefit of the Administrative Agent and the Borrower returning the Erroneous Payment (or portion thereof) to the Administrative Agent.

(c) In the event that an Erroneous Payment (or portion thereof) is not recovered by the Administrative Agent for any reason, after demand therefor by the Administrative Agent in accordance with immediately preceding clause (c), from any Lender that is a Payment Recipient or an Affiliate of a Payment Recipient (such unrecovered amount as to such Lender, an "Erroneous Payment Return Deficiency"), then at the sole discretion of the Administrative Agent and upon the Administrative Agent's written notice to such Lender (i) such Lender shall be deemed to have made a cashless assignment of the full face amount of the portion of its Loans (but not its Term Loan Commitments) of the relevant Class with respect to which such Erroneous Payment was made (the "Erroneous Payment Impacted Class") to the Administrative Agent or, at the option of the Administrative Agent, the Administrative Agent's applicable lending affiliate in an amount that is equal to the Erroneous Payment Return Deficiency (or such lesser amount as the Administrative Agent may specify) (such assignment of the Loans (but not Term Loan Commitments) of the Erroneous Payment Impacted Class, the "Erroneous Payment Deficiency Assignment") plus any

accrued and unpaid interest on such assigned amount, without further consent or approval of any party hereto and without any payment by the Administrative Agent or its applicable lending affiliate as the assignee of such Erroneous Payment Deficiency Assignment. Without limitation of its rights hereunder, the Administrative Agent may cancel any Erroneous Payment Deficiency Assignment at any time by written notice to the applicable assigning Lender and upon such revocation all of the Loans assigned pursuant to such Erroneous Payment Deficiency Assignment shall be reassigned to such Lender without any requirement for payment or other consideration. The parties hereto acknowledge and agree that (1) any assignment contemplated in this clause (d) shall be made without any requirement for any payment or other consideration paid by the applicable assignee or received by the assignor, (2) the provisions of this clause (d) shall govern in the event of any conflict with the terms and conditions of Article XIII and (3) the Administrative Agent may reflect such assignments in the Register without further consent or action by any other Person.

- (d) Each party hereto hereby agrees that (x) in the event an Erroneous Payment (or portion thereof) is not recovered from any Payment Recipient that has received such Erroneous Payment (or portion thereof) for any reason, the Administrative Agent (1) shall be subrogated to all the rights of such Payment Recipient with respect to such amount and (2) is authorized to set off, net and apply any and all amounts at any time owing to such Payment Recipient under any Loan Document, or otherwise payable or distributable by the Administrative Agent to such Payment Recipient from any source, against any amount due to the Administrative Agent under this Section 12.19 or under the indemnification provisions of this Agreement, (y) the receipt of an Erroneous Payment by a Payment Recipient shall not for the purpose of this Agreement be treated as a payment, prepayment, repayment, discharge or other satisfaction of any Obligations owed by the Borrower or any other Loan Party, except, in each case, to the extent such Erroneous Payment is, and solely with respect to the amount of such Erroneous Payment that is, comprised of funds received by the Administrative Agent from the Borrower or any other Credit Party for the purpose of making for a payment on the Obligations and (z) to the extent that an Erroneous Payment was in any way or at any time credited as payment or satisfaction of any of the Obligations, the Obligations or any part thereof that were so credited, and all rights of the Payment Recipient, as the case may be, shall be reinstated and continue in full force and effect as if such payment or satisfaction had never been received.
- (e) Each party's obligations under this <u>Section 12.19</u> shall survive the resignation or replacement of the Administrative Agent or any transfer of right or obligations by, or the replacement of, a Lender, the termination of the Term Loan Commitments or the repayment, satisfaction or discharge of all Obligations (or any portion thereof) under any Loan Document.
- (f) Nothing in this <u>Section 12.19</u> will constitute a waiver or release of any claim of any party hereunder arising from any Payment Recipient's receipt of an Erroneous Payment.

ARTICLE XIII.

BENEFIT OF AGREEMENT; ASSIGNMENTS; PARTICIPATIONS

Successors and Assigns. The terms and provisions of the Loan Documents shall be binding upon and inure to the 13.1. benefit of the Borrower and the Lenders and their respective successors and assigns permitted hereby, except that (i) the Borrower shall not have the right to assign its rights or obligations under the Loan Documents without the prior written consent of each Lender, (ii) any assignment by any Lender must be made in compliance with Section 13.3, and (iii) any transfer by Participation must be made in compliance with Section 13.2. Any attempted assignment or transfer by any party not made in compliance with this Section 13.1 shall be null and void, unless such attempted assignment or transfer is treated as a participation in accordance with Section 13.3. The parties to this Agreement acknowledge that clause (ii) of this Section 13.1 relates only to absolute assignments and this Section 13.1 does not prohibit assignments creating security interests, including, without limitation, (x) any pledge or assignment by any Lender of all or any portion of its rights under this Agreement and any Note to a Federal Reserve Bank or other central banking authority or (y) in the case of a Lender which is a Fund, any pledge or assignment of all or any portion of its rights under this Agreement and any Note to its trustee in support of its obligations to its trustee; provided, however, that no such pledge or assignment creating a security interest shall release the transferor Lender from its obligations hereunder unless and until the parties thereto have complied with the provisions of Section 13.3. The Administrative Agent may treat the Person which made any Loan or which holds any Note as the owner thereof for all purposes hereof unless and until such Person complies with Section 13.3; provided, however, that the Administrative Agent may in its discretion (but shall not be required to) follow instructions from the Person which made any Loan or which holds any Note to direct payments relating to such Loan or Note to another Person. Any assignee of the rights to any Loan or any Note agrees by acceptance of such assignment to be bound by all the terms and provisions of the Loan Documents. Any request, authority or consent of any Person, who at the time of making such request or giving such authority or consent is the owner of the rights to any Loan (whether or not a Note has been issued in evidence thereof), shall be conclusive and binding on any subsequent holder or assignee of the rights to such Loan.

13.2. Participations.

13.2.1 <u>Permitted Participants</u>; <u>Effect</u>. Any Lender may at any time sell to one or more banks or other entities (other than a natural Person, or a holding company, investment vehicle or trust for, or owned and operated for the primary benefit of, a natural Person, any Defaulting Lender, or the Borrower or any of the Borrower's Affiliates or Subsidiaries) ("<u>Participants</u>") participating interests in any Loan owing to such Lender, any Note held by such Lender, any Term Loan Commitment of such Lender or any other interest of such Lender under the Loan Documents. In the event of any such sale by a Lender of participating interests to a Participant, such Lender's obligations under the Loan Documents shall remain unchanged, such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations, such Lender shall remain the owner of its Loans and the holder of any Note issued to it in evidence thereof for all purposes under the Loan Documents, all amounts payable by the Borrower under this Agreement shall be determined as if such Lender had not sold such participating interests, and the Borrower and the

Administrative Agent shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under the Loan Documents.

- 13.2.2 <u>Voting Rights</u>. Each Lender shall retain the sole right to approve, without the consent of any Participant, any amendment, modification or waiver of any provision of the Loan Documents other than any amendment, modification or waiver with respect to any Loan or Term Loan Commitment in which such Participant has an interest which would require consent of all of the Lenders pursuant to the terms of <u>Section 14.13</u> or of any other Loan Document.
- 13.2.3 <u>Benefit of Certain Provisions.</u> The Borrower agrees that each Participant shall be deemed to have the right of setoff provided in <u>Section 14.15(a)</u> in respect of its participating interest in amounts owing under the Loan Documents to the same extent as if the amount of its participating interest were owing directly to it as a Lender under the Loan Documents, <u>provided</u> that each Lender shall retain the right of setoff provided in <u>Section 14.15(a)</u> with respect to the amount of participating interests sold to each Participant. The Lenders agree to share with each Participant, and each Participant, by exercising the right of setoff provided in <u>Section 14.15(a)</u>, agrees to share with each Lender, any amount received pursuant to the exercise of its right of setoff, such amounts to be shared in accordance with <u>Section 14.15(b)</u> as if each Participant were a Lender. The Borrower further agrees that each Participant shall be entitled to the benefits of <u>Sections 4.1</u>, <u>4.2</u>, <u>4.4</u> and <u>4.5</u> to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to <u>Section 13.3</u>, <u>provided</u> that (i) a Participant shall not be entitled to receive any greater payment under <u>Section 4.1</u>, <u>4.2</u>, <u>4.4</u> or <u>4.5</u> than the Lender who sold the participating interest to such Participant would have received had it retained such interest for its own account, unless the sale of such interest to such Participant is made with the prior written consent of the Borrower, and (ii) any Participant not incorporated under the laws of the United States of America or any State thereof agrees to comply with the provisions of Section 4.5 to the same extent as if it were a Lender.
- 13.2.4 <u>Participant Register</u>. Each Lender that sells a participation shall, acting solely for this purpose as a non-fiduciary agent of the Borrower, maintain a register on which it enters the name and address of each Participant and the principal amounts (and stated interest) of each Participant's interest in the Loans or other obligations under this Agreement (the "<u>Participant Register</u>"); provided that no Lender shall have any obligation to disclose all or any portion of the Participant Register to any Person (including the identity of any Participant or any information relating to a Participant's interest in any Term Loan Commitments, Loans or its other obligations under this Agreement) except to the extent that such disclosure is necessary to establish that such Term Loan Commitment, Loan or other obligation is in registered form under Section 5f.103-1(c) of the United States Treasury Regulations. The entries in the Participant Register shall be conclusive absent manifest error, and such Lender shall treat each person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary.

13.3. <u>Assignments</u>; <u>Consents</u>. Any Lender may at any time assign to one or more Eligible Assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Term Loan Commitment and the Loans at the time owing to it); <u>provided</u> that any such assignment shall be subject to the following conditions:

13.3.1 Minimum Amounts.

- (A) in the case of an assignment of the entire remaining amount of an assigning Lender's Term Loan Commitment and the Loans at the time owing to it, or in the case of an assignment to a Lender, an Affiliate of a Lender or an Approved Fund, no minimum amount need be assigned; and
- (B) in any case not described in the immediately preceding subsection (A), the aggregate amount of the Term Loan Commitment (which for this purpose includes Loans outstanding thereunder) or, if the applicable Term Loan Commitment is not then in effect, the principal outstanding balance of the Loans of the assigning Lender subject to each such assignment (determined as of the date the Assignment and Assumption Agreement with respect to such assignment is delivered to the Administrative Agent or, if "Trade Date" is specified in the Assignment and Assumption Agreement, as of the Trade Date) shall not be less than \$5,000,000, unless each of the Administrative Agent and, so long as no Event of Default has occurred and is continuing, the Borrower otherwise consents (each such consent not to be unreasonably withheld or delayed); provided, however, that if, after giving effect to such assignment, the amount of the Term Loan Commitment held by such assigning Lender or the outstanding principal balance of the Loans of such assigning Lender, as applicable, would be less than \$5,000,000, then such assigning Lender shall assign the entire amount of its Term Loan Commitment and the Loans at the time owing to it.
- 13.3.2 <u>Proportionate Amounts</u>. Each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Loan or the Term Loan Commitment assigned.
- 13.3.3 <u>Required Consents</u>. No consent shall be required for any assignment except to the extent required by <u>Section 13.3.1(B)</u> and, in addition:
 - (A) the consent of the Borrower (such consent not to be unreasonably withheld or delayed) shall be required unless (x) an Event of Default has occurred and is continuing at the time of such assignment or (y) such assignment is to a Lender, an Affiliate of a Lender or an Approved Fund; provided that the Borrower shall be deemed to have consented to any such assignment unless it shall object thereto by written notice to the Administrative Agent within five (5) Business Days after having received notice thereof; and
 - (B) the consent of the Administrative Agent (such consent not to be unreasonably withheld or delayed) shall be required for assignments in respect of a Term Loan Commitment or a Loan if such assignment is to a Person that is not

already a Lender, an Affiliate of a Lender or an Approved Fund with respect to a Lender.

- 13.3.4 <u>Assignment and Acceptance; Notes.</u> The parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption Agreement, together with a processing and recordation fee of \$3,500 for each assignment, and the assignee, if it is not a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire. If requested by the transferor Lender or the Assignee, upon the consummation of any assignment, the transferor Lender, the Administrative Agent and the Borrower shall make appropriate arrangements so that new Notes are issued to the Assignee and such transferor Lender, as appropriate.
- 13.3.5 <u>No Assignment to Borrower or Defaulting Lender</u>. No such assignment shall be made to (i) the Borrower or any of the Borrower's Affiliates or Subsidiaries or (ii) any Defaulting Lender.
- 13.3.6 <u>No Assignment to Natural Persons</u>. No such assignment shall be made to a natural person or a holding company, investment vehicle or trust for, or owned and granted for the primary benefit of, a natural Person.

Subject to acceptance and recording thereof by the Administrative Agent pursuant to Section 13.4, from and after the effective date specified in each Assignment and Assumption Agreement, the assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption Agreement, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Assumption Agreement, be released from its obligations under this Agreement (and, in the case of an Assignment and Assumption Agreement covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto) but shall continue to be entitled to the benefits of Sections 12.8 and 14.6 and the other provisions of this Agreement and the other Loan Documents that survive the termination hereof and thereof with respect to facts and circumstances occurring prior to the effective date of such assignment. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this paragraph shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with Section 13.2.

13.4. <u>Register</u>. The Administrative Agent, acting solely for this purpose as a non-fiduciary agent of the Borrower, shall maintain at the Administrative Office a copy of each Assignment and Assumption Agreement delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Term Loan Commitments of, and principal amounts (and stated interest) of the Loans owing to, each Lender pursuant to the terms hereof from time to time (the "<u>Register</u>"). The entries in the Register shall be conclusive, and the Borrower, the Administrative Agent and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by the Borrower and any Lender, at any reasonable time and from time to time upon reasonable prior notice.

13.5. [<u>Reserved</u>].

13.6 Confidentiality. Except as otherwise provided by Applicable Law, the Administrative Agent and each Lender shall maintain the confidentiality of all Information (as defined below) in accordance with its customary procedure for handling confidential information of this nature and in accordance with safe and sound banking practices but in any event may make disclosure: (a) to its Affiliates and to its and its Affiliates' respective partners, directors, officers, employees, agents, advisors and other representatives (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential); (b) subject to an agreement containing provisions substantially the same as those of this Section, to (i) any actual or proposed Eligible Assignee, Participant or other transferee in connection with a potential transfer of any Term Loan Commitment, Loan or participation therein as permitted hereunder, or (ii) any actual or prospective counterparty (or its advisors) to any swap or derivative transaction relating to the Borrower and its obligations; (c) as required or requested by any Governmental Authority or representative thereof or pursuant to legal process or in connection with any legal proceedings, or as otherwise required by Applicable Law; (d) to the Administrative Agent's or such Lender's independent auditors and other professional advisors (provided they shall be notified of the confidential nature of the information); (e) in connection with the exercise of any remedies under any Loan Document or any action or proceeding relating to any Loan Document or the enforcement of rights hereunder or thereunder; (f) to the extent such Information (i) becomes publicly available other than as a result of a breach of this Section actually known by the Administrative Agent or such Lender to be a breach of this Section or (ii) becomes available to the Administrative Agent, any Lender or any Affiliate of the Administrative Agent or any Lender on a nonconfidential basis from a source other than the Borrower or any Affiliate of the Borrower; (g) to the extent requested by, or required to be disclosed to, any nationally recognized rating agency or regulatory or similar authority (including any self-regulatory authority, such as the National Association of Insurance Commissioners) having or purporting to have jurisdiction over it; (h) to bank trade publications, such information to consist of deal terms and other information customarily found in such publications; (i) to any other party hereto; and (j) with the consent of the Borrower. Notwithstanding the foregoing, the Administrative Agent and each Lender may disclose any such confidential information, without notice to the Borrower or the General Partner, to Governmental Authorities in connection with any regulatory examination of the Administrative Agent or such Lender or in accordance with the regulatory compliance policy of the Administrative Agent or such Lender. As used in this Section, the term "Information" means all information received from the Borrower, the General Partner, any Subsidiary or Affiliate of the Borrower relating to the Borrower, the General Partner or any of their Affiliates or any of their respective businesses, other than any such information that is available to the Administrative Agent or any Lender on a nonconfidential basis prior to disclosure by the Borrower, the General Partner, or any Subsidiary or any Affiliate of the Borrower, provided that, in the case of any such information received from the Borrower, the General Partner, or any Subsidiary or any Affiliate of the Borrower after the date hereof, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of

care to maintain the confidentiality of such Information as such Person would accord to its own confidential information.

13.7 <u>Tax Treatment</u>. If any interest in any Loan Document is transferred to any Participant or Purchaser or any other Person acquiring an interest in the Loan Documents by operation of law (each a "<u>Transferee</u>") which is not incorporated under the laws of the United States or any State thereof, the transferor Lender shall cause such Transferee, concurrently with the effectiveness of such transfer, to comply with the provisions of <u>Section 4.5(iv)</u>.

ARTICLE XIV.

GENERAL PROVISIONS

- 14.1. <u>Survival of Representations</u>. All representations and warranties contained in this Agreement shall survive delivery of the Notes and the making of the Borrowings herein contemplated.
- 14.2. <u>Governmental Regulation</u>. Anything contained in this Agreement to the contrary notwithstanding, no Lender shall be obligated to extend credit to the Borrower in violation of any limitation or prohibition provided by any applicable statute or regulation.
- 14.3. <u>Taxes</u>. Any recording and other taxes (excluding franchise, income or similar taxes) or other similar assessments or charges payable or ruled payable by any governmental authority incurred in connection with the consummation of the transactions contemplated by this Agreement shall be paid by the Borrower, together with interest and penalties, if any.
- 14.4. <u>Headings</u>. Section headings in the Loan Documents are for convenience of reference only, and shall not govern the interpretation of any of the provisions of the Loan Documents.
- 14.5. <u>No Third Party Beneficiaries</u>. This Agreement shall not be construed so as to confer any right or benefit upon any Person other than the parties to this Agreement and their respective successors and assigns.
- 14.6. Expenses; Indemnification. Subject to the provisions of this Agreement, the Borrower will pay (a) all out-of-pocket costs and expenses incurred by the Administrative Agent and the Arranger (including the reasonable fees, out-of-pocket expenses and other reasonable expenses of counsel) in connection with the preparation, execution and delivery of this Agreement, the Notes, the Loan Documents and any other agreements or documents referred to herein or therein and any amendments thereto, (b) all out-of-pocket costs and expenses incurred by the Administrative Agent and the Lenders (including the reasonable fees, out-of-pocket expenses and other reasonable expenses of counsel to the Administrative Agent and the Lenders) in connection with the enforcement and protection of the rights of the Lenders under this Agreement, the Notes, the Loan Documents or any other agreement or document referred to herein or therein, and (c) all reasonable and customary costs and expenses of periodic audits by the Administrative Agent's personnel of the Borrower's books and records provided that prior to an Event of Default, the Borrower shall be required to pay for only one such audit during any year. The Borrower further agrees to indemnify the Lenders, their Affiliates, and their respective directors, officers, employees, agents and advisors (each, an "Indemnified Party") against all losses, claims, damages, penalties, judgments, liabilities

and reasonable expenses (including, without limitation, all expenses of litigation or preparation therefor whether or not the Indemnified Party is a party thereto) which any of them may pay or incur arising out of or relating to this Agreement, the other Loan Documents, the transactions contemplated hereby or the direct or indirect application or proposed application of the proceeds of any Borrowing hereunder, except that the foregoing indemnity shall not apply to any Indemnified Party to the extent that any losses, claims, etc. are determined by a court of competent jurisdiction by a final and non-appealable judgment to have resulted from such Indemnified Party's Gross Negligence or willful misconduct. In the case of an investigation, litigation or proceeding to which the indemnity in this Section applies, such indemnity shall be effective whether or not such investigation, litigation or proceeding is brought by the Borrower or the General Partner or the Borrower's or the General Partner's equity holders or creditors or an Indemnified Party is otherwise party thereto. The obligations of the Borrower under this Section shall survive the termination of this Agreement.

- 14.7. <u>Severability of Provisions</u>. Any provision in any Loan Document that is held to be inoperative, unenforceable, or invalid in any jurisdiction shall, as to that jurisdiction, be inoperative, unenforceable, or invalid without affecting the remaining provisions in that jurisdiction or the operation, enforceability, or validity of that provision in any other jurisdiction, and to this end the provisions of all Loan Documents are declared to be severable.
- 14.8. Nonliability of the Lenders. The relationship between the Borrower and the Lenders shall be solely that of borrower and lender. The Lenders shall not have any fiduciary responsibilities to the Borrower. The Lenders undertake no responsibility to the Borrower to review or inform the Borrower of any matter in connection with any phase of the Borrower's business or operations. Moreover, none of the Administrative Agent or any Lender shall be liable to the Borrower or any other parties. None of the Administrative Agent or any Lender, or any Affiliate, officer, director, employee, attorney, or agent of the Administrative Agent or any Lender shall have any liability with respect to, and the Borrower hereby waives, releases, and agrees not to sue any of them upon, any claim for any special, indirect, incidental, or consequential damages suffered or incurred by the Borrower in connection with, arising out of, or in any way related to, this Agreement, any of the other Loan Documents or any fee letter delivered in connection herewith, or any of the transactions contemplated by this Agreement or any of the other Loan Documents.
- 14.9. <u>Choice of Law</u>. THE LOAN DOCUMENTS (OTHER THAN THOSE CONTAINING A CONTRARY EXPRESS CHOICE OF LAW PROVISION) SHALL BE CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS (AND NOT THE LAW OF CONFLICTS) OF THE STATE OF ILLINOIS, BUT GIVING EFFECT TO FEDERAL LAWS APPLICABLE TO NATIONAL BANKS.

- 14.10. Consent to Jurisdiction. THE BORROWER HEREBY IRREVOCABLY SUBMITS TO THE NON EXCLUSIVE JURISDICTION OF ANY UNITED STATES FEDERAL OR ILLINOIS STATE COURT SITTING IN CHICAGO, ILLINOIS IN ANY ACTION OR PROCEEDING ARISING OUT OF OR RELATING TO ANY LOAN DOCUMENTS AND THE BORROWER HEREBY IRREVOCABLY AGREES THAT ALL CLAIMS IN RESPECT OF SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN ANY SUCH COURT AND IRREVOCABLY WAIVES ANY OBJECTION IT MAY NOW OR HEREAFTER HAVE AS TO THE VENUE OF ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN SUCH A COURT OR THAT SUCH COURT IS AN INCONVENIENT FORUM. NOTHING HEREIN SHALL LIMIT THE RIGHT OF THE LENDERS TO BRING PROCEEDINGS AGAINST THE BORROWER IN THE COURTS OF ANY OTHER JURISDICTION. ANY JUDICIAL PROCEEDING BY THE BORROWER AGAINST THE LENDERS OR ANY AFFILIATE OF THE LENDERS INVOLVING, DIRECTLY OR INDIRECTLY, ANY MATTER IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH ANY LOAN DOCUMENT SHALL BE BROUGHT ONLY IN A COURT IN CHICAGO, ILLINOIS.
- 14.11. <u>Waiver of Jury Trial</u>. THE BORROWER, THE GENERAL PARTNER, THE ADMINISTRATIVE AGENT AND THE LENDERS HEREBY WAIVE TRIAL BY JURY IN ANY JUDICIAL PROCEEDING INVOLVING, DIRECTLY OR INDIRECTLY, ANY MATTER (WHETHER SOUNDING IN TORT, CONTRACT OR OTHERWISE) IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH ANY LOAN DOCUMENT OR THE RELATIONSHIP ESTABLISHED THEREUNDER.
- 14.12. <u>Successors and Assigns</u>. The terms and provisions of the Loan Documents shall be binding upon and inure to the benefit of the Borrower and the Lenders and their respective successors and assigns, except that the Borrower shall not have the right to assign its rights or obligations under the Loan Documents. Any assignee or transferee of rights or obligations hereunder or under the Notes agrees by acceptance thereof to be bound by all the terms and provisions of the Loan Documents. Any request, authority or consent of any Person, who at the time of making such request or giving such authority or consent is the holder of a Note, shall be conclusive and binding on any subsequent holder, transferee or assignee of such Note or of any Note or Notes issued in exchange therefor.
- 14.13. <u>Entire Agreement; Modification of Agreement</u>. The Loan Documents embody the entire agreement among the Borrower, the General Partner, Administrative Agent, and Lenders and supersede all prior conversations, agreements, understandings, commitments and term sheets among any or all of such parties with respect to the subject matter hereof. Any provisions of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is signed by the Borrower, and Administrative Agent if the rights or duties of Administrative Agent are affected thereby, and
 - (a) each of the Lenders adversely affected thereby if such amendment or waiver:
 - (i) subject to clause (d) below, reduces or forgives any payment of principal or interest on the Obligations or any fees payable by the Borrower to such Lender hereunder; or

- (ii) postpones the date fixed for any payment of principal of or interest on the Obligations or any fees payable by the Borrower to such Lender hereunder; or
- (iii) increases the amount of such Lender's Term Loan Commitment or the unpaid principal amount of such Lender's Loan (other than pursuant to an assignment permitted under <u>Section 13.3</u>); or
 - (iv) extends the Maturity Date; or
 - (v) modifies or waives all or any portion of Section 2.20;
- (b) all of the Lenders if such amendment or waiver:
 - (i) releases or limits the liability of the General Partner under the Loan Documents; or
- (ii) changes the definition of Required Lenders, or modifies any requirement for consent by each of the Lenders; or
- (iii) reduces the percentage set forth in <u>Section 8.10</u> or modifies or waives the second and third sentences of <u>Section 9.3</u> hereof;
- (c) the Required Lenders, to the extent expressly provided for herein and in the case of all other waivers or amendments if no percentage of Lenders is specified herein; or
- (d) the Administrative Agent and the Borrower may, without the consent of any Lender, enter into amendments or modifications to this Agreement or any of the other Loan Documents or enter into additional Loan Documents as the Administrative Agent reasonably deems appropriate in order to implement any Benchmark Replacement or otherwise effectuate the terms of Section 4.8 in accordance with the terms of Section 4.8 (c).
- 14.14. <u>Dealings with the Borrower</u>. The Lenders and their affiliates may accept deposits from, extend credit to and generally engage in any kind of banking, trust or other business with the Borrower or the General Partner or any of their Affiliates regardless of the capacity of the Lenders hereunder.

14.15. <u>Set-Off</u>.

- (a) Subject to Section 2.20 and in addition to any rights now or hereafter granted under Applicable Law and not by way of limitation of any such rights, the Borrower hereby authorizes the Administrative Agent, each Lender, each Affiliate of the Administrative Agent or any Lender, and each Participant, at any time, or from time to time while, an Event of Default exists, without notice to the Borrower or to any other Person, any such notice being hereby expressly waived, but in the case of a Lender, an Affiliate of a Lender, or a Participant, subject to receipt of the prior written consent of the Required Lenders exercised in their sole discretion, to set off and to appropriate and to apply any and all deposits (general or special, including, but not limited to, indebtedness evidenced by certificates of deposit, whether matured or unmatured) and any other indebtedness at any time held or owing by the Administrative Agent, such Lender, any Affiliate of the Administrative Agent or such Lender, or such Participant, to or for the credit or the account of the Borrower against and on account of any of the Obligations, irrespective of whether or not any or all of the Loans and all other Obligations have been declared to be, or have otherwise become, due and payable as permitted by Section 11.1, and although such Obligations shall be contingent or unmatured.
- (b) Each Lender agrees that if it shall, by exercising any right of set off or counterclaim or otherwise, receive payment of a proportion of the aggregate amount of principal, interest or fees due with respect to any Loan held by it (other than payments received pursuant to Sections 4.1, 4.2, 4.5 and 4.8) which is greater than the proportion received by any other Lender in respect of the aggregate amount of principal, interest or fees due with respect to any Loan held by such other Lender, the Lender receiving such proportionately greater payment shall purchase such participations in the Loans held by the other Lenders and such other adjustments shall be made as may be required so that all such payments of principal, interest or fees with respect to the Loans held by the Lenders shall be shared by the Lenders pro rata according to their respective Percentages.

14.16. <u>Counterparts; Integration, Effectiveness, Electronic Execution</u>.

(a) This Agreement may be executed in counterparts (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute a single contract. This Agreement and the other Loan Documents, and any separate letter agreements with respect to fees payable to the Administrative Agent and/or any Arranger, constitute the entire contract among the parties relating to the subject matter hereof and supersede any and all previous agreements and understandings, oral or written, relating to the subject matter hereof. Except as provided in Section 5.1, this Agreement shall become effective when it shall have been executed by the Administrative Agent and when the Administrative Agent shall have received counterparts hereof that, when taken together, bear the signatures of each of the other parties hereto. Delivery of an executed counterpart of a signature page of this Agreement by facsimile or in electronic (i.e., "pdf" or "tif") format shall be effective as delivery of a manually executed counterpart of this Agreement.

- (b) The words "execute," "execution," "signed," "signature," "delivery" and words of like import in or related to this Agreement, any other Loan Document or any document, amendment, approval, consent, waiver, modification, information, notice, certificate, report, statement, disclosure, or authorization to be signed or delivered in connection with this Agreement or any other Loan Document or the transactions contemplated hereby shall be deemed to include Electronic Signatures or execution in the form of an Electronic Record, and contract formations on electronic platforms approved by the Administrative Agent, deliveries or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system, as the case may be, to the extent and as provided for in any Applicable Law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act. Each party hereto agrees that any Electronic Signature or execution in the form of an Electronic Record shall be valid and binding on itself and each of the other parties hereto to the same extent as a manual, original signature. For the avoidance of doubt, the authorization under this paragraph may include, without limitation, use or acceptance by the parties of a manually signed paper which has been converted into electronic form (such as scanned into PDF format), or an electronically signed paper converted into another format, for transmission, delivery and/or retention. Notwithstanding anything contained herein to the contrary, the Administrative Agent is under no obligation to accept an Electronic Signature in any form or in any format unless expressly agreed to by the Administrative Agent pursuant to procedures approved by it; provided that without limiting the foregoing, (a) to the extent the Administrative Agent has agreed to accept such Electronic Signature from any party hereto, the Administrative Agent and the other parties hereto shall be entitled to rely on any such Electronic Signature purportedly given by or on behalf of the executing party without further verification and (b) upon the request of the Administrative Agent or any Lender, any Electronic Signature shall be promptly followed by an original manually executed counterpart thereof. Without limiting the generality of the foregoing, each party hereto hereby (i) agrees that, for all purposes, including without limitation, in connection with any workout, restructuring, enforcement of remedies, bankruptcy proceedings or litigation among the Administrative Agent, the Lenders and any of the Loan Parties, electronic images of this Agreement or any other Loan Document (in each case, including with respect to any signature pages thereto) shall have the same legal effect, validity and enforceability as any paper original, and (ii) waives any argument, defense or right to contest the validity or enforceability of the Loan Documents based solely on the lack of paper original copies of any Loan Documents, including with respect to any signature pages thereto.
- 14.17. <u>Patriot Act CIP Notice</u>. Each Lender hereby notifies the Borrower and the General Partner that pursuant to the requirements of the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "Act), it is from time to time required to obtain, verify and record information that identifies the Borrower and the General Partner, which information includes the name and address of the Borrower and the General Partner and other information that will allow such Lender to identify the Borrower and the General Partner in accordance with the Act.

- 14.18. Acknowledgement and Consent to Bail-In of Affected Financial Institutions. Notwithstanding anything to the contrary in any Loan Document or in any other agreement, arrangement or understanding among any such parties, each party hereto acknowledges that any liability of any Affected Financial Institution arising under any Loan Document, to the extent such liability is unsecured, may be subject to the Write-Down and Conversion Powers of the applicable Resolution Authority and agrees and consents to, and acknowledges and agrees to be bound by:
 - (a) the application of any Write-Down and Conversion Powers by the applicable Resolution Authority to any such liabilities arising hereunder which may be payable to it by any party hereto that is an Affected Financial Institution; and
 - (b) the effects of any Bail-In Action on any such liability, including, if applicable:
 - (i) a reduction in full or in part or cancellation of any such liability;
 - (ii) a conversion of all, or a portion of, such liability into shares or other instruments of ownership in such Affected Financial Institution, its parent undertaking, or a bridge institution that may be issued to it or otherwise conferred on it, and that such shares or other instruments of ownership will be accepted by it in lieu of any rights with respect to any such liability under this Agreement or any other Loan Document; or
 - (iii) the variation of the terms of such liability in connection with the exercise of the Write-Down and Conversion Powers of the applicable Resolution Authority.
- 14.19. Acknowledgement Regarding Any Supported QFCs. To the extent that the Loan Documents provide support, through a guarantee or otherwise, for Rate Management Transactions or any other agreement or instrument that is a QFC (such support, "QFC Credit Support" and, each such QFC, a "Supported QFC"), the parties acknowledge and agree as follows with respect to the resolution power of the FDIC under the Federal Deposit Insurance Act and Title II of the Dodd-Frank Wall Street Reform and Consumer Protection Act (together with the regulations promulgated thereunder, the "U.S. Special Resolution Regimes") in respect of such Supported QFC and QFC Credit Support (with the provisions below applicable notwithstanding that the Loan Documents and any Supported QFC may in fact be stated to be governed by the laws of the State of New York and/or of the United States or any other state of the United States):
 - (a) In the event a Covered Entity that is party to a Supported QFC (each, a "Covered Party") becomes subject to a proceeding under a U.S. Special Resolution Regime, the transfer of such Supported QFC and the benefit of such QFC Credit Support (and any interest and obligation in or under such Supported QFC and such QFC Credit Support, and any rights in property securing such Supported QFC or such QFC Credit Support) from such Covered Party will be effective to the same extent as the transfer would be effective under the U.S. Special Resolution Regime if the Supported QFC and such QFC Credit Support (and any such interest, obligation and rights in property) were governed by the laws of the United States or a state of the United States. In the event a Covered Party or a BHC Act Affiliate of a Covered Party becomes subject to a proceeding under a U.S. Special Resolution Regime,

Default Rights under the Loan Documents that might otherwise apply to such Supported QFC or any QFC Credit Support that may be exercised against such Covered Party are permitted to be exercised to no greater extent than such Default Rights could be exercised under the U.S. Special Resolution Regime if the Supported QFC and the Loan Documents were governed by the laws of the United States or a state of the United States. Without limitation of the foregoing, it is understood and agreed that rights and remedies of the parties with respect to a Defaulting Lender shall in no event affect the rights of any Covered Party with respect to a Supported QFC or any QFC Credit Support.

(b) As used in this Section 14.19, the following terms have the following meanings:

"BHC Act Affiliate" of a party means an "affiliate" (as such term is defined under, and interpreted in accordance with, 12 U.S.C. 1841(k)) of such party.

"Covered Entity" means any of the following:

- (i) a "covered entity" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 252.82(b);
- (ii) a "covered bank" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 47.3(b); or
- (iii) a "covered FSI" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 382.2(b).

"<u>Default Right</u>" has the meaning assigned to that term in, and shall be interpreted in accordance with, 12 C.F.R. §§ 252.81, 47.2 or 382.1, as applicable.

"QFC" has the meaning assigned to the term "qualified financial contract" in, and shall be interpreted in accordance with, 12 U.S.C. 5390(c)(8)(D).

14.20. Certain ERISA Matters.

- (a) Each Lender (x) represents and warrants, as of the date such Person became a Lender party hereto, to, and (y) covenants, from the date such Person became a Lender party hereto to the date such Person ceases being a Lender party hereto, for the benefit of, the Administrative Agent, each Arranger and their respective Affiliates, and not, for the avoidance of doubt, to or for the benefit of the Borrower or any other Loan Party, that at least one of the following is and will be true:
 - (i) such Lender is not using "plan assets" (within the meaning of Section 3(42) of ERISA or otherwise) of one or more Benefit Plans with respect to such Lender's entrance into, participation in, administration of and performance of the Loans or the Term Loan Commitments;
 - (ii) the transaction exemption set forth in one or more PTEs, such as PTE 84-14 (a class exemption for certain transactions determined by independent qualified professional asset managers), PTE 95-60 (a class exemption for certain transactions involving insurance

company general accounts), PTE 90-1 (a class exemption for certain transactions involving insurance company pooled separate accounts), PTE 91-38 (a class exemption for certain transactions involving bank collective investment funds) or PTE 96-23 (a class exemption for certain transactions determined by in-house asset managers), is applicable with respect to such Lender's entrance into, participation in, administration of and performance of the Loans, the Term Loan Commitments and this Agreement;

- (iii) (A) such Lender is an investment fund managed by a "Qualified Professional Asset Manager" (within the meaning of Part VI of PTE 84-14), (B) such Qualified Professional Asset Manager made the investment decision on behalf of such Lender to enter into, participate in, administer and perform the Loans, the Term Loan Commitments and this Agreement, (C) the entrance into, participation in, administration of and performance of the Loans, the Term Loan Commitments and this Agreement satisfies the requirements of sub-sections (b) through (g) of Part I of PTE 84-14 and (D) to the best knowledge of such Lender, the requirements of subsection (a) of Part I of PTE 84-14 are satisfied with respect to such Lender's entrance into, participation in, administration of and performance of the Loans, the Term Loan Commitments and this Agreement; or
- (iv) such other representation, warranty and covenant as may be agreed in writing between the Administrative Agent, in its sole discretion, and such Lender.

In addition, unless either (1) sub-clause (i) in the immediately preceding clause (a) is true with respect to a Lender or (2) a Lender has provided another representation, warranty and covenant in accordance with sub-clause (iv) in the immediately preceding clause (a), such Lender further (x) represents and warrants, as of the date such Person became a Lender party hereto, to, and (y) covenants, from the date such Person became a Lender party hereto, for the benefit of, the Administrative Agent, each Arranger and their respective Affiliates, and not, for the avoidance of doubt, to or for the benefit of the Borrower or any other Loan Party, that none of the Administrative Agent, any Arranger and their respective Affiliates is a fiduciary with respect to the assets of such Lender involved in such Lender's entrance into, participation in, administration of and performance of the Loans, the Term Loan Commitments and this Agreement (including in connection with the reservation or exercise of any rights by the Administrative Agent under this Agreement, any Loan Document or any documents related hereto or thereto).

ARTICLE XV.

NOTICES

Giving Notice; Electronic Delivery. All notices and other communications provided to any party hereto under this 15.1. Agreement or any other Loan Document shall be in writing or by email and addressed or delivered to such party at its address set forth below or at such other address as may be designated by such party in a notice to the other parties. Any notice, if mailed and properly addressed with postage prepaid, shall be deemed given when received; any notice, if transmitted by email, shall be deemed given when transmitted (provided that, if such notice is transmitted by email, such notice is also promptly mailed in accordance with this paragraph). Notice may be given as follows:

To the Borrower:

First Industrial, L.P. c/o First Industrial Realty Trust, Inc. 1 N. Wacker Drive Suite 4200 Chicago, Illinois 60606 Attention: Mr. Scott Musil

Email: smusil@firstindustrial.com

To the General Partner:

First Industrial Realty Trust, Inc. 1 N. Wacker Drive **Suite 4200** Chicago, Illinois 60606 Attention: Scott A. Musil Email: smusil@firstindustrial.com

Each of the above with a copy to:

Barack Ferrazzano Kirschbaum & Nagelberg LLP 200 West Madison 39th Floor Chicago, Illinois 60606 Attention: Suzanne Bessette-Smith and Matthew A. Jackson

Email: Suzanne.bessette-smith@bfkn.com; matt.jackson@bfkn.com

To each Lender:

At such address as set forth in its Administrative Questionnaire

To the Administrative Agent:

Wells Fargo Bank, National Association, as agent 10 South Wacker Drive, 32nd Floor Chicago, IL 60606 Attention: Scott S. Solis Email: scott.s.solis@wellsfargo.com

with a copy to:

Wells Fargo Bank, National Association, as agent Commercial Banking – Wholesale Loan Center MAC: N9300-085

600 S. 4th Street, Floor 08 Minneapolis, MN 55415-1526 Attention: David DeAngelis

Documents required to be delivered pursuant to the Loan Documents may be delivered by electronic communication and delivery, including, the Internet, e-mail or intranet websites to which the Administrative Agent and each Lender have access (including a commercial, third-party website such as www.sec.gov http://www.sec.gov or a website sponsored or hosted by the Administrative Agent or the Borrower) provided that the foregoing shall not apply to (i) notices to any Lender pursuant to Article II and (ii) any Lender that has notified the Administrative Agent and the Borrower that it cannot or does not want to receive electronic communications. The Administrative Agent or the Borrower may, in its discretion, agree to accept notices and other communications to it hereunder by electronic delivery pursuant to procedures approved by it for all or particular notices or communications. Documents or notices delivered electronically shall be deemed to have been delivered twenty-four (24) hours after the date and time on which the Administrative Agent or the Borrower posts such documents or the documents become available on a commercial website and the Administrative Agent or Borrower notifies each Lender of said posting and provides a link thereto provided if such notice or other communication is not sent or posted during the normal business hours of the recipient, said posting date and time shall be deemed to have commenced as of 11:00 a.m. (Central Time) on the opening of business on the next business day for the recipient. Notwithstanding anything contained herein, in every instance the Borrower shall be required to provide paper copies of the certificate required by Section 8.2(iv) to the Administrative Agent and shall deliver paper copies of any documents to the Administrative Agent or to any Lender that requests such paper copies until a written request to cease delivering paper copies is given by the Administrative Agent or such Lender. Except for the certificates required by Section 8.2(iv), the Administrative Agent shall have no obligation to request the delivery of or to maintain paper copies of the documents delivered electronically, and in any event shall have no responsibility to monitor compliance by the Borrower with any such request for delivery. Each Lender shall be solely responsible for requesting delivery to it of paper copies and maintaining its paper or electronic documents.

Documents required to be delivered pursuant to <u>Article II</u> may be delivered electronically to a website provided for such purpose by the Administrative Agent pursuant to the procedures provided to the Borrower by the Administrative Agent.

15.2.	Change of Address. Each party may change the address for service of notice upon it by a notice in writing to the other
parties hereto.	

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, the undersigned have executed this Agreement as of the date first above written.

BORROWER: FIRST INDUSTRIAL, L.P.

By: FIRST INDUSTRIAL REALTY TRUST, INC., its General Partner

By: /s/ Scott A. Musil
Name: Scott A. Musil
Title: Chief Financial Officer

GENERAL PARTNER: FIRST INDUSTRIAL REALTY TRUST, INC.

By: <u>/s/ Scott A. Musil</u>
Name: Scott A. Musil
Title: Chief Financial Officer

LENDERS:

WELLS FARGO BANK, NATIONAL ASSOCIATION, individually as a Lender and as Administrative Agent

By: <u>/s/ Craig V. Koshkarian</u> Name: Craig V. Koshkarian Title: Director

PNC BANK, NATIONAL ASSOCIATION, individually as a Lender and as Syndication Agent

By: <u>/s/ Margaret K. Grady</u>
Name: Margaret Kilner Grady

Title: Vice President

REGIONS BANK, as a Lender

By: <u>/s/ Susan Wolfe</u> Name: Susan Wolfe

Title: Assistant Vice President

FIFTH THIRD BANK, NATIONAL ASSOCIATION, as a Lender

By: <u>/s/ Michael Glandt</u> Name: Michael Glandt Title: Vice President

U.S. BANK NATIONAL ASSOCIATION, as a Lender

By: <u>/s/ Donald J. Pafford</u>
Name: Donald J. Pafford
Title: Senior Vice President

EXHIBIT A

TERM LOAN COMMITMENT AMOUNTS (As of the Agreement Effective Date)

Bank	Total Commitment
Wells Fargo Bank, National Association	\$67,500,000.00
PNC Bank, National Association	\$67,500,000.00
Regions Bank	\$25,000,000.00
Fifth Third Bank, National Association	\$25,000,000.00
U.S. Bank National Association	\$15,000,000.00
Total	\$200,000,000.00

EXHIBIT B

FORM OF NOTE

TERM LOAN NOTE

1, 2021

On or before the Maturity Date, as defined in that certain Amended and Restated Unsecured Term Loan Agreement dated as of
July 7, 2021 (the "Agreement") between FIRST INDUSTRIAL, L.P., a Delaware limited partnership ("Borrower"), First Industrial
Realty Trust, Inc., a Maryland corporation, PNC Bank, National Association, individually and as Syndication Agent, Wells Fargo
Bank, National Association, individually and as Administrative Agent for the Lenders (as such terms are defined in the Agreement),
and the other Lenders listed on the signature pages of the Agreement, Borrower promises to pay to the order of [] (the
"Lender"), or its successors and assigns, the aggregate unpaid principal amount of all Loans made by the Lender to the Borrower
pursuant to Section 2.1 of the Agreement, in immediately available funds at the office of the Administrative Agent in Minneapolis,
Minnesota, together with interest on the unpaid principal amount hereof at the rates and on the dates set forth in the Agreement. The
Borrower shall pay this Promissory Note ("Note") in full on or before the Maturity Date in accordance with the terms of the

The Lender shall, and is hereby authorized to record on the schedule attached hereto, or to otherwise record in accordance with its usual practice, the date and amount of each Borrowing and the date and amount of each principal payment hereunder; provided, however, that the failure of Lender to so record shall not affect the obligations of the Borrower hereunder or under the other Loan. Documents.

Agreement.

This Note is issued pursuant to, and is entitled to the security under and benefits of, the Agreement and the other Loan Documents, to which Agreement and Loan Documents, as they may be amended from time to time, reference is hereby made for, inter alia, a statement of the terms and conditions under which this Note may be prepaid or its maturity date accelerated. Capitalized terms used herein and not otherwise defined herein are used with the meanings attributed to them in the Agreement.

If there is an Event of Default or Default under the Agreement or any other Loan Document and Lender exercises its remedies provided under the Agreement and/or any of the Loan Documents, then in addition to all amounts recoverable by the Lender under such documents, Lender shall be entitled to receive reasonable attorneys fees and expenses incurred by Lender in exercising such remedies.

Borrower and all endorsers severally waive presentment, protest and demand, notice of protest, demand and of dishonor and nonpayment of this Note (except as otherwise expressly provided for in the Agreement), and any and all lack of diligence or delays in collection or enforcement of this Note, and expressly agree that this Note, or any payment hereunder, may be extended from time to time, and expressly consent to the release of any party liable for the obligation secured by this Note, the release of any of the security of this Note, the acceptance of any other security therefor, or any other indulgence or forbearance whatsoever, all without notice to any party and without affecting the liability of the Borrower and any endorsers hereof.

This Note shall, be governed and construed under the internal laws of the State of Illinois.

BORROWER AND LENDER, BY ITS ACCEPTANCE HEREOF, EACH HEREBY WAIVE ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION OR PROCEEDING TO ENFORCE OR DEFEND ANY RIGHT UNDER THIS PROMISSORY NOTE OR ANY OTHER LOAN DOCUMENT OR RELATING THERETO OR ARISING FROM THE LENDING RELATIONSHIP WHICH IS THE SUBJECT OF THIS NOTE AND AGREE THAT ANY SUCH ACTION OR. PROCEEDING SHALL BE TRIED BEFORE A COURT AND NOT BEFORE A JURY.

[Signature on Following Page]

FIRST INDUSTRIAL, L.P.

By: First Industrial Realty Trust, Inc., its general partne	By: Fir	st Industria	l Realty	Trust,	Inc., i	its general	partner
---	---------	--------------	----------	--------	---------	-------------	---------

Ву:	
Name:	
Title:	

PAYMENTS OF PRINCIPAL

Date Unpaid Principal Balance Notation Made by

EXHIBIT C

DISBURSEMENT INSTRUCTION AGREEMENT

Borrower: First Industrial, L.P.
Administrative Agent: Wells Fargo Bank, National Association
Loan: Loan number [] made pursuant to that certain "Amended and Restated Unsecured Term Loan Agreement" dated as of July 7, 2021 between Borrower, Administrative Agent and Lenders, as amended from time to time (the " <u>Loan Agreement</u> ")
Effective Date: [], 2021
Check applicable box:
New – This is the first Disbursement Instruction Agreement submitted in connection with the Loan. Replace Previous Agreement – This is a replacement Disbursement Instruction Agreement. All prior instructions submitted in connection with this Loan are cancelled as of the Effective Date set forth above.

This Agreement must be signed by the Borrower and is used for the following purposes:

- (1) to designate an individual or individuals with authority to request disbursements of Loan proceeds, whether at the time of Loan closing/origination or thereafter;
- (2) to designate an individual or individuals with authority to request disbursements of funds from Restricted Accounts (as defined in the Terms and Conditions attached to this Agreement), if applicable; and
- (3) to provide Administrative Agent with specific instructions for wiring or transferring funds on Borrower's behalf.

Any of the disbursements, wires or transfers described above are referred to herein as a "Disbursement."

Specific dollar amounts for Disbursements must be provided to Administrative Agent at the time of the applicable Disbursement in the form of a signed closing statement, an email instruction or other written communication, or telephonic request pursuant to 2.7 of the Loan Agreement (each, a "**Disbursement Request**") from an applicable Authorized Representative (as defined in the Terms and Conditions attached to this Agreement).

A new Disbursement Instruction Agreement must be completed and signed by the Borrower if (i) all or any portion of a Disbursement is to be transferred to an account or an entity not described in this Agreement or (ii) Borrower wishes to add or remove any Authorized Representatives.

See the Additional Terms and Conditions attached hereto for additional information and for definitions of certain capitalized terms used in this Agreement.

Disb	ursement of Loan Proceeds at Origination/Closing	
indiv		authorized to accept one or more Disbursement Requests from any of the Authorizer ") to disburse Loan proceeds on or about the date of the Loan ion therewith (each, a " Closing Disbursement "):
	Individual's Name	Title
1.		
2.		
3.		
destin DESC If the	nations, etc.): CRIBE APPLICABLE RESTRICTIONS OR INDICAT	Disbursement Authorizers (dollar amount limits, wire/deposit TE "N/A" Disbursement Authorizer may submit a Disbursement Request for all
DELE	TE FOLLOWING SECTION IF NO WIRE TRANSFE	ERS AT ORIGINATION/CLOSING
amou Admi Rece	int and applicable Receiving Party. Each Receiving Par inistrative Agent is authorized to use the wire instruction	osing Disbursement(s) to be made by wire transfer must specify the ty included in any such Disbursement Request must be listed below. One that have been provided directly to Administrative Agent by the chibit. All wire instructions must be in the format specified on the
	Names of Receiving Parties for the Closing Disburser Receiving Party must be attached as the Closing Exhi	ment(s) (may include as many parties as needed; wire instructions for each
1.		
2.		
3.		
DELE	TE FOLLOWING SECTION IF NO DEPOSITS INTO	O WFB ACCOUNTS AT ORIGINATION/CLOSING
		oursement(s) to be deposited into an account at Wells Fargo Bank, N.A. unt included in any such Disbursement Request must be listed below.
Name	e on Deposit Account:	
Wells	Fargo Bank, N.A. Deposit Account Number:	
Furth	er Credit Information/Instructions:	

Disbu	rsements of Loan Proceeds Subsequent to Loan Closing	Origination (Origination (Origi
the in		uthorized to accept one or more Disbursement Requests from any of at Authorizer ") to disburse Loan proceeds after the date of the Loan erewith (each, a " Subsequent Disbursement "):
	Individual's Name	Title
1.		
2.		
3.		
destir DESC If the	lations, etc.): CRIBE APPLICABLE RESTRICTIONS OR INDICATE "N	Disbursement Authorizers (dollar amount limits, wire/deposit //A" Sbursement Authorizer may submit a Disbursement Request for
DELE	TE FOLLOWING SECTION IF NO SUBSEQUENT WIRE	TRANSFERS ANTICIPATED
amou Admi Recei	nt and applicable Receiving Party. Each Receiving Party inc nistrative Agent is authorized to use the wire instructions tha	Disbursements to be made by wire transfer must specify the luded in any such Disbursement Request must be listed below. It have been provided directly to Administrative Agent by the bursement Exhibit. All wire instructions must be in the format
	Names of Receiving Parties for Subsequent Disbursements Receiving Party must be attached as the Subsequent Disbur	(may include as many parties as needed; wire instructions for each rement Exhibit)
1.	,	·
2.		
3.		
DELE'	TE FOLLOWING SECTION IF NO SUBSEQUENT DEPC	OSITS INTO WFB ACCOUNTS ANTICIPATED
		nents to be deposited into an account at Wells Fargo Bank, N.A. cluded in any such Disbursement Request must be listed below.
Name	on Deposit Account:	
Wells	Fargo Bank, N.A. Deposit Account Number:	
Furth	er Credit Information/Instructions:	

J	and Conditions on the following page.	is correct and agrees to the terms and	conditions set forth heren
FIRST INDUSTRIAL, L.P.			
By: FIRST INDUSTRIAL F	REALTY TRUST, INC.		
General Partner			

By:		
Print Name: _		
Title:		

Additional Terms and Conditions to the Disbursement Instruction Agreement

Definitions. The following capitalized terms shall have the meanings set forth below:

- "Authorized Representative" means any or all of the Closing Disbursement Authorizers, Subsequent Disbursement Authorizers and Restricted Account Disbursement Authorizers, as applicable.
- "Receiving Bank" means the financial institution where a Receiving Party maintains its account.
- "Receiving Party" means the ultimate recipient of funds pursuant to a Disbursement Request.
- "Restricted Account" means an account at Wells Fargo Bank, N.A. associated with the Loan to which Borrower's access is restricted.

Capitalized terms used in these Additional Terms and Conditions to Disbursement Instruction Agreement and not otherwise defined herein shall have the meanings given to such terms in the body of the Agreement.

Disbursement Requests. Except as expressly provided in the Loan Agreement, Administrative Agent must receive Disbursement Requests in writing. Disbursement Requests will only be accepted from the applicable Authorized Representatives designated in the Disbursement Instruction Agreement. Disbursement Requests will be processed subject to satisfactory completion of Administrative Agent's customer verification procedures. Administrative Agent is only responsible for making a good faith effort to execute each Disbursement Request and may use agents of its choice to execute Disbursement Requests. Funds disbursed pursuant to a Disbursement Request may be transmitted directly to the Receiving Bank, or indirectly to the Receiving Bank through another bank, government agency, or other third party that Administrative Agent considers to be reasonable. Administrative Agent will, in its sole discretion, determine the funds transfer system and the means by which each Disbursement will be made. Administrative Agent may delay or refuse to accept a Disbursement Request if the Disbursement would: (i) violate the terms of this Agreement; (ii) require use of a bank unacceptable to Administrative Agent or Lenders or prohibited by government authority; (iii) cause Administrative Agent or Lenders to violate any Federal Reserve or other regulatory risk control program or guideline; or (iv) otherwise cause Administrative Agent or Lenders to violate any Applicable Law or regulation.

Limitation of Liability. Administrative Agent and Lenders shall not be liable to Borrower or any other parties for: (i) errors, acts or failures to act of others, including other entities, banks, communications carriers or clearinghouses, through which Borrower's requested Disbursements may be made or information received or transmitted, and no such entity shall be deemed an agent of the Administrative Agent or any Lender; (ii) any loss, liability or delay caused by fires, earthquakes, wars, civil disturbances, power surges or failures, acts of government, labor disputes, failures in communications networks, legal constraints or other events beyond Administrative Agent's or any Lender's control; or (iii) any special, consequential, indirect or punitive damages, whether or not (A) any claim for these damages is based on tort or contract or (B) Administrative Agent any Lender or Borrower knew or should have known the likelihood of these damages in any situation. Neither

Administrative Agent nor any Lender makes any representations or warranties other than those expressly made in this Agreement. IN NO EVENT WILL ADMINISTRATIVE AGENT OR ANY LENDER BE LIABLE FOR DAMAGES ARISING DIRECTLY OR INDIRECTLY IF A DISBURSEMENT REQUEST IS EXECUTED BY ADMINISTRATIVE AGENT IN GOOD FAITH AN IN ACCORDANCE WITH THE TERMS OF THIS AGREEMENT.

Reliance on Information Provided. Administrative Agent is authorized to rely on the information provided by Borrower or any Authorized Representative in or in accordance with this Agreement when executing a Disbursement Request until Administrative Agent has received a new Agreement signed by Borrower. Borrower agrees to be bound by any Disbursement Request: (i) authorized or transmitted by Borrower; or (ii) made in Borrower's name and accepted by Administrative Agent in good faith and in compliance with this Agreement, even if not properly authorized by Borrower. Administrative Agent may rely solely (i) on the account number of the Receiving Party, rather than the Receiving Party's name, and (ii) on the bank routing number of the Receiving Bank, rather than the Receiving Bank's name, in executing a Disbursement Request. Administrative Agent is not obligated or required in any way to take any actions to detect errors in information provided by Borrower or an Authorized Representative. If Administrative Agent takes any actions in an attempt to detect unauthorized Disbursement Requests, Borrower agrees that, no matter how many times Administrative Agent takes these actions, Administrative Agent will not in any situation be liable for failing to take or correctly perform these actions in the future, and such actions shall not become any part of the Disbursement procedures authorized herein, in the Loan Documents, or in any agreement between Administrative Agent and Borrower.

International Disbursements. A Disbursement Request expressed in US Dollars will be sent in US Dollars, even if the Receiving Party or Receiving Bank is located outside the United States. Administrative Agent will not execute Disbursement Requests expressed in foreign currency unless permitted by the Loan Agreement.

Errors. Borrower agrees to notify Administrative Agent of any errors in the Disbursement of any funds or of any unauthorized or improperly authorized Disbursement Requests within fourteen (14) days after Administrative Agent's confirmation to Borrower of such Disbursement.

Finality of Disbursement Requests. Disbursement Requests will be final and will not be subject to stop payment or recall; provided that Administrative Agent may, at Borrower's request, make an effort to effect a stop payment or recall but will incur no liability whatsoever for its failure or inability to do so.

CLOSING EXHIBIT WIRE INSTRUCTIONS

ADMINISTRATIVE AGENT TO ATTACH WIRE INSTRUCTIONS FROM RECEIVING PARTIES

All wire instructions must contain the following information:

Transfer/Deposit Funds to (Receiving Party Account Name)
Receiving Party Deposit Account Number
Receiving Bank Name, City and State
Receiving Bank Routing (ABA) Number
Further identifying information, if applicable (title escrow number, borrower name, loan number, etc.)

SUBSEQUENT DISBURSEMENT EXHIBIT WIRE INSTRUCTIONS

ADMINISTRATIVE AGENT TO ATTACH WIRE INSTRUCTIONS FROM RECEIVING PARTIES

All wire instructions must contain the following information:

Transfer/Deposit Funds to (Receiving Party Account Name)
Receiving Party Deposit Account Number
Receiving Bank Name, City and State
Receiving Bank Routing (ABA) Number
Further identifying information, if applicable (title escrow number, borrower name, loan number, etc.)

EXHIBIT D

FORM OF GUARANTY

AMENDED AND RESTATED GUARANTY

This Amended and Restated Guaranty (this "<u>Guaranty</u>") is made as of July 7, 2021, by First Industrial Realty Trust, Inc., a Maryland corporation ("<u>Guarantor</u>"), to and for the benefit of Wells Fargo Bank, National Association, a national banking association, individually ("<u>Wells</u>"), and as administrative agent for itself and the lenders listed on the signature pages of the Loan Agreement (as defined below) and their respective successors and assigns (collectively, "<u>Lender</u>").

RECITALS

- A. First Industrial, L.P., a Delaware limited partnership ("<u>Borrower</u>"), and Guarantor have requested that Lender make unsecured term loans available to Borrower in the aggregate principal amount of up to \$200,000,000 (subject to Borrower's right to increase such aggregate principal amount to \$460,000,000 in accordance with <u>Section 2.2</u> of the Loan Agreement (herein defined)) (the "<u>Facility</u>").
- B. Lender has agreed to make available the Facility to Borrower pursuant to the terms and conditions set forth in an Amended and Restated Unsecured Term Loan Agreement bearing even date herewith between Borrower, the Lenders and Guarantor (as amended, restated, supplemented or otherwise modified from time to time, the "Loan Agreement"). All capitalized terms used herein and not otherwise defined shall have the meanings ascribed to such terms in the Loan Agreement.
- C. Guarantor is party to that certain Guaranty, dated July 15, 2020 (as amended, restated, supplemented or otherwise modified from time to time, the "Existing Guaranty").
- D. The parties hereto wish to amend and restate the Existing Guaranty in its entirety and the Guarantor wishes to affirm its obligations under the Existing Guaranty.
- E. Guarantor is the sole general partner of Borrower and, therefore, Guarantor will derive financial benefit from the Facility evidenced by the Loan Agreement and the other Loan Documents. The execution and delivery of this Guaranty by Guarantor is a condition precedent to the performance by Lender of its obligations under the Loan Agreement.

AGREEMENTS

NOW, THEREFORE, Guarantor, in consideration of the matters described in the foregoing Recitals, which Recitals are incorporated herein and made a part hereof, and for other good and valuable consideration, hereby agrees as follows:

1. Guarantor affirms its obligations under, and the terms and conditions of, the Existing

Guaranty and agrees that such obligations remain in full force and effect and are hereby ratified, reaffirmed and confirmed. Guarantor hereby absolutely, unconditionally, and irrevocably guarantees to Lender:

- (a) the full and prompt payment of the principal of and interest on the Loans when due, whether at stated maturity, upon acceleration or otherwise, and at all times thereafter, and the prompt payment of all sums which may now be or may hereafter become due and owing under the Loan Agreement and the other Loan Documents;
 - (b) the payment of all Enforcement Costs (as hereinafter defined in Paragraph 7 hereof);
- (c) the full, complete, and punctual observance, performance, and satisfaction of all of the obligations, duties, covenants, and agreements of Borrower under the Loan Agreement and the Loan Documents; and
- (d) All amounts due, debts, liabilities, and payment obligations described in subparagraphs (a) and (b) of this Paragraph 1 are referred to herein as the "Facility Indebtedness." All obligations described in subparagraph (c) of this Paragraph 1 are referred to herein as the "Obligations."
- 2. In the event of any default by Borrower in making payment of the Facility Indebtedness, or in performance of the Obligations, as aforesaid, in each case beyond the expiration of any applicable grace period, Guarantor agrees, on demand by Lender or any holder of any Note, to pay all the Facility Indebtedness and to perform all the Obligations as are or then or thereafter become due and owing or are to be performed under the terms of the Loan Agreement and the other Loan Documents, and to pay any reasonable expenses incurred by Lender in protecting, preserving, or defending its interest in the Property or in connection with the Facility or under any of the Loan Documents, including, without limitation, all reasonable attorneys' fees and costs. Lender shall have the right, at its option, either before, during or after pursuing any other right or remedy against Borrower or Guarantor, to perform any and all of the Obligations by or through any agent. contractor or subcontractor, or any of their agents, of its selection, all as Lender in its sole discretion deems proper, and Guarantor shall indemnify and hold Lender free and harmless from and against any and all loss, damage, cost, expense, injury, or liability Lender may suffer or incur in connection with the exercise of its rights under this Guaranty or the performance of the Obligations, except to the extent the same arises as determined by a court of competent jurisdiction by a final and non-appealable judgment to have resulted from the Gross Negligence or willful misconduct of Lender.

All of the remedies set forth herein and/or provided by any of the Loan Documents or law or equity shall be equally available to Lender, and the choice by Lender of one such alternative over another shall not be subject to question or challenge by Guarantor or any other person, nor shall any such choice be asserted as a defense, set-off, or failure to mitigate damages in any action, proceeding, or counteraction by Lender to recover or seeking any other remedy under this Guaranty, nor shall such choice preclude Lender from subsequently electing to exercise a different remedy. The parties have agreed to the alternative remedies hereinabove specified in part because they recognize that the choice of remedies in the event of a failure hereunder will necessarily be and should properly be a

matter of business judgment, which the passage of time and events may or may not prove to have been the best choice to maximize recovery by Lender at the lowest cost to Borrower and/or Guarantor. It is the intention of the parties that such choice by Lender be given conclusive effect regardless of such subsequent developments.

- Guarantor does hereby waive (i) notice of acceptance of this Guaranty by Lender and any and all notices and demands of every kind which may be required to be given by any statute, rule or law, (ii) any defense, right of set-off or other claim which Guarantor may have against the Borrower or which Guarantor or Borrower may have against Lender or the holders of any Note (other than defenses relating to payment of the Facility Indebtedness or the correctness of any allegation by Lender that Borrower was in default in the performance of the Obligations), (iii) presentment for payment, demand for payment (other than as provided for in Paragraph 2 above), notice of nonpayment (other than as provided for in Paragraph 2 above) or dishonor, protest and notice of protest, diligence in collection and any and all formalities which otherwise might be legally required to charge Guarantor with liability, (iv) any failure by Lender to inform Guarantor of any facts Lender may now or hereafter know about Borrower, the Facility, or the transactions contemplated by the Loan Agreement, it being understood and agreed that Lender has no duty so to inform and that the Guarantor is fully responsible for being and remaining informed by the Borrower of all circumstances bearing on the existence or creation, or the risk of nonpayment of the Facility Indebtedness or the risk of nonperformance of the Obligations, and (v) any and all right to cause a marshalling of assets of the Borrower or any other action by any court or governmental body with respect thereto, or to cause Lender to proceed against any other security given to Lender in connection with the Facility Indebtedness or the Obligations. Credit may be granted or continued from time to time by Lender to Borrower without notice to or authorization from Guarantor, regardless of the financial or other condition of the Borrower at the time of any such grant or continuation. Lender shall have no obligation to disclose or discuss with Guarantor its assessment of the financial condition of Borrower. Guarantor acknowledges that no representations of any kind whatsoever have been made by Lender to Guarantor. No modification or waiver of any of the provisions of this Guaranty shall be binding upon Lender except as expressly set forth in a writing duly signed and delivered on behalf of Lender. Guarantor further agrees that any exculpatory language contained in the Loan Agreement and the Notes shall in no event apply to this Guaranty, and will not prevent Lender from proceeding against Guarantor to enforce this Guaranty.
- 4. Guarantor further agrees that Guarantor's liability as guarantor shall in no way be impaired by any renewals or extensions which may be made from time to time, with or without the knowledge or consent of Guarantor of the time for payment of interest or principal under the Loan Agreement and the other Loan Documents or by any forbearance or delay in collecting interest or principal under the Loan Agreement and the other Loan Documents, or by any waiver by Lender under the Loan Agreement or any other Loan Documents, or by Lender's failure or election not to pursue any other remedies it may have against Borrower, or by any change or modification in the Loan Agreement or any other Loan Documents, or by the acceptance by Lender of any additional security or any increase, substitution or change therein, or by the release by Lender of any security or any withdrawal thereof or decrease therein, or by the application of payments received from any source to the payment of any obligation other than the Facility Indebtedness, even though Lender might lawfully have elected to apply such payments to any part or all of the Facility Indebtedness, it being

the intent hereof that Guarantor shall remain liable as principal for payment of the Facility Indebtedness and performance of the Obligations until all indebtedness has been paid in full and the other terms, covenants and conditions of the Loan Agreement and other Loan Documents and this Guaranty have been performed, notwithstanding any act or thing which might otherwise operate as a legal or equitable discharge of a surety. For the avoidance of doubt, it is understood and agreed that the Guarantor shall not be a "surety" for purposes of the Securities Act of Illinois (740 ILCS 155 et. seq.) and the Guarantor hereby waives any rights or benefits thereunder. Guarantor further understands and agrees that Lender may at any time enter into agreements with Borrower to amend and modify the Loan Agreement or other Loan Documents, or any thereof (including, without limitation in accordance with Borrower's right to increase the aggregate principal balance of the Loan pursuant to Section 2.2 of the Loan Agreement), and may waive or release any provision or provisions of the Loan Agreement and other Loan Documents or any thereof, and, with reference to such instruments, may make and enter into any such agreement or agreements as Lender and Borrower may deem proper and desirable, without in any manner impairing this Guaranty or any of Lender's rights hereunder or any of the Guarantor's obligations hereunder.

This is an absolute, unconditional, complete, present and continuing guaranty of payment and performance and not of collection. Guarantor agrees that this Guaranty may be enforced by Lender without the necessity at any time of resorting to or exhausting any other security or collateral given in connection herewith or with the Loan Agreement or any of the other Loan Documents, or resorting to any other guaranties, and Guarantor hereby waives the right to require Lender to join Borrower in any action brought hereunder or to commence any action against or obtain any judgment against Borrower or to pursue any other remedy or enforce any other right. Guarantor further agrees that nothing contained herein or otherwise shall prevent Lender from pursuing concurrently or successively all rights and remedies available to it at law and/or in equity or under the Loan Agreement or any other Loan Documents, and the exercise of any of its rights or the completion of any of its remedies shall not constitute a discharge of any of Guarantor's obligations hereunder, it being the purpose and intent of the Guarantor that the obligations of such Guarantor hereunder shall be primary, absolute, independent and unconditional under any and all circumstances whatsoever. Neither Guarantor's obligations under this Guaranty nor any remedy for the enforcement thereof shall be impaired, modified, changed or released in any manner whatsoever by any impairment, modification, change, release or limitation of the liability of Borrower under the Loan Agreement or other Loan Documents or by reason of Borrower's bankruptcy or by reason of any creditor or bankruptcy proceeding instituted by or against Borrower. This Guaranty shall continue to be effective and be deemed to have continued in existence ore reinstated (as the case may be) if at any time payment of all or any part of any sum payable pursuant to the Loan Agreement or any other Loan Document is rescinded or otherwise required to be returned by the payee upon the insolvency, bankruptcy, or reorganization of the payor, all as though such payment to Lender had not been made, regardless of whether Lender contested the order requiring the return of such payment. The obligations of Guarantor pursuant to the preceding sentence shall survive any termination, cancellation, or release of this Guaranty.

- 6. This Guaranty shall be assignable by Lender to any assignee of all or a portion of Lender's rights under the Loan Documents.
- 7. If: (i) this Guaranty, the Loan Agreement or any other Loan Document is placed in the hands of an attorney for collection or is collected through any legal proceeding; (ii) an attorney is retained to represent Lender in any bankruptcy, reorganization, receivership, or other proceedings affecting creditors' rights and involving a claim under this Guaranty, the Loan Agreement or any Loan Document; (iii) an attorney is retained to provide advice or other representation with respect to the Loan Documents in connection with an enforcement action or potential enforcement action; or (iv) an attorney is retained to represent Lender in any other legal proceedings whatsoever in connection with this Guaranty, the Loan Agreement, any of the Loan Documents, or any property subject thereto (other than any action or proceeding brought by any Lender or participant against the Administrative Agent (as defined in the Loan Agreement) alleging a breach by the Administrative Agent of its duties under the Loan Documents), then Guarantor shall pay to Lender upon demand all reasonable attorney's fees, costs and expenses, including, without limitation, court costs, filing fees, recording costs, expenses of foreclosure, title insurance premiums, survey costs, minutes of foreclosure, and all other costs and expenses incurred in connection therewith (all of which are referred to herein as "Enforcement Costs"), in addition to all other amounts due hereunder.
- 8. The parties hereto intend that each provision in this Guaranty comports with all applicable local, state and federal laws and judicial decisions. However, if any provision or provisions, or if any portion of any provision or provisions, in this Guaranty is found by a court of law to be in violation of any applicable local, state or federal ordinance, statute, law, administrative or judicial decision, or public policy, and if such court should declare such portion, provision or provisions of this Guaranty to be illegal, invalid, unlawful, void or unenforceable as written, then it is the intent of all parties hereto that such portion, provision or provisions shall be given force to the fullest possible extent that they are legal, valid and enforceable, that the remainder of this Guaranty shall be construed as if such illegal, invalid, unlawful, void or unenforceable portion, provision or provisions were not contained therein, and that the rights, obligations and interest of Lender or the holder of any Note under the remainder of this Guaranty shall continue in full force and effect.
- 9. Any indebtedness of Borrower to Guarantor now or hereafter existing is hereby subordinated to the Facility Indebtedness. Guarantor agrees that until the entire Facility Indebtedness has been paid in full, (i) Guarantor will not seek, accept, or retain for Guarantor's own account any payment from Borrower on account of such subordinated debt, and (ii) any such payments to Guarantor on account of such subordinated debt shall be collected and received by Guarantor in trust for Lender and shall be paid over to Lender on account of the Facility Indebtedness without impairing or releasing the obligations of Guarantor hereunder.
- 10. Guarantor waives and releases any claim (within the meaning of 11 U.S.C. § 101) which Guarantor may have against Borrower arising from a payment made by Guarantor under this Guaranty and agrees not to assert or take advantage of any subrogation rights of Guarantor or Lender or any right of Guarantor or Lender to proceed against (i) Borrower for reimbursement, or (ii) any other guarantor or any collateral security or guaranty or right of offset held by Lender for the payment of the Facility Indebtedness and performance of the Obligations, nor shall Guarantor seek or be entitled to seek any contribution or reimbursement from Borrower or any other guarantor in respect of payments made by Guarantor hereunder. It is expressly understood that the waivers and

agreements of Guarantor set forth above constitute additional and cumulative benefits given to Lender for its security and as an inducement for its extension of credit to Borrower. Nothing contained in this <u>Paragraph 10</u> is intended to prohibit Guarantor from making all distributions to its constituent shareholders which are required by law from time to time in order for Guarantor to maintain its status as a real estate investment trust in compliance with all applicable provisions of the Code (as defined in the Loan Agreement).

- 11. Any amounts received by Lender from any source on account of any indebtedness may be applied by Lender toward the payment of such indebtedness, and in such order of application, as Lender may from time to time elect.
- 12. The Guarantor hereby submits to personal jurisdiction in the State of Illinois for the enforcement of this Guaranty and waives any and all personal rights to object to such jurisdiction for the purposes litigation to enforce this Guaranty. Guarantor hereby consents to the jurisdiction of either the Circuit Court of Cook County, Illinois, or the United States District Court for the Northern District of Illinois, in any action, suit, or proceeding which Lender may at any time wish to file in connection with this Guaranty or any related matter. Guarantor hereby agrees that an action, suit, or proceeding to enforce this Guaranty may be brought in any state or federal court in the State of Illinois and hereby waives any objection which Guarantor may have to the laying of the venue of any such action, suit, or proceeding in any such court; provided, however, that the provisions of this Paragraph shall not be deemed to preclude Lender from filing any such action, suit, or proceeding in any other appropriate forum.
- 13. Anything contained herein to the contrary notwithstanding, the obligations of Guarantor hereunder at any time shall be limited to an aggregate amount equal to the largest amount that would not render its obligations hereunder subject to avoidance as a fraudulent transfer or conveyance under Section 548 of the United States Bankruptcy Code, as amended, or any comparable provisions of any similar federal or state law.
- 14. All notices and other communications provided to any party hereto under this Agreement or any other Loan Document shall be in writing or by email and addressed or delivered to such party at its address set forth below or at such other address as may be designated by such party in a notice to the other parties. Any notice, if mailed and properly addressed with postage prepaid, shall be deemed given when received; any notice, if transmitted by email, shall be deemed given when transmitted (provided, if such notice is transmitted by email, that such notice is then also promptly mailed and properly addressed with postage prepaid in the cases of email in accordance with this paragraph). Notice may be given as follows:

To the Guarantor: First Industrial Realty Trust, Inc. 1 N. Wacker Drive, Suite 4200 Chicago, Illinois 60606 Attention: Mr. Scott A. Musil

Email: smusil@firstindustrial.com

With a copy to:

and shall inure to the benefit of Lender's successors and assigns.

Barack Ferrazzano Kirschbaum & Nagelberg LLP 200 West Madison 39th Floor Chicago, Illinois 60606 Attention: Suzanne Bessette-Smith and Matthew A. Jackson

Email: Suzanne.bessette-smith@bfkn.com; matt.jackson@bfkn.com

To the Lender:

c/o Wells Fargo Bank, National Association as agent 10 South Wacker Drive, 32nd Floor MAC: N8405-321 Chicago, IL 60606 Attention: Karen T. Skutt Email: berschkt@wellsfargo.com

or at such other address as the party to be served with notice may have furnished in writing to the party seeking or desiring to serve

- notice as a place for the service of notice.

 14. This Guaranty shall be binding upon the heirs, executors, legal and personal representatives, successors and assigns of Guarantor
- 15. This Guaranty shall be construed and enforced under the internal law of the State of Illinois.
- 16. GUARANTOR AND LENDER, BY ITS ACCEPTANCE HEREOF, EACH HEREBY WAIVE ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION OR PROCEEDING TO ENFORCE OR DEFEND ANY RIGHT UNDER THIS GUARANTY OR ANY OTHER LOAN DOCUMENT OR RELATING THERETO OR ARISING FROM THE LENDING RELATIONSHIP WHICH IS THE SUBJECT OF THIS GUARANTY AND AGREE THAT ANY SUCH ACTION OR PROCEEDING SHALL BE TRIED BEFORE A COURT AND NOT BEFORE A JURY.

IN WITNESS WHEREOF, Guarantor has delivered this Guaranty	y in the State of Illinois as of the date first written above
---	---

FIRST INDUSTRIAL REALTY TRUST, INC., a Maryland corporation
By:
Name:
Title:

STATE OF ILLINOIS)
) SS.
COUNTY OF COOK)
I, the undersigned, a Notary Public, in and for said County, in the State aforesaid, DO HEREBY CERTIFY, that of First Industrial Realty Trust, Inc., personally known to me to be the same person whose name is subscribed to the foregoing instrument, appeared before me this day in person and acknowledged that he signed and delivered the said instrument as his own free and voluntary act and as the free and voluntary act of said corporation, for the uses and purposes therein set forth. GIVEN under my hand and Notarial Seal, this day of, 2021.
Notary Public

Acknowledged and Agreed by:
WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent
By:
Name:
Title:

EXHIBIT E

OPINION OF BORROWER'S COUNSEL

Attached

EXHIBIT F OPINION OF GENERAL PARTNER'S COUNSEL

Attached

EXHIBIT G

[RESERVED]

EXHIBIT H

FORM OF COMPLIANCE CERTIFICATE

To: The Administrative Agent and the Lenders who are parties to the Agreement described below

This Compliance Certificate is furnished pursuant to that certain Amended and Restated Unsecured Term Loan Agreement, dated as of July 7, 2021 (as amended, modified, renewed or extended from time to time, the "<u>Agreement</u>") among First Industrial, L.P. (the "<u>Borrower</u>"), First Industrial Realty Trust, Inc. (the "<u>General Partner</u>"), Wells Fargo Bank, National Association, individually and as Administrative Agent, and the Lenders named therein. Unless otherwise defined herein, capitalized terms used in this Compliance Certificate have the meanings ascribed thereto in the Agreement.

THE UNDERSIGNED HEREBY CERTIFIES THAT:

- 1. I am the duly elected [Chief Financial Officer] [Chief Accounting Officer] [Controller] of the [Borrower] [General Partner].
- 2. I have reviewed the terms of the Agreement and I have made, or have caused to be made under my supervision, a detailed review of the transactions and conditions of the General Partner, the Borrower and their respective Subsidiaries and Investment Affiliates during the accounting period covered by the financial statements attached (or most recently delivered to the Administrative Agent if none are attached).
- 3. The examinations described in paragraph 2 did not disclose, and I have no knowledge of, the existence of any condition or event which constitutes a Material Adverse Financial Change, Event of Default or Default during or at the end of the accounting period covered by the attached financial statements or as of the date of this Compliance Certificate, except as set forth below.
- 4. Schedule I (if attached) attached hereto sets forth financial data and computations and other information evidencing the General Partner's and the Borrower's compliance with certain covenants of the Agreement, all of which data, computations and information (or if no Schedule I is attached, the data, computations and information contained in the most recent Schedule I attached to a prior Compliance Certificate) are true, complete and correct in all material respects.
- 5. The financial statements and reports referred to in Section 8.2(i), 8.2(iii) or 8.2(viii), as the case may be, of the Agreement which are delivered concurrently with the delivery of this Compliance Certificate, if any, fairly present in all material respects the consolidated financial condition and operations of the General Partner, the Borrower and their respective Subsidiaries at such date and the consolidated results of their operations for the period then-ended, in accordance with GAAP applied consistently throughout such period and with prior periods and correctly state the amounts of Consolidated Total Indebtedness, Consolidated Secured Debt, Consolidated Senior Unsecured Debt and the Values of all Unencumbered Assets as determined pursuant to the Agreement.

Described below are the exceptions, if any, to paragraph 3 by listing, in detail, the nature of the which it has existed and the action which the Borrower has taken, is taking, or proposes to take event:	condition or event, the period during with respect to each such condition or

ntations and information set forth in Schedule I hereto and the financial n support hereof, are made and delivered this day of
FIRST INDUSTRIAL, L.P. By: FIRST INDUSTRIAL REALTY TRUST, INC. General Partner
By: Print Name: Title:

EXHIBIT I

FORM OF ASSIGNMENT AND ASSUMPTION AGREEMENT

THIS ASSIGNMENT AND ASSUMPTION AGREEMENT dated as of, 20 (the " <u>Agreement</u> ") by and among (the "Assignor"), (the " <u>Assignee</u> "), FIRST INDUSTRIAL, L.P., a
Delaware limited partnership (the "Borrower"), and WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent (the "Administrative Agent").
WHEREAS, the Assignor is a Lender under that certain Amended and Restated Unsecured Term Loan Agreement dated as of July 7, 2021 (as amended, restated, supplemented or otherwise modified from time to time, the "Loan Agreement"), by and among the Borrower, the financial institutions party thereto and their assignees under Section 13.3 thereof, the Administrative Agent, and the other parties thereto;
WHEREAS, the Assignor desires to assign to the Assignee all or a portion of the Assignor's Term Loan Exposure under the Loan Agreement, all on the terms and conditions set forth herein; and
WHEREAS, the [Borrower and the] Administrative Agent consent[s] to such assignment on the terms and conditions set forth herein.
NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which hereby are acknowledged by the parties hereto, the parties hereto hereby agree as follows:
Section 1. <u>Assignment</u> .
(a) Subject to the terms and conditions of this Agreement and in consideration of the payment to be made by the Assignee to the Assignor pursuant to Section 2 of this Agreement, effective as of, 20 (the "Assignment Date") the Assignor hereby irrevocably sells, transfers and assigns to the Assignee, without recourse, a \$ interest (such interest being the "Assigned Exposure") in and to the Assignor's Term Loan Exposure, and all of the other rights and obligations of the Assignor under the Loan Agreement, such Assignor's Note, and the other Loan Documents representing% in respect of the aggregate amount of Total Term Loan Exposure, including without limitation, [a principal amount of outstanding Term Loans equal to \$, all voting rights of the Assignor associated with the Assigned Exposure all rights to receive interest on such amount of Loans and all fees with respect to the [Assigned Exposure] and other rights of the Assignor under the Loan Agreement and the other Loan Documents with respect to the Assigned Exposure, all as if the Assignee were an original Lender under and signatory to the Loan Agreement having a Term Loan Exposure equal to the amount of the Assigned Exposure. The Assignee, subject to the terms and conditions hereof,

hereby assumes all obligations of the Assignor with respect to the Assigned Exposure as if the Assignee were an original Lender under

and signatory to the Loan Agreement having a Term Loan Exposure equal to the Assigned Exposure, which obligations shall

include, but shall not be limited to the obligation to indemnify the Administrative Agent as provided in the Loan Agreement (the foregoing obligations, together with all other similar obligations more particularly set forth in the Loan Agreement and the other Loan Documents, shall be referred to hereinafter, collectively, as the "Assigned Obligations").

(b) The assignment by the Assigner to the Assignee hereunder is without recourse to the Assigner. The Assignee makes and confirms to the Administrative Agent, the Assignor, and the other Lenders all of the representations, warranties and covenants of a Lender under Article XII of the Loan Agreement. Not in limitation of the foregoing, the Assignee acknowledges and agrees that, except as set forth in Section 4, below, the Assignor is making no representations or warranties with respect to, and the Assignee hereby releases and discharges the Assignor for any responsibility or liability for: (i) the present or future solvency or financial condition of the Borrower, any other Loan Party or any other Subsidiary, (ii) any representations, warranties, statements or information made or furnished by the Borrower, any other Loan Party or any other Subsidiary in connection with the Loan Agreement or otherwise, (iii) the validity, efficacy, sufficiency, or enforceability of the Loan Agreement, any Loan Document or any other document or instrument executed in connection therewith, or the collectability of the Assigned Obligations, (iv) the perfection, priority or validity of any Lien with respect to any collateral at any time securing the Obligations or the Assigned Obligations under the Notes or the Loan Agreement and (v) the performance or failure to perform by the Borrower or any other Loan Party of any obligation under the Loan Agreement or any other Loan Document. Further, the Assignee acknowledges that it has, independently and without reliance upon the Administrative Agent, any other Lender or counsel to the Administrative Agent or any of their respective officers, directors, employees and agents and based on the financial statements supplied by the Borrower and such other documents and information as it has deemed appropriate, made its own credit analysis and decision to become a Lender under the Loan Agreement. The Assignee also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Loan Agreement or any Note or pursuant to any other obligation. The Administrative Agent shall have no duty or responsibility whatsoever, either initially or on a continuing basis, to provide the Assignee with any credit or other information with respect to the Borrower, any other Loan Party or any other Subsidiary or to notify the undersigned of any Default or Event of Default except as expressly provided in the Loan Agreement. The Assignee has not relied on the Administrative Agent as to any legal or factual matter in connection therewith or in connection with the transactions contemplated thereunder.

Section 2. <u>Payment by Assignee</u>. In consideration of the assignment made pursuant to Section 1. of this Agreement, the Assignee agrees to pay to the Assignor on the Assignment Date, an amount equal to \$______ representing the aggregate principal amount outstanding of the Loans owing to the Assignor under the Loan Agreement and the other Loan Documents being assigned hereby.

Section 3. <u>Payments by Assignor</u>. The Assignor agrees to pay to the Administrative Agent on the Assignment Date the administrative fee payable under Section 13.3 (c) of the Loan Agreement.

Section 4. Representations and Warranties of Assignor. The Assignor hereby represents and warrants to the Assignee that (a) as of the Assignment Date (i) the Assignor is a Lender under the Loan Agreement having a Term Loan Exposure under the Loan Agreement immediately prior to the Assignment Date, equal to \$ and that the Assignor is not in default of its obligations under the Loan Agreement; and (ii) the outstanding balance of Loans owing to the Assignor (without reduction by any assignments thereof which have not yet become effective) is \$; and (b) it is the legal and beneficial owner of the Assigned Exposure which is free and clear of any adverse claim created by the Assignor.	
Section 5. Representations, Warranties and Agreements of Assignee. The Assignee (a) represents and warrants that it is (i) legally authorized to enter into this Agreement; (ii) an "accredited investor" (as such term is used in Regulation D of the Securities Act) and (iii) an Eligible Assignee; (b) confirms that it has received a copy of the Loan Agreement, together with copies of the most recent financial statements delivered pursuant thereto and such other documents and information (including without limitation the Loan Documents) as it has deemed appropriate to make its own credit analysis and decision to enter into this Agreement; (c) appoints and authorizes the Administrative Agent to take such action as contractual representative on its behalf and to exercise such powers under the Loan Documents as are delegated to the Administrative Agent by the terms thereof together with such powers as are reasonably incidental thereto; (d) agrees that it will become a party to and shall be bound by the Loan Agreement and the other Loan Documents to which the other Lenders are a party on the Assignment Date and will perform in accordance therewith all of the obligations which are required to be performed by it as a Lender; and (e) is either (i) not organized under the laws of a jurisdiction outside the United States of America or (ii) has delivered to the Administrative Agent (with an additional copy for the Borrower) such items required under Section 4.5 of the Loan Agreement.	
Section 6. <u>Recording and Acknowledgment by the Administrative Agent</u> . Following the execution of this Agreement, the Assignor will deliver to the Administrative Agent (a) a duly executed copy of this Agreement for acknowledgment and recording by the Administrative Agent and (b) the Assignor's Note. Upon such acknowledgment and recording, from and after the Assignment Date, the Administrative Agent shall make all payments in respect of the interest assigned hereby (including payments of principal, interest, fees and other amounts) to the Assignee. The Assignor and Assignee shall make all appropriate adjustments in payments under the Loan Agreement for periods prior to the Assignment Date directly between themselves.	
Section 7. <u>Addresses</u> . The Assignee specifies as its address for notices and its Lending Office for all Loans, the offices set forth below:	
Attention:	
Telephone No.:	
Telecopy No.:	

Section 8. <u>Payment Instructions</u> . All payments to be made to the Assignee under this Agreement by the Assignor, and all payments to be made to the Assignee under the Loan Agreement, shall be made as provided in the Loan Agreement in accordance with the following instructions:

Section 9. Effectiveness of Assignment. This Agreement, and the assignment and assumption contemplated herein, shall not be effective until (a) this Agreement is executed and delivered by each of the Assignor, the Assignee, the Administrative Agent and if required, the Borrower, and (b) the payment to the Assignor of the amounts owing by the Assignee pursuant to Section 2. hereof and (c) the payment to the Administrative Agent of the amounts owing by the Assignor pursuant to Section 3. hereof. Upon recording and acknowledgment of this Agreement by the Administrative Agent, from and after the Assignment Date, (i) the Assignee shall be a party to the Loan Agreement and, to the extent provided in this Agreement, have the rights and obligations of a Lender thereunder and (ii) the Assignor shall, to the extent provided in this Agreement, relinquish its rights (except as otherwise provided in Section 13.3 of the Loan Agreement) and be released from its obligations under the Loan Agreement; provided, however, that if the Assignor does not assign its entire interest under the Loan Documents, it shall remain a Lender entitled to all of the benefits and subject to all of the obligations thereunder with respect to its Term Loan Commitment.

Section 10. <u>Governing Law</u>. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF ILLINOIS APPLICABLE TO CONTRACTS EXECUTED, AND TO BE FULLY PERFORMED, IN SUCH STATE.

Section 11. <u>Counterparts</u>. This Agreement may be executed in any number of counterparts each of which, when taken together, shall constitute one and the same agreement.

Section 12. <u>Headings</u>. Section headings have been inserted herein for convenience only and shall not be construed to be a part hereof.

Section 13. <u>Amendments; Waivers</u>. This Agreement may not be amended, changed, waived or modified except by a writing executed by the Assignee and the Assignor.

Section 14. <u>Entire Agreement</u>. This Agreement embodies the entire agreement between the Assignor and the Assignee with respect to the subject matter hereof and supersedes all other prior arrangements and understandings relating to the subject matter hereof.

Section 15. <u>Binding Effect</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns.

Section 16. <u>Definitions</u>. Terms not otherwise defined herein are used herein with the respective meanings given them in the Loan Agreement.

[Include this Section only if the Borrower's consent is required under Section 13.3 of the Loan Agreement] Section 17. Agreements of the Borrower. The Borrower hereby agrees that the Assignee shall be a Lender under the Loan Agreement [having a Term Loan Exposure equal to the Assigned Exposure]. The Borrower agrees that the Assignee shall have all of the rights and remedies of a Lender under the Loan Agreement and the other Loan Documents as if the Assignee were an original Lender under and signatory to the Loan Agreement, including, but not limited to, the right of a Lender to receive payments of principal and interest with respect to the Assigned Obligations, if any, and to the Loans made by the Lenders after the date hereof and to receive the fees payable to the Lenders as provided in the Loan Agreement. Further, the Assignee shall be entitled to the benefit of the indemnification provisions from the Borrower in favor of the Lenders as provided in the Loan Agreement and the other Loan Documents. The Borrower further agrees, upon the execution and delivery of this Agreement, to execute in favor of the Assignee a Note in an initial amount equal to the Assigned Commitment. Further, the Borrower agrees that, upon the execution and delivery of this Agreement, the Borrower shall owe the Assigned Obligations to the Assignee as if the Assignee were the Lender originally making such Loans and entering into such other obligations.

[Signatures on Following Page]

IN WITNESS WHEREOF, the parties hereto have duly executed this Assignment and Assumption Agreement as of the date and year first written above.

ASSIGNOR:
[NAME OF ASSIGNOR]
By: Name:
Title:
Payment Instructions
(D. 11
[Bank] [Address]
ABA No. :
Account No.:
Account Name:
Reference:
ASSIGNEE:
[NAME OF ASSIGNEE]
By:
Name:
Title:
Payment Instructions
[Bank]
[Address]
ABA No.:
Account No.:
Account Name:
Reference:

[Signatures continued on Following Page]

	Agreed and Consented to as of the date first written above.
	[Include signature of the Borrower only if required under Section 13.3(c) of the Loan Agreement]
	BORROWER:
	FIRST INDUSTRIAL, L.P.
	By: First Industrial Realty Trust, Inc., its general partner
	By: Name: Title:
Accepted as of the date first written a	bove.
ADMINISTRATIVE AGENT:	
WELLS FARGO BANK, NATIONA	L ASSOCIATION, as Administrative Agent
By: Name:	
Γitle:	

EXHIBIT J

FORM OF NOTICE OF BORROWING

, 20

Wells Fargo Bank, National Association Minneapolis Loan Center 733 Marquette Avenue, 10th Floor Minneapolis, MN 55402 Attention: Teresa Mager

Ladies and Gentlemen:

Reference is made to that certain Amended and Restated Unsecured Term Loan Agreement dated as of July 7, 2021 (as amended, restated, supplemented or otherwise modified from time to time, the "Loan Agreement"), by and among FIRST INDUSTRIAL, L.P., a Delaware limited partnership (the "Borrower"), the financial institutions party thereto and their assignees under Section 13.3 thereof (the "Lenders"), Wells Fargo Bank, National Association, as Administrative Agent (the "Administrative Agent"), and the other parties thereto. Capitalized terms used herein, and not otherwise defined herein, have their respective meanings given them in the Loan Agreement.

- 1. Pursuant to Section 2.7 of the Loan Agreement, the Borrower hereby requests that the Lenders make Term Loans to the Borrower in an aggregate amount equal to \$______.
- 2. The Borrower requests that such Loans be made available to the Borrower on ________, 20___.
- 3. The Borrower hereby requests that such Loans be of the following Type:

[Check one box only]

Adjusted Base Rate Borrowing Eurocurrency Borrowing, with an initial Interest Period for a duration of:

[Check one box only]

one month
two months
three months
six months
other:

The Borrower hereby certifies to the Administrative Agent and the Lenders that as of the date hereof, as of the date of the making of the requested Loans, and after making such Loans, (a) no Default or Event of Default exists or would exist; and (b) the representations and warranties made or deemed made by the Borrower and each other Loan Party pursuant to Section 5.2(b) of the Loan Agreement, are and shall be true and correct with the same force and effect as if made on and as of such date except to the extent that such representations and warranties expressly relate solely to an earlier date (in which case such representations and warranties shall have been true and accurate on and as of such earlier date). In addition, the Borrower certifies to the Administrative Agent and the Lenders that all conditions to the making of the requested Loans contained in Section 5.2 of the Loan Agreement will have been satisfied at the time such Loans are made.
FIRST INDUSTRIAL, L.P.
By: FIRST INDUSTRIAL REALTY TRUST, INC.

Print Name: ______
Title: _____

General Partner

4. The location and number of Borrower's account to which proceeds of the Requested Loan are to be disbursed:

EXHIBIT K

FORM OF PRICING CERTIFICATE

_____, 20___

Wells Fargo Bank, National Association, as Administrative Agent Minneapolis Loan Center 733 Marquette Avenue, 10th Floor Minneapolis, MN 55402 Attention: Teresa Mager

Ladies and Gentlemen:

Reference is made to that certain Amended and Restated Unsecured Term Loan Agreement dated as of July 7, 2021 (as amended, restated, supplemented or otherwise modified from time to time, the "Loan Agreement"), by and among FIRST INDUSTRIAL, L.P., a Delaware limited partnership (the "Borrower"), the financial institutions party thereto and their assignees under Section 13.3 thereof (the "Lenders"), Wells Fargo Bank, National Association, as Administrative Agent (the "Administrative Agent"), and the other parties thereto. Capitalized terms used herein, and not otherwise defined herein, have their respective meanings given them in the Loan Agreement.

The undersigned hereby certifies solely in [his/her] capacity as [insert title of officer] of the Borrower and not in an individual capacity (and without personal liability) that:

- 1. I am the duly elected [insert title of officer] of the Borrower, and I am authorized to deliver this Certificate on behalf of the Borrower;
- 2. The calculation of the KPI Metric and the Sustainability Margin Adjustment, in each case, for the Reference Year ending ______, 20__ are set forth and attached as Annex A hereto, and are true and correct as of the date hereof and evidences the Borrower's qualification for a Sustainability Margin Adjustment with respect to Eurocurrency Borrowings and Adjusted Base Rate Borrowings equal to [____]% per annum. The LIBOR Applicable Margin and the Base Rate Applicable Margin will never be reduced by more than 0.01% pursuant to the Sustainability Margin Adjustment during any calendar year.
- 3. The public reporting of the Borrower's quarterly "Supplemental Information" report for the final quarterly period of such Reference Year can be found at the following website: [______].

FIRST INDUSTRIAL, L.P.
By: FIRST INDUSTRIAL REALTY TRUST, INC.
General Partner
D.
By:
Print Name:
Title:

SCHEDULE 6.9

Litigation (Borrower)

SCHEDULE 6.19

Environmental Compliance

SCHEDULE 6.24

Trade Names

SCHEDULE 6.25

Subsidiaries (Borrower)

SCHEDULE 6.26

Unencumbered Assets

SCHEDULE 7.8

Litigation (General Partner)

SCHEDULE 7.18

Subsidiaries (General Partner)

THIRD AMENDMENT TO UNSECURED TERM LOAN AGREEMENT

This THIRD AMENDMENT TO UNSECURED TERM LOAN AGREEMENT (this "Third Amendment") is made and entered into as of July 7, 2021 by and among FIRST INDUSTRIAL, L.P., a limited partnership formed under the laws of the State of Delaware (together with its successors and assigns, the "Borrower"), FIRST INDUSTRIAL REALTY TRUST, INC., a corporation formed under the laws of the State of Maryland (the "General Partner"), each of the financial institutions initially a signatory to the Credit Agreement (as defined below) together with their successors and assigns under Section 13.1 of the Credit Agreement (the "Lenders") and WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent (the "Administrative Agent").

WITNESSETH:

WHEREAS, the Borrower, the General Partner, the Administrative Agent and the Lenders are parties to that certain Unsecured Term Loan Agreement dated as of September 11, 2015 (as amended by that certain First Amendment dated as of January 26, 2017 and that certain Second Amendment dated as of October 31, 2017, the "Credit Agreement");

WHEREAS, the Borrower has requested that the Administrative Agent and the Lenders amend certain terms and conditions of the Credit Agreement as described herein; and

WHEREAS, the Administrative Agent and the Lenders party to this Third Amendment have agreed to so amend certain terms and conditions of the Credit Agreement, all on the terms and conditions set forth below in this Third Amendment.

NOW, THEREFORE, for good and valuable consideration, the receipt and adequacy of which is hereby acknowledged, the parties hereto hereby agree as follows:

- 1. <u>Definitions</u>. All capitalized undefined terms used in this Third Amendment shall have the meanings ascribed thereto in the Credit Agreement, as amended hereby.
- 2. <u>Amendments to Credit Agreement</u>. As of the date hereof (such date, the "<u>Amendment Effective Date</u>") the Credit Agreement is hereby amended as set forth in the marked terms on Exhibit B-1 attached hereto including the amended Schedules and Exhibits thereto (the "<u>Amended Credit Agreement</u>"). In Exhibit B-1 hereto, deletions of text in the Amended Credit Agreement are indicated by struck-through text, and insertions of text are indicated by bold, double-underlined text. Exhibit B-2 attached hereto sets forth a clean copy of the Amended Credit Agreement, after giving effect to such amendments.
- 3. <u>Conditions to Effectiveness.</u> This Third Amendment shall not be effective until the Administrative Agent shall have received counterparts of this Third Amendment duly executed and delivered by the Borrower, the General Partner, the Administrative Agent, and the Required Lenders.
- 4. <u>Representations and Warranties</u>. Except for changes in factual circumstances specifically and expressly permitted under the Loan Documents, the representations and warranties of the Borrower and each other Loan Party contained in <u>Article VI</u> of the Amended Credit Agreement or any other Loan Document to which any of them is a party, are true and correct on and as of the date hereof except to the extent any such representation or warranty is stated to relate solely to an

earlier date, in which case such representation or warranty is true and correct on and as of such earlier date.

- 5. <u>Limited Amendment; Ratification of Loan Documents</u>. Except as specifically amended or modified hereby, the terms and conditions of the Credit Agreement and the other Loan Documents shall remain in full force and effect, and are hereby ratified and affirmed in all respects. This Third Amendment shall not be deemed a waiver of, or consent to, or a modification or amendment of, any other term or condition of the Credit Agreement or any other Loan Document, except as expressly set forth herein.
- 6. <u>Governing Law</u>. This Third Amendment shall be governed by and construed in accordance with the internal laws (and not the law of conflicts) of the State of Illinois, but giving effect to federal laws applicable to national banks.
- 7. <u>Miscellaneous</u>. This Third Amendment may be executed in any number of counterparts, which shall together constitute an entire original agreement, and shall be binding upon and inure to the benefit of the parties hereto and their respective successors and assigns. This Third Amendment expresses the entire understanding of the parties with respect to the transactions contemplated hereby. No prior negotiations or discussions shall limit, modify, or otherwise affect the provisions hereof. Any determination that any provision of this Third Amendment or any application hereof is invalid, illegal, or unenforceable in any respect and in any instance shall not affect the validity, legality, or enforceability of such provision in any other instance, or the validity, legality, or enforceability of any other provisions of this Third Amendment. Each of the Borrower and the General Partner represents and warrants that it has consulted with independent legal counsel of its selection in connection herewith and is not relying on any representations or warranties of the Administrative Agent or the Lenders or their counsel in entering into this Third Amendment. This Third Amendment shall constitute a Loan Document.

IN WITNESS WHEREOF, the parties hereto have caused this Third Amendment to be duly executed as of the date first above written.

FIRST INDUSTRIAL, L.P., as the Borrower

By: FIRST INDUSTRIAL REALTY TRUST, INC., its General Partner

By: <u>/s/ Scott A. Musil</u>
Name: Scott A. Musil
Title: Chief Financial Officer

FIRST INDUSTRIAL REALTY TRUST, INC., as General Partner

By: <u>/s/ Scott A. Musil</u> Name: Scott A. Musil

Title: Chief Financial Officer

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent and a Lender

By: <u>/s/ Craig V. Koshkarian</u> Name: Craig V. Koshkarian Title: Director

PNC BANK, NATIONAL ASSOCIATION, as a Lender

By: <u>/s/ Margaret K. Grady</u>
Name: Margaret Kilner Grady
Title: Vice President

REGIONS BANK, as a Lender

By: <u>/s/ Susan Wolfe</u>
Name: Susan Wolfe
Title: Assistant Vice President

U.S. BANK NATIONAL ASSOCIATION, as a Lender

By: <u>/s/ Donald J. Pafford</u>
Name: Donald J. Pafford
Title: Senior Vice President

ASSOCIATED BANK, NATIONAL ASSOCIATION as a Lender

By: <u>/s/ Mitchell Vega</u>
Name: Mitchell Vega
Title: Senior Vice President

FIFTH THIRD BANK, NATIONAL ASSOCIATION, as a Lender

By: <u>/s/ Michael Glandt</u> Name: Michael Glandt Title: Vice President

The undersigned, as Guarantor under that certain Guaranty dated as of September 11, 2015, hereby consents to the foregoing Third Amendment to Credit Agreement and acknowledges and agrees that the Guaranty dated as of September 11, 2015 and executed by the undersigned remains in full force and effect.

FIRST INDUSTRIAL REALTY TRUST, INC., as Guarantor

By: /s/ Scott A. Musil Name: Scott A. Musil Title: Chief Financial Officer

EXHIBIT B-1

MARKED COPY OF AMENDED CREDIT AGREEMENT

as amended by First Amendment to Unsecured Term Loan Agreement, dated as of 1/26/2017; as amended by Second Amendment to Unsecured Term Loan Agreement, dated as of 10/31/17; as amended by Third Amendment to Unsecured Term Loan Agreement, dated as of July 7, 2021

UNSECURED TERM LOAN AGREEMENT

DATED AS OF SEPTEMBER 11, 2015

AMONG

FIRST INDUSTRIAL, L.P., AS BORROWER
FIRST INDUSTRIAL REALTY TRUST, INC.,
AS GENERAL PARTNER AND GUARANTOR
THE LENDERS

AND

WELLS FARGO BANK, NATIONAL ASSOCIATION
AS ADMINISTRATIVE AGENT

AND

PNC BANK, NATIONAL ASSOCIATION,
REGIONS BANK and
U.S. BANK NATIONAL ASSOCIATION
AS CO-SYNDICATION AGENTS

WELLS FARGO SECURITIES, LLC,
PNC CAPITAL MARKETS LLC,
REGIONS CAPITAL MARKETS and
U.S. BANK NATIONAL ASSOCIATION
AS JOINT-LEAD ARRANGERS AND JOINT BOOK RUNNERS

Page

ARTICLE I. DEFINITIONS AND ACCOUNTING TERMS 1 1.1. Definitions 1 1.2. Financial Standards 2324 1.3. Classification of Loans and Borrowings 2324 1.4. General References 2324 1.5. Rates. 24 <u>1.6. Divisions</u> <u>24</u> ARTICLE II. THE FACILITY 2425 2.1. The Facility 2425 2.2. Incremental Loans 2425 2.3. Principal Payments 2526 2.4. Requests for Borrowings; Responsibility for Borrowings 2526 2.5. Evidence of Credit Extensions 2526 2.6. Loans and Borrowings 2627 2.7. Requests for Borrowings 2627 2.8. Interest Elections 2728 2.9. Applicable Margins 2930 2.10. Funds Transfer Disbursements 3132 2.11. Minimum Amount of Each Borrowing 3132 2.12. Interest 3132 2.13. Method of Payment 3132 2.14. Default 3233 2.15. Lending Offices 3233 2.16. Non Receipt of Funds by Administrative Agent 3233 2.17. [Reserved] 3334 2.18. [Reserved]. 3334 2.19. [Reserved] 3334 2.20. Pro Rata Treatment/Sharing of Payments 3334 2.21. [Reserved] 34<u>35</u> 2.22. Application of Moneys Received 3435 2.23. [Reserved] 3435 2.24. Prepayments of Loans 35 ARTICLE III. [RESERVED] 3536 ARTICLE IV. CHANGE IN CIRCUMSTANCES 3536 4.1. Yield Protection 3536 4.2. Changes in Capital Adequacy Regulations 3637 4.3. Availability of Eurocurrency or Adjusted Base Rate Borrowings 3738 4.4. Funding Indemnification 3839

i

(Continued)

 4.5. Taxes 3839 4.6. Lender Statements; Survival of Indemnity 4142 4.7. Replacement of Lenders under Certain Circumstances 4142
ARTICLE V. CONDITIONS PRECEDENT 4243
5.1. Conditions Precedent to Closing 42435.2. Conditions Precedent to Subsequent Borrowings 4445
ARTICLE VI. REPRESENTATIONS AND WARRANTIES 4445
6.1. Existence 4445 6.2. Corporate/Partnership Powers 4546 6.3. Power of Officers 4546 6.4. Government and Other Approvals 4546 6.5. Solvency 4546 6.6. Compliance With Laws 4647 6.7. Enforceability of Agreement 4647 6.8. Title to Property 4647 6.9. Litigation 4647 6.10. Events of Default 4647 6.11. Investment Company Act of 1940 4647 6.12. Public Utility Holding Company Act 47 6.13. Regulation U 47 6.14. No Material Adverse Financial Change 4748 6.15. Financial Information 4748 6.16. Factual Information 4748 6.17. ERISA 4748 6.18. Taxes 4748 6.19. Environmental Matters 4748 6.20. Insurance 49 6.21. No Brokers 4950 6.22. No Violation of Usury Laws 4950 6.23. Not a Foreign Person 4950 6.24. No Trade Name 4950 6.25. Subsidiaries 4950 6.26. Unencumbered Assets 4950 6.27. Anti-Corruption Laws and Sanctions 5152 6.28. EEA Affected Financial Institutions 52
ARTICLE VII. ADDITIONAL REPRESENTATIONS AND WARRANTIES 5253
 7.1. Existence 5253 7.2. Corporate Powers 5253 7.3. Power of Officers 5253 7.4. Government and Other Approvals 5253 7.5. Compliance With Laws 5253 7.6. Enforceability of Agreement 5253

(Continued)

7.7. Liens; Consents 53 7.8. Litigation 5354 7.9. Events of Default 5354 7.10. Investment Company Act of 1940 5354 7.11. Public Utility Holding Company Act 5354 7.12. No Material Adverse Financial Change 5354 7.13. Financial Information 5354 7.14. Factual Information 5354 7.15. ERISA 54 7.16. Taxes 5455 7.17. No Brokers 5455 7.18. Subsidiaries 5455 7.19. Status 5455 7.20. Anti-Corruption Laws and Sanctions 5455		
ARTICLE VIII. AFFIRMATIVE COVENANTS 5556		
8.1. Notices 5556 8.2. Financial Statements, Reports, Etc. 5657 8.3. Existence and Conduct of Operations 59 8.4. Maintenance of Properties 5960 8.5. Insurance 5960 8.6. Payment of Obligations 5960 8.7. Compliance with Laws 60 8.8. Adequate Books 60 8.9. ERISA 60 8.10. Maintenance of Status 6061 8.11. Use of Proceeds 6061 8.12. Distributions 6061 8.13. Guarantors 6061		
ARTICLE IX. NEGATIVE COVENANTS 61		
 9.1. Change in Business 61 9.2. Change of Management of Properties 61 9.3. Change of Borrower Ownership 6162 9.4. Use of Proceeds 6162 9.5. Liens 62 9.6. Regulation U 6263 9.7. Indebtedness and Cash Flow Covenants 63 9.8. Change of Control; Mergers and Dispositions 6465 9.9. Negative Pledge 6465 		
ARTICLE X. DEFAULTS 6465		
 10.1. Nonpayment of Principal 6465 10.2. Certain Covenants 65 10.3. Nonpayment of Interest and Other Obligations 65 		

(Continued)

10.4. Cross Default 65 10.5. Loan Documents 6566 10.6. Representation or Warranty 6566 10.7. Covenants, Agreements and Other Conditions 6566 10.8. No Longer General Partner 6566 10.9. Material Adverse Financial Change 6566 10.10. Bankruptcy 66 10.11. Legal Proceedings 6667
10.12. ERISA 67 10.13. Change in Control 67 10.14. Failure to Satisfy Judgments 67 10.15. Environmental Remediation 6768
ARTICLE XI. ACCELERATION, WAIVERS, AMENDMENTS AND REMEDIES 6768
11.1. Acceleration 676811.2. Preservation of Rights; Amendments 68
ARTICLE XII. THE ADMINISTRATIVE AGENT 6869
12.1. Appointment 6869 12.2. Powers 6869 12.3. General Immunity 6869 12.4. No Responsibility for Loans, Recitals, etc. 69 12.5. Action on Instructions of Lenders 6970 12.6. Employment of Administrative Agents and Counsel 6970 12.7. Reliance on Documents; Counsel 6970 12.8. Administrative Agent's Reimbursement and Indemnification 6970 12.9. Rights as a Lender 7071 12.10. [Reserved]. 7071 12.11. Lender Credit Decision 7071 12.12. Successor Administrative Agent 71 12.13. Notice of Defaults 7172 12.14. Requests for Approval 7172 12.15. Copies of Documents 72 12.16. Defaulting Lenders 7273 12.17. Delegation to Affiliates 7374 12.18. Managing Agents, Syndication Agent, etc. 7374 12.19. Erroneous Payments 74
ARTICLE XIII. BENEFIT OF AGREEMENT; ASSIGNMENTS; PARTICIPATIONS 73.1. Successors and Assigns 73.6. 13.1. Successors and Assigns 73.6. 13.2. Participations 74.7. 13.3. Assignments; Consents 75.78. 13.4. Register 77.80. 13.5. [Reserved]. 77.80. 13.6. Confidentiality 77.80.

(Continued)

13.7 Tax Treatment 78 <u>81</u>	
ARTICLE XIV. GENERAL PROVISIONS 7982	
14.1. Survival of Representations 7982	
14.2. Governmental Regulation 7982	
14.3. Taxes 79 82	
14.4. Headings 79 <u>82</u>	
14.5. No Third Party Beneficiaries 79 82	
14.6. Expenses; Indemnification 7982	
14.7. Severability of Provisions 8083	
14.8. Nonliability of the Lenders 8083	
14.9. Choice of Law <u>80</u> 83	
14.10. Consent to Jurisdiction 8083	
14.11. Waiver of Jury Trial 8184	
14.12. Successors and Assigns 8184	
14.13. Entire Agreement; Modification of Agreement 8184	
14.14. Dealings with the Borrower 8285	
14.15. Set-Off 8285	
14.16. Counterparts-83; <u>Integration</u> , <u>Effectiveness</u> , <u>Electronic Execution</u> 86	
14.17. Patriot Act CIP Notice 8387	
14.18. Acknowledgement and Consent to Bail-In of EEAAffected Financial Institutions.	83 87
14.19. Acknowledgement Regarding Any Supported QFCs 88	
14.20. Certain ERISA Matters 89	
ARTICLE XV. NOTICES 8490	
15.1. Giving Notice; Electronic Delivery 8490	
15.2. Change of Address 8692	

(Continued)

EXHIBITS

- Term Loan Commitment Amounts; Term Loan Exposure Amounts
- Form of Note
- C Form of Disbursement Instruction Agreement
- Form of Guaranty D
- Opinion of Borrower's Counsel Ε
- Opinion of General Partner's Counsel F
- [Reserved] G
- H Form of Compliance Certificate
- I Form of Assignment Agreement
 J Form of Notice of Borrowing

SCHEDULES

- 6.9 Litigation (Borrower)
- 6.19 Environmental Compliance
- 6.24 Trade Names
- 6.25 Subsidiaries (Borrower)
- 6.26 Unencumbered Assets
- 7.8 Litigation (General Partner)
- 7.18 Subsidiaries (General Partner)

UNSECURED TERM LOAN AGREEMENT

THIS UNSECURED TERM LOAN AGREEMENT is entered into as of September 11, 2015 by and among the following:

FIRST INDUSTRIAL, L.P., a Delaware limited partnership having its principal place of business at 311 South 1 N. Wacker Drive, Suite 39004200, Chicago, Illinois 60606, the sole general partner of which is First Industrial Realty Trust, Inc., a Maryland corporation;

FIRST INDUSTRIAL REALTY TRUST, INC., a Maryland corporation that is qualified as a real estate investment trust whose principal place of business is 311 South 1 N. Wacker Drive, Suite 39004200, Chicago, Illinois 60606 ("General Partner");

WELLS FARGO BANK, NATIONAL ASSOCIATION ("<u>Wells</u>"), a national bank organized under the laws of the United States of America having an office at 10 South Wacker Drive, 32nd Floor, Chicago, IL 60606, as Administrative Agent ("<u>Administrative Agent</u>") for the Lenders (as defined below); and

Those Lenders identified on the signature pages hereto.

RECITALS

- A. The Borrower is primarily engaged in the business of acquiring, developing, owning and operating bulk warehouse and light industrial properties.
- B. The Borrower, the Lenders and the Administrative Agent have agreed to enter into this Agreement in order to set forth the terms and conditions under which the Lenders will, on the Closing Date, make term loans to or for the benefit of the Borrower.
- C. The General Partner is fully liable for the obligations of the Borrower hereunder by virtue of its status as the sole general partner of the Borrower and as guarantor under the Guaranty.

NOW, THEREFORE, in consideration of the mutual covenants and agreements herein contained, the parties hereto agree as follows:

ARTICLE I.

DEFINITIONS AND ACCOUNTING TERMS

- 1.1. <u>Definitions</u>. As used in this Agreement, the following terms have the meanings set forth below:
- "Adjusted Base Rate" means a floating interest rate equal to the sum of (i) the LIBOR Market Index Rate, plus (ii) the Base Rate Applicable Margin in effect from time to time.
 - "Adjusted Base Rate Borrowing" means a Borrowing that bears interest at the Adjusted Base Rate.

- "Adjusted EBITDA" means for any Person the sum of EBITDA for such Person and such Person's reported corporate overhead for itself and its Subsidiaries; provided that "Adjusted EBITDA" shall have deducted overhead related to specific properties.
- "Adjusted LIBOR Rate" means, with respect to a Eurocurrency Borrowing for the relevant Interest Period, the sum of (i) the Base LIBOR Rate applicable to such Interest Period, plus, (ii) the LIBOR Applicable Margin in effect from time to time during such Interest Period.
- "<u>Administrative Agent</u>" means Wells, in its capacity as contractual representative of the Lenders pursuant to <u>Article XII</u>, and not in its individual capacity as a Lender, and any successor Administrative Agent appointed pursuant to <u>Article XII</u>.
- "<u>Administrative Office</u>" means the Administrative Agent's office designated on its signature page to this Agreement or such other office as may be designated by the Administrative Agent by written notice to the Borrower and the Lenders.
- "<u>Administrative Questionnaire</u>" means the Administrative Questionnaire completed by each Lender and delivered to the Administrative Agent in a form supplied by the Administrative Agent to the Lenders from time to time.
 - "Affected Financial Institution" means (a) any EEA Financial Institution or (b) any UK Financial Institution.
- "Affiliate" means any Person directly or indirectly controlling, controlled by or under direct or indirect common control with any other Person. A Person shall be deemed to control another Person if the controlling Person owns ten percent (10%) or more of any class of voting securities of the controlled Person or possesses, directly or indirectly, the power to direct or cause the direction of the management or policies of the controlled Person, whether through ownership of stock, by contract or otherwise; <u>provided</u> that in no event shall the Administrative Agent or any Lender be deemed to be an Affiliate of the Borrower.
- "Aggregate Term Loan Commitment" means, as of any date, the total of all Term Loan Commitments, which as of the Agreement Execution Date is \$260,000,000.
 - "Agreement" means this Unsecured Term Loan Agreement and all amendments, modifications and supplements hereto.
- "Agreement Execution Date" shall mean September 11, 2015, the date on which all of the parties hereto have executed this Agreement.
- "Anti-Corruption Laws" means all Applicable Laws of any jurisdiction applicable to the Borrower or its Subsidiaries from time to time concerning or relating to bribery, corruption or money-laundering, including without limitation, the Foreign Corrupt Practices Act of 1977.
 - "Applicable Cap Rate" means 6.255.75%.
- "Applicable Law" means all international, foreign, federal, state and local statutes, treaties, rules, guidelines, regulations, ordinances, codes, directions, executive orders, and administrative

or judicial precedents or authorities, including the interpretation or administration thereof by any Governmental Authority charged with the enforcement, interpretation or administration thereof, and all applicable administrative orders, directed duties, requests, licenses, authorizations and permits of, and agreements with, any Governmental Authority, in each case whether or not having the force of law.

- "Applicable Margin" means the applicable margins set forth in the table in <u>Section 2.9</u> used in calculating the interest rate applicable to the various types of Borrowings, which shall vary from time to time in accordance with the long term, senior unsecured debt ratings of the Borrower and the General Partner in the manner set forth in <u>Section 2.9</u>.
- "Approved Fund" means any Fund that is administered or managed by (a) a Lender, (b) an Affiliate of a Lender or (c) an entity or an Affiliate of an entity that administers or manages a Lender.
- "Arranger" means Wells Fargo Securities, LLC, PNC Capital Markets LLC, Regions Capital Markets and U.S. Bank National Association and their successors in their capacity as Joint Lead Arrangers.
- "Asset Sale" means any sale or other disposition by the Consolidated Operating Partnership of any Property or other assets (excluding any proceeds resulting from the casualty or condemnation of such Property or other assets, to the extent such proceeds are used to rebuild such Properties or assets within 365 days of receipt of such proceeds) that yields gross proceeds equal to the aggregate of (i) all cash proceeds, plus (ii) the initial principal amount of any noncash proceeds consisting of notes or other debt securities, plus (iii) the fair market value of other non-cash proceeds, if and to the extent that the aggregate of (i), (ii) and (iii) exceeds \$500,000.
- "Assets Acquired Not In Service" means, as of any date of determination, any Project which has been acquired and owned for less than 12 months but has not yet been leased to 90%.
- "Assets Under Development" means, as of any date of determination, any Project which is under construction and then treated as an asset under development under GAAP.
- "<u>Assignment and Assumption Agreement</u>" means an Assignment and Assumption Agreement among a Lender, an Eligible Assignee and the Administrative Agent, substantially in the form of <u>Exhibit I.</u>
- "<u>Bail-In Action</u>" means the exercise of any Write-Down and Conversion Powers by the applicable <u>EEA</u>-Resolution Authority in respect of any liability of an <u>EEAAffected</u> Financial Institution.
- "Bail-In Legislation" means, (a) with respect to any EEA Member Country implementing Article 55 of Directive 2014/59/EU of the European Parliament and of the Council of the European Union, the implementing law, regulation, rule or requirement for such EEA Member Country from time to time which is described in the EU Bail-In Legislation Schedule, and (b) with respect to the United Kingdom, Part I of the United Kingdom Banking Act 2009 (as amended from time to time) and any other law, regulation or rule applicable in the United Kingdom relating to the resolution

of unsound or failing banks, investment firms or other financial institutions or their affiliates (other than through liquidation, administration or other insolvency proceedings).

"Base LIBOR Rate" means, subject to the implementation of a Replacement Rate in accordance with Section 4.3(ii), with respect to any Eurocurrency Borrowing for any Interest Period, the rate of interest obtained by dividing (i) the rate of interest per annum determined on the basis of the rate as set by the ICE Benchmark Administration ("ICE") (or the successor thereto if ICE is no longer making such rate available) for deposits in Dollars for a period equal to the applicable Interest Period which appears on Reuters Screen LIBOR01 Page (or any applicable successor page), at approximately 11:00 a.m. (London time) two (2) Business Days prior to the first day of the applicable Interest Period by (ii) a percentage equal to 1 minus the Eurodollar Reserve Percentage. If, for any reason, the rate referred to in the preceding clause (i) does not appear on Reuters Screen LIBOR01 Page (or any applicable successor page), then the rate to be used for such clause (i) shall be determined by the Administrative Agent to be the arithmetic average of the rate per annum at which deposits in Dollars would be offered by first class banks in the London interbank market to the Administrative Agent at approximately 11:00 a.m. (London time) two (2) Business Days prior to the first day of the applicable Interest Period for a period equal to such Interest Period. Any change in the Eurodollar Reserve Percentage shall result in a change in the Base LIBOR Rate on the date on which such change in such Eurodollar Reserve Percentage becomes effective. If the Base LIBOR Rate determined as provided above would be less than zero, the Base LIBOR Rate shall be deemed to be zero. Each calculation by the Administrative Agent of the Base LIBOR Rate shall be conclusive and binding for all purposes, absent manifest error. To the extent a comparable or successor rate is approved by the Administrative Agent in connection herewith, the approved rate shall be applied to the then applicable Interest Period in a manner consistent with market practice as reasonably determined by the Administrative Agent; provided that if such market practice is reasonably determined by the Administrative Agent to not be administratively feasible, such approved rate shall be applied in a manner reasonably determined by the Administrative Agent. All approvals and determinations to be made by the Administrative Agent in the preceding sentence shall be done in consultation with the Borrower. Notwithstanding the foregoing, unless otherwise specified in any amendment to this Agreement entered into in accordance with Section 4.3(ii), in the event that a Replacement Rate with respect to the Base LIBOR Rate is implemented then all references herein to the Base LIBOR Rate shall be deemed references to such Replacement Rate.

"Base Rate Applicable Margin" means the Applicable Margin in effect for an Adjusted Base Rate Borrowing as determined in accordance with Section 2.9 hereof.

"Benefit Plan" means any of (a) an "employee benefit plan" (as defined in ERISA) that is subject to Title I of ERISA, (b) a "plan" as defined in and subject to Section 4975 of the Code or (c) any Person whose assets include (for purposes of ERISA Section 3(42) or otherwise for purposes of Title I of ERISA or Section 4975 of the Code) the assets of any such "employee benefit plan" or "plan".

"Borrower" means First Industrial, L.P., along with its permitted successors and assigns.

"Borrowing" means Loans of the same Type, made, converted or continued on the same date and, in the case of Eurocurrency Loans, as to which a single Interest Period is in effect.

"Borrowing Date" means a Business Day on which a Borrowing is made to the Borrower.

"Borrowing Notice" is defined in Section 2.7 hereof.

"Business Day" means a day, other than a Saturday, Sunday or holiday, on which banks are open for business in New York City; provided that with respect to any borrowings, disbursements and payments in respect of and calculations, interest rates and Interest Periods pertaining to Eurocurrency Loans, such day is also a day on which banks are open for general business in London, England.

"<u>Capital Stock</u>" means any and all shares, interests, participations or other equivalents (however designated) of capital stock of a corporation, any and all equivalent ownership interests in a Person which is not a corporation and any and all warrants or options to purchase any of the foregoing.

"Cash Equivalents" shall mean (i) short term obligations of, or fully guaranteed by, the United States of America, (ii) commercial paper rated A 1 or better by Standard and Poor's Corporation or P1 or better by Moody's Investors Service, Inc., or (iii) certificates of deposit issued by and time deposits with commercial banks (whether domestic or foreign) having capital and surplus in excess of \$100,000,000; provided in each case that the same provides for payment of both principal and interest (and not principal alone or interest alone) and is not subject to any contingency regarding the payment of principal or interest, provided that all such Cash Equivalents would qualify as cash equivalents in accordance with GAAP.

"Change in Control" means (a) the acquisition of ownership, directly or indirectly, beneficially or of record, by any Person or group (within the meaning of the Securities Exchange Act of 1934 and the rules of the Securities and Exchange Commission thereunder as in effect on the date hereof), of more than 40% of the aggregate voting power represented by the then outstanding voting stock of the General Partner, (b) the occupation of a majority of the seats (other than vacant seats) on the board of directors of the General Partner by Persons who were neither (i) nominated by the board of directors of the General Partner, (ii) appointed by directors so nominated, nor (iii) nominated by holders of the preferred stock in the General Partner pursuant to the terms of such stock; or (c) the General Partner shall cease to own, directly or indirectly, fifty-one percent (51%) or more of the Equity Interests of the Borrower.

"Change in Law" means the occurrence, after the date of this Agreement, of any of the following: (a) the adoption or taking effect of any law, rule, regulation or treaty, (b) any change in any law, rule, regulation or treaty or in the administration, interpretation or application thereof by any Governmental Authority, or (c) the making or issuance of any request, rules, guideline, requirement or directive (whether or not having the force of law) by any Governmental Authority; provided however, that notwithstanding anything herein to the contrary, (i) the Dodd-Frank Wall Street Reform and Consumer Protection Act and all requests, rules, guidelines, requirements and directives thereunder, issued in connection therewith or in implementation thereof, and (ii) all requests, rules, guidelines, requirements and directives promulgated by the Bank for International Settlements, the Basel Committee on Banking Supervision (or any successor or similar authority) or the United States or foreign regulatory authorities, in each case pursuant to Basel III, shall in each case be deemed to be a "Change in Law" regardless of the date enacted, adopted, issued or implemented.

"Closing Date" means the date on which the conditions precedent in <u>Section 5.1</u> are satisfied in accordance therewith and this Agreement becomes effective.

"Code" means the Internal Revenue Code of 1986 as amended from time to time, or any replacement or successor statute, and the regulations promulgated thereunder from time to time.

"Consolidated Leverage Ratio" means, as of any date of determination, the ratio of Consolidated Total Indebtedness to Implied Capitalization Value of the Consolidated Operating Partnership.

"Consolidated Leverage Ratio Increase Period" is defined in Section 9.7(b).

"Consolidated Operating Partnership" means the Borrower, the General Partner and any other subsidiary partnerships or entities of either of them which are required under GAAP to be consolidated with the Borrower and the General Partner for financial reporting purposes.

"Consolidated Secured Debt" means as of any date of determination, the sum of (a) the aggregate outstanding principal amount of all Indebtedness of the Consolidated Operating Partnership outstanding at such date which is secured by a Lien on any asset or Capital Stock of Consolidated Operating Partnership, including without limitation loans secured by mortgages, stock, or partnership interests, but excluding Defeased Debt and (b) the amount by which the aggregate principal amount of all Indebtedness of the Subsidiaries of the Borrower or the General Partner outstanding at such date exceeds \$5,000,000 (for the avoidance of doubt, (x) excluding Indebtedness of the Borrower and (y) including Guarantee Obligations (other than customary non-recourse carveout obligations) of the Subsidiaries of the General Partner (other than the Borrower) in respect of primary obligations of the Borrower or the General Partner), without duplication of any Indebtedness included under clause (a). For clarification, Consolidated Secured Debt shall include the Borrower's or the General Partner's Ownership Share of any Investment Affiliate's Indebtedness.

"Consolidated Senior Unsecured Debt" means as of any date of determination, the aggregate outstanding principal amount of all Indebtedness of the Consolidated Operating Partnership (which will include, without limitation, any Indebtedness that is secured by partnership interests and that is recourse to the Borrower or the Guarantor, where such recourse component applies only to the payment of principal and/or interest), outstanding at such date other than (a) Indebtedness which is contractually subordinated to the Indebtedness of the Consolidated Operating Partnership under the Loan Documents on terms acceptable to the Administrative Agent and (b) that portion of Consolidated Secured Debt described in clause (a) of that definition. For clarification, Consolidated Senior Unsecured Debt shall exclude the Borrower's or the General Partner's Ownership Share of any Investment Affiliate's Indebtedness.

"Consolidated Total Indebtedness" means as of any date of determination, all Indebtedness of the Consolidated Operating Partnership outstanding at such date, determined on a consolidated basis in accordance with GAAP, after eliminating intercompany items; provided that for purposes of defining "Consolidated Total Indebtedness" the term "Indebtedness" shall not include the short

term debt (e.g. accounts payable, short term expenses) of the Borrower or the General Partner or Defeased Debt. For clarification, Consolidated Total Indebtedness shall include the Borrower's or the General Partner's Ownership Share of any Investment Affiliate's Indebtedness.

"Controlled Group" means all members of a controlled group of corporations and all trades or businesses (whether or not incorporated) under common control which, together with all or any of the entities in the Consolidated Operating Partnership, are treated as a single employer under Sections 414(b) or 414(c) of the Code.

"Conversion/Continuation Notice" is defined in Section 2.8.

"<u>Debtor Relief Laws</u>" means the United States Bankruptcy Code, and all other liquidation, conservatorship, bankruptcy, assignment for the benefit of creditors, moratorium, rearrangement, receivership, insolvency, reorganization, or similar Applicable Laws relating to the relief of debtors in the United States of America or other applicable jurisdictions from time to time in effect.

"Debt Service" means for any period, (a) Interest Expense for such period <u>plus</u> (b) the aggregate amount of regularly scheduled principal payments of Indebtedness (excluding optional prepayments and balloon principal payments due on maturity in respect of any Indebtedness and mandatory excess cash flow sweeps) required to be made during such period by the Borrower, or any of its consolidated Subsidiaries <u>plus</u> (c) a percentage of all such regularly scheduled principal payments required to be made during such period by any Investment Affiliate on Indebtedness (excluding optional prepayments and balloon principal payments due on maturity in respect of any Indebtedness) taken into account in calculating Interest Expense, such percentage equal to the greater of (x) the percentage of the principal amount of such Indebtedness for which the Borrower or any consolidated Subsidiary is liable and (y) the Ownership Share in such Investment Affiliate held by the Borrower and any consolidated Subsidiaries, in the aggregate, without duplication.

"Default" means an event which, with notice or lapse of time or both, would become an Event of Default.

"<u>Default Rate</u>" means with respect to any Borrowing, a rate equal to the interest rate applicable to such Borrowing plus three percent (3%) per annum.

"<u>Defaulting Lender</u>" means, subject to <u>Section 12.16(f)</u>, any Lender that (a) has failed to (i) fund all or any portion of its Loans within two (2) Business Days of the date such Loans were required to be funded hereunder unless such Lender notifies the Administrative Agent and the Borrower in writing that such failure is the result of such Lender's determination that one or more conditions precedent to funding (each of which conditions precedent, together with any applicable default, shall be specifically identified in such writing) has not been satisfied, or (ii) pay to the Administrative Agent or any other Lender any other amount required to be paid by it hereunder within two (2) Business Days of the date when due, (b) has notified the Borrower or the Administrative Agent in writing that it does not intend to comply with its funding obligations hereunder, or has made a public statement to that effect (unless such writing or public statement relates to such Lender's obligation to fund a Loan hereunder and states that such position is based on such Lender's determination that a condition precedent to funding (which condition precedent, together with any applicable default, shall be specifically identified in such writing or public

statement) cannot be satisfied), (c) has failed, within three (3) Business Days after written request by the Administrative Agent or the Borrower, to confirm in writing to the Administrative Agent and the Borrower that it will comply with its prospective funding obligations hereunder (provided that such Lender shall cease to be a Defaulting Lender pursuant to this clause (c) upon receipt of such written confirmation by the Administrative Agent and the Borrower), (d) has, or has a direct or indirect parent company that has, (i) become the subject of (A) a proceeding under any Debtor Relief Law or (B) a Bail-In Action, or (ii) had appointed for it a receiver, custodian, conservator, trustee, administrator, assignee for the benefit of creditors or similar Person charged with reorganization or liquidation of its business or assets, including the Federal Deposit Insurance Corporation or any other state or federal regulatory authority acting in such a capacity; provided that a Lender shall not be a Defaulting Lender solely by virtue of the ownership or acquisition of any equity interest in that Lender or any direct or indirect parent company thereof by a Governmental Authority so long as such ownership interest does not result in or provide such Lender with immunity from the jurisdiction of courts within the United States of America or from the enforcement of judgments or writs of attachment on its assets or permit such Lender (or such Governmental Authority) to reject, repudiate, disavow or disaffirm any contracts or agreements made with such Lender, (e) has purchased or otherwise acquired all or any portion of another Lender's Term Loan Commitment, excluding any purchase or acquisition in accordance with the terms and conditions set forth in Section 13.1; or (f) has sold or otherwise transferred all or any portion of its Term Loan Commitment, excluding any sale or transfer in accordance with the terms and conditions set forth in Section 13.1. Any determination by the Administrative Agent that a Lender is a Defaulting Lender under clauses (a) through (f) above shall be conclusive and binding absent manifest error, and such Lender shall be deemed to be a Defaulting Lender (subject to Section 12.16(f)) upon delivery of written notice of such determination to the Borrower and each Lender.

"<u>Defeased Debt</u>" means that portion of debt which has already been defeased by depositing collateral in the form of obligations supported by the credit of the United States government in such amounts as are required and permitted under the terms of the applicable loan documents.

"<u>Disbursement Instruction Agreement</u>" means an agreement substantially in the form of <u>Exhibit C</u> to be executed and delivered by the Borrower pursuant to <u>Section 5.1</u>, as the same may be amended, restated or modified from time to time with the prior written approval of the Administrative Agent.

"Dollars" and "\$" mean United States Dollars.

"EBITDA" means, with respect to any Person, income before restructuring charges—in an aggregate amount not to exceed \$10,000,000 during any four fiscal quarter period, non-cash impairment charges and other non-cash, non-recurring items determined in good faith by the Borrower and extraordinary items, without deduction of any losses related to initial offering costs of preferred stock which are written off due to the redemption of such preferred stock, and excluding any gains or losses from pay-off or retirement of debt and gains/losses on sales of Properties and excluding costs incurred in acquiring Properties, where such costs are required to be expensed under ASC 805 Business Combinations 805-10-25-23, as reported by such Person and its Subsidiaries on a consolidated basis in accordance with GAAP (reduced to eliminate any income from Investment Affiliates of such Person, any interest income and, with respect to the

Consolidated Operating Partnership, any income from the assets used for Defeased Debt), plus Interest Expense, depreciation, amortization and income tax (if any) expense plus a percentage of such income (adjusted as described above) of any such Investment Affiliate equal to the allocable economic interest in such Investment Affiliate held by such Person and any Subsidiaries, in the aggregate (provided that no item of income or expense shall be included more than once in such calculation even if it falls within more than one of the foregoing categories).

"EEA Financial Institution" means (a) any credit institution or investment firm established in any EEA Member Country which is subject to the supervision of an EEA Resolution Authority, (b) any entity established in an EEA Member Country which is a parent of an institution described in clause (a) of this definition, or (c) any financial institution established in an EEA Member Country which is a subsidiary of an institution described in clauses (a) or (b) of this definition and is subject to consolidated supervision with its parent.

"EEA Member Country" means any of the member states of the European Union, Iceland, Liechtenstein, and Norway.

"<u>EEA Resolution Authority</u>" means any public administrative authority or any <u>personPerson</u> entrusted with public administrative authority of any EEA Member Country (including any delegee) having responsibility for the resolution of any <u>EEA Financial Institution</u> or investment firm established in any EEA Member Country.

"Effective Date" means each Borrowing Date (other than a continuation or conversion).

"<u>Eligible Assignee</u>" means (a) a Lender, (b) an Affiliate of a Lender, (c) an Approved Fund and (d) any other Person (other than a natural person) approved by the Administrative Agent (such approval not to be unreasonably withheld or delayed by any of the foregoing); provided that notwithstanding the foregoing, "Eligible Assignee" shall not include the Borrower or any of the Borrower's Affiliates or Subsidiaries.

"Environmental Laws" means any and all Federal, state, local or municipal laws, rules, orders, regulations, statutes, ordinances, codes, decrees, requirements of any Governmental Authority having jurisdiction over the Borrower, its Subsidiaries or Investment Affiliates, or their respective assets, and regulating or imposing liability or standards of conduct concerning protection of human health or the environment, as now or may at any time hereafter be in effect, in each case to the extent the foregoing are applicable to the operations of the Borrower, any Investment Affiliate, or any Subsidiary or any of their respective assets or Properties.

"Equity Interests" means, with respect to any Person, any share of capital stock of (or other ownership or profit interests in) such Person, any warrant, option or other right for the purchase or other acquisition from such Person of any share of capital stock of (or other ownership or profit interests in) such Person, any security convertible into or exchangeable for any share of capital stock of (or other ownership or profit interests in) such Person or warrant, right or option for the purchase or other acquisition from such Person of such shares (or such other interests), and any other ownership or profit interest in such Person (including, without limitation, partnership, member or trust interests therein), whether voting or nonvoting, and whether or not such share,

warrant, option, right or other interest is authorized or otherwise existing on any date of determination.

"Equity Value" is defined in Section 10.10 hereof.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and regulations promulgated thereunder from time to time.

"Erroneous Payment" is defined in Section 12.19(a) hereof.

"<u>EU Bail-In Legislation Schedule</u>" means the EU Bail-In Legislation Schedule published by the Loan Market Association (or any successor person), as in effect from time to time.

"<u>Eurocurrency</u>", when used in reference to any Loan or Borrowing, refers to whether such Loan, or the Loans comprising such Borrowing, are bearing interest at a rate determined by reference to the Adjusted LIBOR Rate.

"<u>Eurodollar Reserve Percentage</u>" means, for any day, the percentage (stated as a decimal) which is in effect for such day as prescribed by the Board of Governors of the Federal Reserve System (or any successor) for determining the maximum reserve requirement (including without limitations, any basic, supplemental or emergency reserves) in respect of eurocurrency liabilities or any similar category.

"Event of Default" means any event set forth in Article X hereof.

"Excluded Taxes" means, in the case of each Lender or applicable Lending Office and the Administrative Agent, (a) taxes imposed on its overall net income, and franchise taxes imposed on it, by (i) the jurisdiction under the laws of which such Lender or the Administrative Agent is incorporated or organized or (ii) the jurisdiction in which the Administrative Agent's or such Lender's principal executive office of such Lender's applicable Lending Office is located, and (b) any United States federal withholding taxes imposed by FATCA.

"Existing Revolving Credit Agreement" means that certain ThirdFourth Amended and Restated Unsecured Revolving Credit Agreement dated as of October 31, 2017July 7, 2021 among the Borrower, the General Partner, the lenders party thereto and Wells, as administrative agent, and all amendments, modifications, supplements, refinancing and replacements thereto.

"Existing Term Loan Agreement" means that certain Amended and Restated Unsecured Term Loan Agreement dated as of January 29, 2014 July 7, 2021 among the Borrower, the General Partner, the lenders party thereto and Wells, as administrative agent, and all amendments, modifications, supplements, refinancing and replacements thereto.

"Facility" means the Term Loan Commitments and the extensions of credit made thereunder.

"<u>FATCA</u>" means Sections 1471 through 1474 of the Code, as of the date of this Agreement (or any amended or successor version that is substantively comparable and not materially more onerous to comply with) and any current or future regulations or official interpretations thereof and any agreement entered into pursuant to Section 1471(b)(1) of the Code.

"Federal Funds Effective Rate" means, for any period, a fluctuating interest rate per annum equal for each day during such period to the weighted average of the rates on overnight Federal funds transactions with members of the Federal Reserve System, as published for such day (or, if such day is not a Business Day, for the immediately preceding Business Day) by the Federal Reserve Bank of New York, or, if such rate is not so published for any day which is a Business Day, the average of the quotations for such day on such transactions received by the Administrative Agent from three Federal funds brokers of recognized standing selected by the Administrative Agent in its sole discretion; provided, that, if the Federal Funds Effective Rate shall be less than zero, such rate shall be deemed to be zero for purposes of this Agreement.

"Fixed Charge Coverage Ratio" is defined in Section 9.7(a).

"<u>Fund</u>" means any Person (other than a natural person) that is (or will be) engaged in making, purchasing, holding or otherwise investing in commercial loans and similar extensions of credit in the ordinary course of its business.

"GAAP" means those generally accepted accounting principles in the United States of America as in effect from time to time that are consistent with those utilized in preparing the audited financial statements of the Borrower required hereunder; provided that all financial computations shall be made in accordance with GAAP as in effect on the Agreement Execution Date.

"<u>General Partner</u>" means First Industrial Realty Trust, Inc., a Maryland corporation that is listed on a national securities exchange and is qualified as a real estate investment trust. The General Partner is the sole general partner of the Borrower.

"Governmental Authority" means the government of the United States of America or any other nation, or of any political subdivision thereof, whether state or local, and any agency, authority, instrumentality, regulatory body, court, central bank or other entity exercising executive, legislative, judicial, taxing, regulatory or administrative powers or functions of or pertaining to government (including any supranational bodies such as the European Union or the European Central Bank).

"Gross Negligence" means recklessness, or actions taken or omitted with conscious indifference to or the complete disregard of consequences or rights of others affected. Gross Negligence does not mean the absence of ordinary care or diligence, or an inadvertent act or inadvertent failure to act. If the term "gross negligence" is used with respect to the Administrative Agent or any Lender or any indemnitee in any of the other Loan Documents, it shall have the meaning set forth herein.

"Ground Lease Payments" means, for any period, payment made in cash during such period in respect of any ground lease with respect to which the Borrower or any of its Subsidiaries is a lessee.

"Guarantee Obligation" means as to any Person (the "guaranteeing person"), any obligation (determined without duplication) of (a) the guaranteeing person or (b) another Person (including, without limitation, any bank under any letter of credit) to induce the creation of which the guaranteeing person has issued a reimbursement, counter indemnity or similar obligation, in either case guaranteeing or in effect guaranteeing (but only to the extent that the guaranty applies to the payment of principal or interest due under recourse Indebtedness) any Indebtedness, leases, dividends or other obligations (the "primary obligations") of any other third Person (the "primary obligor") in any manner, whether directly or indirectly, including, without limitation, any obligation of the guaranteeing person, whether or not contingent, (i) to purchase any such primary obligation or any property constituting direct or indirect security therefor, (ii) to advance or supply funds (1) for the purchase or payment of any such primary obligation or (2) to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor, (iii) to purchase property, securities or services primarily for the purpose of assuring the owner of any such primary obligation of the ability of the primary obligor to make payment of such primary obligation or (iv) otherwise to assure or hold harmless the owner of any such primary obligation against loss in respect thereof; provided, however, that the term Guarantee Obligation shall not include endorsements of instruments for deposit or collection in the ordinary course of business. The amount of any Guarantee Obligation of any guaranteeing person shall be deemed to be the maximum stated amount of the primary obligation relating to such Guarantee Obligation (or, if less, the maximum stated liability set forth in the instrument embodying such Guarantee Obligation), provided, that in the absence of any such stated amount or stated liability, the amount of such Guarantee Obligation shall be such guaranteeing person's maximum reasonably anticipated liability in respect thereof as determined by the Borrower in good faith. Notwithstanding the foregoing, a guaranty of customary non-recourse carveouts shall not be deemed a "Guarantee Obligation" for purposes of this Agreement.

" $\underline{\text{Guaranty}}$ " means the Guaranty executed by the General Partner in the form attached hereto as $\underline{\text{Exhibit D}}$ and each guaranty executed and delivered pursuant to Section 8.13.

"Implied Capitalization Value" means for any Person as of any date, the sum (without duplication) of (i) the quotient of (x) the Adjusted EBITDA for such Person during the most recent four fiscal quarters (which Adjusted EBITDA shall exclude any Adjusted EBITDA attributable to all assets of the type described in clause (ii), Assets Under Development, Assets Acquired Not in Service or Rollover Projects, and which Adjusted EBITDA attributable to each Project which was formerly a Rollover Project shall not be less than zero), and (y) the Applicable Cap Rate, plus (ii) the purchase price paid by such Person (less any amounts paid to such Person as a purchase price adjustment, held in escrow, retained as a contingency reserve or in connection with other similar arrangements) for any Property (other than Assets Under Development) acquired by such Person during the immediately preceding period of four consecutive fiscal quarters, plus (iii) an amount equal to the then current book value of each Asset Under Development, plus (iv) the then current book value of Unimproved Land, plus (v) with respect to each Rollover Project, an amount equal to 50% of the then current book value, determined in accordance with GAAP, of such Rollover Project, plus (vi) an amount equal to 100% of unrestricted cash and unrestricted cash equivalents, including any cash on deposit with a qualified intermediary with respect to a deferred tax-free exchange (and specifically excluding any cash or cash equivalents being used to support Defeased Debt), plus (vii) an amount equal to 100% of the then current book value, determined in accordance with GAAP, of all first mortgage receivables on income producing commercial properties.

For purposes of determining Implied Capitalization Value: (v) to the extent the amount of Implied Capitalization Value attributable to Rollover Projects would exceed 10% of Implied Capitalization Value, such excess shall be excluded; (w) to the extent the amount of Implied Capitalization Value attributable to Unimproved Land would exceed 10% of Implied Capitalization Value, such excess shall be excluded; (x) to the extent the amount of Implied Capitalization Value attributable to Asset Under Development and Assets Acquired Not in Service would exceed 20% of Implied Capitalization Value, such excess shall be excluded; (y) to the extent the amount of Implied Capitalization Value attributable to assets of the type described in clause (vii) above would exceed 5% of Implied Capitalization Value, such excess shall be excluded; and (z) to , to the extent the amount of Implied Capitalization Value attributable to unconsolidated joint ventures and partnerships and other assets of the type described in clauses (iii), (iv), (v) and (vii) would, in the aggregate, exceed 30% of Implied Capitalization Value, such excess shall be excluded. The Borrower's Ownership Share of assets held by Investment Affiliates (excluding assets of the type described in the immediately preceding clause (vi)) will be included in Implied Capitalization Value calculations consistent with the above described treatment for wholly owned assets. In the case of a newly formed Investment Affiliate, the Borrower's Ownership Share of assets held by the Investment Affiliate shall be calculated by multiplying (a) total assets plus accumulated depreciation of the Investment Affiliate by (b) the Ownership Share of such Investment Affiliate. This valuation methodology will be used for the first four quarters following the formation of any Investment Affiliate. For purposes of computing the Implied Capitalization Value, Adjusted EBITDA may be increased from quarter to quarter by the amount of (A) net cash flow from new leases of space at the Properties (where such net cash flow has not then been included in EBITDA) which have a minimum term of one year and (B) Properties which were previously Assets Under Development but which have been completed during such four quarter period and have at least some tenants in possession of the respective leased spaces and conducting business operations therein each will be included in the calculation of Implied Capitalization Value using pro forma EBITDA for such four quarter period, so long as a "new opening summary" form is submitted to, and approved by, Administrative Agent for each newly-opened Property during such four quarter period. "New opening summary" forms will only be required for developments to the extent the aggregate thereof exceeds \$600,000,000 during any fiscal year.

"Incremental Loans" is defined in Section 2.2 hereof.

"Indebtedness" of any Person at any date means without duplication, (a) all indebtedness of such Person for borrowed money, (b) all obligations of such Person for the deferred purchase price of property or services (other than current trade liabilities and other accounts payable, and accrued expenses incurred in the ordinary course of business and payable in accordance with customary practices), repurchase obligations, takeout commitments or forward equity commitments, in each case evidenced by a binding agreement (excluding any such obligation to the extent the obligation can be satisfied by the issuance of Equity Interests (other than Mandatorily Redeemable Stock)), to the extent such obligations constitute indebtedness for the purposes of GAAP, (c) any other indebtedness of such Person which is evidenced by a note, bond, debenture or similar instrument, (d) all obligations of such Person under financing leases and capital leases, (e) all obligations of such Person in respect of acceptances issued or created for the account of such Person, (f) all Guarantee Obligations of such Person (excluding in any calculation of consolidated Indebtedness of the Consolidated Operating Partnership, Guarantee Obligations of any member of the Consolidated Operating Partnership in respect of primary obligations of any

other member of the Consolidated Operating Partnership), (g) all reimbursement obligations of such Person for letters of credit and other contingent liabilities, (h) Net Mark-to-Market Exposure under Rate Management Transactions, (i) all liabilities secured by any lien (other than liens for taxes not yet due and payable) on any property owned by such Person even though such Person has not assumed or otherwise become liable for the payment thereof, (j) any repurchase obligation or liability of such Person or any of its Subsidiaries with respect to accounts or notes receivable sold by such Person or any of its Subsidiaries, (k) such Person's Ownership Share of debt in Investment Affiliates and any loans where such Person is liable as a general partner and (l) all obligations of such Person to purchase, redeem, retire, defease or otherwise make any payment in respect of any Mandatorily Redeemable Stock issued by such Person or any other Person, valued at the greater of its voluntary or involuntary liquidation preference plus accrued and unpaid dividends.

"Insolvent" means insolvent as defined in Section 101(32) of the United States Bankruptcy Code, as amended.

"Interest Expense" means all interest expense of the Consolidated Operating Partnership determined in accordance with GAAP <u>plus</u> (i) capitalized interest not covered by an interest reserve from a loan facility, <u>plus</u> (ii) the allocable portion (based on liability) of any interest incurred on any obligation for which the Consolidated Operating Partnership is wholly or partially liable under guaranties covering the payment of principal and/or interest, <u>plus</u> (iii) the allocable percentage of any interest incurred on any Indebtedness of any Investment Affiliate, whether recourse or nonrecourse, equal to the applicable Ownership Share in such Investment Affiliate held by the Consolidated Operating Partnership, in the aggregate, <u>provided</u> that no expense shall be included more than once in such calculation even if it falls within more than one of the foregoing categories; <u>provided</u>, <u>however</u>, that "Interest Expense" shall not include interest on loans after they become Defeased Debt.

"Interest Period" means with respect to a Eurocurrency Borrowing, a period commencing on (and including) the date such Eurocurrency Borrowing is made (or in the case of the continuation of a Eurocurrency Loan, the last day of the preceding Interest Period for such Loan) and ending on (but excluding) the first Business Day in the first, third or sixth calendar month thereafter, as the Borrower may select in the applicable Borrowing Notice or Conversion/Continuation Notice, as the case may be, except that each Interest Period shall begin on the first Business Day of a calendar month, and any Borrowing made on a date other than the first Business Day of a calendar month shall be an Adjusted Base Rate Borrowing unless and until the Borrower shall deliver a Conversion/Continuation Notice providing for the conversion of such Loan to a Eurocurrency Borrowing beginning on the first Business Day of a calendar month. Notwithstanding the foregoing or anything in this Agreement to the contrary, if any Interest Period would otherwise end after the Maturity Date, such Interest Period shall end on the Maturity Date.

"Investment Affiliate" means any Person in which the Consolidated Operating Partnership, directly or indirectly, has an ownership interest, whose financial results are not consolidated under GAAP with the financial results of the Consolidated Operating Partnership on the consolidated financial statements of the Consolidated Operating Partnership.

"Investment Grade Rating" means, for any class of non-credit enhanced long-term senior unsecured debt issued by the Borrower, (a) a rating of BBB- or higher from S&P or (b) a rating of Baa3 or better from Moody's.

"<u>Lenders</u>" means, collectively, Wells, and the other Persons executing this Agreement in such capacity, or any Person which subsequently executes and delivers any amendment hereto in such capacity and each of their respective permitted successors and assigns. Where reference is made to "the Lenders" in any Loan Document it shall be read to mean "all of the Lenders".

"<u>Lending Office</u>" means, for each Lender and for each Type of Loan, the office of such Lender specified in such Lender's Administrative Questionnaire or in the applicable Assignment and Assumption Agreement or such other office of such Lender as such Lender may notify the Administrative Agent in writing from time to time.

"<u>LIBOR Applicable Margin</u>" means, as of any date with respect to any Eurocurrency Borrowing, the Applicable Margin in effect for such Eurocurrency Borrowing as determined in accordance with <u>Section 2.9</u> hereof.

"LIBOR Borrowing" is defined in Section 4.3.

"LIBOR Market Index Rate" means, for any day, the Base LIBOR Rate as of that day that would be applicable for a Eurocurrency Borrowing having a one month Interest Period determined at approximately 10:00 a.m. (Central Time) for such day (rather than 11:00 a.m. (London time) two (2) Business Days prior to the first day of such Interest Period as otherwise provided in the definition of Base LIBOR Rate), or if such day is not a Business Day, the immediately preceding Business Day. The LIBOR Market Index Rate shall be determined on a daily basis.

"<u>Lien</u>" means any mortgage, pledge, security interest, encumbrance, lien or charge of any kind (including, without limitation, any conditional sale or other title retention agreement or lease in the nature thereof, any filing or agreement to file a financing statement as debtor under the Uniform Commercial Code on any property leased to any Person under a lease which is not in the nature of a conditional sale or title retention agreement, or any subordination agreement in favor of another Person).

"Loan" means, with respect to a Lender, such Lender's loan (including any Incremental Loans) made pursuant to Article II (or any conversion or continuation thereof).

"Loan Documents" means this Agreement, the Notes, the Guaranty and any and all other agreements or instruments required and/or provided to Lenders hereunder or thereunder, as any of the foregoing may be amended from time to time.

"<u>Loan Party</u>" means each of the Borrower, the General Partner, each other Person who guarantees all or a portion of the Obligations and/or who at any time pledges any collateral to secure all or a portion of the Obligations.

"Mandatorily Redeemable Stock" means, with respect to any Person, any Equity Interest of such Person which by the terms of such Equity Interest (or by the terms of any security into which it is convertible or for which it is exchangeable or exercisable), upon the happening of any

event or otherwise (a) matures or is mandatorily redeemable, pursuant to a sinking fund obligation or otherwise (other than an Equity Interest to the extent redeemable in exchange for common stock or other equivalent common Equity Interests), (b) is convertible into or exchangeable or exercisable for Indebtedness or Mandatorily Redeemable Stock, or (c) is redeemable at the option of the holder thereof, in whole or in part (other than an Equity Interest which is redeemable solely in exchange for common stock or other equivalent common Equity Interests); in each case, on or prior to the Maturity Date.

"<u>Market Value Net Worth</u>" means at any time, the Implied Capitalization Value of a Person at such time minus the Indebtedness of such Person at such time.

"<u>Material Acquisition</u>" means any acquisition (or series of related acquisitions) permitted by the Loan Documents and consummated in accordance with the terms of the Loan Documents if the aggregate consideration paid in respect of such acquisition (including any Indebtedness assumed in connection therewith) exceeds 10% of the Implied Capitalization Value of the Consolidated Operating Partnership.

"<u>Material Adverse Effect</u>" means, with respect to any matter, that such matter in the Required Lenders' good faith judgment may (x) materially and adversely affect the business, properties, condition or results of operations of the Consolidated Operating Partnership taken as a whole, or (y) constitute a non-frivolous challenge to the validity or enforceability of any material provision of any Loan Document against any obligor party thereto.

"Material Adverse Financial Change" shall be deemed to have occurred if the Required Lenders, in their good faith judgment, determine that a material adverse financial, <u>business or results of operations</u> change has occurred which could prevent timely repayment of any Borrowing hereunder or materially impair the Borrower's ability to perform its obligations under any of the Loan Documents.

"Material Credit Facility" means, as to the Borrower, the General Partner and their Subsidiaries, this Agreement and any other agreement(s) creating or evidencing indebtedness for borrowed money (excluding any Indebtedness which is "non-recourse") entered into on or after the Agreement Execution Date by the Borrower, the General Partner or any of their Subsidiaries, or in respect of which the Borrower, the General Partner or any of their Subsidiaries is an obligor or otherwise provides a guarantee or other credit support (other than a guarantee of customary recourse exceptions) ("Credit Facility"), in a principal amount outstanding or available for borrowing equal to or greater than \$5,000,000 (or the equivalent of such amount in the relevant currency of payment, determined as of the date of the closing of such facility based on the exchange rate of such other currency); and if no Credit Facility or Credit Facilities equal or exceed such amounts, then the largest Credit Facility shall be deemed to be a Material Credit Facility.

"<u>Materials of Environmental Concern</u>" means any gasoline or petroleum (including crude oil or any fraction thereof) or petroleum products or any hazardous or toxic substances, materials or wastes, defined or regulated as such in or under any Environmental Law, including, without limitation, asbestos, radon, polychlorinated biphenyls and urea formaldehyde insulation.

"Maturity Date" means September 11, 2022; or such earlier date on which the principal balance of the Facility and all other sums due in connection with the Facility shall be due as a result of the acceleration of the Facility.

"Monetary Default" means any Default involving the Borrower's failure to pay any of the Obligations when due.

"Moody's" means Moody's Investors Service, Inc. and its successors.

"Net Mark-to-Market Exposure" of a Person means, as of any date of determination, the excess (if any) of all unrealized losses over all unrealized profits of such Person arising from Rate Management Transactions. "Unrealized losses" means the fair market value of the cost to such Person of unwinding such Rate Management Transaction as of the date of determination (assuming the Rate Management Transaction were to be terminated as of that date), and "unrealized profits" means the fair market value of the gain to such Person of unwinding such Rate Management Transaction as of the date of determination (assuming such Rate Management Transaction were to be terminated as of that date).

"Non-Defaulting Lender" means, at any time, each Lender that is not a Defaulting Lender at such time.

"Note" means, to the extent issued pursuant to Section 2.5, with respect to the Facility, the promissory note payable to the order of each Lender in the amount of such Lender's maximum applicable Term Loan Commitment in substantially the form attached hereto as Exhibit B-1 (or such other form approved by the Administrative Agent) (collectively, the "Notes").

"Obligations" means the Borrowings and all accrued and unpaid fees and all other obligations of the Borrower to the Administrative Agent or any or all of the Lenders arising under this Agreement or any of the other Loan Documents.

"OFAC" means the U.S. Department of the Treasury's Office of Foreign Assets Control.

"Other Taxes" is defined in Section 4.5(ii).

"Ownership Share" means, with respect to any Investment Affiliate, the pro rata share of the nominal ownership interests held by the Consolidated Operating Partnership, in the aggregate, in such Investment Affiliate, without duplication (e.g., if the Consolidated Operating Partnership owns 25% of an Investment Affiliate, but receives 90% of the economic benefits from such Investment Affiliate, then the Ownership Share shall be equal to 25%).

"Parent" means, with respect to any Lender, any Person as to which such Lender is, directly or indirectly, a subsidiary.

"Participants" is defined in Section 13.2.1 hereof.

"Payment Recipient" has the meaning assigned thereto in Section 12.19(a).

"PBGC" means the Pension Benefit Guaranty Corporation or any entity succeeding to any or all of its functions under ERISA.

"<u>Percentage</u>" means, with respect to any Lender, the percentage of the Total Term Loan Exposure represented by such Lender's Term Loan Exposure at such time giving effect to any assignments and any Incremental Loans pursuant to <u>Section 2.2</u>.

"Permitted Liens" are defined in Section 9.5 hereof.

"<u>Permitted Negative Pledge</u>" means a "negative pledge" that (a) establishes a maximum ratio of unsecured debt to unencumbered assets, or of secured debt to total assets, or that otherwise conditions a Person's ability to encumber its assets upon the maintenance of one or more specified ratios that limit such Person's ability to encumber its assets but that does not generally prohibit the encumbrance of its assets, or the encumbrance of specific assets or (b) provides for the grant in favor of the holders of any unsecured debt of any equal and ratable Lien in connection with the pledge of any property or asset to secure the Facility.

"<u>Person</u>" means an individual, a corporation, a limited or general partnership, an association, a joint venture or any other entity or organization, including a governmental or political subdivision or an agent or instrumentality thereof.

"Plan" means an employee benefit plan as defined in Section 3(3) of ERISA, whether or not terminated, as to which the Borrower or any member of the Controlled Group may have any liability.

"Project" means any real estate asset which is 100% owned by the Borrower or by any Wholly Owned Subsidiary and which is operated as an industrial property.

"Property" means each parcel of real property owned or operated by the Borrower, any Subsidiary or Investment Affiliate.

"Property Operating Income" means, with respect to any Property, for any period, earnings from rental operations (computed in accordance with GAAP but without deduction for reserves) attributable to such Property <u>plus</u> depreciation, amortization and interest expense with respect to such Property for such period, and, if such period is less than a year, adjusted by straight lining various ordinary operating expenses which are payable less frequently than once during every such period (e.g. real estate taxes and insurance). The earnings from rental operations reported for the immediately preceding fiscal quarter shall be adjusted to include pro forma earnings (as substantiated to the satisfaction of the Administrative Agent) for an entire quarter for any Property acquired or placed in service during such fiscal quarter and to exclude earnings during such quarter from any property not owned as of the end of the quarter.

"Purpose Credit" has the meaning ascribed to it in Regulation U of the Board of Governors of the Federal Reserve System.

"QFC" is defined in Section 14.19.

"Qualified Officer" means, with respect to any entity, the chief financial officer, chief accounting officer, controller or assistant controller of such entity if it is a corporation or of such entity's general partner if it is a partnership.

"Rate Management Obligations" of a Person means any and all obligations of such Person, whether absolute or contingent and howsoever and whensoever created, arising, evidenced or acquired (including all renewals, extensions and modifications thereof and substitutions therefor), under (i) any and all Rate Management Transactions, and (ii) any and all cancellations, buy backs, reversals, terminations or assignments of any Rate Management Transactions.

"Rate Management Transaction" means any transaction (including an agreement with respect thereto) now existing or hereafter entered by the Borrower which is a rate swap, basis swap, forward rate transaction, commodity swap, commodity option, equity or equity index swap, equity or equity index option, interest rate option, foreign exchange transaction, cap transaction, floor transaction, collar transaction, forward transaction, currency swap transaction, cross-currency rate swap transaction, currency option or any other similar transaction (including any option with respect to any of these transactions) or any combination thereof, whether linked to one or more interest rates, foreign currencies, commodity prices, equity prices or other financial measures.

"Rating Agency" means S&P or Moody's.

"Recovery Event" means any settlement of or payment in respect of any property or casualty insurance claim or any condemnation proceeding relating to any Property.

"Register" is defined in Section 13.4.

"Regulation D" means Regulation D of the Board of Governors of the Federal Reserve System as from time to time in effect and any successor or other regulation or official interpretation of said Board of Governors relating to reserve requirements applicable to member banks of the Federal Reserve System.

"Replacement Rate" has the meaning assigned thereto in Section 4.3(ii).

"Reportable Event" means a reportable event as defined in Section 4043 of ERISA and the regulations issued under such section, with respect to a Plan, excluding, however, such events as to which the PBGC by regulation waived the requirement of Section 4043(a) of ERISA that it be notified within 30 days of the occurrence of such event, <u>provided</u> that a failure to meet the minimum funding standard of Section 412 of the Code and of Section 302 of ERISA shall be a Reportable Event regardless of the issuance of any such waivers in accordance with either Section 4043(a) of ERISA or Section 412(d) of the Code.

"Required Lenders" means Lenders in the aggregate having at least 51% of the Aggregate Term Loan Commitment then in effect or, if the Aggregate Term Loan Commitment has been terminated, the Total Term Loan Exposure then outstanding. In determining such percentage at any given time, all then existing Defaulting Lenders will be disregarded and excluded, and at all times when two or more Lenders (excluding Defaulting Lenders) are party to this Agreement, the term "Required Lenders" shall in no event mean less than two Lenders.

"Resolution Authority" means an EEA Resolution Authority or, with respect to any UK Financial Institution, a UK Resolution Authority.

"Rollover Projects" means those Projects which, due to no or low occupancy at such Project, have a value, determined by dividing the Property Operating Income for such a Project for the most recent four fiscal quarters by the Applicable Cap Rate, of less than 50% of book value, provided that a Project shall no longer be treated as a Rollover Project after: (i) a period of six consecutive full fiscal quarters has elapsed since such Project was first included as a Rollover Project, or (ii) such Project has a value, determined by dividing the Property Operating Income for such Project for the most recent four fiscal quarters by the Applicable Cap Rate, of greater than 50% of book value.

"S&P" means Standard & Poor's Ratings Group, a division of the McGraw Hill Companies and its successors.

"Sanctioned Person" means, at any time, (a) any Person that is, or is 50% or more owned by a Person or Persons that are, listed in any Sanctions related list of designated Persons maintained by any Governmental Authority of the United States of America, including without limitation, OFAC or the U.S. Department of State, or by the United Nations Security Council, Her Majesty's Treasury or the European Union or (b) a country, a region within a country, an agency of the government of a country, an organization controlled by a country or a Person resident in a country, that is subject to or the target of a sanctions program applicable to such country, region, agency, organization or Person and identified on any Sanctions-related list maintained by any Governmental Authority of the United States of America, including without limitation, OFAC or the U.S. Department of State.

"Sanctions" means economic or financial sanctions or trade embargoes imposed, administered or enforced by (a) any Governmental Authority of the United States of America, including without limitation, OFAC or the U.S. Department of State, or (b) the United Nations Security Council, Her Majesty's Treasury or the European Union.

"Second Amendment to Unsecured Term Loan Agreement" means that certain Second Amendment to Unsecured Term Loan Agreement dated as of October 31, 2017 among the Borrower, the General Partner, the lenders party thereto and Wells as administrative agent.

"<u>Specified Rate Management Obligations</u>" means all indebtedness, liabilities, obligations, covenants and duties of the Borrower or its Subsidiaries under or in respect of any Specified Rate Management Transaction, whether direct or indirect, absolute or contingent, due or not due, liquidated or unliquidated, and whether or not evidenced by any written confirmation.

"Specified Rate Management Provider" means any Lender, or any Affiliate of a Lender that is a party to a Specified Rate Management Transaction at the time the Specified Rate Management Transaction is entered into.

"Specified Rate Management Transaction" means any Rate Management Transaction that is made or entered into at any time, or in effect at any time now or hereafter, whether as a result of an assignment or transfer or otherwise, between the Borrower or its Subsidiaries and any Specified Rate Management Provider.

"Subsidiary" means as to any Person, a corporation, partnership or other entity of which shares of stock or other ownership interests having ordinary voting power (other than stock or such other ownership interests having such power only by reason of the happening of a contingency) to elect a majority of the board of directors or other managers of such corporation, partnership or other entity are at the time owned, or the management of which is otherwise controlled, directly or indirectly through one or more intermediaries, or both, by such Person, and provided such corporation, partnership or other entity is consolidated with such Person for financial reporting purposes under GAAP.

"Substitute Rate" means a floating rate of interest equal to (a) the Federal Funds Effective Rate from time to time <u>plus</u> one and one-half percent (1.50%) plus (b) the Applicable Margin for Adjusted Base Rate Loans.

"<u>Taxes</u>" means all present or future taxes, levies, imposts, duties, deductions, withholdings (including backup withholding), assessments, fees or other charges imposed by any Governmental Authority, including any interest, additions to tax or penalties applicable thereto.

"Term Loan Commitment" means with respect to each Lender, the commitment, if any, of such Lender to make a Loan hereunder. The amount of each Lender's Term Loan Commitment as of the Agreement Execution Date is set forth on Exhibit A. The aggregate amount of the Term Loan Commitments is \$260,000,000 as of the Agreement Execution Date. After giving effect to the Second Third Amendment to Unsecured Term Loan Agreement, the amount of each Lender's Term Loan Exposure shall be as set forth on Exhibit A. The Aggregate Term Loan Exposure is \$260,000,000 as of the effective date of the Second Third Amendment to Unsecured Term Loan Agreement.

"<u>Term Loan Exposure</u>" means with respect to any Lender at any time, the sum of the outstanding principal amount of such Lender's Loan at such time.

"Third Amendment to Unsecured Term Loan Agreement" means that certain Third Amendment to Unsecured Term Loan Agreement dated as of July 7, 2021 among the Borrower, the General Partner, the lenders party thereto and Wells as administrative agent.

"Total Term Loan Exposure" means at any time, the sum of the aggregate Term Loan Exposures for each of the Lenders.

"Transferee" is defined in Section 13.7 hereof.

"TRS" means a taxable REIT Subsidiary of the General Partner.

"Type" when used in reference to any Loan or Borrowing, refers to the rate by reference to which interest on such Loan, or on the Loans comprising such Borrowing, is determined. For purposes hereof, "rate" shall include the Adjusted LIBOR Rate and Adjusted Base Rate.

"UK Financial Institution" means any BRRD Undertaking (as such term is defined under the PRA Rulebook (as amended from time to time) promulgated by the United Kingdom Prudential Regulation Authority) or any person falling within IFPRU 11.6 of the FCA Handbook (as amended from time to time) promulgated by the United Kingdom Financial Conduct Authority, which

includes certain credit institutions and investment firms, and certain affiliates of such credit institutions or investment firms.

<u>"UK Resolution Authority" means the Bank of England or any other public administrative authority having responsibility for the resolution of any UK Financial Institution.</u>

"Unencumbered Asset" means any Project which as of any date of determination, (a) is not subject to any Liens other than Permitted Liens set forth in Sections 9.5(i) through 9.5(v), (b) is not subject to any agreement (including any agreement governing Indebtedness incurred in order to finance or refinance the acquisition of such asset) which prohibits or limits the ability of the Borrower, or its Wholly-Owned Subsidiaries, as the case may be, to create, incur, assume or suffer to exist any Lien upon any assets or Capital Stock of the Borrower, or any of its Wholly-Owned Subsidiaries other than a Permitted Negative Pledge, (c) is not subject to any agreement (including any agreement governing Indebtedness incurred in order to finance or refinance the acquisition of such asset) which entitles any Person to the benefit of any Lien (but not subject to any Liens other than Permitted Liens set forth in Sections 9.5(i) through 9.5(v)) on any assets or Capital Stock of the Borrower or any of its Wholly-Owned Subsidiaries or would entitle any Person to the benefit of any Lien (but excluding the Permitted Liens set forth in Sections 9.5(i) through 9.5(v)) on such assets or Capital Stock upon the occurrence of any contingency (including, except in the case of a Material Credit Facility, pursuant to an "equal and ratable" clause), (d) is not the subject of any material architectural/engineering issue, as evidenced by a certification of the Borrower, and (e) is materially compliant with the representations and warranties in Article VI below. Notwithstanding the foregoing, if any Project is a "Superfund" site under federal law or a site identified in writing by the jurisdiction in which such Project is located as having significant environmental contamination under applicable state law, the Borrower shall so advise the Lenders in writing and the Required Lenders shall have the right to request from the Borrower a current detailed environmental assessment (or one which is not more than two years old for Unencumbered Assets owned as of the Agreement Execution Date), and, if applicable, a written estimate of any remediation costs from a recognized environmental contractor and to exclude any such Project from Unencumbered Assets at their election. No Project of a Wholly-Owned Subsidiary shall be deemed to be unencumbered unless such Project and all Capital Stock of such Wholly-Owned Subsidiary or any other intervening Wholly-Owned Subsidiary between the Borrower and such Wholly-Owned Subsidiary is unencumbered and neither such Wholly-Owned Subsidiary nor any other intervening Wholly-Owned Subsidiary between the Borrower and such Wholly-Owned Subsidiary has any Indebtedness for borrowed money (other than Indebtedness due to the Borrower).

"Unencumbered Leverage Ratio" is defined in Section 9.7(c).

"Unencumbered Leverage Ratio Increase Period" is defined in Section 9.7(c).

"<u>Unimproved Land</u>" means land which constitutes a single tax parcel or separately platted lot and on which construction of an industrial building has not commenced.

"<u>Value of Unencumbered Assets</u>" means, for any Person as of any date, the sum (without duplication) of (a) the value of all Unencumbered Assets that are not Assets Under Development, Assets Acquired Not in Service or assets of the type described in clause (b) (determined in the

manner set forth below), plus (b) the purchase price paid by such Person (less any amounts paid to such Person as a purchase price adjustment, held in escrow, retained as a contingency reserve or in connection with other similar arrangements) for any Property (other than Assets Under Development) that constitutes an Unencumbered Asset and acquired by such Person during the immediately preceding period of four consecutive fiscal quarters, plus (c) any unrestricted cash, including any cash on deposit with a qualified intermediary with respect to a deferred tax-free exchange, plus (d) an amount equal to 100% of the then-current book value, determined in accordance with GAAP, of each first mortgage receivable secured by an income producing commercial property, provided that such first mortgage receivable is not subject to any Lien, plus (e) 100% of the then current book value of each Asset Under Development that constitutes an Unencumbered Asset plus (f) with respect to each Rollover Project, an amount equal to 50% of the then current book value, determined in accordance with GAAP, of each Rollover Project; provided that to the extent the aggregate amount of Value of Unencumbered Assets from Assets Acquired Not in Service and the other items set forth in clauses (d), (e) and (f) exceed 20% of the total Value of Unencumbered Assets, such excess shall be excluded. Unencumbered Assets that are not Assets Under Development or assets of the type described in clause (b) above shall be valued by dividing the Property Operating Income for such Project for the most recent four fiscal quarters by the Applicable Cap Rate (provided that for the purpose of such calculation, the Property Operating Income of each Unencumbered Asset that was formerly a Rollover Project shall in no event be less than zero). If a Project is no longer owned as of the date of calculation, then no value shall be included based on capitalizing Property Operating Income from such Project, except for purposes of such financial covenant comparing the Property Operating Income from Unencumbered Assets during a quarter to Debt Service for such quarter.

"Wells" means Wells Fargo Bank, National Association.

"<u>Wholly-Owned Subsidiary</u>" means a member of the Consolidated Operating Partnership 100% of the ownership interests in which are owned, directly or indirectly, by the Borrower and the General Partner in the aggregate.

"Write-Down and Conversion Powers" means, (a) with respect to any EEA Resolution Authority, the write-down and conversion powers of such EEA Resolution Authority from time to time under the Bail-In Legislation for the applicable EEA Member Country, which write-down and conversion powers are described in the EU Bail-In Legislation Schedule, and (b) with respect to the United Kingdom, any powers of the applicable Resolution Authority under the Bail-In Legislation to cancel, reduce, modify or change the form of a liability of any UK Financial Institution or any contract or instrument under which that liability arises, to convert all or part of that liability into shares, securities or obligations of that person or any other person, to provide that any such contract or instrument is to have effect as if a right had been exercised under it or to suspend any obligation in respect of that liability or any of the powers under that Bail-In Legislation that are related to or ancillary to any of those powers.

The foregoing definitions shall be equally applicable to both the singular and the plural forms of the defined terms.

1.2. <u>Financial Standards</u>. All financial computations required of a Person under this Agreement shall be made in accordance with GAAP as in effect on the date of this Agreement,

and all financial information required under this Agreement shall be prepared in accordance with GAAP, except that if any Person's financial statements are not audited, such Person's financial statements shall be prepared in accordance with the same sound accounting principles utilized in connection with the financial information submitted to Lenders with respect to the Borrower or the General Partner or the Properties in connection with this Agreement and shall be certified by an authorized representative of such Person. Moreover, all financial computations required of a Person under this Agreement shall be calculated (i) without giving effect to any election under Accounting Standards Codification 825-10-25 (or any other Accounting Standards Codification or Financial Accounting Standard having a similar result or effect) to value any Indebtedness or other liabilities of the Borrower, the General Partner or any Wholly Owned Subsidiary at "fair value", as defined therein and (ii) without giving effect to any treatment of Indebtedness in respect of convertible debt instruments under Accounting Standards Codification 470-20 (or any other Accounting Standards Codification or Financial Accounting Standard having a similar result or effect) to value any such Indebtedness in a reduced or bifurcated manner as described therein, and such Indebtedness shall at all times be valued at the full stated principal amount thereof. To the extent the Consolidated Operating Partnership has Defeased Debt, both the underlying debt and interest payable thereon and the financial assets used to defease such debt and interest earned thereon shall be excluded from calculations of the foregoing financial covenants.

- 1.3. <u>Classification of Loans and Borrowings</u>. For purposes of this Agreement, Loans may be classified and referred to by Type (e.g., a "<u>Eurocurrency Loan</u>"). Borrowings also may be classified and referred to by Type (e.g., a "<u>Eurocurrency Borrowing</u>").
- 1.4. <u>General References</u>. References in this Agreement to "Sections", "Articles", "Exhibits" and "Schedules" are to sections, articles, exhibits and schedules herein and hereto unless otherwise indicated. References in this Agreement to any document, instrument or agreement (a) shall include all exhibits, schedules and other attachments thereto, (b) shall include all documents, instruments or agreements issued or executed in replacement thereof, to the extent permitted hereby and (c) shall mean such document, instrument or agreement, or replacement or predecessor thereto, as amended, supplemented, restated or otherwise modified from time to time to the extent not otherwise stated herein or prohibited hereby and in effect at any given time. Wherever from the context it appears appropriate, each term stated in either the singular or plural shall include the singular and plural, and pronouns stated in the masculine, feminine or neuter gender shall include the masculine, the feminine and the neuter. Unless otherwise indicated, all references to time are references to Central Time.

1.5. Rates

- . The Administrative Agent does not warrant or accept responsibility for, and shall not have any liability with respect to, the administration, submission or any other matter related to the rates in the definition of "Base LIBOR Rate".
- 1.6. Divisions. For all purposes under the Loan Documents, in connection with any division or plan of division under Delaware law (or any comparable event under a different jurisdiction's laws): (a) if any asset, right, obligation or liability of any Person becomes the asset, right, obligation or liability of a different Person, then it shall be deemed to have been transferred from the original Person to the subsequent Person, and (b) if any new Person comes into existence, such new Person shall be deemed to have been organized on the first date of its existence by the holders of its Equity Interests at such time.

ARTICLE II.

THE FACILITY

2.1. The Facility.

- (a) Subject to the terms and conditions of this Agreement and in reliance upon the representations and warranties of the Borrower and the General Partner contained herein, each Lender agrees, severally and not jointly, to make Loans in Dollars to the Borrower on the Closing Date in an aggregate principal amount that will not result in (x) such Lender's Term Loan Exposure exceeding such Lender's Term Loan Commitment or (y) the Total Term Loan Exposure exceeding the Aggregate Term Loan Commitment. The Borrowings may be Adjusted Base Rate Borrowings or Eurocurrency Borrowings.
- (b) The Facility created by this Agreement shall terminate on the Maturity Date, unless sooner terminated in accordance with the terms of this Agreement. The Term Loan Commitment of each Lender shall terminate upon the earlier of (i) the funding of such Lender's Loan hereunder on the Closing Date and (ii) 5:00 p.m. (Central Time) on the Closing Date. Amounts repaid in respect of the Loans may not be reborrowed.
- Incremental Loans. The Borrower shall have the right to request one or more additional tranches of additional Loans (the "Incremental Loans") by providing written notice to the Administrative Agent, which notice shall be irrevocable once given; provided, however, that the aggregate amount of such Incremental Loans shall not exceed One Hundred Million Dollars (\$100,000,000). Each such Incremental Loan must be an aggregate minimum amount of \$20,000,000 and integral multiples of \$5,000,000 in excess thereof. The Administrative Agent and the Arrangers, in consultation with the Borrower, shall manage all aspects of such Incremental Loans, including decisions as to the selection of the existing Lenders and/or other banks, financial institutions and other institutional lenders to be approached with respect to such increase and the allocations of Incremental Loans among such existing Lenders and/or other banks, financial institutions and other institutional lenders. No Lender shall be obligated in any way whatsoever to provide an Incremental Loan, and any new Lender becoming a party to this Agreement in connection with any such requested Incremental Loan must be an Eligible Assignee. Effecting the Incremental Loans under this Section is subject to the following conditions precedent: (w) the Administrative Agent shall have consented thereto (such consent not to be unreasonably withheld or delayed), (x) no Default or Event of Default shall be in existence on the effective date of such Incremental Loans, (y) the representations and warranties made or deemed made by the Borrower or any other Loan Party in any Loan Document to which such Loan Party is a party shall be true and correct on the effective date of such Incremental Loans except to the extent that such representations and warranties expressly relate solely to an earlier date (in which case such representations and warranties shall have been true and correct on and as of such earlier date) and except for changes in factual circumstances specifically and expressly permitted hereunder, and (z) the Administrative Agent shall have received each of the following, in form and substance satisfactory to the Administrative Agent: (i) if not previously delivered to the Administrative Agent, copies certified by a Qualified Officer of the Borrower of (A) all partnership or other necessary action taken by the Borrower to authorize such Incremental Loans and (B) all corporate or other necessary action taken by Guarantor authorizing the guaranty of such Incremental Loans;

and (ii) an opinion of counsel to the Borrower and the Guarantor, and addressed to the Administrative Agent and the Lenders covering such matters as reasonably requested by the Administrative Agent; and (iii) if requested, new Notes executed by the Borrower, payable to any new Lenders and replacement Notes, if requested by any applicable Lender, executed by the Borrower, payable to any existing Lenders making Incremental Loans, in the amount of each such Lender's Term Loan Exposure at the time of the effectiveness of the applicable Incremental Loans. In connection with any Incremental Loans pursuant to this Section 2.2 any Lender becoming a party hereto shall execute such documents and agreements as the Administrative Agent may reasonably request. The Incremental Loans (i) shall rank pari passu in right of payment with the existing Loans, (ii) shall not mature earlier than the Maturity Date and (iii) shall be treated substantially the same as (and in any event no more favorably than) the existing Loans. Incremental Loans may be made hereunder pursuant to an amendment or restatement of this Agreement and, as appropriate, the other Loan Documents, executed by the Borrower, each new Lender participating in such tranche (if any), each existing Lender participating in such tranche (if any) and the Administrative Agent. The Administrative Agent may, without the consent of any other Lenders (except as expressly required pursuant to Section 14.3), effect such amendments to this Agreement and the other Loan Documents as may be necessary or appropriate, in the reasonable opinion of the Administrative Agent, to effect the provisions of this Section 2.2.

- 2.3. <u>Principal Payments</u>. Any outstanding Borrowings and all other unpaid Obligations shall be paid in full by the Borrower on the Maturity Date.
- 2.4. <u>Requests for Borrowings; Responsibility for Borrowings</u>. Borrowings shall be made available to the Borrower by Administrative Agent in accordance with <u>Section 2.1(a)</u> and <u>Section 2.7</u> hereof. The obligation of each Lender to fund its Percentage of each Borrowing shall be several and not joint.
- 2.5. Evidence of Credit Extensions. Any Lender may request that Loans made by it be evidenced by a Note. In such event, the Borrower shall prepare, execute and deliver to such Lender a Note payable to the order of such Lender (or, if requested by such Lender, to such Lender and its registered assigns). Thereafter, the Loans evidenced by such Note and interest thereon shall at all times (including after assignment pursuant to Section 13.3) be represented by one or more Notes in such form payable to the order of the payee named therein (or, if such Note is a registered note, to such payee and its registered assigns). Each Lender may record Borrowings and principal payments thereof on the schedule attached to its Note or, at its option, in its records, and each Lender's record thereof shall be conclusive absent the Borrower furnishing to such Lender conclusive and irrefutable evidence of an error made by such Lender with respect to that Lender's records. Notwithstanding the foregoing, the failure to make, or an error in making, a notation with respect to any Borrowing shall not limit or otherwise affect the obligations of the Borrower hereunder or under the Notes to pay the amount actually owed by the Borrower to Lenders.

2.6. Loans and Borrowings.

(a) Each Loan shall be made as part of a Borrowing consisting of Loans of the same Type made by the Lenders ratably in accordance with their respective Percentages. The failure of any Lender to make any Loan required to be made by it shall not relieve any other Lender of its obligations hereunder.

- (b) Subject to Section 4.3, each Borrowing shall be comprised entirely of Adjusted Base Rate Loans or Eurocurrency Loans as the Borrower may request in accordance herewith. Each Lender at its option may make any Eurocurrency Loan by causing any domestic or foreign branch or Affiliate of such Lender to make such Loan; provided that any exercise of such option shall not affect the obligation of the Borrower to repay such Loan in accordance with the terms of this Agreement.
 - (c) No more than ten (10) Eurocurrency Borrowings may be outstanding at any one time under the Facility.
- (d) Notwithstanding any other provision of this Agreement, the Borrower shall not be entitled to request, or to elect to convert or continue, any Borrowing if the Interest Period requested with respect thereto would end after the Maturity Date.

2.7. Requests for Borrowings.

(a) To request a Borrowing, the Borrower shall notify the Administrative Agent of such request by telephone (a "Borrowing Notice") (A) in the case of a Eurocurrency Borrowing, not later than 11:00 a.m. (Central Time), three (3) Business Days before the date of the proposed Borrowing, or (B) in the case of an Adjusted Base Rate Borrowing, not later than 11:00 a.m. (Central Time), one (1) Business Day before the date of the proposed Borrowing. Each such telephonic Borrowing Notice shall be irrevocable and, if requested by the Administrative Agent, shall be confirmed promptly by delivery to the Administrative Agent of a written Borrowing Notice in substantially the form of Exhibit J in the case of any Borrowing. Each such telephonic and written Borrowing Notice shall specify the following information in compliance with Section 2.7: (i) the Borrower requesting such Borrowing; (ii) the Type of the requested Borrowing; (iii) the aggregate amount of such Borrowing; (iv) the date of such Borrowing, which shall be a Business Day; (v) in the case of a Eurocurrency Borrowing, the initial Interest Period to be applicable thereto; and (vi) the location and number of the Borrower's account to which funds are to be disbursed, which shall comply with the requirements of Section 2.7. If no election as to the Type of Borrowing is specified, then the requested Borrowing shall be an Adjusted Base Rate Borrowing. If no Interest Period is specified with respect to any requested Eurocurrency Borrowing, then the Borrower shall be deemed to have selected an Interest Period of one month's duration. Promptly following receipt of a Borrowing Notice in accordance with this Section, the Administrative Agent shall advise each relevant Lender of the details thereof and of the amount of such Lender's Loan to be made as part of the requested Borrowing.

The Borrower shall also deliver together with each Borrowing Notice the compliance certificate required in <u>Section 5.2</u> and otherwise comply with the conditions set forth in <u>Section 5.2</u> for Borrowings.

Not later than 11:00 a.m. (Central Time) on each Borrowing Date, each Lender shall make available its Loan or Loans, in funds immediately available at the Administrative Office. Administrative Agent will promptly make the funds so received

from the Lenders available to the Borrower in the account specified by the Borrower in the Disbursement Instruction Agreement.

- (b) Administrative Agent shall, as soon as practicable after receipt of a Borrowing Notice, determine the Adjusted LIBOR Rate applicable to the requested Eurocurrency Borrowing and inform the Borrower and Lenders of the same. Each determination of the Adjusted LIBOR Rate by Administrative Agent shall be conclusive and binding upon the Borrower in the absence of manifest error.
- (c) If the Borrower shall prepay a Eurocurrency Borrowing other than on the last day of the Interest Period applicable thereto, the Borrower shall be responsible to pay all amounts due to Lenders as required by <u>Section 4.4</u> hereof. The Lenders shall not be obligated to match fund their Eurocurrency Borrowings.
- (d) The right of the Borrower to select the Adjusted LIBOR Rate and the Adjusted Base Rate for a Borrowing pursuant to this Agreement is subject to the availability to Lenders, as applicable, of a similar option. If Administrative Agent determines that (i) deposits of Dollars in an amount approximately equal to the Borrowing for which the Borrower wishes to select the Adjusted LIBOR Rate or Adjusted Base Rate are not generally available at such time in the London interbank eurodollar market, or (ii) the rate at which the deposits described in subsection (i) herein are being offered will not adequately and fairly reflect the costs to Lenders, as applicable, of maintaining an Adjusted LIBOR Rate or Adjusted Base Rate on a Borrowing or of funding the same in such market for such Interest Period, or (iii) reasonable means do not exist for determining an Adjusted LIBOR Rate or Adjusted Base Rate, or (iv) the Adjusted LIBOR Rate or Adjusted Base Rate would be in excess of the maximum interest rate which the Borrower may by law pay, then in any of such events, Administrative Agent shall so notify the Borrower and Lenders and notwithstanding anything herein to the contrary, such Borrowing shall bear interest at the Substitute Rate.

2.8. <u>Interest Elections</u>.

(a) Each Borrowing initially shall be of the Type specified in the applicable Borrowing Request and, in the case of a Eurocurrency Borrowing, shall have an initial Interest Period as specified in such Borrowing Request (or as set forth in Section 2.7 if no Interest Period is specified). Thereafter, the Borrower may elect to convert such Borrowing to a different Type or to continue such Borrowing and, in the case of a Eurocurrency Borrowing, may elect Interest Periods therefor, all as provided in this Section. The Borrower may elect different options with respect to different portions of the affected Borrowing, in which case each such portion shall be allocated ratably among the Lenders holding the Loans comprising such Borrowing, and the Loans comprising each such portion shall be considered a separate Borrowing. Notwithstanding the foregoing, the Borrower may not (i) elect an Interest Period for Eurocurrency Loans that does not comply with Section 2.6(d), (ii) elect to convert any Adjusted Base Rate Loans to Eurocurrency Loans that would result in the number of Eurocurrency Borrowings exceeding the maximum number of Eurocurrency Borrowings permitted under Section 2.6(c) or (iii) elect an Interest Period for Eurocurrency Loans unless the aggregate outstanding principal amount of Eurocurrency Loans to which such Interest Period will apply complies with the requirements as to minimum principal amount set forth in Section 2.11.

- (b) To make an election pursuant to this Section (an "<u>Interest Election Request</u>"), the Borrower shall notify the Administrative Agent of such election by telephone by the time that a Borrowing Request would be required under <u>Section 2.7</u> if the Borrower were requesting a Borrowing of the Type resulting from such election to be made on the effective date of such election. Each such telephonic Interest Election Request shall be irrevocable and shall be confirmed promptly by delivery to the Administrative Agent of a written Interest Election Request in a form approved by the Administrative Agent (hereinafter referred to as a "Conversion/Continuation Notice").
- (c) Each telephonic and written Conversion/Continuation Notice shall specify the following information in compliance with Section 2.6 and paragraph (a) of this Section: (i) the Borrowing to which such Conversion/Continuation Notice applies; (ii) the effective date of the election made pursuant to such Conversion/Continuation Notice, which shall be a Business Day; (iii) whether the resulting Borrowing is to be an Adjusted Base Rate Borrowing or a Eurocurrency Borrowing; and (iv) if the resulting Borrowing is a Eurocurrency Borrowing, the Interest Period to be applicable thereto after giving effect to such election. If any such Conversion/Continuation Notice requests a Eurocurrency Borrowing but does not specify an Interest Period, then the Borrower shall be deemed to have selected an Interest Period of one month's duration.
- (d) Promptly following receipt of a Conversion/Continuation Notice, the Administrative Agent shall advise each relevant Lender of the details thereof and of such Lender's portion of each resulting Borrowing.
- (e) If the Borrower fails to timely deliver a Conversion/Continuation Notice with respect to a Eurocurrency Borrowing in accordance with Section 2.8(b) prior to the end of the Interest Period applicable thereto, then, unless such Borrowing is repaid as provided herein, at the end of such Interest Period such Borrowing shall be continued as a Eurocurrency Borrowing with the same Interest Period as the Eurocurrency Borrowing being continued (unless such deemed election shall result in an Interest Period ending after the Maturity Date, in which case the Borrower shall be deemed to have elected a one (1) month Interest Period which will not end after the Maturity Date; provided however that if less than 30 days remain until the Maturity Date, such Eurocurrency Borrowing shall bear interest at the Adjusted Base Rate). The Administrative Agent shall advise each relevant Lender and the Borrower of the details of any such deemed continuation under this clause (e).
- (f) Notwithstanding anything to the contrary contained in this <u>Section 2.8</u>, no Borrowing may be converted into a Eurocurrency Borrowing or continued as a Eurocurrency Borrowing (except with the consent of the Required Lenders) when any Monetary Default or Event of Default has occurred and is continuing.

2.9. <u>Applicable Margins</u>.

(a) The Base Rate Applicable Margin and the LIBOR Applicable Margin to be used in calculating the interest rate applicable to different Types of Borrowings, shall vary from time to time in accordance with the Consolidated Leverage Ratio as follows:

Consolidated Leverage Ratio	LIBOR Applicable Margin	Base Rate Applicable Margin
Less than 35%	1.20%	1.20%
35% or greater but less than 40%	1.30%	1.30%
40% or greater but less than 45%	1.35%	1.35%
45% or greater but less than 50%	1.40%	1.40%
50% or greater but less than 55%	1.50%	1.50%
55% or greater	1.70%	1.70%

The LIBOR Applicable Margin and Base Rate Applicable Margin shall be determined by the Administrative Agent from time to time, based on the Consolidated Leverage Ratio as set forth in the compliance certificate most recently delivered by the Borrower pursuant to Section 8.2(iv). Any adjustment to the LIBOR Applicable Margin and Base Rate Applicable Margin shall be effective as of the first day of the calendar month immediately following the month during which the Borrower delivers to the Administrative Agent the applicable compliance certificate pursuant to Section 8.2(iv). If the Borrower fails to deliver a compliance certificate in accordance with Section 8.2(iv), the LIBOR Applicable Margin and the Base Rate Applicable Margin shall equal the percentages corresponding to a Consolidated Leverage Ratio of 55% or greater until the first day of the calendar month immediately following the month that the required compliance certificate is delivered. Notwithstanding the foregoing, for the period from the Agreement Execution Date through but excluding the date on which the Administrative Agent first determines the LIBOR Applicable Margin and the Base Rate Applicable Margin as set forth above, the LIBOR Applicable Margin and the Base Rate Applicable Margin shall be determined based on a Consolidated Leverage Ratio of "Less than 35%". Thereafter, such LIBOR Applicable Margin and Base Rate Applicable Margin shall be adjusted from time to time as set forth in this definition. It is understood and agreed that each change in pricing level shall apply during the period commencing on the effective date of such change and ending on the date immediately preceding the effective date of the next such change.

The parties understand that the applicable interest rate for the Obligations and certain fees set forth herein may be determined and/or adjusted from time to time based upon certain financial ratios and/or other information to be provided or certified to the Lenders by the Borrower (the "Borrower Information"). If it is subsequently determined that any such Borrower Information was incorrect at the time it was delivered to the Administrative Agent as the result of fraud or intentional misstatement thereof, and if the applicable interest rate or fees calculated for any period were lower than they should have been had the correct information been timely provided, then, such interest rate and such fees for such period shall be automatically recalculated using correct Borrower Information. The Administrative Agent shall promptly notify the Borrower in writing of

any additional interest and fees due because of such recalculation, and the Borrower shall pay such additional interest or fees due to the Administrative Agent, for the account of each Lender, within five (5) Business Days of receipt of such written notice. Any recalculation of interest or fees required by this provision shall survive the termination of this Agreement, and this provision shall not in any way limit any of the Administrative Agent's or any Lender's other rights under this Agreement.

(b) If the Borrower has an Investment Grade Rating during the term of the Facility, at the one-time election of the Borrower upon prior irrevocable written notice to Administrative Agent and the Lenders, which election shall relate solely to this Agreement and shall have no effect on elections to be made with respect to the Existing Revolving Credit Agreement or the Existing Term Loan Agreement, from and after such election the Base Rate Applicable Margin and the LIBOR Applicable Margin shall vary from time to time in accordance with the Investment Grade Rating as follows (such that the Applicable Margin shall change from time to time as and when the Investment Grade Rating changes, which changes shall be effective from and after the date that notice is delivered from the Borrower to the Administrative Agent of the applicable Investment Grade Rating change):

Rating Level	LIBOR Applicable Margin	Base Rate Applicable Margin
A-/A3	0.90%	0.90%
BBB+/Baa1	0.95%	0.95%
BBB/Baa2	1.10%	1.10%
BBB-/Baa3	1.35%	1.35%
Below BBB- and Baa3 (" <u>Level V</u> ")	1.75%	1.75%

The Applicable Margin shall be determined by the higher of the two ratings from S&P or Moody's. In the event that such two ratings are more than one rating level apart and both are Investment Grade Ratings, then the rating level one level above the lower of the two ratings shall apply. If only one Investment Grade Ratings has been issued, the Applicable Margin shall be determined based on the sole Investment Grade Rating then in effect. If Investment Grade Ratings shall have been assigned by both rating agencies and thereafter the Borrower does not have an Investment Grade Rating from either Rating Agency, the Applicable Margin shall be determined based on Level V of the forgoing table in this Section 2.9(b). The Borrower may not return to pricing determinations based on the table in Section 2.9(a) after selecting to use the table in Section 2.9(b).

- 2.10. <u>Funds Transfer Disbursements</u>. The Borrower hereby authorizes the Administrative Agent to disburse the proceeds of any Loan made by the Lenders or any of their Affiliates pursuant to the Loan Documents as requested by an authorized representative of the Borrower to any of the accounts designated in the Disbursement Instruction Agreement.
- 2.11. <u>Minimum Amount of Each Borrowing</u>. Each Eurocurrency Borrowing shall be in the minimum amount of \$2,000,000 (and in multiples of \$100,000 if in excess thereof), and each

Adjusted Base Rate Borrowing shall be in the minimum amount of \$1,000,000 (and in multiples of \$100,000 if in excess thereof).

2.12. <u>Interest</u>.

- (a) Subject to <u>Section 2.14</u>, the outstanding principal balance of the Loans shall bear interest from time to time at a rate per annum equal to:
 - (i) the Adjusted Base Rate; or
 - (ii) at the election of the Borrower in accordance with <u>Section 2.8</u> with respect to all or portions of the Obligations, the Adjusted LIBOR Rate.
- (b) All interest shall be calculated for actual days elapsed on the basis of a 360-day year. Interest accrued on each Borrowing shall be payable on the first day of each calendar month in arrears from time to time while such Borrowing is outstanding (or, if earlier, on the Maturity Date). Interest shall not be payable for the day of any payment on the amount paid if payment is received by Administrative Agent prior to 1:00 p.m. (Central Time). If any payment of principal or interest on the Loans, or fees hereunder or under any other Loan Document, shall become due on a day that is not a Business Day, such payment shall be made on the next succeeding Business Day and, in the case of a payment of principal, such extension of time shall be included in computing interest due in connection with such payment; provided that for purposes of Section 10.1 hereof, any payments of principal described in this sentence shall be considered to be "due" on such next succeeding Business Day.

2.13. Method of Payment.

- (a) All payments of the Obligations hereunder shall be made, without set off, deduction, or counterclaim, in immediately available funds to Administrative Agent at the Administrative Office or at any other Lending Office of Administrative Agent specified in writing by Administrative Agent to the Borrower, by 11:00 a.m. (Central Time) on the date when due and shall be applied ratably by Administrative Agent among Lenders in the Facility. Each such payment shall be made in Dollars. Each payment delivered to Administrative Agent for the account of any Lender shall be delivered promptly by Administrative Agent to such Lender in the same type of funds that Administrative Agent received at its address specified herein or at any Lending Office specified in a notice received by Administrative Agent from such Lender. Payments not made by Administrative Agent within one (1) Business Day after receipt shall accrue interest at Federal Funds Effective Rate. Administrative Agent is hereby authorized to charge the account of the Borrower maintained with Wells for each payment of principal, interest and fees as it becomes due hereunder.
- (b) If any Lender shall fail to make any payment required to be made by it pursuant to <u>Sections 2.7</u>, <u>2.16</u> or <u>12.8</u>, then the Administrative Agent shall, notwithstanding any contrary provision hereof, (i) apply any amounts thereafter received by the Administrative Agent for the account of such Lender for the benefit of the Administrative

Agent to satisfy such Lender's obligations to it under such Section until all such unsatisfied obligations are fully paid, and/or (ii) hold any such amounts in a segregated account as cash collateral for, and application to, any future funding obligations of such Lender under any such Section, in the case of each of clauses (i) and (ii) above, in any order as determined by the Administrative Agent in its discretion.

- 2.14. <u>Default</u>. Notwithstanding the foregoing, during the continuance of a Monetary Default or an Event of Default, the Borrower shall not have the right to request a Eurocurrency Borrowing, select a new Interest Period for an existing Eurocurrency Borrowing or convert any Adjusted Base Rate Borrowing to a Eurocurrency Borrowing. During the continuance of a Monetary Default or an Event of Default, at the election of the Required Lenders, by notice to the Borrower, outstanding Borrowings shall bear interest at the applicable Default Rates until such Monetary Default or Event of Default ceases to exist or the Obligations are paid in full.
- 2.15. <u>Lending Offices</u>. Each Lender may book its Borrowings at any Lending Office selected by such Lender and may change its Lending Office from time to time. All terms of this Agreement shall apply to any such Lending Office and any Note shall be deemed held by the applicable Lender for the benefit of such Lending Office. Each Lender may, by written or telex notice to the Administrative Agent and the Borrower, designate a Lending Office through which Borrowings will be made by it and for whose account payments are to be made.
- 2.16. Non Receipt of Funds by Administrative Agent. Unless the Borrower or a Lender, as the case may be, notifies Administrative Agent prior to the date on which it is scheduled to make payment to Administrative Agent of (i) in the case of a Lender, a Borrowing, or (ii) in the case of the Borrower, a payment of principal, interest or fees to the Administrative Agent for the account of the Lenders, that it does not intend to make such payment, Administrative Agent may assume that such payment has been made. Administrative Agent may, but shall not be obligated to, make the amount of such payment available to the intended recipient in reliance upon such assumption. If such Lender or the Borrower, as the case may be, has not in fact made such payment to Administrative Agent, the recipient of such payment shall, promptly after demand by Administrative Agent, repay to Administrative Agent the amount so made available together with interest thereon in respect of each day during the period commencing on the date such amount was so made available by Administrative Agent until the date Administrative Agent recovers such amount at a rate per annum equal to (i) in the case of payment by a Lender, the Federal Funds Effective Rate (as determined by Administrative Agent) or (ii) in the case of payment by the Borrower, the interest rate applicable to Adjusted Base Rate Borrowings. If the Borrower and any Lender shall each pay any such amount made available by the Administrative Agent on behalf of such Lender for the same or overlapping period, the Administrative Agent shall promptly remit to the Borrower the amount paid by the Borrower for such period. Any payment by the Borrower shall be without prejudice to any claim the Borrower may have against a Lender that shall have failed to make available the proceeds of a Loan to be made by such Lender.
 - 2.17. [<u>Reserved</u>].
 - 2.18. [<u>Reserved</u>].
 - 2.19. [<u>Reserved</u>].

2.20. Pro Rata Treatment/Sharing of Payments.

- (a) Except to the extent otherwise provided herein: (i) each Borrowing from the Lenders under Sections 2.7 shall be made from the Lenders pro rata according to the amounts of their respective Percentages; (ii) each payment or prepayment of principal of Loans shall be made for the account of the Lenders pro rata in accordance with the respective unpaid principal amounts of the Loans held by them, provided that, subject to Section 12.16, if immediately prior to giving effect to any such payment in respect of any Loans the outstanding principal amount of the Loans shall not be held by the Lenders pro rata in accordance with their respective Percentages in effect at the time such Loans were made, then such payment shall be applied to the Loans in such manner as shall result, as nearly as is practicable, in the outstanding principal amount of the Loans being held by the Lenders pro rata in accordance with their respective Percentages; (iii) each payment of interest on Loans shall be made for the account of the Lenders, pro rata in accordance with the amounts of interest on such Loans, then due and payable to the respective Lenders and (iv) the making, conversion and continuation of Loans of a particular Type shall be made pro rata among the Lenders, according to the amounts of their respective Loans, and the then current Interest Period for each Lender's portion of each such Loan of such Type shall be coterminous.
- If a Lender shall obtain payment of any principal of, or interest on, any Loan made by it to the Borrower under this Agreement or shall obtain payment on any other Obligation owing by the Borrower or any other Loan Party through the exercise of any right of set-off, banker's lien, counterclaim or similar right or otherwise or through voluntary prepayments directly to a Lender or other payments made by or on behalf the Borrower or any other Loan Party to a Lender (other than any payment in respect of Specified Rate Management Obligations) not in accordance with the terms of this Agreement and such payment should be distributed to the Lenders in accordance with Section 2.20(a) or Section 2.22, as applicable, such Lender shall promptly purchase from the other Lenders participations in (or, if and to the extent specified by such Lender, direct interests in) the Loans made by the other Lenders or other Obligations owed to such other Lenders in such amounts, and make such other adjustments from time to time as shall be equitable, to the end that all the Lenders shall share the benefit of such payment (net of any reasonable expenses which may actually be incurred by such Lender in obtaining or preserving such benefit) in accordance with the requirements of Section 2.20(a) or Section 2.22, as applicable. To such end, all the Lenders shall make appropriate adjustments among themselves (by the resale of participations sold or otherwise) if such payment is rescinded or must otherwise be restored. The Borrower agrees that any Lender so purchasing a participation (or direct interest) in the Loans or other Obligations owed to such other Lenders may exercise all rights of set-off, banker's lien, counterclaim or similar rights with respect to such participation as fully as if such Lender were a direct holder of Loans in the amount of such participation. Nothing contained herein shall require any Lender to exercise any such right or shall affect the right of any Lender to exercise and retain the benefits of exercising, any such right with respect to any other indebtedness or obligation of the Borrower.

2.21. [<u>Reserved</u>].

- 2.22. <u>Application of Moneys Received</u>. Subject to <u>Section 2.13(b)</u> hereof, all moneys collected or received by the Administrative Agent on account of the Facility directly or indirectly, shall be applied in the following order of priority:
 - (i) to the payment of all reasonable costs incurred in the collection of such moneys of which the Administrative Agent shall have given notice to the Borrower;
 - (ii) to the reimbursement of any yield protection due to any of the Lenders in accordance with Section 4.1;
 - (iii) to the payment of all fees then due to the Administrative Agent;
 - (iv) [reserved];
 - (v) first to interest until paid in full and then to principal for all Lenders (x) for Borrowings (unless an Event of Default exists) (the amount allocated to Borrowings to be distributed in accordance with the applicable pro rata shares of the outstanding amounts of the Lenders) or (y) if an Event of Default exists, in accordance with the respective Percentages of the Lenders until principal is paid in full, each Lenders' share of such payment to be allocated pro rata among the outstanding Types of Loans owed to such Lender; and
 - (vi) any other sums due to the Administrative Agent or any Lender under any of the Loan Documents.

2.23. [<u>Reserved</u>].

2.24. <u>Prepayments of Loans</u>.

- (a) The Borrower shall have the right at any time and from time to time to prepay any of its Borrowings in whole or in part, subject to prior notice in accordance with paragraph (b) of this Section and payment of a prepayment premium in accordance with paragraph (c) of this Section.
- (b) The Borrower shall notify the Administrative Agent by telecopyemail of any prepayment under clause (a) above not later than 11:00 a.m. (Central Time) on the date of prepayment. Each such notice shall be irrevocable and shall specify the prepayment date and the principal amount of each Borrowing or portion thereof to be prepaid. Promptly following receipt of any such notice relating to a Borrowing, the Administrative Agent shall advise the Lenders of the contents thereof. Each partial prepayment pursuant to clause (a) above of any Borrowing shall be in an amount that would be permitted in the case of an advance of a Borrowing of the same Type as provided in Section 2.11 hereof. Each prepayment of a Borrowing shall be applied ratably to the Loans included in the applicable prepaid Borrowing. Prepayments shall be accompanied by accrued interest to the extent required by Section 2.12 hereof and by breakage costs to the extent required by Section 4.4 hereof.

(c) Upon the prepayment (whether voluntary or otherwise) of any Borrowing hereunder, the Borrower shall pay a prepayment premium (the "<u>Prepayment Premium</u>") equal to the percentage of the principal amount so prepaid set forth in the following table corresponding to the period during which such prepayment is made:

Period	Prepayment Premium
After the Closing Date and on or prior to the 1-year anniversary of the Closing Date	2.0%
After the 1-year anniversary of the Closing Date and on or prior to the 2-year anniversary of the Closing Date	1.0%
After the 2-year anniversary of the Closing Date	0.0%

The Prepayment Premium shall be due and payable on the date of any such prepayment.

ARTICLE III.

[RESERVED]

ARTICLE IV.

CHANGE IN CIRCUMSTANCES

4.1. <u>Yield Protection</u>. If any Change in Law:

- (i) subjects the Administrative Agent, any Lender or any applicable Lending Office to any taxes, duties, levies, imposts, deductions, assessments, fees, charges or withholdings, and all liabilities with respect thereto (other than (A) Taxes, (B) Excluded Taxes or (C) Other Taxes) on its loans, loan principal, letters of credit, commitments, or other obligations, or its deposits, reserves, other liabilities, capital or liquidity attributable to making, converting, funding or maintaining its Borrowings or its Term Loan Commitment, or
- (ii) imposes or increases or deems applicable any reserve, assessment, insurance charge, special deposit, liquidity or similar requirement against assets of, deposits with or for the account of, or credit extended by, any Lender or any applicable Lending Office (other than reserves and assessments taken into account in determining the interest rate applicable to Eurocurrency Borrowings), or
- (iii) imposes any other condition, and the result is to increase the cost of any Lender or any applicable Lending Office of making, funding or maintaining loans or reduces any amount receivable by any Lender or any applicable Lending Office in connection with loans, or requires any Lender or any applicable Lending Office to make any payment calculated by reference to the amount of loans held or interest received by it, by an amount deemed material by such Lender,

then, within fifteen (15) days of demand by the Administrative Agent or such Lender, the Borrower shall pay the Administrative Agent or such Lender that portion of such increased expense incurred or reduction in an amount received which the Administrative Agent or such Lender determines is attributable to making, funding and maintaining its Borrowings and its Term Loan Commitment (which determination shall be made in good faith (and not on an arbitrary or capricious basis) and consistent with similarly situated customers of the applicable Lender after consideration of such factors as such Lender then reasonably determines to be relevant).

4.2. <u>Changes in Capital Adequacy Regulations</u>. If a Lender determines the amount of capital or liquidity required or expected to be maintained by such Lender, any Lending Office of such Lender or any corporate entity controlling such Lender is increased as a result of a Change in Law (which determination shall be made in good faith (and not on an arbitrary or capricious basis) and consistent with similarly situated customers of the applicable Lender after consideration of such factors as such Lender then reasonably determines to be relevant), then, within fifteen (15) days of demand by such Lender, the Borrower shall pay such Lender the amount necessary to compensate for any shortfall in the rate of return on the portion of such increased capital which is attributable to this Agreement, its Borrowings or its obligation to make Borrowings hereunder (after taking into account such Lender's policies as to capital adequacy). Without in any way affecting the Borrower's obligation to pay compensation actually claimed by a Lender under this Section 4.2, the Borrower shall have the right to replace any Lender which has demanded such compensation provided that the Borrower notifies such Lender that it has elected to replace such Lender and notifies such Lender and the Administrative Agent of the identity of the proposed replacement Lender not more than six (6) months after the date of such Lender's most recent demand for compensation under this Section 4.2, and further provided that such replacement is otherwise in accordance with Section 4.7. The Lender being replaced shall assign its Term Loan Exposure and its rights and obligations under this Facility to the replacement Lender in accordance with the requirements of Section 13.3 hereof and the replacement Lender shall assume such Term Loan Exposure, all pursuant to an assignment agreement substantially in the form of Exhibit I hereto. The purchase by the replacement Lender shall be at par (plus all accrued and unpaid interest and any other sums owed to such Lender being replaced hereunder) which shall be paid to the Lender being replaced upon the execution and delivery of the assignment. The Lender being replaced shall continue to be entitled to the benefits of Sections 4.1, 4.2, 4.4, 4.5 and 14.6 for events recurring prior to assignment to the replacement Lender.

4.3. Availability of Eurocurrency or Adjusted Base Rate Borrowings.

(i) Unless and until a Replacement Rate is implemented in accordance with clause (ii) below, if (a) any Lender determines that maintenance of any of its Eurocurrency Borrowings or Adjusted Base Rate Borrowings (sometimes collectively referred to as "LIBOR Borrowings") at a suitable Lending Office would violate any Applicable Law, the Administrative Agent shall suspend by written notice to the Borrower (with a copy thereof being delivered contemporaneously to Lenders) the availability of LIBOR Borrowings and require any LIBOR Borrowings to be repaid; or (b) if the Required Lenders determine that (1) deposits of a type or maturity appropriate to match fund LIBOR Borrowings are not available, the Administrative Agent shall suspend by written notice to the

Borrower (with a copy thereof being delivered contemporaneously to Lenders) the availability of LIBOR Borrowings with respect to any LIBOR Borrowings made after the date of any such determination, or (2) an interest rate applicable to a LIBOR Borrowing does not accurately reflect the cost of making a Eurocurrency Borrowing, and, if for any reason whatsoever the provisions of Section 4.1 are inapplicable, the Administrative Agent shall suspend by written notice to the Borrower (with a copy thereof being delivered contemporaneously to Lenders) the availability of LIBOR Borrowings with respect to any LIBOR Borrowings made after the date of any such determination. If this occurs, Borrowings shall, to the extent permissible under Applicable Law, bear interest at the Substitute Rate, rather than any Loans becoming payable.

Notwithstanding anything to the contrary in Section 4.3(i) above, if the Administrative Agent has made the determination (such determination to be conclusive absent manifest error) that (i) the circumstances described in Section 4.3(i)(b) have arisen and that such circumstances are unlikely to be temporary, (ii) any applicable interest rate specified herein is no longer a widely recognized benchmark rate for newly originated loans in Dollars in the U.S. market or (iii) the applicable supervisor or administrator (if any) of any applicable interest rate specified herein or any Governmental Authority having or purporting to have jurisdiction over the Administrative Agent has made a public statement identifying a specific date after which any applicable interest rate specified herein shall no longer be used for determining interest rates for loans in Dollars in the U.S. market, then the Administrative Agent may, to the extent practicable (in consultation with the Borrower and as determined by the Administrative Agent to be generally in accordance with similar situations in other transactions in which it is serving as administrative agent or otherwise consistent with market practice generally), establish a replacement interest rate (the "Replacement Rate"), in which case, the Replacement Rate shall, subject to the next two sentences, replace such applicable interest rate for all purposes under the Loan Documents unless and until (A) an event described in Section 4.3(i)(b) occurs with respect to the Replacement Rate or (B) the Administrative Agent (or the Required Lenders through the Administrative Agent) notifies the Borrower that the Replacement Rate does not adequately and fairly reflect the cost to the Lenders of funding the Loans bearing interest at the Replacement Rate. In connection with the establishment and application of the Replacement Rate, this Agreement and the other Loan Documents shall be amended solely with the consent of the Administrative Agent and the Borrower, as may be necessary or appropriate, in the opinion of the Administrative Agent, to effect the provisions of this Section 4.3(ii). Notwithstanding anything to the contrary in this Agreement or the other Loan Documents (including, without limitation, Section 14.13), such amendment shall become effective without any further action or consent of any other party to this Agreement so long as the Administrative Agent shall not have received, within five (5) Business Days of the delivery of such amendment to the Lenders, a written notice signed by Lenders constituting Required Lenders stating that such Lenders object to such amendment (which such notice shall note with specificity the particular provisions of the amendment to which such Lenders object). To the extent the Replacement Rate is

approved by the Administrative Agent in connection with this clause (ii), the Replacement Rate shall be applied in a manner consistent with market practice.

4.4. <u>Funding Indemnification</u>. If any payment of a LIBOR Borrowing occurs on a date which is not the last day of the applicable Interest Period, whether because of acceleration, prepayment or otherwise, or a LIBOR Borrowing is not made, continued or converted on the date specified by the Borrower for any reason other than default by one or more of the Lenders, the Borrower will indemnify each Lender for any loss or cost incurred by such Lender, as applicable, resulting therefrom, including, without limitation, any loss or cost in liquidating or employing deposits acquired to fund or maintain the LIBOR Borrowing.

4.5. <u>Taxes</u>.

- (i) All payments by the Borrower to or for the account of any Lender or the Administrative Agent hereunder or under any Note shall be made free and clear of and without deduction for any and all Taxes. If the Borrower shall be required by law to deduct any Taxes from or in respect of any sum payable hereunder to any Lender or the Administrative Agent, (a) the sum payable shall be increased as necessary so that after making all required deductions (including deductions applicable to additional sums payable under this Section 4.5) such Lender or the Administrative Agent (as the case may be) receives an amount equal to the sum it would have received had no such deductions been made, (b) the Borrower shall make such deductions, (c) the Borrower shall pay the full amount deducted to the relevant authority in accordance with Applicable Law, and (d) the Borrower shall furnish to the Administrative Agent the original copy of a receipt evidencing payment thereof within 30 days after such payment is made.
- (ii) In addition, the Borrower hereby agrees to pay any present or future stamp or documentary taxes and any other excise or property taxes, charges or similar levies which arise from any payment made hereunder or under any Note or from the execution or delivery of, or otherwise with respect to, this Agreement or any Note ("Other Taxes").
- (iii) The Borrower hereby agrees to indemnify the Administrative Agent and each Lender for the full amount of Taxes or Other Taxes (including, without limitation, any Taxes or Other Taxes imposed on amounts payable under this Section 4.5) paid by the Administrative Agent or such Lender as a result of its Term Loan Commitment, any Loans made by it hereunder, or otherwise in connection with its participation in this Agreement and any liability (including penalties, interest and expenses) arising therefrom or with respect thereto. Payments due under this indemnification shall be made within 30 days of the date the Administrative Agent or such Lender makes demand therefor pursuant to Section 4.6.
- (iv) Each Lender that is not incorporated under the laws of the United States of America or a state thereof (each a "Non-U.S. Lender") agrees that it will, not more than ten (10) Business Days after the date of this Agreement, (i) deliver

to the Administrative Agent two duly completed copies of United States Internal Revenue Service Form W-8BEN, W-8BEN-E or W-8ECI, certifying in either case that such Lender is entitled to receive payments under this Agreement without deduction or withholding of any United States federal income taxes, and (ii) deliver to the Administrative Agent a United States Internal Revenue Form W-8 or W-9, as the case may be, and certify that it is entitled to an exemption from United States backup withholding tax. Each Non-U.S. Lender further undertakes to deliver to each of the Borrower and the Administrative Agent (x) renewals or additional copies of such form (or any successor form) on or before the date that such form expires or becomes obsolete, and (y) after the occurrence of any event requiring a change in the most recent forms so delivered by it, such additional forms or amendments thereto as may be reasonably requested by the Borrower or the Administrative Agent. All forms or amendments described in the preceding sentence shall certify that such Lender is entitled to receive payments under this Agreement without deduction or withholding of any United States federal income taxes, unless an event (including without limitation any change in treaty, law or regulation) has occurred prior to the date on which any such delivery would otherwise be required which renders all such forms inapplicable or which would prevent such Lender from duly completing and delivering any such form or amendment with respect to it and such Lender advises the Borrower and the Administrative Agent that it is not capable of receiving payments without any deduction or withholding of United States federal income tax.

- (v) For any period during which a Non-U.S. Lender has failed to provide the Borrower with an appropriate form pursuant to clause (iv), above (unless such failure is due to a change in treaty, law or regulation, or any change in the interpretation or administration thereof by any governmental authority, occurring subsequent to the date on which a form originally was required to be provided), such Non-U.S. Lender shall not be entitled to indemnification under this Section 4.5 with respect to Taxes imposed by the United States; provided that, should a Non-U.S. Lender which is otherwise exempt from or subject to a reduced rate of withholding tax become subject to Taxes because of its failure to deliver a form required under clause (iv), above, the Borrower shall take such steps as such Non-U.S. Lender shall reasonably request to assist such Non-U.S. Lender to recover such Taxes.
- (vi) Any Lender that is entitled to an exemption from or reduction of withholding tax with respect to payments under this Agreement or any Note pursuant to the law of any relevant jurisdiction or any treaty shall deliver to the Borrower (with a copy to the Administrative Agent), at the time or times prescribed by Applicable Law, such properly completed and executed documentation prescribed by Applicable Law as will permit such payments to be made without withholding or at a reduced rate.
- (vii) If the U.S. Internal Revenue Service or any other governmental authority of the United States or any other country or any political subdivision thereof asserts a claim that the Administrative Agent did not properly withhold tax

from amounts paid to or for the account of any Lender (because the appropriate form was not delivered or properly completed, because such Lender failed to notify the Administrative Agent of a change in circumstances which rendered its exemption from withholding ineffective, or for any other reason), such Lender shall indemnify the Administrative Agent fully for all amounts paid, directly or indirectly, by the Administrative Agent as tax, withholding therefor, or otherwise, including penalties and interest, and including taxes imposed by any jurisdiction on amounts payable to the Administrative Agent under this subsection, together with all costs and expenses related thereto (including attorneys fees and time charges of attorneys for the Administrative Agent, which attorneys may be employees of the Administrative Agent). The obligations of the Lenders under this Section 4.5(vii) shall survive the payment of the Obligations and termination of this Agreement.

- (viii) If a payment made to a Lender hereunder or under any Note would be subject to United States federal withholding tax imposed by FATCA if such Lender were to fail to comply with the applicable reporting requirements of FATCA (including those contained in Section 1471(b) or 1472(b) of the Code, as applicable) such Lender shall deliver to the Borrower and the Administrative Agent at the time or times prescribed by law and at such time or times reasonably requested by the Borrower or the Administrative Agent such documentation prescribed by Applicable Law (including as prescribed by Section 1471(b)(3)(C)(i) of the Code) and such additional documentation reasonably requested by the Borrower or the Administrative Agent as may be necessary for the Borrower and the Administrative Agent to comply with their obligations under FATCA and to determine that such Lender has complied with such Lender's obligations under FATCA or to determine the amount to deduct and withhold from such payment. Solely for purposes of this Section 4.5(viii), "FATCA" shall include any amendments made to FATCA after the date of this Agreement.
- (ix) Each of the Lenders represents that as of the Agreement Execution Date it is not aware of any facts that would give rise to a claim for additional payments under this <u>Section 4.5</u>.
- 4.6. <u>Lender Statements</u>; <u>Survival of Indemnity</u>. To the extent reasonably possible, each Lender shall designate an alternate Lending Office with respect to its LIBOR Borrowings to reduce any liability of the Borrower to such Lender under <u>Sections 4.1</u>, <u>4.2</u> and <u>4.5</u> or to avoid the unavailability of a LIBOR Borrowing, so long as such designation is not disadvantageous to such Lender. Each Lender shall deliver a written statement of such Lender as to the amount due, if any, under <u>Sections 4.1</u>, <u>4.2</u>, <u>4.4</u> and <u>4.5</u> hereof. Such written statement shall set forth in reasonable detail the calculations upon which such Lender determined such amount and shall be final, conclusive and binding on the Borrower in the absence of manifest error. The amount due in such statement shall not include amounts due under <u>Section 4.5</u> that are either attributable to facts known to the Lender as of the Agreement Execution Date or that relate to a time period more than ninety (90) days prior to the giving of such written statement. Determination of amounts payable under such Sections in connection with a LIBOR Borrowing shall be calculated as though each Lender funded its LIBOR Borrowing through the purchase of a deposit of the

type and maturity corresponding to the deposit used as a reference in determining the Adjusted LIBOR Rate applicable to such Borrowing, whether in fact that is the case or not. Unless otherwise provided herein, the amount specified in the written statement shall be payable on demand after receipt by the Borrower of the written statement. The obligations of the Borrower under <u>Sections 4.1</u>, <u>4.2</u>, <u>4.4</u> and <u>4.5</u> hereof shall survive payment of the Obligations and termination of this Agreement.

4.7. Replacement of Lenders under Certain Circumstances. The Borrower shall be permitted to replace any Lender which (a) is not capable of receiving payments without any deduction or withholding of United States federal income tax pursuant to Section 4.5, or (b) cannot maintain its LIBOR Borrowings at a suitable Lending Office pursuant to Section 4.6, with a replacement bank or other financial institution or (c) becomes a Defaulting Lender; provided that (i) such replacement does not conflict with any applicable legal or regulatory requirements affecting the remaining Lenders, (ii) no Event of Default or (after notice thereof to the Borrower) no Default shall have occurred and be continuing at the time of such replacement, (iii) the replacement bank or institution shall purchase at par all Loans and the Borrower or the replacement bank or institution shall repay other amounts owing to such replaced Lender prior to the date of replacement, (iv) the Borrower shall be liable to such replaced Lender under Sections 4.4 and 4.6 if any LIBOR Borrowing owing to such replaced Lender shall be prepaid (or purchased) other than on the last day of the Interest Period relating thereto, (v) the replacement bank or institution, if not already a Lender, and the terms and conditions of such replacement, shall be reasonably satisfactory to the Administrative Agent, (vi) the replaced Lender shall be obligated to make such replacement in accordance with the provisions of Section 13.3 (provided that the Borrower shall be obligated to pay the processing fee referred to therein and the replaced Lender's consent shall not be required), (vii) until such time as such replacement shall be consummated, the Borrower shall continue to pay all amounts payable hereunder without setoff, deduction, counterclaim or withholding and (viii) any such replacement shall not be deemed to be a waiver of any rights which the Borrower, the Administrative Agent or any other Lender shall have against the replaced Lender.

ARTICLE V.

CONDITIONS PRECEDENT

- 5.1. <u>Conditions Precedent to Closing</u>. The Lenders shall not be required to make the initial Loans pursuant to Section 2.1 hereunder unless (i) the Borrower shall have paid all fees then due and payable to the Lenders, the Arrangers and the Administrative Agent hereunder, including reasonable legal fees and out-of-pocket expenses, (ii) all of the conditions set forth in <u>Section 5.2</u> are satisfied, and (iii) the Borrower shall have furnished to the Administrative Agent, in form and substance satisfactory to the Lenders and their counsel and with sufficient copies for the Lenders, the following:
 - (a) <u>Certificates of Limited Partnership/Incorporation</u>. A copy of the Certificate of Limited Partnership for the Borrower and a copy of the articles of incorporation of the General Partner, each certified by the appropriate Secretary of State or equivalent state official.

- (b) <u>Agreements of Limited Partnership/Bylaws</u>. A copy of the Agreement of Limited Partnership for the Borrower and a copy of the bylaws of the General Partner, including all amendments thereto, each certified by the Secretary or an Assistant Secretary of the General Partner as being in full force and effect on the Agreement Execution Date.
- (c) <u>Good Standing Certificates</u>. A certified copy of a certificate from the Secretary of State or equivalent state official of the states where the Borrower and the General Partner are organized, dated as of the most recent practicable date, showing the good standing or partnership qualification (if issued) of (i) the Borrower, and (ii) the General Partner.
- (d) <u>Foreign Qualification Certificates</u>. A certified copy of a certificate from the Secretary of State or equivalent state official of the state where the Borrower and the General Partner maintain their principal place of business, dated as of the most recent practicable date, showing the qualification to transact business in such state as a foreign limited partnership or foreign corporation, as the case may be, for (i) the Borrower, and (ii) the General Partner.
- (e) <u>Resolutions</u>. A copy of a resolution or resolutions adopted by the Board of Directors of the General Partner, certified by the Secretary or an Assistant Secretary of the General Partner as being in full force and effect on the Agreement Execution Date, authorizing the Borrowings provided for herein and the execution, delivery and performance of the Loan Documents by the General Partner to be executed and delivered by it hereunder on behalf of itself and the Borrower.
- (f) <u>Incumbency Certificate</u>. A certificate, signed by the Secretary or an Assistant Secretary of the General Partner and dated the Agreement Execution Date, as to the incumbency, and containing the specimen signature or signatures, of the Persons authorized to execute and deliver the Loan Documents to be executed and delivered by it and the Borrower hereunder.
- (g) <u>Loan Documents</u>. Originals of the Loan Documents (in such quantities as the Lenders may reasonably request), duly executed by authorized officers of the appropriate entity, including this Agreement duly executed by authorized officers of each Lender.
- (h) <u>Opinion of Borrower's Counsel</u>. A written opinion, dated the Agreement Execution Date, from outside counsel for the Borrower which counsel is reasonably satisfactory to Administrative Agent, substantially in the form attached hereto as <u>Exhibit E</u>.
- (i) <u>Opinion of General Partner's Counsel</u>. A written opinion, dated the Agreement Execution Date, from outside counsel for the General Partner which counsel is reasonably satisfactory to Administrative Agent, substantially in the form attached hereto as <u>Exhibit F</u>.

(j) <u>Financial and Related Information</u>. The following information:

- (i) The most recent financial statements of the Borrower and the General Partner and a certificate from a Qualified Officer of the Borrower that no change in the Borrower's financial condition that would have a Material Adverse Effect has occurred since December 31, 2014; and
- (ii) A Disbursement Instruction Agreement, in substantially the form of <u>Exhibit C</u> hereto, addressed to the Administrative Agent and signed by a Qualified Officer, together with such other related money transfer authorizations as the Administrative Agent may have reasonably requested.
- (k) <u>Closing Certificate</u>. A certificate, signed by an officer of the Borrower, stating that on the Agreement Execution Date no Default or Event of Default has occurred and is continuing and that all representations and warranties of the Borrower contained herein are true and correct as of the Agreement Execution Date as and to the extent set forth herein.
- (l) <u>Compliance Certificate</u>. A certificate substantially in the form of <u>Exhibit H</u>, signed by an officer of the Borrower certifying as to compliance with the financial covenants set forth in <u>Section 9.7</u> on a pro-forma basis on the Agreement Execution Date using the financial results as of the last day of the most recently completed calendar quarter and giving effect to the incurrence of the Loans, which certificate shall include calculations in reasonable detail demonstrating such compliance.
- (m) Other Evidence as any Lender May Require. Such other evidence as any Lender may reasonably request to establish the consummation of the transactions contemplated hereby, the taking of all necessary actions in any proceedings in connection herewith and compliance with the conditions set forth in this Agreement.

When all such conditions have been fulfilled (or, in the Lenders' sole discretion, waived by Lenders), the Administrative Agent shall confirm in writing to the Borrower that the initial Borrowings are then available to the Borrower hereunder.

- 5.2. <u>Conditions Precedent to Subsequent Borrowings</u>. Borrowings after the initial Borrowing (other than conversions or continuations) shall be made from time to time as requested by the Borrower, and the obligation of each Lender to make any Borrowing is subject to the following terms and conditions:
 - (a) prior to and after giving effect to each such Borrowing, no Default or Event of Default shall have occurred and be continuing under this Agreement or any of the Loan Documents, and the Borrower shall deliver a certificate of the Borrower to such effect; and
 - (b) The representations and warranties contained in <u>Article VI</u> and <u>VII</u> are true and correct as of such borrowing date, as and to the extent set forth therein, except to the extent any such representation or warranty is stated to relate solely to an earlier date, in which case such representation or warranty shall be true and correct on and as of such earlier date.

Subject to the last grammatical paragraphs of <u>Article VI</u> and <u>VII</u> hereof, each Borrowing Notice shall constitute a representation and warranty by the Borrower that the conditions contained in <u>Sections 5.2(a)</u> and <u>(b)</u> have been satisfied. Unless set forth in writing to the contrary, the making of its initial Loan or Incremental Loan by a Lender shall constitute a certification by such Lender to the Administrative Agent and the other Lenders that the conditions precedent for initial Loans set forth in <u>Section 5.1</u> and <u>Section 5.2</u> (if applicable) that have not previously been waived by the Lenders in accordance with the terms of this Agreement have been satisfied.

ARTICLE VI.

REPRESENTATIONS AND WARRANTIES

The Borrower hereby represents and warrants to the Lenders that:

- 6.1. Existence. The Borrower is a limited partnership duly organized and existing under the laws of the State of Delaware, with its principal place of business in the State of Illinois, and is duly qualified as a foreign limited partnership, properly licensed (if required), in good standing and has all requisite authority to conduct its business in each jurisdiction in which it owns Properties and, except where the failure to be so qualified or to obtain such authority would not have a Material Adverse Effect, in each other jurisdiction in which its business is conducted. Each of its Subsidiaries is duly organized, validly existing and in good standing under the laws of its jurisdiction of organization and has all requisite authority to conduct its business in each jurisdiction in which it owns Property, and except where the failure to be so qualified or to obtain such authority would not have a Material Adverse Effect, in each other jurisdiction in which it conducts business.
- 6.2. <u>Corporate/Partnership Powers</u>. The execution, delivery and performance of the Loan Documents required to be delivered by the Borrower hereunder are within the partnership authority of such entity and the corporate powers of the general partners of such entity, have been duly authorized by all requisite action, and are not in conflict with the terms of any organizational instruments of such entity, or any instrument or agreement to which the Borrower or the General Partner is a party or by which the Borrower, the General Partner or any of their respective assets may be bound or affected.
- 6.3. <u>Power of Officers</u>. The officers of the General Partner executing the Loan Documents required to be delivered by such entities hereunder have been duly elected or appointed and were fully authorized to execute the same at the time each such agreement, certificate or instrument was executed.
- 6.4. <u>Government and Other Approvals</u>. No approval, consent, exemption or other action by, or notice to or filing with, any governmental authority is necessary in connection with the execution, delivery or performance of the Loan Documents required hereunder.

6.5. <u>Solvency</u>.

(i) Immediately after the Agreement Execution Date and immediately following the making of each Loan and after giving effect to the application of the

proceeds of such Loans, (a) the fair value of the assets of the Borrower and its Subsidiaries on a consolidated basis, at a fair valuation, will exceed the debts and liabilities, subordinated, contingent or otherwise, of the Borrower and its Subsidiaries on a consolidated basis; (b) the present fair saleable value of the Properties of the Borrower and its Subsidiaries on a consolidated basis will be greater than the amount that will be required to pay the probable liability of the Borrower and its Subsidiaries on a consolidated basis on their debts and other liabilities, subordinated, contingent or otherwise, as such debts and other liabilities become absolute and matured; (c) the Borrower and its Subsidiaries on a consolidated basis will be able to pay their debts and liabilities, subordinated, contingent or otherwise, as such debts and liabilities become absolute and matured; and (d) the Borrower and its Subsidiaries on a consolidated basis will not have unreasonably small capital with which to conduct the businesses in which they are engaged as such businesses are now conducted and are proposed to be conducted after the date hereof.

- (ii) The Borrower does not intend to, or to permit any of its Subsidiaries to incur debts beyond its ability to pay such debts as they mature, taking into account the timing of and amounts of cash to be received by it or any such Subsidiary and the timing of the amounts of cash to be payable on or in respect of its Indebtedness or the Indebtedness of any such Subsidiary.
- 6.6. <u>Compliance With Laws</u>. There is no judgment, decree or order or any law, rule or regulation of any court or governmental authority binding on the Borrower or any of its Subsidiaries which would be contravened by the execution, delivery or performance of the Loan Documents required hereunder.
- 6.7. <u>Enforceability of Agreement</u>. This Agreement is the legal, valid and binding agreement of the Borrower, and the Notes when executed and delivered will be the legal, valid and binding obligations of the Borrower, enforceable against the Borrower in accordance with their respective terms, and the Loan Documents required hereunder, when executed and delivered, will be similarly legal, valid, binding and enforceable except to the extent that such enforcement may be limited by applicable bankruptcy, insolvency, reorganization or other similar laws affecting the rights of creditors generally.
- 6.8. <u>Title to Property</u>. To the best of the Borrower's knowledge after due inquiry, the Borrower or its Subsidiaries has good and marketable title to the Properties and assets reflected in the financial statements as owned by it or any such Subsidiary free and clear of Liens except for the Permitted Liens. The execution, delivery or performance of the Loan Documents required to be delivered by the Borrower hereunder will not result in the creation of any Lien on the Properties. No consent to the transactions contemplated hereunder is required from any ground lessor or mortgagee or beneficiary under a deed of trust or any other party except as has been delivered to the Lenders.
- 6.9. <u>Litigation</u>. There are no suits, arbitrations, claims, disputes or other proceedings (including, without limitation, any civil, criminal, administrative or environmental proceedings), pending or, to the best of the Borrower's knowledge, threatened against or affecting the Borrower

or any of the Properties, the adverse determination of which individually or in the aggregate would have a Material Adverse Effect on the Borrower and/or would cause a Material Adverse Financial Change of the Borrower or materially impair the Borrower's ability to perform its obligations hereunder or under any instrument or agreement required hereunder, except as disclosed on <u>Schedule 6.9</u> hereto, or otherwise disclosed to Lenders in accordance with the terms hereof.

- 6.10. <u>Events of Default</u>. No Default or Event of Default has occurred and is continuing or would result from the incurring of obligations by the Borrower under any of the Loan Documents or any other document to which the Borrower is a party.
- 6.11. <u>Investment Company Act of 1940</u>. The Borrower is not and will by such acts as may be necessary continue not to be, an investment company within the meaning of the Investment Company Act of 1940.
- 6.12. <u>Public Utility Holding Company Act</u>. The Borrower is not a "holding company" or a "subsidiary company" of a "holding company," or an "affiliate" of a "holding company," or of a "subsidiary company" of a "holding company," within the definitions of the Public Utility Holding Company Act of 1935, as amended.
- 6.13. <u>Regulation U</u>. The proceeds of the Borrowings will not be used, directly or indirectly, in a manner which would cause the Facility to be treated as a "Purpose Credit."
- 6.14. <u>No Material Adverse Financial Change</u>. To the best knowledge of the Borrower, there has been no Material Adverse Financial Change in the condition of the Borrower since the date of the financial and/or operating statements most recently submitted to the Lenders December 31, 2020.
- 6.15. <u>Financial Information</u>. All financial statements furnished to the Lenders by or at the direction of the Borrower and all other financial information and data furnished by the Borrower to the Lenders are complete and correct in all material respects as of the date thereof, and such financial statements have been prepared in accordance with GAAP and fairly present the consolidated financial condition and results of operations of the Borrower as of such date. The Borrower has no contingent obligations, liabilities for taxes or other outstanding financial obligations which are material in the aggregate, except as disclosed in such statements, information and data.
- 6.16. <u>Factual Information</u>. All factual information heretofore or contemporaneously furnished by or on behalf of the Borrower to the Lenders for purposes of or in connection with this Agreement and the other Loan Documents and the transactions contemplated therein is, and all other such factual information hereafter furnished by or on behalf of the Borrower to the Lenders will be, true and accurate (taken as a whole) in all material respects on the date as of which such information is dated or certified and not incomplete by omitting to state any material fact necessary to make such information (taken as a whole) not misleading at such time.
- 6.17. <u>ERISA</u>. (i) The Borrower is not an entity deemed to hold "plan assets" within the meaning of ERISA or any regulations promulgated thereunder of an employee benefit plan (as defined in Section 3(3) of ERISA) which is subject to Title I of ERISA or any plan within the meaning of Section 4975 of the Code, and (ii) the execution of this Agreement and the transactions

contemplated hereunder do not give rise to a prohibited transaction within the meaning of Section 406 of ERISA or Section 4975 of the Code.

- 6.18. <u>Taxes</u>. All required Federal, state and other material tax returns have been filed by the Borrower with the appropriate authorities except (i) to the extent that extensions of time to file have been requested, granted and have not expired and (ii) to the extent such taxes are being contested in good faith and for which adequate reserves, in accordance with GAAP, are being maintained.
- 6.19. <u>Environmental Matters</u>. Except as disclosed in <u>Schedule 6.19</u>, each of the following representations and warranties is true and correct except to the extent that the facts and circumstances giving rise to any such failure to be so true and correct, in the aggregate, could not reasonably be expected to have a Material Adverse Effect:
 - (i) To the knowledge of the Borrower, the Properties of the Borrower, its Subsidiaries, and Investment Affiliates do not contain any Materials of Environmental Concern in amounts or concentrations which constitute a violation of, or could reasonably give rise to liability under, Environmental Laws.
 - (ii) The Borrower has not received any written notice alleging that any or all of the Properties of the Borrower and its Subsidiaries and Investment Affiliates and all operations at the Properties are not currently in compliance with all applicable Environmental Laws. Further, the Borrower has not received any written notice alleging the current existence of any contamination at or under such Properties in amounts or concentrations which constitute a violation of any Environmental Law, or any violation of any Environmental Law with respect to such Properties for which the Borrower, its Subsidiaries or Investment Affiliates is or could be liable.
 - (iii) Neither the Borrower nor any of its Subsidiaries or Investment Affiliates has received any written notice of current non-compliance, liability or potential liability regarding Environmental Laws with regard to any of the Properties, nor does it have knowledge that any such notice will be received or is being threatened.
 - (iv) To the knowledge of the Borrower during the ownership of the Properties by any or all of the Borrower, its Subsidiaries and Investment Affiliates, Materials of Environmental Concern have not been transported or disposed of from the Properties of the Borrower and its Subsidiaries and Investment Affiliates in violation of, or in a manner or to a location which could reasonably give rise to liability of the Borrower, any Subsidiary, or any Investment Affiliate under, Environmental Laws, nor during the ownership of the Properties by any or all of the Borrower, its Subsidiaries and Investment Affiliates have any Materials of Environmental Concern been generated, treated, stored or disposed of at, on or under any of such Properties in violation of, or in a manner that could give rise to liability of the Borrower, any Subsidiary or any Investment Affiliate under, any applicable Environmental Laws.

- (v) No judicial proceedings or governmental or administrative action is pending, or, to the knowledge of the Borrower, threatened, under any Environmental Law to which the Borrower, any of its Subsidiaries, or any Investment Affiliate, is named as a party with respect to the Properties of such entity, nor are there any consent decrees or other decrees, consent orders, administrative order or other orders, or other administrative or judicial requirements outstanding under any Environmental Law with respect to such Properties for which the Borrower, its Subsidiaries, or any Investment Affiliate is or could be liable.
- (vi) To the knowledge of the Borrower during the ownership of the Properties by any or all of the Borrower, its Subsidiaries and Investment Affiliates, there has been no release or threat of release of Materials of Environmental Concern at or from the Properties of the Borrower and its Subsidiaries and Investment Affiliates, or arising from or related to the operations of such entity in connection with the Properties in violation of or in amounts or in a manner that could give rise to liability under Environmental Laws.
- 6.20. <u>Insurance</u>. The Borrower maintains insurance on its properties consistent with the insurance maintained by other institutional owners of similar properties.
- 6.21. No Brokers. The Borrower has dealt with no brokers in connection with this Facility, and no brokerage fees or commissions are payable by or to any Person in connection with this Agreement or the Borrowings. Lenders shall not be responsible for the payment of any fees or commissions to any broker and the Borrower shall indemnify, defend and hold Lenders harmless from and against any claims, liabilities, obligations, damages, costs and expenses (including reasonable attorneys' fees and disbursements) made against or incurred by Lenders as a result of claims made or actions instituted by any broker or Person claiming by, through or under the Borrower in connection with the Facility.
- 6.22. <u>No Violation of Usury Laws</u>. No aspect of any of the transactions contemplated herein violate or will violate any usury laws or laws regarding the validity of agreements to pay interest in effect on the date hereof.
- 6.23. <u>Not a Foreign Person</u>. The Borrower is not a "foreign person" within the meaning of Section 1445 or 7701 of the Internal Revenue Code.
- 6.24. <u>No Trade Name</u>. Except for the name "First Industrial," and except as otherwise set forth on <u>Schedule 6.24</u> attached hereto, the Borrower does not use any trade name and has not and does not do business under any name other than their actual names set forth herein. The principal place of business of the Borrower is as stated in the recitals hereto.
- 6.25. <u>Subsidiaries</u>. <u>Schedule 6.25</u> hereto contains an accurate list of all of the presently existing Subsidiaries of the Borrower, setting forth the percentage of their respective Capital Stock owned by it or its Subsidiaries. All of the issued and outstanding shares of Capital Stock of such Subsidiaries have been duly authorized and issued and are fully paid and nonassessable.

- 6.26. <u>Unencumbered Assets</u>. <u>Schedule 6.26</u> hereto contains a complete and accurate description of Unencumbered Assets as of September 30, 2017 and as supplemented from time to time including the entity that owns each Unencumbered Asset. With respect to each Project identified from time to time as an Unencumbered Asset, the Borrower hereby represents and warrants as follows except to the extent disclosed in writing to the Lenders and approved by the Required Lenders (which approval shall not be unreasonably withheld) or except to the extent the failure of such representation and warranty to be true would not materially adversely affect the use and operation of such Project for its intended use or its marketability or value:
 - (a) No portion of any improvement on the Unencumbered Asset is located in an area identified by the Secretary of Housing and Urban Development or any successor thereto as an area having special flood hazards pursuant to the National Flood Insurance Act of 1968 or the Flood Disaster Protection Act of 1973, as amended, or any successor law, or, if located within any such area, the Borrower has obtained and will maintain the insurance prescribed in Section 6.20 hereof.
 - (b) To the Borrower's knowledge, the Unencumbered Asset and the present use and occupancy thereof are in material compliance with all Applicable Laws (including all Environmental Laws).
 - (c) The Unencumbered Asset is served by all utilities required for the current or contemplated use thereof. All utility service is provided by public utilities and the Unencumbered Asset has accepted or is equipped to accept such utility service.
 - (d) All public roads and streets necessary for service of and access to the Unencumbered Asset for the current or contemplated use thereof have been completed, are serviceable and all-weather and are physically and legally open for use by the public.
 - (e) The Unencumbered Asset is served by public water and sewer systems or, if the Unencumbered Asset is not serviced by a public water and sewer system, such alternate systems are adequate and meet, in all material respects, all requirements and regulations of, and otherwise complies in all material respects with, all Applicable Laws with respect to such alternate systems.
 - (f) The Borrower is not aware of any latent or patent structural or other significant deficiency of the Unencumbered Asset. The Unencumbered Asset is free of damage and waste that would materially and adversely affect the value of the Unencumbered Asset, is in good repair and there is no deferred maintenance other than ordinary wear and tear. The Unencumbered Asset is free from damage caused by fire or other casualty. There is no pending or, to the actual knowledge of the Borrower threatened condemnation proceedings affecting the Unencumbered Asset, or any material part thereof.
 - (g) To the Borrower's knowledge, all liquid and solid waste disposal, septic and sewer systems located on the Unencumbered Asset are in a good and safe condition and repair and to the Borrower's knowledge, in material compliance with all Applicable Laws with respect to such systems.

- (h) All improvements on the Unencumbered Asset lie within the boundaries and building restrictions of the legal description of record of the Unencumbered Asset, no such improvements encroach upon easements benefiting the Unencumbered Asset other than encroachments that do not materially adversely affect the use or occupancy of the Unencumbered Asset and no improvements on adjoining properties encroach upon the Unencumbered Asset or easements benefiting the Unencumbered Asset other than encroachments that do not materially adversely affect the use or occupancy of the Unencumbered Asset. All amenities, access routes or other items that materially benefit the Unencumbered Asset are under direct control of the Borrower, constitute permanent easements that benefit all or part of the Unencumbered Asset or are public property, and the Unencumbered Asset, by virtue of such easements or otherwise, is contiguous to a physically open, dedicated all weather public street, and has the necessary permits for ingress and egress.
- (i) There are no delinquent taxes, ground rents, water charges, sewer rents, assessments, insurance premiums, leasehold payments, or other outstanding charges affecting the Unencumbered Asset except to the extent such items are being contested in good faith and as to which adequate reserves have been provided.
- (j) The Unencumbered Asset satisfies each of the requirements for an Unencumbered Asset as set forth in the definition thereof.

A breach of any of the representations and warranties contained in this <u>Section 6.26</u> with respect to a Project shall disqualify such Project from being an Unencumbered Asset for so long as such breach continues (unless otherwise approved by the Required Lenders) but shall not constitute a Default (unless the elimination of such Property as an Unencumbered Asset results in a Default under one of the other provisions of this Agreement).

6.27. <u>Anti-Corruption Laws and Sanctions</u>.

None of the Borrower or any of its Subsidiaries, nor, to the knowledge of the Borrower, any of the directors, officers, employees or agents of the Borrower or of any of its Subsidiaries (i) is a Sanctioned Person, or (ii) to the knowledge of the Borrower, except as may be disclosed by the Borrower to the Administrative Agent from time to time, derives any of its assets or operating income from investments in or transactions with any Sanctioned Person. The Borrower and its Subsidiaries and, to the knowledge of the Borrower, their respective officers, directors, employees and agents, are in compliance with Anti-Corruption Laws and applicable Sanctions in all material respects. No Loan, use of the proceeds of any Loan, or other transactions contemplated hereby will violate (x) Anti-Corruption Laws or (y) any Sanctions applicable to any party hereto. Neither the making of the Loans nor the use of the proceeds thereof will violate the Patriot Act, the Trading with the Enemy Act, as amended, or any of the foreign assets control regulations of the United States Treasury Department (31 C.F.R., Subtitle B, Chapter V, as amended) or any enabling legislation or executive order relating thereto or successor statute thereto. The Borrower and its Subsidiaries are in compliance in all material respects with the Patriot Act.

The Borrower agrees that all of its representations and warranties set forth in <u>Article VI</u> of this Agreement and elsewhere in this Agreement are true as of the effective date of the

Second Third Amendment to Unsecured Term Loan Agreement, and will be true on each Effective Date in all material respects (except with respect to matters which have been disclosed in writing to and approved by the Required Lenders), and will be true in all material respects (except with respect to matters which have been disclosed in writing to and approved by the Required Lenders) upon the date of each request for disbursement of a Borrowing, provided that the Borrower shall only be obligated to update any Schedules referred to in this Article VI and the financial statements required under Section 8.2(i) on a quarterly basis, unless any change otherwise required to be disclosed could reasonably be expected to have a Material Adverse Effect. Each request for disbursement hereunder shall constitute a reaffirmation of such representations and warranties as deemed modified in accordance with the disclosures made and approved, as aforesaid, as of the date of such request and disbursement.

6.28. **EEA**Affected Financial Institutions 6.29. No Loan Party is an **EEA**Affected Financial Institution.

ARTICLE VII.

ADDITIONAL REPRESENTATIONS AND WARRANTIES

The General Partner hereby represents and warrants to the Lenders that:

- 7.1. Existence. The General Partner is a corporation duly organized and existing under the laws of the State of Maryland, with its principal place of business in the State of Illinois, is duly qualified as a foreign corporation and properly licensed (if required) and in good standing in each jurisdiction where the failure to qualify or be licensed (if required) would constitute a Material Adverse Financial Change with respect to the General Partner or have a Material Adverse Effect on the business or properties of the General Partner.
- 7.2. <u>Corporate Powers</u>. The execution, delivery and performance of the Loan Documents required to be delivered by the General Partner hereunder are within the corporate powers of the General Partner, have been duly authorized by all requisite corporate action, and are not in conflict with the terms of any organizational instruments of the General Partner, or any instrument or agreement to which the General Partner is a party or by which the General Partner or any of its assets is bound or affected.
- 7.3. <u>Power of Officers</u>. The officers of the General Partner executing the Loan Documents required to be delivered by the General Partner hereunder have been duly elected or appointed and were fully authorized to execute the same at the time each such agreement, certificate or instrument was executed.
- 7.4. <u>Government and Other Approvals</u>. No approval, consent, exemption or other action by, or notice to or filing with, any governmental authority is necessary in connection with the execution, delivery or performance of the Loan Documents required hereunder.
- 7.5. <u>Compliance With Laws</u>. There is no judgment, decree or order or any law, rule or regulation of any court or governmental authority binding on the General Partner which would be contravened by the execution, delivery or performance of the Loan Documents required hereunder.

- 7.6. <u>Enforceability of Agreement</u>. This Agreement is the legal, valid and binding agreement of the General Partner, as the general partner of the Borrower, enforceable against the General Partner in accordance with its respective terms, and the Loan Documents required hereunder, when executed and delivered, will be similarly legal, valid, binding and enforceable except to the extent that such enforcement may be limited by applicable bankruptcy, insolvency, reorganization or other similar laws affecting the rights of creditors generally.
- 7.7. <u>Liens; Consents</u>. The execution, delivery or performance of the Loan Documents required to be delivered by the General Partner hereunder will not result in the creation of any Lien on the Properties other than in favor of the Lenders. No consent to the transactions hereunder is required from any ground lessor or mortgagee or beneficiary under a deed of trust or any other party except as has been delivered to the Lenders.
- 7.8. <u>Litigation</u>. There are no suits, arbitrations, claims, disputes or other proceedings (including, without limitation, any civil, criminal, administrative or environmental proceedings), pending or, to the best of the General Partner's knowledge, threatened against or affecting the General Partner or any of the Properties, the adverse determination of which individually or in the aggregate would have a Material Adverse Effect on the General Partner and/or would cause a Material Adverse Financial Change with respect to the General Partner or materially impair the General Partner's ability to perform its obligations hereunder or under any instrument or agreement required hereunder, except as disclosed on <u>Schedule 7.8</u> hereto, or otherwise disclosed to Lenders in accordance with the terms hereof.
- 7.9. <u>Events of Default</u>. No Default or Event of Default has occurred and is continuing or would result from the incurring of obligations by the General Partner under any of the Loan Documents or any other document to which the General Partner is a party.
- 7.10. <u>Investment Company Act of 1940</u>. The General Partner is not, and will by such acts as may be necessary continue not to be, an investment company within the meaning of the Investment Company Act of 1940.
- 7.11. <u>Public Utility Holding Company Act</u>. The General Partner is not a "holding company" or a "subsidiary company" of a "holding company," or an "affiliate" of a "holding company," or of a "subsidiary company" of a "holding company," within the definitions of the Public Utility Holding Company Act of 1935, as amended.
- 7.12. <u>No Material Adverse Financial Change</u>. There has been no Material Adverse Financial Change in the condition of the General Partner since the last date on which the financial and/or operating statements were submitted to the Lenders December 31, 2020.
- 7.13. <u>Financial Information</u>. All financial statements furnished to the Lenders by or on behalf of the General Partner and all other financial information and data furnished by or on behalf of the General Partner to the Lenders are complete and correct in all material respects as of the date thereof, and such financial statements have been prepared in accordance with GAAP and fairly present the consolidated financial condition and results of operations of the General Partner as of such date. The General Partner has no contingent obligations, liabilities for taxes or other

outstanding financial obligations which are material in the aggregate, except as disclosed in such statements, information and data.

- 7.14. <u>Factual Information</u>. All factual information heretofore or contemporaneously furnished by or on behalf of the General Partner to the Lenders for purposes of or in connection with this Agreement and the other Loan Documents and the transactions contemplated therein is, and all other such factual information hereafter furnished by or on behalf of the General Partner to the Lenders will be, true and accurate in all material respects (taken as a whole) on the date as of which such information is dated or certified and not incomplete by omitting to state any material fact necessary to make such information (taken as a whole) not misleading at such time.
- 7.15. ERISA. (i) The General Partner is not an entity deemed to hold "plan assets" within the meaning of ERISA or any regulations promulgated thereunder of an employee benefit plan (as defined in Section 3(3) of ERISA) which is subject to Title I of ERISA or any plan within the meaning of Section 4975 of the Code, and (ii) the execution of this Agreement and the transactions contemplated hereunder do not give rise to a prohibited transaction within the meaning of Section 406 of ERISA or Section 4975 of the Code.
- 7.16. <u>Taxes</u>. All required Federal, state and other material tax returns have been filed by the General Partner with the appropriate authorities except (i) to the extent that extensions of time to file have been requested, granted and have not expired and (ii) to the extent such taxes are being contested in good faith and for which adequate reserves, in accordance with GAAP, are being maintained.
- 7.17. No Brokers. The General Partner has dealt with no brokers in connection with this Facility, and no brokerage fees or commissions are payable by or to any Person in connection with this Agreement or the Borrowings. Lenders shall not be responsible for the payment of any fees or commissions to any broker and the General Partner shall indemnify, defend and hold Lender harmless from and against any claims, liabilities, obligations, damages, costs and expenses (including reasonable attorneys' fees and disbursements) made against or incurred by Lender as a result of claims made or actions instituted by any broker or Person claiming by, through or under the General Partner in connection with the Facility.
- 7.18. <u>Subsidiaries</u>. <u>Schedule 7.18</u> hereto contains an accurate list of all of the presently existing Subsidiaries of the General Partner, setting forth their respective jurisdictions of formation, the percentage of their respective Capital Stock owned by it or its Subsidiaries and the Properties owned by them. All of the issued and outstanding shares of Capital Stock of such Subsidiaries have been duly authorized and issued and are fully paid and non-assessable.
- 7.19. <u>Status</u>. General Partner is a corporation listed and in good standing on a national securities exchange and is currently qualified as a real estate investment trust under the Code.
- 7.20. <u>Anti-Corruption Laws and Sanctions</u>. None of the General Partner, the Borrower, any of the other Subsidiaries of the General Partner, or, to the knowledge of the General Partner, any other Affiliate of the General Partner or any of their respective directors, officers, employees or agents: (i) is a Sanctioned Person; or (ii) to the knowledge of the General Partner, except as may be disclosed by the General Partner to the Administrative Agent from time to time, derives any of

its assets or operating income from investments in or transactions with any Sanctioned Person. The General Partner and its Affiliates and, to the knowledge of the General Partner, their respective officers, directors, employees and agents, are in compliance with Anti-Corruption Laws and applicable Sanctions in all material respects. No Loan, use of the proceeds of any Loan, or other transactions contemplated hereby will violate (x) Anti-Corruption Laws or (y) any Sanctions applicable to any party hereto. Neither the making of the Loans nor the use of the proceeds thereof will violate the Patriot Act, the Trading with the Enemy Act, as amended, or any of the foreign assets control regulations of the United States Treasury Department (31 C.F.R., Subtitle B, Chapter V, as amended) or any enabling legislation or executive order relating thereto or successor statute thereto. The Borrower and its Subsidiaries are in compliance in all material respects with the Patriot Act.

The General Partner agrees that all of its representations and warranties set forth in <u>Article VII</u> of this Agreement and elsewhere in this Agreement are true as of the effective date of the <u>SecondThird</u> Amendment to Unsecured Term Loan Agreement, and will be true on each Effective Date in all material respects (except with respect to matters which have been disclosed in writing to and approved by the Required Lenders), and will be true in all material respects (except with respect to matters which have been disclosed in writing to and approved by the Required Lenders) upon each request for disbursement of a Borrowing, <u>provided</u> that the General Partner shall only be obligated to update any Schedules referred to in this <u>Article VII</u> and the financial statements required under <u>Section 8.2(i)</u> on a quarterly basis, unless any change otherwise required to be disclosed could reasonably be expected to have a Material Adverse Effect. Each request for disbursement hereunder shall constitute a reaffirmation of such representations and warranties as deemed modified in accordance with the disclosures made and approved, as aforesaid, as of the date of such request and disbursement.

ARTICLE VIII.

AFFIRMATIVE COVENANTS

The Borrower (and the General Partner, if expressly included in Sections contained in this Article) covenant and agree that so long as the Term Loan Commitment of any Lender shall remain available and until the full and final payment of all Obligations incurred under the Loan Documents each will:

- 8.1. Notices. Promptly give written notice to Administrative Agent (who will promptly send such notice to Lenders) of:
- (a) all litigation or arbitration proceedings affecting the Borrower, the General Partner or any Subsidiary where the amount claimed is \$25,000,000,000 or more;
- (b) any Default or Event of Default, specifying the nature and the period of existence thereof and what action has been taken or been proposed to be taken with respect thereto;

- (c) all claims filed against any property owned by the Borrower or the General Partner which, if adversely determined, could have a Material Adverse Effect on the ability of the Borrower or the General Partner to meet any of their obligations under the Loan Documents;
- (d) the occurrence of any other event which might have a Material Adverse Effect or cause a Material Adverse Financial Change on or with respect to the Borrower or the General Partner;
- (e) any Reportable Event or any "prohibited transaction" (as such term is defined in Section 4975 of the Code) in connection with any Plan or any trust created thereunder, which may, singly or in the aggregate materially impair the ability of the Borrower or the General Partner to repay any of its obligations under the Loan Documents, describing the nature of each such event and the action, if any, the Borrower or the General Partner, as the case may be, proposes to take with respect thereto;
- (f) any notice from any Federal, state, local or foreign authority regarding any Hazardous Material, asbestos, or other environmental condition, proceeding, order, claim or violation affecting any of the Properties where such condition, proceeding, order, claim or violation would reasonably be expected to result in a liability to the Borrower, the General Partner or any Subsidiary in an amount, individually or in the aggregate, of \$50,000,000 or more.
- 8.2. <u>Financial Statements, Reports, Etc.</u>. The Borrower and the General Partner each shall maintain, for itself and each Subsidiary, a system of accounting established and administered in accordance with GAAP, and shall furnish to the Administrative Agent (and the Administrative Agent shall thereafter promptly furnish to the Lenders):
 - (i) quarterly financial statements (including a balance sheet, income statement, and cash flow statement) and related reports in form and substance satisfactory to the Lenders not later than forty-five (45) days after the end of each of the first three fiscal quarters (commencing with the fiscal quarter ending September 30, 2015), and not later than ninety (90) days after the end of each fiscal year (commencing with the fiscal year ending December 31, 2015), annual audited financial statements, audited by an accounting firm as reasonably approved by Administrative Agent (which audit report shall be without a "going concern" or like qualification or exception), provided, however, Administrative Agent shall only have the right to approve such accounting firm if the accounting firm is not a big 4 accounting firm, all certified by the Borrower's or the General Partner's, as applicable, chief financial officer, chief accounting officer or controller, calculation of the financial covenants described below, a description of Unencumbered Assets, a listing of capital expenditures (in the level of detail as currently disclosed in the Borrower's "Supplemental Information"), a report listing and describing all newly acquired Properties, including their cash flow, cost and secured or unsecured Indebtedness assumed in connection with such acquisition, if any, summary Property information for all Properties, including, without limitation, their Property Operating Income, occupancy rates, square footage, property type and date

acquired or built, and such other information as may be requested to evaluate the quarterly compliance certificate delivered as provided below;

- (ii) copies of all Form 10-Ks, 10-Qs, 8-Ks, and any other public information filed with the Securities Exchange Commission by the Borrower or the General Partner once a quarter simultaneously with delivering the compliance certificate described below, along with any other materials distributed to the shareholders of the General Partner or the partners of the Borrower from time to time, including a copy of the General Partner's annual report; provided that, to the extent any of such reports contains information required under the other subsections of this Section 8.2, the information need not be furnished separately under the other subsections; provided, further that the Borrower may comply with this Section 8.2(ii) by posting or causing to be posted, the foregoing information on either the Securities Exchange Commission public website or on the Borrower's or the General Partner's public website, so long as the Lenders have access to and are timely referred to any such website by the Borrower;
- (iii) not later than forty-five (45) days after the end of the first three fiscal quarters (commencing with the fiscal quarter ending September 30, 2015), and not later than ninety (90) days after the end of the fiscal year (commencing with the fiscal year ending December 31, 2015), a report certified by the entity's chief financial officer, chief accounting officer or controller, containing Property Operating Income from individual properties owned by the Borrower or a Wholly-Owned Subsidiary and included as Unencumbered Assets.
- (iv) Not later than forty-five (45) days after the end of each of the first three fiscal quarters (commencing with the fiscal quarter ending September 30, 2015), and not later than ninety (90) days after the end of the fiscal year (commencing with the fiscal year ending December 31, 2015), a compliance certificate in substantially the form of Exhibit H hereto signed by the Borrower's chief financial officer, chief accounting officer or controller confirming that the Borrower is in compliance with all of the covenants of the Loan Documents, showing the calculations and computations necessary to determine compliance with the financial covenants contained in this Agreement (including such schedules and backup information as may be necessary to demonstrate such compliance) and stating that to such officer's best knowledge, there is no other Default or Event of Default exists, or if any Default or Event of Default exists, stating the nature and status thereof;
- (v) As soon as possible and in any event within ten (10) Business Days after the Borrower knows that any Reportable Event has occurred with respect to any Plan, a statement, signed by the chief financial officer of the Borrower, describing said Reportable Event and within twenty (20) days after such Reportable Event, a statement signed by such chief financial officer describing the action which the Borrower proposes to take with respect thereto; and (b) within ten (10) Business Days of receipt, any notice from the Internal Revenue Service, PBGC or Department of Labor with respect to a Plan regarding any excise tax, proposed

termination of a Plan, prohibited transaction or fiduciary violation under ERISA or the Code which could result in any liability to the Borrower or any member of the Controlled Group in excess of \$100,000; and (c) within ten (10) Business Days of filing, any Form 5500 filed by the Borrower with respect to a Plan, or any member of the Controlled Group which includes a qualified accountant's opinion;

- (vi) As soon as possible and in any event within thirty (30) days after receipt by the Borrower, a copy of (a) any notice or claim to the effect that the Borrower or any of its Subsidiaries is or may be liable to any Person as a result of the release by such entity, or any of its Subsidiaries, or any other Person of any toxic or hazardous waste or substance into the environment, and (b) any notice alleging any violation of any federal, state or local environmental, health or safety law or regulation by the Borrower or any of its Subsidiaries or Investment Affiliates, which, in either case, could be reasonably likely to have a Material Adverse Effect;
- (vii) Not later than ninety (90) days after the end of each fiscal year, a balance sheet and cash flow projection for the current fiscal year based on future plans, expectations and strategies of the Borrower and the General Partner; provided that all of the parties hereto acknowledge and agree that the foregoing shall (A) be based on certain assumptions of the Borrower; (B) only reflect the Borrower's outlook as of the date that the Borrower delivers the same; and (C) not be deemed to be a prediction of any results or the actual effect of future plans or strategies of the Borrower; [Reserved];
- (Viii) Promptly upon the furnishing thereof to the shareholders of the Borrower, copies of all financial statements, reports and proxy statements so furnished; provided, further that the Borrower may comply with this <u>Section 8.2(viii)</u> by posting or having posted the foregoing information on either the Securities Exchange Commission public website or on the Borrower's or the General Partner's public website, so long as the Lenders have access to and are timely referred to any such website by the Borrower;
- (ix) Promptly upon the distribution thereof to the press or the public, copies of all press releases; provided, further that the Borrower may comply with this <u>Section 8.2(ix)</u> by posting or having posted the foregoing information on either the Securities Exchange Commission public website or on the Borrower's or the General Partner's public website, so long as the Lenders have access to and are timely referred to any such website by the Borrower;
- (x) Promptly upon receipt thereof, notices with respect to the ratings for the Borrower's or the General Partner's long-term, senior unsecured debt, the effect of which may be to change the Base Rate Applicable Margin and/or the LIBOR Applicable Margin;
- (Xi) As soon as possible, and in any event within ten (10) days after the Borrower knows of any fire or other casualty or any pending or threatened

condemnation or eminent domain proceeding with respect to all or any material portion of any Unencumbered Asset, a statement signed by the Chief Financial Officer of the Borrower, describing such fire, casualty or condemnation and the action the Borrower intends to take with respect thereto; and

- (Xii) Such other information (including, without limitation, nonfinancial information) as the Administrative Agent or any Lender may from time to time reasonably request.
- 8.3. Existence and Conduct of Operations. Except as permitted herein, maintain and preserve its existence and all rights, privileges and franchises now enjoyed and necessary for the operation of its business, including remaining in good standing in each jurisdiction in which business is currently operated. The Borrower and the General Partner shall carry on and conduct their respective businesses in substantially the same manner and in substantially the same fields of enterprise as presently conducted. The Borrower will do, and will cause each of its Subsidiaries to do, all things necessary to remain duly incorporated and/or duly qualified, validly existing and in good standing as a real estate investment trust, corporation, general partnership, limited liability company or limited partnership, as the case may be, in its jurisdiction of incorporation/formation, except and to the extent that the Borrower determines that causing any of its Subsidiaries (other than any Loan Party) to remain duly incorporated and/or duly qualified, validly existing and in good standing is not in the best interests of the Borrower and such failure to remain duly incorporated and/or qualified, validly existing and in good standing could not reasonably be expected to have a Material Adverse Effect. The Borrower will maintain all requisite authority to conduct its business in each jurisdiction in which the Properties are located and, except where the failure to be so qualified would not have a Material Adverse Effect, in each jurisdiction required to carry on and conduct its businesses in substantially the same manner as it is presently conducted, and, specifically, neither the Borrower nor its Subsidiaries will undertake any business other than the acquisition, development, ownership, management, operation and leasing of industrial/warehouse properties and business activities ancillary, incidental or otherwise reasonably related thereto.
- 8.4. <u>Maintenance of Properties</u>. Maintain, preserve, protect and keep the Properties in good repair, working order and condition, and make all necessary and proper repairs, renewals and replacements, normal wear and tear excepted.
- 8.5. <u>Insurance</u>. Upon request of the Administrative Agent, provide a certificate of insurance from all insurance carriers who maintain policies with respect to the Properties within thirty (30) days after the end of each fiscal year, evidencing that the insurance required to be furnished to Lenders pursuant to <u>Section 6.20</u> hereof is in full force and effect. The Borrower shall timely pay, or cause to be paid, all premiums on all insurance policies required under this Agreement from time to time. The Borrower shall promptly notify its insurance carrier or agent therefor (with a copy of such notification being provided simultaneously to Administrative Agent) if there is any occurrence which, under the terms of any insurance policy then in effect with respect to the Properties, requires such notification.
- 8.6. <u>Payment of Obligations</u>. Pay and discharge when due (i) all Federal, state and other material taxes, assessments, governmental charges and (ii) all other obligations which, if unpaid,

would by law become a Lien upon its property, except, in the case of each of clauses (i) and (ii), such as may be contested in good faith or as to which a bona fide dispute may exist, and for which adequate reserves have been provided in accordance with sound accounting principles used by the Borrower on the date hereof.

- 8.7. <u>Compliance with Laws</u>. Comply in all material respects with all Applicable Laws, rules, regulations, orders and directions (including, without limitation, Anti-Corruption Laws and Sanctions) of any governmental authority having jurisdiction over the Borrower, the General Partner, or any of their respective businesses.
- 8.8. <u>Adequate Books</u>. Maintain adequate books, accounts and records in order to provide financial statements in accordance with GAAP and, if requested by any Lender, permit employees or representatives of such Lender at any reasonable time and upon reasonable notice to inspect and audit the properties of the Borrower and of the Consolidated Operating Partnership, and to examine or audit the inventory, books, accounts and records of each of them and make copies and memoranda thereof.
 - 8.9. ERISA. Comply in all material respects with all requirements of ERISA applicable to it with respect to each Plan.
- 8.10. <u>Maintenance of Status</u>. The General Partner shall at all times (i) remain as a corporation listed and in good standing on a national securities exchange, and (ii) take all steps to maintain the General Partner's status as a real estate investment trust in compliance with all applicable provisions of the Code.
- 8.11. <u>Use of Proceeds</u>. Use the proceeds of the Facility for the general business purposes of the Borrower, including without limitation working capital needs, closing costs, interim funding for property acquisitions and construction of new industrial properties, and/or payment of other debts and obligations of the Borrower.
- 8.12. <u>Distributions</u>. Provided there is no declared Monetary Default then existing and provided there is not an Event of Default relating to a breach of the financial covenants contained in <u>Section 9.7</u> below, the Borrower may make distributions to the General Partner and the General Partner may make distributions to its shareholders. Notwithstanding the foregoing, the Borrower shall be permitted to make distributions to the General Partner and the General Partner shall be permitted at all times to make distributions of whatever amount is necessary to maintain its tax status as a real estate investment trust.
- 8.13. <u>Guarantors</u>. To the extent any Person executes and delivers a guaranty of, or otherwise becomes obligated in respect of, the Indebtedness of the Parent or the Borrower under the Existing Revolving Credit Agreement, the Borrower shall, within five (5) Business Days (or such longer period as the Administrative Agent may reasonably determine) cause such Person to deliver to the Administrative Agent each of the following in form and substance satisfactory to the Administrative Agent: (a) a Guaranty executed by such Person substantially similar to the form of guaranty delivered by such Person with respect to the Existing Revolving Credit Agreement and (b) to the extent reasonably requested by the Administrative Agent, supporting documentation

comparable in scope and type to any supporting documentation delivered by such Person with respect to the Existing Revolving Credit Agreement.

ARTICLE IX.

NEGATIVE COVENANTS

The Borrower covenants and agrees that, so long as the Term Loan Commitment shall remain available and until full and final payment of all Obligations incurred under the Loan Documents, without the prior written consent of either all of the Lenders pursuant to Section 14.13(b)(iii) or the consent of the Required Lenders in all other cases, it will not, and the General Partner will not and, in the case of Sections 9.9, the Borrower's Subsidiaries will not:

- 9.1. <u>Change in Business</u>. Engage in any business if, as a result, the general nature of the business in which the Borrower and the General Partner, taken as a whole, would then be engaged would be substantially changed from the general nature of the business in which the Borrower and the General Partner, taken as a whole, are engaged on the Closing Date.
- 9.2. <u>Change of Management of Properties</u>. Change the management of the Properties, except that (a) any Affiliate of the Borrower or the General Partner shall be permitted to manage any of the Properties and (b) the Borrower, the General Partner or any of their Subsidiaries may engage and/or replace third-party property managers with respect to any Property, provided that the aggregate book value of all Properties at any one time under third-party management shall not exceed 30% of the aggregate book value of all of the Properties at such time.
- 9.3. <u>Change of Borrower Ownership</u>. Without the consent of the Required Lenders, allow (i) the General Partner to own less than fifty-one percent (51%) of the partnership interests in the Borrower, (ii) the Borrower to be controlled by a Person other than the General Partner, (iii) a Change in Control to occur, or (iv) any pledge of, other encumbrance on, or conversion to limited partnership interests of, any of the general partnership interests in the Borrower. If any of the foregoing occurs with the consent of the Required Lenders, each non-consenting Lender shall have the option to demand repayment of any outstanding Obligations due to such Lender prior to or concurrent with the occurrence of any of the foregoing events. Upon any Lender's demand in accordance with the foregoing provision, the Borrower shall repay any outstanding Obligations due to such Lender prior to or concurrently with the occurrence of any of the foregoing events.
- 9.4. <u>Use of Proceeds</u>. Apply or permit to be applied any proceeds of any Borrowing directly or indirectly, to the funding of any purchase of, or offer for, any share of capital stock of any publicly held corporation unless the board of directors of such corporation has consented to such offer prior to any public announcements relating thereto and all of the Lenders have consented to such use of the proceeds of the Facility. The Borrower shall not request any Loan, shall not use, and shall ensure that its Subsidiaries and its or their respective directors, officers, employees and agents do not use the proceeds of any Loan (i) in furtherance of an offer, payment, promise to pay, or authorization of the payment or giving of money, or anything else of value, to any Person in violation of any Anti-Corruption Laws, (ii) for the purpose of funding, financing or facilitating any activities, business or transaction of or with any Sanctioned Person to the extent such activities, business or transaction would be prohibited by Sanctions if conducted by a corporation

incorporated in the United States or in any European Union member state or (iii) in any manner that would result in the violation of any Sanctions applicable to any party hereto.

- 9.5. <u>Liens</u>. Create, incur, or suffer to exist (or permit any of its Subsidiaries to create, incur, or suffer to exist) any Lien in, of or on the Property of any member of the Consolidated Operating Partnership other than:
 - (i) Liens for taxes, assessments or governmental charges or levies on their Property if the same shall not at the time be delinquent or thereafter can be paid without penalty, or are being contested in good faith and by appropriate proceedings and for which adequate reserves shall have been set aside on their books;
 - (ii) Liens which arise by operation of law, such as carriers', warehousemen's, landlords', materialmen and mechanics' liens and other similar liens arising in the ordinary course of business which secure payment of obligations not more than thirty (30) days past due or which are being contested in good faith by appropriate proceedings and for which adequate reserves shall have been set aside on its books;
 - (iii) Liens arising out of pledges or deposits under worker's compensation laws, unemployment insurance, old age pensions, or other social security or retirement benefits, or similar legislation;
 - (iv) Utility easements, building restrictions, zoning restrictions, easements and such other encumbrances or charges against real property as are of a nature generally existing with respect to properties of a similar character and which do not in any material way affect the marketability of the same or interfere with the use thereof in the business of the Borrower or its Subsidiaries;
 - (v) Liens of any Subsidiary in favor of the Borrower or the General Partner; and
 - (vi) So long as immediately prior to the creation, assumption or incurring of such Lien, and immediately thereafter, no Default or Event of Default would be in existence, Liens on Properties securing Indebtedness not prohibited hereunder.

Liens permitted pursuant to this <u>Section 9.5</u> shall be deemed to be "Permitted Liens".

9.6. <u>Regulation U</u>. Use the proceeds of the Loans for any other purpose than as provided for in <u>Section 8.11</u> and <u>Section 9.4</u> hereof. No part of the proceeds of any Loan will be used, whether directly or indirectly for any purpose that entails a violation of any of the regulations of the Board of Governors of the Federal Reserve System, including Regulations T, U and X. The proceeds will not be used in a manner which would cause the Facility to be treated as a "Purpose Credit."

9.7. <u>Indebtedness and Cash Flow Covenants</u>. Permit or suffer:

- (a) as of the last day of any fiscal quarter, the ratio of (A) the sum of (1) EBITDA of the Consolidated Operating Partnership plus (2) interest income (other than any interest income from assets being used to support Defeased Debt) to (B) the sum of (1) Debt Service plus, without duplication, (2) all payments on account of preferred stock or preferred partnership units of any member of the Consolidated Operating Partnership for such quarter plus (3) all Ground Lease Payments due from any member of the Consolidated Operating Partnership to the extent not deducted as an expense in calculating EBITDA of the Consolidated Operating Partnership (the "Fixed Charge Coverage Ratio"), to be less than (i) 1.50 to 1.0 from the Agreement Execution Date through the Maturity Date, with all such calculations in clauses (A) and (B) above based on the results of the four (4) consecutive fiscal quarters then ended;
- (b) as of the last day of any fiscal quarter, the Consolidated Leverage Ratio to exceed sixty percent (60%); provided that, if any Material Acquisition shall occur then, at the election of the Borrower upon delivery of prior written notice to the Administrative Agent, concurrently with or prior to the delivery of a Compliance Certificate pursuant to Section 8.2 (iv), and provided that no Default or Event of Default shall have occurred and be continuing (other than as a result of the Consolidated Leverage Ratio as of the end of such fiscal quarter being greater than sixty percent (60%) but less than or equal to sixty-five percent (65%)), the Consolidated Leverage Ratio shall be increased to sixty-five percent (65%) for such fiscal quarter and the next three succeeding fiscal quarterquarters (any period with such increase a "Consolidated Leverage Ratio Increase Period"), provided that (i) no more than two (2) Consolidated Leverage Ratio Increase Periods may be elected by the Borrower during the term of this Agreement and (ii) any such Consolidated Leverage Ratio Increase Periods shall be non-consecutive;
- (c) as of the last day of any fiscal quarter, the ratio of outstanding Consolidated Senior Unsecured Debt to Value of Unencumbered Assets (such ratio, the "<u>Unencumbered Leverage Ratio</u>") to exceed sixty percent (60%); <u>provided</u> that, if any Material Acquisition shall occur then, at the election of the Borrower upon delivery of prior written notice to the Administrative Agent, concurrently with or prior to the delivery of a Compliance Certificate pursuant to <u>Section 8.2 (iv)</u>, and provided that no Default or Event of Default shall have occurred and be continuing (other than as a result of the Unencumbered Leverage Ratio as of the end of such fiscal quarter being greater than sixty percent (60%) but less than or equal to sixty-five percent (65%)), the Unencumbered Leverage Ratio shall be increased to sixty-five percent (65%) for such fiscal quarter and the next <u>three</u> succeeding fiscal <u>quarterguarters</u> (any period with such increase, an "<u>Unencumbered Leverage Ratio Increase Period</u>"), <u>provided</u> further that (i) no more than two (2) Unencumbered Leverage Ratio Increase Periods may be elected by the Borrower during the term of this Agreement and (ii) any such Unencumbered Leverage Ratio Increase Periods shall be non-consecutive;
- (d) as of the last day of any fiscal quarter, Consolidated Secured Debt to exceed 40% of Implied Capitalization Value of the Consolidated Operating Partnership; or

(e) as of the last day of any fiscal quarter, the ratio of (A) Property Operating Income from Unencumbered Assets that are not Assets Under Development for such fiscal quarter to (B) interest expense on all Consolidated Senior Unsecured Debt for such fiscal quarter to be less than (x) 1.75 to 1.0 from the Agreement Execution Date through the Maturity Date.

The foregoing covenants set forth in paragraphs (c) and (e) above shall be tested at the end of each fiscal quarter (for the applicable reporting period), on a pro-forma basis at the time of each Borrowing under the Facility (using the latest quarterly financial statements then available and taking into account the proposed Borrowing), and on a pro-forma basis upon any Asset Sale or incurrence of any Indebtedness by the Consolidated Operating Partnership (using the latest quarterly financial statements then available and taking into account the proposed incurrence of Indebtedness). To the extent the Consolidated Operating Partnership has Defeased Debt, both the underlying debt and interest payable thereon and the financial assets used to defease such debt and interest earned thereon shall be excluded from calculations of the foregoing financial covenants. All financial computations required under this Section 9.7 shall be made in accordance with GAAP as in effect on the Agreement Execution Date.

- 9.8. <u>Change of Control; Mergers and Dispositions</u>. Enter into any merger, consolidation, reorganization, <u>division</u> or liquidation or transfer or otherwise dispose of all or a substantial portion of its properties, except for: such transactions that occur between wholly owned Subsidiaries; transactions where the Borrower and the General Partner are the surviving entities and there is no change in business conducted, loss of an Investment Grade Rating or Change in Control, and no Default or Event of Default under the Loan Documents results from such transaction; or as otherwise approved in advance by the Lenders. To the extent not reported on General Partner's Form 8-K filings, the Borrower will notify the Administrative Agent (who will promptly notify Lenders) of any single event acquisition (whether individually or consummated as a series of related acquisitions pursuant to documentation closed concurrently), disposition, merger, <u>division</u> or asset purchase involving assets valued in excess of 25% of the Consolidated Operating Partnership's then current Market Value Net Worth and certify compliance with covenants after giving effect to such proposed acquisition, disposition, merger, <u>division</u> or asset purchase.
- 9.9. <u>Negative Pledge</u>. The Borrower agrees that throughout the term of this Facility, no "negative pledge" on any Project then included in Unencumbered Assets restricting the Borrower's (or wholly owned Subsidiary's) right to sell or encumber such Project shall be given to any other lender or creditor other than a Permitted Negative Pledge or, if such a "negative pledge" is given, the Project affected shall be immediately excluded from Unencumbered Assets.

ARTICLE X.

DEFAULTS

The occurrence of any one or more of the following events shall constitute an Event of Default:

10.1. <u>Nonpayment of Principal</u>. The Borrower fails to pay any principal portion of the Obligations when due, whether on the Maturity Date or otherwise.

- 10.2. <u>Certain Covenants</u>. The Borrower, the General Partner and/or Consolidated Operating Partnership, as the case may be, is not in compliance with any one or more of <u>Sections 8.10</u>, <u>8.12</u>, <u>9.3</u>, <u>9.4</u>, <u>9.5</u>, <u>9.7</u>, <u>9.8</u>, or <u>9.9</u> hereof.
- 10.3. <u>Nonpayment of Interest and Other Obligations</u>. The Borrower fails to pay any interest or other portion of the Obligations, other than payments of principal, and such failure continues for a period of five (5) days after the date such payment is due.
- 10.4. <u>Cross Default</u>. Any monetary default occurs (after giving effect to any applicable cure period) under any other Indebtedness (which includes liability under Guaranties) of the Borrower or the General Partner, singly or in the aggregate, in excess of Seventy-Five Million Dollars (\$75,000,000), other than (i) Indebtedness arising from the purchase of personal property or the provision of services, the amount of which is being contested by the Borrower or (ii) Indebtedness which is "non recourse", i.e., which is not recoverable by the creditor thereof from the general assets of the Borrower, the General Partner or any of their Affiliates, but is limited to the proceeds of certain real estate, improvements and related personal property.
- 10.5. <u>Loan Documents</u>. Any Loan Document is not in full force and effect or a default has occurred and is continuing thereunder after giving effect to any cure or grace period in any such document.
- 10.6. <u>Representation or Warranty</u>. At any time or times hereafter any representation or warranty set forth in <u>Articles VI</u> or <u>VII</u> of this Agreement or in any other Loan Document or in any statement, report or certificate now or hereafter made by the Borrower or the General Partner to the Lenders or the Administrative Agent is not true and correct in any material respect.
- 10.7. <u>Covenants, Agreements and Other Conditions</u>. The Borrower or the General Partner fails to perform or observe any of the other covenants, agreements and conditions contained in <u>Articles VIII</u> and <u>IX</u> (except for <u>Sections 8.10, 8.12, 9.3, 9.4, 9.5, 9.7, 9.8, or 9.9 hereof) and elsewhere in this Agreement or any of the other Loan Documents in accordance with the terms hereof or thereof, not specifically referred to herein, and such Default continues unremedied for a period of thirty (30) days after written notice from Administrative Agent, <u>provided, however</u>, that if such Default is susceptible of cure but cannot by the use of reasonable efforts be cured within such thirty (30) day period, such Default shall not constitute an Event of Default under this <u>Section 10.7</u> so long as (i) the Borrower or the General Partner, as the case may be, has commenced a cure within such thirty day period and (ii) thereafter, the Borrower or the General Partner, as the case may be, is proceeding to cure such default continuously and diligently and in a manner reasonably satisfactory to Lenders and (iii) such default is cured not later than sixty (60) days after the expiration of such thirty (30) day period.</u>
 - 10.8. No Longer General Partner. The General Partner shall no longer be the sole general partner of the Borrower.
- 10.9. <u>Material Adverse Financial Change</u>. The Borrower or the General Partner has suffered a Material Adverse Financial Change or is Insolvent.

10.10. Bankruptcy.

- (a) The General Partner, the Borrower or one or more Subsidiaries in which the Equity Value of the Subsidiary or Subsidiaries is more than ten percent (10%), in the aggregate, of Implied Capitalization Value of the Consolidated Operating Partnership shall (i) have an order for relief entered with respect to it under the Federal bankruptcy laws as now or hereafter in effect, (ii) make an assignment for the benefit of creditors, (iii) apply for, seek, consent to, or acquiesce in, the appointment of a receiver, custodian, trustee, examiner, liquidator or similar official for it or any substantial portion of its Property, (iv) institute any proceeding seeking an order for relief under the Federal bankruptcy laws as now or hereafter in effect or seeking to adjudicate it as a bankrupt or insolvent, or seeking dissolution, winding up, liquidation, reorganization, arrangement, adjustment or composition of it or its debts under any law relating to bankruptcy, insolvency or reorganization or relief of debtors or fail to file an answer or other pleading denying the material allegations of any such proceeding filed against it, (v) take any corporate action to authorize or effect any of the foregoing actions set forth in this Section 10.10(a), (vi) fail to contest in good faith any appointment or proceeding described in Section 10.10(b) or (vii) not pay, or admit in writing its inability to pay, its debts generally as they become due. As used herein, the term "Equity Value" of a Subsidiary shall mean the lesser of: (1) Property Operating Income of such Subsidiary's Properties owned as of the Agreement Execution Date capitalized at a 6.5% rate, less any Indebtedness of such Subsidiary; or (2) the appraised value of any of such Subsidiary's Properties, less any Indebtedness of such Subsidiary.
- (b) A receiver, trustee, examiner, liquidator or similar official shall be appointed for (i) the General Partner, the Borrower or one or more Subsidiaries in which the Equity Value of the Subsidiary or Subsidiaries is more than ten percent (10%), in the aggregate, of Implied Capitalization Value of the Consolidated Operating Partnership or (ii) all or substantially all of the Consolidated Operating Partnership's Properties.
- (c) A proceeding described in <u>Section 10.10(a)(iv)</u> shall be instituted against the General Partner, the Borrower or one or more Subsidiaries in which the Equity Value of the Subsidiary or Subsidiaries is more than ten percent (10%), in the aggregate, of Implied Capitalization Value of the Consolidated Operating Partnership, and such proceeding continues undismissed or unstayed for a period of sixty (60) consecutive days.
- 10.11. <u>Legal Proceedings</u>. The Borrower or the General Partner is enjoined, restrained or in any way prevented by any court order or judgment or if a notice of lien, levy, or assessment is filed of record with respect to all or any part of the Properties by any governmental department, office or agency, which could materially adversely affect the performance of the obligations of such parties hereunder or under the Loan Documents, as the case may be, or if any proceeding is filed or commenced seeking to enjoin, restrain or in any way prevent the foregoing parties from conducting all or a substantial part of their respective business affairs and failure to vacate, stay, dismiss, set aside or remedy the same within ninety (90) days after the occurrence thereof.
- 10.12. <u>ERISA</u>. The Borrower or the General Partner is deemed to hold "plan assets" within the meaning of ERISA or any regulations promulgated thereunder of an employee benefit

plan (as defined in Section 3(3) of ERISA) which is subject to Title I of ERISA or any plan (within the meaning of Section 4975 of the Code).

- 10.13. <u>Change in Control</u>. A Change in Control has occurred.
- 10.14. Failure to Satisfy Judgments. The General Partner, the Borrower or any of its Subsidiaries shall fail within sixty (60) days to pay, bond or otherwise discharge any judgments or orders for the payment of money (other than with respect to any Indebtedness which is "non recourse", i.e., which is not recoverable by the creditor thereof from the general assets of the Borrower, the General Partner or any of their Affiliates, but is limited to the proceeds of certain real estate, improvements and related personal property) in an amount which, when added to all other judgments or orders outstanding against the General Partner, the Borrower or any Subsidiary would exceed Seventy-Five Million Dollars (\$75,000,000) in the aggregate, which have not been stayed on appeal or otherwise appropriately contested in good faith, unless the liability is insured against and the insurer has not challenged coverage of such liability.
- 10.15. Environmental Remediation. Failure to remediate (or pursue the remediation process with due diligence and good faith) within the time period required by law or governmental order, (or within a reasonable time in light of the nature of the problem if no specific time period is so established), environmental problems in violation of Applicable Law related to Properties of the Borrower and/or its Subsidiaries where the estimated cost of remediation is in the aggregate in excess of Seventy-Five Million Dollars (\$75,000,000), in each case after all administrative hearings and appeals have been concluded.

ARTICLE XI.

ACCELERATION, WAIVERS, AMENDMENTS AND REMEDIES

- 11.1. Acceleration. If any Event of Default described in Section 10.10 hereof occurs, or the Borrower or the General Partner becomes Insolvent, the Term Loan Commitments and obligation of the Lenders to make Borrowings hereunder shall automatically terminate and the Obligations shall immediately become due and payable without presentment, demand, protest or other notice of any kind, all of which are hereby expressly waived by the Borrower. If any other Event of Default described in Article X hereof occurs and is continuing, the Administrative Agent, with the consent of the Required Lenders, may, and at the request of the Required Lenders shall, by notice to the Borrower, take either or both of the following actions, at the same or different times: (i) terminate the Term Loan Commitments, and thereupon the Term Loan Commitments shall terminate immediately, and (ii) declare the Loans then outstanding to be due and payable in whole (or in part, in which case any principal not so declared to be due and payable may thereafter be declared to be due and payable), and thereupon the principal of the Loans so declared to be due and payable, together with accrued interest thereon and all fees and other Obligations of the Borrower accrued hereunder, shall become due and payable immediately, without presentment, demand, protest or other notice of any kind, all of which are hereby waived by the Borrower.
- 11.2. <u>Preservation of Rights; Amendments</u>. No delay or omission of the Lenders in exercising any right under the Loan Documents shall impair such right or be construed to be a

waiver of any Default or an acquiescence therein, and the making of a Borrowing notwithstanding the existence of a Default or the inability of the Borrower to satisfy the conditions precedent to such Borrowing shall not constitute any waiver or acquiescence. Any single or partial exercise of any such right shall not preclude other or further exercise thereof or the exercise of any other right, and no waiver, amendment or other variation of the terms, conditions or provisions of the Loan Documents whatsoever shall be valid unless in writing signed by the Administrative Agent and the number of Lenders required hereunder and then only to the extent in such writing specifically set forth. All remedies contained in the Loan Documents or by law afforded shall be cumulative and all shall be available to the Lenders until the Obligations have been paid in full.

ARTICLE XII.

THE ADMINISTRATIVE AGENT

- 12.1. Appointment. Wells Fargo Bank, National Association is hereby appointed by each of the Lenders as its contractual representative (herein referred to as the "Administrative Agent") hereunder and under each other Loan Document, and each of the Lenders irrevocably authorizes the Administrative Agent to act as the contractual representative of such Lender with the rights and duties expressly set forth herein and in the other Loan Documents. The Administrative Agent agrees to act as such contractual representative upon the express conditions contained in this Article XII. Notwithstanding the use of the defined term "Administrative Agent," it is expressly understood and agreed that the Administrative Agent shall not have any fiduciary responsibilities to any Lender by reason of this Agreement or any other Loan Document and that the Administrative Agent is merely acting as the contractual representative of the Lenders with only those duties as are expressly set forth in this Agreement and the other Loan Documents. In its capacity as the Lenders' contractual representative, the Administrative Agent (i) does not hereby assume any fiduciary duties to any of the Lenders, (ii) is a "representative" of the Lenders within the meaning of the term "secured party" as defined in the Illinois Uniform Commercial Code and (iii) is acting as an independent contractor, the rights and duties of which are limited to those expressly set forth in this Agreement and the other Loan Documents. Each of the Lenders hereby agrees to assert no claim against the Administrative Agent on any agency theory or any other theory of liability for breach of fiduciary duty, all of which claims each Lender hereby waives.
- 12.2. <u>Powers</u>. The Administrative Agent shall have and may exercise such powers under the Loan Documents as are specifically delegated to the Administrative Agent by the terms of each thereof, together with such powers as are reasonably incidental thereto. The Administrative Agent shall have no implied duties to the Lenders, or any obligation to the Lenders to take any action thereunder except any action specifically provided by the Loan Documents to be taken by the Administrative Agent.
- 12.3. <u>General Immunity</u>. Neither the Administrative Agent (in its capacity as Administrative Agent) nor any of its directors, officers, agents or employees shall be liable to the Borrower, the Lenders or any Lender for any action taken or omitted to be taken by it or them hereunder or under any other Loan Document or in connection herewith or therewith, except for

its or their own Gross Negligence or willful misconduct, as determined by a court of competent jurisdiction by a final and non-appealable judgment.

- 12.4. No Responsibility for Loans, Recitals, etc.. Neither the Administrative Agent (in its capacity as Administrative Agent) nor any of its directors, officers, agents or employees shall be responsible for or have any duty to ascertain, inquire into, or verify (i) any statement, warranty or representation made in connection with any Loan Document or any borrowing hereunder; (ii) the performance or observance of any of the covenants or agreements of any obligor under any Loan Document; (iii) the satisfaction of any condition specified in Article V, except receipt of items required to be delivered to the Administrative Agent; or (iv) the validity, effectiveness or genuineness of any Loan Document or any other instrument or writing furnished in connection therewith. Except as expressly set forth herein, the Administrative Agent shall not have any duty to disclose, and shall not be liable for the failure to disclose, any information relating to the Borrower or any of its Subsidiaries that is communicated to or obtained by the bank serving as Administrative Agent or any of its Affiliates in any capacity.
- 12.5. Action on Instructions of Lenders. The Administrative Agent shall in all cases be fully protected in acting, or in refraining from acting, hereunder and under any other Loan Document in accordance with written instructions signed by the Required Lenders or all Lenders, as the case may be, and such instructions and any action taken or failure to act pursuant thereto shall be binding on all of the Lenders and on all holders of Notes. The Administrative Agent shall be fully justified in failing or refusing to take any action hereunder and under any other Loan Document unless it shall be indemnified to its satisfaction by the Lenders pro rata against any and all liability, cost and expense that it may incur by reason of taking or continuing to take any such action.
- 12.6. Employment of Administrative Agents and Counsel. The Administrative Agent may execute any of its duties as Administrative Agent hereunder and under any other Loan Document by or through employees, agents, and attorneys in fact and shall not be answerable to the Lenders, except as to money or securities received by it or its authorized agents, for the default or misconduct of any such agents or attorneys in fact selected by it with reasonable care. The Administrative Agent shall be entitled to advice of counsel concerning all matters pertaining to the agency hereby created and its duties hereunder and under any other Loan Document. Without limitation of the extent of any liability any Lender may otherwise have pursuant to this Agreement, the Administrative Agent shall have no liability for any action taken or omitted in good faith as a result of advice of counsel.
- 12.7. <u>Reliance on Documents; Counsel</u>. The Administrative Agent shall be entitled to rely upon any Note, notice, consent, certificate, affidavit, letter, telegram, statement, paper or document believed by it to be genuine and correct and to have been signed or sent by the proper person or persons, and, in respect to legal matters, upon the opinion of outside counsel selected by the Administrative Agent.
- 12.8. <u>Administrative Agent's Reimbursement and Indemnification</u>. The Lenders agree to reimburse and indemnify the Administrative Agent ratably in accordance with their respective Percentages (i) for any amounts not reimbursed by the Borrower (and without limiting the obligation of the Borrower to do so) for which the Administrative Agent is entitled to

reimbursement by the Borrower under the Loan Documents, (ii) for any other reasonable expenses incurred by the Administrative Agent on behalf of the Lenders, in connection with the preparation, execution, delivery, administration and enforcement of the Loan Documents, if not paid by the Borrower (and without limiting the obligation of the Borrower to do so), and (iii) for any liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind and nature whatsoever which may be imposed on, incurred by or asserted against the Administrative Agent (in its capacity as Administrative Agent and not as a Lender) in any way relating to or arising out of the Loan Documents or any other document delivered in connection therewith or the transactions contemplated thereby, or the enforcement of any of the terms thereof or of any such other documents, <u>provided</u> that no Lender shall be liable for any of the foregoing to the extent they arise from the Gross Negligence or willful misconduct of the Administrative Agent as determined by a court of competent jurisdiction in a final, non-appealable judgment; <u>provided</u>, <u>however</u>, that no action taken in accordance with the directions of the Required Lenders (or all of the Lenders, if expressly required hereunder) shall be deemed to constitute Gross Negligence or willful misconduct for purposes of this Section. The agreements in this Section shall survive the payment of the Loans and all other amounts payable hereunder or under the other Loan Documents and the termination of this Agreement. If the Borrower shall reimburse the Administrative Agent for any indemnifiable amount pursuant to this Section, the Administrative Agent shall share such reimbursement on a ratable basis with each Lender making any such payment.

12.9. Rights as a Lender. With respect to its Term Loan Commitment, Borrowings made by it and any Note issued to it, the Administrative Agent shall have the same rights and powers hereunder and under any other Loan Document as any Lender and may exercise the same as though it were not the Administrative Agent, and the term "Lender" or "Lenders" shall, unless the context otherwise indicates, include the Administrative Agent in its individual capacity. The Administrative Agent, in its individual capacity, may accept deposits from, lend money to, and generally engage in any kind of trust, debt, equity or other transaction, in addition to those contemplated by this Agreement or any other Loan Document, with the Borrower or any of its Subsidiaries in which the Borrower or such Subsidiary is not restricted hereby from engaging with any other Person. Each of the Lenders acknowledges that the Administrative Agent's legal counsel in connection with the transactions contemplated by this Agreement is only acting as counsel to the Administrative Agent and is not acting as counsel to any Lender.

12.10. [<u>Reserved</u>].

12.11. <u>Lender Credit Decision</u>. Each Lender acknowledges that it has, independently and without reliance upon the Administrative Agent or any other Lender and based on the financial statements prepared by the Borrower and such other documents and information as it has deemed appropriate, made its own credit analysis and decision to enter into this Agreement and the other Loan Documents. Each Lender also acknowledges that it will, independently and without reliance upon the Administrative Agent or any other Lender and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Agreement and the other Loan Documents.

- 12.12. Successor Administrative Agent. The Administrative Agent may resign at any time as Administrative Agent under the Loan Documents by giving written notice thereof to the Lenders and the Borrower. Upon any such resignation, the Required Lenders shall have the right to appoint a successor Administrative Agent which appointment shall, provided no Default or Event of Default exists, be subject to the Borrower's approval, which approval shall not be unreasonably withheld or delayed (except that the Borrower shall, in all events, be deemed to have approved each Lender and any of its Affiliates as a successor Administrative Agent). If no successor Administrative Agent shall have been so appointed in accordance with the immediately preceding sentence, and shall have accepted such appointment, within 30 days after the current Administrative Agent's giving of notice of resignation, then the current Administrative Agent may, on behalf of the Lenders, appoint a successor Administrative Agent, which shall be a Lender, if any Lender shall be willing to serve, and otherwise shall be an Eligible Assignee. Upon the acceptance of any appointment as Administrative Agent hereunder by a successor Administrative Agent, such successor Administrative Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the current Administrative Agent, and the current Administrative Agent shall be discharged from its duties and obligations under the Loan Documents. After any Administrative Agent's resignation hereunder as Administrative Agent, the provisions of this Article XII. shall continue to inure to its benefit as to any actions taken or omitted to be taken by it while it was Administrative Agent under the Loan Documents. Notwithstanding anything contained herein to the contrary, the Administrative Agent may assign its rights and duties under the Loan Documents to any of its Affiliates by giving the Borrower and each Lender prior written notice.
- 12.13. Notice of Defaults. The Administrative Agent shall not be deemed to have knowledge or notice of the occurrence of a Default or Event of Default unless the Administrative Agent has received notice from a Lender or the Borrower referring to this Agreement, describing with reasonable specificity such Default or Event of Default and stating that such notice is a "notice of default." If a Lender becomes aware of a Default or Event of Default, such Lender shall notify the Administrative Agent of such fact. Upon receipt of such notice that a Default or Event of Default has occurred, the Administrative Agent shall notify each of the Lenders of such fact.
- 12.14. Requests for Approval. If the Administrative Agent requests in writing the consent or approval of a Lender, such Lender shall, or, in the case of any request for consent or approval that is subject to Section 14.13(a) or (b), shall use commercially reasonable efforts to, respond and either approve or disapprove definitively in writing to the Administrative Agent within ten (10) Business Days (or sooner if such notice specifies a shorter period, but in no event less than five (5) Business Days for responses based on Administrative Agent's good faith determination that circumstances exist warranting its request for an earlier response) after such written request from the Administrative Agent provided that the request for approval states the time by which a response is needed before approval is deemed given. Solely with respect to any request for consent or approval that is subject to Section 14.13(c), if the Lender does not so respond, that Lender shall be deemed to have approved the request. Upon request, the Administrative Agent shall notify the Lenders which Lenders, if any, failed to respond to a request for approval.

- 12.15. <u>Copies of Documents</u>. Administrative Agent shall promptly deliver to each of the Lenders copies of all notices of default and other formal notices sent or received and according to <u>Section 15.1</u> of this Agreement. Administrative Agent shall deliver to Lenders within fifteen (15) Business Days following receipt, copies of all financial statements, certificates and notices received regarding the General Partner's ratings except to the extent such items are required to be furnished directly to the Lenders by the Borrower hereunder. Within fifteen (15) Business Days after a request by a Lender to the Administrative Agent for other documents furnished to the Administrative Agent by the Borrower, the Administrative Agent shall provide copies of such documents to such Lender except where this Agreement obligates Administrative Agent to provide copies in a shorter period of time.
- 12.16. <u>Defaulting Lenders</u>. Notwithstanding any provision of this Agreement to the contrary, if any Lender becomes a Defaulting Lender, then to the extent permitted by Applicable Law, the following provisions shall apply for so long as such Lender is a Defaulting Lender:
- (a) <u>Waivers and Amendments</u>. Such Defaulting Lender's right to approve or disapprove any amendment, waiver or consent with respect to this Agreement shall be restricted as set forth in the definition of Required Lenders.
- (b) <u>Defaulting Lender Waterfall</u>. Any payment of principal, interest, fees or other amounts received by the Administrative Agent for the account of such Defaulting Lender (whether voluntary or mandatory, at maturity, pursuant to Section 11.1 or otherwise) or received by the Administrative Agent from a Defaulting Lender pursuant to Section 2.20(b) shall be applied at such time or times as may be determined by the Administrative Agent as follows: first, to the payment of any amounts owing by such Defaulting Lender to the Administrative Agent hereunder; second, [reserved]; third, [reserved]; fourth, as the Borrower may request (so long as no Default or Event of Default exists), to the funding of any Loan in respect of which such Defaulting Lender has failed to fund its portion thereof as required by this Agreement, as determined by the Administrative Agent; fifth, [reserved]; sixth, to the payment of any amounts owing to the Lenders as a result of any judgment of a court of competent jurisdiction obtained by any Lender against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; seventh, so long as no Default or Event of Default exists, to the payment of any amounts owing to the Borrower as a result of any judgment of a court of competent jurisdiction obtained by the Borrower against such Defaulting Lender as a result of such Defaulting Lender's breach of its obligations under this Agreement; and eighth, to such Defaulting Lender or as otherwise directed by a court of competent jurisdiction; provided that if such Loans were made at a time when the conditions set forth in Article IV were satisfied or waived, such payment shall be applied solely to pay the Loans of all Non-Defaulting Lenders on a pro rata basis prior to being applied to the payment of any Loans of such Defaulting Lender until such time as all Loans are held by the Lenders pro rata in accordance with their respective Percentages (determined without giving effect to the immediately following subsection (d)). Any payments, prepayments or other amounts paid or payable to a Defaulting Lender that are applied (or held) to pay amounts owed by a Defaulting Lender pursuant to this subsection shall be deemed paid to and redirected by such Defaulting Lender, and each Lender irrevocably consents hereto.
 - (c) [Reserved]

(d) [Reserved]

(e) [Reserved]

- (f) <u>Defaulting Lender Cure</u>. If the Borrower and the Administrative Agent agree in writing that a Lender is no longer a Defaulting Lender, the Administrative Agent will so notify the parties hereto, whereupon such Lender will cease to be a Defaulting Lender; <u>provided</u> that except to the extent otherwise expressly agreed by the affected parties, no change hereunder from Defaulting Lender to Lender will constitute a waiver or release of any claim of any party hereunder arising from that Lender's having been a Defaulting Lender.
- 12.17. <u>Delegation to Affiliates</u>. The Borrower and the Lenders agree that the Administrative Agent may delegate any of its duties under this Agreement to any of its Affiliates. Any such Affiliate (and such Affiliate's directors, officers, agents and employees) which performs duties in connection with this Agreement shall be entitled to the same benefits of the indemnification, waiver and other protective provisions to which the Administrative Agent is entitled under <u>Articles XII</u> and <u>XIV</u>.
- 12.18. <u>Managing Agents, Syndication Agent, etc.</u>. Neither any of the Lenders identified in this Agreement as a "managing agent" nor the Syndication Agent shall have any right, power, obligation, liability, responsibility or duty under this Agreement other than those applicable to all Lenders as such. Without limiting the foregoing, none of such Lenders shall have or be deemed to have a fiduciary relationship with any Lender. Each Lender hereby makes the same acknowledgments with respect to such Lenders as it makes with respect to the Administrative Agent in <u>Section 12.11</u>.

12.19. Erroneous Payments.

(a) Each Lender and any other party hereto hereby severally agrees that if (i) the Administrative Agent notifies (which such notice shall be conclusive absent manifest error) such Lender or any other Person that has received funds from the Administrative Agent or any of its Affiliates, either for its own account or on behalf of a Lender (each such recipient, a "Payment Recipient") that the Administrative Agent has determined in its sole discretion that any funds received by such Payment Recipient were erroneously transmitted to, or otherwise erroneously or mistakenly received by, such Payment Recipient (whether or not known to such Payment Recipient) or (ii) any Payment Recipient receives any payment from the Administrative Agent (or any of its Affiliates) (x) that is in a different amount than, or on a different date from, that specified in a notice of payment, prepayment or repayment sent by the Administrative Agent (or any of its Affiliates) with respect to such payment, prepayment or repayment, as applicable, (y) that was not preceded or accompanied by a notice of payment, prepayment or repayment, as applicable, or (z) that such Payment Recipient otherwise becomes aware was transmitted or received in error or by mistake (in whole or in part) then, in each case, an error in payment shall be presumed to have been made (any such amounts specified in clauses (i) or (ii) of this Section 12.19(a), whether received as a payment, prepayment or repayment of principal, interest, fees, distribution or otherwise; individually and collectively, an "Erroneous Payment"), then, in each case, such Payment Recipient is deemed to have knowledge of such error at the time

of its receipt of such Erroneous Payment; provided that nothing in this Section shall require the Administrative Agent to provide any of the notices specified in clauses (i) or (ii) above. Each Payment Recipient agrees that it shall not assert any right or claim to any Erroneous Payment, and hereby waives any claim, counterclaim, defense or right of set-off or recoupment with respect to any demand, claim or counterclaim by the Administrative Agent for the return of any Erroneous Payments, including without limitation waiver of any defense based on "discharge for value" or any similar doctrine.

(b) Without limiting the immediately preceding clause (a), each Payment Recipient agrees that, in the case of clause (a)(ii) above, it shall promptly notify the Administrative Agent in writing of such occurrence.

In the case of either clause (a)(i) or (a)(ii) above, such Erroneous Payment shall at all times remain the property of the Administrative Agent and shall be segregated by the Payment Recipient and held in trust for the benefit of the Administrative Agent, and upon demand from the Administrative Agent such Payment Recipient shall (or, shall cause any Person who received any portion of an Erroneous Payment on its behalf to), promptly, but in all events no later than two Business Days thereafter, return to the Administrative Agent the amount of any such Erroneous Payment (or portion thereof) as to which such a demand was made in same day funds and in the currency so received, together with interest thereon in respect of each day from and including the date such Erroneous Payment (or portion thereof) was received by such Payment Recipient to the date such amount is repaid to the Administrative Agent at the greater of the Federal Funds Effective Rate and a rate determined by the Administrative Agent in accordance with banking industry rules on interbank compensation from time to time in effect; provided that, if the Payment Recipient is the Borrower, (a) the accrual of such interest shall be at the Federal Funds Effective Rate and shall commence to accrue from and including the second Business Day following the date such Erroneous Payment (or portion thereof) was received by the Borrower (the "Borrower EP Interest Accrual Date") to the date such amount is repaid to the Administrative Agent and (b) if such Erroneous Payment (or portion thereof) shall have been returned to the Administrative Agent prior to the Borrower EP Interest Accrual Date, then the Borrower shall not have any further obligations or liability with respect to such Erroneous Payment, including for any interest, charges, fees, costs or expenses incurred as a result of the Administrative Agent making the Erroneous Payment to the Borrower, the Borrower holding the Erroneous Payment (or portion thereof) in trust for the benefit of the Administrative Agent and the Borrower returning the Erroneous Payment (or portion thereof) to the Administrative Agent.

(c) In the event that an Erroneous Payment (or portion thereof) is not recovered by the Administrative Agent for any reason, after demand therefor by the Administrative Agent in accordance with immediately preceding clause (c), from any Lender that is a Payment Recipient or an Affiliate of a Payment Recipient (such unrecovered amount as to such Lender, an "Erroneous Payment Return Deficiency"), then at the sole discretion of the Administrative Agent and upon the Administrative Agent's written notice to such Lender (i) such Lender shall be deemed to have made a cashless assignment of the full face amount of the portion of its Loans (but not its Term Loan Commitments) of the relevant Class with respect to which such Erroneous Payment was made (the "Erroneous Payment Impacted Class") to the Administrative Agent or, at the option of the Administrative Agent, the Administrative Agent's applicable lending affiliate in an amount that is equal to the Erroneous Payment Return Deficiency (or such lesser amount as the

Administrative Agent may specify) (such assignment of the Loans (but not Term Loan Commitments) of the Erroneous Payment Impacted Class, the "Erroneous Payment Deficiency Assignment") plus any accrued and unpaid interest on such assigned amount, without further consent or approval of any party hereto and without any payment by the Administrative Agent or its applicable lending affiliate as the assignee of such Erroneous Payment Deficiency Assignment. Without limitation of its rights hereunder, the Administrative Agent may cancel any Erroneous Payment Deficiency Assignment at any time by written notice to the applicable assigning Lender and upon such revocation all of the Loans assigned pursuant to such Erroneous Payment Deficiency Assignment shall be reassigned to such Lender without any requirement for payment or other consideration. The parties hereto acknowledge and agree that (1) any assignment contemplated in this clause (d) shall be made without any requirement for any payment or other consideration paid by the applicable assignee or received by the assignor, (2) the provisions of this clause (d) shall govern in the event of any conflict with the terms and conditions of Article XIII and (3) the Administrative Agent may reflect such assignments in the Register without further consent or action by any other Person.

- (d) Each party hereto hereby agrees that (x) in the event an Erroneous Payment (or portion thereof) is not recovered from any Payment Recipient that has received such Erroneous Payment (or portion thereof) for any reason, the Administrative Agent (1) shall be subrogated to all the rights of such Payment Recipient with respect to such amount and (2) is authorized to set off, net and apply any and all amounts at any time owing to such Payment Recipient under any Loan Document, or otherwise payable or distributable by the Administrative Agent to such Payment Recipient from any source, against any amount due to the Administrative Agent under this Section 12.19 or under the indemnification provisions of this Agreement, (y) the receipt of an Erroneous Payment by a Payment Recipient shall not for the purpose of this Agreement be treated as a payment, prepayment, repayment, discharge or other satisfaction of any Obligations owed by the Borrower or any other Loan Party, except, in each case, to the extent such Erroneous Payment is, and solely with respect to the amount of such Erroneous Payment that is, comprised of funds received by the Administrative Agent from the Borrower or any other Credit Party for the purpose of making for a payment on the Obligations and (z) to the extent that an Erroneous Payment was in any way or at any time credited as payment or satisfaction of any of the Obligations, the Obligations or any part thereof that were so credited, and all rights of the Payment Recipient, as the case may be, shall be reinstated and continue in full force and effect as if such payment or satisfaction had never been received.
- (e) Each party's obligations under this Section 12.19 shall survive the resignation or replacement of the Administrative Agent or any transfer of right or obligations by, or the replacement of, a Lender, the termination of the Term Loan Commitments or the repayment, satisfaction or discharge of all Obligations (or any portion thereof) under any Loan Document.
- (f) Nothing in this Section 12.19 will constitute a waiver or release of any claim of any party hereunder arising from any Payment Recipient's receipt of an Erroneous Payment.

ARTICLE XIII.

BENEFIT OF AGREEMENT; ASSIGNMENTS; PARTICIPATIONS

13.1. Successors and Assigns. The terms and provisions of the Loan Documents shall be binding upon and inure to the benefit of the Borrower and the Lenders and their respective successors and assigns permitted hereby, except that (i) the Borrower shall not have the right to assign its rights or obligations under the Loan Documents without the prior written consent of each Lender, (ii) any assignment by any Lender must be made in compliance with Section 13.3, and (iii) any transfer by Participation must be made in compliance with Section 13.2. Any attempted assignment or transfer by any party not made in compliance with this Section 13.1 shall be null and void, unless such attempted assignment or transfer is treated as a participation in accordance with Section 13.3. The parties to this Agreement acknowledge that clause (ii) of this Section 13.1 relates only to absolute assignments and this Section 13.1 does not prohibit assignments creating security interests, including, without limitation, (x) any pledge or assignment by any Lender of all or any portion of its rights under this Agreement and any Note to a Federal Reserve Bank or other central banking authority or (y) in the case of a Lender which is a Fund, any pledge or assignment of all or any portion of its rights under this Agreement and any Note to its trustee in support of its obligations to its trustee; provided, however, that no such pledge or assignment creating a security interest shall release the transferor Lender from its obligations hereunder unless and until the parties thereto have complied with the provisions of Section 13.3. The Administrative Agent may treat the Person which made any Loan or which holds any Note as the owner thereof for all purposes hereof unless and until such Person complies with Section 13.3; provided, however, that the Administrative Agent may in its discretion (but shall not be required to) follow instructions from the Person which made any Loan or which holds any Note to direct payments relating to such Loan or Note to another Person. Any assignee of the rights to any Loan or any Note agrees by acceptance of such assignment to be bound by all the terms and provisions of the Loan Documents. Any request, authority or consent of any Person, who at the time of making such request or giving such authority or consent is the owner of the rights to any Loan (whether or not a Note has been issued in evidence thereof), shall be conclusive and binding on any subsequent holder or assignee of the rights to such Loan.

13.2. Participations.

13.2.1 <u>Permitted Participants</u>; <u>Effect</u>. Any Lender may at any time sell to one or more banks or other entities (other than a natural Person, or a holding company, investment vehicle or trust for, or owned and operated for the primary benefit of, a natural Person, any Defaulting Lender, or the Borrower or any of the Borrower's Affiliates or Subsidiaries) ("<u>Participants</u>") participating interests in any Loan owing to such Lender, any Note held by such Lender, any Term Loan Commitment of such Lender or any other interest of such Lender under the Loan Documents. In the event of any such sale by a Lender of participating interests to a Participant, such Lender's obligations under the Loan Documents shall remain unchanged, such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations, such Lender shall remain the owner of its Loans and the holder of any Note issued to it in evidence thereof for all purposes under the Loan Documents, all amounts payable by the Borrower under

this Agreement shall be determined as if such Lender had not sold such participating interests, and the Borrower and the Administrative Agent shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under the Loan Documents.

- 13.2.2 <u>Voting Rights</u>. Each Lender shall retain the sole right to approve, without the consent of any Participant, any amendment, modification or waiver of any provision of the Loan Documents other than any amendment, modification or waiver with respect to any Loan or Term Loan Commitment in which such Participant has an interest which would require consent of all of the Lenders pursuant to the terms of <u>Section 14.13</u> or of any other Loan Document.
- 13.2.3 <u>Benefit of Certain Provisions</u>. The Borrower agrees that each Participant shall be deemed to have the right of setoff provided in <u>Section 14.15(a)</u> in respect of its participating interest in amounts owing under the Loan Documents to the same extent as if the amount of its participating interest were owing directly to it as a Lender under the Loan Documents, <u>provided</u> that each Lender shall retain the right of setoff provided in <u>Section 14.15(a)</u> with respect to the amount of participating interests sold to each Participant. The Lenders agree to share with each Participant, and each Participant, by exercising the right of setoff provided in <u>Section 14.15(a)</u>, agrees to share with each Lender, any amount received pursuant to the exercise of its right of setoff, such amounts to be shared in accordance with <u>Section 14.15(b)</u> as if each Participant were a Lender. The Borrower further agrees that each Participant shall be entitled to the benefits of <u>Sections 4.1, 4.2, 4.4</u> and <u>4.5</u> to the same extent as if it were a Lender and had acquired its interest by assignment pursuant to <u>Section 13.3</u>, <u>provided</u> that (i) a Participant shall not be entitled to receive any greater payment under <u>Section 4.1, 4.2, 4.4</u> or <u>4.5</u> than the Lender who sold the participating interest to such Participant would have received had it retained such interest for its own account, unless the sale of such interest to such Participant is made with the prior written consent of the Borrower, and (ii) any Participant not incorporated under the laws of the United States of America or any State thereof agrees to comply with the provisions of <u>Section 4.5</u> to the same extent as if it were a Lender.
- 13.2.4 <u>Participant Register</u>. Each Lender that sells a participation shall, acting solely for this purpose as a non-fiduciary agent of the Borrower, maintain a register on which it enters the name and address of each Participant and the principal amounts (and stated interest) of each Participant's interest in the Loans or other obligations under this Agreement (the "<u>Participant Register</u>"); provided that no Lender shall have any obligation to disclose all or any portion of the Participant Register to any Person (including the identity of any Participant or any information relating to a Participant's interest in any Term Loan Commitments, Loans or its other obligations under this Agreement) except to the extent that such disclosure is necessary to establish that such Term Loan Commitment, Loan or other obligation is in registered form under Section 5f.103-1(c) of the United States Treasury Regulations. The entries in the Participant Register shall be conclusive absent manifest error, and such Lender shall treat each person whose name is recorded in the Participant Register as the owner of such participation for all purposes of this Agreement notwithstanding any notice to the contrary.

13.3. <u>Assignments; Consents</u>. Any Lender may at any time assign to one or more Eligible Assignees all or a portion of its rights and obligations under this Agreement (including all or a portion of its Term Loan Commitment and the Loans at the time owing to it); <u>provided</u> that any such assignment shall be subject to the following conditions:

13.3.1 Minimum Amounts.

- (A) in the case of an assignment of the entire remaining amount of an assigning Lender's Term Loan Commitment and the Loans at the time owing to it, or in the case of an assignment to a Lender, an Affiliate of a Lender or an Approved Fund, no minimum amount need be assigned; and
- (B) in any case not described in the immediately preceding subsection (A), the aggregate amount of the Term Loan Commitment (which for this purpose includes Loans outstanding thereunder) or, if the applicable Term Loan Commitment is not then in effect, the principal outstanding balance of the Loans of the assigning Lender subject to each such assignment (determined as of the date the Assignment and Assumption Agreement with respect to such assignment is delivered to the Administrative Agent or, if "Trade Date" is specified in the Assignment and Assumption Agreement, as of the Trade Date) shall not be less than \$5,000,000, unless each of the Administrative Agent and, so long as no Event of Default has occurred and is continuing, the Borrower otherwise consents (each such consent not to be unreasonably withheld or delayed); provided, however, that if, after giving effect to such assignment, the amount of the Term Loan Commitment held by such assigning Lender or the outstanding principal balance of the Loans of such assigning Lender, as applicable, would be less than \$5,000,000, then such assigning Lender shall assign the entire amount of its Term Loan Commitment and the Loans at the time owing to it.
- 13.3.2 <u>Proportionate Amounts</u>. Each partial assignment shall be made as an assignment of a proportionate part of all the assigning Lender's rights and obligations under this Agreement with respect to the Loan or the Term Loan Commitment assigned.
- 13.3.3 <u>Required Consents</u>. No consent shall be required for any assignment except to the extent required by <u>Section 13.3.1(B)</u> and, in addition:
 - (A) the consent of the Borrower (such consent not to be unreasonably withheld or delayed) shall be required unless (x) an Event of Default has occurred and is continuing at the time of such assignment or (y) such assignment is to a Lender, an Affiliate of a Lender or an Approved Fund; provided that the Borrower shall be deemed to have consented to any such assignment unless it shall object thereto by written notice to the Administrative Agent within five (5) Business Days after having received notice thereof; and
 - (B) the consent of the Administrative Agent (such consent not to be unreasonably withheld or delayed) shall be required for assignments in respect of a Term Loan Commitment or a Loan if such assignment is to a Person that is not already a Lender, an Affiliate of a Lender or an Approved Fund with respect to a Lender.

- 13.3.4 <u>Assignment and Acceptance; Notes.</u> The parties to each assignment shall execute and deliver to the Administrative Agent an Assignment and Assumption Agreement, together with a processing and recordation fee of \$3,500 for each assignment, and the assignee, if it is not a Lender, shall deliver to the Administrative Agent an Administrative Questionnaire. If requested by the transferor Lender or the Assignee, upon the consummation of any assignment, the transferor Lender, the Administrative Agent and the Borrower shall make appropriate arrangements so that new Notes are issued to the Assignee and such transferor Lender, as appropriate.
- 13.3.5 <u>No Assignment to Borrower or Defaulting Lender</u>. No such assignment shall be made to (i) the Borrower or any of the Borrower's Affiliates or Subsidiaries or (ii) any Defaulting Lender.
- 13.3.6 <u>No Assignment to Natural Persons</u>. No such assignment shall be made to a natural person or a holding company, investment vehicle or trust for, or owned and granted for the primary benefit of, a natural Person.

Subject to acceptance and recording thereof by the Administrative Agent pursuant to Section 13.4, from and after the effective date specified in each Assignment and Assumption Agreement, the assignee thereunder shall be a party to this Agreement and, to the extent of the interest assigned by such Assignment and Assumption Agreement, have the rights and obligations of a Lender under this Agreement, and the assigning Lender thereunder shall, to the extent of the interest assigned by such Assignment and Assumption Agreement, be released from its obligations under this Agreement (and, in the case of an Assignment and Assumption Agreement covering all of the assigning Lender's rights and obligations under this Agreement, such Lender shall cease to be a party hereto) but shall continue to be entitled to the benefits of Sections 12.8 and 14.6 and the other provisions of this Agreement and the other Loan Documents that survive the termination hereof and thereof with respect to facts and circumstances occurring prior to the effective date of such assignment. Any assignment or transfer by a Lender of rights or obligations under this Agreement that does not comply with this paragraph shall be treated for purposes of this Agreement as a sale by such Lender of a participation in such rights and obligations in accordance with Section 13.2.

13.4. Register. The Administrative Agent, acting solely for this purpose as a non-fiduciary agent of the Borrower, shall maintain at the Administrative Office a copy of each Assignment and Assumption Agreement delivered to it and a register for the recordation of the names and addresses of the Lenders, and the Term Loan Commitments of, and principal amounts (and stated interest) of the Loans owing to, each Lender pursuant to the terms hereof from time to time (the "Register"). The entries in the Register shall be conclusive, and the Borrower, the Administrative Agent and the Lenders may treat each Person whose name is recorded in the Register pursuant to the terms hereof as a Lender hereunder for all purposes of this Agreement, notwithstanding notice to the contrary. The Register shall be available for inspection by the Borrower and any Lender, at any reasonable time and from time to time upon reasonable prior notice.

13.5. [<u>Reserved</u>].

13.6 <u>Confidentiality</u>. Except as otherwise provided by Applicable Law, the Administrative Agent and each Lender shall maintain the confidentiality of all Information (as defined below) in accordance with its customary procedure for handling confidential information of this nature and in accordance with safe and sound banking practices but in any event may make disclosure: (a) to its Affiliates and to its and its Affiliates' respective partners, directors, officers, employees, agents, advisors and other representatives (it being understood that the Persons to whom such disclosure is made will be informed of the confidential nature of such Information and instructed to keep such Information confidential); (b) subject to an agreement containing provisions substantially the same as those of this Section, to (i) any actual or proposed Eligible Assignee, Participant or other transferee in connection with a potential transfer of any Term Loan Commitment, Loan or participation therein as permitted hereunder, or (ii) any actual or prospective counterparty (or its advisors) to any swap or derivative transaction relating to the Borrower and its obligations; (c) as required or requested by any Governmental Authority or representative thereof or pursuant to legal process or in connection with any legal proceedings, or as otherwise required by Applicable Law; (d) to the Administrative Agent's or such Lender's independent auditors and other professional advisors (provided they shall be notified of the confidential nature of the information); (e) in connection with the exercise of any remedies under any Loan Document or any action or proceeding relating to any Loan Document or the enforcement of rights hereunder or thereunder; (f) to the extent such Information (i) becomes publicly available other than as a result of a breach of this Section actually known by the Administrative Agent or such Lender to be a breach of this Section or (ii) becomes available to the Administrative Agent, any Lender or any Affiliate of the Administrative Agent or any Lender on a nonconfidential basis from a source other than the Borrower or any Affiliate of the Borrower; (g) to the extent requested by, or required to be disclosed to, any nationally recognized rating agency or regulatory or similar authority (including any self-regulatory authority, such as the National Association of Insurance Commissioners) having or purporting to have jurisdiction over it; (h) to bank trade publications, such information to consist of deal terms and other information customarily found in such publications; (i) to any other party hereto; and (j) with the consent of the Borrower. Notwithstanding the foregoing, the Administrative Agent and each Lender may disclose any such confidential information, without notice to the Borrower or the General Partner, to Governmental Authorities in connection with any regulatory examination of the Administrative Agent or such Lender or in accordance with the regulatory compliance policy of the Administrative Agent or such Lender. As used in this Section, the term "Information" means all information received from the Borrower, the General Partner, any Subsidiary or Affiliate of the Borrower relating to the Borrower, the General Partner or any of their Affiliates or any of their respective businesses, other than any such information that is available to the Administrative Agent or any Lender on a nonconfidential basis prior to disclosure by the Borrower, the General Partner, or any Subsidiary or any Affiliate of the Borrower, provided that, in the case of any such information received from the Borrower, the General Partner, or any Subsidiary or any Affiliate of the Borrower after the date hereof, such information is clearly identified at the time of delivery as confidential. Any Person required to maintain the

confidentiality of Information as provided in this Section shall be considered to have complied with its obligation to do so if such Person has exercised the same degree of care to maintain the confidentiality of such Information as such Person would accord to its own confidential information.

13.7 <u>Tax Treatment</u>. If any interest in any Loan Document is transferred to any Participant or Purchaser or any other Person acquiring an interest in the Loan Documents by operation of law (each a "<u>Transferee</u>") which is not incorporated under the laws of the United States or any State thereof, the transferor Lender shall cause such Transferee, concurrently with the effectiveness of such transfer, to comply with the provisions of <u>Section 4.5(iv</u>).

ARTICLE XIV.

GENERAL PROVISIONS

- 14.1. <u>Survival of Representations</u>. All representations and warranties contained in this Agreement shall survive delivery of the Notes and the making of the Borrowings herein contemplated.
- 14.2. <u>Governmental Regulation</u>. Anything contained in this Agreement to the contrary notwithstanding, no Lender shall be obligated to extend credit to the Borrower in violation of any limitation or prohibition provided by any applicable statute or regulation.
- 14.3. <u>Taxes</u>. Any recording and other taxes (excluding franchise, income or similar taxes) or other similar assessments or charges payable or ruled payable by any governmental authority incurred in connection with the consummation of the transactions contemplated by this Agreement shall be paid by the Borrower, together with interest and penalties, if any.
- 14.4. <u>Headings</u>. Section headings in the Loan Documents are for convenience of reference only, and shall not govern the interpretation of any of the provisions of the Loan Documents.
- 14.5. <u>No Third Party Beneficiaries</u>. This Agreement shall not be construed so as to confer any right or benefit upon any Person other than the parties to this Agreement and their respective successors and assigns.
- 14.6. Expenses; Indemnification. Subject to the provisions of this Agreement, the Borrower will pay (a) all out-of-pocket costs and expenses incurred by the Administrative Agent and the Arranger (including the reasonable fees, out-of-pocket expenses and other reasonable expenses of counsel) in connection with the preparation, execution and delivery of this Agreement, the Notes, the Loan Documents and any other agreements or documents referred to herein or therein and any amendments thereto, (b) all out-of-pocket costs and expenses incurred by the Administrative Agent and the Lenders (including the reasonable fees, out-of-pocket expenses and other reasonable expenses of counsel to the Administrative Agent and the Lenders) in connection with the enforcement and protection of the rights of the Lenders under this Agreement, the Notes, the Loan Documents or any other agreement or document referred to herein or therein, and (c) all reasonable and customary costs and expenses of periodic audits by the Administrative Agent's personnel of the Borrower's books and records provided that prior to an Event of Default, the Borrower shall be required to pay for only one such audit during any year. The Borrower further agrees to indemnify the Lenders, their Affiliates, and their respective directors, officers, employees, agents and advisors (each, an "Indemnified Party") against all losses, claims, damages, penalties, judgments, liabilities and reasonable expenses (including, without limitation, all expenses of litigation or preparation therefor whether or not the Indemnified Party is a party thereto) which any of

them may pay or incur arising out of or relating to this Agreement, the other Loan Documents, the transactions contemplated hereby or the direct or indirect application or proposed application of the proceeds of any Borrowing hereunder, except that the foregoing indemnity shall not apply to any Indemnified Party to the extent that any losses, claims, etc. are determined by a court of competent jurisdiction by a final and non-appealable judgment to have resulted from such Indemnified Party's Gross Negligence or willful misconduct. In the case of an investigation, litigation or proceeding to which the indemnity in this Section applies, such indemnity shall be effective whether or not such investigation, litigation or proceeding is brought by the Borrower or the General Partner or the Borrower's or the General Partner's equity holders or creditors or an Indemnified Party is otherwise party thereto. The obligations of the Borrower under this Section shall survive the termination of this Agreement.

- 14.7. <u>Severability of Provisions</u>. Any provision in any Loan Document that is held to be inoperative, unenforceable, or invalid in any jurisdiction shall, as to that jurisdiction, be inoperative, unenforceable, or invalid without affecting the remaining provisions in that jurisdiction or the operation, enforceability, or validity of that provision in any other jurisdiction, and to this end the provisions of all Loan Documents are declared to be severable.
- 14.8. Nonliability of the Lenders. The relationship between the Borrower and the Lenders shall be solely that of borrower and lender. The Lenders shall not have any fiduciary responsibilities to the Borrower. The Lenders undertake no responsibility to the Borrower to review or inform the Borrower of any matter in connection with any phase of the Borrower's business or operations. Moreover, none of the Administrative Agent or any Lender shall be liable to the Borrower or any other parties. None of the Administrative Agent or any Lender, or any Affiliate, officer, director, employee, attorney, or agent of the Administrative Agent or any Lender shall have any liability with respect to, and the Borrower hereby waives, releases, and agrees not to sue any of them upon, any claim for any special, indirect, incidental, or consequential damages suffered or incurred by the Borrower in connection with, arising out of, or in any way related to, this Agreement, any of the other Loan Documents or any fee letter delivered in connection herewith, or any of the transactions contemplated by this Agreement or any of the other Loan Documents.
- 14.9. <u>Choice of Law</u>. THE LOAN DOCUMENTS (OTHER THAN THOSE CONTAINING A CONTRARY EXPRESS CHOICE OF LAW PROVISION) SHALL BE CONSTRUED IN ACCORDANCE WITH THE INTERNAL LAWS (AND NOT THE LAW OF CONFLICTS) OF THE STATE OF ILLINOIS, BUT GIVING EFFECT TO FEDERAL LAWS APPLICABLE TO NATIONAL BANKS.
- 14.10. <u>Consent to Jurisdiction</u>. THE BORROWER HEREBY IRREVOCABLY SUBMITS TO THE NON EXCLUSIVE JURISDICTION OF ANY UNITED STATES FEDERAL OR ILLINOIS STATE COURT SITTING IN CHICAGO, ILLINOIS IN ANY ACTION OR PROCEEDING ARISING

OUT OF OR RELATING TO ANY LOAN DOCUMENTS AND THE BORROWER HEREBY IRREVOCABLY AGREES THAT ALL CLAIMS IN RESPECT OF SUCH ACTION OR PROCEEDING MAY BE HEARD AND DETERMINED IN ANY SUCH COURT AND IRREVOCABLY WAIVES ANY OBJECTION IT MAY NOW OR HEREAFTER HAVE AS TO THE VENUE OF ANY SUCH SUIT, ACTION OR PROCEEDING BROUGHT IN SUCH A COURT OR THAT SUCH COURT IS AN INCONVENIENT FORUM. NOTHING HEREIN SHALL LIMIT THE RIGHT OF THE LENDERS TO BRING PROCEEDINGS AGAINST THE BORROWER IN THE COURTS OF ANY OTHER JURISDICTION. ANY JUDICIAL PROCEEDING BY THE BORROWER AGAINST THE LENDERS OR ANY AFFILIATE OF THE LENDERS INVOLVING, DIRECTLY OR INDIRECTLY, ANY MATTER IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH ANY LOAN DOCUMENT SHALL BE BROUGHT ONLY IN A COURT IN CHICAGO, ILLINOIS.

- 14.11. <u>Waiver of Jury Trial</u>. THE BORROWER, THE GENERAL PARTNER, THE ADMINISTRATIVE AGENT AND THE LENDERS HEREBY WAIVE TRIAL BY JURY IN ANY JUDICIAL PROCEEDING INVOLVING, DIRECTLY OR INDIRECTLY, ANY MATTER (WHETHER SOUNDING IN TORT, CONTRACT OR OTHERWISE) IN ANY WAY ARISING OUT OF, RELATED TO, OR CONNECTED WITH ANY LOAN DOCUMENT OR THE RELATIONSHIP ESTABLISHED THEREUNDER.
- 14.12. <u>Successors and Assigns</u>. The terms and provisions of the Loan Documents shall be binding upon and inure to the benefit of the Borrower and the Lenders and their respective successors and assigns, except that the Borrower shall not have the right to assign its rights or obligations under the Loan Documents. Any assignee or transferee of rights or obligations hereunder or under the Notes agrees by acceptance thereof to be bound by all the terms and provisions of the Loan Documents. Any request, authority or consent of any Person, who at the time of making such request or giving such authority or consent is the holder of a Note, shall be conclusive and binding on any subsequent holder, transferee or assignee of such Note or of any Note or Notes issued in exchange therefor.
- 14.13. <u>Entire Agreement; Modification of Agreement</u>. The Loan Documents embody the entire agreement among the Borrower, the General Partner, Administrative Agent, and Lenders and supersede all prior conversations, agreements, understandings, commitments and term sheets among any or all of such parties with respect to the subject matter hereof. Any provisions of this Agreement may be amended or waived if, but only if, such amendment or waiver is in writing and is signed by the Borrower, and Administrative Agent if the rights or duties of Administrative Agent are affected thereby, and
 - (a) each of the Lenders adversely affected thereby if such amendment or waiver:
 - (i) subject to clause (d) below, reduces or forgives any payment of principal or interest on the Obligations or any fees (including the Prepayment Premium pursuant to Section 2.24(c)) payable by the Borrower to such Lender hereunder; or

- (ii) postpones the date fixed for any payment of principal of or interest on the Obligations or any fees (including the Prepayment Premium pursuant to Section 2.24(c)) payable by the Borrower to such Lender hereunder; or
- (iii) increases the amount of such Lender's Term Loan Commitment or the unpaid principal amount of such Lender's Loan (other than pursuant to an assignment permitted under <u>Section 13.3</u>); or
 - (iv) extends the Maturity Date; or
 - (v) modifies or waives all or any portion of Section 2.20;
- (b) all of the Lenders if such amendment or waiver:
 - (i) releases or limits the liability of the General Partner under the Loan Documents; or
 - (ii) changes the definition of Required Lenders, or modifies any requirement for consent by each of the Lenders; or
 - (iii) modifies or waives the second and third sentences of Section 9.3 hereof;
- (c) the Required Lenders, to the extent expressly provided for herein and in the case of all other waivers or amendments if no percentage of Lenders is specified herein; or
- (d) the Administrative Agent and the Borrower may, without the consent of any Lender, enter into amendments or modifications to this Agreement or any of the other Loan Documents or enter into additional Loan Documents as the Administrative Agent reasonably deems appropriate in order to implement any Replacement Rate or otherwise effectuate the terms of Section 4.3(ii) in accordance with the terms of Section 4.3(ii).
- 14.14. <u>Dealings with the Borrower</u>. The Lenders and their affiliates may accept deposits from, extend credit to and generally engage in any kind of banking, trust or other business with the Borrower or the General Partner or any of their Affiliates regardless of the capacity of the Lenders hereunder.

14.15. Set-Off.

(a) Subject to Section 2.20 and in addition to any rights now or hereafter granted under Applicable Law and not by way of limitation of any such rights, the Borrower hereby authorizes the Administrative Agent, each Lender, each Affiliate of the Administrative Agent or any Lender, and each Participant, at any time, or from time to time while, an Event of Default exists, without notice to the Borrower or to any other Person, any such notice being hereby expressly waived, but in the case of a Lender, an Affiliate of a Lender, or a Participant, subject to receipt of the prior written consent of the Required Lenders exercised in their sole discretion, to set off and to appropriate and to apply any and all deposits (general or special, including, but not limited to, indebtedness evidenced by certificates of deposit, whether matured or unmatured) and any other indebtedness at any time held or owing by the Administrative Agent, such Lender, any Affiliate of the Administrative Agent or such Lender, or such Participant, to or for the credit or the account of the Borrower against and on account of any of the Obligations, irrespective of whether or not any or all of the Loans and all other Obligations have been declared to be, or have

otherwise become, due and payable as permitted by Section 11.1, and although such Obligations shall be contingent or unmatured.

(b) Each Lender agrees that if it shall, by exercising any right of set off or counterclaim or otherwise, receive payment of a proportion of the aggregate amount of principal, interest or fees due with respect to any Loan held by it (other than payments received pursuant to Sections 4.1, 4.2, 4.3 and 4.5) which is greater than the proportion received by any other Lender in respect of the aggregate amount of principal, interest or fees due with respect to any Loan held by such other Lender, the Lender receiving such proportionately greater payment shall purchase such participations in the Loans held by the other Lenders and such other adjustments shall be made as may be required so that all such payments of principal, interest or fees with respect to the Loans held by the Lenders shall be shared by the Lenders pro rata according to their respective Percentages.

14.16. Counterparts; Integration, Effectiveness, Electronic Execution.

- (a) ___14.16. Counterparts. This Agreement may be executed in any number of counterparts, all of which (and by different parties hereto in different counterparts), each of which shall constitute an original, but all of which when taken together shall constitute one agreement, and any of the parties hereto may execute this Agreement by signing any such counterpart. This Agreement shall be a single contract. This Agreement and the other Loan Documents, and any separate letter agreements with respect to fees payable to the Administrative Agent and/or any Arranger, constitute the entire contract among the parties relating to the subject matter hereof and supersede any and all previous agreements and understandings, oral or written, relating to the subject matter hereof. Except as provided in Section 5.1, this Agreement shall become effective when it has shall have been executed by the Borrower and each of the Lenders shown on the signature pages hereof. Administrative Agent and when the Administrative Agent shall have received counterparts hereof that, when taken together, bear the signatures of each of the other parties hereto. Delivery of an executed counterpart of a signature page of this Agreement by facsimile or in electronic (i.e., "pdf" or "tif") format shall be effective as delivery of a manually executed counterpart of this Agreement.
- (b) The words "execute," "execution," "signed," "signature," "delivery" and words of like import in or related to this Agreement, any other Loan Document or any document, amendment, approval, consent, waiver, modification, information, notice, certificate, report, statement, disclosure, or authorization to be signed or delivered in connection with this Agreement or any other Loan Document or the transactions contemplated hereby shall be deemed to include Electronic Signatures or execution in the form of an Electronic Record, and contract formations on electronic platforms approved by the Administrative Agent, deliveries or the keeping of records in electronic form, each of which shall be of the same legal effect, validity or enforceability as a manually executed signature or the use of a paper-based recordkeeping system,

as the case may be, to the extent and as provided for in any Applicable Law, including the Federal Electronic Signatures in Global and National Commerce Act, the New York State Electronic Signatures and Records Act, or any other similar state laws based on the Uniform Electronic Transactions Act. Each party hereto agrees that any Electronic Signature or execution in the form of an Electronic Record shall be valid and binding on itself and each of the other parties hereto to the same extent as a manual, original signature. For the avoidance of doubt, the authorization under this paragraph may include, without limitation, use or acceptance by the parties of a manually signed paper which has been converted into electronic form (such as scanned into PDF format), or an electronically signed paper converted into another format, for transmission, delivery and/or retention. Notwithstanding anything contained herein to the contrary, the Administrative Agent is under no obligation to accept an Electronic Signature in any form or in any format unless expressly agreed to by the Administrative Agent pursuant to procedures approved by it; provided that without limiting the foregoing, (a) to the extent the Administrative Agent has agreed to accept such Electronic Signature from any party hereto, the Administrative Agent and the other parties hereto shall be entitled to rely on any such Electronic Signature purportedly given by or on behalf of the executing party without further verification and (b) upon the request of the Administrative Agent or any Lender, any Electronic Signature shall be promptly followed by an original manually executed counterpart thereof. Without limiting the generality of the foregoing, each party hereto hereby (i) agrees that, for all purposes, including without limitation, in connection with any workout, restructuring, enforcement of remedies, bankruptcy proceedings or litigation among the Administrative Agent, the Lenders and any of the Loan Parties, electronic images of this Agreement or any other Loan Document (in each case, including with respect to any signature pages thereto) shall have the same legal effect, validity and enforceability as any paper original, and (ii) waives any argument, defense or right to contest the validity or enforceability of the Loan Documents based solely on the lack of paper original copies of any Loan Documents, including with respect to any signature pages thereto.

14.17. <u>Patriot Act CIP Notice</u>. Each Lender hereby notifies the Borrower and the General Partner that pursuant to the requirements of the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)) (the "Act), it is from time to time required to obtain, verify and record information that identifies the Borrower and the General Partner, which information includes the name and address of the Borrower and the General Partner and other information that will allow such Lender to identify the Borrower and the General Partner in accordance with the Act.

14.18. Acknowledgement and Consent to Bail-In of **EEA**Affected Financial Institutions.

Notwithstanding anything to the contrary in any Loan Document or in any other agreement, arrangement or understanding among any such parties, each party hereto acknowledges that any liability of any EEAAffected Financial Institution arising under any Loan Document, to the extent such liability is unsecured, may be subject to the Write-Down and Conversion Powers of am EEAthe applicable Resolution Authority and agrees and consents to, and acknowledges and agrees to be bound by:

- (a) (a) the application of any Write-Down and Conversion Powers by an EEAthe applicable Resolution Authority to any such liabilities arising hereunder which may be payable to it by any party hereto that is an EEAAffected Financial Institution; and
 - (b) (b) the effects of any Bail-In Action on any such liability, including, if applicable:
 - (i) a reduction in full or in part or cancellation of any such liability;
 - (ii) a conversion of all, or a portion of, such liability into shares or other instruments of ownership in such EEAAffected Financial Institution, its parent undertaking, or a bridge institution that may be issued to it or otherwise conferred on it, and that such shares or other instruments of ownership will be accepted by it in lieu of any rights with respect to any such liability under this Agreement or any other Loan Document; or
 - (iii) (iii) the variation of the terms of such liability in connection with the exercise of the Write-Down and Conversion Powers of any EEAthe applicable Resolution Authority.
- 14.19. Acknowledgement Regarding Any Supported QFCs. To the extent that the Loan Documents provide support, through a guarantee or otherwise, for Rate Management Transactions or any other agreement or instrument that is a QFC (such support, "QFC Credit Support" and, each such QFC, a "Supported QFC"), the parties acknowledge and agree as follows with respect to the resolution power of the FDIC under the Federal Deposit Insurance Act and Title II of the Dodd-Frank Wall Street Reform and Consumer Protection Act (together with the regulations promulgated thereunder, the "U.S. Special Resolution Regimes") in respect of such Supported QFC and QFC Credit Support (with the provisions below applicable notwithstanding that the Loan Documents and any Supported QFC may in fact be stated to be governed by the laws of the State of New York and/or of the United States or any other state of the United States):
 - (a) In the event a Covered Entity that is party to a Supported QFC (each, a "Covered Party") becomes subject to a proceeding under a U.S. Special Resolution Regime, the transfer of such Supported QFC and the benefit of such QFC Credit Support (and any interest and obligation in or under such Supported QFC and such QFC Credit Support, and any rights in property securing such Supported QFC or such QFC Credit Support) from such Covered Party will be effective to the same extent as the transfer would be effective under the U.S. Special Resolution Regime if the Supported QFC and such QFC Credit Support (and any such interest, obligation and rights in property) were governed by the laws of the United States or a state of the United States. In the event a Covered Party or a BHC Act Affiliate of a Covered Party becomes subject to a proceeding under a U.S. Special Resolution Regime, Default Rights under the Loan Documents that might otherwise apply to such Supported QFC or any QFC Credit Support that may be exercised against such Covered Party are permitted to be exercised to no greater extent

than such Default Rights could be exercised under the U.S. Special Resolution Regime if the Supported QFC and the Loan Documents were governed by the laws of the United States or a state of the United States. Without limitation of the foregoing, it is understood and agreed that rights and remedies of the parties with respect to a Defaulting Lender shall in no event affect the rights of any Covered Party with respect to a Supported QFC or any QFC Credit Support.

(b) As used in this Section 14.19, the following terms have the following meanings:

"BHC Act Affiliate" of a party means an "affiliate" (as such term is defined under, and interpreted in accordance with, 12 U.S.C. 1841(k)) of such party.

"Covered Entity" means any of the following:

- (i) a "covered entity" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 252.82(b);
- (ii) a "covered bank" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 47.3(b); or
- (iii) a "covered FSI" as that term is defined in, and interpreted in accordance with, 12 C.F.R. § 382.2(b).

"Default Right" has the meaning assigned to that term in, and shall be interpreted in accordance with, 12 C.F.R. §§ 252.81, 47.2 or 382.1, as applicable.

"QFC" has the meaning assigned to the term "qualified financial contract" in, and shall be interpreted in accordance with, 12 U.S.C. 5390(c)(8)(D).

14.20. Certain ERISA Matters.

- (g) Each Lender (x) represents and warrants, as of the date such Person became a Lender party hereto, to, and (y) covenants, from the date such Person became a Lender party hereto to the date such Person ceases being a Lender party hereto, for the benefit of, the Administrative Agent, each Arranger and their respective Affiliates, and not, for the avoidance of doubt, to or for the benefit of the Borrower or any other Loan Party, that at least one of the following is and will be true:
 - (i) such Lender is not using "plan assets" (within the meaning of Section 3(42) of ERISA or otherwise) of one or more Benefit Plans with respect to such Lender's entrance into, participation in, administration of and performance of the Loans or the Term Loan Commitments;
 - (ii) the transaction exemption set forth in one or more PTEs, such as PTE 84-14 (a class exemption for certain transactions determined by independent qualified professional asset managers), PTE 95-60 (a class exemption for certain transactions involving insurance company general accounts), PTE 90-1 (a class exemption for certain transactions involving insurance company pooled separate accounts), PTE 91-38 (a class exemption for certain transactions involving bank collective investment funds) or PTE 96-23 (a class exemption for certain transactions determined by in-house asset managers), is applicable with respect to such Lender's entrance into, participation in, administration of and performance of the Loans, the Term Loan Commitments and this Agreement;

(iii) (A) such Lender is an investment fund managed by a "Qualified Professional Asset Manager" (within the meaning of Part VI of PTE 84-14), (B) such Qualified Professional Asset Manager made the investment decision on behalf of such Lender to enter into, participate in, administer and perform the Loans, the Term Loan Commitments and this Agreement, (C) the entrance into, participation in, administration of and performance of the Loans, the Term Loan Commitments and this Agreement satisfies the requirements of sub-sections (b) through (g) of Part I of PTE 84-14 and (D) to the best knowledge of such Lender, the requirements of subsection (a) of Part I of PTE 84-14 are satisfied with respect to such Lender's entrance into, participation in, administration of and performance of the Loans, the Term Loan Commitments and this Agreement; or

(iv) such other representation, warranty and covenant as may be agreed in writing between the Administrative Agent, in its sole discretion, and such Lender.

In addition, unless either (1) sub-clause (i) in the immediately preceding clause (a) is true with respect to a Lender or (2) a Lender has provided another representation, warranty and covenant in accordance with sub-clause (iv) in the immediately preceding clause (a), such Lender further (x) represents and warrants, as of the date such Person became a Lender party hereto, to, and (y) covenants, from the date such Person became a Lender party hereto, for the benefit of, the Administrative Agent, each Arranger and their respective Affiliates, and not, for the avoidance of doubt, to or for the benefit of the Borrower or any other Loan Party, that none of the Administrative Agent, any Arranger and their respective Affiliates is a fiduciary with respect to the assets of such Lender involved in such Lender's entrance into, participation in, administration of and performance of the Loans, the Term Loan Commitments and this Agreement (including in connection with the reservation or exercise of any rights by the Administrative Agent under this Agreement, any Loan Document or any documents related hereto or thereto).

ARTICLE XV.

NOTICES

15.1. <u>Giving Notice</u>; <u>Electronic Delivery</u>. All notices and other communications provided to any party hereto under this Agreement or any other Loan Document shall be in writing or by <u>telex or by facsimileemail</u> and addressed or delivered to such party at its address set forth below or at such other address as may be designated by such party in a notice to the other parties. Any notice, if mailed and properly addressed with postage prepaid, shall be deemed given when received; any notice, if transmitted by <u>telex or facsimileemail</u>, shall be deemed given when transmitted (<u>answerback confirmed in the case of telexesprovided that, if such notice is transmitted by email, such notice is also promptly mailed in accordance with this <u>paragraph</u>). Notice may be given as follows:</u>

To the Borrower:

First Industrial, L.P. c/o First Industrial Realty Trust, Inc. 311 South 1 N. Wacker Drive Suite 40004200 Chicago, Illinois 60606 Attention: Mr. Scott Musil Telecopy: (312) 895-9380 Email: smusil@firstindustrial.com

To the General Partner:

First Industrial Realty Trust, Inc. 311 South 1 N. Wacker Drive Suite 39004200 Chicago, Illinois 60606 Attention: Scott A. Musil Telecopy: (312) 922 9851 Email: smusil@firstindustrial.com

Each of the above with a copy to:

Barack Ferrazzano Kirschbaum & Nagelberg LLP 200 West Madison 39th Floor Chicago, Illinois 60606 Attention: Suzanne Bessette-Smith and Matthew A. Jackson Telecopy: (312) 984 3150 Email: Suzanne.bessette-smith@bfkn.com; matt.jackson@bfkn.com

To each Lender:

At such address as set forth in its Administrative Questionnaire

To the Administrative Agent:

Wells Fargo Bank, National Association, as agent 10 South Wacker Drive, 32nd Floor Chicago, IL 60606 Attention: Karen T. Scutt Scott S. Solis Telecopy: (312)782-0969 Email: scott.s.solis@wellsfargo.com

with a copy to:

Wells Fargo Bank, National Association, as agent

333 South Grand Avenue, 9th Floor Los Angeles, CA 90071 Attention: Rita M. Swayne

Telecopy: (866)600-0942

Documents required to be delivered pursuant to the Loan Documents may be delivered by electronic communication and delivery, including, the Internet, e-mail or intranet websites to which the Administrative Agent and each Lender have access (including a commercial, third-party website such as www.sec.gov http://www.sec.gov or a website sponsored or hosted by the Administrative Agent or the Borrower) provided that the foregoing shall not apply to (i) notices to any Lender pursuant to Article II and (ii) any Lender that has notified the Administrative Agent and the Borrower that it cannot or does not want to receive electronic communications. The Administrative Agent or the Borrower may, in its discretion, agree to accept notices and other communications to it hereunder by electronic delivery pursuant to procedures approved by it for all or particular notices or communications. Documents or notices delivered electronically shall be deemed to have been delivered twenty-four (24) hours after the date and time on which the Administrative Agent or the Borrower posts such documents or the documents become available on a commercial website and the Administrative Agent or Borrower notifies each Lender of said posting and provides a link thereto provided if such notice or other communication is not sent or posted during the normal business hours of the recipient, said posting date and time shall be deemed to have commenced as of 11:00 a.m. (Central Time) on the opening of business on the next business day for the recipient. Notwithstanding anything contained herein, in every instance the Borrower shall be required to provide paper copies of the certificate required by Section 8.2(iv) to the Administrative Agent and shall deliver paper copies of any documents to the Administrative Agent or to any Lender that requests such paper copies until a written request to cease delivering paper copies is given by the Administrative Agent or such Lender. Except for the certificates required by Section 8.2(iv), the Administrative Agent shall have no obligation to request the delivery of or to maintain paper copies of the documents delivered electronically, and in any event shall have no responsibility to monitor compliance by the Borrower with any such request for delivery. Each Lender shall be solely responsible for requesting delivery to it of paper copies and maintaining its paper or electronic documents.

Documents required to be delivered pursuant to Article II may be delivered electronically to a website provided for such purpose by the Administrative Agent pursuant to the procedures provided to the Borrower by the Administrative Agent.

15.2. <u>Change of Address</u>. Each party may change the address for service of notice upon it by a notice in writing to the other parties hereto.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, the undersigned have executed this Agreement as of the date first above written.

BORROWER: FIRST INDUSTRIAL, L.P.

By: FIRST INDUSTRIAL REALTY TRUST, INC., its General Partner

By:	
Name:	
Title	

GENERAL PARTNER: FIRST INDUSTRIAL REALTY TRUST, INC.

By: _	
Name:	
Title:	

SIGNATURE PAGE TO FIRST INDUSTRIAL TERM LOAN CREDIT AGREEMENT

By: Name: Title:
PNC BANK, NATIONAL ASSOCIATION, individually as a Lender and as Syndication Agent
By: Name: Title:
REGIONS BANK, individually as a Lender and as Syndication Agent
By: Name: Title:
U.S. BANK NATIONAL ASSOCIATION, individually as a Lender and as Syndication Agent
By: Name: Title:

LENDERS: WELLS FARGO BANK, NATIONAL ASSOCIATION, individually as a

Lender and as Administrative Agent

SIGNATURE PAGE TO FIRST INDUSTRIAL TERM LOAN CREDIT AGREEMENT

[Include Other Lender Signature Blocks]

EXHIBIT A

TERM LOAN COMMITMENT AMOUNTS (As of the Agreement Effective Date)

Bank	Total Commitment
Wells Fargo Bank, National Association	\$50,000,000.00
PNC Bank, National Association	\$50,000,000.00
U.S. Bank National Association	\$50,000,000.00
Regions Bank	\$45,000,000.00
Associated Bank	\$40,000,000.00
Fifth Third Bank	\$25,000,000.00
Total	\$260,000,000.00

TERM LOAN EXPOSURE AMOUNTS

(After giving effect to Second Third Amendment to Unsecured Term Loan Agreement)

Bank	Term Loan Amount
Wells Fargo Bank, National Association	\$50,000,000
PNC Bank, National Association	\$50,000,000
U.S. Bank National Association	\$50,000,000
Regions Bank	\$45,000,000
Associated Bank	\$40,000,000
Fifth Third Bank	\$25,000,000
Total	\$260,000,000.00

EXHIBIT B

FORM OF NOTE

TERM LOAN NOTE

1, 2015

On or before the Maturity Date, as defined in that certain Unsecured Term Loan Agreement dated as of September 11,
2015 (the " <u>Agreement</u> ") between FIRST INDUSTRIAL, L.P., a Delaware limited partnership (" <u>Borrower</u> "), First Industrial
Realty Trust, Inc., a Maryland corporation, Wells Fargo Bank, National Association, individually and as Administrative Agent for
the Lenders (as such terms are defined in the Agreement), and the other Lenders listed on the signature pages of the Agreement,
Borrower promises to pay to the order of [] (the "Lender"), or its successors and assigns, the aggregate unpaid
principal amount of all Loans made by the Lender to the Borrower pursuant to Section 2.1 of the Agreement, in immediately
available funds at the office of the Administrative Agent in Minneapolis, Minnesota, together with interest on the unpaid
principal amount hereof at the rates and on the dates set forth in the Agreement. The Borrower shall pay this Promissory Note
("Note") in full on or before the Maturity Date in accordance with the terms of the Agreement.

The Lender shall, and is hereby authorized to record on the schedule attached hereto, or to otherwise record in accordance with its usual practice, the date and amount of each Borrowing and the date and amount of each principal payment hereunder; provided, however, that the failure of Lender to so record shall not affect the obligations of the Borrower hereunder or under the other Loan. Documents.

This Note is issued pursuant to, and is entitled to the security under and benefits of, the Agreement and the other Loan Documents, to which Agreement and Loan Documents, as they may be amended from time to time, reference is hereby made for, inter alia, a statement of the terms and conditions under which this Note may be prepaid or its maturity date accelerated. Capitalized terms used herein and not otherwise defined herein are used with the meanings attributed to them in the Agreement.

If there is an Event of Default or Default under the Agreement or any other Loan Document and Lender exercises its remedies provided under the Agreement and/or any of the Loan Documents, then in addition to all amounts recoverable by the Lender under such documents, Lender shall be entitled to receive reasonable attorneys fees and expenses incurred by Lender in exercising such remedies.

Borrower and all endorsers severally waive presentment, protest and demand, notice of protest, demand and of dishonor and nonpayment of this Note (except as otherwise expressly provided for in the Agreement), and any and all lack of diligence or delays in collection or enforcement of this Note, and expressly agree that this Note, or any payment hereunder, may be extended from time to time, and expressly consent to the release of any party liable for the obligation secured by this Note, the release of any of the security of this Note, the acceptance of any other security therefor, or any other indulgence or forbearance whatsoever, all without notice to any party and without affecting the liability of the Borrower and any endorsers hereof.

This Note shall, be governed and construed under the internal laws of the State of Illinois.

BORROWER AND LENDER, BY ITS ACCEPTANCE HEREOF, EACH HEREBY WAIVE ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION OR PROCEEDING TO ENFORCE OR DEFEND ANY RIGHT UNDER THIS PROMISSORY NOTE OR ANY OTHER LOAN DOCUMENT OR RELATING THERETO OR ARISING FROM THE LENDING RELATIONSHIP WHICH IS THE SUBJECT OF THIS NOTE AND AGREE THAT ANY SUCH ACTION OR. PROCEEDING SHALL BE TRIED BEFORE A COURT AND NOT BEFORE A JURY.

[Signature on Following Page]

FIRST INDUSTRIAL, L.P.

By: First Industrial Realty Trust, Inc., its general partner
By: Name:
Title:

PAYMENTS OF PRINCIPAL

Date Unpaid Principal Balance Notation Made by

EXHIBIT C

DISBURSEMENT INSTRUCTION AGREEMENT

Borrower: FIRST INDUSTRIAL, L.P.		
Administrative Agent: Wells Fargo Bank, National Association		
Loan: Loan number [] made pursuant to that certain Unsecured Term Loan Agreement ("Credit Agreement") dated as of September 11, 2015 between Borrower, FIRST INDUSTRIAL REALTY TRUST, INC., a Maryland corporation, as General Partner, Administrative Agent and Lenders, as amended from time to time		
Effective Date: September 11, 2015		
Check applicable box:		
New – This is the first Disbursement Instruction Agreement submitted in connection with the Loan. Replace Previous Agreement – This is a replacement Disbursement Instruction Agreement. All prior instructions submitted in connection with this Loan are cancelled as of the Effective Date set forth above.		

This Agreement must be signed by the Borrower and is used for the following purposes:

- (1) to designate an individual or individuals with authority to request disbursements of Loan proceeds, whether at the time of Loan closing/origination or thereafter;
- (2) to designate an individual or individuals with authority to request disbursements of funds from Restricted Accounts (as defined in the Terms and Conditions attached to this Agreement), if applicable; and
- (3) to provide Administrative Agent with specific instructions for wiring or transferring funds on Borrower's behalf.

Any of the disbursements, wires or transfers described above are referred to herein as a "Disbursement."

Specific dollar amounts for Disbursements must be provided to Administrative Agent at the time of the applicable Disbursement in the form of a signed closing statement, an email instruction or other written communication, or telephonic request pursuant to 2.7 of the Credit Agreement (each, a "**Disbursement Request**") from an applicable Authorized Representative (as defined in the Terms and Conditions attached to this Agreement).

A new Disbursement Instruction Agreement must be completed and signed by the Borrower if (i) all or any portion of a Disbursement is to be transferred to an account or an entity not described in this Agreement or (ii) Borrower wishes to add or remove any Authorized Representatives.

See the Additional Terms and Conditions attached hereto for additional information and for definitions of certain capitalized terms used in this Agreement.

Disbu	rsement of Loan Proceeds at Origination/Closing	
indivi	ng <u>Disbursement Authorizers</u> : Administrative Agent is authorized duals named below (each, a " Closing Disbursement Authorized ation/closing and to initiate Disbursements in connection therev	
	Individual's Name	Title
1.		
2.		
3.		
etc.): DESC If the	ibe Restrictions, if any, on the authority of the Closing Disburse CRIBE APPLICABLE RESTRICTIONS OR INDICATE "N/A" re are no restrictions described here, any Closing Disbursenable Loan proceeds.	ment Authorizers (dollar amount limits, wire/deposit destinations, nent Authorizer may submit a Disbursement Request for all
DELE	ΓΕ FOLLOWING SECTION IF NO WIRE TRANSFERS AT C	PRIGINATION/CLOSING
and ap		
	Names of Receiving Parties for the Closing Disbursement(s) (Receiving Party must be attached as the Closing Exhibit)	may include as many parties as needed; wire instructions for each
1.		
2.		
3.		
DELE	TE FOLLOWING SECTION IF NO DEPOSITS INTO WFB A	CCOUNTS AT ORIGINATION/CLOSING
	<u>t Deposit:</u> Disbursement Requests for the Closing Disbursement by the amount and applicable account. Each account included in	(s) to be deposited into an account at Wells Fargo Bank, N.A. must any such Disbursement Request must be listed below.
Name	on Deposit Account:	
Wells	Fargo Bank, N.A. Deposit Account Number:	
Furthe	er Credit Information/Instructions:	

Disbursements of Loan Proceeds Subsequent to Loan Closing/Origination Subsequent Disbursement Authorizers: Administrative Agent is authorized to accept one or more Disbursement Requests from any of the individuals named below (each, a "Subsequent Disbursement Authorizer") to disburse Loan proceeds after the date of the Loan origination/closing and to initiate Disbursements in connection therewith (each, a "Subsequent Disbursement"): Individual's Name Individual's Name Title Describe Restrictions, if any, on the authority of the Subsequent Disbursement Authorizers (dollar amount limits, wire/deposit destination etc.): DESCRIBE APPLICABLE RESTRICTIONS OR INDICATE "N/A" If there are no restrictions described here, any Subsequent Disbursement Authorizer may submit a Disbursement Request for all available Loan proceeds. DELETE FOLLOWING SECTION IF NO SUBSEQUENT WIRE TRANSFERS ANTICIPATED Permitted Wire Transfers: Disbursement Requests for Subsequent Disbursements to be made by wire transfer must specify the amount an applicable Receiving Party. Each Receiving Party included in any such Disbursement Request must be listed below. Administrative Agen is authorized to use the wire instructions that have been provided directly to Administrative Agen by the Receiving Party or Borrower are attached as the Subsequent Disbursement Exhibit. Names of Receiving Parties for Subsequent Disbursements (may include as many parties as needed; wire instructions for each Receiving Party must be attached as the Subsequent Disbursement Exhibit. Names of Receiving Parties for Subsequent Disbursement Exhibit. Names of Receiving Parties for Subsequent Disbursement Exhibit. Names of Beceiving Parties for Subsequent Disbursement Exhibit. Direct Deposit: Disbursement Requests for Subsequent Disbursement Exhibit. Direct Deposit: Disbursement Requests for Subsequent Disbursements to be deposited into an account at Wells Fargo Bank, N.A. must specify the amount and applicable account. Each account included in any such Disbursement	1		
individuals named below (each, a "Subsequent Disbursement Authorizer") to disburse Loan proceeds after the date of the Loan origination/closing and to initiate Disbursements in connection therewith (each, a "Subsequent Disbursement"): Individual's Name Title 1. 2. 3. Describe Restrictions, if any, on the authority of the Subsequent Disbursement Authorizers (dollar amount limits, wire/deposit destination etc.): DESCRIBE APPLICABLE RESTRICTIONS OR INDICATE "N/A" If there are no restrictions described here, any Subsequent Disbursement Authorizer may submit a Disbursement Request for all available Loan proceeds. DELETE FOLLOWING SECTION IF NO SUBSEQUENT WIRE TRANSFERS ANTICIPATED Permitted Wire Transfers: Disbursement Requests for Subsequent Disbursements to be made by wire transfer must specify the amount an applicable Receiving Party. Each Receiving Party included in any such Disbursement Request must be listed below. Administrative Agen is authorized to use the wire instructions that have been provided directly to Administrative Agent by the Receiving Party or Borrower and attached as the Subsequent Disbursement Exhibit. Names of Receiving Parties for Subsequent Disbursements (may include as many parties as needed; wire instructions for each Receiving Party must be attached as the Subsequent Disbursement Exhibit) Names of Receiving Parties for Subsequent Disbursements (may include as many parties as needed; wire instructions for each Receiving Party must be attached as the Subsequent Disbursement Exhibit) Direct Deposit; Disbursement Requests for Subsequent Disbursements to be deposited into an account at Wells Fargo Bank, N.A. must specify the amount and applicable account. Each account included in any such Disbursement Request must be listed below. Name on Deposit Account: Wells Fargo Bank, N.A. Deposit Account Number:	Disbu	irsements of Loan Proceeds Subsequent to Lo	oan Closing/Origination
1. 2. 3.	indiv	iduals named below (each, a "Subsequent Disbu	ursement Authorizer") to disburse Loan proceeds after the date of the Loan
2. 3. Describe Restrictions, if any, on the authority of the Subsequent Disbursement Authorizers (dollar amount limits, wire/deposit destination etc.): DESCRIBE APPLICABLE RESTRICTIONS OR INDICATE "N/A" If there are no restrictions described here, any Subsequent Disbursement Authorizer may submit a Disbursement Request for all available Loan proceeds. DELETE FOLLOWING SECTION IF NO SUBSEQUENT WIRE TRANSFERS ANTICIPATED Permitted Wire Transfers; Disbursement Requests for Subsequent Disbursements to be made by wire transfer must specify the amount an applicable Receiving Party. Each Receiving Party included in any such Disbursement Request must be listed below. Administrative Agen is authorized to use the wire instructions that have been provided directly to Administrative Agent by the Receiving Party or Borrower an attached as the Subsequent Disbursement Exhibit. Names of Receiving Parties for Subsequent Disbursements (may include as many parties as needed; wire instructions for each Receiving Party must be attached as the Subsequent Disbursement Exhibit) Names of Receiving Parties for Subsequent Disbursement Exhibit) Direct Deposit; Disbursement Requests for Subsequent Disbursements to be deposited into an account at Wells Fargo Bank, N.A. must specify the amount and applicable account. Each account included in any such Disbursement Request must be listed below. Name on Deposit Account: Wells Fargo Bank, N.A. Deposit Account Number:		Individual's Name	Title
Describe Restrictions, if any, on the authority of the Subsequent Disbursement Authorizers (dollar amount limits, wire/deposit destination etc.): DESCRIBE APPLICABLE RESTRICTIONS OR INDICATE "N/A" If there are no restrictions described here, any Subsequent Disbursement Authorizer may submit a Disbursement Request for all available Loan proceeds. DELETE FOLLOWING SECTION IF NO SUBSEQUENT WIRE TRANSFERS ANTICIPATED Permitted Wire Transfers: Disbursement Requests for Subsequent Disbursements to be made by wire transfer must specify the amount an applicable Receiving Party. Each Receiving Party included in any such Disbursement Request must be listed below. Administrative Agen is authorized to use the wire instructions that have been provided directly to Administrative Agent by the Receiving Party or Borrower an attached as the Subsequent Disbursement Exhibit. Names of Receiving Parties for Subsequent Disbursements (may include as many parties as needed; wire instructions for each Receiving Party must be attached as the Subsequent Disbursement Exhibit) Names of Receiving Parties for Subsequent Disbursement Exhibit) Direct Deposit: Disbursement Requests for Subsequent Disbursements to be deposited into an account at Wells Fargo Bank, N.A. must specify the amount and applicable account. Each account included in any such Disbursement Request must be listed below. Name on Deposit Account: Wells Fargo Bank, N.A. Deposit Account Number:	1.		
Describe Restrictions, if any, on the authority of the Subsequent Disbursement Authorizers (dollar amount limits, wire/deposit destination etc.): DESCRIBE APPLICABLE RESTRICTIONS OR INDICATE "N/A" If there are no restrictions described here, any Subsequent Disbursement Authorizer may submit a Disbursement Request for all available Loan proceeds. DELETE FOLLOWING SECTION IF NO SUBSEQUENT WIRE TRANSFERS ANTICIPATED Permitted Wire Transfers: Disbursement Requests for Subsequent Disbursements to be made by wire transfer must specify the amount an applicable Receiving Party. Each Receiving Party included in any such Disbursement Request must be listed below. Administrative Agen is authorized to use the wire instructions that have been provided directly to Administrative Agent by the Receiving Party or Borrower an attached as the Subsequent Disbursement Exhibit. All wire instructions must be in the format specified on the Subsequent Disbursement Exhibit. Names of Receiving Parties for Subsequent Disbursements (may include as many parties as needed; wire instructions for each Receiving Party must be attached as the Subsequent Disbursement Exhibit) Names of Receiving Party must be attached as the Subsequent Disbursement Exhibit) DELETE FOLLOWING SECTION IF NO SUBSEQUENT DEPOSITS INTO WFB ACCOUNTS ANTICIPATED Direct Deposit: Disbursement Requests for Subsequent Disbursements to be deposited into an account at Wells Fargo Bank, N.A. must specify the amount and applicable account. Each account included in any such Disbursement Request must be listed below. Name on Deposit Account: Wells Fargo Bank, N.A. Deposit Account Number:	2.		
etc.): DESCRIBE APPLICABLE RESTRICTIONS OR INDICATE "N/A" If there are no restrictions described here, any Subsequent Disbursement Authorizer may submit a Disbursement Request for all available Loan proceeds. DELETE FOLLOWING SECTION IF NO SUBSEQUENT WIRE TRANSFERS ANTICIPATED Permitted Wire Transfers: Disbursement Requests for Subsequent Disbursements to be made by wire transfer must specify the amount an applicable Receiving Party. Each Receiving Party included in any such Disbursement Request must be listed below. Administrative Agen is authorized to use the wire instructions that have been provided directly to Administrative Agent by the Receiving Party or Borrower at attached as the Subsequent Disbursement Exhibit. All wire instructions must be in the format specified on the Subsequent Disbursement Exhibit. Names of Receiving Parties for Subsequent Disbursements (may include as many parties as needed; wire instructions for each Receiving Party must be attached as the Subsequent Disbursement Exhibit) Names of Receiving Party must be attached as the Subsequent Disbursement Exhibit) DELETE FOLLOWING SECTION IF NO SUBSEQUENT DEPOSITS INTO WFB ACCOUNTS ANTICIPATED Direct Deposit: Disbursement Requests for Subsequent Disbursements to be deposited into an account at Wells Fargo Bank, N.A. must specify the amount and applicable account. Each account included in any such Disbursement Request must be listed below. Name on Deposit Account: Wells Fargo Bank, N.A. Deposit Account Number:	3.		
Permitted Wire Transfers: Disbursement Requests for Subsequent Disbursements to be made by wire transfer must specify the amount an applicable Receiving Party. Each Receiving Party included in any such Disbursement Request must be listed below. Administrative Agen is authorized to use the wire instructions that have been provided directly to Administrative Agent by the Receiving Party or Borrower an attached as the Subsequent Disbursement Exhibit. All wire instructions must be in the format specified on the Subsequent Disbursement Exhibit. Names of Receiving Parties for Subsequent Disbursements (may include as many parties as needed; wire instructions for each Receiving Party must be attached as the Subsequent Disbursement Exhibit) 1. 2. 3. DELETE FOLLOWING SECTION IF NO SUBSEQUENT DEPOSITS INTO WFB ACCOUNTS ANTICIPATED Direct Deposit: Disbursement Requests for Subsequent Disbursements to be deposited into an account at Wells Fargo Bank, N.A. must specify the amount and applicable account. Each account included in any such Disbursement Request must be listed below. Name on Deposit Account: Wells Fargo Bank, N.A. Deposit Account Number:	etc.): DESO If the	CRIBE APPLICABLE RESTRICTIONS OR IN ere are no restrictions described here, any Sub	DICATE "N/A"
applicable Receiving Party. Each Receiving Party included in any such Disbursement Request must be listed below. Administrative Agent is authorized to use the wire instructions that have been provided directly to Administrative Agent by the Receiving Party or Borrower and attached as the Subsequent Disbursement Exhibit. All wire instructions must be in the format specified on the Subsequent Disbursement Exhibit. Names of Receiving Parties for Subsequent Disbursements (may include as many parties as needed; wire instructions for each Receiving Party must be attached as the Subsequent Disbursement Exhibit) 1.	DELE	TE FOLLOWING SECTION IF NO SUBSEQU	JENT WIRE TRANSFERS ANTICIPATED
Receiving Party must be attached as the Subsequent Disbursement Exhibit) 1.	applio is aut attach	cable Receiving Party. Each Receiving Party inc horized to use the wire instructions that have been ned as the Subsequent Disbursement Exhibit. All	luded in any such Disbursement Request must be listed below. Administrative Agent en provided directly to Administrative Agent by the Receiving Party or Borrower and
2. 3. DELETE FOLLOWING SECTION IF NO SUBSEQUENT DEPOSITS INTO WFB ACCOUNTS ANTICIPATED Direct Deposit: Disbursement Requests for Subsequent Disbursements to be deposited into an account at Wells Fargo Bank, N.A. must specify the amount and applicable account. Each account included in any such Disbursement Request must be listed below. Name on Deposit Account: Wells Fargo Bank, N.A. Deposit Account Number:			
3. DELETE FOLLOWING SECTION IF NO SUBSEQUENT DEPOSITS INTO WFB ACCOUNTS ANTICIPATED Direct Deposit: Disbursement Requests for Subsequent Disbursements to be deposited into an account at Wells Fargo Bank, N.A. must specify the amount and applicable account. Each account included in any such Disbursement Request must be listed below. Name on Deposit Account: Wells Fargo Bank, N.A. Deposit Account Number:	1.		
DELETE FOLLOWING SECTION IF NO SUBSEQUENT DEPOSITS INTO WFB ACCOUNTS ANTICIPATED Direct Deposit: Disbursement Requests for Subsequent Disbursements to be deposited into an account at Wells Fargo Bank, N.A. must specify the amount and applicable account. Each account included in any such Disbursement Request must be listed below. Name on Deposit Account: Wells Fargo Bank, N.A. Deposit Account Number:	2.		
<u>Direct Deposit:</u> Disbursement Requests for Subsequent Disbursements to be deposited into an account at Wells Fargo Bank, N.A. must specify the amount and applicable account. Each account included in any such Disbursement Request must be listed below. Name on Deposit Account: Wells Fargo Bank, N.A. Deposit Account Number:	3.		
specify the amount and applicable account. Each account included in any such Disbursement Request must be listed below. Name on Deposit Account: Wells Fargo Bank, N.A. Deposit Account Number:	DELE	TE FOLLOWING SECTION IF NO SUBSEQU	JENT DEPOSITS INTO WFB ACCOUNTS ANTICIPATED
Wells Fargo Bank, N.A. Deposit Account Number:	speci	fy the amount and applicable account. Each acco	
•		*	
Further Credit Information/Instructions:			
	Furth	er Credit Information/Instructions:	

Borrower acknowledges that all of the information in this Agreement is correct and agrees to the term	is and conditions set forth herein and ii
the Additional Terms and Conditions on the following page.	
FIRST INDUSTRIAL, L.P.	
By: FIRST INDUSTRIAL REALTY TRUST, INC.	
General Partner	

By:	
Print Name: _	
Title:	

Additional Terms and Conditions to the Disbursement Instruction Agreement

Definitions. The following capitalized terms shall have the meanings set forth below:

- "Authorized Representative" means any or all of the Closing Disbursement Authorizers, Subsequent Disbursement Authorizers and Restricted Account Disbursement Authorizers, as applicable.
- "Receiving Bank" means the financial institution where a Receiving Party maintains its account.
- "Receiving Party" means the ultimate recipient of funds pursuant to a Disbursement Request.
- "Restricted Account" means an account at Wells Fargo Bank, N.A. associated with the Loan to which Borrower's access is restricted.

Capitalized terms used in these Additional Terms and Conditions to Disbursement Instruction Agreement and not otherwise defined herein shall have the meanings given to such terms in the body of the Agreement.

Disbursement Requests. Except as expressly provided in the Credit Agreement, Administrative Agent must receive Disbursement Requests in writing. Disbursement Requests will only be accepted from the applicable Authorized Representatives designated in the Disbursement Instruction Agreement. Disbursement Requests will be processed subject to satisfactory completion of Administrative Agent's customer verification procedures. Administrative Agent is only responsible for making a good faith effort to execute each Disbursement Request and may use agents of its choice to execute Disbursement Requests. Funds disbursed pursuant to a Disbursement Request may be transmitted directly to the Receiving Bank, or indirectly to the Receiving Bank through another bank, government agency, or other third party that Administrative Agent considers to be reasonable. Administrative Agent will, in its sole discretion, determine the funds transfer system and the means by which each Disbursement will be made. Administrative Agent may delay or refuse to accept a Disbursement Request if the Disbursement would: (i) violate the terms of this Agreement; (ii) require use of a bank unacceptable to Administrative Agent or Lenders or prohibited by government authority; (iii) cause Administrative Agent or Lenders to violate any Federal Reserve or other regulatory risk control program or guideline; or (iv) otherwise cause Administrative Agent or Lenders to violate any applicable law or regulation.

Limitation of Liability. Administrative Agent and Lenders shall not be liable to Borrower or any other parties for: (i) errors, acts or failures to act of others, including other entities, banks, communications carriers or clearinghouses, through which Borrower's requested Disbursements may be made or information received or transmitted, and no such entity shall be deemed an agent of the Administrative Agent or any Lender; (ii) any loss, liability or delay caused by fires, earthquakes, wars, civil disturbances, power surges or failures, acts of government, labor disputes, failures in communications networks, legal constraints or other events beyond Administrative Agent's or any Lender's control; or (iii) any special, consequential, indirect or punitive damages, whether or not (A) any claim for these damages is based on tort or contract or (B) Administrative Agent any Lender or Borrower knew or should have known the likelihood of these damages in any situation. Neither Administrative Agent nor any Lender makes any representations or warranties other than those expressly made in this Agreement. IN NO EVENT WILL ADMINISTRATIVE AGENT OR ANY LENDER BE LIABLE FOR DAMAGES ARISING DIRECTLY OR INDIRECTLY IF A DISBURSEMENT REQUEST IS EXECUTED BY ADMINISTRATIVE AGENT IN GOOD FAITH AN IN ACCORDANCE WITH THE TERMS OF THIS AGREEMENT.

Reliance on Information Provided. Administrative Agent is authorized to rely on the information provided by Borrower or any Authorized Representative in or in accordance with this Agreement when executing a Disbursement Request until Administrative Agent has received a new Agreement signed by Borrower. Borrower agrees to be bound by any Disbursement Request: (i) authorized or transmitted by Borrower; or (ii) made in Borrower's name and accepted by Administrative Agent in good faith and in compliance with this Agreement, even if not properly authorized by Borrower. Administrative Agent may rely solely (i) on the account number of the Receiving Party, rather than the Receiving Party's name, and (ii) on the bank routing number of the Receiving Bank, rather than the Receiving Bank's name, in executing a Disbursement Request. Administrative Agent is not obligated or required in any way to take any actions to detect errors in information provided by Borrower or an Authorized Representative. If Administrative Agent takes any actions in an attempt to detect errors in the transmission or content of transfers or requests or takes any actions in an attempt to detect unauthorized Disbursement Requests, Borrower agrees that, no matter how many times Administrative Agent takes these actions, Administrative Agent will not in any situation be liable for failing to take or correctly perform these actions in the future, and such actions shall not become any part of the Disbursement procedures authorized herein, in the Loan Documents, or in any agreement between Administrative Agent and Borrower.

International Disbursements. A Disbursement Request expressed in US Dollars will be sent in US Dollars, even if the Receiving Party or Receiving Bank is located outside the United States. Administrative Agent will not execute Disbursement Requests expressed in foreign currency unless permitted by the Credit Agreement.

Errors. Borrower agrees to notify Administrative Agent of any errors in the Disbursement of any funds or of any unauthorized or improperly authorized Disbursement Requests within fourteen (14) days after Administrative Agent's confirmation to Borrower of such Disbursement.

Finality of Disbursement Requests. Disbursement Requests will be final and will not be subject to stop payment or recall; provided that Administrative Agent may, at Borrower's request, make an effort to effect a stop payment or recall but will incur no liability whatsoever for its failure or inability to do so.

CLOSING EXHIBIT WIRE INSTRUCTIONS

ADMINISTRATIVE AGENT TO ATTACH WIRE INSTRUCTIONS FROM RECEIVING PARTIES

All wire instructions must contain the following information:

Transfer/Deposit Funds to (Receiving Party Account Name)
Receiving Party Deposit Account Number
Receiving Bank Name, City and State
Receiving Bank Routing (ABA) Number
Further identifying information, if applicable (title escrow number, borrower name, loan number, etc.)

SUBSEQUENT DISBURSEMENT EXHIBIT WIRE INSTRUCTIONS

ADMINISTRATIVE AGENT TO ATTACH WIRE INSTRUCTIONS FROM RECEIVING PARTIES

All wire instructions must contain the following information:

Transfer/Deposit Funds to (Receiving Party Account Name)
Receiving Party Deposit Account Number
Receiving Bank Name, City and State
Receiving Bank Routing (ABA) Number
Further identifying information, if applicable (title escrow number, borrower name, loan number, etc.)

EXHIBIT D

FORM OF GUARANTY

GUARANTY

This Guaranty is made as of September 11, 2015, by First Industrial Realty Trust, Inc., a Maryland corporation ("<u>Guarantor</u>"), to and for the benefit of Wells Fargo Bank, National Association, a national banking association, individually ("<u>Wells</u>"), and as administrative agent for itself and the lenders listed on the signature pages of the Credit Agreement (as defined below) and their respective successors and assigns (collectively, "<u>Lender</u>").

RECITALS

- A. First Industrial, L.P., a Delaware limited partnership ("<u>Borrower</u>"), and Guarantor have requested that Lender make unsecured term loans available to Borrower in the aggregate principal amount of up to \$260,000,000 (subject to Borrower's right to increase such aggregate principal amount to \$360,000,000 in accordance with <u>Section 2.2</u> of the Credit Agreement (herein defined)) (the "<u>Facility</u>").
- B. Lender has agreed to make available the Facility to Borrower pursuant to the terms and conditions set forth in an Unsecured Term Loan Agreement bearing even date herewith between Borrower, the Lenders and Guarantor (as amended, restated, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>"). All capitalized terms used herein and not otherwise defined shall have the meanings ascribed to such terms in the Credit Agreement.
- C. Guarantor is the sole general partner of Borrower and, therefore, Guarantor will derive financial benefit from the Facility evidenced by the Credit Agreement and the other Loan Documents. The execution and delivery of this Guaranty by Guarantor is a condition precedent to the performance by Lender of its obligations under the Credit Agreement.

AGREEMENTS

NOW, THEREFORE, Guarantor, in consideration of the matters described in the foregoing Recitals, which Recitals are incorporated herein and made a part hereof, and for other good and valuable consideration, hereby agrees as follows:

- 1. Guarantor hereby absolutely, unconditionally, and irrevocably guarantees to Lender:
- (a) the full and prompt payment of the principal of and interest on the Loans when due, whether at stated maturity, upon acceleration or otherwise, and at all times thereafter, and the prompt payment of all sums which may now be or may hereafter become due and owing under the Credit Agreement and the other Loan Documents;
 - (b) the payment of all Enforcement Costs (as hereinafter defined in Paragraph 7 hereof);
- (c) the full, complete, and punctual observance, performance, and satisfaction of all of the obligations, duties, covenants, and agreements of Borrower under the Credit Agreement and the Loan

Documents; and

- (d) All amounts due, debts, liabilities, and payment obligations described in subparagraphs (a) and (b) of this Paragraph 1 are referred to herein as the "Facility Indebtedness." All obligations described in subparagraph (c) of this Paragraph 1 are referred to herein as the "Obligations."
- 2. In the event of any default by Borrower in making payment of the Facility Indebtedness, or in performance of the Obligations, as aforesaid, in each case beyond the expiration of any applicable grace period, Guarantor agrees, on demand by Lender or any holder of any Note, to pay all the Facility Indebtedness and to perform all the Obligations as are or then or thereafter become due and owing or are to be performed under the terms of the Credit Agreement and the other Loan Documents, and to pay any reasonable expenses incurred by Lender in protecting, preserving, or defending its interest in the Property or in connection with the Facility or under any of the Loan Documents, including, without limitation, all reasonable attorneys' fees and costs. Lender shall have the right, at its option, either before, during or after pursuing any other right or remedy against Borrower or Guarantor, to perform any and all of the Obligations by or through any agent. contractor or subcontractor, or any of their agents, of its selection, all as Lender in its sole discretion deems proper, and Guarantor shall indemnify and hold Lender free and harmless from and against any and all loss, damage, cost, expense, injury, or liability Lender may suffer or incur in connection with the exercise of its rights under this Guaranty or the performance of the Obligations, except to the extent the same arises as determined by a court of competent jurisdiction by a final and non-appealable judgment to have resulted from the Gross Negligence or willful misconduct of Lender.

All of the remedies set forth herein and/or provided by any of the Loan Documents or law or equity shall be equally available to Lender, and the choice by Lender of one such alternative over another shall not be subject to question or challenge by Guarantor or any other person, nor shall any such choice be asserted as a defense, set-off, or failure to mitigate damages in any action, proceeding, or counteraction by Lender to recover or seeking any other remedy under this Guaranty, nor shall such choice preclude Lender from subsequently electing to exercise a different remedy. The parties have agreed to the alternative remedies hereinabove specified in part because they recognize that the choice of remedies in the event of a failure hereunder will necessarily be and should properly be a matter of business judgment, which the passage of time and events may or may not prove to have been the best choice to maximize recovery by Lender at the lowest cost to Borrower and/or Guarantor. It is the intention of the parties that such choice by Lender be given conclusive effect regardless of such subsequent developments.

Guarantor does hereby waive (i) notice of acceptance of this Guaranty by Lender and any and all notices and demands of every kind which may be required to be given by any statute, rule or law, (ii) any defense, right of set-off or other claim which Guarantor may have against the Borrower or which Guarantor or Borrower may have against Lender or the holders of any Note (other than defenses relating to payment of the Facility Indebtedness or the correctness of any allegation by Lender that Borrower was in default in the performance of the Obligations), (iii) presentment for payment, demand for payment (other than as provided for in Paragraph 2 above), notice of nonpayment (other than as provided for in Paragraph 2 above) or dishonor, protest and notice of protest, diligence in collection and any and all formalities which otherwise might be legally required to charge Guarantor with liability, (iv) any failure by Lender to inform Guarantor of any facts Lender may now or hereafter know about Borrower, the Facility, or the transactions contemplated by the Credit Agreement, it being understood and agreed that Lender has no duty so to inform and that the Guarantor is fully responsible for being and remaining informed by the Borrower of all circumstances bearing on the existence or creation, or the risk of nonpayment of the Facility Indebtedness or the risk of nonperformance of the Obligations, and (v) any and all right to cause a marshalling of assets of the Borrower or any other action by any court or governmental body with respect thereto, or to cause Lender to proceed against any other security given to Lender in connection with the Facility Indebtedness or the Obligations. Credit may be granted or continued from time to time by Lender to Borrower without notice to or authorization from Guarantor, regardless of the financial or other condition of the Borrower at the time of any such grant or continuation. Lender shall have no obligation to disclose or discuss with Guarantor its assessment of the financial condition of Borrower. Guarantor acknowledges that no representations of any kind whatsoever have been made by Lender to Guarantor. No modification or waiver of any of the provisions of this Guaranty shall be binding upon Lender except as expressly set forth in a writing duly signed and delivered on behalf of Lender. Guarantor further agrees that any exculpatory

language contained in the Credit Agreement and the Notes shall in no event apply to this Guaranty, and will not prevent Lender from proceeding against Guarantor to enforce this Guaranty.

- 4. Guarantor further agrees that Guarantor's liability as guarantor shall in no way be impaired by any renewals or extensions which may be made from time to time, with or without the knowledge or consent of Guarantor of the time for payment of interest or principal under the Credit Agreement and the other Loan Documents or by any forbearance or delay in collecting interest or principal under the Credit Agreement and the other Loan Documents, or by any waiver by Lender under the Credit Agreement or any other Loan Documents, or by Lender's failure or election not to pursue any other remedies it may have against Borrower, or by any change or modification in the Credit Agreement or any other Loan Documents, or by the acceptance by Lender of any additional security or any increase, substitution or change therein, or by the release by Lender of any security or any withdrawal thereof or decrease therein, or by the application of payments received from any source to the payment of any obligation other than the Facility Indebtedness, even though Lender might lawfully have elected to apply such payments to any part or all of the Facility Indebtedness, it being the intent hereof that Guarantor shall remain liable as principal for payment of the Facility Indebtedness and performance of the Obligations until all indebtedness has been paid in full and the other terms, covenants and conditions of the Credit Agreement and other Loan Documents and this Guaranty have been performed, notwithstanding any act or thing which might otherwise operate as a legal or equitable discharge of a surety. For the avoidance of doubt, it is understood and agreed that the Guarantor shall not be a "surety" for purposes of the Securities Act of Illinois (740 ILCS 155 et. seq.) and the Guarantor hereby waives any rights or benefits thereunder. Guarantor further understands and agrees that Lender may at any time enter into agreements with Borrower to amend and modify the Credit Agreement or other Loan Documents, or any thereof (including, without limitation in accordance with Borrower's right to increase the aggregate principal balance of the Loan pursuant to Section 2.2 of the Credit Agreement), and may waive or release any provision or provisions of the Credit Agreement and other Loan Documents or any thereof, and, with reference to such instruments, may make and enter into any such agreement or agreements as Lender and Borrower may deem proper and desirable, without in any manner impairing this Guaranty or any of Lender's rights hereunder or any of the Guarantor's obligations hereunder.
- 5. This is an absolute, unconditional, complete, present and continuing guaranty of payment and performance and not of collection. Guarantor agrees that this Guaranty may be enforced by Lender without the necessity at any time of resorting to or exhausting any other security or collateral given in connection herewith or with the Credit Agreement or any of the other Loan Documents, or resorting to any other guaranties, and Guarantor hereby waives the right to require Lender to join Borrower in any action brought hereunder or to commence any action against or obtain any judgment against Borrower or to pursue any other remedy or enforce any other right. Guarantor further agrees that nothing contained herein or otherwise shall prevent Lender from pursuing concurrently or successively all rights and remedies available to it at law and/or in equity or under the Credit Agreement or any other Loan Documents, and the exercise of any of its rights or the completion of any of its remedies shall not constitute a discharge of any of Guarantor's obligations hereunder, it being the purpose and intent of the Guarantor that the obligations of such Guarantor hereunder shall be primary, absolute, independent and unconditional under any and all circumstances whatsoever. Neither Guarantor's obligations under this Guaranty nor any remedy for the enforcement thereof shall be impaired, modified, changed or released in any manner whatsoever by any impairment, modification, change, release or limitation of the liability of Borrower under the Credit Agreement or other Loan Documents or by reason of Borrower's bankruptcy or by reason of any creditor or bankruptcy proceeding instituted by or against Borrower. This Guaranty shall continue to be effective and be deemed to have continued in existence ore reinstated (as the case may be) if at any time payment of all or any part of any sum payable pursuant to the Credit Agreement or any other Loan Document is rescinded or otherwise required to be returned by the payee upon the insolvency, bankruptcy, or reorganization of the payor, all as though such payment to Lender had not been made, regardless of whether Lender contested the order requiring the return of such payment. The obligations of Guarantor pursuant to the preceding sentence shall survive any termination, cancellation, or release of this Guaranty.
- 6. This Guaranty shall be assignable by Lender to any assignee of all or a portion of Lender's rights under the Loan Documents.
- 7. If: (i) this Guaranty, the Credit Agreement or any other Loan Document is placed in the hands of an attorney for collection or is collected through any legal proceeding; (ii) an attorney is retained to represent Lender in any

bankruptcy, reorganization, receivership, or other proceedings affecting creditors' rights and involving a claim under this Guaranty, the Credit Agreement or any Loan Document; (iii) an attorney is retained to provide advice or other representation with respect to the Loan Documents in connection with an enforcement action or potential enforcement action; or (iv) an attorney is retained to represent Lender in any other legal proceedings whatsoever in connection with this Guaranty, the Credit Agreement, any of the Loan Documents, or any property subject thereto (other than any action or proceeding brought by any Lender or participant against the Administrative Agent (as defined in the Credit Agreement) alleging a breach by the Administrative Agent of its duties under the Loan Documents), then Guarantor shall pay to Lender upon demand all reasonable attorney's fees, costs and expenses, including, without limitation, court costs, filing fees, recording costs, expenses of foreclosure, title insurance premiums, survey costs, minutes of foreclosure, and all other costs and expenses incurred in connection therewith (all of which are referred to herein as "Enforcement Costs"), in addition to all other amounts due hereunder.

- 8. The parties hereto intend that each provision in this Guaranty comports with all applicable local, state and federal laws and judicial decisions. However, if any provision or provisions, or if any portion of any provision or provisions, in this Guaranty is found by a court of law to be in violation of any applicable local, state or federal ordinance, statute, law, administrative or judicial decision, or public policy, and if such court should declare such portion, provision or provisions of this Guaranty to be illegal, invalid, unlawful, void or unenforceable as written, then it is the intent of all parties hereto that such portion, provision or provisions shall be given force to the fullest possible extent that they are legal, valid and enforceable, that the remainder of this Guaranty shall be construed as if such illegal, invalid, unlawful, void or unenforceable portion, provision or provisions were not contained therein, and that the rights, obligations and interest of Lender or the holder of any Note under the remainder of this Guaranty shall continue in full force and effect.
- 9. Any indebtedness of Borrower to Guarantor now or hereafter existing is hereby subordinated to the Facility Indebtedness. Guarantor agrees that until the entire Facility Indebtedness has been paid in full, (i) Guarantor will not seek, accept, or retain for Guarantor's own account any payment from Borrower on account of such subordinated debt, and (ii) any such payments to Guarantor on account of such subordinated debt shall be collected and received by Guarantor in trust for Lender and shall be paid over to Lender on account of the Facility Indebtedness without impairing or releasing the obligations of Guarantor hereunder.

- 10. Guarantor waives and releases any claim (within the meaning of 11 U.S.C. § 101) which Guarantor may have against Borrower arising from a payment made by Guarantor under this Guaranty and agrees not to assert or take advantage of any subrogation rights of Guarantor or Lender or any right of Guarantor or Lender to proceed against (i) Borrower for reimbursement, or (ii) any other guarantor or any collateral security or guaranty or right of offset held by Lender for the payment of the Facility Indebtedness and performance of the Obligations, nor shall Guarantor seek or be entitled to seek any contribution or reimbursement from Borrower or any other guarantor in respect of payments made by Guarantor hereunder. It is expressly understood that the waivers and agreements of Guarantor set forth above constitute additional and cumulative benefits given to Lender for its security and as an inducement for its extension of credit to Borrower. Nothing contained in this <u>Paragraph 10</u> is intended to prohibit Guarantor from making all distributions to its constituent shareholders which are required by law from time to time in order for Guarantor to maintain its status as a real estate investment trust in compliance with all applicable provisions of the Code (as defined in the Credit Agreement).
- 11. Any amounts received by Lender from any source on account of any indebtedness may be applied by Lender toward the payment of such indebtedness, and in such order of application, as Lender may from time to time elect.
- 12. The Guarantor hereby submits to personal jurisdiction in the State of Illinois for the enforcement of this Guaranty and waives any and all personal rights to object to such jurisdiction for the purposes litigation to enforce this Guaranty. Guarantor hereby consents to the jurisdiction of either the Circuit Court of Cook County, Illinois, or the United States District Court for the Northern District of Illinois, in any action, suit, or proceeding which Lender may at any time wish to file in connection with this Guaranty or any related matter. Guarantor hereby agrees that an action, suit, or proceeding to enforce this Guaranty may be brought in any state or federal court in the State of Illinois and hereby waives any objection which Guarantor may have to the laying of the venue of any such action, suit, or proceeding in any such court; provided, however, that the provisions of this Paragraph shall not be deemed to preclude Lender from filing any such action, suit, or proceeding in any other appropriate forum.
- 13. Anything contained herein to the contrary notwithstanding, the obligations of Guarantor hereunder at any time shall be limited to an aggregate amount equal to the largest amount that would not render its obligations hereunder subject to avoidance as a fraudulent transfer or conveyance under Section 548 of the United States Bankruptcy Code, as amended, or any comparable provisions of any similar federal or state law.
- 14. All notices and other communications provided to any party hereto under this Agreement or any other Loan Document shall be in writing or by telex or by facsimile and addressed or delivered to such party at its address set forth below or at such other address as may be designated by such party in a notice to the other parties. Any notice, if mailed and properly addressed with postage prepaid, shall be deemed given when received; any notice, if transmitted by telex or facsimile, shall be deemed given when transmitted (answerback confirmed in the case of telexes). Notice may be given as follows:

To the Guarantor:

First Industrial Realty Trust, Inc.

311 South 1 N. Wacker Drive, Suite 39004200

Chicago, Illinois 60606 Attention: Mr. Scott A. Musil

Telecopy: (312) 922 9851Email: smusil@firstindustrial.com

With a copy to:

Barack Ferrazzano Kirschbaum & Nagelberg LLP 200 West Madison 39th Floor Chicago, Illinois 60606

Attention: Suzanne Bessette-Smith and Matthew A. Jackson

Telecopy: (312) 984-3150 Email: Suzanne.bessette-smith@bfkn.com; matt.jackson@bfkn.com

To the Lender:

c/o Wells Fargo Bank, National Association as agent 10 South Wacker Drive, 32nd Floor Chicago, IL 60606 Attention: Scott S. Solis

Email: scott.s.solis@wellsfargo.com

Telecopy: (312)782-0969

or at such other address as the party to be served with notice may have furnished in writing to the party seeking or desiring to serve notice as a place for the service of notice.

- 15. This Guaranty shall be binding upon the heirs, executors, legal and personal representatives, successors and assigns of Guarantor and shall inure to the benefit of Lender's successors and assigns.
- 16. This Guaranty shall be construed and enforced under the internal law of the State of Illinois.
- 17. GUARANTOR AND LENDER, BY ITS ACCEPTANCE HEREOF, EACH HEREBY WAIVE ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION OR PROCEEDING TO ENFORCE OR DEFEND ANY RIGHT UNDER THIS GUARANTY OR ANY OTHER LOAN DOCUMENT OR RELATING THERETO OR ARISING FROM THE LENDING RELATIONSHIP WHICH IS THE SUBJECT OF THIS GUARANTY AND AGREE THAT ANY SUCH ACTION OR PROCEEDING SHALL BE TRIED BEFORE A COURT AND NOT BEFORE A JURY.

IN WITNESS WHEREOF, Guarantor has delivered this Guaranty in the State of Illinois as of the date first written above.

FIRST INDUSTRIAL REALTY TRUST, INC.,	, a
Maryland corporation	

By:		
Name:		
Title:		

STATE OF ILLINOIS)
) SS.
COUNTY OF COOK)
I, the undersigned, a Notary Public, in and for said County, in the State aforesaid, DO HEREBY CERTIFY, that of First Industrial Realty Trust, Inc., personally known to me to be the same person whose name is subscribed to the foregoing instrument, appeared before me this day in person and acknowledged that he signed and delivered the said instrument as his own free and voluntary act and as the free and voluntary act of said corporation, for the uses and purposes therein set forth. GIVEN under my hand and Notarial Seal, this day of, 2015.
Notary Public

Acknowledged and Agreed by:
WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent
By: Name: Title:

EXHIBIT E

OPINION OF BORROWER'S COUNSEL

Attached

EXHIBIT F

OPINION OF GENERAL PARTNER'S COUNSEL

Attached

EXHIBIT G

[RESERVED]

EXHIBIT H

FORM OF COMPLIANCE CERTIFICATE

To: The Administrative Agent and the Lenders who are parties to the Agreement described below

This Compliance Certificate is furnished pursuant to that certain Unsecured Term Loan Agreement, dated as of September 11, 2015 (as amended, modified, renewed or extended from time to time, the "<u>Agreement</u>") among First Industrial, L.P. (the "<u>Borrower</u>"), First Industrial Realty Trust, Inc. (the "<u>General Partner</u>"), Wells Fargo Bank, National Association, individually and as Administrative Agent, and the Lenders named therein. Unless otherwise defined herein, capitalized terms used in this Compliance Certificate have the meanings ascribed thereto in the Agreement.

THE UNDERSIGNED HEREBY CERTIFIES THAT:

- 1. I am the duly elected [Chief Financial Officer] [Chief Accounting Officer] [Controller] of the [Borrower] [General Partner].
- 2. I have reviewed the terms of the Agreement and I have made, or have caused to be made under my supervision, a detailed review of the transactions and conditions of the General Partner, the Borrower and their respective Subsidiaries and Investment Affiliates during the accounting period covered by the financial statements attached (or most recently delivered to the Administrative Agent if none are attached).
- 3. The examinations described in paragraph 2 did not disclose, and I have no knowledge of, the existence of any condition or event which constitutes a Material Adverse Financial Change, Event of Default or Default during or at the end of the accounting period covered by the attached financial statements or as of the date of this Compliance Certificate, except as set forth below.
- 4. Schedule I (if attached) attached hereto sets forth financial data and computations and other information evidencing the General Partner's and the Borrower's compliance with certain covenants of the Agreement, all of which data, computations and information (or if no Schedule I is attached, the data, computations and information contained in the most recent Schedule I attached to a prior Compliance Certificate) are true, complete and correct in all material respects.
- 5. The financial statements and reports referred to in Section 8.2(i), 8.2(ii) or 8.2(viii), as the case may be, of the Agreement which are delivered concurrently with the delivery of this Compliance Certificate, if any, fairly present in all material respects the consolidated financial condition and operations of the General Partner, the Borrower and their respective Subsidiaries at such date and the consolidated results of their operations for the period then-ended, in accordance with GAAP applied consistently throughout such period and with prior periods and correctly state the amounts of Consolidated Total Indebtedness, Consolidated Secured Debt, Consolidated Senior Unsecured Debt and the Values of all Unencumbered Assets as determined pursuant to the Agreement.

Described below are the exceptions, if any, to paragraph 3 by listing, in detail, the nature of the condition or event, the period during which it has existed and the action which the Borrower has taken, is taking, or proposes to take with respect to each such condition or event:

	ions and information set forth in Schedule I hereto and the financial statements
delivered with this Compliance Certificate in support hered	of, are made and delivered this day of, 20
	FIRST INDUSTRIAL, L.P.
	By: FIRST INDUSTRIAL REALTY TRUST, INC.
	General Partner
	By:
	Print Name:
	Title:

EXHIBIT I

FORM OF ASSIGNMENT AND ASSUMPTION AGREEMENT

THIS ASSIGNMENT AND ASSUMPTION AGREEMENT dated as of, 20 (the "Agreement") by and among (the "Assignor"). (the "Assignee"), FIRST INDUSTRIAL, L.P., a Delaware
(the "Assignor"), (the "Assignee"), FIRST INDUSTRIAL, L.P., a Delaware limited partnership (the "Borrower"), and WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent (the "Administrative Agent").
WHEREAS, the Assignor is a Lender under that certain Unsecured Term Loan Agreement dated as of September 11, 2015 (as amended, restated, supplemented or otherwise modified from time to time, the "Credit Agreement"), by and among the Borrower, the financial institutions party thereto and their assignees under Section 13.3 thereof, the Administrative Agent, and the other parties thereto;
WHEREAS, the Assignor desires to assign to the Assignee all or a portion of the Assignor's Term Loan Exposure under the Credit Agreement, all on the terms and conditions set forth herein; and
WHEREAS, the [Borrower and the] Administrative Agent consent[s] to such assignment on the terms and conditions set forth herein
NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which hereby are acknowledged by the parties hereto, the parties hereto hereby agree as follows:
Section 1. <u>Assignment</u> .
(a) Subject to the terms and conditions of this Agreement and in consideration of the payment to be made by the Assignee to the Assignor pursuant to Section 2 of this Agreement, effective as of, 20 (the "Assignment Date") the Assignor hereby irrevocably sells, transfers and assigns to the Assignee, without recourse, a \$ interest (such interest being the "Assigned Exposure") in and to the Assignor's Term Loan Exposure, and all of the other rights and obligations of the Assignor under the Credit Agreement, such Assignor's Note, and the other Loan Documents representing% in respect of the aggregate amount of Total Term Loan Exposure, including without limitation, [a principal amount of outstanding Term Loans equal to \$, all voting rights of the Assignor associated with the Assigned Exposure all rights to receive interest on such amount of Loans and all fees with respect to the [Assigned Exposure] and other rights of the Assignor under the Credit Agreement and the other Loan Documents with respect to the Assigned Exposure, all as if the Assignee were an original Lender under and signatory to the Credit Agreement having a Term Loan Exposure equal to the amount of the Assigned Exposure. The Assignee, subject to the terms and conditions hereof, hereby assumes all
obligations of the Assignor with respect to the Assigned Exposure as if the Assignee were an original Lender under and signatory to the

Credit Agreement having a Term Loan Exposure equal to the Assigned Exposure, which obligations shall include, but shall not be limited to the obligation to indemnify the Administrative Agent as provided in the Credit Agreement (the foregoing obligations, together with all other similar obligations more particularly set forth in the Credit Agreement and the other Loan Documents, shall be referred to hereinafter,

collectively, as the "Assigned Obligations").

(b) The assignment by the Assignor to the Assignee hereunder is without recourse to the Assignor. The Assignee makes and
confirms to the Administrative Agent, the Assignor, and the other Lenders all of the representations, warranties and covenants of a Lender
under Article XII of the Credit Agreement. Not in limitation of the foregoing, the Assignee acknowledges and agrees that, except as set forth
in Section 4. below, the Assignor is making no representations or warranties with respect to, and the Assignee hereby releases and discharges
the Assignor for any responsibility or liability for: (i) the present or future solvency or financial condition of the Borrower, any other Loan
Party or any other Subsidiary, (ii) any representations, warranties, statements or information made or furnished by the Borrower, any other
Loan Party or any other Subsidiary in connection with the Credit Agreement or otherwise, (iii) the validity, efficacy, sufficiency, or
enforceability of the Credit Agreement, any Loan Document or any other document or instrument executed in connection therewith, or the
collectability of the Assigned Obligations, (iv) the perfection, priority or validity of any Lien with respect to any collateral at any time
securing the Obligations or the Assigned Obligations under the Notes or the Credit Agreement and (v) the performance or failure to perform
by the Borrower or any other Loan Party of any obligation under the Credit Agreement or any other Loan Document. Further, the Assignee
acknowledges that it has, independently and without reliance upon the Administrative Agent, any other Lender or counsel to the
Administrative Agent or any of their respective officers, directors, employees and agents and based on the financial statements supplied by
the Borrower and such other documents and information as it has deemed appropriate, made its own credit analysis and decision to become a
Lender under the Credit Agreement. The Assignee also acknowledges that it will, independently and without reliance upon the
Administrative Agent or any other Lender and based on such documents and information as it shall deem appropriate at the time, continue to
make its own credit decisions in taking or not taking action under the Credit Agreement or any Note or pursuant to any other obligation. The
Administrative Agent shall have no duty or responsibility whatsoever, either initially or on a continuing basis, to provide the Assignee with
any credit or other information with respect to the Borrower, any other Loan Party or any other Subsidiary or to notify the undersigned of any
Default or Event of Default except as expressly provided in the Credit Agreement. The Assignee has not relied on the Administrative Agent
as to any legal or factual matter in connection therewith or in connection with the transactions contemplated thereunder.

Section 2. <u>Payment by Assignee</u>. In consideration of the assignment made pursuant to Section 1. of this Agreement, the Assignee agrees to pay to the Assignor on the Assignment Date, an amount equal to \$______ representing the aggregate principal amount outstanding of the Loans owing to the Assignor under the Credit Agreement and the other Loan Documents being assigned hereby.

Section 3. <u>Payments by Assignor</u>. The Assignor agrees to pay to the Administrative Agent on the Assignment Date the administrative fee payable under Section 13.3 (c) of the Credit Agreement.

Section 4. Representations and Warranties of Assignor. The Assignor hereby represents and warrants to the Assignee that (a) as of the Assignment Date (i) the Assignor is a Lender under the Credit Agreement having a Term Loan Exposure under the Credit Agreement immediately prior to the Assignment Date, equal to \$ and that the Assignor is not in default of its obligations under the Credit Agreement; and (ii) the outstanding balance of Loans owing to the Assignor (without reduction by any assignments thereof which have not yet become effective) is \$; and (b) it is the legal and beneficial owner of the Assigned Exposure which is free and clear of any adverse claim created by the Assignor.
Section 5. Representations, Warranties and Agreements of Assignee. The Assignee (a) represents and warrants that it is (i) legally authorized to enter into this Agreement; (ii) an "accredited investor" (as such term is used in Regulation D of the Securities Act) and (iii) an Eligible Assignee; (b) confirms that it has received a copy of the Credit Agreement, together with copies of the most recent financial statements delivered pursuant thereto and such other documents and information (including without limitation the Loan Documents) as it has deemed appropriate to make its own credit analysis and decision to enter into this Agreement; (c) appoints and authorizes the Administrative Agent to take such action as contractual representative on its behalf and to exercise such powers under the Loan Documents as are delegated to the Administrative Agent by the terms thereof together with such powers as are reasonably incidental thereto; (d) agrees that it will become a party to and shall be bound by the Credit Agreement and the other Loan Documents to which the other Lenders are a party on the Assignment Date and will perform in accordance therewith all of the obligations which are required to be performed by it as a Lender; and (e) is either (i) not organized under the laws of a jurisdiction outside the United States of America or (ii) has delivered to the Administrative Agent (with an additional copy for the Borrower) such items required under Section 4.5 of the Credit Agreement.
Section 6. Recording and Acknowledgment by the Administrative Agent. Following the execution of this Agreement, the Assignor will deliver to the Administrative Agent (a) a duly executed copy of this Agreement for acknowledgment and recording by the Administrative Agent and (b) the Assignor's Note. Upon such acknowledgment and recording, from and after the Assignment Date, the Administrative Agent shall make all payments in respect of the interest assigned hereby (including payments of principal, interest, fees and other amounts) to the Assignee. The Assignor and Assignee shall make all appropriate adjustments in payments under the Credit Agreement for periods prior to the Assignment Date directly between themselves.
Section 7. <u>Addresses</u> . The Assignee specifies as its address for notices and its Lending Office for all Loans, the offices set forth below:
Attention: Telephone No.: Telecopy No.:

Section 8. <u>Payment Instructions</u> . All payments to be made to the Assignee under this Agreement by the Assignor, and all payments to be made to the Assignee under the Credit Agreement, shall be made as provided in the Credit Agreement in accordance with the following instructions:
Section 9. Effectiveness of Assignment. This Agreement, and the assignment and assumption contemplated herein, shall not be

Section 9. Effectiveness of Assignment. This Agreement, and the assignment and assumption contemplated herein, shall not be effective until (a) this Agreement is executed and delivered by each of the Assignor, the Assignee, the Administrative Agent and if required, the Borrower, and (b) the payment to the Assignor of the amounts owing by the Assignee pursuant to Section 2. hereof and (c) the payment to the Administrative Agent of the amounts owing by the Assignor pursuant to Section 3. hereof. Upon recording and acknowledgment of this Agreement by the Administrative Agent, from and after the Assignment Date, (i) the Assignee shall be a party to the Credit Agreement and, to the extent provided in this Agreement, have the rights and obligations of a Lender thereunder and (ii) the Assignor shall, to the extent provided in this Agreement, relinquish its rights (except as otherwise provided in Section 13.3 of the Credit Agreement) and be released from its obligations under the Credit Agreement; provided, however, that if the Assignor does not assign its entire interest under the Loan Documents, it shall remain a Lender entitled to all of the benefits and subject to all of the obligations thereunder with respect to its Commitment.

Section 10. <u>Governing Law</u>. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF ILLINOIS APPLICABLE TO CONTRACTS EXECUTED, AND TO BE FULLY PERFORMED, IN SUCH STATE.

- Section 11. <u>Counterparts</u>. This Agreement may be executed in any number of counterparts each of which, when taken together, shall constitute one and the same agreement.
 - Section 12. <u>Headings</u>. Section headings have been inserted herein for convenience only and shall not be construed to be a part hereof.
- Section 13. <u>Amendments; Waivers</u>. This Agreement may not be amended, changed, waived or modified except by a writing executed by the Assignee and the Assignor.
- Section 14. <u>Entire Agreement</u>. This Agreement embodies the entire agreement between the Assignor and the Assignee with respect to the subject matter hereof and supersedes all other prior arrangements and understandings relating to the subject matter hereof.
- Section 15. <u>Binding Effect</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns.

Section 16. <u>Definitions</u>. Terms not otherwise defined herein are used herein with the respective meanings given them in the Credit Agreement.

[Include this Section only if the Borrower's consent is required under Section 13.3 of the Credit Agreement]. Section 17. Agreements of the Borrower. The Borrower hereby agrees that the Assignee shall be a Lender under the Credit Agreement [having a Term Loan Exposure equal to the Assigned Exposure]. The Borrower agrees that the Assignee shall have all of the rights and remedies of a Lender under the Credit Agreement and the other Loan Documents as if the Assignee were an original Lender under and signatory to the Credit Agreement, including, but not limited to, the right of a Lender to receive payments of principal and interest with respect to the Assigned Obligations, if any, and to the Loans made by the Lenders after the date hereof and to receive the fees payable to the Lenders as provided in the Credit Agreement. Further, the Assignee shall be entitled to the benefit of the indemnification provisions from the Borrower in favor of the Lenders as provided in the Credit Agreement and the other Loan Documents. The Borrower further agrees, upon the execution and delivery of this Agreement, to execute in favor of the Assignee a Note in an initial amount equal to the Assigned Commitment. Further, the Borrower agrees that, upon the execution and delivery of this Agreement, the Borrower shall owe the Assigned Obligations to the Assignee as if the Assignee were the Lender originally making such Loans and entering into such other obligations.

[Signatures on Following Page]

IN WITNESS	WHEREOF, the par	ties hereto have duly exe	cuted this Assignmen	nt and Assumption A	greement as of the o	date and year first
written above.						

ASSIGNOR:
[NAME OF ASSIGNOR]
By:
Name: Title:
Payment Instructions
[Bank] [Address] ABA No.: Account No.: Account Name: Reference:
ASSIGNEE:
[NAME OF ASSIGNEE]
By: Name: Title: Payment Instructions
[Bank] [Address] ABA No.: Account No.: Account Name: Reference:

[Signatures continued on Following Page]

	Agreed and Consented to as of the date first written above.
	[Include signature of the Borrower only if required under Section 13.3(c) of the Credit Agreement]
	BORROWER:
	FIRST INDUSTRIAL, L.P.
	By: First Industrial Realty Trust, Inc., its general partner
	By:
	Name: Title:
Accepted as of the date first written abo	ove.
ADMINISTRATIVE AGENT:	
WELLS FARGO BANK, NATIONAL	ASSOCIATION, as Administrative Agent
Ву:	
Name:	
Title:	

EXHIBIT J

FORM OF NOTICE OF BORROWING

Wells Fargo Bank, National Association Minneapolis Loan Center 733 Marquette Avenue, 10th Floor Minneapolis, MN 55402 Attention: Teresa Mager

Ladies and Gentlemen:

Reference is made to that certain Unsecured Term Loan Agreement dated as of September 11, 2015 (as amended, restated, supplemented or otherwise modified from time to time, the "<u>Credit Agreement</u>"), by and among FIRST INDUSTRIAL, L.P., a Delaware limited partnership (the "<u>Borrower</u>"), the financial institutions party thereto and their assignees under Section 13.3 thereof (the "<u>Lenders</u>"), Wells Fargo Bank, National Association, as Administrative Agent (the "<u>Administrative Agent</u>"), and the other parties thereto. Capitalized terms used herein, and not otherwise defined herein, have their respective meanings given them in the Credit Agreement.

1.	Pursuant to Section 2.7 of the Credit Agreement, the Borrower hereby requests that the Lenders make Term Loans to the Borrower in an aggregate amount equal to \$	
2.	. The Borrower requests that such Loans be made available to the Borrower on, 20	
3.	The Borrower hereby requests that such Loans be of the following Type:	
	 [Check one box only] □ Adjusted Base Rate Borrowing □ Eurocurrency Borrowing, with an initial Interest Period for a duration of: 	
	[Check one box only] one month two months three months	

□ other: _____

The Borrower hereby certifies to the Administrative Agent and the Lenders that as of the date hereof, as of the date of the making of the requested Loans, and after making such Loans, (a) no Default or Event of Default exists or would exist; and (b) the representations and warranties made or deemed made by the Borrower and each other Loan Party pursuant to Section 5.2(b) of the Credit Agreement, are and shall be true and correct with the same force and effect as if made on and as of such date except to the extent that such representations and warranties expressly relate solely to an earlier date (in which case such representations and warranties shall have been true and accurate on and as of such earlier date). In addition, the Borrower certifies to the Administrative Agent and the Lenders that all conditions to the making of the requested Loans contained in Section 5.2 of the Credit Agreement will have been satisfied at the time such Loans are made.		
FI	RST INDUSTRIAL, L.P.	
Ву	7: FIRST INDUSTRIAL REALTY TRUST, INC.	
Ge	eneral Partner	
Ву	7:	
Pr	int Name:	
Ti	tle:	

The location and number of Borrower's account to which proceeds of the Requested Loan are to be disbursed:

Litigation (Borrower)

Environmental Compliance

Trade Names

Subsidiaries (Borrower)

Unencumbered Assets

Litigation (General Partner)

Subsidiaries (General Partner)



First Industrial Realty Trust, Inc. 1 North Wacker Drive Suite 4200 Chicago, IL 60606 312/344-4300

MEDIA RELEASE

First Industrial Realty Trust Closes \$750 Million Unsecured Revolving Credit Facility and \$200 Million Unsecured Term Loan

CHICAGO, July 7, 2021 – First Industrial Realty Trust, Inc. (NYSE: FR), a leading fully integrated owner, operator and developer of industrial real estate, today announced it has closed a \$750 million senior unsecured revolving credit facility. The facility amends and restates the Company's previous revolving credit facility.

The new revolving credit facility matures on July 7, 2025, with two six-month extension options subject to certain conditions. The facility provides for interest-only payments initially at an interest rate of LIBOR plus 77.5 basis points based on the Company's current consolidated leverage ratio and credit ratings, a 32.5 basis point improvement in its credit spread compared to the previous facility. The facility also provides for a facility fee of 15 basis points and includes an accordion feature that allows First Industrial to increase the aggregate revolving borrowing capacity to \$1 billion, subject to certain conditions.

Wells Fargo Securities, LLC, BofA Securities, Inc., PNC Capital Markets LLC, Regions Capital Markets and U.S. Bank National Association were the Joint Lead Arrangers, with Wells Fargo Bank, National Association as Administrative Agent and Bank of America, N.A. and PNC Bank, National Association as Co-Syndication Agents. Others lenders include JPMorgan Chase Bank, N.A., Citibank, N.A., Royal Bank of Canada and Fifth Third Bank.

First Industrial also announced the refinancing of its \$200 million unsecured term loan with a maturity date of July 7, 2026. The loan provides for interest-only payments initially at an interest rate of LIBOR plus 85 basis points based on the Company's current consolidated leverage ratio and credit ratings, a 65 basis point improvement in its credit spread from the previous term loan.

Wells Fargo Securities, LLC, PNC Capital Markets LLC, Fifth Third Bank, National Association and Regions Capital Markets served as the Joint Lead Arrangers for the \$200 million term loan, with Wells Fargo Bank, National Association as the Administrative Agent and PNC Bank, National Association as the Syndication Agent. U.S. Bank National Association also participated in the term loan.

Given the strength of the Company's key credit metrics, initial pricing for both the senior unsecured revolving credit facility and the unsecured term loan is based on the BBB+/Baa1 credit ratings level, even though the Company's current ratings are BBB/Baa2. This favorable pricing level will be maintained provided that the Company's consolidated leverage ratio, as defined in the applicable agreements, remains less than 32.5%.

"Our new credit facility and term loan provide us with lower borrowing costs and greater term that support our long-term growth," said Scott Musil, chief financial officer of First Industrial Realty Trust, Inc. "We thank our banking partners for their commitments and support."

About First Industrial Realty Trust, Inc.

First Industrial Realty Trust, Inc. (NYSE: FR) is a leading fully integrated owner, operator, and developer of industrial real estate with a track record of providing industry-leading customer service to multinational corporations and regional customers. Across major markets in the United States, our local market experts manage, lease, buy, (re)develop, and sell bulk and regional distribution centers, light industrial, and other industrial facility types. In total, we own and have under development approximately 64.4 million square feet of industrial space as of March 31, 2021. For more information, please visit us at www.firstindustrial.com.

Forward-Looking Information

This press release may contain forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, and Section 21E of the Securities Exchange Act of 1934. We intend for such forward-looking statements to be covered by the safe harbor provisions for forward-looking statements contained in the Private Securities Litigation Reform Act of 1995. Forwardlooking statements are based on certain assumptions and describe our future plans, strategies and expectations, and are generally identifiable by use of the words "believe," "expect," "plan," "intend," "anticipate," "estimate," "project," "seek," "target," "potential," "focus," "may," "will," "should" or similar words. Although we believe the expectations reflected in forward-looking statements are based upon reasonable assumptions, we can give no assurance that our expectations will be attained or that results will not materially differ. Factors which could have a materially adverse effect on our operations and future prospects include, but are not limited to: changes in national, international, regional and local economic conditions generally and real estate markets specifically; changes in legislation/regulation (including changes to laws governing the taxation of real estate investment trusts) local economic conditions generally and real estate markets specifically; changes in legislation/regulation (including changes to laws governing the taxation of real estate investment trusts) and actions of regulatory authorities; the uncertainty and economic impact of pandemics, epidemics or other public health emergencies or fear of such events, such as the recent outbreak of coronavirus disease 2019 (COVID-19); our ability to qualify and maintain our status as a real estate investment trust; the availability and attractiveness of financing (including both public and private capital) and changes in interest rates; the availability and attractiveness of terms of additional debt repurchases; our ability to retain our credit agency ratings; our ability to comply with applicable financial covenants; our competitive environment; changes in supply, demand and valuation of industrial properties and land in our current and potential market areas; our ability to identify, acquire, develop and/or manage properties on favorable terms; our ability to dispose of properties on favorable terms; our ability to manage the integration of properties we acquire; potential liability relating to environmental matters; defaults on or non-renewal of leases by our tenants; decreased rental rates or increased vacancy rates; higher-than-expected real estate construction costs and delays in development or lease-up schedules; potential natural disasters and other potentially catastrophic events such as acts of war and/or terrorism; litigation, including costs associated with prosecuting or defending claims and any adverse

outcomes; risks associated with our investments in joint ventures, including our lack of sole decision-making authority; and other risks and uncertainties described under the heading "Risk Factors" and elsewhere in our annual report on Form 10-K for the year ended December 31, 2020, as well as those risks and uncertainties discussed from time to time in our other Exchange Act reports and in our other public filings with the SEC. We caution you not to place undue reliance on forward-looking statements, which reflect our outlook only and speak only as of the date of this press release or the dates indicated in the statements. We assume no obligation to update or supplement forward-looking statements. For further information on these and other factors that could impact us and the statements contained herein, reference should be made to our filings with the SEC.

Contact: Art Harmon
Vice President, Investor Relations and Marketing
(312) 344-4320

###